

Baden Trunk Sanitary Sewer Class EA Schedule B

Environmental Study Report

Project Location: Baden, Ontario

Prepared for: Township of Wilmot 60 Snyder's Road West Baden, ON N3A 1A1

Prepared by: MTE Consultants 520 Bingemans Centre Drive Kitchener, ON N2B 3X9

Revised April 6, 2023 November 10, 2022

MTE File No.: 48730-100

Engineers, Scientists, Surveyors



Contents

| Exe | cuti | ve Si | ummary | 1 |
|-----|------|-------|---|----|
| 1.0 | lr | ntrod | uction | 1 |
| 1 | .1 | Bac | kground and Context | 1 |
| 1 | .2 | Env | ironmental Study Report | 2 |
| 1 | .3 | Purp | pose of the Project | 2 |
| | 1.3. | 1 | Background to Baden Trunk Sanitary Sewer EA | 2 |
| | 1.3. | 2 | Project Rationale | 4 |
| 1 | .4 | Prob | blem and Opportunity Statement | 4 |
| 2.0 | S | tudy | Approach | 5 |
| 2 | .1 | Clas | s Environmental Assessment (EA) Process | 5 |
| | 2.1. | 1 | Project Classification | 5 |
| | 2.1. | 2 | Scope of Environmental Assessment | 8 |
| 2 | .2 | Proj | ect Organization | 9 |
| 2 | .3 | Age | ncy Steering Committee | 9 |
| 2 | .4 | Con | nmunication Plan | 10 |
| | 2.4. | 1 | Consultation Mechanisms and Tools | 10 |
| | 2.4. | 2 | Consultation with Government Agencies | 12 |
| | 2.4. | 3 | Public Consultation | 13 |
| | 2.4. | 4 | First Nations Consultation | 14 |
| 3.0 | E | xisti | ng Conditions | 16 |
| 3 | .1 | Phy | sical Setting & Topography | 16 |
| 3 | .2 | Natu | ural Environment | 16 |
| | 3.2. | 1 | Terrestrial Ecosystems | 16 |
| | 3.2. | 2 | Wetland Ecosystems | 18 |
| | 3.2. | 3 | Aquatic Ecosystems | 18 |
| | 3.2. | 4 | Surface Water | 19 |
| | 3.2. | 5 | Groundwater Resources | 19 |
| | 3.2. | 6 | Surface Drainage | 19 |
| 3 | .3 | Soc | io-Cultural Environment | 20 |
| | 3.3. | 1 | Land Use Policy | 20 |
| | 3.3. | 2 | Archeological Resources | 21 |



| 3 | 3.3.3 | Built Heritage and Cultural Landscapes | 21 |
|-----|---|---|----|
| 3 | 3.3.4 Noise | | 22 |
| 3.4 | | Transportation/Municipal Services and Utilities | 22 |
| 3 | 6.4.1 | Transportation Network | 22 |
| 3 | 8.4.2 | Municipal Services and Utilities | 22 |
| 3.5 | | Financial | 22 |
| 3.6 | - | Technical | 22 |
| 4.0 | As | sessment of Alternatives | 23 |
| 4.1 | | Development of Alternatives | 23 |
| 4 | .1.1 | Alternative 1 – Do-Nothing (2014 Environmental Assessment) | 23 |
| 4 | .1.2 | Alternative 2 - Snyder's Road West to Foundry Street | 24 |
| 4 | .1.3 | Alternative 3 - Christian/Charlotta/Mill/Charles/Foundry | 24 |
| 4 | .1.4 | Alternative 4 - Christian/Charlotta/Queen and Baden Creek | 24 |
| 4 | .1.5 | Alternative 5 - Christian Street and Baden Creek | 25 |
| 4.2 | | Evaluation Criteria | 25 |
| 4.3 | " | 'Must 10" Score Methodology | 28 |
| 4.4 | .4 Supporting Technical Memorandums | | |
| 4.5 | | Results of Evaluation | 29 |
| 4.6 | | Selection of Preferred Alternative | 29 |
| 5.0 | Ge | eneral Description and Design Features of the Preferred Alignment | 30 |
| 5.1 | | General Trunk Sewer Description | 30 |
| 5.2 | Geotechnical and Groundwater Assessment | | |
| 5.3 | 5.3 Easement Requirements | | |
| 5.4 | .4 Road Reconstruction Considerations | | |
| 5.5 | • | Trenchless Crossings | 31 |
| 6.0 | Po | tential Environmental Effects and Proposed Mitigation | 32 |
| 6.1 | | Direct Impacts | 32 |
| 6.2 | | Indirect Impacts | 33 |
| 6.3 | | Other Potential Mitigative Measures | 33 |
| 6.4 | | Climate Change | 34 |
| 6 | 5.4.1 | Climate Change Mitigation | 34 |
| 6 | 6.4.2 | Climate Change Adaptation and Resiliency | 34 |
| 6 | 5.4.3 | Minimizing negative impacts during project implementation | 34 |



| 6.5 | Summary of Environmental Impacts | 35 |
|-----|--|----|
| 7.0 | Approvals and Implementation Recommendations | 35 |
| 8.0 | Conclusion | 36 |

Figures

| Figure 1.0 – Primary Study Area | Enclosed |
|--|----------|
| Figure 2.0 – Detailed Study Area | Enclosed |
| Figure 3.0 – Existing Sanitary Drainage Areas | Enclosed |
| Figure 4.1 – Alternative 1 | Enclosed |
| Figure 4.2 – Alternative 1 Sanitary Drainage Areas | Enclosed |
| Figure 5.1 – Alternative 2 | Enclosed |
| Figure 5.2 – Alternative 2 Sanitary Drainage Areas | Enclosed |
| Figure 6.1 – Alternative 3 | Enclosed |
| Figure 6.2 – Alternative 3 Sanitary Drainage Areas | Enclosed |
| Figure 7.1 – Alternative 4 | Enclosed |
| Figure 7.2 – Alternative 4 Sanitary Drainage Areas | Enclosed |
| Figure 8.1 – Alternative 5 | Enclosed |
| Figure 8.2 – Alternative 5 Sanitary Drainage Areas | Enclosed |

Tables

| Table 2.1 – The Five Phases of the Class EA Planning Process | 6 |
|---|----|
| Table 2.2 – Class EA Scope of Investigations/Review | 8 |
| Table 2.3 – Project Team | 9 |
| Table 2.4 – Agency Steering Committee | 9 |
| Table 2.5 – Summary of Government Agencies and Other Stakeholders Contacted | 12 |
| Table 2.6 – Summary of Notification Dates | 13 |
| Table 4.1 – Alignment Alternatives Evaluation Criteria | 25 |
| Table 4.2 – Summary of Evaluation Matrix | 29 |
| Table 6.1 – Summary of greenhouse gas emitting sources | 34 |



Appendices

- Appendix A Figures Conestoga Rovers & Associates 2014 (CRA 2014)
- Appendix B Notifications
- Appendix C Communication Meeting Minutes, Signed Agreements, emails
- Appendix D Public Information Centre Slides, Comments
- Appendix E Technical Memos Evaluation of Alternatives, Project Assessment Reports
- Appendix F Detailed Evaluation Matrix
- Appendix G Preliminary Plan and Profile Drawings



Executive Summary

The Wastewater Servicing Study for New Growth Areas in Baden and New Hamburg was prepared by Conestoga Rovers & Associates (CRA) in 2014 to assist Township staff and Council in the decision-making process to address existing wastewater servicing constraints and future wastewater servicing needs associated with new growth areas. Changes to the Provincial Policy Statement (2020), the Places to Grow Plan (2020), the amended Regional Official Plan (2014) and the Township of Wilmot Official Plan (2019) have prompted the preferred wastewater management strategy to be re-evaluated.

The Township of Wilmot (herein referred to as the 'Township') has entered into a Schedule B Municipal Class Environmental Assessment (Class EA) for the Baden Trunk Sanitary Sewer. This Environmental Study Report (ESR) documents the study process and evaluation of five sanitary trunk alignments, resulting in the selection of the preferred trunk sewer alignment and preliminary design.

The following Problem and Opportunity Statement was developed for the project:

The existing sanitary infrastructure lacks capacity and depth for the planned growth within the Settlement of Baden. The proposed sanitary trunk will accommodate current and future growth in the area. The study will determine the corridor servicing alignment, size and depth of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.

A total of five alternatives were identified, each representing a separate alignment. The alternatives were evaluated in the context of several factors/criteria that have been grouped into 5 categories including: natural environment, socio-cultural environment, transportation/municipal services and utilities, financial, and technical. Alternative 1, (the 'Do-Nothing' alternative) is representative of the previously approved alignment of the 2014 CRA report.

The five alternatives were presented to the public in the form of a virtual Public Information Centre (PIC). It consisted of a pre-recorded presentation that clearly stated the Problem and Opportunity Statement, identified the five alternative alignments being considered, listed the criteria in which the alternatives were to be evaluated, and provided the next steps in the Class Environmental Assessment process. The PIC was hosted on the Township of Wilmot website on November 24, 2021. The web-hosted PIC provided stakeholders, indigenous groups, steering committee members and the general public an opportunity to comment on the project.

The numerical evaluation ("Must 10" scoring system) approach was used for the selection of the preferred alternative. Each factor group was scored, and commentary prepared through technical memos to document the differentiating factors or impact mitigation of the proposed sanitary trunk alternatives.

The result of the evaluation revealed Alternative 5 as the preferred alternative. It was presented to the stakeholders, indigenous groups, steering committee members, and the general public in the form of second virtual PIC, which provided an opportunity to comment on the preliminary preferred alternative.



The following summarizes the principal rationale in support of Alternative 5 as the Preferred Alignment for the Baden Sanitary Trunk:

- Provides flexibility for servicing the entire Primary Study Area
- Provides timely project completion due to minimal coordination with Regional projects
- Minimal impact to road network
- Minimal impact to traffic
- Least impacts to existing businesses and residents
- Least impacts to cultural heritage
- Eliminates two (2) existing Wastewater Pumping Stations (WWPS)
- Eliminates the need for a future WWPS
- Eliminates noise and odour associated with the removal of existing WWPS
- Reduces long-term operational/maintenance cost
- Minimizes impacts to significant watercourses (Baden Creek at the Baden Creek Pond and the Huehn Award Drain)
- No impacts to locally significant wetlands
- No impacts to Species at Risk or Significant Wildlife Habitats (woodlands, Butternuts)
- Does not impact flow regime of any surface water feature
- Does not impact the aquatic habitat of Baden Creek, the Baden Creek Pond or the Huehn Award Drain
- Eliminates impacts to the coldwater fisheries tributary (Huehn Award Drain)
- Right-of-ways have been secured or committed to
- Represents the lowest cost alternative

This alternative consists of replacing the local sanitary sewer and installing a new trunk sanitary sewer along Christian Street, through the open fields adjacent to the Baden Creek Tributary 'A' and Baden Creek, and upgrading the sewer from the intersection of Gingerich Road and Foundry Street to the Baden WWPS. A complete reconstruction of Christian Street will be required.

The final design may explore further mitigation through establishing alignment parallel to Foundry Street and directly to the Region's WWPS.

1.0 Introduction

1.1 Background and Context

The Township of Wilmot (Township) is a municipality situated in southwestern Ontario, within the Waterloo Region. It is responsible for the long-term operation and maintenance of the wastewater collection system for the settlement of Baden. In 2012, Conestoga Rovers & Associates commenced a study to evaluate the sanitary infrastructure needs as suggested by the 2011 *Baden and New Hamburg Water and Wastewater Master Plan Update* and to evaluate development options for the forecasted population growth. The study divided the predicted growth sanitary catchment areas into several parcels and identified them with letter designations from A to Z (**Appendix A**). *The Wastewater Servicing Study for the Growth Areas of Baden and New Hamburg*, completed in 2014 (CRA, 2014), revealed that the existing sanitary infrastructure did not support existing peak flows and lacked capacity and depth for the planned growth within the Settlement of Baden. This study identified Alternative D (**Appendix A**) as the preferred servicing alignment and was endorsed by the Township.

Changes to the Provincial Policy Statement (2020), the Places to Grow Plan (2020), the amended Regional Official Plan (2015), and the Township of Wilmot Official Plan (2019), has prompted the Township to re-evaluate the wastewater servicing strategy to better service existing residents and accommodate the anticipated future growth areas for the Settlement of Baden.

In 2019, MTE engaged in consultation with the Township with respect to updating the wastewater servicing strategy for Baden and New Hamburg resulting in three additional studies currently underway at Township and Regional planning levels.

These studies include;

- Morningside Trunk Sewer Class Environmental Assessment, GM BluePlan
- Baden / New Hamburg Water and Wastewater System Servicing Review, Stantec
- Baden Trunk Class Environmental Assessment, MTE

MTE Consultants Inc. (MTE) has been retained by the Township to carry out a Schedule B Municipal Class EA. On June 17, 2021, the Notice of Commencement was issued (**Appendix B**). The study proposed five sanitary trunk alignments that would provide sanitary services to future growth areas and improve services to the existing residents of Baden. The project team conducted studies to establish existing conditions of the natural, socio-cultural, transportation/municipal services and utilities, financial, and technical environments. Technical memorandums were prepared to assist in the evaluation of the alternative alignments. Each factor was numerically evaluated by means of the 'Must 10' scoring system and ranked accordingly. The alternative with the highest score was selected as the preferred sanitary servicing alignment. Upon Township endorsement of the preferred alignment, a preliminary design of the preferred sanitary sewer alignment was prepared. The purpose of this Environmental Study Report (ESR) is to document all activities undertaken as outlined in the Schedule B, Municipal Class EA process. The publication of this ESR represents the conclusion of the Class EA, including public and agency consultation.

1.2 Environmental Study Report

Municipal infrastructure projects in Ontario are subject to Ontario's Environmental Assessment Act (EA Act). The Municipal Engineers Association's Municipal Class Environmental Assessment (October 2000, as amended in 2007, 2011 and 2015) (MCEA) process for planning, design, and construction of municipal infrastructure was developed in accordance with the EA Act and is described in more detail in Section 2.1 of this report. This ESR documents the planning process conducted in accordance with the MCEA and the Planning Act for the Baden Trunk Sanitary Sewer, hereafter referred to as the "Baden Trunk".

This ESR is organized as follows:

- Section 1 introduces the study, including background and the context of the project, and introduces the rationale for the project and related study objectives.
- Section 2 summarizes the Class EA process.
- Section 3 provides an overview of existing conditions within the Primary Study Area (PSA) to establish baseline conditions against which alignment alternatives have been assessed.
- Section 4 presents the development and assessment of alignment alternatives and design solutions, including the identification of the Preliminary Preferred Alternative for the Baden Trunk. Section 4 also describes the public process and input received relative to the preliminary preferred alignment alternative.
- Section 5 describes the recommended Alternative and design features in some detail.
- Section 6 describes potential environmental effects, proposed mitigation and monitoring.
- Section 7 provides a summary of approvals required prior to construction.
- Section 8 describes the procedure to amend the ESR and includes interpretation provisions.

The Appendices attached to this ESR provide more detailed information relating to the communication record, supporting technical reports, evaluation matrix and the proposed design of the Baden Trunk alignment.

1.3 Purpose of the Project

1.3.1 Background to Baden Trunk Sanitary Sewer EA

In 2003 the Region developed a Long-Term Regional Growth Management Strategy (RGMS) to plan for and guide future population and employment growth within the Region of Waterloo (the Region). The RGMS anticipated the Region will grow to approximately 700,000 by 2041. Subsequently, the Province of Ontario identified the Region as a "Future Growth Area" in the Places to Grow Act, passed in June 2005. The RGMS helps to ensure that "Smart Growth" principles are realized in the Region. As part of this, the RGMS recommended the WWTMP (1995) be updated to provide a plan for wastewater treatment services to meet the needs of customers in a safe, efficient, cost effective and fiscally responsible manner.

In 2007, Earth Tech updated the Region's Waste Water Treatment Management Plan (WWTMP). The study identified and evaluated wastewater treatment projects, technologies, and servicing strategies to meet the long-term needs of residents and business until 2041.

The updated WWTMP provided an overall plan for the upgrade and expansion of facilities to ensure adequate wastewater treatment service capacity for the Region's wastewater treatment plants until 2041. The key goal of the proposed strategy is to ensure adequate wastewater treatment servicing to protect human health and the natural environment, meeting legislative requirements and future growth, and being fiscally responsible and sustainable. The study revealed that the average daily influent flow at the New Hamburg WWTP would increase to 5600 m³/d by 2016, exceeding the rated capacity.

In 2008, the Region initiated a study to update the previous Baden/New Hamburg Wastewater Treatment Class Environmental Assessment (Class EA), and the Baden New Hamburg Water Supply Master Plan Update completed in 1997 and 1996, respectively. The goal of the Master Plan Study was to update the previous Master Plan and Class EA studies with recent works, revised population projections and policies, and changes in the environment. It identified strategies that would optimize existing systems and increase capacities of wastewater infrastructure to provide servicing into the future. The study resulted in the *Baden-New Hamburg Wastewater Master Plan Update* (BNHWWMPU) completed in 2011 (AECOM, 2011). This study revealed that the capacity of the Baden WWPS is sufficient under dry conditions however during very high wet weather flows, the station capacity was exceeded. These high flow events are resolved by contracting vacuum trucks and tankers to haul the excess wastewater directly to the New Hamburg WWTP. The BNHWWMPU recommended that Infiltration and Inflow (I/I) be reduced or additional equalization storage capacity be provided at the Baden SPS to accommodate the very high flow events.

The Baden and New Hamburg wastewater collection systems were evaluated to identify opportunities and constraints of the Region's infrastructure based on the current installed capacity and 2041 flow requirements (Low and High Scenarios). The initial analysis compared the design flow with the projected 2041 (Low and High) flows for the current service areas. Additionally, the possibility of incorporating the Foxboro Green wastewater system into the Baden and New Hamburg System was evaluated. Opportunities and constraints for the wastewater collection system are summarized below:

- Optimization of the Baden WWPS and related equalization storage requirements to handle 2041 wet weather flows need to be confirmed as part of further work. Additional pumping capacity required along with storage to address wet weather flow impacts.
- Additional pumping capacity may be required necessitating replacement of the impellers and an additional pump at the Morningside WWPS for the 2041 (High) scenario;
- New trunk sewers and pumping stations by the Township would be required to discharge wastewater to the Morningside WWPS for areas to be developed;
- Possibility for converting the Foxboro Wastewater Treatment System to a pumping station with connection to the Baden New Hamburg collection system.

In 2014, Conestoga Rovers & Associates were retained by the Township to address the BNHWWMPU recommendations. Their study evaluated sanitary infrastructure needs, formulated development options, and evaluated the options to maximize the development of new growth areas.

1.3.2 Project Rationale

The purpose of this study is to identify and evaluate possible trunk sewer alternatives that can provide sanitary services to the existing residents and future growth areas of Baden. Alternatives are to be acceptable to the Township, the Region, the Grand River Conservation Authority (GRCA), and the Ministry of the Environment, Conservation and Parks (MECP).

An important objective of the EA process is to provide ample opportunity for public participation including dialogue between parties with differing interests. This approach allows the following objectives to be achieved:

- 1. Protection of the environment through the wise management of resources (as defined in the Environmental Assessment Act). This goal will be met through an understanding of existing conditions, extensive consultation with all affected and interested parties, monitoring, and mitigation.
- 2. Minimal disruption during construction to existing residents, farms, and commuters that rely on these roadways/intersections.
- 3. Participation of a broad range of stakeholders in the study process to allow for sharing of ideas, education, testing/analysis of solutions including consideration of alternatives.
- 4. Documentation of the study process in compliance with all phases of the Municipal Class EA process and the Planning Act.

1.4 Problem and Opportunity Statement

Based on the sanitary servicing needs outlined from previous studies, the following Problem and Opportunity Statement was developed for the Baden Trunk EA.

The existing sanitary infrastructure lacks capacity and depth for the planned growth within the Settlement of Baden. The proposed sanitary trunk will accommodate current and future growth in the area. The study will determine the corridor servicing alignment, size and depth of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.

The Problem and Opportunity Statement was first presented at Public Information Centre (PIC) No.1 for public input. Once comments were received by all interested parties, it was finalized and presented at the subsequent PIC No. 2.

2.0 Study Approach

2.1 Class Environmental Assessment (EA) Process

This study has been conducted in accordance with the (MCEA). The Class EA process was developed in accordance with the EA Act for a 'class' of projects, including municipal road, water, wastewater, and transit projects. These types of projects are recurring, similar in nature, limited in scale, responsive to mitigation measures, and have a predictable range of environmental effects. The purpose of the Class EA is to provide input that supports the protection, conservation, and wise management of the environment. This includes the natural, social, cultural, built, and economic environment.

This study will be accomplished through comprehensive planning and informed decisionmaking. It allows municipalities to meet the requirements of the EA Act while following a streamlined, self-administered process.

The key principles that were taken into account throughout the planning process include:

- Consultation with affected parties early and throughout the process. Consultation occurred throughout the Planning process as detailed throughout Section 2.3 of this study,
- A reasonable range of alternatives were considered and systematically evaluated,
- Identification and consideration of the effects of each alternative on all aspects of the environment,
- The systematic evaluation of the alternatives in terms of their benefits and constraints to determine their net environmental effects,
- Provide clear and complete documentation of the planning process to ensure traceability of decision-making with respect to the project.

2.1.1 **Project Classification**

The Class EA process recognizes that potential environmental impacts may vary depending on the nature of the project and classifies projects into four schedules as follows:

Schedule A - municipal maintenance, operational and emergency activities. These projects are pre-approved and, therefore, allow the municipality to proceed without further approval under the EA Act.

Schedule A+ - the environmental effects are usually minimal. However, the public is to be advised prior to implementation. These projects are also pre-approved.

Schedule B - projects that may have minimal environmental effects on the environment. These projects are approved subject to a screening process, including consultation with directly affected public and agencies.

Schedule C - projects that may have significant environmental effects on the environment and must proceed under the full planning and documentation procedures outlined in the Municipal Class EA Document.

This study was conducted as a Schedule 'B' Class EA in accordance with the project schedule criteria in Appendix 1 of the Municipal Class EA. This classification recognizes that the Baden Sanitary Trunk is designated as a Municipal Wastewater Project and is subject to the screening

process. Schedule 'B' projects involve the completion of Phases 1 and 2 of the Municipal Class EA process as identified and summarized below:

| Phase 1 Problem or Opportunity | Identify the problem or opportunity. |
|--|---|
| Phase 2 Alternative Solutions | Identify alternative solutions to address the problem or opportunity by taking into consideration the existing environment, select the project Schedule, and establish the preferred solution taking into account public and review agency input. |
| Phase 3 Alternative Design Concepts for Preferred Alternative | Examine the alternative methods of implementing the preferred solution, based upon the existing environment, public and review agency input, anticipated environmental effects, and methods of minimizing negative effects and maximizing positive effects. |
| Phase 4 Environmental Study Report | Document, in an Environmental Study Report, a summary of the rationale, and the planning, design and consultation process of the project as established through the above Phases, and make such documentation available for scrutiny by review agencies and the public. |
| Phase 5 Implementation | Complete contract drawings and documents and proceed to construction and operation. Monitor construction for adherence to environmental provisions and commitments. Where special conditions dictate, also monitor the operation of the completed facilities. |

Successful completion of Phase 2 and endorsement of the preferred alternative by Township Council will result in an ESR that documents the Class EA process and includes the preliminary design of the preferred alternative. The Class EA process is illustrated in the following flowchart.



2.1.2 Scope of Environmental Assessment

The ESR includes a comprehensive set of studies prepared by a multi-disciplinary project team aimed at establishing baseline existing conditions within the PSA. These studies included inventories, impact assessments, and mitigation recommendations. The evaluation of alternatives was based on the studies, consultation with review agency staff, and stakeholders.

Table 2.2 identifies the categories and the factors that were assessed.

| Natural Environment | Hydrology/Hydrogeology/Geotechnical |
|--|--|
| Wildlife surveys Vegetation inventories Aquatic habitat and fish community assessment Establish appropriate reference habitats/communities for preparation of mitigations | Surface water quality/quantity assessment Erosion assessment Groundwater resources assessment Geotechnical assessment |
| Socio-Cultural Environment | Heritage Resources |
| Land use designations & policiesNoise impacts | Archaeological Assessment Heritage Resource Assessment Recommendations for further archaeological investigation |
| Transportation | Municipal |
| Traffic operations assessment Transportation policy assessment | Infrastructure survey & assessment Utility survey & assessment |
| Technical | Financial |
| Constructability Complexity Serviceability Land requirement assessment | Capital CostOperational Cost |

2.2 Project Organization

This EA was completed in cooperation with a multidisciplinary team. The Project Team consisted of:

Table 2.3 – Project Team

| Project Team | | | |
|--|---|--|--|
| Detritus Consulting | Archeology – Garth Grimes | | |
| MHBC Planning (MHBC) | Process and Planning - Paul Britton, Pierre Chauvin | | |
| MTE Consultants Inc. (MTE-C) | Civil Engineering - Jeff Martens, Valentina Lazic, Alex Cressman, Carla Illman | | |
| MTE Consultants Inc. (MTE-H) | Geotechnical and Hydrogeology - Dan Gonser, Fraser Cummings | | |
| Natural Resource Solutions Inc. (NRSI) | Ecology – Ryan Archer | | |
| Water's Edge Environmental Solutions Team (WEEST) | Erosion Assessment & Fluvial Geomorphology Nik Gazendam | | |

2.3 Agency Steering Committee

The Steering Committee was comprised of:

 Table 2.4 – Agency Steering Committee

| Steering Committee | | |
|--|---|--|
| Township of Wilmot | | |
| Engineering, Planning and Support Staff | Jeff Molenhuis – Director of Public Works and Engineering Harold O'Krafka - Director of Development Services Mark Jeffery - Engineering Steve Scott - GIS Analyst, Infrastructure Services Cameron Miller - Planner | |
| Grand River Conservation Authority | | |
| Planning | Trevor Heywood – Resource Planner | |
| Regional Municipality of Waterloo | | |
| Engineering | Kaoru Yajima | |

2.4 Communication Plan

Public, regulatory agency, and other stakeholder consultation was recognized as an important component of the Baden Trunk Class EA study. The communications plan was designed to capture a full range of opinions and perspectives at milestone junctures throughout the study. It provided multiple and ongoing opportunities for feedback during the planning and design process meeting the requirements for mandatory contact prescribed by the Class EA process.

2.4.1 Consultation Mechanisms and Tools

The Communication Plan was implemented by employing the following mechanisms:

Notice of Commencement

Potential study participants, affected governing agencies, and stakeholders were formally notified of the study commencement via email on June 17, 2021. The Notice of Commencement also appeared in a published advertisement for the Wilmot Post and New Hamburg Independent on two separate dates; June 24, 2021 and June 30, 2021, as well as being published on the Township's website at

(https://www.wilmot.ca/en/doing-business/resources/Documents/Current_Projects/Baden-Sanitary-Trunk/48730-100-Notice-of-Commencement_June-17-2021.pdf)

for the general public (**Appendix B**). This is a mandatory contact point in the Municipal Class EA process. The notification included information on the study background, purpose, location, and opportunities/contacts for obtaining information and providing input.

To facilitate public access to the Project Team, project notifications identified the Township's representative (Jeff Molenhuis), and Project Manager (Dot Roga, MTE) as points of contact which included their mail, phone, fax, and email information.

Project Team

The Project Team had meetings with specialists, independent of the regular Project Team meetings, as required to convey information and discuss various technical matters in additional detail.

Public Information Centres (PIC)

Two PICs were held at mandatory points of contact, as indicated in the MCEA. The PICs were formally announced through advertisements in the Wilmot Post and New Hamburg Independent, postage mail, and emailed to all study participants, affected governing agencies, and stakeholders. The notification provided information on the PIC purpose, date, time, location and Project Team contacts; for obtaining information and providing input. In addition, letter invitations were distributed to property owners within the area bounded by Nafziger Road, Snyder's Road West, Foundry Street, Gingerich Road and to prescribed agencies.

As requested by the Township of Wilmot, PIC presentations were pre-recorded and uploaded to their website located at the following URL address:

(https://www.wilmot.ca/en/doing-business/current-projects.aspx#Baden-Trunk-Sanitary-Sewer)

for public viewing and comments. The website provided continuous on-line public access to the PIC, as well as the opportunity for comments and input from the public, stakeholders, and indigenous groups throughout the Class EA process.

On-line Consultation

The Township's website provided on-line public access to information and opportunities for input throughout the Class EA study at the following URL address:

(https://www.wilmot.ca/en/doing-business/current-projects.aspx#Baden-Trunk-Sanitary-Sewer)

Documents provided on the website mirrored the information and documentation provided through milestone notifications and the PICs.

Property Owner Meetings

The alignment alternatives, plan and profile drawings, and detailed information were reviewed with the landowners directly affected and/or their representatives.

Agency Consultation

Federal, Provincial, and Municipal agencies with policy or regulatory mandates that may affect the project development or implementation were identified and contacted with respect to obtaining technical information and their desire to participate in the studies. Refer to Section 2.4 of this report for additional detail.

Indigenous Communities Consultation

Selected study information was provided to First Nations with an invitation to participate throughout the Class EA process.

Notice of Completion

The final Class EA mandatory contact point is the Notice of Completion, which will also be published in the Wilmot Post and New Hamburg Independent on two separate dates. This notice will advise that the Class EA study documentation (this ESR) has been finalized and placed in the public record for a review period of 30 calendar days. The notice will also provide information on the location(s) where the ESR is available for review, the period within which the document will be available for review and comments must be submitted, the Project Team contacts for obtaining information and providing input, and the MECP contact for submitting objections to the project.

2.4.2 Consultation with Government Agencies

Full communication and participation by the review agencies was encouraged from the onset of the study. **Appendix C** includes a detailed listing of principal agency staff and other stakeholders that were engaged during the study. Table 2.4 provides a general summary of the agencies and stakeholders contacted for input on the project.

Table 2.5 – Summary of Government Agencies and Other Stakeholders Contacted

| First Nations |
|---|
| Six Nations of the Grand River |
| Six Nations of the Grand River Elected Council |
| Haudenosaunee Development Institute (HDI) |
| Haudenosaunee Confederacy Chiefs Council |
| Mississaugas of the Credit First Nation (MCFN) |
| Mississaugas of the New Credit First Nation |
| Federal |
| Environment and Climate Change Canada (ECCC) Department of Fisheries and Oceans Canada (DFO) Indigenous and Northern Affairs Canada (changed 2017 – Indigenous Relations and Northern Affairs Canada: Indigenous Services) Transport Canada |
| Provincial |
| Ministry of Agriculture, Food and Rural Affairs (OMAFRA) Ministry of Economic Development, Job Creation and Trade Ministry of the Environment, Conservation and Parks (MECP) Ministry of Indigenous Affairs Ministry of Municipal Affairs and Housing (MMAH) Ministry of Northern Development, Mines, Natural Resources and Forestry (MNDMNRF) Ministry of Heritage, Sport, Tourism and Culture Industries (MHSTCI) Ministry of Transportation (MTO) Ministry of Infrastructure |
| Municipal |
| Grand River Conservation Authority Regional Municipality of Waterloo Township of Wilmot |
| Utilities |
| Bell Canada Enbridge Gas Hydro One Enova Power (formerly Kitchener-Wilmot Hydro) PacketWorks |

| Rogers Cable |
|---|
| Railroads |
| CN Railway |
| Developers |
| Nideva Properties Inc. |
| Activa |
| Schmidt Estates |
| Consultants |
| MHBC – Planning Consultant for Nideva and Activa |
| GSP – Planning Consultant for Schmidt Estates |
| Walter Fedy – Engineering Consultant for Schmidt Estates |
| Stantec – Engineering Consultant for the Region's BNHWWMSP |
| GM BluePlan – Engineering Consultant for the Township's Morningside Sanitary Trunk Class EA |

The aforementioned agencies and stakeholders were notified via mail or email of the following events:

- Notice of Commencement
- Public Information Centre No. 1
- Public Information Centre No. 2
- Notice of Study Completion (not yet completed)

2.4.3 Public Consultation

The following section summarizes all public consultation efforts including virtual PIC publication dates, the information provided, and any comments received from the public, stakeholders, or governing agencies.

| Notification | Circulation Date | Date of Placement in Record | |
|----------------------------|--------------------|-----------------------------|--|
| Notice of Commencement | June 17, 2021 | June 24, 2021 | |
| PIC No. 1 | November 17, 2021 | November 24, 2021 | |
| PIC No. 2 | September 19, 2022 | September 26, 2022 | |
| Notice of Study Completion | TBD | TBD | |

Table 2.6 – Summary of Notification Dates

As previously mentioned, two PICs were conducted to provide opportunities to present the information gained and to solicit comment and input from the public and reviewing agencies. Both PICs consisted of a pre-recorded presentation hosted on the Township's website. The notices of the PICs were circulated to the steering committee, interested parties, and the public via email, direct mail and door-to-door distribution on the respective dates listed in **Table 2.6**.

Public Information Centre No. 1

PIC No. 1 was uploaded to the Township of Wilmot website on November 24, 2021. The purpose of PIC No. 1 was to describe and confirm the intent to follow the Schedule B of the Class EA process and solicit public/agency input to the problem statement and the identification of alternative solutions.

The information presented at the PIC included (see **Appendix D** for slides):

- Welcoming message
- Introduction to the objectives of PIC No. 1
- Problem and Opportunity statement
- Outline of the Class EA process
- Identification of the study area and the lands to be serviced
- Existing sanitary servicing conditions
- Introduction of the five sanitary sewer trunk alignment alternatives
- Next steps in the study process
- Proposed evaluation criteria
- The preferred evaluation method
- Project Team members contact information and information on how to submit comments/concerns

The PIC No.1 did not generate any comments from the public. Agency comments were submitted directly and can be found in **Appendix D**.

Public Information Centre No. 2

PIC No.2 was uploaded to the Township's website on September 26, 2021. The purpose of PIC No.2 was to provide a summary of the project to date, identify at what stage in the Class EA process the project was, and solicit public/agency input regarding the preliminary preferred alignment.

The information presented at the PIC included (see **Appendix D** for slides):

- Welcoming message
- Introduction to the objectives of PIC No. 2
- The final Problem and Opportunity statement
- Summary of existing environments
- Trunk sewer alignments with associated impacts
- Evaluation summary of the alignments
- The preliminary preferred alignment
- EA project timeline and next steps
- Project Team members contact information and information on how to submit comments/concerns

The PIC No. 2 did not generate any comments from the public. Agency comments were submitted directly and can be found in **Appendix D**.

To remain transparent, the PICs have remained online via the Township's website for the duration of the project.

2.4.4 First Nations Consultation

The Ontario Ministry of Indigenous Affairs and the Federal Department of Indigenous Relations and Northern Affairs Canada were contacted with respect to information on established or potential Indigenous and treaty rights in the vicinity of the project, and advice on consultation with First Nations communities and groups representing First Nations.

MTE Consultants | 48730-100 | Baden Sanitary Trunk EA | April 6, 2023

The Notice of Commencement was emailed directly to Six Nations Land & Resources, Haudenosaunee Development Institute and the Mississaugas of the Credit First Nation (MCFN) on June 17, 2021.

Copies of reports and background information were prepared and forwarded to Megan DeVries, Archaeological Operation Supervisor for the Department of Consultation and Accommodation (DOCA) and the MCFN. The following agreements were signed and can be found in **Appendix C**:

- Six Nations of the Grand River Elected Council (SNGREC) Archaeological Monitoring Agreement was signed on July 23, 2021.
- Archaeological Review Agreement between The Mississaugas of the Credit First Nation (MCFN) and Township was signed on July 23, 2021.
- Field Liaison Representative Participation Agreement between The Mississaugas of the Credit First Nation and Township was signed on July 23, 2021

Notices of Commencement, PIC notices, and Notices of Completion were directly communicated with all First Nation communities and groups representing First Nations.

3.0 Existing Conditions

The project team carried out detailed background studies to establish the baseline existing conditions. These studies were utilized for assessing the impacts and benefits of the Baden Trunk alternative alignments. The following sections describe the existing environments within the Detailed Study Area (DSA) and the area impacted.

3.1 Physical Setting & Topography

The study area is located within the physiographic region of southern Ontario known as the Waterloo Hills. This region occupies approximately 777 km², centred primarily in the Region. The PSA (**Figure 1.0**) is located within the Grand River watershed and is predominantly comprised of actively cultivated agricultural lands and regenerating fallow fields. Smaller areas of rural residential and commercial land cover also fall within the study area. The dominant natural feature within the study area is Baden Creek and its tributaries.

Baden Creek drains from a dammed pond within the northeast end of the DSA (**Figure 2.0**), and meanders through the DSA from north to south along historically channelized alignments used to optimize agricultural land use opportunities. Baden Creek flows downstream of the study area into the Nith River, which is a tributary of the Grand River.

Three tributaries confluence with Baden Creek within the DSA: The Huehn Award Drain, which generally drains lands east of Foundry Street, and two unnamed tributaries of Baden Creek. Baden Creek Tributary A drains lands to the northwest and originates from a stormwater management pond located within a subdivision northwest of the study area. Baden Creek Tributary B drains lands to the northeast of the study area and converges with Baden Creek east of Foundry Street.

The topography of the study area is gently sloping with the high point at ~359m amsl at Snyder's Road West to ~338m at the culvert on Baden Creek at Gingerich Road.

3.2 Natural Environment

NRSI prepared the Natural Environment Assessment (NEA) report in 2022, which summarizes background information on natural heritage features within the study area. The field survey results were used to characterize the existing terrestrial, wetland, and aquatic ecosystems of the natural environment conditions. Information on existing conditions was also gathered from background and field data collected for the Schmidt Estate Environmental Impact Study (EIS) collected by NRSI in 2017. The EIS is still ongoing.

The terrestrial, wetland, and aquatic ecosystems detailing the existing natural features and their functions, including surveys and assessments of their ecological significance and sensitivity, are provided in the NEA report and is summarized herein.

3.2.1 Terrestrial Ecosystems

The Terrestrial Ecosystems being evaluated include 'Vegetation Communities and Flora' and 'Wildlife Habitat'. This group summarizes significant and sensitive natural features and habitats present within the PSA and DSA that may represent constraints to site alteration or areas requiring mitigative measures. The ecological sensitivities of these natural features vary and are described further in the NEA.

Vegetation Communities

The DSA is predominantly comprised of active agricultural fields. According to the Ecological Land Classification (ELC) system (Lee et al., 1998), there exists four distinct cultural ecosites and one Woodland ecosite within the PSA; all of which are summarized below. The delineations can be found on Map 2 and are described in Table 1 and Table 2 of the NEA report (**Appendix E**):

Cultural Ecosites:

- Mineral Cultural Meadow (CUM1)
- Mineral Cultural Woodland (CUW1)
- Mineral Cultural Thicket (CUT1)
- Meadow Marsh (MAM)

Woodland Ecosite:

• Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3) - Locally significant woodlands

Of the 141 plant species inventoried, the Butternut (Junglans cinerea) was found to be the only federally and provincially significant plant species and is listed as Endangered (Government of Canada 2021, MECP 2020). As a Species At Risk (SAR), Ontario Regulation 830/21 Section 30 states that, in regards to the construction or installation of infrastructure, no permanent structure should be constructed or installed within 25m of a Butternut tree (Refer to Map 3 of the NEA for more details) unless approved by the MECP in writing.

The only regionally significant species within the PSA was the Running Serviceberry. It was observed within the CUM1 community along the rail corridor in the northwest study area.

Wildlife Habitat

The terrestrial ecosystems within the PSA provide a range of habitats for a variety of fauna species.

Of the 74 bird species reported, only two were identified as SAR; they include the Bank Swallow and the Barn Swallow. Bank Swallows are listed as Threatened both federally and provincially (Government of Canada 2021, MECP 2020). Lands representing existing foraging habitat include; existing agricultural fields, Mineral Cultural Meadows (CUM1), and the Baden Creek Pond. Though this is considered to represent evidence of possible breeding (Birds Canada 2021); habitat areas within the DSA do not provide suitable nesting habitat and are not impacted by the temporary construction of the buried infrastructure.

Four regionally significant bird species (Martin 1996) showing evidence of 'possible' breeding were observed within specific ecosites of the PSA, as listed below:

- Mineral Cultural Meadow (CUM1)
 - Brown Thrasher
- Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3)
 - Red-breasted Nuthatch
 - Hooded Merganser
- Pond
 - Great Blue Heron within the study area.

Herpetofauna species were not observed within the PSA during the targeted snake emergence survey.

There were no federally, provincially, or regionally significant mammal species reported within a 10km radius of the PSA. However, previous NRSI surveys observed three regionally significant species, directly and indirectly within the PSA, and should be assumed to still be present. They include the American Mink and Coyote, which are also considered to be 'Scarce', and the Virginia Opossum, considered to be 'Rare' (Region of Waterloo 1985). Neither of these species may still be considered significant.

Candidate Significant Wildlife Habitat (SWH) were identified for bats and the Species of Conservation (SCC) Snapping Turtle. Although a bat habitat assessment was not completed, all wooded areas have the potential to provide suitable habitat for SAR bats and should be avoided. The SCC Snapping Turtle was not observed during field studies. However, the pond is a potential significant habitat for the SCC Snapping Turtle and a potential overwintering habitat area.

Ecological connectivity within the study area is primarily associated with the Baden Creek riparian corridor. This corridor represents the primary naturalized feature within the study area that is otherwise dominated by active agriculture. Linkages within the study area may be of minor or localized ecological significance.

3.2.2 Wetland Ecosystems

Wetlands

Four, previously unmapped, wetland features have been identified. Two wetland features were identified within the Mineral Cultural Meadow (CUM1) community of the DSA and are believed to be anthropogenic in origin, as they do not meet the criteria to be defined as a wetland but would require confirmation by the GRCA. Refer to Map 2 of the NEA for more details.

Within the greater extent of the PSA, northeast of the Snyder's Road East and Foundry Street intersection, two small meadow marsh wetland features, associated with Baden Creek Tributary B, have been identified. All wetland features are considered to be 'locally significant' (Township of Wilmot 2019) and represent Supporting Environmental Features.

3.2.3 Aquatic Ecosystems

Four prominent aquatic features exist within the PSA. These include Baden Creek, the Huehn Award Drain, and the two Baden Creek Tributaries (A and B). The study area watercourses are considered Supporting Environmental Features as described by the policies of the Township Official Plan (OP). Refer to Map 1 of the NEA for more details.

Aquatic Habitat

Baden Creek and the lower portion of the Huehn Award Drain, up to the perched culvert at Foundry Street, provide direct year-round fish habitat including suitable spawning, foraging and rearing habitats for a variety of cool and warmwater species. Baden Creek is considered a warmwater feature, but water temperatures are maintained at the lower range due to groundwater influence and the shade provided by the adjacent woodlots. The Huehn Award Drain is a coldwater feature, providing coldwater inputs within the stream and downstream of the confluence with Baden Creek, which maintains cooler water temperatures within the lower reaches of Baden Creek.

No fish were observed within Tributary B during the assessment. However, the watercourse contains suitable habitat to support fish.

The upper portion of Baden Creek Tributary A is considered ephemeral and does not support fish habitat. The lower portion of tributary A, near its confluence with Baden Creek, is considered intermittent and provides direct fish habitat seasonally when flows are elevated within the low flow channel.

3.2.4 Surface Water

Water quality measurements of dissolved oxygen, conductivity, pH, and total dissolved solids were taken at three locations within Baden Creek. All measurements fall within the recommended Canadian Environmental Quality Guidelines (CEQG) for the protection of aquatic life. Based on available background information, pH levels, and water temperature measurements, Baden Creek is considered to have a warmwater thermal regime.

3.2.5 Groundwater Resources

Mapping illustrates that neither the PSA or DSA are located within the Regional Recharge Area (RRA), a Wellhead Protection Area (WHPA), or above a Highly Vulnerable Aquifer (HVA). Small portions in the northern part of the PSA are located within a Significant Groundwater Recharge Area (SGRA) as seen in Figure 4 of *Hydrogeological Assessment of Alternatives* (HGAA) (MTE, 2021) in **Appendix E**.

Based on the borehole logs from the previous and current investigations, the water table is present in the DSA within the Upper Maryhill Till. As this represents a Regional aquitard, shallow groundwater flow is expected to be predominately vertical. In isolated locations where sand seams are present (at MW103-21 for example), there may be a stronger horizontal component to groundwater flow.

Using water levels collected from the previous and current investigations, a map of theoretical water table contours has been generated and is presented in Figure 7 of the HGAA. Note that this representation of the water table is theoretical only, as it has been compiled using water levels taken at different times over a 3-year period. Regional water table mapping available from the GRCA is presented in Figure 8 of the HGAA. Both figures suggest that some component of shallow groundwater flow is directed towards the Baden Creek.

3.2.6 Surface Drainage

Descriptions from the NEA report and the *Baden Trunk Sanitary Sewer Environmental Assessment Fluvial Geomorphological Assessment* by Water's Edge Environmental Solutions Team (2022) were used to summarize the Surface Drainage factor group along Baden Creek, its tributaries, and the Huehn Award Drain.

Baden Creek is a permanent watercourse within the DSA. The feature flows southwest from a dammed reservoir located in the northeast portion of the study area before turning east towards Foundry Street. It then flows southeast and parallel to Foundry Street, crossing Gingerich Road, and eventually exits the DSA. Baden Creek predominantly flows through actively cultivated agricultural lands, with lesser areas of fallow, regenerating fields. Small areas of rural residential land are located to the north and east, while commercial land also exists to the south. The floodplain and extent of natural vegetation is greater than 10.0m on both sides of the channel throughout the upper portion. Baden Creek appears channelized throughout its lower portion, from where the channel flows in an easterly direction to where it runs alongside Foundry Street and under Gingerich Road. Active and historical erosion was also noted, which suggests that the channel experiences high water levels and velocities at times throughout the year. The channel banks have poor to fair stability and the bank height also vary from 0.2m to over 2.5m. The high channel banks are not stable and are often found on outside meander bends with evidence of erosion.

Baden Creek Tributary A starts at a stormwater management facility just beyond the northwest corner of the PSA. The upper reach of this tributary is channelized for approximately 320m downstream of the railway tracks. It is characterized as both ephemeral and intermittent. It runs between two row crop fields and collects runoff from the fields through tile drainage. The riparian zone (approximately 10m) is exclusively composed of grasses which help protect the channel banks' loose soil. The lower reach extends from the bend in the tributary to where it confluences with Baden Creek. The channel shows significant signs of bank erosion and lateral migration. Sections along the reach are unstable and susceptible to erosion due to the shallow rooting depth of the riparian zone.

Baden Creek Tributary B enters the study area from the north and generally flows in a south/southeasterly direction before its confluence with Baden Creek approximately 120m east of Foundry Street. This watercourse primarily drains a combination of urbanized-residential and agricultural lands within the north end of Baden. The thermal regime of this watercourse has not been determined. Evidence suggests that it is a permanent watercourse feature with minimal groundwater flow inputs which is reinforced by the presence of Watercress. There is evidence of erosion, particularly upstream of the trail.

The Huehn Award Drain is a permanent watercourse that flows west through residential land and crosses Foundry Street before flowing to Baden Creek at the east side of the study area. Only a small portion (approximately 40m) of the feature located within the study area boundary. The Huehn Award Drain is considered to be a coldwater watercourse by the GRCA. Erosion and sedimentation were noted within the aquatic features on site due to their close proximity to active agricultural lands and location downstream of highly developed areas, including stormwater management facilities; which influence the natural hydrologic regimes of each feature. In addition, historical channelization along Baden Creek and Baden Creek Tributary A influence the flow regimes of the features and increase erosion and sedimentation.

The wetlands north of Snyder's Road West and Foundry Street support the year-round flow of the Baden Creek Tributary B and contributes to Baden Creek.

3.3 Socio-Cultural Environment

The following sections provide a summary of existing and planned land uses within the study area, as well as existing cultural and heritage resources.

The Class EA is consistent with the Provincial Policy Statement of 2020 and conforms with the policies of the Places to Grow Plan of 2020, the amended Regional Official Plan of 2015 and the Township of Wilmot Official Plan of 2019. As such, it supports an integrated approach to land use planning, infrastructure investments, and environmental protection to achieve the outcomes of the Plan, ensuring the efficient use of land and infrastructure. Municipal water and wastewater systems are to be planned, designed, constructed, or expanded to serve growth in a manner that supports achievement of the minimum intensification and density targets of the Growth Plan.

3.3.1 Land Use Policy

MHBC undertook a *Land Use Planning Evaluation* for the Baden Trunk Class EA to identify potential land use impacts for each alignment alternative. Refer to their report in **Appendix E** for more details. Baden is designated as a Township Urban Area in the Regional Official Plan, consisting primarily of urban residential, urban core, light industrial, and general industrial lands to the south.

Recent updates to the Township's OP have altered future growth patterns from an 'outward' development concept to an 'in-filling' development concept between Baden and New Hamburg.

The approximate west half of the Study Area is designated agricultural lands located outside the Urban Area and partially out of the Countryside Line; as defined by the approved Township OP. The Countryside Line is the long-term boundary between the existing Urban Areas and the Countryside. Future expansions of the boundaries of the Baden Urban Areas are only permitted onto lands within the Countryside Line.

The main concerns associated with temporary land use impacts include:

- Access to agricultural/livestock operations
- Disruptions to cropping and other agricultural activities
- Many residential and commercial uses along Foundry Street and Snyder's Road West
- Traffic disruption at busy intersections combined with railroad crossing
- Access to Township's Fire Hall

3.3.2 Archeological Resources

The Township is located on the traditional territory of the Neutral Anishnaabeg, Haudenosaunee, and Mississauga peoples and acknowledges the importance of the Dish with One Spoon Covenant; which is a peace agreement made between Indigenous groups before the arrival of Euro-Canadian settlers.

Detritus Consulting was retained to complete a Stage 1 and 2 Archeological Assessment of the DSA.

The Stage 1 investigation revealed that portions of the DSA exhibited moderate to high potential for the identification and recovery of archaeological resources. These portions corresponded with the agricultural field.

Previous Stage 1-2 assessments identified two findspots that extend into the DSA. These areas will require monitoring during construction of the Baden Trunk for Alternatives 4 and 5. Stage 3 assessments are recommended for alignments that intersect with the 20m protective buffer. Currently, no alignment traverses the 20m protective buffer. However, if the preferred alignment is within the 50m monitoring buffer of an existing findspot, the Township's Consultant will engage the First Nations agencies to arrange for monitors to attend the site during excavation.

Any additional finds during the excavation may result in needing further assessments.

3.3.3 Built Heritage and Cultural Landscapes

The Region has identified that Baden is a potential Cultural Heritage Landscape; Foundry Street and Snyder's Road have been identified as scenic roads. Further, the project meets the criteria of the Evaluating Potential for Built Heritage Resources and Cultural Heritage Landscapes Checklist of the Ministry of Heritage, Sport, Tourism and Culture Industries. MHBC prepared a *Cultural Heritage Evaluation Report* (CHER) (**Appendix E**) to identify cultural heritage resources within the broader and detailed study area to provide input into the evaluation of the Baden Trunk alignments.

The impacts related to Built Heritage and Cultural Landscapes are associated to construction work that is both subterranean and undertaken for a limited period of time.

Eleven identified cultural heritage resources, which are included in the Township's Heritage Register, have been reviewed. No alternatives are considered likely to result in the removal of cultural heritage resources given that they are either located within the right-of-way or are sited away from identified cultural heritage resources.

The CHER recommends that a Heritage Impact Assessment be prepared should construction of the preferred alignment be coordinated with proposed road reconstruction and the design of the sewer change the profile or character of streets which are either scenic (i.e., Foundry Street and Snyder's Road) or identified heritage attributes in the forthcoming Cultural Heritage Landscape Study by the Region.

3.3.4 Noise

There are currently two WWPS located within Baden. The Baden West WWPS, which is maintained and operated by the Township, and the Waterloo-Oxford District WWPS, which is privately owned. A noise barrier is present along Charlotta Street to mitigate noise generated from the Baden West WWPS.

3.4 Transportation/Municipal Services and Utilities

3.4.1 Transportation Network

Snyder's Road West and Foundry Street are arterial roads in Baden. The regional roads are extensively used by commuters. Foundry Street receives traffic to and from Highway 8 while Snyder's Road West connects Kitchener to Highway 8 to the west. The intersection of Snyder's Road West and Foundry Street is bisected by a CN Railway crossing.

The intersection of Gingerich Road and Foundry Street provides access to Highway 8 along Foundry Street and access to Snyder's Road East and Kitchener along Gingerich Road.

All local roads connect to the arterial roads. The Kiwanis specialized transit and the Grand River Transit line 77 bus rely on the arterial roads. The Kiwanis specialized transit services individual residences, generally utilizing the various local roads to access the residences as requested.

3.4.2 Municipal Services and Utilities

The Baden sanitary sewer system is comprised of gravity sewers, trunk sewers, forcemains, and wastewater pumping stations. Two sanitary pumping stations direct flow to gravity sewers where it is conveyed to the Baden WWPS before being pumped to the Morningside WWPS and ultimately to the New Hamburg WWTP for treatment. **Figure 3.0** identifies the areas currently serviced by the existing sanitary sewer system.

Additional service infrastructure includes local watermains, storm drains, rights-of-way, and a Regionally owned 450mm diameter watermain within the local roads of Christian Street, Charlotta Street, Mill Street, Beck Street, Foundry Street, and Gingerich Road. Utilities include hydro, gas, and bell utilities. The primary utility corridors which service Baden are located within Snyder's Road West and Foundry Street.

3.5 Financial

Existing financial obligations stem from operational and maintenance costs associated with the Baden WWPS, local sewers, easements/rights-of-way, and traffic control required for maintenance operations.

3.6 Technical

There are existing obstacles that pose technical challenges during construction. They include crossing the CN Railroad, crossing Baden Creek, disruption of existing Regional and local infrastructure, narrow rights-of-way, and land acquisition.

4.0 Assessment of Alternatives

4.1 Development of Alternatives

Five alternatives were developed to potentially service the existing residents and future growth areas within Baden and to relieve stress on the existing sanitary infrastructure. Special consideration was given to the following key components:

- Integration with the Existing Community
- Avoidance or Mitigation of Environmental Impacts
- Connectivity to Existing Sanitary Sewer System
- Feasibility and Constructability

Sufficient depth will allow more lands to be serviced directly by gravity drainage, thereby decreasing costs associated with operations and maintenance of pumping stations. Special attention was given to the natural features that support SAR habitat.

Factors considered were cost, timing, capacity, topography, net benefits, and constructability. Evaluation criteria are described within Section 4.2.

4.1.1 Alternative 1 – Do-Nothing (2014 Environmental Assessment)

In order to establish a baseline for the assessment, MTE has defined the approved Preferred Alternative D (**Appendix A**) of the 2014 CRA report as the 'Do Nothing' Alternative. It has been endorsed by the Township and documents the preferred servicing strategy at the time the CRA report was approved.

The western portion of the alignment begins at the intersection of Christian Street and Snyder's Road West, at the previously approved obvert of 353.70m⁺ (**Figure 4.1**). It continues east along Snyder's Road West to Foundry Street. The eastern portion of the alignment runs along the eastern boundary of the Activa Lands (Parcel O), north of Snyder's Road East, southwest along the railway to Foundry Street, where it connects to the western section. The sewer then runs south along Foundry Street and ultimately to the Baden WWPS. This alternative calls for the construction of a pumping station within Parcel O, the installation of approximately 1,555m of forcemain for the eastern portion of the alignment, approximately 1,800m of sanitary sewer for the western portion of the alignment ranging in depths from 3.0m to 5.0m and associated manholes and apertances.

Construction considerations for Alternative 1 include a trenchless railway crossing, a trenchless crossing of the Baden Creek Tributary B north of Snyder's Road East, crossing the Baden Creek upstream of the Baden Creek Pond, crossing the Huehn Award Drain, upgrading existing sewers, reconstructing large portions of existing road infrastructure, and integrating it with future road construction along the Regional Roads of Snyder's Road West and Foundry Street.

Alternative 1 will service the growth areas of Parcels BB, O₁, O₂, P₁, P₂, R, S, T, U, V, Y₂ through the Nideva lands at 1012 Snyder's Road West, parts of Parcel AA₁ to AA₃, and Foxboro. Refer to **Figure 4.2** for reference to the growth area Parcels and the sanitary sewer drainage areas.

4.1.2 Alternative 2 - Snyder's Road West to Foundry Street

Alternative 2 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 5.1**), continues east on Snyder's Road West to Foundry Street, and south on Foundry Street to the Baden WWPS. This alignment is primarily within the existing rights-of-way for Snyder's Road West and Foundry Street. It calls for approximately 1,800m+ of sanitary sewer varying in depth from 3.0m to 9.3m and associated manholes.

Construction considerations for Alternative 2 include a trenchless railway crossing, crossing the Baden Creek upstream of the Baden Creek Pond, crossing the Huehn Award Drain, upgrading existing sewers along Snyder's Road West and Foundry Street, reconstructing large portions of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street.

The primary difference between Alternative 1 and Alternative 2 is that the pump station and forcemain on Activa's lands will not be required. Due to increased sewer depth, the deep sewer will provide additional sanitary services to Parcel Y_{1} , Parcel Z, and eliminates the Waterloo-Oxford District WWPS (**Figure 5.2**).

4.1.3 Alternative 3 - Christian/Charlotta/Mill/Charles/Foundry

Alternative 3 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 6.1**). It lies within the existing rights-of-way along Christian Street, Charlotta Street, Mill Street, Charles Street, and Foundry Street. This alignment requires approximately 1,985m of sanitary sewer varying in depth from 3.3m to 9.3m and associated manholes.

Construction considerations for this alignment include a trenchless railway crossing, crossing the Baden Creek just upstream of the Baden Creek Pond, crossing the Huehn Award Drain, upgrading existing sewers along Christian Street, Charlotta Street, Mill Street, Charles Street, and Foundry Street, reconstructing large portions of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street. As part of the development of the 1012 Snyder's Road West Subdivision, a portion of this alternative along Christian Street will be reconstructed.

This alternative will eliminate two WWPS; the Waterloo-Oxford District WWPS (private), and the Baden West WWPS (public) owned and operated by the Township, and will additionally service Parcels Q, X, and W (**Figure 6.2**).

4.1.4 Alternative 4 - Christian/Charlotta/Queen and Baden Creek

Alternative 4 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 7.1**). It lies within the existing rights-of-way along Christian Street, Charlotta Street, Queen Street, through Parcels AA_1 , AA_2 , and BB_2 to Gingerich Road, and then south on Foundry Street to connect to the Baden WWPS. This alignment requires approximately 2,095m of sanitary sewer, varying in depth from 3.0m to 9.3m and associated manholes.

Construction considerations for this alignment include a trenchless railroad crossing, trenchless crossing of the Baden Creek Tributary A, trenchless crossing of Baden Creek just north of Gingerich Road, upgrading existing sewers along Christian Street, Charlotta Street and Queen Street, reconstructing a small portion of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street. As part of the development of the 1012 Snyder's Road West Subdivision, a portion of this alternative along Christian Street will be reconstructed.

The existing forcemain along Queen Street will be maintained should this Alignment be selected.

This alternative benefits from following the natural topography of the Schmidt Lands and will eliminate the two WWPS as identified in Alternative 3, as well as servicing all identified growth areas (**Figures 7.2**).

4.1.5 Alternative 5 - Christian Street and Baden Creek

Alternative 5 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 8.1**). It lies within the existing right-of-way along Christian Street. It runs through Parcels AA_1 , AA_2 , and BB_2 to Gingerich Road, and then south on Foundry Street to connect to the Baden WWPS. It has the potential to run along a future easement adjacent to the Schmidt Lands (parallel to Baden Creek and Tributary A), parallel to Baden Creek, east along Gingerich Road then south on Foundry Street to connect to the Baden WWPS. This alignment requires approximately 1,905m of sanitary sewer varying in depth from 3.0m to 9.3m and associated manholes.

Construction considerations for this alignment include a trenchless railroad crossing, trenchless crossing of the Baden Creek Tributary A, trenchless crossing of Baden Creek north of Gingerich Road, upgrading existing sewers along Christian Street and Charlotta Street, reconstructing a small portion of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street. As part of the development of the 1012 Snyder's Road West Subdivision, a portion of this alternative along Christian Street will be reconstructed.

This alternative benefits from following the natural topography of the Schmidt Lands and will eliminate two WWPS as identified in Alternative 3 as well as servicing all identified growth areas (**Figures 8.2**).

4.2 Evaluation Criteria

Comprehensive evaluation criteria were developed according to the requirements of the MCEA. **Table 4.1** includes the evaluation of factors which have been grouped into five factor categories. These Factor Categories include natural environment, socio-cultural environment, transportation/municipal services and utilities, financial, and technical. Each factor category is further divided into various factor groups and factors for a more thorough evaluation.

| Factor Category/Factor | Indicator/Measure |
|--|---|
| | |
| 1 NATURAL ENVIRONMENT | |
| 1.A Terrestrial Ecosystems | |
| Vegetation Communities and Flora | Impact on feature and function of vegetation communities and flora -ecological significance (type, size, composition, species of conservation concern) -sensitivity/ susceptibility (e.g., mortality, stress, composition change). |
| • Wildlife Habitat | Impact on feature and function of wildlife habitat - significance (e.g., size, presence of specialized habitat elements, species of conservation concern) - sensitivity (e.g., noise, light) Effects on wildlife movement corridors (e.g., habitat removal, |

| able 4.1 – Alignment Alternatives | Evaluation Criteria |
|-----------------------------------|----------------------------|
|-----------------------------------|----------------------------|

| | fragmentation, potential increases in animal-vehicle | | |
|---|--|--|--|
| 1 B Wetland Ecosystems | | | |
| Wetlands/watercourse | Impact on feature and function of wetlands (e.g., area, | | |
| | classification and guality, relative extent, function in broader | | |
| | wetland complex ecosystem). | | |
| 1.C Aquatic Ecosystems | | | |
| Aquatic Habitat | - Number of water bodies and watercourse crossings. | | |
| | - Sensitivity of fish/fish habitat and thermal regime (warm, | | |
| | cool, or cold water). | | |
| | - Extent (area) and function of riparian habitat removed. | | |
| | - Extent and type of fish habitat (in-stream) altered/displaced | | |
| | in watercourse, including importance to aquatic ecosystem | | |
| | (e.g., spawning, nursery areas, habitat for species of | | |
| | conservation concern or less tolerant species). | | |
| 1.D Surface Water (Quality - Im | pact to Water Quality/Thermal/Chlorides) | | |
| Water Quality | - Impact on water quality of surface water features (TSS, | | |
| | gas/oil fueling, construction operations). | | |
| Thermal Regime | - Impact in the thermal regime of surface water features. | | |
| Chlorides | - Impact on water quality of surface water features. | | |
| 1.E Groundwater Resources | | | |
| Groundwater Recharge | - Encroachment into the Regional Recharge Area. | | |
| Areas | (removal/disruption of function – area, depth). | | |
| Groundwater Quality - | - Potential for impacts to vulnerable areas (Area) as defined | | |
| Clean Water Act (WHPA, | in Source Water Protection Areas (SWPA). | | |
| SGRA, IPZ, HVA) | | | |
| Shallow Groundwater | - Potential for interference with existing flow patterns | | |
| Movement/Contributions | (baseflow) relative to proximity to significant surface water | | |
| (Quality and Quantity) | feature and significant groundwater feature discharge areas. | | |
| | - Consider short-term impact. | | |
| | - Regional shallow groundwater elevation 346m - 353m | | |
| 1.F Surface Drainage (Proximity | / - Erosion) | | |
| Quantity (flows) | - Quantity Control and Flow Regime | | |
| Flood Plain Function | - Impacts to Regulatory Floodplain and changes to floodplain | | |
| | | | |
| | (During Construction), including need for GRCA work permit | | |
| Stormwater Management | - Opportunities to ennance roadway stormwater | | |
| | management measures, including coordination with/use of | | |
| Mater Delerer | aujacent development lacilities. | | |
| vvater Balance | | | |
| Erosion Assessment | - Impact on slope stability due to stream disturbances and | | |
| | from increased TSS during construction. | | |

| 2 SOCIO-CULTURAL ENVIRONMENT | | | |
|--|--|--|--|
| 2.A Land Use Policy | | | |
| Land Use Designations and Policies | Aligned with Provincial, Regional, and Township growth and development goals, objectives, or policies (high, medium, low). Upper tier and lower tier infrastructure and corridor owner. Efficiency of resulting development patterns and timing. | | |
| Impact of the Sanitary Sewer on Adjacent Lands | - Impacts to residence and businesses | | |
| 2.B Archaeological Resources | | | |
| Known and/or Potential Archaeological Sites | Number/type/significance of direct/indirect impacts to known archaeological resources and/or areas of high archaeological potential. | | |
| 2.C Built Heritage | | | |
| Built Heritage Features | Number/type/significance of direct/indirect impacts to built heritage resources (based on presence of built heritage resources identified by the Region). | | |
| 2.D Noise | | | |
| Noise | Noise impacts due to addition/removal of wastewater pumping stations | | |
| 3 TRANSPORTATION/MUNIC | CIPAL SERVICES AND UTILITIES | | |
| 3.A Transportation Network (Re | oads and Pedestrians) | | |
| Roads and Intersections | Potential impacts to roads/intersections. | | |
| Traffic Operations | Road safety and accessibility (sight distance, turning movements). Capacity/capability to accommodate forecast traffic demand. Provides for appropriate access to existing properties. Impact on access to existing properties. | | |
| Transportations Policy Initiatives | Capability to support municipal policy initiatives (transit, active transportation, roundabouts) | | |
| Railroad Crossings | - Ease of railroad crossing, approval | | |
| 3.B Municipal Services and Uti | lities | | |
| Municipal Services | - Opportunities for connections to the broader sanitary | | |
| • (sanitary, water, storm) | system. Impacts to existing sanitary, water, and storm infrastructure during construction. Undermines existing infrastructure. | | |
| Municipal Infrastructure | Wasterwater Pump Stations Noise Wall | | |
| Utilities (Bell, Hydro, Gas) | Degree of exposure of utilities and/or utility conflicts with road design. | | |

| 4 FINANCIAL | |
|---|---|
| Capital Cost | - Total Construction costs |
| Operational Cost | - Operational costs due to traffic during construction, pumping |
| (Sanitary sewer) | station |
| 5 TECHNICAL | |
| Constructability | Physical constructability (adequate access to construction area). Potential conflicts with existing infrastructure (other than utilities). Depth, trenchless crossings Water table and groundwater dewatering (short-term impact). |
| Trenchless Cost | - The complex issues associated with the construction of the trunk sewer |
| Property Cost | - Property requirements (number, area) and ease of |
| (Land/acquisition) | acquisition |
| Serviceability (Long- term) | Percentage of Baden to be serviced by Trunk sewer. |

4.3 "Must 10" Score Methodology

The Numerical Evaluation approach was selected as the preferred scoring methodology. This methodology requires a "Must 10" assessment of the five alternatives in a matrix form. For each factor, the least preferred alternative received a score of 1 and the most preferred alternative received a score of 10. All other alternatives were scored between these values with ties strongly discouraged. The factors receiving less than a 5 were deemed less desirable and were used to identify potential issues requiring additional mitigation. Each factor group was analyzed and commented on to clarify differentiating factors and commonalities between the impacts of the proposed Baden Sanitary Trunk alternatives.

4.4 Supporting Technical Memorandums

Technical Memorandums were prepared to provide a description of each factor being scored, the score each alternative received, and the justification for the score.

The following are the prepared Technical Memorandums which can be found in Appendix E.

- Natural Environment
- Geotechnical Site Characterization
- Hydrogeological Assessment
- Fluvial Geomorphological Assessment
- Socio-Cultural Environment
- Transportation/Municipal Services & Utilities
- Financial
- Technical
- Selection of Preferred Alternative

4.5 Results of Evaluation

The alternatives were independently evaluated according to the "Must 10" scoring methodology as described in section 4.4. The factor group score was determined by taking the average score of all the factors within the group. Scores for each factor group were then averaged to get the total category score. Final scores were calculated by summing all five category scores and ranking them accordingly. The following **Table 4.2** represents a summary of the evaluation completed by the project team and reviewed by the steering committee. The evaluation matrix can be found in **Appendix F**.

| THE ADDRESS CALLS AND DRAME OF THE PARTY AND | | | | | |
|--|----------|------|------|--------|-------|
| from Sector I have been from | Planet . | | | | |
| an and the | | - | | - | - |
| 2 C 2 C | | No. | | Read . | 1.100 |
| A defining management | | - 16 | - | 144 | 84 |
| 1 SHOOD THE BUILDING OF | | 11 | 1000 | | #811 |
| · | 19.1 | 44 | | 100 | 10 |
| Contraction of the second s | 1.44 | | N | 44 | 141 |
| C HOMAN . | | | | | W., |
| The second secon | - | | ÷ | | |

Table 4.2 – Summary of Evaluation Matrix

4.6 Selection of Preferred Alternative

The preferred sanitary sewer trunk Alternative was selected by following a Schedule B framework outlined in the Municipal Engineers Class EA manual. Based on the preceding analysis and the outcome of the evaluation matrix, Alternative 5, with a total score of 47, was selected as the preferred alternative. The sanitary sewer construction of the preferred alternative has the following benefits:

- Least impacts to existing businesses and residents
- Least impacts to cultural heritage
- Provides timely project completion due to minimal Regional Road corridors
- Provides flexibility for servicing the entire PSA
- Eliminates two existing pumping stations
- Eliminates the need for a future WWPS
- Eliminates noise and odour associated with the removal of existing pumping stations
- Reduces long-term operational/maintenance cost
- Minimal impact to road network
- Minimal impact to traffic
- Minimizes impacts to significant watercourses (Baden Creek at the Baden Creek pond and the Huehn Award Drain)
- Avoids wetlands, woodlands, Butternuts
- Does not impact flow regime of any water feature
- Eliminates impacts to the coldwater fisheries tributary (Huehn Award Drain)
- Represents the lowest cost solution

MTE Consultants | 48730-100 | Baden Sanitary Trunk EA | April 6, 2023
5.0 General Description and Design Features of the Preferred Alignment

5.1 General Trunk Sewer Description

Based on the wastewater flow projections for the Baden Trunk service area (**Figure 8.2**), the size of the Baden Trunk was established to be in the range of 450mm to 750mm in diameter. It is approximately 2,000m in length and has three trenchless crossings. Preliminary plan and profile design drawings have been prepared for the preferred Baden Trunk alignment (**Appendix G**). The drawings illustrate the existing topographic elevations, proposed sewer depth, existing utilities (to be confirmed), and surface features including buildings, roadway, and property lines.

A preliminary sanitary sewer design sheet has been prepared with pipe diameters and slopes for Alternative 5. The design has been prepared per Township and Region (DGSSMS) standards. The design uses a previously established residential population density of 60 people per hectare (ppha), as well as institutional, commercial, and industrial zonings where appropriate. The sanitary sewer design sheet is provided in **Appendix G**.

The Baden Trunk will be a gravity trunk sanitary sewer constructed of pre-cast concrete pipe or PVC. It will begin at the intersection of Snyder's Road West and Christian Street at an obvert at 349.40m and continue down Christian Street for approximately 255m at a 1% slope to Charlotta Street. The alignment will cross Charlotta Street and make a 90degree turn and continue adjacent to the Baden West WWPS for 68m at a 1% slope. The Baden Trunk can efficiently cross under the CN Railway perpendicularly, without encroaching into potential SAR bat habitat on the other side of the railway. The alignment will cross under the Baden Creek Tributary A and continue downstream, parallel to Baden Creek, for approximately 1,300m, crossing Baden Creek immediately upstream of Gingerich Road. The alignment will again cross under Baden Creek and connect with the existing trunk sanitary sewer at the intersection of Foundry Street and Gingerich Road. The existing trunk sanitary sewer will require upsizing along Foundry Street, up to the entrance of the Baden WWPS, located west of Foundry Street between Gingerich Road and Highway 8. The final invert elevation will be at 335.92m.

5.2 Geotechnical and Groundwater Assessment

Further geotechnical investigations will be required down Christian Street and through the open field.

Through the final design with further geotechnical investigation, the construction of the sanitary sewer along Christian Street will be assessed, comparing an open cut with trenchless construction techniques.

The geotechnical study will also be used to estimate the amount of dewatering required during construction. This study will establish if a Permit to Take Water (PTTW) is required or if an EASR is sufficient.

5.3 Easement Requirements

A portion of Alternative 5 will require the dedication of an easement through the northwest portion of the Schmidt Lands where the Township has not yet secured land necessary for the implementation of the preferred Alternative. The width of the easement will depend on the depth of the Baden Trunk final design.

5.4 Road Reconstruction Considerations

The upstream portion of Alternative 5 has an alignment generally following the centreline of the Christian Street right-of-way. Due to the depth of the Baden Trunk, Christian Street will be completely excavated requiring a complete road reconstruction. Road reconstruction considerations include whether to maintain Christian Street as a rural road, updating it to an urban road, burying existing hydro lines and hanging transformer or leaving it alone. Attention is required during construction to prevent damage to existing underground infrastructure, including a 450mm diameter Regional watermain located along Christian Street and Gingerich Road, as well as the existing underground utilities and infrastructure at the intersection of Foundry Street and Gingerich Road. Coordination efforts are required with the Region during final design to avoid conflicts with ongoing and proposed projects.

5.5 Trenchless Crossings

Different trenchless technologies will be utilized depending on the crossing. The railway crossing will utilize a combination of pipe hammering (Jack and Bore) and Horizontal Directional Drilling (HDD). Christian Street may implement HDD should trenchless installation be more cost effective as compared to an open cut given existing infrastructure within the Christian Street right-of-way. Pipe hammering (Jack and Bore techniques) may be utilized for the stream crossing of Tributary A and for the shallow Baden Creek crossing at Gingerich Road.

6.0 Potential Environmental Effects and Proposed Mitigation

6.1 Direct Impacts

In the long-term, Alternative 5 does not directly impact core environmental features, watercourses (no watercourses will be crossed directly, and no in-water works are proposed) or designated heritage resources.

The direct impacts are those occurring during the construction of the sanitary trunk.

Environmental impacts include:

- Possible impact to a portion of a Mineral Cultural Woodland (CUW1) located south of Christian Street and the railroad tracks. Trees requiring removal may represent roosting habitat for SAR bats. These trees should be inspected for the presence of suitable bat habitat prior to removal. If suitable bat habitat is present, MECP consultation may be required to ensure that steps are taken to avoid or mitigate impacts to bat SAR or their habitat. At a minimum, potential bat habitat trees should be removed outside of the bat active period (April 1-September 30).
- Potential edge tree removal, or disturbance to tree root zones where the alignment passes through the Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3). Trees should be inspected for the presence of suitable bat habitat prior to removal. If suitable bat habitat is present, MECP consultation may be required to ensure that steps are taken to avoid or mitigate impacts to bat SAR or their habitat. At a minimum, potential bat habitat trees should be removed outside of the bat active period (April 1-September 30). The sewer alignment should be designed to avoid direct impacts to these woodland habitats and should be offset to the extent feasible from the woodland edge.
- Direct removal of habitat for wildlife occupying CUM1 meadow will be required (e.g., small mammals, snakes, open habitat birds). However, this habitat removal will be temporary, as CUM1 meadow habitats are resilient to disturbance and will readily renaturalize. The majority of species occupying this habitat type are generally disturbance tolerant and mobile therefore they will not be negatively impacted. Vegetation removals should be timed to occur outside of the period April 1 August 31 in order to avoid contravention of the *Migratory Birds Convention Act.* Removal within the bird nesting period may be possible subject to prior inspection of the vegetation for the presence of active nests by a qualified biologist.
- The crossing of the upstream ephemeral reach of Tributary A can occur with no impacts if done during dry conditions.
- Baden Creek provides direct fish habitat therefore any activity that may cause the Harmful Alteration, Disruption or Destruction (HADD) of fish or fish habitat is subject to the requirements of the federal Fisheries Act. Details of the proposed sewer installation activities will be subject to review by the federal Department of Fisheries and Oceans (DFO) through submission of a Request for Review prior to construction. The DFO will provide a response with recommended Best Management Practices to suitably avoid or mitigate impacts to fish and fish habitat or will determine that the proposed undertaking will require a Fisheries Act Authorization. Acquisition of an authorization under the Fisheries Act will require additional measures to provide an Overall Benefit to fish or fish habitat, such as through the restoration or enhancement of aquatic ecosystems.

MTE Consultants | 48730-100 | Baden Sanitary Trunk EA | April 6, 2023

 Portions of the proposed alignment will occur in close proximity to Baden Creek. Increased proximity to the watercourse may result in greater potential for constructionstage disturbances or potential future leakage of the sewer infrastructure that may affect the watercourse. To the extent possible, the sewer alignment should be maintained outside of the recommended 15.0m watercourse buffer identified in the NEA to mitigate these potential impacts. Sewer construction below the water table will be constructed with isolation collars and enhanced gaskets to ensure a water tight construction limiting inflow and infiltration (I/I).

6.2 Indirect Impacts

Tree removal within the Cultural Woodland (CUW1) or Forest (FOD7-3) may impact the wildlife corridor surrounding Baden Creek.

Open-cut trenching along Baden Creek and Tributary A may introduce additional sediments to the water courses resulting in increased erosion and decreasing water quality if mitigation measures are not followed.

If dewatering is required, it may impact groundwater inflow to Baden Creek and Tributary A resulting in lower baseflow and an increase in water temperature. Any de-watering measures should provide sediment removal and flows should be returned to the Creek with thermal mitigation.

With the removal of the existing two WWPS, the odour and noise associated with the WWPS will be eliminated and represent a community benefit.

6.3 Other Potential Mitigative Measures

Alternative 5 can be implemented with the securing of easements from the Schmidt Lands for the north portion of the alignment. The balance of the alignment is within lands already owned by the Township or within Township and Region road rights-of-way.

The lower sections of the Alternative 5 alignment cross Baden Creek twice, once upstream of Gingerich Road and the second within the Baden WWPS site. These two crossings could be eliminated entirely with an easement across the OK Tire site located on Gingerich Road (**Figure 8.1b**).

The modified alignment follows the same route as Alternative 5 however, instead of crossing Baden Creek near Gingerich Road, it will continue parallel to Baden Creek, across Gingerich Road, and connect directly to the Baden WWPS. The advantages of this alignment are the following:

- Eliminate two crossings of Baden Creek -
 - > Does not require DFO approval for stream crossing
 - > Reduces construction costs associated with trenchless crossing
- Does not interfere with Regional projects and infrastructure through the intersection of Foundry Street and Gingerich Road
- Eliminates upsizing the existing trunk sanitary sewer along Foundry Street to the Baden WWPS
- Will not impact traffic at busy intersection of Foundry Street and Gingerich Road

This adjusted alignment is dependent upon securing an easement from OK Tire owners.

6.4 Climate Change

The Considering Climate Change in the Environmental Assessment Process guide, produced by the MECP, is now included in the Environmental Assessment program's Guides and Codes of Practice. The directions and methods outlined in the guide, complement and support the climate-focused policies of the 2014 Provincial Policy Statement.

Even though Climate Change was not specifically included into the Evaluation Matrix, its impact on each alternative was considered during the evaluation stage.

The following sections describe the effects of climate change through climate change mitigation and climate change adaptation/resiliency.

6.4.1 Climate Change Mitigation

With respect to the Baden Trunk Sanitary Sewer, mitigating greenhouse gas emissions would best be achieved by reducing the existing number of pumping stations. Currently, there are three pumping stations operating in Baden. The following table summarizes the number of pumping stations required for each alternative.

| <u></u> | . g | | |
|---------------|-------------------------------------|---------------------------------------|-------------------------------------|
| Alternative | Number of pumping stations added | Number of pumping stations removed | Total number of pumping stations |
| Alternative 1 | 1 | 0 | 4 |
| Alternative 2 | 0 | 1 | 2 |
| Alternative 3 | 0 | 2 | 1 |
| Alternative 4 | 0 | 2 | 1 |
| Alternative 5 | 0 | 2 | 1 |

 Table 6.1 - Summary of greenhouse gas emitting sources

*Note: Total number of pumping stations includes the Baden WWPS

The preferred alternative, Alternative 5, reduces redundancy of infrastructure thus mitigating climate change by reducing greenhouse gas emissions.

6.4.2 Climate Change Adaptation and Resiliency

According to the *Localized Climate Projections for Waterloo Region (Dec 2015)*, if things progress in a business-as-usual fashion, the total annual precipitation is projected to increase by 12% by 2080.

One approach to adapting a sanitary trunk sewer to climate change is by sizing the pipes to accommodate the increased I/I from the anticipated increase in precipitation. Considering the preferred alternative is to be a new construction, there is an opportunity for the Baden Trunk to be sized to include the anticipated 12% increase in I/I without exceeding the 85% capacity requirement as identified in the DGSSMS.

Going forward with the project into Preliminary and Detailed Design, it will be important to specify efficient materials, enhanced gaskets, and sealed manholes to mitigate I/I thus increasing its resiliency to climate change.

6.4.3 Minimizing negative impacts during project implementation

To minimize potential effects during construction, the idling of construction equipment will be avoided, and equipment will be in good working order to reduce inefficiencies in the operation of equipment.

6.5 Summary of Environmental Impacts

The proposed alignment of Alternative 5 has considered the significant and sensitive natural features along the alignment in order to avoid, minimize, and mitigate potential impacts. Overall, Alternative 5 presented the least impactful alignment to the natural environment.

7.0 Approvals and Implementation Recommendations

Final approval of the design will be required from municipal authorities and regulatory agencies prior to commencing construction. During the detailed design stage of the project, consultation with these approval authorities will ensure that their requirements are adequately addressed on the design drawings and restrictions are included in the contract specifications.

Prior to construction, several approvals/clearances will be required to be obtained from the Township, which are described below:

Township of Wilmot

Construction of the Baden Trunk will occur along alignments within agricultural fields adjacent to Baden Creek and along local and Regional Roads. Reviews and approvals will be required by the respective municipality to ensure that municipal standards, guidelines, and By-laws are followed. Detailed Traffic Management Plans will be developed during the detailed design stage, including an outline of all lane closures and detour routes, which will be subject to the review and approval of the Township.

Grand River Conservation Authority

The preferred alignment is partially located within a GRCA regulated area. In accordance with Ontario Regulation 97/04, under Section 28 of the Conservation Authorities Act, a permit will be required from the GRCA.

Department of Fisheries and Oceans (DFO)

A permit may be required from DFO for stream crossings.

Ministry of Environment, Conservation and Parks

Tree removals may require MECP with respect to SAR bats and/or nesting birds. Tree removal should be timed to occur outside of the period April 1 – August 31 in order to avoid contravention of the *Migratory Birds Convention Act*. Removal within the bird nesting period may be possible subject to prior inspection of the vegetation for the presence of active nests by a qualified biologist.

Registration on the Environmental Activity and Sector Registry (EASR) is required for construction in the likely event that dewatering will be required. Registration replaces the need to obtain a PTTW. A hydrogeological/geotechnical investigation will determine dewatering requirements.

CN Railway

A crossing permit from CN Railway is required for crossing the railroad south of Charlotta Street. Following CN Railway's design criteria, the primary casing of the sewer pipe must be sufficient to withstand the loading applied by passing trains. Additional requirements include a geotechnical report that includes comments and recommendations with respect to construction methodology, a settlement monitoring program by a Geotechnical Engineer with experience in large diameter pipe installation and the name and phone number of the qualified site inspector who will be on the job site, full-time, for the duration of construction.

MTE Consultants | 48730-100 | Baden Sanitary Trunk EA | April 6, 2023

Archeology

The preferred alignment falls within a specified 50m monitoring buffer zone for two archeological findspots. Special permission, along with an appointed representative, is required to oversee construction within the 50m monitoring buffer zone.

8.0 Conclusion

The following conclusions and recommendations provide a summary of the key findings presented in this report:

- Based on the projected population growth for the Settlement of Baden, a new trunk sanitary sewer is required to convey sanitary sewage flows from the proposed growth areas. The estimated size of the Baden Trunk ranges from 450mm to 750mm diameter (size to be confirmed during detailed design).
- The preferred alignment for the Baden Trunk is Alternative 5, as described in this report. The selection of the preferred Alternative was based on the following criteria:
 - Impacts to the Natural Environment
 - Social and cultural impacts
 - Impacts to transportation/municipal services and utilities
 - Financial impacts and
 - Technical constraints
- Alternative 5 is the least impactful to the natural, socio-cultural, and technical environments. However, the potential impacts to the natural environment trigger a need for an Environmental Mitigation Plan to develop mitigative measures for those impacts.
- The use of trenchless technology will be utilized for the railway crossing and the two stream crossings. Furthermore, certain sections of the alignment that are excessive in depth and have a narrow right-of-way may also utilize trenchless technology.
- Permanent and working easements will have to be acquired from private property owners for the construction of the Baden Trunk. A permanent easement, approximately 20m wide, through the agricultural lands, has been secured.
- Geotechnical Investigations are recommended during the detailed design stage.
- Securing an easement from OK Tire owners is recommended to mitigate the number of crossings of Baden Creek and the potential impacts associated with the construction along Gingerich Road and Foundry Street.

The conclusion of this study is that the Township of Wilmot Department of Public Works and Engineering recommends Alternative 5 as the preferred Baden Trunk Sanitary Sewer alignment alternative.

The conclusion is based on the detailed evaluation completed through the Class EA process.

All (Exciting the second educed in the second educed educed in the second educed educe

Del Rega, G.E.T. Pojast Managlar Stisla (IS-coult apt. 1280stisla (IS-coult apt. 1280-





Velen, no Lozic, P.Eng. Design Engineer S19-740-Stoff est 1433 Olarister 1455-com

A REAL PROPERTY AND A REAL PROPERTY AND

Figures



| LE | GEND |
|-------|--|
| | - PROPERTY LINE |
| | EXISTING SANITARY TRUNK |
| | EXISTING SANITARY FORCEMAIN |
| 326 | EXISTING 2015 GRCA CONTOURS |
| | POTENTIAL SERVICEABLE LANDS TO THE BADEN TRUNK |
| | DRAINAGE BY GRAVITY TO NAFZIGER ROAD TRUNK SEWER |
| | DRAINAGE TO FOREST GLEN WWPS |
| | DRAINAGE TO WILMOT EMPLOYMENT LANDS TRUNK SEWER |
| | LANDS NOT TO BE DEVELOPED |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | HUEHN AWARD DRAIN 30m BUFFER |
| | GRCA EXISTING WATERCOURSE |
| ***** | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | · DETAILED STUDY AREA |
| 0 | PROPOSED WASTEWATER PUMPING STATION |
| 0 | EXISTING WASTEWATER PUMPING STATION |
| ALTER | RNATIVES |
| | 2014 ENVIRONMENTAL ASSESSMENT PREFERRED ALTERNATIVE (DO NOTHING) |
| 2 | SNYDER'S/FOUNDRY |
| 3 | CHRISTIAN/CHARLOTTA/MILL/ CHARLES/FOUNDRY |
| 4 | CHRISTIAN/CHARLOTTA/QUEEN AND BADEN CREEK |
| 5 | CHRISTIAN AND BADEN CREEK |
| | |















P:∖P\48730\100\48730−100−F3.1.dwg





| | EGEND |
|--------|--|
| | |
| | EXISTING SANITARY TRUNK |
| | EXISTING SANITARY SEWER |
| | EXISTING SANITARY FORCEMAIN |
| 326 | EXISTING 2015 GRCA CONTOURS |
| | POTENTIAL SERVICEABLE LANDS TO THE BADEN TRUNK |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| | BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER |
| | ARCHAEOLOGICAL 20m NO-G0 BUFFER |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | HUEHN AWARD DRAIN 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND |
| | LOCAL 15m WETLAND BUFFER |
| | POTENTIAL BAT (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | DETAILED STUDY AREA |
| 0 | EXISTING WASTEWATER PUMPING STATION |
| ALTE | RNATIVES |
| 2 | SNYDER'S/FOUNDRY |
| \sim | |



P:\P\48730\100\48730−100−F3.2.dwg





| | LEGEND | | |
|--------------|--|--|--|
| | PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN | | |
| 326 | EXISTING 2015 GRCA CONTOURS | | |
| | POTENTIAL SERVICEABLE LANDS TO THE BADEN TRUNK | | |
| | GRCA REGULATORY FLOODPLAIN | | |
| | GRCA EXISTING WATERBODY | | |
| (\bigcirc) | BUTTERNUT 25m BUFFER | | |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER | | |
| | ARCHAEOLOGICAL 20m NO-GO BUFFER | | |
| | TRIBUTARY 'A' 10m BUFFER | | |
| | BADEN CREEK 15m BUFFER | | |
| | HUEHN AWARD DRAIN 30m BUFFER | | |
| | LOCALLY SIGNIFICANT WETLAND | | |
| | LOCAL 15m WETLAND BUFFER | | |
| | POTENTIAL BAT (SAR) HABITAT | | |
| | GRCA EXISTING WATERCOURSE | | |
| | TRENCHLESS CROSSING | | |
| | TOWNSHIP URBAN AREA | | |
| | COUNTRYSIDE LINE | | |
| | PRIMARY STUDY AREA | | |
| | DETAILED STUDY AREA | | |
| 0 | EXISTING WASTEWATER PUMPING STATION | | |
| ALTERNATIVES | | | |
| 3 | CHRISTIAN/CHARLOTTA/MILL/ CHARLES/FOUNDRY | | |



P:\P\48730\100\48730−100−F3.3.dwg





| | LEGEND | | |
|------------|---|--|--|
| | PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOURS | | |
| | POTENTIAL SERVICEABLE LANDS TO THE BADEN TRUNK GRCA REGULATORY FLOODPLAIN | | |
| | GRCA EXISTING WATERBODY | | |
| \bigcirc | BUTTERNUT 25m BUFFER | | |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER | | |
| | ARCHAEOLOGICAL 20m NO-G0 BUFFER | | |
| | TRIBUTARY 'A' 10m BUFFER BADEN CREEK 15m BUFFER HUEHN AWARD DRAIN 30m BUFFER | | |
| | LOCALLY SIGNIFICANT WETLAND | | |
| | LOCAL 15m WETLAND BUFFER | | |
| | POTENTIAL BAT (SAR) HABITAT | | |
| | GRCA EXISTING WATERCOURSE | | |
| | TRENCHLESS CROSSING | | |
| | TOWNSHIP URBAN AREA | | |
| | COUNTRYSIDE LINE | | |
| | PRIMARY STUDY AREA | | |
| | DETAILED STUDY AREA | | |
| | EXISTING WASTEWATER PUMPING STATION | | |
| ALTE | CHRISTIAN/CHARLOTTA/QUEEN AND BADEN CREEK | | |
| | | | |



P:\P\48730\100\48730−100−F3.4.dwg





| LI | EGEND |
|--------------|--|
| | |
| | |
| | |
| | |
| | EXISTING SANTART FORCEMAIN |
| 326 | EXISTING 2015 GRCA CONTOUR |
| | POTENTIAL SERVICEABLE LANDS TO THE BADEN TRUNK |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| (\bigcirc) | BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER |
| | ARCHAEOLOGICAL 20m NO-G0 BUFFER |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | HUEHN AWARD DRAIN 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND |
| *********** | LOCAL 15m WETLAND BUFFER |
| | POTENTIAL BAT (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | DETAILED STUDY AREA |
| | EXISTING WASTEWATER PUMPING STATION |
| ALTE | RNATIVES |
| 5 | CHRISTIAN AND BADEN CREEK |
| | |







| LE | - PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER |
|--------------|---|
| | PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER |
| | EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOUR |
| | POTENTIAL SERVICEABLE LANDS TO THE BADEN TRUNK |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| (\bigcirc) | BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER ARCHAEOLOGICAL 20m NO-GO |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER HUEHN AWARD DRAIN 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND |
| *********** | LOCAL 15m WETLAND BUFFER |
| | POTENTIAL BAT (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | DETAILED STUDY AREA |
| | EXISTING WASTEWATER PUMPING STATION |
| | CHRISTIAN AND BADEN CREEK |



P:\P\48730\100\48730−100−F3.5.dwg



Figures from 2014 CRA Report











Notifications







Notice of Study Commencement

Municipal Class Environmental Assessment

The Study

The Township of Wilmot has initiated a Class Environmental Assessment Study for the Baden Sanitary Trunk Main required in the settlement area of Baden. This sanitary trunk is proposed to accommodate current and future growth in the area. The study will determine the alignment and size of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.

The Process

This project is being carried out in accordance with the requirements of the Environmental Assessment Act by following the Municipal Class EA document (as amended in 2007, 2011, 2015 and 2020) and it is being planned under Schedule 'B' of the Municipal Class Environmental Assessment (Class EA). This process is being integrated with the Planning Act for active development applications in Baden.

MTE Consultants will undertake the study, which involves an evaluation of alternatives, selection of preferred alternative, and evaluation of environmental impacts and their mitigation measures. At the end of the study, a conceptual design will be prepared by MTE Consultants.

Comments Invited

Public, Indigenous, and agency consultation is a key component of the Class EA process, and we value your input during the planning process. To help facilitate this input, a Public Information Centre where some of the alternative solutions will be presented is scheduled to take place later in the study. If you wish to be placed on the study mailing list to receive notices and information, or, if you wish to provide comments at any time during the Class EA process, you can do so by contacting:

Dot Roga, C.E.T.

Project Manager

MTE Consultants Inc. Phone: (519) 743-6500 ext. 1269 Cell Phone: (519) 505-7507 Email: <u>droga@mte85.com</u>

Jeff Molenhuis, P. Eng

Director of Public Works & Engineering Township of Wilmot Phone: (519) 634-8444 Email: jeff.molenhuis@wilmot.ca

Please note that all correspondence will be kept on file for use during the decision-making process throughout the project and will become part of the public record. Under the Municipal Freedom of Information and Protection of Privacy Act and the Environmental Assessment Act, any personal information such as name, address, and telephone number included in a submission may become part of the public record unless otherwise requested in the submission.

Figure 1. This figure illustrates the limits of the primary study area and the detailed study area.



This notice first published on June 17, 2021.





November 17, 2021

Dear Resident:

RE: Baden Sanitary Trunk Main Class Environment Assessment - Notice of Public Information Centre No. 1

The Township of Wilmot has initialized a Schedule 'B' Municipal Class Environmental Assessment (EA) Study to identify the size and alignment for the Baden Sanitary Trunk Main. The Study will review and provide recommendations including:

- Determining the system capacity requirements to appropriately size the Baden Sanitary Trunk Main; and
- Identify a route for the new sewer that minimizes impacts to the environment, optimizes existing infrastructure in the community.

Effective consultation with the public and stakeholders is key to the success of this Class EA Study. The communications and consultation plan for this Class EA has been developed to ensure that key stakeholders and the public have a voice at each step along the way to help select the sewer alignment. Your property or business is located in proximity to the existing Baden Sanitary Trunk Main or a possible future conveyance alignment.

We invite you to participate in a Virtual Public Information Centre (PIC) available on the Township's website <u>www.wilmot.ca/en/doing-business/current-projects.aspx</u> **November 24, 2021.** The PIC will present information about the Study including alternative sanitary conveyance alignments and evaluation methodology, and will welcome your comments and feedback. Additional information about the project and details related to the Virtual PIC are included in the attached Notice.

Please contact a member of the project team below, if you have any concerns regarding this study.

We look forward to hearing from you.

Yours truly,

Dot Roga, C.E.T.

Project Manager MTE Consultants Inc. Phone: (519) 743-6500 ext. 1269 Cell Phone: (519) 505-7507 Email: <u>DRoga@mte85.com</u>

Jeff Molenhuis, P.Eng.

Director of Public Works & Engineering Township of Wilmot Phone: (519) 634-8444

Email: jeff.molenhuis@wilmot.ca





November 23, 2021

RE: Baden Sanitary Trunk Main Class Environment Assessment - Notice of Public Information Centre No. 1

The Township of Wilmot has initialized a Schedule 'B' Municipal Class Environmental Assessment (EA) Study to identify the size and alignment for the Baden Sanitary Trunk Main. The Study will review and provide recommendations including:

- Determining the system capacity requirements to appropriately size the Baden Sanitary Trunk Main; and
- Identify a route for the new sewer that minimizes impacts to the environment, optimize existing infrastructure within the community.

Effective consultation with the public and stakeholders is key to the success of this Class EA Study. The communications and consultation plan for this Class EA has been developed to ensure that key stakeholders and the public have a voice at each step along the way to help select the sewer alignment.

We invite you to participate in a Virtual Public Information Centre (PIC) available on the Township's website <u>www.wilmot.ca/en/doing-business/current-projects.aspx</u> **November 24, 2021**. The PIC will present information about the Study including alternative sanitary conveyance alignments and evaluation methodology, and will welcome your comments and feedback. Additional information about the project and details related to the Virtual PIC are included in the attached Notice.

Please contact a member of the project team below, if you have any concerns regarding this study.

We look forward to hearing from you.

Yours truly,

Dot Roga, C.E.T.

Project Manager MTE Consultants Inc. Phone: (519) 743-6500 ext. 1269 Cell Phone: (519) 505-7507 Email: <u>DRoga@mte85.com</u>

Jeff Molenhuis, P.Eng.

Director of Public Works & Engineering Township of Wilmot Phone: (519) 634-8444

Email: jeff.molenhuis@wilmot.ca





Municipal Class Environmental Assessment (Class EA) Baden Sanitary Trunk Main

Notice of Public Information Centre No. 1

The Township of Wilmot has initiated a Class Environmental Assessment Study for the Baden Sanitary Trunk Main required in the settlement area of Baden. This sanitary trunk is proposed to accommodate current and future growth in the area. The study will determine the alignment and size of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.

This project is being carried out in accordance with the requirements of the Environmental Assessment Act by following the Municipal Class EA document (as amended in 2007, 2011, 2015 and 2020) and it is being planned under Schedule 'B' of the Municipal Class Environmental Assessment (Class EA). This process is being integrated with the Planning Act for active development applications in Baden.

MTE Consultants and the Township has undertaken the study, which involves an evaluation of alternatives, selection of preferred alternative, and evaluation of environmental impacts and their mitigation measures. MTE has identified alternative sanitary servicing solutions and conceptual design.



Figure 1: Limits of the Primary Study Area and Detailed Study Area

Presentation, discussion, and input on the identified alternatives will be presented at the Public Information Centre (PIC). MTE Consultants and the Township of Wilmot invite all interested parties to participate in a Virtual PIC available on the **Township's Website** www.wilmot.ca/en/doing-business/current-projects.aspx on November 24, 2021.

Please note that comments will be maintained for reference throughout the project and will become part of the public record. Under the Municipal Freedom of Information and Protection of Privacy Act and the Environmental Assessment Act, any personal information such as name, address, and telephone number included in a submission will become part of the public record unless the comments specifically request that such personal details not be included in the public record.

Project information will be made available on the Township of Wilmot's website.

This Notice was issued on November 17, 2021.





September 19, 2022

Dear Resident:

RE: Baden Sanitary Trunk Main Class Environment Assessment - Notice of Public Information Centre No. 2

The Township of Wilmot has initialized a Schedule 'B' Municipal Class Environmental Assessment (EA) Study to identify the size and alignment for the Baden Sanitary Trunk Main. The Study will review and provide recommendations including:

- Determining the system capacity requirements to appropriately size the Baden Sanitary Trunk Main; and
- Identify a route for the new sewer that minimizes impacts to the environment, optimizes existing infrastructure in the community.

Effective consultation with the public and stakeholders is key to the success of this Class EA Study. The communications and consultation plan for this Class EA has been developed to ensure that key stakeholders and the public have a voice at each step along the way to help select the sewer alignment. Your property or business is located in proximity to the existing Baden Sanitary Trunk Main or a possible future conveyance alignment.

We invite you to participate in a Virtual Public Information Centre (PIC) available on the Township's website <u>www.wilmot.ca/en/doing-business/current-projects.aspx</u> **September 26, 2022.** The PIC will present information about the Study including alternative sanitary conveyance alignments and evaluation methodology, and will welcome your comments and feedback. Additional information about the project and details related to the Virtual PIC are included in the attached Notice.

Please contact a member of the project team below, if you have any concerns regarding this study.

We look forward to hearing from you.

Yours truly,

Dot Roga, C.E.T.

Project Manager MTE Consultants Inc. Phone: (519) 743-6500 ext. 1269 Cell Phone: (519) 505-7507 Email: <u>DRoga@mte85.com</u>

Jeff Molenhuis, P.Eng.

Director of Public Works & Engineering Township of Wilmot Phone: (519) 634-8444

Email: jeff.molenhuis@wilmot.ca





Municipal Class Environmental Assessment (Class EA) Baden Sanitary Trunk Main

Notice of Public Information Centre No. 2

The Township of Wilmot has initiated a Class Environmental Assessment Study for the Baden Sanitary Trunk Main required in the settlement area of Baden. This sanitary trunk is proposed to accommodate current and future growth in the area. The study will determine the alignment and size of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.

This project is being carried out in accordance with the requirements of the Environmental Assessment Act by following the Municipal Class EA document (as amended in 2007, 2011, 2015 and 2020) and it is being planned under Schedule 'B' of the Municipal Class Environmental Assessment (Class EA). This process is being integrated with the Planning Act for active development applications in Baden.

MTE Consultants and the Township has undertaken the study, which involves an evaluation of alternatives, selection of preferred alternative, and evaluation of environmental impacts and their mitigation measures. MTE has identified alternative sanitary servicing solutions and conceptual design.



Figure 1: Limits of the Primary Study Area and Detailed Study Area

Presentation, discussion, and input on the identified alternatives will be presented at the Public Information Centre (PIC). MTE Consultants and the Township of Wilmot invite all interested parties to participate in a Virtual PIC available on the **Township's Website** www.wilmot.ca/en/doing-business/current-projects.aspx on September 26, 2022.

Please note that comments will be maintained for reference throughout the project and will become part of the public record. Under the Municipal Freedom of Information and Protection of Privacy Act and the Environmental Assessment Act, any personal information such as name, address, and telephone number included in a submission will become part of the public record unless the comments specifically request that such personal details not be included in the public record.

Project information will be made available on the Township of Wilmot's website.

This Notice was issued on September 19, 2022.


Communication Mailing List, Meeting Minutes, and Signed Agreements



DERRY JOHN EDWARD 89 SNYDER'S RD W BADEN ON N3A 2M2

BECKER EARL JOHN 95 CHARLES ST BADEN ON N3A 2N9

PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

JC MILLER ANTIQUES LIMITED 8 FOUNDRY ST BADEN ON N3A 2P5

PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

ERB ENTERPRISES INC 290 HAMILTON RD SUITE 605 NEW HAMBURG ON N3A 1A2

WEICKER WILLIAM ANGUS 1578 GINGERICH RD BADEN ON N3A 3J8

PFEIFER MURRAY C/O KELLY & CO. BARRISTERS AND 50 QUEEN ST N UNIT 903 KITCHENER ON N2H 6P4

SCHMIDT ENID ELIZABETH C/O KELLY & CO. BARRISTERS AND SOLICITOR 50 QUEEN ST N UNIT 903 KITCHENER ON N2H 6P4

STRUYK TYLER 249 CHARLOTTA ST BADEN ON N3A 4M8 PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

SOP ZVONKO 29 PAULANDER DR UNIT 16 KITCHENER ON N2M 5L2

SCHMIDT ENID ELIZABETH C/O KELLY & CO BARRISTERS AND SOLICITOR 50 QUEEN ST N SUITE 903 KITCHENER ON N2H 6P4

PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

MILLER BOYS INC C/O JC MILLER ANTIQUES 10 FOUNDRY ST BADEN ON N3A 2P5

PFEIFER MURRAY C/O KELLY & CO. BARRISTERS AND SOLICITOR 50 QUEEN ST N UNIT 903 KITCHENER ON N2H 6P4

PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

STREMMA DEVELOPMENTS (BADEN 7050 MAJOR MACKENZIE DR KLEINBURG ON LOJ 1C0

MILLER WILLIAM ROY 60 VILLAGE GREEN WAY BADEN ON N3A 2V7 PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

PFEIFER MURRAY 148 FOUNDRY ST PO BOX 130 BADEN ON N3A 2P7

SCHMIDT ENID ELIZABETH PO BOX 138 148 FOUNDRY ST BADEN ON N3A 2P7

MILLER BOYS INC 10 FOUNDRY ST BADEN ON N3A 2P5

CONSERVATION AUTHORITY GRAND PO BOX 729 STN GALT CAMBRIDGE ON N1R 5W6

PESTELL LAND CORP 141 HAMILTON RD NEW HAMBURG ON N3A 2H1

WEICKER WILLIAM ANGUS 1578 GINGERICH RD BADEN ON N3A 3J8

STREMMA DEVELOPMENTS (BADEN 7050 MAJOR MACKENZIE DR KLEINBURG ON LOJ 1C0

STREMMA DEVELOPMENTS (BADEN 7050 MAJOR MACKENZIE DR KLEINBURG ON LOJ 1C0

BRAGA MICHAEL FAGUNDES 210 CHARLOTTA ST BADEN ON N3A 4M8 KRULICKI MARK ANDREW 93 BRENNEMAN DR BADEN ON N3A 4N1

GOODING CARL DAVID PO BOX 771 110 CHARLES ST BADEN ON N3A 2N8

FARKAS SANDOR PAUL 35 BRENNEMAN DR BADEN ON N3A 4M9

OSTOJSKI DONNA REGINA 25 SNYDER'S RD W BADEN ON N3A 2M1

APPLETON CHRISTIAN ALEXANDER 43 BRENNEMAN DR BADEN ON N3A 4M9

HINSCHBERGER DALTON CODY 242 BRENNEMAN DR BADEN ON N3A 4N2

PROCTER KRISTEN KAREN 59 BRENNEMAN DR BADEN ON N3A 4M9

SIDDORN MARK DAVID 218 CHARLOTTA ST BADEN ON N3A 4M8

STUMPF JORDAN DAVID 12 QUEEN ST BADEN ON N3A 2N1

WAECHTER TIMOTHY BRUCE 121 SNYDER'S RD W BADEN ON N3A 2M3 HOWARD GRANT ANTONY 233 CHARLOTTA ST BADEN ON N3A 4M8

ROTH LEOLA GINGERICH 46 CHRISTIAN ST BADEN ON N3A 2M6

VAILLANCOURT DAVID ROGER G 12 BRENNEMAN DR BADEN ON N3A 4M9

STANLEY JAMES R D 40 BRENNEMAN DR BADEN ON N3A 4M9

HASKETT WILLIAM CHARLES 80 BRENNEMAN DR BADEN ON N3A 4M9

LINES TODD MICHAEL 242 CHARLOTTA ST BADEN ON N3A 4M8

WUERZ KELLY 129 SNYDER'S RD W BADEN ON N3A 2M3

GALLIAN SHANE BRADLEY 68 BRENNEMAN DR BADEN ON N3A 4M9

GERBER BRUCE D 145 QUEEN ST BADEN ON N3A 2N3

RADMAN SANDRA 260 CHARLOTTA ST BADEN ON N3A 4M8 HADLAND CHARLES RICHARD 69 SNYDER'S RD W BADEN ON N3A 2M2

WATERS CHRISTINE MARIE 62 BECK ST BADEN ON N3A 2P4

DAVIS CLIFFORD AMOS WESLEY 36 BRENNEMAN DR BADEN ON N3A 4M9

RUHLMAN AMY LYNN 63 BRENNEMAN DR BADEN ON N3A 4M9

BRUCE MICHAEL GORDON 52 BRENNEMAN DR BADEN ON N3A 4M9

LINDNER MARILYN LOUISE 46 MILLER ST BADEN ON N3A 2M8

KUTTSCHRUTTER JOHN WILLIAM 202 CHARLOTTA ST BADEN ON N3A 4M8

FISCHER JOSEPH ALLAN 23 BRENNEMAN DR BADEN ON N3A 4M9

WATTERS STEVEN ARNOLD 89 BRENNEMAN DR BADEN ON N3A 4N1

LUCKHART PATRICIA LYNN 1081 SNYDER'S RD W RR 2 BADEN ON N3A 3K9 WILDFONG SCOTT RICHARD 85 BRENNEMAN DR BADEN ON N3A 4M9

LEE GREGORY THOMAS 100 CHARLOTTA ST BADEN ON N3A 2R4

PETERS DIEDRICH PENNER 105 BRENNEMAN DR BADEN ON N3A 4N1

WILKER AMIE COLLEEN 248 CHARLOTTA ST BADEN ON N3A 4M8

KERR CHARLES ARTHUR BOX 281 66 CHRISTIAN ST BADEN ON N3A 2M6

GOMES ALBERTO VALENTE 206 CHARLOTTA ST BADEN ON N3A 4M8

CORNING WILLIAM CHRISTI 95 SNYDER'S RD W BADEN ON N3A 2M2

ABDELMONIM TARIG 16 BRENNEMAN DR BADEN ON N3A 4M9

VORSTENBOSCH DARLENE MARIE 150 CHARLOTTA ST BADEN ON N3A 2R5

SHINNIE ADRIENNE ANNE 53 SNYDER'S RD W BADEN ON N3A 2M2 CHARLOTTA HOLDINGS INC 410 PINEBUSH RD CAMBRIDGE ON N1T 1Z6

SNYDER DEREK KENNETH 174 CHARLOTTA ST BADEN ON N3A 2R6

LAVALLEE COREY MARCEL 252 CHARLOTTA ST BADEN ON N3A 4M8

SIDER BRETT RICHARD 126 CHARLOTTA ST BADEN ON N3A 2R5

SKOWRON KATHRYN SHEILA 1203 SNYDER'S RD W BADEN ON N3A 3K9

RUDY MARY GENEVIEVE 256 CHARLOTTA ST BADEN ON N3A 4M8

YAVNER JONATHAN ARTHUR 30 FOUNDRY ST BADEN ON N3A 2P6

WEICKER WAYNE GRANT 165 SNYDER'S RD W BADEN ON N3A 2M4

WIESCHOLLEK JEREMY F P 90 CHARLOTTA ST BADEN ON N3A 2R4

SMITH PATRICIA PO BOX 239 24 MILLER ST BADEN ON N3A 2M8 STURGEON STEVEN EDWARD 27 BRENNEMAN DR BADEN ON N3A 4M9

ZINGER JASON GEORGE 230 CHARLOTTA ST BADEN ON N3A 4M8

OGASAWARA MORIO 32 MILLER ST BADEN ON N3A 2M8

MCSHANNOCK ADAM DAVID 277 CHARLOTTA ST BADEN ON N3A 4M8

DETZLER MARK GERARD 54 MILL ST BADEN ON N3A 2N6

BURNETT KYLE KENNETH 40 QUEEN ST BADEN ON N3A 2N1

OSINKOWSKI WALTER FRANK 67 CHRISTIAN ST BADEN ON N3A 2M6

JIMENEZ DANILO ALEXANDER 48 BRENNEMAN DR BADEN ON N3A 4M9

BAILLARGEON JEANETTE 72 BRENNEMAN DR BADEN ON N3A 4M9

DREDGE ALISON JOY 76 CHRISTIAN ST BADEN ON N3A 2M6 MCCLEAR MARY-EILEEN 89 SNYDER'S RD W BADEN ON N3A 2M2

FREY KENTON ERIC 24 BREWERY ST BADEN ON N3A 2S7

BEAVER DARYL KENNETH 225 CHARLOTTA ST BADEN ON N3A 4M8

PUCAN JERALD 241 CHARLOTTA ST BADEN ON N3A 4M8

REINHARDT JULIA ANNE 62 MILL ST BADEN ON N3A 2N6

FUREY PAUL JAMES 75 BRENNEMAN DR BADEN ON N3A 4M9

TARR JEFFREY BENJAMIN 44 BRENNEMAN DR BADEN ON N3A 4M9

SEIFRIED KEVIN KENNETH 237 CHARLOTTA ST BADEN ON N3A 4M8

DOMM MARK WAYNE 101 BRENNEMAN DR BADEN ON N3A 4N1

LONDRY JASON QUINTON 229 CHARLOTTA ST BADEN ON N3A 4M8 BROSOWSKI GLENN CALVIN 143 SNYDER'S RD W BADEN ON N3A 2M3

HENNIGE COLIN 31 BRENNEMAN DR BADEN ON N3A 4M9

HARRICKEY NEIL 253 CHARLOTTA ST BADEN ON N3A 4M8

EBY PATRICIA ANNE 111 SNYDER'S RD W BADEN ON N3A 2M2

SZUSZ RYAN DAVID 261 CHARLOTTA ST BADEN ON N3A 4M8

BEATON ANDREW GERARD 94 CHARLOTTA ST BADEN ON N3A 2R4

SMITH KATHLEEN JEAN 203 SNYDER'S RD W BADEN ON N3A 2M5

PHAM TRUONG XUAN 1547 GINGERICH RD BADEN ON N3A 3J7

DANAHY BRIAN ANDREW 245 CHARLOTTA ST BADEN ON N3A 4M8

MAHN ADAM THOMAS 64 BRENNEMAN DR BADEN ON N3A 4M9 BESTER KENNETH JAMES 56 CHRISTIAN ST BADEN ON N3A 2M6

GASCHO ADAM GERALD 39 CHRISTIAN ST BADEN ON N3A 2M7

BENDER LEON JOSEPH 41 MILLER ST BADEN ON N3A 2M9

MACKAY DAVID 281 CHARLOTTA ST BADEN ON N3A 4M8

GROENESTEGE JULIA 28 BRENNEMAN DR BADEN ON N3A 4M9

BECHTHOLD BLAIN IVAN 199 SNYDER'S RD W BADEN ON N3A 2M5

NIELSEN KARSTEN LEIF 14 FOUNDRY ST BADEN ON N3A 2P5

WINGER ELIZABETH JEAN 90 ST GEORGE ST BADEN ON N3A 2T5

NEILSON MYRNA LYNN 25 MILLER ST BADEN ON N3A 2M9

AMATO MARC ANTHONY 51 BRENNEMAN DR BADEN ON N3A 4M9 RANK JODY WILLIAM 33 MILLER ST BADEN ON N3A 2M9

OSINKOWSKI AMY LISE 178 CHARLOTTA ST BADEN ON N3A 4M8

VAN BEKKUM ALI DRIKJE 197 SNYDER'S RD W BADEN ON N3A 2M5

TURNER CHRISTOPHER ALAN 32 BRENNEMAN DR BADEN ON N3A 4M9

HOFFMAN WAID ALEXANDER 51 CHRISTIAN ST BADEN ON N3A 2M7

MILLER BOYS INC 10 FOUNDRY ST BADEN ON N3A 2P5

RUSSON MARLENE CATHERINE 111 CHARLES ST BADEN ON N3A 2N9

BARDOUNIOTIS MARIAN 47 BRENNEMAN DR BADEN ON N3A 4M9

ALBAYATY ARKAN 55 BRENNEMAN DR BADEN ON N3A 4M9

SPENCER MURRAY DOUGLAS 18 QUEEN ST BADEN ON N3A 2N1 SNOOK JONATHAN HOWARD 56 BRENNEMAN DR BADEN ON N3A 4M9

KIESWETTER KIMBERLEY 1221 SNYDER'S RD W BADEN ON N3A 3K9

ARCHER DARRYL 226 CHARLOTTA ST BADEN ON N3A 4M8

LIVINGSTON LAWRENCE EARLE BOX 226 36 MILLER ST BADEN ON N3A 2M8

LARIVIERE PAMELA JAYNE 41 QUEEN ST BADEN ON N3A 2N2

WALED KHANI HEWA 248 BRENNEMAN DR BADEN ON N3A 4N3

SCHOLL CHRISTOPHER MATTHEW 81 BRENNEMAN DR BADEN ON N3A 4M9

BOLEN-DEJONG LISA ANNE 24 BRENNEMAN DR BADEN ON N3A 4M9

EGERDEE VERNON DALE 109 BRENNEMAN DR BADEN ON N3A 4N1

WALL COLIN PATRICK 265 CHARLOTTA ST BADEN ON N3A 4M8 BRENNEMAN KAYLA BARBARA 268 CHARLOTTA ST BADEN ON N3A 4M8

FEWKES MICHAEL DAWS 38 COUNTRY CREEK DR BADEN ON N3A 2V2

HUGHES CHRISTOPHER G A 113 BRENNEMAN DR BADEN ON N3A 4N1

NILES TYLER DANE JOSEPH 20 BRENNEMAN DR BADEN ON N3A 4M9

SCHIRA STEVEN ANDREW 42 MILLER ST BADEN ON N3A 2M8

HOFSTETTER DWAYNE 39 BRENNEMAN DR BADEN ON N3A 4M9

LUSZCZKI JAKUB 76 BRENNEMAN DR BADEN ON N3A 4M9

BURNS THOMAS MARK 9 BRENNEMAN DR BADEN ON N3A 4M9

TAYLOR CATHERINE ANNE 1717 SNYDERS RD PETERSBURG ON NOB 2H0

BRICKER ROBERT ISAAC 153 QUEEN ST BADEN ON N3A 2N3 CARNEY WAYNE 60 BRENNEMAN DR BADEN ON N3A 4M9

SKOWRON KLAUS ALBERT 1209 SNYDER'S RD W BADEN ON N3A 3K9

BURGESS JONATHON DAVID 139 QUEEN ST BADEN ON N3A 2N3

SNYDER BRIAN PAUL 82 CHARLOTTA ST BADEN ON N3A 2R4

ROPP DALE ANDREW 8 CHRISTIAN ST BADEN ON N3A 2M6

INCH JAMES MARCUS 269 CHARLOTTA ST BADEN ON N3A 4M8

REIDT JUDITH ELAINE 8 BRENNEMAN DR BADEN ON N3A 4M9

MURRAY KENNETH RICHARD 67 BRENNEMAN DR BADEN ON N3A 4M9

BECKER KATHERINE MAE 95 CHARLES ST BADEN ON N3A 2N9

ALBRECHT DAVID RAY 1199 SNYDER'S RD W BADEN ON N3A 3K9 LINDSAY SCOTT CHARLES 132 CHARLOTTA ST BADEN ON N3A 2R5

WINDLEY RYAN JOHN 264 CHARLOTTA ST BADEN ON N3A 4M8

PAVEY CHARLES BRENT 147 SNYDER'S RD W BADEN ON N3A 2M3

VANDENAKERBOOM JOE HAROLD 146 FOUNDRY ST BADEN ON N3A 2P7

RAYMOND PATRICK BOX 242 34 CHRISTIAN ST BADEN ON N3A 2M6

DEBOER ANDREW RALPH 276 CHARLOTTA ST BADEN ON N3A 4M8

CARROLL JAMES L M 99 SNYDER'S RD W BADEN ON N3A 2M2

PARSONS PHILLIP JOHN-PAUL 222 CHARLOTTA ST BADEN ON N3A 4M8

2290816 ONTARIO LIMITED 88 ROOSEVELT AVE WATERLOO ON N2L 2N5

VANGYSSEL JESSICA MAY 24 BECK ST BADEN ON N3A 2P3 NETTERFIELD PAULETTE DARLENE 257 CHARLOTTA ST BADEN ON N3A 4M8

KRAEHLING WILBERT VINCENT 198 CHARLOTTA ST BADEN ON N3A 4M8

STUART LAWRENCE BYRON 60 MILLER ST BADEN ON N3A 2M8

BAILEY TIMOTHY MARSON 255 BRENNEMAN DR BADEN ON N3A 4N2

TSCHIRHART CHRISTOPHER T 33 QUEEN ST BADEN ON N3A 2N2

ASSMAN KARLA JESSICA 16 BECK ST BADEN ON N3A 2P3

TAYLOR COLIN FREDERICK 104 CHARLES ST BADEN ON N3A 2N8

PATTEN DONALD JOHN 19 BRENNEMAN DR BADEN ON N3A 4M9

LAWRENCE GEORGE WALLACE 91 CHARLES ST BADEN ON N3A 2N9

MARTIN ROBERT JAMES 273 CHARLOTTA ST BADEN ON N3A 4M8 DAVIES ALAN BUD 236 CHARLOTTA ST BADEN ON N3A 4M8

JUTZI BRENDA LYNN 57 CHRISTIAN ST BADEN ON N3A 2M7

THE TRUSTEES OF THE 44 BECK ST PO BOX 291 BADEN ON N3A 2P3

STONE MICHAEL ORVILLE 4 BRENNEMAN DR BADEN ON N3A 4M9

VERMA PRADEEP KUMAR 15 BRENNEMAN DR BADEN ON N3A 4M9

HALLMAN PENNY PO BOX 5144 BADEN ON N3A 4J3

DENSTEDT REID STEWART 195 BRENNEMAN DR BADEN ON N3A 4N2

WILHELM MARK GREGORY 230 BRENNEMAN DR BADEN ON N3A 4N2

ZETTEL TIMOTHY WILLIAM 212 BRENNEMAN DR BADEN ON N3A 4N2

WALSER DARYL 191 BRENNEMAN DR BADEN ON N3A 4N3 ABBOTT KEVIN ROBERT ANTHONY 97 BRENNEMAN DR BADEN ON N3A 4N1

ARMSTRONG DONALD CRAIG 1215 SNYDER'S RD E BADEN ON N3A 3L2

LONG DAVID JOSEPH 96 ST GEORGE ST BADEN ON N3A 2T5

BENDER RETA MAGDALENE 159 SNYDER'S RD W BADEN ON N3A 2M4

MILLER BOYS INC 10 FOUNDRY ST BADEN ON N3A 2P5

RILEY DONALD JAMES 1423 GINGERICH RD PO BOX 5153 NEW HAMBURG STN BADEN BADEN ON N3A 4J3

RYAN ERIN LEIGHANN 190 BRENNEMAN DR BADEN ON N3A 4N3

WALL PETER 200 BRENNEMAN DR BADEN ON N3A 4N2

MCNICHOL DONNA JANE 227 BRENNEMAN DR BADEN ON N3A 4N2

BROADFOOT KEVIN SCOTT 129 BRENNEMAN DR BADEN ON N3A 4N1 GRAVILL ROBERT CHADWICK 1529 NAFZIGER RD NEW HAMBURG ON N3A 3H3

MILLER BOYS INC 10 FOUNDRY ST BADEN ON N3A 2P5

GLAUSER PAUL GORDON 214 CHARLOTTA ST BADEN ON N3A 4M8

SANTOS PAUL FERNANDO 26 QUEEN ST BADEN ON N3A 2N1

KRISSAK TIMOTHY JOHN JOSEPH 75 SNYDER'S RD W BADEN ON N3A 2M2

SMITH NIGEL CRAIG 29 FIRST AVE KITCHENER ON N2C 1N3

DICKSON STEPHEN ROBERT 179 BRENNEMAN DR BADEN ON N3A 4N3

BRUZZESE CONOR DANIEL 220 BRENNEMAN DR BADEN ON N3A 4N2

NICHOLLS SARAH ELIZABETH 141 BRENNEMAN DR BADEN ON N3A 4N1

MACKLAM DIANE MICHELLE 143 BRENNEMAN DR BADEN ON N3A 4N1 ALEXANDER ERIKA 239 BRENNEMAN DR BADEN ON N3A 4N2

ACKLES RYAN DAVID 196 BRENNEMAN DR BADEN ON N3A 4N2

AKIFOVSKI ERDJAN 189 BRENNEMAN DR BADEN ON N3A 4N3

GRAVILL MICHELLE 155 BRENNEMAN DR BADEN ON N3A 4N1

GOUGH JESSE CLIFFORD 164 BRENNEMAN DR BADEN ON N3A 4N1

JOSEPH NATHAN PHILIP 44 HUNSBERGER DR BADEN ON N3A 4S5

CLARKE STEPHEN 161 BRENNEMAN DR BADEN ON N3A 4N1

PEARSON KIMBERLEY THERESA 158 BRENNEMAN DR BADEN ON N3A 4N1

KINDLER LAURA MARIE 224 BRENNEMAN DR BADEN ON N3A 4N2

RIVEROS JOSE LUIS 122 BRENNEMAN DR BADEN ON N3A 4N1 KROPF JOHN MATTHEW 188 BRENNEMAN DR BADEN ON N3A 4N3

MAINLAND DEANNA GRACE 50 GEORGE ST NEW HAMBURG ON N3A 1W4

MILLER JESSICA LYNN 214 BRENNEMAN DR BADEN ON N3A 4N2

ARACHCHIGE BOSITH 243 BRENNEMAN DR BADEN ON N3A 4N2

RAMSDEN MICHAELA DESIREE 147 BRENNEMAN DR BADEN ON N3A 4N1

HORNER TINA ELIZABETH 209 BRENNEMAN DR BADEN ON N3A 4N2

JANTZI BRENDA LEE 128 BRENNEMAN DR BADEN ON N3A 4N1

CAIAZZO DOMINIC JONATHAN 236 BRENNEMAN DR BADEN ON N3A 4N2

WHITELAW JOHN THOMAS 237 BRENNEMAN DR BADEN ON N3A 4N2

SCHWARTZENTRUBER GREGORY 14 CHRISTIAN ST BADEN ON N3A 2M6 ROTONDO LUKE ANTHONY 232 BRENNEMAN DR BADEN ON N3A 4N2

SMITH JENNIFER GWENDOLYNN 166 BRENNEMAN DR BADEN ON N3A 4N1

MALLON LAWRENCE ANTHONY 32 BENT WILLOW DR KITCHENER ON N2N 2K9

SAUNDERS DENISE MARY 176 BRENNEMAN DR BADEN ON N3A 4N3

HRYCYK MATTHEW ROBERT 226 BRENNEMAN DR BADEN ON N3A 4N2

SMAGLINSKI JULIE LYNNE 130 BRENNEMAN DR BADEN ON N3A 4N1

QUEHL BARBARA FRANCES 120 BRENNEMAN DR BADEN ON N3A 4N1

MAY ADAM JAMES BRENT 225 BRENNEMAN DR BADEN ON N3A 4N2

BLAZIK LEEANN JOAN 131 BRENNEMAN DR BADEN ON N3A 4N1

SCHMIDT RICHARD ALLAN 218 BRENNEMAN DR BADEN ON N3A 4N2 COOTE JAIME LYN 202 BRENNEMAN DR BADEN ON N3A 4N2

LEIS KATHRYN ELIZABETH 22 CHRISTIAN ST BADEN ON N3A 2M6

RYERSON RICHARD 203 BRENNEMAN DR BADEN ON N3A 4N2

FERRO OMAR TOMAS 110 CHARLOTTA ST BADEN ON N3A 2R4

HIGGINS CRAIG PAUL CHARLES 108 CHARLOTTA ST BADEN ON N3A 2R4

WALSH NICOLE RAE 149 BRENNEMAN DR BADEN ON N3A 4N1

EVANS JODIE LYNN 233 BRENNEMAN DR BADEN ON N3A 4N2

VALCANOFF TYLER MICHAEL 206 BRENNEMAN DR BADEN ON N3A 4N2

LOUCKS AARON JAMES 125 BRENNEMAN DR BADEN ON N3A 4N1

LONERGAN BRIAN R D 167 BRENNEMAN DR BADEN ON N3A 4N3 HALLMAN KEVIN BARRY 16 CHRISTIAN ST BADEN ON N3A 2M6

LYON ADAM PAUL 165 BRENNEMAN DR BADEN ON N3A 4N1

MALTHANER ADAM ROLF 184 BRENNEMAN DR BADEN ON N3A 4N3

PADMANATHAN CHENTHIVELNATHAN 213 BRENNEMAN DR BADEN ON N3A 4N2

MCKAY STEVEN JAMES 249 BRENNEMAN DR BADEN ON N3A 4N2

ELVINS BARRY CLEASON 20 CHRISTIAN ST BADEN ON N3A 2M6

WAGLER TREVOR DALE 117 BRENNEMAN DR BADEN ON N3A 4N1

KOCH KRISTOPHER ALLAN 194 BRENNEMAN DR BADEN ON N3A 4N2

BOHNERT JONATHAN 135 BRENNEMAN DR BADEN ON N3A 4N1

FAIRHALL KENNETH CHARLES 177 BRENNEMAN DR BADEN ON N3A 4N3 YIU YUEN KI 238 BRENNEMAN DR BADEN ON N3A 4N2

KOVACS GRACE 178 BRENNEMAN DR BADEN ON N3A 4N3

CLUNIES IAN OSMOND 215 BRENNEMAN DR BADEN ON N3A 4N2

FARQUHAR BRADLEY EDWARD 231 BRENNEMAN DR BADEN ON N3A 4N2

REIS FERNANDO AMARAL 201 BRENNEMAN DR BADEN ON N3A 4N2

STRAUS ANTHONY JOHN 160 BRENNEMAN DR BADEN ON N3A 4N1

ALBRECHT MATTHEW JAMES 197 BRENNEMAN DR BADEN ON N3A 4N2

BRICKER JASON JACK 106 CHARLOTTA ST BADEN ON N3A 2R4

TOTH THOMAS ROBERT 172 BRENNEMAN DR BADEN ON N3A 4N3

BURKE ROBYN HARMONY 170 BRENNEMAN DR BADEN ON N3A 4N3 GRASSI JEFFREY MICHAEL 105 SNYDER'S RD W BADEN ON N3A 2M2

POHL NICOLE IIONA 30 CHARLES ST BADEN ON N3A 2N7

MANUEL COLLEEN ELLEN 221 BRENNEMAN DR BADEN ON N3A 4N2

GRAHAM ROBERT ANDREW 183 BRENNEMAN DR BADEN ON N3A 4N3

STEINMANN RICHARD 499 FOXHILL PL WATERLOO ON N2T 1X6

BUCHANAN BONNIE LYNN 153 BRENNEMAN DR BADEN ON N3A 4N1

CAPRIGLIONE MASONRY INC 20 LASCHINGER BLVD NEW HAMBURG ON N3A 2G7

ARNOTT RICHARD WILLIAM 136 FOUNDRY ST BADEN ON N3A 2P7

HALLMAN KEITH ALLEN 63 SNYDER'S RD W BADEN ON N3A 2M2

MILLER ETHAN ROBERT 130 QUEEN ST BADEN ON N3A 2P1 VAN HERK KAITLIN ROSE 245 BRENNEMAN DR BADEN ON N3A 4N2

JEFFERIES LOIS YVONNE 251 BRENNEMAN DR BADEN ON N3A 4N2

HYNES DAVID CRAIG 185 BRENNEMAN DR BADEN ON N3A 4N3

YIP JASON CHI-SING 103 SNYDER'S RD W BADEN ON N3A 2M2

ORGAN JEREMY FABIAN 159 BRENNEMAN DR BADEN ON N3A 4N1

BECKER EDWARD WILLIAM 4357 WILMOT EASTHOPE RD NEW HAMBURG ON N3A 3T2

BRENNEMAN KATY MADELYN 58 MILL ST BADEN ON N3A 2N6

CAPRIGLIONE FRANCESCO 20 LASCHINGER BLVD NEW HAMBURG ON N3A 2G7

HALLMAN PENNY PO BOX 5144 BADEN ON N3A 4J3

SNYDER HEATHER KATHERINE 135 SNYDER'S RD W BADEN ON N3A 2M3 FRITZ PATRICK 434 FAIRVIEW ST NEW HAMBURG ON N3A 1M2

CLAYTON TIMOTHY PAUL 182 BRENNEMAN DR BADEN ON N3A 4N3

BEEKSMA BRIAN 219 BRENNEMAN DR BADEN ON N3A 4N2

WILKINSON CATHERINE HELEN 119 BRENNEMAN DR BADEN ON N3A 4N1

KORNELSEN WAYNE ALAN 347 NORTHLAKE DR WATERLOO ON N2V 1R1

CAPRIGLIONE MASONRY INC 20 LASCHINGER BLVD NEW HAMBURG ON NOB 2G0

MCKAY ANGELA 18 BECK ST BADEN ON N3A 2P3

SNYDER JAYME 92 CHARLES ST BADEN ON N3A 2N8

JAMTECH HOLDINGS INC 478 UNDERWOOD CRES OAKVILLE ON L6L 5P1

2769934 ONTARIO INC 2544 NEW JERUSALEM RD ELMIRA ON N3B 2Z2 2082970 ONTARIO INC 1457 GINGERICH RD BADEN ON N3A 3S7

TAV-PRO RURAL HOLDINGS LTD 25 HOPE ST E TAVISTOCK ON NOB 2R0

MILLER & MILLER AUCTIONS LTD 8 FOUNDRY ST BADEN ON N3A 2P5

925384 ONTARIO INC 1457 GINGERICH RD BADEN ON N3A 3J7

2238269 ONTARIO INC 76 MILL ST UNIT 1 BADEN ON NOB 1G0

2113102 ONTARIO INC PO BOX 5145 BADEN ON N3A 4J3

1011399 ONTARIO LIMITED 292 GREENBROOK DR KITCHENER ON N2M 4K2

ST JAMES LUTHERAN CHURCH 66 MILL ST BADEN ON N3A 2N6 SATURN POWER INC 100 MILL ST UNIT F PO BOX 6087 NEW HAMBURG ON N3A 2K6

2238269 ONTARIO INC 76 MILL ST BADEN ON N3A 2N6

MCINALLY CRAIG 55 SNYDER'S RD W BADEN ON N3A 2M2

925384 ONTARIO INC PO BOX 839 NEW HAMBURG ON NOB 2G0

C V K HOLDINGS INC 78 LASCHINGER BLVD NEW HAMBURG ON N3A 2G7

MILLER & MILLER AUCTIONS LTD 8 FOUNDRY ST BADEN ON N3A 2P5

KITCHENER-WILMOT HYDRO INC PO BOX 9010 STN C KITCHENER ON N2G 4L2

LIVINGSTON PRESBYTERIAN 44 BECK ST BADEN ON N3A 2P3 2813708 ONTARIO LIMITED 26 FOUNDRY ST BADEN ON N3A 2P6

THE BADEN HOTEL CORP 39 SNYDER'S RD W BADEN ON N3A 2M1

HOPE HOLDING CO LTD 4476 SANDHILLS RD RR 1 ST AGATHA ON NOB 2L0

WEST AG RESOURCES INC 280 WILMOT ST NEW HAMBURG ON N3A 1H5

ERB ENTERPRISES INC 290 HAMILTON RD 290 HAMILTON RD NEW HAMBURG ON N3A 1A2

PESTELL LAND CORP 141 HAMILTON RD NEW HAMBURG ON N3A 2H1

WATERLOO REGIONAL 150 FREDERICK ST KITCHENER ON N2H 6T1

STEINMANN MENNONITE CHURCH 1317 SNYDER'S RD W RR 2 BADEN ON NOB 1G0 Stakeholders Mailing Labels

Ava Hill, Chief Six Nations of the Grand River 1695 Chiefswood Road, PO Box 5000 Ohsweken, NOA 1M0

Environment Canada 4905 Dufferin Street Toronto, M3H 5T4

Bryan LaForme, Chief Mississaugas of the New Credit First Nation R.R. 6, 2789 Mississauga Road Hagersville, ON NOA 1H0

Allison Berman, Manager Indigenous and Northern Affairs Canada 10 Wellington Street, 8th Floor Gatineau, K1A 0H4

Monique Mousseau, Regional Manager Transport Canada 4900 Yonge Street, Suite 300 Toronto, ON M2N 6A5

Rachael Manson-Smith, Manager Acting Ministry of Indigenous Affairs 160 Bloor Street East, 9th Floor Toronto, M7A 2E6

Carol Neumann, Rural Planner Ministry of Agriculture, Food and Rural Affairs, 6484 Wellington Road 7, Unit 10 Elora, ON NOB 1S0

Marion-Frances Cabral, Planner Ministry of Municipal Affairs & Housing 659 Exeter Road, 2nd Floor London, ON N6E 1L3

Joseph Muller, Heritage Planner Ministry of Heritage, Sport, Tourism and Culture Industries 401 Bay Street, Suite 1700 Toronto ON M7A 0A7

Joseph Muller, Heritage Planner Ministry of Heritage, Sport, Tourism & Culture 401 Bay Street, Suite 1700 Toronto ON M7A 0A7 Terra McKenna, District Planner Ministry of Natural Resources and Forestry I Stone Road West Guelph ON N1G 4Y2 Cyrus Elmpal-Mackie Hydro One Networks 483 Bay Street, 15th Floor Toronto ON M5G 2P5

Ian Hagman, District Planner Ministry of Natural Resources and Forestry I Stone Road West Guelph ON N1G 4Y2

Kathleen Hedley, Director Ministry of Environment, Conservation and Parks 135 St. Clair Avenue West Toronto ON M4V 1P5

Dolly Goyette, District Manager Ministry of Environment, Conservation and Parks 1 Stone Road West Guelph, ON N1G 4Y2

Jane Glassco, Manager Ministry of Environment, Conservation and Parks 1 Stone Road West Guelph, ON N1G 4Y2

Clare Barnett, Director Ministry of Economic Development, Job Creation and Trade 900 Bay Street, 8th Floor Toronto, ON M7A 2E1

Sandra Crooke, Senior Water Quality Supervisor Grand River Conservation Authority 400 Clyde Road, PO Box 729 Cambridge, ON N1R 5W6

Kevin Schimus Manager, Waterloo District Enbridge Gas Inc. 603 Kumpf Drive Waterloo ON N2V 1K3

Ted Hancocks Rogers Cable 85 Grand Crest Place, PO Box 488 Kitchener ON N2G 4A8

Gayle Widmeyer, Manager Bell Canada 575 Riverbend Drive, Floor 2 Kitchener ON N2K 3S3 **Meeting Minutes**



| Project Name: | Beckdale Sanitary Trunk EA | MTE File No.: | C48730-100 |
|-------------------|----------------------------|---------------|----------------|
| Purpose: | Initiation Meeting | Date: | March 16, 2021 |
| Time: | 10 am | | |
| Meeting Location: | Teleconference | | |

| Attendees | Initials | Representation |
|-------------------|----------|----------------------|
| Valentina Lazic | JL | MTE Consultants Inc. |
| Jeff Martens | JM | MTE Consultants Inc. |
| Dot Roga | DR | MTE Consultants Inc. |
| Amy Domartzki | AD | MTE Consultants Inc. |
| Kassandra Wallace | KW | MTE Consultants Inc. |
| Chelsea Kochany | СК | MTE Consultants Inc. |
| Paul Britton | PB | MHBC |
| Ryan Archer | RA | NRSI |
| | | |

| ltem No. | Discussion | Action |
|-------------|---|--------|
| 1.0 | Purpose of Meeting | |
| | Prepare to begin the EA process. | |
| | Confirm Project Team | |
| | Ryan and Garth were consultants for Schmidt Lands; their information along with our scope of work will round out the natural environment and archaeological needs of the project. | |
| | Initiate Studies | |
| | Prepare for external meeting with Township Schmidt Lands is stakeholder, other developers are Activa and Nadiva Kickoff meeting to involve land owners | |
| | Steering committee - GRCA | |
| | Jeff Mollenhuis is waiting to meet – Township meeting first, then invite landowners to attend subsequent meeting | |
| 2.0 | Studies Update | |



| ltem No. | Discussion | Action |
|-------------|--|--------|
| | NRSI – RA will complete the necessary Natural Environment Study, confirm any sensitive areas, natural habitats, etc. Was able to confirm scope in meeting. | RA |
| | Also will work with archaeologist for the Schmidt Lands Do we require groundwater monitoring for the last 2 years? (Schmidt) Seasonal studies are good to proceed | |
| | Pinchin doing drilling – access information – Steve P will collect this GSP | |
| | Delritus – Garth Grimes will complete the archaeological study for the areas of concern | |
| | MHBC – Dan Currie will complete the Cultural Heritage Review | |
| | ROPP and OP mapping by MHBC will be repurposed by MHBC | |
| | GRCA – to request modelling | |
| | Peto – Bill Loghrin has not been confirmed for geotechnical work as of yet but assumed | |
| | MTE – AD will complete the hydrogeo study with scope yet to be defined; broader study area, lands that could be serviced by trunk MTE will gather modelling for the Trib/hydraulic modelling | AD |
| | MHBC – PB will contribute planning information and updates Process flow chart for EA committed to landowner involvement Staggered draft plans expected | РВ |
| | PB – says to hitch the EA to this draft plan | |
| | PB et al will draft problem statement | |
| | CRA background study – 2014 – draw from that background and Region – water/wastewater study Pinchin/Region for Geotech | PB |
| 3.0 | EA Process | |
| | In meeting, it was stated that this is Schedule 'B', but based on the integrated process, will be treated as Schedule 'C'. | |
| | A Problem Statement will need to be devised and Notice of Project Commencement shared shortly. To be done before meeting with Township | |
| | Confirm stakeholders list and maintain communication plan. | |
| | GRCA – to be contacted | ГD |



| ltem No. | Discussion | Action |
|-------------|--|----------|
| | Baden Creek is categorized as a warm water feature with some cold water (groundwater) input, NRSI recommended 15m buffer in the EIS for the Schmidt Lands | |
| | CN / GEXR leased – contact through AECOM? / Spur Line/Pestills Dot to check with Peter Johnson from AECOM | |
| | Confirm method of Public Consultation and quantity of meetings with the public and stakeholders. | |
| | Share study area with Township | |
| | Key base plan with contours/ownership areas | VI /AI I |
| | Title block developed for consistency – logo to be added to the title block from all | V L// LL |
| 4.0 | Alternatives to Consider | |
| | MTE has developed a few options for the alignment of this sanitary trunk. These options will be further developed prior to considering evaluation. We have 7 alternatives to consider at this point. | |
| | Technical assessment is an important component as well as the impacts to existing natural features (tributary crossings, alignment near tributary) | |
| | GRCA to be advised early on for buffer limits and general buy in. | |
| | 15m in Schmidt Lands | |
| | Cost assessments including Capital, Operational and Maintenance cost to be evaluated | |
| | Payment plan – final cost sharing to Nadiva and Activa | |
| | Letter to Township confirming the above Reimbursement to developers | |
| 5.0 | Schedule | |
| | Filing ESR by end of 2021 requires that these studies be initiated immediately after meeting with the Township which will be scheduled within the next couple of weeks. | |
| | Planning process also aiming to be completed by end of 2021 | |

M:\48730\100\Meetings\48730-100_Minutes_2021-03-16_Project Initiation Meeting - Internal_cek.docx



| Project Name: | Baden Sanitary Trunk EA | MTE File No.: | C48730-100 |
|-------------------|-------------------------|---------------|----------------|
| Purpose: | Initiation Meeting | Date: | April 21, 2021 |
| Time: | 9:30AM | | |
| Meeting Location: | Teleconference | | |

| Attendees | Initials | Representation |
|------------------|----------|----------------------|
| Jeff Molenhuis | JM | Township of Wilmot |
| Harold O'Krafka | НО | Township of Wilmot |
| Bryan Bishop | BB | Township of Wilmot |
| Nick Gougoulias | NG | Nideva Properties |
| Peter Armbruster | PA | Activa |
| Hitesh Lad | HL | Stantec |
| Jeff Martens | JXM | MTE Consultants Inc. |
| Dot Roga | DR | MTE Consultants Inc. |
| Chelsea Kochany | CK | MTE Consultants Inc. |
| Paul Britton | PB | MHBC |
| Pierre Chauvin | PC | MHBC |
| Kevin Brousseau | KB | Stantec |
| | | |

| ltem No. | Discussion | Action |
|-------------|--|--------|
| 1.0 | Cost Sharing | |
| | PB has received documents from JM Funds available from Township The Township has insufficient funds to cover entire study cost. Nideva and Activa will be responsible for shortfall. In addition, Nideva and Activa will initially upfront the Township funds that are available (\$150,000.00) and will be reimbursed by the Township in proportion to their upfront contributions (Nideva – 56.5% and Activa – 45.5%) PB confirmed with Nideva and Activa developers for cost sharing Without servicing outlet, project development not possible PB proposes segregated file for accounting purposes PB looks for guidance for invoices from Township Send monthly? Total amount included and then show breakdown on invoice for developers? | |



| ltem No. | Discussion | Action |
|-------------|---|--------|
| | Developers initially share township's share, and then reimbursed from Township, with reimbursement at 2 separate milestones (80% costs incurred by PIC with EA recommendation, balance reimbursed with Notice of Study Completion) JM, HO to weigh in for the authorization from Township to proceed, invoice preference, formalizing reimbursements JM – prefers to be kept in loop on project/payment status; if monthly invoice from MTE, provide copy to Township HO – recouping costs done as draft approval condition (preferred way due to scheduling, and can usually be done at building permit stage), could be done as DC credits (depending on status), payment at milestone stages As agreement impacts Schmidt lands, they should be invited to the table as donation of land MAY be required PB – ensure formalized understanding and recaps Township to receive copies of invoices Reimbursement requires proof of payment; offset through DC funding at building permit stage | |

M:\48730\100\Meetings\48730-100_Minutes_2021-04-21_Cost Sharing Meeting_cek.docx



| Project Name: | Baden Sanitary Trunk EA | MTE File No.: | C48730-100 |
|-------------------|-------------------------|---------------|---------------|
| Purpose: | Stakeholder's Meeting | Date: | July 20, 2022 |
| Time: | 3:00pm-4:30pm | | |
| Meeting Location: | Teleconference | | |

| Attendees | Initials | Representation |
|-----------------|----------|----------------------|
| Jeff Molenhuis | JM | Township of Wilmot |
| Harold O'Krafka | НО | Township of Wilmot |
| Bryan Bishop | BB | Township of Wilmot |
| Nick Gougoulias | NG | Nideva Properties |
| Jeff Martens | JXM | MTE Consultants Inc. |
| Dot Roga | DR | MTE Consultants Inc. |
| Valentina Lazic | VAL | MTE Consultants Inc. |
| Joanne McDaniel | JM | MTE Consultants Inc. |
| Paul Britton | PB | MHBC |
| Pierre Chauvin | PC | MHBC |
| Hitesh Lad | HL | Stantec |
| Dan Ferguson | DF | Walter Fedy |
| | | |

| ltem No. | Discussion | Action |
|-------------|--|--------|
| 1.0 | Purpose of Meeting | |
| | Review the evaluation of alternative routes for the Baden Sanitary Trunk alignment. | |
| | Confirm next steps and prepare PIC #2 for public input. | |
| 2.0 | EA Process | |
| | The Problem Statement was defined as: | - |
| | "To determine the preferred size, alignment and drainage area of the Baden Trunk Sanitary Sewer that will outlet to the Existing Baden SPS." | |
| | The Project Team completed the various studies required to properly evaluate the proposed route alternatives as outlined in the PIC #1 presentation to find a solution to the defined Problem Statement. | |
| | These studies included; Natural Environment, Archaeological, Hydrogeological, | |

Geotechnical and Geomorphological as it relates to the Baden Creek.



| ltem No. | Discussion | Action |
|-------------|--|--------|
| 3.0 | Studies Update | |
| | NRSI –Completed the necessary Natural Environment Study, confirmed any sensitive areas, natural habitats, etc. and evaluated the alternatives based on their findings. | |
| | Delritus –Completed the archaeological study for the areas of concern. Three First Nations groups were invited to participate and after several months of delay due to wet weather conditions, the study has been completed. Final report will be sent to the First Nations for review and then that requirement is satisfied. Post Meeting Note: adjustment to the alignment prompted additional archaeological investigation to be conducted. | |
| | MHBC –Completed the Cultural Heritage Review | |
| | MTE – Hydrogeo Study and Geotech Investigation were completed after a review of completed studies yielded the scope of work. | |
| | MHBC – Has contributed planning information and updates | |
| | Technical Memos for each Study will be included in the Environmental Study Report that is being prepared currently with plans to publish in late August. | |
| | GRCA has provided buffer zones and information on areas of concern. | |
| 4.0 | Alternatives to Consider | |
| | MTE has developed five separate alignments of this sanitary trunk for evaluation. These alternatives were presented in the virtual PIC held in November of 2021. | |
| | Review and evaluation of the five alternatives | |
| | The alternatives will be assessed for overall Cost including Capital, Operational and Maintenance. | |
| 5.0 | Schedule | |
| | We intend to go out to the public again in late July with a virtual presentation of our process to date.Tech memos are being finalized and the ESR will be circulated in draft for comments in August with the intent of filing it by the end of August/early | |
| | September. | |

M:\48730\100\Meetings\48730-100_Minutes_2022-07-20_Stakeholders Meeting Pre-PIC #2.docx

SIGNED AGREEMENTS

Stx Nations of the Grand River Riccard Council Archaeological Monitoring Agreement

The purpose of this synoarcest is to ensure that his Nations of the Grant Silver Elected Council ("SNGRUC") with the extraordistic for the reasonable cover (b) ONE Adduesdogical menium ("Monther") is connection with archaeotopical work required for the <u>Borton Fact here</u> Transition (respectively, let <u>Borton</u>), 2021 (weather dependent). The developse brighters is <u>To reaction of the sort</u>

The Worston newloce will be to marrier and work directly Will the a clear-logical frem Oscibility <u>Constitute</u> state to ensure that SNORDO's properties and priorities are care dereand to provide any applicable feedback of SNORDO's projections and priorities are care dereto statingfel contract on the Croject is part of its cher diligence as related for the Millibery of Halland, Spint. Touchers and Colours Industries (MIRST CI)² wardwards and guidelines, and the Appriment Ingagement Process.

<u>Formational 64 and — and SNGROUT (regetter, the "Paralies (Lapracition price Cases and SNGROUT (regetter, the "Paralies (Lapracities (Cases and Social International Control (Cases and Cases)) (Cases and Cases and Social International Cases and C</u>

- The Monitor will be selected by SNGREUT and will have appropriate quality at one for the required work, for controlle, militagin accurcibing and a working intervieting of field work web masses.
- 2. The Difference will be negrenerated the biddles over personal safety equipment with essays affety based, safety plasses, surgery west, based and, the The Member will strend affet appriate uncertained by the fail of a strength of exceptions and a presenced by the fail of the constraints affect to safety to say meanings presenced by the fail of the constraints affect to safety to say meanings presenced by the fail of the constraints affect to safety to say meanings presenced by the fail of the constraints affect to safety to say meanings presenced by the fail of the constraints affect to safety to say meanings presenced by the fail of the constraints affect to safety to saf
- 3 Definities is field finator will execting the meeting locations rate cause circle with SMGREC's pain, of period period, Taryo Hill Meeting, Andractingy Supervised, or the Mentler once designs of the fire site. See may be perioded at 519-352-0665 or work small address <u>period</u> ill proceeding in the period.
- <u>bCT6</u> will reliable with the manual rap work in the Monther reasonable expression in connection with the manual rap work in the anomal of \$125.00 per scar sinking monther. Tak account will cover all costs account with this project such activity reliable costs account of an expression with the project such activity of the administrative inclusion of the State CPP cost and construct the system (a), the administrative inclusion of the State State CPP cost and construct the system (a), the administrative inclusion of the State State CPP cost and construct the system (a), the administrative inclusion of the State St
- 2. <u>ATE</u> will create payment to 9NOREC by chopes or back rapet: with 3E days of reacipt of investor. The loss jee will be addressed in <u>Det. Rep. as inducting</u> repairing a O the Project will be noted in the test of the interves. The prochase order, so will be provided for the lost of the lost jee and then will be with a submitted electronically to the following gradies dress. <u>absorption prinches</u> days.

This foregoing accurately reflects the terms of the symmetry it is the Parties Jacoby agree or erer true and the and engines syntactic he legally brand hereby.

| Asserted h | Township of Wimot these 22 or July these |
|------------------------|---|
| 11000000 | (Developer/Promercel) (Dev) (Month) |
| By Joff | alentura Dun. |
| 89-11-11 | (Prine Name) (9 |
| $t \in \mathbb{R}^{2}$ | zor of Public Works and Engineering |
| | |
| Arbgredi | 1000.50 at 231 at 75 k 2021 |
| By: Lu | ismisary gisteller |
| Thie Dip Six | of, Lends and Receivering sizes of the Cornel River Bleaked Contacil |
| | |

This agreement will capite on Dearster 31, 2021

ίά.



CALCHER AND IS NUMBER OF STREET

July 19,2021

VIA EMAIL

Chelsea Kochany MTE ckochany@mte85.com

Dear Chelsea Kochany,

RE: MCFN Archaeological Review for

Beckdale Sanitary Sewer in Waterloo

Confirmation of Receipt

I am writing in follow up to an email received from Amanda McCall of Detritus dated July 16 2021 sent to the Department of Consultation and Accommodation ("DOCA") representing the Mississaugas of the Credit First Nation ("MCFN").

Outline of MCFN Rights and Territory

In 1792, the Crown and MCFN entered into Between the Lakes, No. 3 (1792) regarding the lands in which your project is situated.

MCFN has formed the Department of Consultation and Accommodation ("DOCA") to represent its interests in consultation and accommodation matters. In this regard, it is DOCA's mandate to ensure that we are directly involved in all planning and development that impacts the integrity of our Territory. DOCA will assess and help alleviate impacts on our rights, land claims, and ways of life by building relationships with governments and private sector proponents. We share a mutual interest in ensuring that projects in the Territory are planned, reviewed, and developed in a manner which ensures healthy communities, ecological protection, and sustainable development for present and future generations in the Territory.

MCFN has a stewardship responsibility over its Territory and asserts that our Aboriginal and treaty rights fundamentally entitle us to preserve our culture and heritage, including



archaeological materials and human burials. Our Territory is the source of our identity as a First Nation and the basis for many cultural activities and spiritual ceremonies. It is home to sacred sites, burial grounds, traditional teachings and meeting places, and sites of profound archaeological and historical significance. We assert that our Aboriginal and treaty rights fundamentally entitle us to preserve our cultural and heritage.

Too much of our cultural objects and the remains of our ancestors have been lost already through development of the most intensely urbanized lands in Canada and we have a strong interest in ensuring that no more of it becomes bulldozed and desecrated.

MCFN Standards and Guidelines for Archaeology

In April 2018, MCFN Chief and Council adopted the MCFN *Standards and Guidelines for Archaeology*, a document aimed to provide guidance to consultant archaeologists, proponents, governments, etc. who are conducting archaeological assessment activities within MCFN's Territory. It sets out, in MCFN's own words, what engagement with our Nation should entail for archaeology as well as technical expectations for fieldwork, in relation to the provincial regulations which were created without our input and feedback. It is important to note that MCFN holds all archaeological resources present within its Territory as of interest to the Nation as part of their cultural patrimony. Resources, regardless of size, frequency, condition, etc., should not be interpreted by non-MCFN representatives in such a way as to remove the requirement for engagement with our Nation.

We are attaching a copy for your reference. We expect compliance with these *Standards and Guidelines* as any fieldwork you will be conducting will have the potential of disturbing MCFN's cultural artifacts or its ancestors' remains.

MCFN Expectations Regarding Ancestors' Remains

MCFN has obligations under Anishinaabe law to protect burials within its Territory and MCFN maintains its right to do so. Our ancestors buried their loved ones in our Territory with the understanding that they would not be disturbed.

We would also like to draw your attention to our expectation that at any time that ancestral remains are encountered during fieldwork, we expect all activities on site to stop and that MCFN be contacted immediately to determine a proper course of action.

Technical Review

In the exercise of its stewardship responsibility, DOCA seeks to work together with project proponents and their archaeological consultants to ensure that archaeological work is done properly and respectfully. DOCA has retained technical advisers with expertise in the field of archaeology. These experts will review the technical aspects and cultural appropriateness of the archaeological assessments and strategies associated with your project. Upon completion of these reviews, MCFN will identify, if





necessary, mitigation measures to address any project impacts upon MCFN rights. For cultural materials and human remains, DOCA may advise that this includes ceremonies required by Anishinaabe law, as well as request adjustments to the proposed fieldwork strategy.

The proponent is expected to pay the costs for MCFN to engage in a technical review of the project. DOCA anticipates at this time that all archaeological review will be undertaken by in-house technical experts, but will advise the proponent if an outside peer-review is required. Please find attached the agreement that covers MCFN's inhouse technical review of the archaeological assessments and strategies associated with your project. Please fill in the additional required information, highlighted in yellow, and return to us a signed copy.

Please note that capacity at DOCA is limited. We maintain the right to review all material that comes to our office as part of our consultation process. If you have specific filing deadlines, please advise us as soon as possible. However, it is MCFN's assertion that part of the process of meaningful engagement is allowing our Nation a reasonable amount of time to review, reflect upon, and respond to reports and recommendations. On average, this process can be accomplished in 4-6 weeks. It is our position that no archaeological assessment – but especially Stage 4 mitigation – should begin until DOCA has completed our review and is in agreement that with the proposed strategy for fieldwork.

Request for Missing Information

In order to complete our project record, we ask that you provide the following information:

- 1. Is an archaeological assessment required for this project? If no, why not?
- 2. Have any archaeological assessments already been completed for this project and/or its study area? If yes, please provide all documentation including reports, supplementary documentation, etc.
- 3. Has the MHSTCI issued a letter of entry into register for some or all of the study area? If yes, please provide all documentation, including letter, communications to and from MHSTCI, etc.
- 4. Is there any archaeological activity (e.g. assessment, excavation, monitoring) that has not yet been completed for the project?
- 5. If the answer to #4 is yes, please provide the following:
 - a. A description of the outstanding archaeological activity/activities.
 - b. Anticipated date of the activity/activities.
 - c. The appropriate contact person overseeing the archaeological activity/activities.



<u>Closing</u>

The review of project-related archaeological assessments is only one part of the consultation process that may be required for your development. Please contact DOCA's Consultation Coordinator, Fawn Sault, if you have any questions about the process.

We ask that you respond with the above requested information and executed agreement within fourteen days following receipt of this letter. We thank you in advance for your attention to our requirements and we look forward to working with you further to shape the planning for development in our Territory.

Sincerely,

Low Tellier

Megan DeVries Archaeological Operations Supervisor megan.devries@mncfn.ca

Attachment(s)

MCFN *Standards and Guidelines for Archaeology* [2018] DOCA Archaeological Review Agreement [2020]





Archaeological Review Agreement between: The Mississaugas of the Credit First Nation ("MCFN") and Township of Wilmot

A - Background

- The purpose of this agreement is to provide the Mississaugas of the Credit First Nation (hereinafter, "MCFN") with capacity assistance to review reports and other materials in connection with all archaeological assessments required for the Baden Sanitary Trunk EA (hereinafter, "the Project") located at in Baden, Ontario, owned by Township of Wilmot, (hereinafter, "the Proponent").
- 2. The Proponent understands that MCFN wishes its designated representatives at the Department of Consultation and Accommodation (hereinafter, "DOCA") to provide timely and meaningful comment on the Project via its established review process.
- 3. The Proponent, or their consultant(s), will therefore provide all reports in draft form to MCFN (via DOCA) for review and comment prior to their submission to other approval or regulatory authorities. The Proponent and their consultant(s) agree to provide reasonable and adequate time for DOCA to complete its review and provide comments on draft reports. DOCA requires a minimum of three weeks to complete any review of materials and provide comments on draft reports.
- 4. For archaeological assessments, the Proponent agrees that their consultant(s) will provide, if applicable, both the Supplementary Documentation and the Indigenous Engagement report alongside the draft archaeological report. The Indigenous Engagement report must contain the consultant's full account of MCFN's participation in and comments on the archaeological assessment.
- 5. For archaeological assessments, the Proponent agrees that no new fieldwork will commence until MCFN has completed its review and has provided comments on the previous Stage of assessment.
- MCFN agrees that MCFN representatives will have appropriate qualifications for the work required – for example, education in environmental and/or archaeological assessments – and experience in bridging Indigenous perspectives with Western approaches, as reasonably determined by MCFN.

B-Fees and Cost Structure

- 7. The Proponent will provide capacity funding for the designated DOCA staff representative in the amount of \$150.00 per hour for all activities relating to review of Project materials. An estimate of costs is provided in Schedule B.
- 8. If MCFN is of the view, that designated DOCA staff are unable to complete a comprehensive technical review of Project materials, the Proponent agrees to pay costs incurred by MCFN to retain an external expert in the appropriate field to be chosen at MCFN's sole discretion. The Parties agree that a review by an external expert will commence following mutual acceptance by both Parties of an estimate of work provided by the expert.

C – Additional Conditions

- 9. All archaeological work in connection with any Project in the Territory will be carried out in accordance with the *Ontario Heritage Act* and its Regulations. The Archaeological work will meet or exceed the Ontario Ministry of Heritage, Sport, Tourism, and Culture Industries (hereinafter, "MHSTCI") standards and guidelines for consultant archaeologists as amended, including the *Terms and Conditions for Archaeological Licences, Standards and Guidelines for Consultant Archaeologists (2011)* and the Draft *Engaging Aboriginal Communities in Archaeology Technical Bulletin (2011),* (hereinafter collectively, "MHSTCI Standards 2011"), and any subsequent or updated guidelines or bulletins from MHTSCI.
- 10. The Proponent agrees that all archaeological work conducted for the Project will comply with the MCFN Standards and Guidelines for Archaeology (published April 2, 2018), (hereinafter, "MCFN Standards") as long as the MCFN Standards do not fall below MHSTCI Standards 2011. The MHSTCI Standards 2011 will be paramount in the event of a direct conflict between MCFN Standards and the MHSTCI Standards 2011.
- 11. The Proponent shall make best efforts to avoid and protect archaeological sites, artifacts, and/or features. The Parties agree that the preferred option for human remains that may be of Aboriginal ancestry is that the human remains stay where they are found with appropriate protections.
- 12. If archaeological resources are encountered at any time during construction or other Project-related activity, all excavation or other activity that could disturb the site shall

immediately cease, and the Proponent shall immediately notify MCFN's duly appointed Archaeological Operations Supervisor or designate. The Parties shall work collaboratively to minimize impacts and ensure respectful treatment of any archaeological resources in accordance with the practices and values of MCFN as identified by MCFN.

- 13. If human remains are encountered at any time during construction or other Project-related activity, the following steps shall be taken:
 - a. All excavation or other activity that could disturb the site shall immediately cease, and the area shall be secured in a manner which protects the site location and prevents public access and trespass; and
 - b. In addition to any notifications required under the *Funeral, Burial and Cremation* Services Act, 2002, SO 2002, C 33, the Proponent shall immediately contact MCFN's duly appointed Archaeological Operations Supervisor or designate; and
 - c. MCFN shall be permitted to conduct any ceremonies on site in relation to the human remains that may be of Aboriginal ancestry; and
 - d. MCFN shall be consulted about all steps in the investigation and any decisions or agreements to be made regarding human remains that may be of Aboriginal ancestry.
- 14. Nothing in this Agreement shall be interpreted or implemented so as to derogate or abrogate from any MCFN Aboriginal or Treaty right or claim, or to indicate consent to the Project.

D - Method of Payment

15. The Parties agree that the Proponent will pay the capacity funding as agreed to above by cheque or bank transfer and upon receipt of an invoice from MCFN. All invoices will be addressed directly to the Proponent, the Project will be noted in the text of each invoice, and all invoices will be prepared as per MCFN-DOCA's standard invoicing format. Invoices should be submitted electronically to the following address:

Email address: droga@mte85.com Attention: Dot Roga on behalf of Township of Wilmot MTE Consultants Inc. 520 Bingemens Centre Drive Kitchener ON N2B 3X9 (519) 743-6500 ext 1269 16. All payment should be made to the MCFN Department of Consultation and Accommodation to the following address. For additional information, please call the office at 905-768-4260.

> Email address: DOCA.Admin@mncfn.ca Attention: MCFN-DOCA 4065 Highway 6 Hagersville, Ontario N0A 1H0

17. After thirty [30] days, a 5% monthly compounded interest rate will be charged on outstanding invoices. After six [6] months of non-payment, a 20% monthly compounded interest rate will be charged on outstanding invoices.

F – Disclaimer

- 18. The Parties agree that the capacity funding payments for the FLRs will be used only for the purposes described in this Agreement and will not be paid for the improper personal gain of any individual or for any other purpose that might violate any Canadian anti-corruption law.
- 19. This agreement may be executed in counterparts, including via electronic signature.
- 20. This agreement is legally binding on MCFN and the Proponent..
- 21. This agreement is effective as of the date of execution by the Parties. In the event Projectrelated activities continue beyond, April 1, 2023, the Parties agree to conclude a new agreement to address MCFN involvement in the Project after April 1, 2023.

[The remainder of this page is intentionally left blank.]

Authorized Signatory on behalf of The Proponent

[Jeff Molenhuis Director Public Works and Engineering Township of Wilmot Authorized Signatory on behalf of Mississaugas of the Credit First Nation

Mark LaForme Director Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

Witness

Witness

| [printed name of witness] |
|-----------------------------|
| [<mark>job title</mark>] |
| [<mark>department</mark>] |
| [name of the proponent] |

Adam LaForme Archaeological Operations Supervisor Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

Schedule A



Municipalities within Mississaugus of the Grecil Treaty Lands and Territory

Schedule B

Approx. Quote for Technical Review (Reference Only)

For review of materials and communications associated with Stage 1 AAs.

| | Number | Rate | 2 | Total |
|---------------------|--------|--------------|----|--------|
| review hours | 4.0 | \$ 150.00 | \$ | 600.00 |
| contingency (@ 20%) | | | \$ | 120.00 |
| Total | | | \$ | 720.00 |

For review of materials and communications associated with Stage 2 AAs.

| | Number | Rate | 2 | Total |
|---------------------|--------|--------------|----|--------|
| review hours | 4.0 | \$ 150.00 | \$ | 600.00 |
| contingency (@ 20%) | | | \$ | 120.00 |
| Total | | | \$ | 720.00 |

For review of materials and communications associated with Stage 3 AAs.

| | Number | Rat | е | Total |
|---------------------|--------|--------------|----|----------|
| review hours | 8.0 | \$ 150.00 | \$ | 1,200.00 |
| contingency (@ 20%) | | | \$ | 240.00 |
| Total | | | \$ | 1,440.00 |

For review of materials and communications associated with Stage 4 AAs.

| | Number | Rate | Total |
|---------------------|--------|--------------|----------------|
| review hours | 8.0 | \$ 150.00 | \$ 1,200.00 |
| contingency (@ 20%) | | | \$ 240.00 |
| Total | | | \$ 1,440.00 |


DALO INSTANUCIS NATIONAL A SUBJECT A SUBJECT SUBJECT SUB-

July 19,2021

VIA EMAIL

Chelsea Kochany MTE ckochany@mte85.com

Dear Chelsea Kochany,

RE: MCFN FLR Participation for

Beckdale Sanitary Sewer in Waterloo

Confirmation of Receipt

I am writing in follow up to an email received from Amanda McCall of Detritus dated July 16 2021 sent to the Department of Consultation and Accommodation ("DOCA") representing the Mississaugas of the Credit First Nation ("MCFN").

Outline of MCFN Rights and Territory

In 1792, the Crown and MCFN entered into Between the Lakes, No. 3 (1792) regarding the lands in which your project is situated.

MCFN has formed the Department of Consultation and Accommodation ("DOCA") to represent its interests in consultation and accommodation matters. In this regard, it is DOCA's mandate to ensure that we are directly involved in all planning and development that impacts the integrity of our Territory. DOCA will assess and help alleviate impacts on our rights, land claims, and ways of life by building relationships with governments and private sector proponents. We share a mutual interest in ensuring that projects in the Territory are planned, reviewed, and developed in a manner which ensures healthy communities, ecological protection, and sustainable development for present and future generations in the Territory.

One of the ways we require proponents to engage with us is in providing transparency during the environmental survey and archaeological assessment process. The best way



to accomplish this is by having Field Liaison Representatives ("FLRs") on location while fieldwork is occurring, who can ensure that the Nation's special interests and concerns are respected and considered during fieldwork. The cultural and natural resources in question are part of MCFN's territory and heritage and it is our responsibility to ensure their protection, on behalf of the Nation. MCFN's stewardship of its territory extends through the life of any development project and beyond.

It is our expectation that no project-related fieldwork will take place without the participation of our FLRs. MCFN considers it disrespectful of our rights as Indigenous peoples if our natural and cultural heritage is interfered with without our involvement.

FLR Participation

DOCA deploys FLRs to be boots on the ground so that fieldwork by a proponent and their consultants/contractors is carried out with appropriate care, thoroughness, and respect. In the context of MCFN's Territory, where so much natural and cultural heritage has already been lost or destroyed, MCFN's monitoring of fieldwork is of utmost importance to ensure that the trail of desecration stops. FLRs are deployed to observe fieldwork, provide cultural advice, act as a direct link back to DOCA and MCFN, and assist with compliance.

FLRs are MCFN band members who have received training in environmental and archaeological assessments, traditional medicine identification and use, Anishinaabe burial practices, and more throughout their employment with DOCA.

DOCA requires, at minimum, FLR participation during the following project-related studies and/or activities:

- ecological and natural heritage technical studies
- archaeological assessments (Stages 2 through 4) and site visits
- monitoring of activities within 50m of areas of special concern (e.g. waterways and wetlands, archaeological sites, species at risk)
- post-construction remediation activities and follow-up impact monitoring

Agreement for FLR Participation

The cost for the participation of our FLRs is covered by the proponent, not the consultant, whom we view as having the ultimate responsibility to consult with, and accommodate, the Nation. Therefore, please find attached the agreement that covers MCFN's participation in the upcoming fieldwork. The costs associated with this involvement reflect a number of expenses not visible at first glance: payment for the FLRs themselves, operational costs for DOCA, and efforts to engage the community to garner feedback on development projects. If you could please fill in the additional required information, highlighted in yellow, and return to us a signed copy so that we may arrange for FLR participation on your project, that would be greatly appreciated.



Once a signed agreement is in place, DOCA generally arranges scheduling and other related matters directly with the consultant conducting the fieldwork, unless you prefer otherwise.

Please note that MCFN requires two of its FLRs to be on location whenever fieldwork is taking place within its territory. The reason for this is so that FLRs can provide support and security for each other in the field. This has become a requirement in light of uncommon, but unfortunate, occurrences when FLRs have felt pressured or intimidated from external persons while at work locations. We ask that you would respect this request.

Request for Missing Information

In order to complete our project record, we ask that you provide the following information:

- 1. Please provide a list of all completed technical studies for the project, their date of completion, and the contact information of the consultant who completed each study.
- 2. Please provide a list of all incomplete and/or upcoming technical studies for the project, the anticipated date of fieldwork for each, and the contact information for the consultant who will complete them.
- 3. Are there any short-term and/or long-term avoidance and protection strategies currently in place for the natural and/or cultural resources in the study area for this project? If yes, what are they?

<u>Closing</u>

The participation of FLRs in project fieldwork is only one part of the consultation process that may be required for your development. Please contact DOCA's Consultation Coordinator, Fawn Sault, if you have any questions about the process.

We ask that you respond with the above requested information and executed agreement within fourteen days following receipt of this letter. We thank you in advance for your attention to our requirements and we look forward to working with you further to shape the planning for development in our Territory.

Sincerely,

Tallier

Megan DeVries Archaeological Operations Supervisor <u>megan.devries@mncfn.ca</u>

DEPARTMENT OF EDNSIGTATION AND ACCOMMODATION WEINGKOGENET IN COVER Pass Patient New Investigation (Internet Pass Patient)



Attachment(s)

MCFN *Standards and Guidelines for Archaeology* [2018] FLR Participation Agreement [2020]



DEPARTMENT OF CONSIGNATION AND ACCOMMODATION Visual sectors of the former line. Name Consider, Sal Name Statistical Constants



A - Background

- The purpose of this agreement is to provide the Mississaugas of the Credit First Nation (hereinafter, "MCFN") with capacity assistance to its Field Liaison Representatives (hereinafter, "FLRs") in connection with all environmental and/or archaeological assessments required for the Baden Sanitary Trunk EA (hereinafter, "the Project") located in Baden, Ontario, owned by Township of Wilmot, (hereinafter, "the Proponent").
- 2. The Proponent understands that MCFN wishes to send its FLRs to participate in and monitor the assessments associated with the Project, and that the FLRs' mandate will be to ensure that MCFN's perspectives and priorities are considered and to enable MCFN to provide timely and meaningful comment on the Project.
- 3. All archaeological work in connection with any Project in the Territory will be carried out in accordance with the Ontario Heritage Act and its Regulations. The archaeological work will meet or exceed the Ontario Ministry of Heritage, Sport, Tourism, and Culture Industries (hereinafter, "MHSTCI") standards and guidelines for consultant archaeologists as amended, including the Terms and Conditions for Archaeological Licences, Standards and Guidelines for Consultant Archaeologists (2011) and the Draft Engaging Aboriginal Communities in Archaeology Technical Bulletin (2011), (hereinafter collectively, "MHSTCI Standards 2011"), and any subsequent or updated guidelines or bulletins from MHTSCI.
- 4. The Proponent agrees that all archaeological work conducted for the Project will comply with the MCFN *Standards and Guidelines for Archaeology* (published April 2, 2018), (hereinafter, "MCFN Standards") as long as the MCFN Standards do not fall below MHSTCI Standards 2011. The MHSTCI Standards 2011 will be paramount in the event of a direct conflict between MCFN Standards and the MHSTCI Standards 2011.
- 5. Nothing in this Agreement shall be interpreted or implemented so as to derogate or abrogate from any MCFN Aboriginal or Treaty right or claim, or to indicate MCFN's consent to the Project.

B-Fees and Cost Structure

- 6. The Proponent will provide capacity funding for each FLR in the amount of \$85.00 per hour for all activities relating to the Project. Activities relating to the Project include, but are not limited to:
 - a. Time spent on site monitoring assessment or predetermined construction-related activities;
 - b. Time spent completing data or artifact processing, identification, analysis, and interpretation activities alongside their consultant(s);
 - c. Actual travel time at the beginning of, during, and/or end of each day;
 - d. Time completing daily notes relating to the Project;
 - e. Time spent on standby at the request of the Proponent or their consultant(s); and
 - f. Time completing mandatory training at the request of the Proponent or their consultant(s).
- 7. The Proponent will pay a supervisory fee of 3.5%, based on the number of hours charged to the Proponent, to provide MCFN with the capacity to facilitate in-field technical support for the FLRs via the Field Archaeologist.
- The Proponent will reimburse the FLRs for reasonable mileage and meals in accordance with current Federal Canada Treasury Board guidelines, over and above the hourly rate [see Schedule B]. Mileage rates are determined using the MCFN Department of Consultation and Accommodation as the place of departure.
- 9. The Proponent will provide capacity funding for each FLR in the amount of \$125.00 per hour for any work exceeding eight hours per day and/or forty hours per week. The above noted mileage and meal allowance remains in effect.
- 10. The Proponent will provide capacity funding for each FLR in the amount of \$125.00 per hour for any work occurring on the following holidays: New Year's Day, Family Day, Good Friday, Victoria Day, Indigenous Solidarity Day (June 21), Canada Day, Civic Holiday, Labour Day, the National Day for Truth and Reconciliation (September 30), Thanksgiving Day, Remembrance Day, Christmas Day, and Boxing Day. The above noted mileage and meal allowance rates remain in effect.
- 11. The Proponent agrees that the FLRs will be paid for a minimum of three hours, plus actual travel time, mileage, and meal allowance rates as noted above, on any day when

work is cancelled by the Proponent or their consultant(s) while FLRs are en route to the work site or after the FLRs have already arrived.

- 12. If its use is deemed necessary by both Parties, the Proponent agrees to reimburse the FLRs for their use of the 407ETR upon receipt of a copy of the bill. This agreement will be provided in writing to MCFN's Field Coordinator.
- 13. If deemed reasonable by both Parties, the Proponent agrees to cover the cost of overnight accommodation for FLRs participating in environmental and/or archaeological fieldwork at locations which would otherwise require more than 90 minutes of travel time at both the beginning and end of the work day, using the MCFN Department of Consultation and Accommodation as the place of departure. An additional Incidental Allowance fee is required for any work which requires overnight accommodations, as set out in Schedule B. This agreement will be provided in writing to MCFN's Field Coordinator.

<u>C – Additional Conditions</u>

- 14. The parties acknowledge that the Project, in whole or in part, takes place within MCFN Territory and agree that the Proponent shall provide capacity funding for FLR participation on the Project for the duration of the Project.
- 15. The Proponent agrees that two FLRs shall be on location whenever Project-related activities are taking place within its Territory, as set out in Schedule A.
- 16. Furthermore, additional FLRs are required if the number of field personnel utilized by the consultant exceeds fourteen (14) individuals and the Proponent agrees to provide capacity funding for additional FLRs as required. MCFN requires one additional FLR per five additional field crew, as outlined in the chart below:

| Number of Field Personnel | Number of FLRs Required |
|---------------------------|-------------------------|
| 1 to 14 | 2 |
| 15 to 19 | 3 |
| 20 to 24 | 4 |
| 25 to 29 | 5 |
| 30 to 34 | 6 |
| 35 to 39 | 7 |
| 40+ | 8+ |

- 17. The Parties acknowledge that the FLRs' time and travel will be recorded and verified using the ClockShark Time Tracking Software System and that invoicing will be prepared using these records, not those of a third party.
- 18. If archaeological resources are encountered at any time during construction or other Project-related activity, all excavation or other activity that could disturb the site shall immediately cease, and the Proponent shall immediately notify MCFN's Archaeological Operations Supervisor or designate. The Parties shall work collaboratively to minimize impacts and ensure respectful treatment of any archaeological resources in accordance with the practices and values of MCFN as identified by MCFN.
- 19. If human remains are encountered at any time during construction or other Project-related activity, the following steps shall be taken:
 - a. All excavation or other activity that could disturb the site shall immediately cease, and the area shall be secured in a manner which protects the site location and prevents public access and trespass; and
 - In addition to any notifications required under the *Funeral, Burial and Cremation* Services Act, 2002, SO 2002, C 33, the Proponent shall immediately contact MCFN's duly appointed Archaeological Operations Supervisor or designate; and
 - c. MCFN shall be permitted to conduct any ceremonies on site in relation to the human remains that may be of Aboriginal ancestry ("Ancestral Remains"); and
 - d. MCFN shall be consulted about all steps in the investigation and any decisions or agreements to be made regarding Ancestral Remains.

D - Coordination of the FLRs

- 20. The Parties agree that the FLRs will follow the reasonable instructions of the Proponent and their consultant firm(s) conducting the environmental and/or archaeological work concerning safety practices, and that the FLRs will attend "tailgate" safety meetings if requested.
- 21. The contact person for activities relating to the environmental assessment portion of the Project is [name of contact person #1] from [name of consultant]. Contact information for this person is as follows:

[insert contact information here]

22. The contact person for activities relating to the archaeological assessment portion of the Project is Garth Grimes from Detritus Consulting Ltd. Contact information for this person is as follows:

Detritus Consulting Ltd. www.detcon.net (519) 744-7018

23. The Parties agree that the contact person for the consultant firm(s) will coordinate site meeting locations and times through MCFN's duly appointed Field Coordinator. Contact information for the Field Coordinator is as follows:

Mariah Sault Telephone: 905-768-4260 Cell: 905-870-2918 Email: field.coordinator@mncfn.ca

<u>E - Status of the FLRs</u>

- 24. The FLRs selected by MCFN have appropriate qualifications for the work required for example, training in environmental and/or archaeological monitoring and experience in bridging Indigenous perspectives with Western approaches, as reasonably determined by MCFN.
- 25. The Parties agree that the FLRs are not employees, contractors, or sub-contractors of the Proponent or their consultant(s) and that the FLRs will be responsible for their own personal protective equipment, such as hard hats, safety boots, and safety vests, unless specific or otherwise unique personal protective equipment is required, which will therefore be provided or reimbursed by the Proponent.
- 26. FLRs take direction from MCFN. MCFN pays Workplace Safety and Insurance Board ("WSIB") contributions in respect of the FLRs and will, at its own expense, maintain for the term of this agreement a comprehensive general liability ("CGL") policy or policies with a limit of at least \$5 million. MCFN will, at its own expense, maintain a policy for automobile insurance with a combined single limit of not less that \$2 million.
- 27. MCFN shall provide the Proponent with evidence of such insurance above in the form of a memorandum of insurance issued by the insurer, upon request.
- 28. MCFN agrees that FLRs will perform their activities safely, in a good and competent manner, in compliance with all applicable laws, regulations, and guidelines.

- 29. MCFN expects that the Proponent will comply with the *Occupational Health and Safety Act*, R.S.O. 1990, C. 0.1, the Ontario *Human Rights Code*, R. S. O. 1990, c. H.19, and maintain a safe, harassment-free work environment.
- 30. The Proponent is responsible for negligence or other failure to maintain a safe and harassment-free work environment. To the extent that the Proponent is responsible for negligence or other failure to maintain a safe and harassment-free work environment, the Proponent is liable and shall indemnify MCFN claims or demands related to injury, accident, discrimination, or harassment by the Proponent's employees, agents, consultants, or other parties under the control or direction of the Proponent.

F - Method of Payment

31. The Parties agree that the Proponent will pay the capacity funding as agreed to above by cheque or bank transfer and upon receipt of an invoice from MCFN. All invoices will be addressed directly to the Proponent, the Project will be noted in the text of each invoice, and all invoices will be prepared as per MCFN-DOCA's standard invoicing format. Invoices should be submitted electronically to the following address:

Email address: droga@mte85.com Attention: Dot Roga on behalf of Township of Wilmot MTE Consultants Inc. 520 Bingemens Centre Drive Kitchener ON N2B 3X9 (519) 743-6500 ext 1269

32. All payment should be made to the MCFN Department of Consultation and Accommodation to the following address. For additional information, please call the office at 905-768-4260.

Email address: DOCA.Admin@mncfn.ca Attention: MCFN-DOCA 4065 Highway 6 Hagersville, Ontario N0A 1H0

33. After thirty [30] days, a 5% monthly compounded interest rate will be charged on outstanding invoices. After six [6] months of non-payment, a 20% monthly compounded interest rate will be charged on outstanding invoices.

<u>G – Disclaimer</u>

- 34. The Parties agree that the capacity funding payments for the FLRs will be used only for the purposes described in this Agreement and will not be paid for the improper personal gain of any individual or for any other purpose that might violate any Canadian anti-corruption law.
- 35. This agreement may be executed in counterparts, including via electronic signature.
- 36. This agreement is legally binding on MCFN and the Proponent. This agreement is signed by authorized representatives of the Parties on the date set out in this agreement below.

This agreement is effective as of the date of execution by the Parties. In the event Project-related activities continue beyond, April 1, 2023, the Parties agree to conclude a new agreement to address MCFN involvement in the Project after April 1, 2023.

[The remainder of this page is intentionally left blank.]

Authorized Signatory on behalf of The Proponent

Jeff Molenhuis Director Public Works and Engineering Township of Wilmot Authorized Signatory on behalf of Mississaugas of the Credit First Nation

Mark LaForme Director Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

Witness

Witness

[printed name of witness] [job title] [department] [name of the proponent] Adam LaForme Archaeological Operations Supervisor Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

Schedule A



Municipalities within Mississaugus of the Grecil Treaty Lands and Territory

Schedule B

| MISSISSING | NAS OF THE | CREDET FORST NO | ADON MELASE & ALLOWANCE CHART |
|-------------------------|------------|-------------------|--|
| | EFFEC | THEAPPRILL A | 122 to MARCH 32, 2023 |
| MILEAGE | | \$-0.575 per KM | |
| | #Rec/hey | ea rije oner de c | Adment/we/Ad100/x238/1658/en |
| MEAL ALLOWANCE | | | not eligible if: |
| 082A87A37 | | 22.98 | Ruse after Bare gel horse before 7.30 av- |
| 10NCH | 38 | 22.23 | leave after 1:13pm, get horns before 12:15pm |
| OINNER: | 15 | 59.50 | ingve after 7pm, get horse before 6:30pm |
| INCIDENTALS | 15 | 1750 | ing stability of the second state of the secon |
| V 101 - 100 - 100 - 100 | attau/hey | south store and | Manutaer/0202/v200/1685/jmp |
| NIGHTIME! | 5 | 54.42 | *Applies only to nightfine surveys that would not otherwise trigger clanes or breakfort |

Field Liaison Representative Participation Agreement between: The Mississaugas of the Credit First Nation and Township of Wilmot

A - Background

- The purpose of this agreement is to provide the Mississaugas of the Credit First Nation (hereinafter, "MCFN") with capacity assistance to its Field Liaison Representatives (hereinafter, "FLRs") in connection with all environmental and/or archaeological assessments required for the Baden Sanitary Trunk EA (hereinafter, "the Project") located in Baden, Ontario, owned by Township of Wilmot, (hereinafter, "the Proponent").
- 2. The Proponent understands that MCFN wishes to send its FLRs to participate in and monitor the assessments associated with the Project, and that the FLRs' mandate will be to ensure that MCFN's perspectives and priorities are considered and to enable MCFN to provide timely and meaningful comment on the Project.
- 3. All archaeological work in connection with any Project in the Territory will be carried out in accordance with the *Ontario Heritage Act* and its Regulations. The archaeological work will meet or exceed the Ontario Ministry of Heritage, Sport, Tourism, and Culture Industries (hereinafter, "MHSTCI") standards and guidelines for consultant archaeologists as amended, including the *Terms and Conditions for Archaeological Licences, Standards and Guidelines for Consultant Archaeologists (2011)* and the Draft *Engaging Aboriginal Communities in Archaeology Technical Bulletin (2011),* (hereinafter collectively, "MHSTCI Standards 2011").
- 4. The Proponent agrees that all archaeological work conducted for the Project will comply with the MCFN *Standards and Guidelines for Archaeology* (published April 2, 2018), (hereinafter, "MCFN Standards") as long as the MCFN Standards do not fall below MHSTCI Standards 2011. The MHSTCI Standards 2011 will be paramount in the event of a direct conflict between MCFN Standards and the MHSTCI Standards 2011.
- 5. Nothing in this Agreement shall be interpreted or implemented so as to derogate or abrogate from any MCFN Aboriginal or Treaty right or claim, or to indicate consent to the Project.

B – Fees and Cost Structure

- 6. The Proponent will provide capacity funding for each FLR in the amount of \$85.00 per hour for all activities relating to the Project. Activities relating to the Project include, but are not limited to:
 - a. Time spent on site monitoring assessment or predetermined construction-related activities;
 - b. Time spent completing data or artifact processing, identification, analysis, and interpretation activities alongside their consultant(s);
 - c. Actual travel time at the beginning of, during, and/or end of each day;
 - d. Time completing daily notes relating to the Project;
 - e. Time spent on standby at the request of the Proponent or their consultant(s); and
 - f. Time completing mandatory training at the request of the Proponent or their consultant(s).
- 7. The Proponent will pay a supervisory fee of 3.5%, based on the number of hours charged to the Proponent, to provide MCFN with the capacity to facilitate in-field technical support for the FLRs via the Field Archaeologist.
- The Proponent will reimburse the FLRs for reasonable mileage and meals in accordance with current Federal Canada Treasury Board guidelines, over and above the hourly rate [see Schedule B]. Mileage rates are determined using the MCFN Department of Consultation and Accommodation as the place of departure.
- 9. The Proponent will provide capacity funding for each FLR in the amount of \$125.00 per hour for any work exceeding eight hours per day and/or forty hours per week. The above noted mileage and meal allowance remains in effect.
- 10. The Proponent will provide capacity funding for each FLR in the amount of \$125.00 per hour for any work occurring on the following holidays: New Year's Day, Family Day, Good Friday, Victoria Day, Indigenous Solidarity Day (June 21), Canada Day, Civic Holiday, Labour Day, Thanksgiving Day, Remembrance Day, Christmas Day, and Boxing Day. The above noted mileage and meal allowance rates remain in effect.
- 11. The Proponent agrees that the FLRs will be paid for a minimum of three hours, plus actual travel time, mileage, and meal allowance rates as noted above, on any day when work is cancelled by the Proponent or their consultant(s) while FLRs are en route to the work site or after the FLRs have already arrived.

- 12. If its use is deemed necessary by both Parties, the Proponent agrees to reimburse the FLRs for their use of the 407ETR upon receipt of a copy of the bill. This agreement will be provided in writing to MCFN's Field Coordinator.
- 13. If deemed reasonable by both Parties, the Proponent agrees to cover the cost of overnight accommodation for FLRs participating in environmental and/or archaeological fieldwork at locations which would otherwise require more than 90 minutes of travel time at both the beginning and end of the work day, as determined using the MCFN Department of Consultation and Accommodation as the place of departure. An additional Incidental Allowance fee is required for any work which requires overnight accommodations, as set out in Schedule B. This agreement will be provided in writing to MCFN's Field Coordinator.

<u>C – Additional Conditions</u>

- 14. The parties acknowledge that the Project, in whole or in part, takes place within MCFN Territory and agree that the Proponent shall provide capacity funding for FLR participation on the Project for the duration of the Project.
- 15. The Proponent agrees that two FLRs shall be on location whenever Project-related activities are taking place within its Territory, as set out in Schedule A.
- 16. Furthermore, additional FLRs are required if the number of field personnel utilized by the consultant exceeds fourteen (14) individuals and the Proponent agrees to provide capacity funding for additional FLRs as required. MCFN requires one additional FLR per five additional field crew, as outlined in the chart below:

| Number of Field Personnel | Number of FLRs Required |
|---------------------------|-------------------------|
| 1 to 14 | 2 |
| 15 to 19 | 3 |
| 20 to 24 | 4 |
| 25 to 29 | 5 |
| 30 to 34 | 6 |
| 35 to 39 | 7 |
| 40+ | 8+ |

17. The Parties acknowledge that the FLRs time and travel will be recorded and verified using the ClockShark Time Tracking Software System and that invoicing will be prepared using these records, not those of a third party.

- 18. If archaeological resources are encountered at any time during construction or other Project-related activity, all excavation or other activity that could disturb the site shall immediately cease, and the Proponent shall immediately notify MCFN's Archaeological Operations Supervisor or designate. The Parties shall work collaboratively to minimize impacts and ensure respectful treatment of any archaeological resources in accordance with the practices and values of MCFN as identified by MCFN.
- 19. If human remains are encountered at any time during construction or other Project-related activity, the following steps shall be taken:
 - a. All excavation or other activity that could disturb the site shall immediately cease, and the area shall be secured in a manner which protects the site location and prevents public access and trespass; and
 - In addition to any notifications required under the *Funeral, Burial and Cremation* Services Act, 2002, SO 2002, C 33, the Proponent shall immediately contact MCFN's duly appointed Archaeological Operations Supervisor or designate; and
 - c. MCFN shall be permitted to conduct any ceremonies on site in relation to the human remains that may be of Aboriginal ancestry ("Ancestral Remains"); and
 - d. MCFN shall be consulted about all steps in the investigation and any decisions or agreements to be made regarding Ancestral Remains.

D - Coordination of the FLRs

- 20. The Parties agree that the FLRs will follow the reasonable instructions of the Proponent and their consultant firm(s) conducting the environmental and/or archaeological work concerning safety practices, and that the FLRs will attend "tailgate" safety meetings if requested.
- 21. The contact person for activities relating to the environmental assessment portion of the Project is [name of contact person #1] from [name of consultant]. Contact information for this person is as follows:

[insert contact information here]

22. The contact person for activities relating to the archaeological assessment portion of the Project is Garth Grimes from Detritus Consulting Ltd. Contact information for this person is as follows:

Detritus Consulting Ltd. www.detcon.net (519) 744-7018 23. The Parties agree that the contact person for the consultant firm(s) will coordinate site meeting locations and times through MCFN's duly appointed Field Coordinator. Contact information for the Field Coordinator is as follows:

Joelle Williams Telephone: 905-768-4260 Cell: 905-870-2918 Email: joelle.williams@mncfn.ca

E - Status of the FLRs

- 24. The FLRs selected by MCFN have appropriate qualifications for the work required for example, training in environmental and/or archaeological monitoring – and experience in bridging Indigenous perspectives with Western approaches, as reasonably determined by MCFN.
- 25. The Parties agree that the FLRs are not employees, contractors, or sub-contractors of the Proponent or their consultant(s) and that the FLRs will be responsible for their own personal protective equipment, such as hard hats, safety boots, and safety vests, unless specific or otherwise unique personal protective equipment is required, which will therefore be provided or reimbursed by the Proponent.
- 26. FLRs take direction from MCFN. MCFN pays Workplace Safety and Insurance Board ("WSIB") contributions in respect of the FLRs and will, at its own expense, maintain for the term of this agreement a comprehensive general liability ("CGL") policy or policies with a limit of at least \$1 million and shall provide the Proponent with evidence of such insurance, upon request. MCFN agrees that FLRs will perform their activities safely, in a good and competent manner, in compliance with all applicable laws, regulations, and guidelines.
- 27. MCFN expects that the Proponent will comply with the *Occupational Health and Safety Act*, R.S.O. 1990, C. 0.1, the Ontario *Human Rights Code*, R. S. O. 1990, c. H.19, and maintain a safe, harassment-free work environment.
- 28. The Proponent is responsible for negligence or other failure to maintain a safe and harassment-free work environment. To the extent that the Proponent is responsible for negligence or other failure to maintain a safe and harassment-free work environment, the Proponent is liable and shall indemnify MCFN claims or demands related to injury, accident, discrimination, or harassment by the Proponent's employees, agents, consultants, or other parties under the control or direction of the Proponent.

F - Method of Payment

29. The Parties agree that the Proponent will pay the capacity funding as agreed to above by cheque or bank transfer and upon receipt of an invoice from MCFN. All invoices will be addressed directly to the Proponent, the Project will be noted in the text of each invoice, and all invoices will be prepared as per MCFN-DOCA's standard invoicing format. Invoices should be submitted electronically to the following address:

Email address: droga@mte85.com Attention: Dot Roga on behalf of Township of Wilmot MTE Consultants Inc. 520 Bingemens Centre Drive Kitchener ON N2B 3X9 (519) 743-6500 ext 1269

30. All payment should be made to the MCFN Department of Consultation and Accommodation to the following address. For additional information, please call the office at 905-768-4260.

> Email address: <u>nicole.laforme-hess@mncfn.ca</u> Attention: MCFN-DOCA 4065 Highway 6 Hagersville, Ontario N0A 1H0

31. After thirty [30] days, a 5% monthly compounded interest rate will be charged on outstanding invoices. After six [6] months of non-payment, a 20% monthly compounded interest rate will be charged on outstanding invoices.

<u>G – Disclaimer</u>

- 32. The Parties agree that the capacity funding payments for the FLRs will be used only for the purposes described in this Agreement and will not be paid for the improper personal gain of any individual or for any other purpose that might violate any Canadian anti-corruption law.
- 33. This agreement may be executed in counterparts.
- 34. This agreement is legally binding on MCFN and the Proponent. This agreement is signed by authorized representatives of the Parties on the date set out in this agreement below.

35. The term of this agreement expires on April 1, 2022. In the event that Project-related activities requiring FLR participation continue past this termination date, a new agreement will be executed between Parties.

[The remainder of this page is intentionally left blank.]

Signed this _____ day of _____, 2021,



Authorized Signatory on behalf of The Proponent

Jeff Molenhuis Director Public Works and Engineering Township of Wilmot



Authorized Signatory on behalf of Mississaugas of the Credit First Nation

per

Mark LaForme Director Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

2 Elver

Witness

Witness

[printed name of witness] [job title] [department] [name of the proponent]

Megan DeVries Archaeological Operations Supervisor Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation





Municipalities within Mastessugas of the Greell Treaty Lands and Territory

Schedule B

| MISSISSAL | NAS DE TRE | CREAT HAS INC | ATCM MILEAGE & ATLCWARCE CHART |
|---------------|------------------|------------------|---|
| | 6656 | тия арки 1, 202 | 2 to MARCH 81, 2022 |
| MULTING | | 5 Printper KM | |
| | http://w | ww.njo-chin.goza | yoli edil eq/d10/v206/s050/em |
| MEALAILOW/MCE | | | non eligible it: |
| DREAKTAST | 5 | 21.10 | lesse aller Ban; gel hone beiten 7:30an |
| LUNCH | \$ | 21.35 | leave after 1:15pm; get home before 12.15pm |
| O IN MER | 5 | 52,440 | leave after April get home before 5.40 pm |
| INCIDENTALS. | 5 | 12.50 | not sitying overnight |
| | <u>https://w</u> | est for the form | ienen 1 we/d100/2208/5050/en |
| NEUTOMP* | 5 | 52.20 | "applies only to a girth me sameys that would not |
| | | | otherwise trigger dinner or breaking). |

Archaeological Review Agreement between: The Mississaugas of the Credit First Nation ("MCFN") and Township of Wilmot

A - Background

- The purpose of this agreement is to provide the Mississaugas of the Credit First Nation (hereinafter, "MCFN") with capacity assistance to review reports and other materials in connection with all archaeological assessments required for the Baden Sanitary Trunk EA (hereinafter, "the Project") located at in Baden, Ontario, owned by Township of Wilmot, (hereinafter, "the Proponent").
- 2. The Proponent understands that MCFN wishes its designated representatives at the Department of Consultation and Accommodation (hereinafter, "DOCA") to provide timely and meaningful comment on the Project via its established review process.
- 3. The Proponent, or their consultant(s), will therefore provide all reports in draft form to MCFN (via DOCA) for review and comment prior to their submission to other approval or regulatory authorities. The Proponent and their consultant(s) agree to provide reasonable and adequate time for MCFN to complete its review and provide comments on draft reports. MCFN is unable to review of any material in less than one week.
- 4. For archaeological assessments, the Proponent agrees that their consultant(s) will provide, if applicable, both the Supplementary Documentation and the Indigenous Engagement report alongside the draft archaeological report. The Indigenous Engagement report must contain the consultant's full account of MCFN's participation in and comments on the archaeological assessment.
- 5. For archaeological assessments, the Proponent agrees that no new fieldwork will commence until MCFN has completed its review and has provided comments on the previous Stage of assessment.
- MCFN agrees that MCFN representatives will have appropriate qualifications for the work required – for example, education in environmental and/or archaeological assessments – and experience in bridging Indigenous perspectives with Western approaches, as reasonably determined by MCFN.

B-Fees and Cost Structure

- 7. The Proponent will provide capacity funding for the designated DOCA staff representative in the amount of \$150.00 per hour for all activities relating to review of Project materials. An estimate of costs is provided in Schedule B.
- 8. If MCFN is of the view, that designated DOCA staff are unable to complete a comprehensive technical review of Project materials, the Proponent agrees to pay costs incurred by MCFN to retain an external expert in the appropriate field to be chosen at MCFN's sole discretion. The Parties agree that a review by an external expert will commence following mutual acceptance by both Parties of an estimate of work provided by the expert.

<u>C – Additional Conditions</u>

- 9. All archaeological work in connection with any Project in the Territory will be carried out in accordance with the *Ontario Heritage Act* and its Regulations. The Archaeological work will meet or exceed the Ontario Ministry of Heritage, Sport, Tourism, and Culture Industries (hereinafter, "MHSTCI") standards and guidelines for consultant archaeologists as amended, including the *Terms and Conditions for Archaeological Licences, Standards and Guidelines for Consultant Archaeologists (2011)* and the Draft *Engaging Aboriginal Communities in Archaeology Technical Bulletin (2011),* (hereinafter collectively, "MHSTCI Standards 2011").
- 10. The Proponent agrees that all archaeological work conducted for the Project will comply with the MCFN Standards and Guidelines for Archaeology (published April 2, 2018), (hereinafter, "MCFN Standards") as long as the MCFN Standards do not fall below MHSTCI Standards 2011. The MHSTCI Standards 2011 will be paramount in the event of a direct conflict between MCFN Standards and the MHSTCI Standards 2011.
- 11. The Proponent shall make best efforts to avoid and protect archaeological sites, artifacts, and/or features. The Parties agree that the preferred option for human remains that may be of Aboriginal ancestry is that they remain where they are found with appropriate protections.
- 12. If archaeological resources are encountered at any time during construction or other Project-related activity, all excavation or other activity that could disturb the site shall immediately cease, and the Proponent shall immediately notify MCFN's duly appointed Archaeological Operations Supervisor or designate. The Parties shall work

collaboratively to minimize impacts and ensure respectful treatment of any archaeological resources in accordance with the practices and values of MCFN as identified by MCFN.

- 13. If human remains are encountered at any time during construction or other Project-related activity, the following steps shall be taken:
 - a. All excavation or other activity that could disturb the site shall immediately cease, and the area shall be secured in a manner which protects the site location and prevents public access and trespass; and
 - b. In addition to any notifications required under the *Funeral, Burial and Cremation* Services Act, 2002, SO 2002, C 33, the Proponent shall immediately contact MCFN's duly appointed Archaeological Operations Supervisor or designate; and
 - c. MCFN shall be permitted to conduct any ceremonies on site in relation to the human remains that may be of Aboriginal ancestry; and
 - d. MCFN shall be consulted about all steps in the investigation and any decisions or agreements to be made regarding human remains that may be of Aboriginal ancestry.
- 14. Nothing in this Agreement shall be interpreted or implemented so as to derogate or abrogate from any MCFN Aboriginal or Treaty right or claim, or to indicate consent to the Project.

D - Method of Payment

15. The Parties agree that the Proponent will pay the capacity funding as agreed to above by cheque or bank transfer and upon receipt of an invoice from MCFN. All invoices will be addressed directly to the Proponent, the Project will be noted in the text of each invoice, and all invoices will be prepared as per MCFN-DOCA's standard invoicing format. Invoices should be submitted electronically to the following address:

Email address: droga@mte85.com Attention: Dot Roga on behalf of Township of Wilmot MTE Consultants Inc. 520 Bingemens Centre Drive Kitchener ON N2B 3X9 (519) 743-6500 ext 1269

16. All payment should be made to the MCFN Department of Consultation and Accommodation to the following address. For additional information, please call the office at 905-768-4260.

Email address: <u>nicole.laforme-hess@mncfn.ca</u>

Attention: MCFN-DOCA 4065 Highway 6 Hagersville, Ontario N0A 1H0

17. After thirty [30] days, a 5% monthly compounded interest rate will be charged on outstanding invoices. After six [6] months of non-payment, a 20% monthly compounded interest rate will be charged on outstanding invoices.

<u>F – Disclaimer</u>

- 18. The Parties agree that the capacity funding payments for the FLRs will be used only for the purposes described in this Agreement and will not be paid for the improper personal gain of any individual or for any other purpose that might violate any Canadian anticorruption law.
- 19. This agreement may be executed in counterparts.
- 20. This agreement is legally binding on MCFN and the Proponent. This agreement is legally binding on MCFN and the Proponent. This agreement is signed by authorized representatives of the Parties on the date set out in this agreement below.
- 21. The term of this agreement expires on April 1, 2022. In the event that Project-related activities continue past this termination date, a new agreement will be executed between Parties.

[The remainder of this page is intentionally left blank.]

Signed this _____ day of _____, 2021,



Authorized Signatory on behalf of The Proponent





Authorized Signatory on behalf of Mississaugas of the Credit First Nation

per

Mark LaForme Director Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

2 Elvier

Witness

Witness

[printed name of witness] [job title] [department] [name of the proponent]

Megan DeVries Archaeological Operations Supervisor Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation





Municipalities within Mastersugae of the Grecil Treaty Lands and Territory

Schedule B

Approx. Quote for Technical Review (Reference Only)

For review of materials and communications associated with Stage 1 AAs.

| | Number | Rate | Total |
|---------------------|--------|--------------|--------------|
| review hours | 4.0 | \$ 150.00 | \$ 600.00 |
| contingency (@ 20%) | | | \$ 120.00 |
| Total | | | \$ 720.00 |

For review of materials and communications associated with Stage 2 AAs.

| | Number | Rate | Total |
|---------------------|--------|--------------|--------------|
| review hours | 4.0 | \$ 150.00 | \$ 600.00 |
| contingency (@ 20%) | | | \$ 120.00 |
| Total | | | \$ 720.00 |

For review of materials and communications associated with Stage 3 AAs.

| | Number | Rate | 2 | Total |
|---------------------|--------|--------------|----|----------|
| review hours | 8.0 | \$ 150.00 | \$ | 1,200.00 |
| contingency (@ 20%) | | | \$ | 240.00 |
| Total | | | \$ | 1,440.00 |

For review of materials and communications associated with Stage 4 AAs.

| | Number | Rate | Total |
|---------------------|--------|--------------|----------------|
| review hours | 8.0 | \$ 150.00 | \$ 1,200.00 |
| contingency (@ 20%) | | | \$ 240.00 |
| Total | | | \$ 1,440.00 |

Field Liaison Representative Participation Agreement between: The Mississaugas of the Credit First Nation and Township of Wilmot

A - Background

- The purpose of this agreement is to provide the Mississaugas of the Credit First Nation (hereinafter, "MCFN") with capacity assistance to its Field Liaison Representatives (hereinafter, "FLRs") in connection with all environmental and/or archaeological assessments required for the Baden Sanitary Trunk EA (hereinafter, "the Project") located in Baden, Ontario, owned by Township of Wilmot, (hereinafter, "the Proponent").
- 2. The Proponent understands that MCFN wishes to send its FLRs to participate in and monitor the assessments associated with the Project, and that the FLRs' mandate will be to ensure that MCFN's perspectives and priorities are considered and to enable MCFN to provide timely and meaningful comment on the Project.
- 3. All archaeological work in connection with any Project in the Territory will be carried out in accordance with the *Ontario Heritage Act* and its Regulations. The archaeological work will meet or exceed the Ontario Ministry of Heritage, Sport, Tourism, and Culture Industries (hereinafter, "MHSTCI") standards and guidelines for consultant archaeologists as amended, including the *Terms and Conditions for Archaeological Licences, Standards and Guidelines for Consultant Archaeologists (2011)* and the Draft *Engaging Aboriginal Communities in Archaeology Technical Bulletin (2011),* (hereinafter collectively, "MHSTCI Standards 2011").
- 4. The Proponent agrees that all archaeological work conducted for the Project will comply with the MCFN *Standards and Guidelines for Archaeology* (published April 2, 2018), (hereinafter, "MCFN Standards") as long as the MCFN Standards do not fall below MHSTCI Standards 2011. The MHSTCI Standards 2011 will be paramount in the event of a direct conflict between MCFN Standards and the MHSTCI Standards 2011.
- 5. Nothing in this Agreement shall be interpreted or implemented so as to derogate or abrogate from any MCFN Aboriginal or Treaty right or claim, or to indicate consent to the Project.

B – Fees and Cost Structure

- 6. The Proponent will provide capacity funding for each FLR in the amount of \$85.00 per hour for all activities relating to the Project. Activities relating to the Project include, but are not limited to:
 - a. Time spent on site monitoring assessment or predetermined construction-related activities;
 - b. Time spent completing data or artifact processing, identification, analysis, and interpretation activities alongside their consultant(s);
 - c. Actual travel time at the beginning of, during, and/or end of each day;
 - d. Time completing daily notes relating to the Project;
 - e. Time spent on standby at the request of the Proponent or their consultant(s); and
 - f. Time completing mandatory training at the request of the Proponent or their consultant(s).
- 7. The Proponent will pay a supervisory fee of 3.5%, based on the number of hours charged to the Proponent, to provide MCFN with the capacity to facilitate in-field technical support for the FLRs via the Field Archaeologist.
- The Proponent will reimburse the FLRs for reasonable mileage and meals in accordance with current Federal Canada Treasury Board guidelines, over and above the hourly rate [see Schedule B]. Mileage rates are determined using the MCFN Department of Consultation and Accommodation as the place of departure.
- 9. The Proponent will provide capacity funding for each FLR in the amount of \$125.00 per hour for any work exceeding eight hours per day and/or forty hours per week. The above noted mileage and meal allowance remains in effect.
- 10. The Proponent will provide capacity funding for each FLR in the amount of \$125.00 per hour for any work occurring on the following holidays: New Year's Day, Family Day, Good Friday, Victoria Day, Indigenous Solidarity Day (June 21), Canada Day, Civic Holiday, Labour Day, Thanksgiving Day, Remembrance Day, Christmas Day, and Boxing Day. The above noted mileage and meal allowance rates remain in effect.
- 11. The Proponent agrees that the FLRs will be paid for a minimum of three hours, plus actual travel time, mileage, and meal allowance rates as noted above, on any day when work is cancelled by the Proponent or their consultant(s) while FLRs are en route to the work site or after the FLRs have already arrived.

- 12. If its use is deemed necessary by both Parties, the Proponent agrees to reimburse the FLRs for their use of the 407ETR upon receipt of a copy of the bill. This agreement will be provided in writing to MCFN's Field Coordinator.
- 13. If deemed reasonable by both Parties, the Proponent agrees to cover the cost of overnight accommodation for FLRs participating in environmental and/or archaeological fieldwork at locations which would otherwise require more than 90 minutes of travel time at both the beginning and end of the work day, as determined using the MCFN Department of Consultation and Accommodation as the place of departure. An additional Incidental Allowance fee is required for any work which requires overnight accommodations, as set out in Schedule B. This agreement will be provided in writing to MCFN's Field Coordinator.

<u>C – Additional Conditions</u>

- 14. The parties acknowledge that the Project, in whole or in part, takes place within MCFN Territory and agree that the Proponent shall provide capacity funding for FLR participation on the Project for the duration of the Project.
- 15. The Proponent agrees that two FLRs shall be on location whenever Project-related activities are taking place within its Territory, as set out in Schedule A.
- 16. Furthermore, additional FLRs are required if the number of field personnel utilized by the consultant exceeds fourteen (14) individuals and the Proponent agrees to provide capacity funding for additional FLRs as required. MCFN requires one additional FLR per five additional field crew, as outlined in the chart below:

| Number of Field Personnel | Number of FLRs Required |
|---------------------------|-------------------------|
| 1 to 14 | 2 |
| 15 to 19 | 3 |
| 20 to 24 | 4 |
| 25 to 29 | 5 |
| 30 to 34 | 6 |
| 35 to 39 | 7 |
| 40+ | 8+ |

17. The Parties acknowledge that the FLRs time and travel will be recorded and verified using the ClockShark Time Tracking Software System and that invoicing will be prepared using these records, not those of a third party.

- 18. If archaeological resources are encountered at any time during construction or other Project-related activity, all excavation or other activity that could disturb the site shall immediately cease, and the Proponent shall immediately notify MCFN's Archaeological Operations Supervisor or designate. The Parties shall work collaboratively to minimize impacts and ensure respectful treatment of any archaeological resources in accordance with the practices and values of MCFN as identified by MCFN.
- 19. If human remains are encountered at any time during construction or other Project-related activity, the following steps shall be taken:
 - a. All excavation or other activity that could disturb the site shall immediately cease, and the area shall be secured in a manner which protects the site location and prevents public access and trespass; and
 - In addition to any notifications required under the *Funeral, Burial and Cremation* Services Act, 2002, SO 2002, C 33, the Proponent shall immediately contact MCFN's duly appointed Archaeological Operations Supervisor or designate; and
 - c. MCFN shall be permitted to conduct any ceremonies on site in relation to the human remains that may be of Aboriginal ancestry ("Ancestral Remains"); and
 - d. MCFN shall be consulted about all steps in the investigation and any decisions or agreements to be made regarding Ancestral Remains.

D - Coordination of the FLRs

- 20. The Parties agree that the FLRs will follow the reasonable instructions of the Proponent and their consultant firm(s) conducting the environmental and/or archaeological work concerning safety practices, and that the FLRs will attend "tailgate" safety meetings if requested.
- 21. The contact person for activities relating to the environmental assessment portion of the Project is [name of contact person #1] from [name of consultant]. Contact information for this person is as follows:

[insert contact information here]

22. The contact person for activities relating to the archaeological assessment portion of the Project is Garth Grimes from Detritus Consulting Ltd. Contact information for this person is as follows:

Detritus Consulting Ltd. www.detcon.net (519) 744-7018 23. The Parties agree that the contact person for the consultant firm(s) will coordinate site meeting locations and times through MCFN's duly appointed Field Coordinator. Contact information for the Field Coordinator is as follows:

Joelle Williams Telephone: 905-768-4260 Cell: 905-870-2918 Email: joelle.williams@mncfn.ca

E - Status of the FLRs

- 24. The FLRs selected by MCFN have appropriate qualifications for the work required for example, training in environmental and/or archaeological monitoring – and experience in bridging Indigenous perspectives with Western approaches, as reasonably determined by MCFN.
- 25. The Parties agree that the FLRs are not employees, contractors, or sub-contractors of the Proponent or their consultant(s) and that the FLRs will be responsible for their own personal protective equipment, such as hard hats, safety boots, and safety vests, unless specific or otherwise unique personal protective equipment is required, which will therefore be provided or reimbursed by the Proponent.
- 26. FLRs take direction from MCFN. MCFN pays Workplace Safety and Insurance Board ("WSIB") contributions in respect of the FLRs and will, at its own expense, maintain for the term of this agreement a comprehensive general liability ("CGL") policy or policies with a limit of at least \$1 million and shall provide the Proponent with evidence of such insurance, upon request. MCFN agrees that FLRs will perform their activities safely, in a good and competent manner, in compliance with all applicable laws, regulations, and guidelines.
- 27. MCFN expects that the Proponent will comply with the *Occupational Health and Safety Act*, R.S.O. 1990, C. 0.1, the Ontario *Human Rights Code*, R. S. O. 1990, c. H.19, and maintain a safe, harassment-free work environment.
- 28. The Proponent is responsible for negligence or other failure to maintain a safe and harassment-free work environment. To the extent that the Proponent is responsible for negligence or other failure to maintain a safe and harassment-free work environment, the Proponent is liable and shall indemnify MCFN claims or demands related to injury, accident, discrimination, or harassment by the Proponent's employees, agents, consultants, or other parties under the control or direction of the Proponent.
F - Method of Payment

29. The Parties agree that the Proponent will pay the capacity funding as agreed to above by cheque or bank transfer and upon receipt of an invoice from MCFN. All invoices will be addressed directly to the Proponent, the Project will be noted in the text of each invoice, and all invoices will be prepared as per MCFN-DOCA's standard invoicing format. Invoices should be submitted electronically to the following address:

Email address: droga@mte85.com Attention: Dot Roga on behalf of Township of Wilmot MTE Consultants Inc. 520 Bingemens Centre Drive Kitchener ON N2B 3X9 (519) 743-6500 ext 1269

30. All payment should be made to the MCFN Department of Consultation and Accommodation to the following address. For additional information, please call the office at 905-768-4260.

> Email address: <u>nicole.laforme-hess@mncfn.ca</u> Attention: MCFN-DOCA 4065 Highway 6 Hagersville, Ontario N0A 1H0

31. After thirty [30] days, a 5% monthly compounded interest rate will be charged on outstanding invoices. After six [6] months of non-payment, a 20% monthly compounded interest rate will be charged on outstanding invoices.

<u>G – Disclaimer</u>

- 32. The Parties agree that the capacity funding payments for the FLRs will be used only for the purposes described in this Agreement and will not be paid for the improper personal gain of any individual or for any other purpose that might violate any Canadian anti-corruption law.
- 33. This agreement may be executed in counterparts.
- 34. This agreement is legally binding on MCFN and the Proponent. This agreement is signed by authorized representatives of the Parties on the date set out in this agreement below.

35. The term of this agreement expires on April 1, 2022. In the event that Project-related activities requiring FLR participation continue past this termination date, a new agreement will be executed between Parties.

[The remainder of this page is intentionally left blank.]

Signed this ______, 2021,



Authorized Signatory on behalf of The Proponent

Jeff Molenhuis Director Public Works and Engineering Township of Wilmot Authorized Signatory on behalf of Mississaugas of the Credit First Nation

Mark LaForme Director Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation

Witness

Witness

[printed name of witness] [job title] [department] [name of the proponent]

Megan DeVries Archaeological Operations Supervisor Dept. of Consultation and Accommodation Mississaugas of the Credit First Nation





Municipalities within Mastersugae of the Grecil Treaty Lands and Territory

Schedule B

| MISSISSAL | 645 DE 16 | ECKEDI HISI MA | TON MUSASE & ALLOWARCE CHART |
|---------------|-------------------|--------------------|---|
| | 6656 | стия арри 1, 202 | 1 to AMBCH 81, 2022 |
| MURANE | анылан ўслагрегом | | |
| | http://w | www.njo-annugazaj | /cline.1/we/d10/v206/s250/en |
| MEALALLOWMARE | | | nor eligible if: |
| DECARTAGE | 5 | 21.10 | kowe a tar dang gat bena belora 7:00 m |
| LUNCH | 2 | 31.55 | leave after 1:15pm; get home before 12:15pm |
| O IN MER | 5 | 52,40 | leave after April get home before 5.40 pm |
| INCODERTAIS. | 5 | 12.50 | non strying overnight |
| | <u>hlinció</u> | ever the time pass | ienee.5we/4007/2007/2007/090/en |
| NEUTOMET | \$ | 52.40 | "applies only to regist time surveys that would not |
| | | | otherwise trigger dinner or breaktast. |



Public Information Centre Slides and Comments





Welcome to the first Virtual Public Information Centre for the Baden Trunk Sanitary Sewer Class Environmental Assessment.

This study has been initiated by the Township of Wilmot. MTE Consultants Inc. has been retained by the Township to serve as the lead consultant on the project.

This Public Information Centre will by published virtually and will remain online.



The goals of our first Public Information Centre are:

- To introduce the objective of the study by defining the problem or opportunity statement,
- To Outline the Class Environmental Assessment Process,
- Identify the study area and the lands to be serviced
- Present existing sanitary conditions
- Point out the possible sanitary sewer trunk alignments
- State the next steps in the study process
- Define the criteria used for evaluating the alternatives
- Confirm the preferred evaluation method
- And to provide the general public, Indigenous groups and stakeholders with an opportunity to review, provide comments and submit questions



The Problem/Opportunity statement for the Baden Trunk Sanitary Sewer states that

The existing sanitary infrastructure lacks capacity and depth for the planned growth within the Settlement of Baden. The proposed sanitary trunk will accommodate current and future growth in the area. The study will determine the corridor servicing alignment, size and depth of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.



The Township of Wilmot is responsible for the long-term operation and maintenance of the wastewater collection system for the Settlement of Baden. In 2014, a wastewater servicing study for the growth areas of Baden and New Hamburg was prepared for the Township of Wilmot.

The study evaluated the sanitary infrastructure needs presented in the 2011 Baden – New Hamburg Water and Wastewater Master Plan and evaluated the development options of these new growth areas.

The additional expected growth identified in the updated 2019 Official Plan for the Township of Wilmot has prompted an overall review of the servicing strategy for the entire Baden – New Hamburg area.

The Township has entered into a Schedule B Municipal Class EA for the Baden Trunk Sanitary Sewer. The Notice of Commencement was issued on June 17, 2021

This is the first Public Information Centre for the Baden Trunk Sanitary Sewer EA. We look forward to your comments and participation

| Baden Trunk Sanitary Sewer Class EA | | | | | |
|--|---|--|--|--|--|
| WHAT? | What is an environmental assessment (EA)? A process used to predict environmental effects before project implementation What are the benefits of the EA process? Incorporates environmental factors into the decision-making process Minimizes or avoids adverse environmental effects before they occur Provides an opportunity for meaningful public and stakeholder participation Provides comprehensive decision making that considers a range of factors | | | | |
| WHY? | Why are we undertaking an EA for the Baden Sanitary Sewer Trunk? Identify and evaluate a range of potential sanitary sewer trunk alignments that satisfy the needs of the growing Settlement of Baden | | | | |
| HOW? | How will we determine the best alignment? The Municipal Engineers Class Environmental Assessment – Schedule B framework Specialized Technical studies to minimize impacts Meaningful public and stakeholder engagement | | | | |

An environmental assessment is a process used to identify potential environmental impacts and avoid them where possible.

The benefits of the environmental assessment process include

- Incorporating environmental factors into the decision-making process
- Minimizing or avoiding adverse environmental effects before the project is initiated
- Providing an opportunity for meaningful public and stakeholder participation
- Providing comprehensive decision making that considers a range of factors

We are undertaking an EA for the Baden Trunk Sanitary Sewer to

• Identify and evaluate a range of potential sanitary sewer trunk alignments that satisfy the needs of the growing Settlement of Baden

It will be accomplished by

- Following a Schedule B framework outlined in the Municipal Engineers Class Environmental Assessment manual
- Preforming and reviewing technical studies to minimize impacts
- And by having meaningful engagements with the public and stakeholders

Municipal Class Environmental Assessment Process

The Municipal Class Environmental Assessment is a process that provides a decision-making framework enabling the requirements of the Environmental Assessment Act to be met in an effective and efficient manner.

What is a Schedule B?

- Schedule B projects refer to the magnitude of anticipated environmental impacts and are subject to the prescribed screening criteria as outlined in the Municipal Class Environmental Assessment manual.
- Provides for comprehensive decision making that considers a range of factors
- Enables consultation with all those who may be affected and the relevant review agencies is a critical part of the process.

The Municipal Class Environmental Assessment is a process that provides a decisionmaking framework enabling the requirements of the Environmental Assessment Act to be met in an effective and efficient manner.

Broadly defined, 'environment' includes the natural, social, cultural, built and economic environments.

It applies to municipal infrastructure projects including roads, water and wastewater projects.

'Schedule B' projects refer to the magnitude of anticipated environmental impacts and are subject to the prescribed screening criteria as outlined in the Municipal Class Environmental Assessment manual.

It provides for comprehensive decision making that considers a range of factors

A critical part of the process is consulting with all those who may be affected and the relevant review agencies.

12



To best identify and assess possible environmental impacts, a project team was put together. Each team member will perform a detailed study in the area of their expertise. Their studies will assist in identifying which alternative best satisfies the Problem/opportunity statement during the evaluation process.



There are 3 key phases to a schedule B Class EA

Phase 1 begins with the Notice of Commencement where problems or deficiencies, as well as opportunities are addressed and formalized into a Problem/Opportunity Statement



The process in Phase 2 helps to determine which schedule to use.

For this study, the first step includes identifying the Alternative Alignments, this is performed by

- Gathering Background Data on the natural, social and economic environments
- Documenting Existing Conditions
- Identifying constraints



The second step evaluates the alternative alignments. Once the project teams have completed their detailed studies, their assessments become the basis for evaluating each alternative alignment in the form of an evaluation matrix. Consulting with the review agencies and the public to solicit comment and input is key.



The final step of phase 2 will be to select the preferred alignment identified from the matrix, and taking into consideration the input and comments received from the review agencies and the public.



The final phase of the Schedule B Class EA process will be to present the preliminary design of the preferred alignment and submit a Notice of Completion.



Throughout the EA process, the public and stakeholders have an opportunity to actively engage and participate in the decisions affecting their community.

Currently, we are at the beginning stage of phase 2



This study is located within the Settlement of Baden

It takes into consideration the future development of Baden as prescribed by the Provincial Policy Statement of 2020, the Places to Grow Plan of 2020, the amended Regional Official Plan of 2015 and the Township Of Wilmot Official Plan of 2019



The Urban Areas, as seen in gold, are the primary focus for current population and employment growth

Lands located outside the urban areas are deemed countryside, however, the countryside line, as seen in silver, represents the future expansion to the Baden urban area boundary to accommodate for future growth

The proposed Baden Trunk Sanitary Sewer will service the areas east of Nafziger Rd within the countryside line.



Aside from the first alternative, The Baden Trunk alignment alternatives will begin and end within the boundaries of the detailed study area.

The alternatives located within the detailed study area, will begin at the intersection of Snyder's Road West and Christian Street, and ultimately connect to the Baden wasterwater pumping station, adjacent to Foundry Street just south of Gingerich Road.

Existing Conditions



The existing Baden Sanitary Sewer network is comprised of gravity sewers, pressure sewers and trunk sewers

There are 2 wastewater pumping stations.

The Baden West Wastewater Pumping Station is located on Charlotta Street, and the Baden Wastewater Pumping Station, located adjacent to Foundry Street, just south of Gingerich Road.

The wastewater servicing study of 2014 identified that the existing sanitary sewer did not have the capacity to support future growth.

It also identified that, under existing conditions in 2012, the sanitary sewer and the Baden wastewater pumping station on Foundry Street, were over-capacity with respect to peak flows.



The existing Baden Sanitary Sewer network is comprised of gravity sewers, pressure sewers and trunk sewers

There are 2 wastewater pumping stations.

The Baden West Wastewater Pumping Station is located on Charlotta Street, and the Baden Wastewater Pumping Station, located adjacent to Foundry Street, just south of Gingerich Road.

The wastewater servicing study of 2014 identified that the existing sanitary sewer did not have the capacity to support future growth.

It also identified that, under existing conditions in 2012, the sanitary sewer and the Baden wastewater pumping station on Foundry Street, were over-capacity with respect to peak flows.



Five sanitary conveyance alignments, including a "Do Nothing" alternative, have been identified as possible alignments for the Baden Trunk sanitary sewer.

Each alternative has its own unique set of impacts

The 'Do Nothing' alternative begins outside the detailed study area, bordering the eastern boundary of Baden, and terminating within the detailed study area

The remaining 4 alternative alignments lie within the detailed study area.



Alternative l is the 'Do-Nothing' alternative

It examines what may happen if no alternative is selected.

In this case, if none of the alternatives are selected, then the current wastewater servicing strategy of 2014 will be carried out.

This alignment begins at the north eastern urban boundary of Baden, runs southwest along the railroad and down Foundry Street to the Baden wastewater pumping station

This alternative would require constructing a pumping station, constructing a pressure sewer along the eastern Urban Boundary, crossing a railroad, upgrading existing sewer and integrating it along the Regional Rd.



This alignment begins at the north eastern urban boundary of Baden, runs southwest along the railroad and down Foundry Street to the Baden wastewater pumping station

This alternative would require constructing a pumping station, constructing a pressure sewer along the eastern Urban Boundary, crossing a railroad, upgrading existing sewer and integrating it along the Regional Rd.



Alternative 2 is predominantly lies within the existing right-of-ways of Snyder's Road West and Foundry Street

This alternative would require reconstructing large portions of existing road infrastructure, crossing a railroad,

Upgrading existing sewers as well as upgrading and integrating sewers along a Regional road



Alternative 3 lies within the existing right-of-ways of Christian St, Charlotta St, Mill St, Charles St. and Foundry St

This alternative would require reconstructing large portions of existing road infrastructure, crossing a railroad

Upgrading existing sewers as well as upgrading and integrating sewers along a Regional road



Alternative 4 predominantly lies within the existing right-of-ways of Christian St, Charlotta St, Queen St, and Foundry St

This alternative would require reconstructing existing road infrastructure, crossing a railroad, multiple stream crossings

Constructing new sewers, upgrading existing sewers as well as upgrading and integrating sewers along a Regional road



Alternative 5 affects the existing right-of-way of Christian St.

This alternative would require reconstructing existing road infrastructure, crossing a railroad, a stream crossings

Construction and upgrading existing sewers as well as upgrading and integrating sewers along a Regional road



The next step of the environmental assessment process is to evaluate the alternatives based on the following criteria:

- **The impacts on the Natural Environment
- **The impacts on the Socio-Cultural Environment
- **The impacts on Transportation, Municipal Services & Utilities
- **The Financial Impact
- **Technical Impacts**

Each alignment will undergo a numerical evaluation. The Numerical Evaluation approach was selected as the preferred scoring methodology. This methodology requires a "Must 10" scoring assessment of the alternatives in a matrix form. Comments justifying the scoring will be included for each alternative.

For each factor, the least preferred alternative receives a score of 1 and the most preferred alternative receives a score of 10. All scores will fall between these 2 values. The relative score of the alternatives between the high (10) and the low (1) will be justified in the commentary utilizing the professional judgement of the project team and the reviewing agencies.

The recommended alignment will then be identified based on the highest score.

| We W | ant to Hear from | m You |
|--|--|---|
| THANK YOU | FOR YOUR PART | |
| Any comments, questions or suggestions relevant to this study should be directed to either of the following project team | Jeff Molenhuis, P.Eng Director of Public Works and Engineering Township of Wilmot 60 Snyder's Road West Baden, ON N3A 1A1 | Dot Roga, C.E.T Division Manager MTE Consultants Inc. 520 Bingemans Centre Drive Kitchener, Ontario N2B 3X9 |
| members: | Phone: (519)634-8519 ext.9238 Fax: (519)634-5522 Email: jeff.molenhuis@wilmot.ca | Phone: (519)743-6500 Fax: (519)743-6513 Email: droga@mte85.com |
| with with | ioe 🖏 MTE 👫 | Detritus Consulting Ltd. Archaeology |

Thank you for participating in the Baden Trunk Sanitary Sewer Class (EA) Online Public Information Centre (PIC).

Any comments, questions or suggestions relevant to this study should be directed to either Jeff Molenhuis with the Township of Wilmot or to Dot Roga with MTE Consultants.

This Public Information Centre is relying on web-based communications, please feel free to fill out the comment sheet located on this website. Any comments received will be collected under the Environmental Assessment Act and, with the exception of personal information, will become part of the public record.



Welcome to the second Virtual Public Information Centre for the Baden Trunk Sanitary Sewer Class Environmental Assessment.

This study has been initiated by the Township of Wilmot. MTE Consultants has been retained by the Township to serve as the lead consultant on the project and we will be guiding you through this presentation on behalf of the Township of Wilmot and the study team.

This Public Information Centre will by published virtually on the Township of Wilmot's website and will remain online for the duration of the project.



The goals of our second Public Information Centre are to:

Provide a summary of the Project to date ...

Identify where we are in the Municipal Class Environmental Assessment Process...

Review the study area

Define the existing sanitary design system...

Summarize the existing natural, social, cultural/heritage, built, and economical environments...

Present the wastewater servicing alternatives and their associated impacts...

Summarize the evaluation of the alternative solutions...

Identify the preliminary preferred alternative...

And to provide a timeline for this study and the next steps.

We encourage the general public, Indigenous groups and stakeholders to review the presentation, provide comments and/or submit questions on a comment sheet available on the Township's website. Information on how to submit your comments will be provided at the end of the presentation. Your input is appreciated and will be considered by the study team in the evaluation of the alternative solution.


The Township of Wilmot has identified that the existing sanitary infrastructure lacks capacity and depth for the planned growth within the Settlement of Baden. The proposed sanitary trunk will accommodate current and future growth in the area. This study will determine the corridor servicing alignment, size and depth of the trunk sewer to facilitate growth and optimize existing Township infrastructure in the area.



Providing sewage collection to accommodate existing and future growth of Baden will require changes to the current sewage collection system. Under the Municipal Class EA process, establishing, extending or enlarging a sewage collection system is considered a Schedule B project as defined in the Municipal Engineers Association Class EA document.

As a Schedule B project, the project planning proceeds under the planning and documentation procedures of Phases 1 and 2 of the Municipal Class Environmental Assessment process. Through this process, feasible solutions are identified and evaluated with input from agencies, Indigenous communities and stakeholders toward a recommendation for a preferred solution. At the conclusion of Phase 2, the appropriate EA planning Schedule is confirmed and, if no Part II order requests are received by the Minister, the proponent may proceed to design and implementation of the project.

An overview of the EA process is illustrated on the flow chart. The study is currently at the stage where we engage with interested parties and the public about the project and the alternative solutions, as highlighted in the yellow box under Phase 2. We are currently midway through Phase 2 of the study. Following the confirmation of the preferred solution, Project File Report (PFR) will be prepared to summarize the decision-making process for the public and stakeholders. Once Township council has endorsed the preferred alternative, a 30-day review period will be provided for the public and stakeholders to review and comment on the Report.



The primary study area is located within the Settlement of Baden and consists primarily of agricultural and residential land uses with some commercial and institutional land uses.

Several development proposals within the Study Area are in pre-consultation and are at various stages of approval.

Key considerations include:

Future development per the Township's Official Plan, topography, capacity of existing sewer, encroachment on natural features and planning policy and regulation.

The Township, in conjunction with the Region, will ensure that all new development is serviced by adequate and effective water, wastewater and storm water facilities.

The Detailed Study Area defines the boundary for the proposed Baden trunk alternative alignments.

It is defined as the area south of Snyder's Road West, west of Foundry Street to the Baden pumping station, north of Gingerich Road and east of Christian Street including the agricultural lands it crosses.

All alternative alignments will begin at the intersection of Snyder's Road West and Christian Street and ultimately connect to the Baden wastewater pumping station.

There are four prominent water features, existing archeological findspots and sensitive vegetation communities within the detailed study area



The Detailed Study Area defines the boundary for the proposed Baden trunk alternative alignments.

It is defined as the area south of Snyder's Road West, west of Foundry Street to the Baden pumping station, north of Gingerich Road and east of Christian Street including the agricultural lands it crosses.

All alternative alignments will begin at the intersection of Snyder's Road West and Christian Street and ultimately connect to the Baden wastewater pumping station.

There are four prominent water features, existing archeological findspots and sensitive vegetation communities within the detailed study area



The Settlement of Baden is currently serviced by a two-tier wastewater system by Township of Wilmot and the Region of Waterloo.

The Township is responsible for the sanitary sewers and one local pumping stations, the Baden West wastewater pumping stations. The Region of Waterloo is responsible for the Baden wastewater pumping station.

The areas outlined in orange are the current areas serviced by the Township.

The Baden wastewater pumping station and associated sewer system, are under capacity for peak flows and do not support the future growth demands of Baden.

Areas in orange represent the current

The existing Baden Sanitary Sewer network is comprised of gravity sewers, pressure sewers, trunk sewers and wastewater pumping stations

Pressure sewers and local gravity sewers connect with existing sanitary trunk sewers along Foundry Street where it flows southeast to the Baden wastewater pumping station then pumped to the New Hamburg wastewater treatment plant.

Based on the elevations and locations of existing infrastructure, extended sanitary service to the Primary Study Area would not be achievable by gravity.



The Settlement of Baden is currently serviced by a two-tier wastewater system by Township of Wilmot and the Region of Waterloo.

The Township is responsible for the sanitary sewers and one local pumping stations, the Baden West wastewater pumping stations. The Region of Waterloo is responsible for the Baden wastewater pumping station.

The areas outlined in orange are the current areas serviced by the Township.

The Baden wastewater pumping station and associated sewer system, are under capacity for peak flows and do not support the future growth demands of Baden.

Areas in orange represent the current

The existing Baden Sanitary Sewer network is comprised of gravity sewers, pressure sewers, trunk sewers and wastewater pumping stations

Pressure sewers and local gravity sewers connect with existing sanitary trunk sewers along Foundry Street where it flows southeast to the Baden wastewater pumping station then pumped to the New Hamburg wastewater treatment plant.

Based on the elevations and locations of existing infrastructure, extended sanitary service to the Primary Study Area would not be achievable by gravity.



An assessment of the natural environment was completed to characterize the vegetation communities and the potential for habitat of Species at Risk.

Vegetation communities within the Detailed Study Area and along Alternative 1 were inventoried to assess potential direct and/or indirect impacts to significant and sensitive vegetation species as well as terrestrial habitat features and functions. Although the vegetation communities within the Detailed Study Area are considered to be relatively common in Ontario, there are vegetation communities of interest, these include forest, woodland, cultural thicket and cultural meadow.

The significant and sensitive vegetation species identified include the Species at Risk (SAR) Butternut and the regionally significant Running Serviceberry.

Significant natural features include Locally Significant Woodlands and nonsignificant woodlands.



Impacts to wildlife habitats, including habitat for Species at Risk and Significant Wildlife Habitat (SWH), were assessed.

Two Candidate Significant Wildlife Habitats were identified, they include:

• All wooded areas, which have the potential to provide suitable habitat for Species at Risk bats.

and

• The Baden Creek pond, which provides suitable habitat for the Species for Conservation Concern Snapping Turtle and is a possible Turtle Wintering Area

The ecological connectivity associated with the Baden Creek riparian zone provides a potential corridor function for land-based animals.



Wetland cover within the study area is limited to small, isolated features that are anthropogenic in nature which may have been highly altered by past land uses. Wetlands that are not provincially significant are generally considered "locally significant" in the Township. Locally Significant Wetlands are considered a form of Supporting Environmental Feature (Township of Wilmot 2019).



There are five main water features within the Primary Study Area, they include Baden Creek, a permanent watercourse with groundwater inflow, The Baden Creek tributary A, an ephemeral/intermittent stream, The Baden Creek tributary B, a permanent watercourse with groundwater inflow, The Huehn Award Drain, a coldwater stream and the Baden Creek pond

Baden Creek provides direct fish habitat including suitable spawning, foraging and rearing habitats for a variety of warmwater and coolwater fish.

All watercourses provide direct fish habitat except tributary A.

Baden Creek provides habitat for a variety of warmwater and coolwater fish due to groundwater inflow, vegetation cover and the Huehn Award Drain.



The hydrogeologic assessment was based on geologic and geotechnical information, was used to assess impacts to:

- Shallow Groundwater Movement
- Groundwater Recharge areas
- Groundwater Quality by means of
 - > SGRA Significant Groundwater Recharge Areas
 - > HVA Highly Vulnerable Aquifers
 - > WHPA Well Head Protection Areas and
 - > IPZ Intake Protection Zones

No significant impacts to groundwater resources were revealed



The hydrogeologic assessment was based on geologic and geotechnical information, was used to assess impacts to:

- Shallow Groundwater Movement
- Groundwater Recharge areas
- Groundwater Quality by means of
 - > SGRA Significant Groundwater Recharge Areas
 - > HVA Highly Vulnerable Aquifers
 - > WHPA Well Head Protection Areas and
 - > IPZ Intake Protection Zones

No significant impacts to groundwater resources were revealed



Baden is designated as a Township Urban Area according to the Regional Official Plan of 2015.

The land use designations within Baden consist of environmentally constrained Lands, urban residential, urban core, open space, major recreational, light industrial with general industrial lands to the south and agricultural lands, which are primarily located outside the Urban Area, within the Countryside.

Each alternative was assessed for potential land use impacts.



The countryside are lands located outside the urban areas.

The countryside line, as seen in silver, is the long-term growth boundary between the existing Urban Area boundaries and the Countryside. It represents the future urban area boundary to accommodate for projected growth, and is consistent with the Provincial Policy Statement of 2020, the Places to Grow Plan of 2020, the amended Regional Official Plan of 2015 and the Township Of Wilmot Official Plan of 2019.



Stage 1 and Stage 2 archaeological assessments have been performed along the alternatives traversing open fields.

The Stage 1 background research indicated the Detailed Study Area exhibited moderate to high potential for the identification and recovery of archeological resources.

The Stage 2 investigations consisted of a pedestrian surveys at 3m intervals and test pits.

Archeological finds require a 20m protective buffer and a 50m monitoring buffer.



Baden has been identified as a potential Cultural Heritage Landscape.

As such, a cultural heritage assessment of the Primary Study Area was performed. Of the Eleven Cultural Heritage Resources identified within and/or adjacent to the Detailed Study Area, 7 were listed Heritage Properties and 4 were designated Heritage Properties within close proximity to the Alternatives.

In addition, Foundry Street and Snyder's Road West have been identified as scenic roads and have been included in the assessment.

Adverse impacts were evaluated as designated in the Ontario Heritage Toolkit (OHTK).

All cultural heritage resources identified are included in the Township's Cultural Heritage Register.



Five alternative alignments were selected for this study.

- 1. Do Nothing 2014 preapproved alignment
- 2. Snyder's/Foundry requires upgrading the existing sewers down two main regional roads
- 3. Christian/Charlotta/Mill/Charles/Foundry Upgrade existing sewers along local and regional roads
- 4. Christian/Charlotta/Queen and Baden Creek Upgrade local sewers and construct new sewer within agricultural lands
- 5. Christian and Baden Creek Upgrade one local sewer and construct new sewer within agricultural lands

The alternatives were evaluated relative to each other against a set of criteria to meet the project objectives outline in the Problem/Opportunity Statement.



Alternative 1 is the 'Do-Nothing' alternative

The eastern portion of the alignment begins at the proposed wastewater pumping station, runs southeast along the eastern boundary of the Primary Study Area, then southwest along the railway to Foundry Street. The western portion begins at the intersection of Christian Street and Snyder's Road West, proceeds east to Foundry Street, where it connects with the eastern portion, then southeast to the Baden wastewater pumping station.

The Do-Nothing alternative is a Mandatory requirement for consideration in a Class EA. It examines what may happen if no alternative is selected. In this case, if no alternative is selected, then the approved wastewater servicing strategy of 2014 will be carried out.



Aside from this alternative being the preapproved sanitary servicing strategy, the benefits of this alignment are that it does not impact the Significant Wildlife Habitat woodlands or the Species at Risk Butternut Trees. It also does not impact any known archeological findspots.



The constraints of this alternative are that it does not support the planned future development of Baden, it hinders timely project completion due to regional road corridors, it greatly disrupts the road network and traffic along Snyder's Road West and Foundry Street impacting access to existing businesses and residents as well as the Township Fire Hall. It will impact the largest number of cultural heritage resources during construction.

Installation of the **trenchless sewer pipes under the railway** will be technically complex due to **the narrow right-of-way and the various connections at the busy intersection** of Snyder's Road West and Foundry Street.



Impacts to flow regime at the stream crossings will lead to water quality and water quantity issues affecting the aquatic habitat of Baden Creek, the Baden Creek pond and the Huehn Award Drain.

The additional pumping station required for this alternative will increase noise and odour as well as long-term operation and maintenance costs. The right-of-way to service the eastern portion of the alignment has not been secured and is currently the most costly alternative.



Alternative 2 begins at the intersection of Snyder's Road West and Christian Street, proceeds east on Snyder's Road West to Foundry Street where it continues southeast to the Baden wastewater pumping station.

This alternative has negligible impacts to vegetation communities, wildlife habitat and wetlands. It does not impact any known archeological findings. It eliminates the school WWPS which reduces the noise, odour and long-term operation and maintenance costs associated with the pumping station. No land acquisitions are required for this alternative.



The constraints of this alternative are that it does not support the planned future development of Baden and hinders timely project completion due to regional road corridors. It greatly it disrupts the road network and traffic along Snyder's Road West and Foundry Street, impacting access to the existing businesses and residents as well as the Township Fire Hall. It will impact a large number of cultural heritage resources during construction.



Impacts to flow regime at the stream crossings will lead to water quality and water quantity issues which will affect the aquatic habitat of Baden Creek, the Baden Creek pond and the Huehn Award Drain.

Installation of the trenchless sewer pipes under the railway will be technically complex due to the deep sewer, the narrow railway right-of-way and the various connections at the busy intersection of Snyder's Road West and Foundry Street.



This alignment begins at the intersection Snyder's Road West and Christian Street, proceeds southeast to Charlotta Street, northeast to Mill Street, southeast to Charles Street, northeast to Foundry Street then southeast to the Baden WWPS.

This alternative has negligible impacts to vegetation communities, wildlife habitat and wetlands. It does not impact any known archeological findings. It eliminates two waste water pumping stations, the school WWPS and the Baden West WWPS which will reduce noise, odour and long-term operation and maintenance costs associated with the pumping station and the Baden West noise wall. No land acquisitions are required for this alternative.



The constraints of this alternative are that it does not support the planned future development of Baden and hinders timely project completion due to regional road corridors. It greatly disrupts the road network and traffic along local roads and Foundry Street impacting access to the existing businesses and residents as well as the Township Fire Hall. Various cultural heritage resources are also impacted.

Impacts to flow regime at the stream crossings will lead to water quality and water quantity issues which affect the aquatic habitat of Baden Creek, the Baden Creek pond and the Huehn Award Drain.



This alignment begins at the intersection of Snyder's Road West and Christian Street, it proceeds southeast to Charlotta Street, northeast to Queen Street, southeast on Queen Street, along Baden Creek to Gingerich Road, northeast to Foundry Street and southeast to the Baden WWPS

This alternative provides flexibility for servicing the entire Primary Study area and allows for timely project completion due to minimal regional road corridors. It eliminates two waste water pumping stations, the school WWPS and the Baden West WWPS which will reduce noise, odour and long-term operation and maintenance costs associated with the pumping station and the Baden West noise wall.



There are minimal impacts to Baden Creek, the Baden Creek pond and the Huehn Award Drain. It does not impact flow regime due to trenchless stream crossings or to the aquatic habitat of any water feature. There are minimal impacts to cultural heritage resources and the right-of-ways for this alignment have been secured.



The constraints of this alternative are that it disrupts the local road network and traffic along Christian Street, Charlotta Street and Queen Street impacting access to the existing local residents. It also infringes on the 50m monitoring buffer for two archeological findspots and may impact the locally significant wetlands at the trenchless railway crossing



This alignment begins at the intersection of Snyder's Road West and Christian Street, proceeds southeast to Charlotta Street, crosses the railway and runs adjacent to Baden Creek tributary A and Baden Creek to Gingerich Road, northeast to Foundry Street and southeast to the Baden WWPS

This alternative provides flexibility for servicing the entire Primary Study area and allows for timely project completion due to minimal regional road corridors. There are minimal disruptions to the road network and traffic and impacts access to existing businesses and residents the least. There are no impacts to cultural heritage resources.



It eliminates two waste water pumping stations, the school WWPS and the Baden West WWPS which will reduce noise, odour and long-term operation and maintenance costs associated with the pumping station and the Baden West noise wall. This alternative does not impact any locally significant wetlands, Species at Risk or Significant Wildlife Habitat.



There are negligible impacts to Baden Creek, the Baden Creek pond and the Huehn Award Drain. It does not impact flow regime due to trenchless stream crossings or aquatic habitat of any water feature. The right-of-ways for this alignment have been secured and represents the least cost solution.



This alternative disrupts local traffic along Christian Street and a small section of Charlotta Street impacting access to existing local residents. Will require monitoring within the 50m archeological buffer during construction.

| Evaluation of Alternative Solutions | | | | | | | | | | |
|--|---------------|------------------|--------|------|------|----------------------|------|--|--|--|
| THE RE | 1 Disedant | | | | | | | | | |
| The second distance from | - | No. Andrew | - 01 | | | | | | | |
| | | | | - | | Income of Parameters | | | | |
| 1 mar at | | | | | | | 121 | | | |
| Common Cale Summer State | 1.1 | 1 | | -3 | 112 | - | | | | |
| house have | - 1 • | | _1 | -* | | | | | | |
| Transmitting . | | 1 - 1 - 1 | | - N. | | | | | | |
| Provide in the second statement of | 1 | | - 10 m | -4- | 1.2 | | | | | |
| land the second se | | | 1. | - 8 | | | | | | |
| Name Artes | | 1000 | _ | _ | | | | | | |
| Conception in the local division of the loca | | | | | | | | | | |
| And the second s | | | | | | | | | | |
| And a second second second | 1.0.2 | | | | | | | | | |
| Construction of the local division of the lo | In the second | and some | - 2 | - 3 | 1000 | | - 10 | | | |

The impacts of the Alternatives were evaluated against the inventory of the natural, social/cultural, economic and technical environment, including possible mitigating measures.

| Evaluation of Alternative Solutions | | | | | | | | | | |
|--|---------------|------------------|--------|------|------|----------------------|------|--|--|--|
| THE RE | 1 Disedant | | | | | | | | | |
| The second distance from | - | No. Andrew | - 01 | | | | | | | |
| | | | | - | | Income of Parameters | | | | |
| 1 mar at | | | | | | | 121 | | | |
| Common Cale Summer State | 1.1 | 1 | | -3 | 112 | - | | | | |
| house have | - 1 • | | _1 | -* | | | | | | |
| Transmitting . | | 1 - 1 - 1 | | - N. | | | | | | |
| Provide in the second statement of | 1 | | - 10 m | -4- | 1.2 | | | | | |
| land the second se | | | 1. | - 8 | | | | | | |
| Name Artes | | 1000 | _ | _ | | | | | | |
| Conception in the local division of the loca | | | | | | | | | | |
| And the second s | | | | | | | | | | |
| And a second second second | 1.0.2 | | | | | | | | | |
| Construction of the local division of the lo | In the second | and some | - 2 | - 3 | 1000 | | - 10 | | | |

Each alternative underwent a numerical evaluation. The alternatives were then compared to each other by applying a ranking from most preferred to least preferred based on the level of anticipated impact for each factor.
| Evaluation of Alternative Solutions | | | | | | |
|--|---------|-----------------------|--------------|----------|----------------|---|
| 141.81 | 0010100 | CATTORN (SAD | A MARCHITCH. | 4 () | | Stand Palance |
| Terroran Station from | and a | | | | Aurente C. | |
| | and and | | | | | Inter Presson |
| 1 married | | | - | | | |
| Contract of the Association | | 1 | | | | Concession of the |
| Print Party 1 | | | | - * | | |
| reasoning . | • | 4 | | - 16 - C | | |
| Provide in the second state of the second | 1 | | | - 4 - | | |
| land the sector | | | 1 | - 8 - | | |
| Address Australia | | and the second | | | | |
| Concession of the local division of the loca | : | - 5 - | | | - | |
| Television . | | | | | | |
| Statute and statute and state | 1.0.2 | | - 1 - I | | | 1 I I I I I I I I I I I I I I I I I I I |
| 100.00.001.001.001 | | | - 2. | 1 | and the second | |
| Colors - other sections - the section of the | | and the second second | | | | |

A graphical representation of the numerical evaluation is shown to best illustrate the results of the analysis. A full pie represents the greatest anticipated impact, and therefore the least preferred. The empty pie represents the least anticipated impact, and therefore most preferred.



The table provides a summary of the Evaluation of Alternatives as an average range under each environment category.

The outcome of the evaluation illustrates that Alternative 5 received the most favourable overall ranking when compared to the other alternatives.



Alternative 5 has been selected as the preliminary preferred alignment for the Baden Sanitary Trunk.

The alignment will improve current sanitary services to the existing Baden residents, reduce operations and maintenance costs and provide sanitary services to the future growth areas with minimal impacts to the natural, social, cultural, built and economic environments.



This alternative consists of replacing the local sanitary sewer along Christian Street, installing a new trunk sanitary sewer along Christian Street,

through the open fields adjacent to the Baden Creek Tributary 'A' and Baden Creek, and upgrading the sewer from the intersection of Gingerich Road and Foundry Street to the Baden WWPS. A complete reconstruction of Christian Street will be required.



The timeline of this study is provided on this slide. Following the comments received from this PIC, the Project Team will confirm the preferred alternative. The Municipal Class Environmental Assessment process, including the input from the public, agencies, Indigenous communities and stakeholders will be documented in a Project File Report. Once council endorsement is received, the Project File Report will be made available for public review and comment for a period of 30-days following the Notice of Completion. At this time, it is anticipated that the Notice of Completion and the Project File Report will be available in the Winter of 2022.



Thank you for participating in the Baden Trunk Sanitary Sewer Class (EA) Online Public Information Centre (PIC).

Please direct any comments, questions or suggestions relevant to this study to either Jeff Molenhuis with the Township of Wilmot or to Dot Roga with MTE Consultants by October 11, 2022.

This Public Information Centre is relying on web-based communications, please feel free to fill out the comment sheet located on this website. Commenting period for this Public Information Centre will extend to Tuesday Oct 10, 2022. All comments received will be collected under the Environmental Assessment Act and, with the exception of personal information, will become part of the public record.



Advestration Centre: 400 Clock Instit, HD Inte 229 Cambridge, CN N18 5986

Phone 310 423 2011. Tol: Free: 1-888-832-4122. Fax: 119-621-4844. www.gtasubterr.ca

January 29, 2021

Jeff Molenhuis Director of Public Works and Engineering Township of Wilmot 60 Snyder's Road West Baden ON N3A 1A1 jeff.molenhuis@wilmot.ca Dot Roga Project Manager MTE 520 Bingemans Centre Drive Kitchener ON N2B 3X9 <u>droga@mte85.com</u>

Re: Municipal Class Environmental Assessment Baden Sanitary Trunk Sewer Township of Wilmot

Dear Mr. Molenhuis and Ms. Roga,

The Grand River Conservation Authority (GRCA) has received the Virtual Public Information Centre 1 materials for the above-noted Class Environmental Assessment (Class EA). We have reviewed the proposed alternatives and can offer the following comments.

- 1. While generally permitted, GRCA policy emphasizes exploring all feasible alternative sites for public infrastructure outside of flooding and erosion hazards. If alternatives 4 or 5 are chosen, a scoped environmental impact study (EIS) will be required to demonstrate, where unavoidable, intrusions on significant natural features or hydrologic / ecological functions are minimized, and the proposed design will adequately restore and enhance features and functions. The design should also consider maintenance and emergency repair requirements, and ensure those activities can also minimize intrusions into features / functions.
- 2. Any part of the preferred alternative in the flooding hazard (e.g. watercourse crossings) should ensure existing grades are generally maintained, and new hydraulic crossing obstructions are minimized.

- 3. Consideration should also be given to potential erosion from Baden Creek, and demonstrate that there will be no impacts on existing and future slope stability.
- 4. Detailed design plans will be requested subsequent to the Class EA process in support of a GRCA permit. An erosion and sediment control plan will also be required, which should include any isolation works.

Please continue to include us on the mailing list for stakeholder consultation, and the GRCA appreciates the opportunity to comment on the preferred alternative. If you have any questions or require additional information, please contact me at 519-621-2763 ext. 2292 or <u>theywood@grandriver.ca</u>.

Sincerely,

Trevor Heywood Resource Planner Grand River Conservation Authority



Adventeration Centre: 400 Clyin/Instit, HD Inix 729 Centerings, ON NUMBER

Phone 318 423 2001. Tol free: 1-888-855-4122. Fee: 115-623-4884. www.gtanabier.ca

September 28, 2022

Jeff Molenhuis Director of Public Works and Engineering Township of Wilmot 60 Snyder's Road West Baden ON N3A 1A1 jeff.molenhuis@wilmot.ca

Dot Roga Project Manager MTE 520 Bingemans Centre Drive Kitchener ON N2B 3X9 <u>droga@mte85.com</u>

Re: Municipal Class Environmental Assessment Baden Sanitary Trunk Sewer Township of Wilmot

Dear Mr. Molenhuis and Ms. Roga,

The Grand River Conservation Authority (GRCA) has received the Draft Environmental Study Report Appendices for the above-noted Class Environmental Assessment (Class EA). We have reviewed Preferred Alternative 5 and can offer the following comments.

- 1. We agree with the evaluation that trenchless stream crossings will not impact Baden Creek, as long as standard erosion and sediment control measures are implemented at the drill pits.
- 2. The evaluation notes impacts to the floodplain may occur. We do not anticipate impacts to the floodplain if existing grades are restored. If significant grade changes are anticipated, please consult with us early during detailed design.
- 3. The final Class EA and/or detailed design process should:
 - a. Identify and confirm the size of the wetland features and properly screen them for compliance with GRCA Policy 8.4.4 / 8.4.5;
 - b. Identify and interpret the proposed construction sequencing and techniques proposed to install the trunk sewer.

4. A GRCA permit will be required prior to construction. Detailed design and erosion / sediment control plans are requested in support of an application.

We trust this information is of assistance. If you have any questions or require additional information, please contact me at 519-621-2763 ext. 2292 or <u>theywood@grandriver.ca</u>.

Sincerely,

Trevor Heywood Resource Planner Grand River Conservation Authority



Map Centre (UTM NAD83 z17): 526,866.18 4,805,181.89

This map is not to be used for navigation | 2020 Ortho (ON)



60 Snyder's Road West, Baden, ON N3A 1A1

Public Works and Engineering Services Department T: 519-634-8444 F:519-634-5522

To Kim Hallman

We are in receipt of your letter dated January 9, 2023, regarding the Baden Trunk Sanitary Main Project.

We will attempt to address your concerns within this letter, however, if at any time you would like a more fulsome discussion on the proposed project and the problem statement, we would be more than happy to schedule a meeting to elaborate.

You note, no information has been presented to show how the preferred alternative addresses the capacity issue for growth.

As per the introduction of the Baden Trunk Sanitary Sewer Class EA,

The Wastewater Servicing Study for the Growth Areas of Baden and New Hamburg, completed in 2014 (CRA, 2014), revealed that the existing sanitary infrastructure did not support existing peak flows and lacked capacity and depth for the planned growth within the Settlement of Baden.

At the time of the 2014 study, the report recommended a preferred alternative that would have seen a trunk sanitary sewer installed on Sandhills Road (Township Road 16) towards the Baden WWPS, and a second trunk sewer on Snyders Road East towards New Hamburg.

Changes to the Provincial Policy Statement (2020), the Places to Grow Plan (2020), the amended Regional Official Plan (2015), and the Township of Wilmot Official Plan (2019), has prompted the Township to re-evaluate the wastewater servicing strategy to better service existing residents and accommodate the anticipated future growth areas for the Settlement of Baden.

The new line being proposed within this Environmental Assessment will be sized in such away to accommodate the flows from the proposed growth area. Presently the flows from Baden are directed to the existing Baden WWPS. Plans are underway with the region, and the Township to address the capacity at the pumping station to direct flows ultimately to the New Hamburg Waste Water Treatment Plant.

Of the Five proposed alternatives presented within the Baden Trunk Sewer EA, the selected Alternative 5 represents the most comprehensive solution that will provide long-term servicing for the largest anticipated growth area possible within Baden while also maintaining servicing for the existing

This information is available in accessible formats upon request

community. Preliminary sizing of the trunk sanitary sewer takes into account all existing drainage to the Baden wastewater pumping station (WWPS) (approximately 275 hectares) and accepts potential future drainage from approximately 380 hectares, including residential, commercial, institutional, and industrial flows.

In regard to the alignment of the proposed trunk sanitary sewer near the existing Baden Creek, we note that sanitary trunk sewers are often laid at the lowest area to maximize the gravity flows to this trunk sewer while minimizing the depth and costs for installing these services. The Ministry of the Environment has stringent requirements for the construction of sanitary sewers to prevent exfiltration of sanitary sewage. Further, while the trunk sewer will be in close proximity to the existing watercourse, it will be located away from the watercourse for ability to maintain the infrastructure as well as provide sufficient separation from any environmentally sensitive areas.

In regard to the construction on Christian Street, we note the proposal calls for a 450mm sanitary trunk sewer at a significant depth. Through detailed design we will be looking at ways to minimize the impacts on local residents during construction including investigating trenchless technologies for these works. We do note that reconstruction of Christian Street in a fulsome manner would be expected to take several months, however, these timelines and disturbances can be managed and minimized through proper contract administration and contract act requirements. These will all be investigated through the detailed design portion of this project.

In summary, we have taken a wholesome evaluation of the existing sanitary capacity constraints within the community and the proposed alternative will accommodate the sanitary growth for the foreseeable future within the current urban boundary. We also note, similar capacity and planning investigations are being undertaken for the New Hamburg growth area.

Again, if you would like to discuss this more fully, please do not hesitate to contact me at the numbers provided below.

Sincerely

Ken VanderWal P. Eng Manager of Engineering Infrastructure Services Township of Wilmot Office: 519-634-8519 x. 239 Cell: 519-778-0449

Cc: Jeff Molenhuis Township of Wilmot Dot Roga MTE

This information is available in accessible formats upon request

Wilmot is a cohesive, vibrant and welcoming countryside community

January 9, 2023

To whom it may concern,

I received your Notice of Study Completion for the Baden Sanitary Trunk Main project in our mailbox around December 9, 2022. I am writing to provide my feedback on this report within the 30 days as requested.

After watching the online presentation and reviewing the information provided on the Township website, I don't agree with any of the proposed alternatives. The problem statement indicates that the existing infrastructure lacks capacity and depth for future planned growth. No information has been presented to show how this preferred alternative addresses the capacity issue or future growth. We know several new housing developments are under consideration, and you can see these areas have been captured in the Primary Study Area on slide 5 of the presentation, but are we to believe that this one new line will support all of this new growth? This report proposes to pump sewage all the way across town to the Baden Pumping station then to New Hamburg for treatment, even for all the new developments being considered. Are we sure this solution will adequately support the future planned growth? The preferred alternative seems a bit short sighted and lacks focus on managing capacity for future growth.

I am also concerned that we are honestly considering running sewage lines along or near any type of waterway, either creek, tributary or pond. This does not sound like a good idea, and a disaster just waiting to happen. If a pipe broker or worse, we could disrupt the entire water system which we all rely upon. Are we really going to push construction through and potentially compromise our community and our surrounding neighbors? Has there been any sort of assessment done to determine the level of risk associated with running these pipes along a waterway to the pumping station? I don't believe any of these details were provided in this report or presentation. As we continue to lose agricultural land to new housing developments, I think it's imperative that council and leaders work to preserve and protect the habitats and wildlife who call this area home. Yes, the report clearly articulates that running a new line down Christian St would be the easiest option, but is this really the best decision for our community in the long term?

I am also a resident of Christian Street, and I will go on record to say that I do not support a 'full reconstruction' of my street, because I don't feel this preferred alternative addresses the problem. Although there were no details provided as to what a total reconstruction would mean, I can assume it will be hugely disruptive, an inconvenience to every resident on the street and as previous construction projects in Baden have shown us, it will probably not be completed in a timely manner. I certainly do want to live in a construction zone for months, especially if this will not be a sound long term investment in our infrastructure. I also work at home and I believe it will be extremely challenging to do my job living in a construction zone.

My husband and I moved to Baden over a decade ago because it was a small community. We picked this street because it was quiet and not busy. We are not opposed to new developments coming into the Township but we are concerned that they're not being planned or developed in a

smart way that will benefit the community moving forward. The main street has been under construction for several years now and it is still not fully done. How will our Township handle new proposed developments especially since there is a known lack of infrastructure in place to support them. I expect a tremendous amount of disruption at Christian St and Snyder's Rd for many years to come due to these developments, but I hope this solution isn't being pushed through without proper consideration and the long term plan for Baden.

In summary, I would urge this council to reconsider the proposed recommendations and take a hard look at the long term implications. Does this preferred alternative truly address the problem statement, or will this project simply need to be redone in a few years time because proper due diligence and planning was not done properly. I know I am only one individual in this community, but please know that your decision will have a direct impact on me and our household.

Thank you for your time. Warm Regards, Kim Hallman



| Ministry of the Environment, Conservation and Parks | Ministère de l'Environnement, de la Protection de la nature et des Parcs | | | |
|--|--|--|--|--|
| Environmental Assessment Branch | Direction des évaluations environnementales | | | |
| 1 st Floor | Rez-de-chaussée | | | |
| 135 St. Clair Avenue W | 135, avenue St. Clair Ouest | | | |
| Toronto ON M4V 1P5 | Toronto ON M4V 1P5 | | | |
| Tel. : 416 314-8001 | Tél. : 416 314-8001 | | | |
| Fax.: 416 314-8452 | Téléc. : 416 314-8452 | | | |

January 25, 2023

Ken VanderWal Manager of Engineering – Infrastructure Services Township of Wilmot <u>ken.vanderwal@wilmot.ca</u>

(Via Email Only)

Dot Roga Project Manager MTE Consultants Inc. <u>droga@mte85.com</u>

Re: Baden Sanitary Trunk Main Township of Wilmot Municipal Class Environmental Assessment – Schedule B Project Review Unit Comments – Final Project File Report

Dear Ken VanderWal and Dot Roga,

Thank you for providing the ministry with an opportunity to comment on the final Project File Report (Report) for the above noted Class Environmental Assessment (EA) project. Our understanding is that in order accommodate current and future growth in the area, the Township of Wilmot (the proponent) has determined that the preferred alternative, Alternative 5, consists of replacing the local sanitary sewer and installing a new trunk sanitary sewer along Christian Street, through the open fields adjacent to the Baden Creek Tributary 'A' and Baden Creek, and upgrading the sewer from the intersection of Gingerich Road and Foundry Street to the Baden WWPS. The Ministry of the Environment, Conservation and Parks (ministry) provides the following comments for your consideration.

General

Please note schedule B projects require the completion of a Project file Report rather than an Environmental Study Report. Please refer to section A.4.1 Schedule B project file of the MECA available here: <u>A.4.1 SCHEDULE B PROJECT FILE (municipalclassea.ca)</u>

Consultation Records

Any follow-up correspondence should be included in the final Project File Report.

Indigenous Consultation

Further to any follow-up during the review period for the EA, the proponent should continue reaching out to all communities previously engaged if there any substantial changes to the project/process or if they are applying for subsequent permits from the ministry that may be of interest or concern to communities.

Species at Risk

Please note that it is the responsibility of the proponent to ensure that Species at Risk (SAR) are not killed, harmed, or harassed, and that their habitat is not damaged or destroyed through the proposed activities to be carried out on the site. If the proposed activities cannot avoid impacting protected species and their habitats, then the proponent will need to apply for an authorization under the Endangered Species Act (ESA). As is noted in the Report, if the proponent believes that their proposed activities are going to have an impact or are uncertain about the impacts, they should contact <u>SAROntario@ontario.ca</u> to undergo a formal review under the ESA.

Climate Change

Climate change considerations have not been documented in the Project File Report. The document "Considering Climate Change in the Environmental Assessment Process" (Guide) (www.ontario.ca/page/considering-climate-change-environmental-assessment-process) is now a part of the EA program's Guides and Codes of Practice. The Guide sets out the ministry's expectation for considering climate change in the preparation, execution and documentation of environmental assessment studies and processes. The guide provides examples, approaches, resources, and references to assist proponents with consideration of climate change in EA. The proponent should review this Guide in detail. The ministry expects proponents of Class EA projects to:

- Consider the project's expected production of greenhouse gas emissions and impacts on carbon sinks (climate change mitigation), as well as resilience or vulnerability of the undertaking to changing climatic conditions (climate change adaptation).
- Include a discrete section in the ESR detailing how climate change was considered in the EA.

 How climate change is considered can be qualitative or quantitative in nature and should be scaled to the project's level of environmental effect. In all instances, both a project's impacts on climate change (mitigation) and impacts of climate change on a project (adaptation) should be considered.

Groundwater

Current geotechnical conditions are adequately described in the report. The hydrogeological assessment provided an analysis with respect to the impacts on existing shallow flow(baseflow) and interflow patterns relative to surface water features and groundwater discharge areas. The recommended alternative 5: Christian Street and Baden Creek has minimal groundwater interaction at the stream crossing of ephemeral Tributary A and the Baden. The groundwater potentially discharged to Baden Creek (BH2, BH8, & BH13) was observed at depths of 0.25 mbgs.

MW101-21 (clayey silt), MW102-21 (silt and sand), and MW104-21(clayey silt) were drilled along Alternative 5. A shallow sand layer is present between 5.6 and 6.1 mbgs at MW102-21, which is below the proposed sanitary sewer installation of 5.4 mbgs and may indicate higher dewatering volumes at this location.

If dewatering is required, it may impact groundwater inflow to Baden Creek and Tributary A resulting in lower baseflow and an increase in water temperature. However, it can occur with no impacts if done during dry conditions and/or by securing an easement from OK Tire owners to mitigate the number of crossings of Baden Creek and the potential impacts associated with the construction along Gingerich Road and Foundry Street.

The Site is connected to the municipal drinking water system and there were no active PTTW observed within 500 m according to the database. Therefore, there are no private water users that may be impacted by dewatering requirements during construction.

A hydrogeological/geotechnical investigation during the detailed design stage will determine dewatering requirements during construction and will depend on the depth of the Baden Trunk final design. This study will establish if a Permit to Take Water (PTTW) is required or if the registration on the Environmental Activity and Sector Registry (EASR) is sufficient.

If the construction of this project requires groundwater dewatering exceeds 50,000 L/d and is <400,000 L/d, the prescribed activity will be registered as an Environmental Activity and Sector Registry (EASR) regulation O. Reg. 63/16. If EASR is required, a calculation of the amount of groundwater that is expected to be taken from the area of influence of each dewatering pit, and from any overlapping areas of influence should be provided to evaluate the level of groundwater impacts. As well, as a notification protocol that identifies water users that may be impacted, including a written notice with details of the water taking (time, location, etc.) and a contingency plan with measures to mitigate identified risks.

If the groundwater dewatering is more than 400,000 L/d, a Permit to Take Water (PTTW) should be obtained (Sections 34 and 98 Ontario Water Resources Act R.S.O. 1990 and Water Taking Regulation O. Reg. 387/04). If a PTTW is required, a hydrogeological environment assessment should be provided in detail to allow for a reasonable evaluation of the level of groundwater impacts that might occur as a result of the undertaking during sanitary trunk installation.

If a PTTW is required, an investigation should be conducted during the detailed design stage regarding potential interference due to construction with other water users along this project. The hydrogeologic investigation includes drilling and installation of monitoring wells, streambed mini-piezometers, groundwater levels, and water quality monitoring (O. Reg. 169/03: Ontario Water Quality Standards). This investigation should include the related monitoring and mitigation plan regarding project interference with other water users within the zone of influence of this project. As well, as impact prediction models, mitigation measures and monitoring, and compliance reporting if they are applicable.

If a PTTW is required, an environmental assessment report prepared by a qualified professional (PGO or equivalent) and a geotechnical note prepared by a geotechnical engineer should be included in the supporting documentation of the PTTW application. The geotechnical note should address potential surrounding structural damage due to settlement from the required groundwater taking and the proposed monitoring/contingency and mitigation plan.

If a PTTW is required, MECP requires a discussion of the potential impacts on the surrounding natural environment and adjacent waterbody feature(s), any risks posed to nearby structures from subsidence, and the potential for contaminated groundwater migration from construction dewatering (O. Reg. 153/04: Records of Site Conditions – Part XV.1 of the Act). PTTW applications must also provide details regarding the proposed discharge plan and discuss how the dewatering effluent/surface water (i.e., rain) will be managed and treated to meet water quality criteria based on the final discharge location (i.e., sanitary sewer or natural environment).

Excess Material and Waste

- Activities involving the management of excess soil should be completed in accordance with O. Reg. 406/19 and the ministry's current guidance document titled *Management* of Excess Soil – A Guide for Best Management Practices, available online at ontario.ca/page/management-excess-soil-guide-best-management-practices.
- waste generated during construction must be disposed of in accordance with ministry requirements.

Thank you for circulating this Report for the ministry's consideration. Should you or any members of your project team have any questions regarding the material above, please contact me at <u>joan.delvillarcuicas@ontario.ca</u>

Sincerely,

Joan Del Villar C Regional Environmental Planner Project Review Unit, Environmental Assessment Branch Ontario Ministry of the Environment, Conservation and Parks

Cc: Zenova Gentles, Administrative Assistant, MTE Consultants Inc



Technical Memo



Table of Contents

INTRODUCTION

- Appendix A CRA 2014 Growth Areas
- Appendix B CRA 2014 Alternative D

1.0 NATURAL ENVIRONMENT

• Appendix A Natural Environmental Assessment Report (April 2022)

Evaluation of Alternatives Natural Environment Criteria Technical Memorandum (July 11, 2022)

- Appendix B Baden Trunk EA Geotechnical Site Characterization Report (May 3, 2022)
- Appendix C Hydrogeological Assessment of Alternatives Technical Memorandum (April 26, 2022)
- Appendix D Fluval Geomorphological Assessment Technical Memorandum (August 30, 2022)

2.0 SOCIO-CULTURAL ENVIRONMENT

- Appendix A Land Use Planning Evaluation, Baden Trunk Sanitary Sewer EA Technical Memorandum (April 28, 2022)
- Appendix B Stage 1-2 Archaeological Assessment Baden Sanitary Sewer Replacement, Baden (November 8, 2022) Supplementary Documentation- Stage 1-2 Archaeological Assessment, Baden Sanitary Sewer Replacement, Baden (November 8, 2022)
- Appendix C Cultural Heritage Evaluation Report (CHER) (December 2021)

3.0 TRANSPORTATION/MUNICIPAL SERVICES & UTILITIES

4.0 FINANCIAL

Appendix A Cost Estimate

5.0 TECHNICAL

6.0 SELECTION OF THE PREFERRED ALTERNATIVE & EVALUATION MATRIX

• Appendix A Evaluation Matrix

Figures

- Figure 1.0 Primary Study Area
- Figure 2.0 Detailed Study Area
- Figure 3.0 Existing Sanitary Sewer System and Drainage Areas
- Figure 4.1 Alternative 1 2014 Environmental Assessment referred Alternative (Do Nothing)
- Figure 4.2 Alternative 1 Sanitary Drainage Area
- Figure 5.1 Alternative 2 Snyder's Road West to Foundry Street
- Figure 5.2 Alternative 2 Sanitary Drainage Area
- Figure 6.1 Alternative 3 Christian/Charlotta/Mill/Charles/Foundry
- Figure 6.2 Alternative 3 Sanitary Drainage Area
- Figure 7.1 Alternative 4 Christian/Charlotta/Queen and Baden Creek
- Figure 7.2 Alternative 4 Sanitary Drainage Area
- Figure 8.1 Alternative 5 Christian Street and Baden Creek
- Figure 8.2 Alternative 5 Sanitary Drainage Area



Project Name:Baden Trunk
Sanitary Sewer EATo:Jeff Molenhuis, Township of Wilmot
Bryan Bishop, Township of Wilmot
Trevor Heywood, GRCA
Kaoru Yajimi, ROW

cc: Pierre Chauvin, MHBC Paul Britton, MHBC MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENTAL ASSESSMENT Introduction

1.0 Introduction

The Township of Wilmot (Township) is responsible for the long-term operation and maintenance of the wastewater collection system for the communities of Baden and New Hamburg. In 2014, a Wastewater Servicing Study for the Growth Areas of Baden and New Hamburg was prepared for the Township by Conestoga Rovers & Associates (CRA) and can be found in **Appendix A**. The study evaluated the sanitary infrastructure needs presented in the 2011 Baden – New Hamburg Water and Wastewater Master Plan Update (BNHWWMPU), and evaluated the development options of these new growth areas. Of the four overall servicing strategies considered in the CRA study, it was concluded that Alternative D (**Appendix B**) was the preferred option. Growth since 2014 has generally followed the 2014 strategy with infrastructure being built from downstream to future growth areas. Recent studies regarding sanitary services to Foxboro and discussions with the Township have resulted in the expansion of the growth areas to include Parcel P₂ and Parcel Y₂ (**Figure 1.0**).

Changes to the Provincial Policy Statement of 2020, the Places to Grow Plan of 2020, the amended Regional Official Plan of 2015 and the updated Township of Wilmot Official Plan of 2019 has prompted an overall review of the servicing strategy for the entire Baden - New Hamburg area in support of the additional expected growth. As a result, the following three studies have been implemented:

- Morningside Trunk Sewer EA, GM BluePlan 2021
- Baden-New Hamburg Water and Wastewater System Servicing Review 2021
- Baden Trunk Class EA, MTE 2021

MTE Consultants Inc. (MTE) has been retained to complete a Schedule B, Municipal Class Environmental Assessment (EA) for a new sanitary trunk sewer to service the current and future growth of Baden. The study will include an evaluation of the alternatives based on their environmental impacts and mitigation measures, and the selection of a preferred alternative. At the end of the study, a conceptual design will be prepared by MTE. This proposed trunk sanitary sewer is herein referred to as the 'Baden Sanitary Trunk'.

The purpose of this technical memo is to summarize the five alternatives, identify their environmental, socio-cultural and economic impacts and to identify the preferred alternative as defined by the EA process.



2.0 Study Area for the Proposed Sanitary Trunk Alignment

2.1 Primary Study Area

The Primary Study Area (PSA) is located within the Settlement of Baden (**Figure 1.0**). The Urban Areas are the primary focus for population and employment growth per the Township's Official Plan. Lands located outside the urban areas are deemed countryside and represent the future expansion to the Baden Urban Area boundary. The proposed Baden Sanitary Trunk will service all areas east of Nafziger Road within the countryside line taking into consideration the future growth of Baden.

2.2 Detailed Study Area

The Detailed Study Area (DSA) (**Figure 2.0**) encompasses all alternatives except the eastern portion of Alternative 1. It is defined by the area south of Snyder's Road West, west of Foundry Street to the Baden pumping station, north of Gingerich Road and east of Christian Street including the agricultural lands it crosses. All alternative alignments will begin at the intersection of Snyder's Road West and Christian Street at a previously agreed upon sanitary sewer obvert of 353.7m⁺ for Alternative 1 and 349.40m⁺ for the remaining alternatives. They will ultimately connect to the manhole at the entrance of the Baden wastewater pumping station (WWPS), within the Foundry Street right-of-way, at an obvert of 335.92m⁺.

The DSA is comprised of various land uses within the urban boundary, agricultural fields and four prominent water features that are regulated by the Grand River Conservation Authority.

2.3 Existing Sanitary Drainage System

The Settlement of Baden is currently serviced by a two-tier wastewater system. The Township is responsible for the operations and maintenance of the gravity sewers, pressure sewers, trunk sewers, the School Block WWPS and the Baden West WWPS. The Region of Waterloo is responsible for the operations and maintenance of the Baden WWPS (**Figure 3.0**). Special coordination efforts and permissions are required for maintenance of Township sanitary infrastructure along the Snyder's Road West, Snyder's Road East and Foundry Street Regional roads.



3.0 Baden Sanitary Trunk Alignment Alternatives

The Baden Sanitary Trunk alignment alternatives were developed in accordance with the requirements of the Environmental Assessment Act by following the Municipal Class EA document (as amended in 2007, 2011, 2015 and 2020) and is being planned under Schedule 'B' of the Municipal Class Environmental Assessment.

Features common to all alternatives include a trenchless railway crossing and the section of sewer from the Baden WWPS along Foundry Street to Gingerich Road, which will require coordination and approval with the Region of Waterloo.

3.1 Alternative 1 – Modified 2014 Environmental Assessment Preferred Alternative

MTE has defined the approved Preferred Alternative D (**Appendix A**) of the CRA 2014 study as the 'Do Nothing' Alternative. It has been endorsed by the Township of Wilmot and as such, has been designated as the 'Do Nothing' alternative. This alignment documents the preferred servicing strategy at the time the CRA report was approved.

The western portion of the alignment begins at the intersection of Christian Street and Snyder's Road West, at the previously approved invert of $353.70m \pm$ (**Figure 4.1**). It continues east along Snyder's Road West to Foundry Street. The eastern portion of the alignment runs along the eastern boundary of the Activa MDS Lands north of Snyder's Road East, southwest along the railway to Foundry Street, where it connects to the western section. The sewer then runs south along Foundry Street and ultimately to the Baden WWPS. This alternative calls for the construction of a pumping station within Parcel O, the installation of approximately 1,555m of forcemain for the eastern portion of the alignment, approximately 1,800m of sanitary sewer for the western portion of the alignment ranging in depths from 3.0m to 5.0m and associated manholes.

Construction considerations for Alternative 1 include a trenchless railway crossing, a trenchless crossing of the Baden Creek Tributary B, north of Snyder's Road East, crossing the Baden Creek just upstream of the Baden Creek pond, crossing the Huehn Award Drain, upgrading existing sewers, reconstructing large portions of existing road infrastructure, and integrating it with future road construction along the Regional Roads of Snyder's Road West and Foundry Street.

Alternative 1 will service the growth areas of Parcels BB, O₁, O₂, P₁, P₂, R, S, T, U, V, Y₂ through the Nideva lands at 1012 Snyder's Road West, parts of Parcels AA₁ to AA₃ and Foxboro. Refer to **Figure 4.2** for reference to the growth area Parcels and the sanitary sewer drainage areas.

3.2 Alternative 2 – Snyder's Road West to Foundry Street

Alternative 2 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 5.1**), continues east on Snyder's Road West to Foundry Street, and south on Foundry Street to the Baden WWPS. This alignment is mainly within the existing rights-of-way of Snyder's Road West and Foundry Street. It calls for approximately 1,800m+ of sanitary sewer varying in depth from 3.0m to 9.3m and associated manholes.

Construction considerations for Alternative 2 include a trenchless railway crossing, crossing the Baden Creek just upstream of the Baden Creek pond, crossing the Huehn Award Drain, upgrading existing sewers along Snyder's Road West and Foundry Street, reconstructing large portions of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street.

The main difference between Alternative 1 and Alternative 2 is that the pump station and forcemain on Activa's lands will not be required. The deep sewer along Snyder's Road West will provide additional sanitary services to Parcel Y_1 , Parcel Z and the School Block, eliminating the need for the Waterloo-Oxfor School WWPS (**Figure 5.2**).



3.3 Alternative 3 - Christian/Charlotta/Mill/Charles/Foundry

Alternative 3 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 6.1**). It lies within the existing right-of-ways of Christian Street, Charlotta Street, Mill Street, Charles Street, and Foundry Street. This alignment requires approximately 1,985m of sanitary sewer varying in depth from 3.3m to 9.3m and associated manholes.

Construction considerations for this alignment include a trenchless railway crossing, crossing the Baden Creek just upstream of the Baden Creek pond, crossing the Huehn Award Drain, upgrading existing sewers along Christian Street, Charlotta Street, Mill Street, Charles Street, and Foundry Street, reconstructing large portions of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street. As part of the development of the 1012 Snyder's Road West Subdivision, a portion of this alternative along Christian Street will be reconstructed.

This alternative will eliminate two WWPS; the privately owned, Waterloo-Oxford District WWPS, and the Baden West WWPS owned and operated by the Township, and will additionally service Parcels Q, X, and W (**Figure 6.2**).

3.4 Alternative 4 - Christian/Charlotta/Queen and Baden Creek (Future Right-of-way to Schmidt Lands)

Alternative 4 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 7.1**). It lies within the existing right-of-ways of Christian Street, Charlotta Street, Queen Street and through Parcels AA₁, AA₂, and BB₂ to Gingerich Road, and then south on Foundry Street to connect to the Baden WWPS. This alignment requires approximately 2,095m of sanitary sewer, varying in depth from 3.0m to 9.3m and associated manholes.

Construction considerations for this alignment include a trenchless railroad crossing, trenchless crossing of the Baden Creek Tributary A, trenchless crossing of Baden Creek just north of Gingerich Road, upgrading existing sewers along Christian Street, Charlotta Street and Queen Street, reconstructing a small portion of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street. As part of the development of the 1012 Snyder's Road West Subdivision, a portion of this alternative along Christian Street will be reconstructed. The forcemain along Queen Street will remain intact.

This alternative benefits in following the natural topography of the Schmidt Lands and will eliminate the two WWPS owned and operated by the Township of Wilmot as well as servicing all identified growth areas (**Figures 7.2**).

3.5 Alternative 5 – Christian Street and Baden Creek

Alternative 4 begins at the intersection of Snyder's Road West and Christian Street at an obvert of 349.40m (**Figure 8.1**). It lies within the existing right-of-ways along Christian Street. It runs through portions of Charlotta Street and through Parcels AA₁, AA₂, and BB₂ to Gingerich Road, and then south on Foundry Street to connect to the Baden WWPS. This alignment requires approximately 1,905m of sanitary sewer varying in depth from 3.0m to 9.3m and associated manholes.

Construction considerations for this alignment include a trenchless railroad crossing, trenchless crossing of the Baden Creek Tributary A, trenchless crossing of Baden Creek just north of Gingerich Road, upgrading existing sewers along Christian Street and a small section of Charlotta Street, reconstructing a small portion of existing road infrastructure, and integrating with future road construction along Snyder's Road and Foundry Street. As part of the development of the 1012 Snyder's Road West Subdivision, a portion of this alternative along Christian Street will be reconstructed.



This alternative benefits in following the natural topography of the Schmidt Lands and will eliminate the two WWPS as identified in Alternative 3 as well as servicing all identified growth areas (**Figures 8.2**).

4.0 Assessment of Sanitary Trunk Alternatives

The Numerical Evaluation approach was selected as the preferred scoring methodology. This methodology requires a "must score" assessment of the five alternatives in a matrix form. Factors used to assess the Alternatives include:

- Natural Environment Terrestrial Ecosystems, Wetland Ecosystems, Aquatic Ecosystems, Surface Water Quality, Groundwater Resources, Surface Water Drainage
- Socio-Cultural Environment Land Use Policy, Archaeological Resources, Built Heritage, Noise
- Transportation/Municipal Services and Utilities
- Financial
- Technical

For each factor, the least preferred alternative received a score of one (1) and the most preferred alternative received a score of ten (10). All other alternatives were scored between these values with ties strongly discouraged. Those factors receiving less than a five (5) were deemed unacceptable and used to identify potential issues requiring additional mitigation. Each factor group will be analyzed and commented on to clarify differentiating factors and commonalities between the impacts of the proposed Baden Sanitary Trunk alternatives.



Wastewater Servicing Study for the Growth Areas of Baden and New Hamburg (CRA 2014)







78251-10(MEMO002)GN-WA001 MAR 26/2013



CRA 2014 - Alternative D





List of Figures

- Figure 1.0 Primary Study Area
- Figure 2.0 Detailed Study Area
- Figure 3.0 Existing Sanitary Sewer System and Drainage Areas
- Figure 4.1: Alternative 1 Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing)
- Figure 4.2: Alternative 1 Sanitary Drainage Area
- Figure 5.1: Alternative 2 Snyder's Road West to Foundry Street
- Figure 5.2: Alternative 2 Sanitary Drainage Area
- Figure 6.1: Alternative 3 Christian/Charlotta/Mill/Charles/Foundry
- Figure 6.2: Alternative 3 Sanitary Drainage Area
- Figure 7.1: Alternative 4 Christian/Charlotta/Queen and Baden Creek
- Figure 7.2: Alternative 4 Sanitary Drainage Area
- Figure 8.1: Alternative 5 Christian Street and Baden Creek
- Figure 8.2: Alternative 5 Sanitary Drainage Area



| LE | GEND |
|-------------|--|
| | |
| | EXISTING SANITARY TRUNK |
| | EXISTING SANITARY FORCEMAIN |
| 326 | EXISTING 2015 GRCA CONTOURS |
| | DRAINAGE TO BADEN SANITARY TRUNK SEWER |
| | DRAINAGE BY GRAVITY TO NAFZIGER ROAD TRUNK SEWER |
| | DRAINAGE TO FOREST GLEN WWPS |
| | DRAINAGE TO WILMOT EMPLOYMENT LANDS TRUNK SEWER |
| | LANDS NOT TO BE DEVELOPED |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| | BUTTERNUT 25m BUFFER |
| ***** | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER |
| | ARCHAEOLOGICAL 20m NO-GO BUFFER |
| ********** | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | |
| | WETLAND |
| | LOCAL 15m WETLAND BUFFER |
| | (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| *********** | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | · DETAILED STUDY AREA |
| | PROPOSED WASTEWATER PUMPING STATION |
| \bigcirc | EXISTING WASTEWATER PUMPING STATION |
| ALTEF | RNATIVES |
| 1 | 2014 ENVIRONMENTAL ASSESSMENT PREFERRED ALTERNATIVE (DO NOTHING) |
| 2 | SNYDER'S/FOUNDRY |
| 3 | CHRISTIAN/CHARLOTTA/MILL/ CHARLES/FOUNDRY |
| 4 | CHRISTIAN/CHARLOTTA/QUEEN AND BADEN CREEK |
| 5 | CHRISTIAN AND BADEN CREEK |
| | |



P:\P\48730\100\48730-100-F2.0.dwg

| LEGEND | | | | | |
|--------------|--|--|--|--|--|
| | | | | | |
| | EXISTING SANITARY TRUNK | | | | |
| | EXISTING SANITARY SEWER | | | | |
| | EXISTING SANITARY FORCEMAIN | | | | |
| 326 | EXISTING 2015 GRCA CONTOURS | | | | |
| | DRAINAGE TO BADEN SANITARY TRUNK SEWER | | | | |
| | GRCA REGULATORY FLOODPLAIN | | | | |
| | GRCA EXISTING WATERBODY | | | | |
| | BUTTERNUT 25m BUFFER | | | | |
| ****** | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER | | | | |
| | ARCHAEOLOGICAL 20m NO-G0 BUFFER | | | | |
| | TRIBUTARY 'A' 10m BUFFER | | | | |
| | BADEN CREEK 15m BUFFER | | | | |
| | HUEHN AWARD DRAIN 30m BUFFER | | | | |
| | LOCALLY SIGNIFICANT WETLAND | | | | |
| | LOCAL 15m WETLAND BUFFER | | | | |
| | POTENTIAL BAT (SAR) HABITAT | | | | |
| | GRCA EXISTING WATERCOURSE | | | | |
| | TRENCHLESS CROSSING | | | | |
| | TOWNSHIP URBAN AREA | | | | |
| | COUNTRYSIDE LINE | | | | |
| | PRIMARY STUDY AREA | | | | |
| | DETAILED STUDY AREA | | | | |
| | EXISTING WASTEWATER PUMPING STATION | | | | |
| ALTERNATIVES | | | | | |
| 1 | 2014 ENVIRONMENTAL ASSESSMENT PREFERRED ALTERNATIVE (DO NOTHING) | | | | |
| 2 | SNYDER'S/FOUNDRY | | | | |
| 3 | CHRISTIAN/CHARLOTTA/MILL/ CHARLES/FOUNDRY | | | | |
| 4 | CHRISTIAN/CHARLOTTA/QUEEN AND BADEN CREEK | | | | |
| (5) | CHRISTIAN AND BADEN CREEK | | | | |
| Γ | | | | | |






| | EGEND |
|-------------|--|
| | PROPERTY LINE |
| | EXISTING SANITARY TRUNK |
| | EXISTING SANITARY SEWER |
| | EXISTING SANITARY FORCEMAIN |
| 326 | EXISTING 2015 GRCA CONTOURS |
| | DRAINAGE TO BADEN SANITARY TRUNK SEWER |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| \bigcirc | BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION |
| | ARCHAEOLOGICAL 20m NO-GO BUFFER |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | HUEHN AWARD DRAIN 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND |
| *********** | LOCAL 15m WETLAND BUFFER |
| | POTENTIAL BAT (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | DETAILED STUDY AREA |
| 0 | PROPOSED WASTEWATER PUMPING STATION |
| • | EXISTING WASTEWATER PUMPING STATION |
| ALTE | RNATIVES |
| | 2014 ENVIRONMENTAL ASSESSMENT PREFERRED ALTERNATIVE (DO NOTHING) |
| | |



P:\P\48730\100\48730−100−F3.1.dwg





P:\P\48730\100\48730-100-MS3.dwg

| | EGEND |
|------------|--|
| | |
| 325 | EXISTING 2015 GRCA CONTOURS |
| | DRAINAGE TO BADEN SANITARY TRUNK SEWER |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| \bigcirc | BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER |
| | ARCHAEOLOGICAL 20m NO-G0 BUFFER |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | HUEHN AWARD DRAIN 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND |
| | LOCAL 15m WETLAND BUFFER |
| | POTENTIAL BAT (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| | TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | DETAILED STUDY AREA |
| | EXISTING WASTEWATER PUMPING STATION |
| ALTE | RNATIVES |
| 2 | SNYDER'S/FOUNDRY |



P:\P\48730\100\48730−100−F3.2.dwg





P:\P\48730\100\48730-100-MS3.dwg

| | GEND |
|-------|--|
| | PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOURS DRAINAGE TO BADEN SANITARY TRUNK SEWER GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER ARCHAEOLOGICAL 20m NO-GO BUFFER TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER HUEHN AWARD DRAIN 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND LOCAL 15m WETLAND BUFFER POTENTIAL BAT |
| | (SAR) HABITAT GRCA EXISTING WATERCOURSE TRENCHLESS CROSSING |
| | TOWNSHIP URBAN AREA COUNTRYSIDE LINE PRIMARY STUDY AREA DETAILED STUDY AREA |
| ALTER | EXISTING WASTEWATER PUMPING STATION |
| | |



P:\P\48730\100\48730−100−F3.3.dwg





| PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOUR INAGE TO BADEN SANITARY NK SEWER A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION ITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
|--|
| PROPERTY LINE EXISTING SANITARY TRUNK EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOUR INAGE TO BADEN SANITARY INAGE TO BADEN SANITARY A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION ITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| EXISTING SANITARY TRUNK EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOUR INAGE TO BADEN SANITARY NK SEWER A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION ITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| EXISTING SANITARY SEWER EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOUR INAGE TO BADEN SANITARY NK SEWER A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION ITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| EXISTING SANITARY FORCEMAIN EXISTING 2015 GRCA CONTOUR INAGE TO BADEN SANITARY NK SEWER A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION ITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| EXISTING 2015 GRCA CONTOUR INAGE TO BADEN SANITARY NK SEWER A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION ITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| INAGE TO BADEN SANITARY NK SEWER A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION IITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| A REGULATORY FLOODPLAIN A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION IITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| A EXISTING WATERBODY TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION IITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| TERNUT 25m BUFFER HAEOLOGICAL 50m CONSTRUCTION IITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| HAEOLOGICAL 50m CONSTRUCTION IITORING BUFFER HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| HAEOLOGICAL 20m NO-GO FER BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| BUTARY 'A' 10m BUFFER EN CREEK 15m BUFFER |
| EN CREEK 15m BUFFER |
| |
| HN AWARD DRAIN 1 BUFFER |
| ALLY SIGNIFICANT LAND |
| AL 15m WETLAND BUFFER |
| ENTIAL BAT R) HABITAT |
| A EXISTING WATERCOURSE |
| NCHLESS CROSSING |
| TOWNSHIP URBAN AREA |
| COUNTRYSIDE LINE |
| PRIMARY STUDY AREA |
| DETAILED STUDY AREA |
| EXISTING WASTEWATER PUMPING |
| |
| AIIVES |
| |



P:\P\48730\100\48730−100−F3.4.dwg









| LEC | END |
|----------------|--|
| | PROPERTY LINE |
| | EXISTING SANITARY TRUNK |
| | EXISTING SANITARY SEWER |
| | EXISTING SANITARY FORCEMAIN |
| 326 | EXISTING 2015 GRCA CONTOURS |
| | SANITARY DRAINAGE AREA BOUNDARY |
| | FLOW DIRECTION |
| | DRAINAGE BY GRAVITY TO NAFZIGER ROAD TRUNK SEWER |
| | DRAINAGE TO FOREST GLEN WWPS |
| | DRAINAGE TO WILMOT EMPLOYMENT LANDS TRUNK SEWER |
| | LANDS NOT TO BE DEVELOPED |
| | DRAINAGE TO FOUNDRY STREET TRUNK SEWER |
| | GRCA REGULATORY FLOODPLAIN |
| | GRCA EXISTING WATERBODY |
| (\mathbb{C}) | BUTTERNUT 25m BUFFER |
| | ARCHAEOLOGICAL 50m CONSTRUCTION MONITORING BUFFER ARCHAEOLOGICAL 20m NO-G0 BUFFER |
| | TRIBUTARY 'A' 10m BUFFER |
| | BADEN CREEK 15m BUFFER |
| | 30m BUFFER |
| | LOCALLY SIGNIFICANT WETLAND |
| | LOCAL 15m WETLAND BUFFER |
| | POTENTIAL BAT (SAR) HABITAT |
| | GRCA EXISTING WATERCOURSE |
| | TOWNSHIP URBAN AREA |
| | COUNTRYSIDE LINE |
| | PRIMARY STUDY AREA |
| | DETAILED STUDY AREA |
| | PROPOSED WASTEWATER PUMPING STATION |
| 0 | EXISTING WASTEWATER PUMPING STATION |
| ALTE | RNATIVES |
| 5 | CHRISTIAN AND BADEN CREEK |
| | |



P:\P\48730\100\48730-100-MS3.dwg

Technical Memo 1 Natural Environment



Project Name: Baden Trunk Sanitary Sewer EA Jeff Molenhuis, Township of Wilmot

- To: Bryan Bishop, Township of Wilmot Trevor Heywood, GRCA Kaoru Yajimi, ROW
- cc: Pierre Chauvin, MHBC Paul Britton, MHBC

MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENTAL ASSESSMENT Natural Environment Assessment of Alternatives

1.0 Natural Environment

The Natural Environment Factor was assessed for each alternative based on the number of sensitive environmental features crossed and/or their potential need for mitigative measures during construction and post-construction operations.

Natural Resource Solutions Inc. (NRSI) was retained on behalf of MTE to complete a Natural Environmental Assessment (NEA). The report, *Baden Trunk Sanitary Sewer Environmental Assessment*, *(April 2022)*, can be found in **Appendix A**. The study details the existing natural features and their functions, including assessments of their ecological significance and sensitivity as it pertains to the terrestrial ecosystems, wetland ecosystems and aquatic ecosystems of the DSA and the potential impacted areas of the PSA. NRSI prepared a technical memorandum, *Evaluation of Alternatives – Natural Environment Criteria* (July 11, 2022) (**Appendix A**), that summarizes the potential for direct or indirect impacts to the natural environment features for each alternative and their scoring is listed below for sections 1.A - 1.C.

MTE assessed the Factor Categories pertaining to 1.D - Surface Water Quality, 1.E - Groundwater 1.F - Resources and Surface Drainage.

1.A Terrestrial Ecosystems

Vegetation Communities and Flora

<u>Alternative 2</u> received a maximum score of 10. <u>Alternative 3</u> received a score of 8. <u>Alternative 5</u> received a score of 5. <u>Alternative 1</u> received a score of 4. <u>Alternative 4</u> received a minimum score of 1.

Wildlife Habitat

Alternative 5 received a maximum score of 10.

<u>Alternative 2</u> received a score of 6.

<u>Alternative 3</u> received a score of 5.

<u>Alternative 4</u> received a score of 4.

<u>Alternative 1</u> received a minimum score of 1.



1.B Wetland Ecosystems – refer to NRSI technical memo for justification of scoring

<u>Alternative 5</u> received a maximum score of 10. <u>Alternative 3</u> received a score of 9. <u>Alternative 2</u> received a score of 8.

<u>Alternative 4</u> received a score of 4.

Alternative 1 received a minimum score of 1.

1.C Aquatic Ecosystems – refer to NRSI technical memo for justification of scoring

<u>Alternative 5</u> received a maximum score of 10. Alternative 4 received a score of 7.

<u>Alternative 2</u> received a score of 4.

Alternative 3 received a score of 4.

Alternative 1 received a minimum score of 1.

1.D Surface Water (Quality - Impact to Water Quality / Thermal Regime)

Water Quality (TSS-oil/gas, Temperature)

This Factor Group evaluated the potential impacts to water quality due to the Baden Sanitary Trunk construction. Factors identified to alter water quality include Total Suspended Solids (TSS), deleterious substances such as oil/gas and temperature.

The impacts stem from the possibility of additional sediments being deposited into nearby watercourses/bodies during excavation, the type of sediment (road substrate, natural soil), and the proximity of construction vehicles to watercourses/bodies with respect to gas/oil leaking and refueling during the construction phase.

Increases in TSS and turbidity results in reduced photosynthesis, reduced oxygen, increased water temperature, carries toxins, and promotes erosion and resuspension of sediment within the stream bed. Rapid sedimentation can interfere with groundwater inflow to surface water features as well as burying benthic organisms important to the lifecycle of Baden Creek.

Where dewatering is required, design Alternatives will provide enhanced TSS removal before entering back into the stream. Changes to flow regime due to water rerouting can increase turbulence and contribute to particle resuspension leading to further water quality issues. Monitoring of flow rates may be required where rerouted water re-enters Baden Creek.

The following categories were scored based on the number of sensitive environmental feature crossings, both trenchless and open-cut, and/or the potential need for mitigative measures.

It should be noted that the potential impacts are limited to the construction phase of the project. Scoring was based on the proximity to Baden Creek pond, the Huehn Award Drain, Baden Creek, the Baden Creek Tributary A, and the Baden Creek Tributary B while taking into account the sensitivity of the water feature.

<u>Alternative 5</u> received a maximum score of 10. This alternative scored the highest because of the impact to Baden Creek and its tributaries. There are a total of 2 stream crossings. The combined trenchless stream/railway crossing that occurs at the Baden Creek Tributary A (**Figure 8.1**), is categorized as an ephemeral/intermittent stream with no evidence of groundwater interaction. Therefore, little to no water quality impacts are expected to occur during construction, and no dewatering would likely be required. The second trenchless stream crossing located north of Gingerich Road, is not anticipated to impact the Creek, and in turn, the water quality will not be affected.



<u>Alternative 4</u> received a score of 8. This alternative is similar to Alternative 5 in that it crosses the intermittent Baden Creek tributary A. However, it crosses further downstream, just before the confluence with Baden Creek. Refer to **Figure 7.1** for alignment location. This trenchless stream crossing is located within a reach that has been identified to have possible groundwater interaction. Therefore, increased TSS contributions may occur during construction due to dewatering. The trenchless crossing north of Gingerich Road is not anticipated to impact the Creek, and in turn, the water quality will not be affected.

<u>Alternative 2</u> received a score of 3. This alternative has two open-cut stream crossings located along Foundry Street. One stream crossing is next to the Baden Creek pond and the other at the Huehn Award Drain. Refer to **Figure 5.1** for alignment location. Flow rates (while rerouting during the dewatering process) would have to be established to prevent increased flow as well as provide sediment control during the dewatering process. The potential for contaminated sediments from the road (asphalt) and oil/gas getting into the surface water also exist. Mitigative measures would need to be implemented.

<u>Alternative 3</u> received a score of 2. This alternative is similar to Alternative 2 with the exception that the proposed sewer at the stream crossings along Foundry Street are deeper than Alternative 2. Refer to **Figure 6.1** for alignment location.

<u>Alternative 1</u> received the minimum score of 1. Aside from the same two (2) open-cut stream crossings as indicated in Alternatives 2 and 3, there is an additional trenchless stream crossing further upstream of Baden Creek, north of Snyder's Road East, with a potential to impact the water quality of the Locally Significant Wetland as well as Baden Creek. Refer to **Figure 4.1** for alignment location. This upstream reach has the potential to further increase TSS levels to Baden Creek and impacting its water quality.

Thermal Regime

The thermal regime of Baden Creek is based on field measurements of water temperature, taking into consideration the amount of vegetative cover and groundwater inflow. The Township recognizes that groundwater discharge to streams is a vital hydrogeological function contributing to Baden Creek's thermal regime.

Baden Creek is considered to have a warmwater thermal regime with evidence of groundwater inflow. pH levels provide further evidence of warm freshwater habitats. The Baden Creek Pond is a candidate for significant wildlife habitat for SCC Snapping Turtle with evidence of groundwater inflow as well as providing wintering conditions for the snapping turtle. Refer to the NEA report in **Appendix A** for more information.

The Huehn Award Drain, located along Foundry Street, is classified as a coldwater watercourse by the GRCA with evidence of groundwater inflow. The coldwater contribution maintain cooler water temperatures during summer months within downstream sections of Baden Creek from the confluence past the BWWPS.

The Baden Creek Tributary A is an ephemeral stream supporting seasonal flows and originating from the stormwater management facility, northwest of the detailed study area. Baseflow appears to be provided by tile drains with no signs of groundwater input.

An increase in TSS increases turbidity and can lead to an increase in water temperature as it absorbs additional heat from the sun. This can also cause dissolved oxygen levels to drop below the thermocline, creating hypoxic conditions. Scoring is based on the greatest potential for an increase to the thermal regime.

<u>Alternative 5</u> received a maximum score of 10. This alternative scored the highest because it does not impact the sensitive water features of the Baden Creek pond and the Huehn Award Drain (**Figure 8.1**). The northern portion of the alignment is along the intermittent Baden Creek Tributary A, where threnchless construction will have no impact on the thermal regime of significant water features due to



no groundwater interaction. The proposed Baden Creek trenchless crossing, north of Gingerich Road, are not anticipated to impact the thermal regime due to no anticipated dewatering.

<u>Alternative 4</u> received a score of 8. This alternative is similar to southern portion of Alternative 5 however, the area between Queen Street and the trenchless crossing of Tributary A, adjacent to Baden Creek, may require dewatering resulting in an increase in water temperature. The increased flow rates from rerouting may also lead to additional suspended sediments and turbidity, further impacting the thermal regime. Impacted vegetation would reduce vegetative cover and not be able to assist in regulating the temperature of Baden Creek.

<u>Alternative 2</u> received a score of 3. Open-cut trenching along Foundry Street at the Baden Creek pond and the Huehn Award Drain would require water rerouting and dewatering during construction resulting in a possible increase to the water temperature. These activities along with the disruption of groundwater inflow of cold water would alter the thermal regime of the pond and Baden Creek, affecting the vegetation and cool water fish habitat. The increased flow rates from rerouting may also lead to additional suspended sediments and turbidity, further impacting the thermal regime. Impacted vegetation would reduce vegetative cover and not be able to assist in regulating the temperature of Baden Creek.

<u>Alternative 3</u> received a score of 2. This alternative is similar to Alternative 2, but the proposed sewer is deeper, further increasing potential impacts to the thermal regime during construction.

<u>Alternative 1</u> received the minimum score of 1. This alternative is the same as Alternative 2 except that there is an additional stream crossing north of Snyder's Road East (**Figure 4.1**) that can lead to increased flow and sediments, thus altering the temperature upstream of the Baden Creek pond.

Chlorides

Chlorides are not a significant issue. All Alternatives are scored as 10.

1.E Groundwater Resources

This Factor Group assesses the impacts related to groundwater resources based on the following factors:

- Groundwater Recharge areas: Encroachment into the Regional Recharge Area.
- Groundwater Quality: Impacts to Vulnerable Areas (IPZ, SGRA, HVA and WHPA):
 - IPZ Intake Protection Zone;
 - o SGRA Significant Groundwater Recharge Areas;
 - HVA Highly Vulnerable Aquifers; and
 - WHPA Well Head Protection Areas.
- Shallow Groundwater Movement: Impacts to existing shallow flow and interflow patterns relative to surface water features and groundwater discharge areas.

For the purpose of this evaluation, the Groundwater Resources Factor Group will be comprised of Groundwater Regional Recharge Areas, Groundwater Quality - as it pertains to the Clean Water Act, and Shallow Groundwater Movement.

MTE performed a geotechnical site characterization investigation to determine the general soil and groundwater conditions within the DSA. In addition to the extensive historical geotechnical data, additional borehole samples were taken within the DSA to further characterize soils and groundwater condition. The geotechnical report, *Baden Sanitary Trunk EA, Geotechnical Site Characterization (May 3, 2022)*, is provided in **Appendix B.** This report was used to validate the hydrogeological assessment of the DSA and is summarized in the *Hydrogeological Assessment of Alternatives - Technical Memo* (April 26, 2022) (**Appendix C**) prepared by MTE.

Engineers, Scientists, Surveyors.



The scoring was based on the trunk sewer being located in or in close proximity to regional recharge areas. The scoring was not based on the potential for infiltration and/or groundwater interference as those elements are covered under categories of shallow groundwater movement and shallow groundwater quality.

Groundwater Recharge Areas (Regional Recharge Areas)

This Factor Group was scored according to the proximity and possible encroachment to regional recharge areas due to dewatering. Baden Creek and the Huehn Award Drain both connect to the regional recharge area to the east. Refer to Figure 5 of the *Hydrogeological Assessment of Alternatives - Technical Memo* (**Appendix C**) for locally mapped Regional Recharge Areas.

<u>Alternative 5</u> received a maximum score of 10. Alternative 5 (**Figure 8.1**) does not impact any regional recharge areas. The two trenchless crossings occur in areas that do not require dewatering.

<u>Alternative 4</u> received a score of 9. This alternative is similar to Alternative 5 however there is a possibility that the trenchless stream crossing of Baden Creek Tributary A (**Figure 7.1**) may require dewatering due to possible groundwater inflow. As stated above, there is some connectivity to the Regional Recharge Area to the east of the PSA.

<u>Alternative 2</u> received a score of 7. This alternative has two open-cut crossings along Foundry Street at the Baden Creek Pond and the Huehn Award Drain (**Figure 5.1**). These water features are connected to the regional recharge area to the east. Construction of the sewer in this area may affect groundwater recharge temporarily due to dewatering.

<u>Alternative 3</u> received a score of 6. This alternative is similar to Alternative 2 with respect to the open-cut stream crossing along Foundry Street (**Figure 6.1**), but the increased depth would increase the potential of impacting the Regional Recharge Area to the east.

<u>Alternative 1</u> received the minimum score of 1. Though this alternative has the same open-cut stream crossings along Foundry Street as Alternatives 2 and 3 (**Figure 4.1**), it also has a proposed forcemain and pumping station, to service parcel Area O, which is in close proximity to the regional recharge area to the northeast.

Groundwater – Clean Water Act (WHPA, SGRA)

This Factor Group assesses the potential impact of each alternative with respect to WHPAs, HVAs and/or SGRAs as described in the Clean Water Act. The scoring was based on the proximity of the proposed Baden Trunk Alternative to a vulnerable area. The scoring was not based on the potential for infiltration and/or groundwater interference as those elements are covered under categories of shallow groundwater movement and shallow groundwater quality. Refer to Figure 4 and Figure 6 of **Appendix C** for locally mapped WHPA and SGRAs.

<u>Alternative 5</u> received a maximum score of 10. This alternative scored the highest because it is the furthest alignment from the WHPAs and SGRAs located to the southeast of the PSA.

<u>Alternative 4</u> received a score of 9. This alternative is similar to Alternative 5, however portions of the alignment are slightly closer to the WHPAs and SGRAs.

<u>Alternative 3</u> received a score of 6. This alternative scored lower due to the proximity of the headwaters of the Baden Creek and Huehn Award Drain to the WHPAs and SGRAs located to the southeast of the PSA.

<u>Alternative 2</u> received a score of 5. This alternative is similar to Alternative 3.

<u>Alternative 1</u> received the minimum score of 1. Not only are the headwaters of Baden Creek and the Huehn Award Drain in close proximity to the southeastern WHPAs and SGRAs, the forcemain and



proposed pumping station in Parcel O (**Figure 4.1**) is also in close proximity to the WHPAs and SGRAs to the northeast.

Shallow Groundwater Movement

The evaluation of this Factor is based on the potential for interference with existing shallow groundwater flow patterns (baseflow) and its proximity to significant groundwater feature discharge areas. Short-term impact is considered. Refer to Figure 7 of the *Hydrogeological Assessment of Alternatives - Technical Memo* (Appendix C) regarding local existing groundwater movement and contours.

<u>Alternative 5</u> received a maximum score of 10. This alternative scored the highest because there is minimal groundwater interaction at the stream crossing of Tributary A. A shallow sand layer is present between 5.6m and 6.1m bgs at MW102-21, which is below the proposed obvert of 5.4m bgs, refer to Figure 1 and borehole log in **Appendix B** for further details. The soil profile at MW104-21, located at the stream crossing just north of Gingerich Road, is primarily clayey silt with no sand, implying little groundwater movement through that area. It is also furthest away from the locally significant wetlands (**Figure 8.1**).

<u>Alternative 2</u> received a score of 5. This alternative does not impede shallow groundwater flow along Snyder's Road West due to the soils being predominantly silty clay till. There are some saturated sand seams located at the proposed trunk sewer depth next to the Baden Creek pond which may influence shallow groundwater movement during construction.

<u>Alternative 3</u> received a score of 4. This alternative passes through a sand layer at MW103-21. Construction may impact groundwater movement to a locally significant wetland in this area. Refer to **Figure 6.1**. The increased depth of construction of this alternative may impact shallow groundwater to the Huehn Award Drain and Baden Creek pond.

<u>Alternative 4</u> received a score of 2. This alternative, located below an existing sand layer at MW103-21 (between 4.0m and 5.8m bgs), could possibly impact groundwater movement to local significant wetlands during construction if dewatering is required. Refer to Figure 1 and borehole log in **Appendix B** for further details. Groundwater movement to the Baden Creek Tributary A may be temporarily impacted during trenchless construction due to potential dewatering.

<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 with the additional shallow forcemain and stream crossing north of Snyder's Road East. Groundwater influence has been observed in this area and is associated with two meadow marsh wetland features located in this area (**Figure 4.1**). Alternative 1 has the largest extent of construction which may influence groundwater movement during construction.

1.F Surface Drainage (Quantity and Erosion)

This Factor Group assess how surface drainage impacts surface water features, the hydrologic functions within the Regulatory Floodplain and the slope stability of the watercourses.

Quantity Control and Flow Regime

The Baden Creek flow regime is most impacted during the dewatering phase or rerouting of Baden Creek during construction. This occurs when extracted water during open-cut trenching or dewatering is returned to Baden Creek at a flow rate different than its current flow rate. The added volume and change in flow rate can impact the flow regime leading to erosion and water quality issues. Temporary damming of the Baden Creek may also result in upstream flooding if open cut trenching is used without rerouting. Mitigative measures such as flow monitoring of rerouted water and TSS removal may need to be implemented.



Scoring of this Factor is based on the number of locations that would potentially require mitigative measures.

<u>Alternative 5</u> received a maximum score of 10. Given the nature of intermittent/ephemeral streams and the soil structure along Baden Creek Tributary A, it is reasonable to conclude dewatering will not be required at the trenchless railway/stream crossing (**Figure 8.1**). The Baden Creek trenchless crossing north of Gingerich Road is also not anticipated to require dewatering due to the soil structure at that location. Therefore, no impacts to quantity of flow or flow regime within the Baden Creek are anticipated.

<u>Alternative 4</u> received a score of 8. This alignment is similar to Alternative 5 with respect to the crossing of Baden Creek north of Gingerich Road however, there is a possibility of increasing volume to Baden Creek due to redirecting water if dewatering is required at the trenchless railway crossing (**Figure 7.1**). This trenchless crossing occurs through a sand layer and is in close proximity to locally significant wetlands. The area between the trenchless crossing of Baden Creek Tributary A and Queen Street may also require dewatering resulting in an increasing water quantity to Baden Creek if significant.

<u>Alternative 1</u> received a score of 7. This alternative may have additional water added to Baden Creek Tributary B during dewatering at the stream crossing north of Snyder's Road East (**Figure 4.1**). The open cut trenches required at the Baden Creek pond and the Huehn Award Drain crossing will require flow diversion, resulting in potential increased flows. Flow regime should be monitored while rerouting to mitigate significant impacts.

<u>Alternative 2</u> received a score of 6. This alternative is similar to Alternative 1 except that it does not cross Baden Creek north of Snyder's Road East. It scored lower because this alignment is deeper than Alternative 1, and would possibly require further dewatering.

<u>Alternative 3</u> received the minimum score of 1. This alternative's is the similar as Alternative 2 except that the proposed alignment is much deeper resulting in increased volume and flow due to additional dewatering and flow rerouting.

Floodplain Function

This Factor assesses impacts to the Regulatory Floodplain and changes to the hydrologic functions of the floodplain during construction. Scoring is based on construction within the floodplain, primarily where stream crossings occur and potential impacts to the floodplain.

<u>Alternative 2</u> received a score of 10. This alternative impacts the floodplain at the two open-cut stream crossings along Foundry Street (**Figure 5.1**).

<u>Alternative 3</u> received a score of 9. This alternative is similar to Alternative 3 except the deeper construction will impact more of the floodplain during construction.

<u>Alternative 1</u> received a score of 7. This alternative is similar to Alternative 2 except it has an additional stream crossing within the regulated floodplain north of Snyder's Road East (**Figure 4.1**).

<u>Alternative 5</u> received a score of 6. This alternative runs along the floodplain boundary and impacts the GRCA regulated floodplain along the southern, horizontal section of Baden Creek and south to Gingerich Road (**Figure 8.1**). The trenchless stream crossing at the uppermost reach of Baden Creek Tributary A, does not have a regulated floodplain and will not impact the GRCA regulated floodplain.

<u>Alternative 4</u> received a score of 1. This alignment is similar to Alternative 5 except the trenchless stream crossing of Baden Creek Tributary A is close to the GRCA regulated floodplain and may be impacted during construction (**Figure 7.1**).

Stormwater Management

Stormwater management was not a significant issue and all Alternatives are scored as 10.

Engineers, Scientists, Surveyors.



Water Balance

Water Balance was not a significant issue and all Alternatives are scored as 10.

Erosion Assessment

This Factor Group assessed each alternative based on the potential erodibility of Baden Creek and its tributaries during construction. Factors considered to contribute to bank erosion include increased TSS due to construction activities, open-cut trenching across stream and flow regime due to rerouting or dewatering.

The creek shows many signs of active bank erosion and lateral migration, particularly in grass areas where the root depth is not sufficient to protect the loose banks from eroding. Results for the rapid geomorphic assessment (RGA) show that each reach is either stressed or unstable.

The *Fluvial Geomorphological Assessment (August 30, 2022),* prepared by Water's Edge (**Appendix D**), was used to score this factor. Preliminary design of the Baden Sanitary Trunk will reflect the required erosion hazard limits.

<u>Alternative 5</u> received the maximum score of 10. Two trenchless crossings at Baden Creek are proposed. There is a trenchless crossing along the intermittent/ephemeral reach of the Baden Creek Tributary A and railway, which will not require dewatering. The trenchless crossing of Baden Creek, north of Gingerich Road is not anticipated to require dewatering. The alignment should be outside the erosion hazard setback, as it is predominantly a sand system and shows obvious signs of ongoing erosion.

<u>Alternative 4</u> received a score of 9. This alignment is similar to Alternative 5 except the trenchless crossing near the confluence of tributary A and Baden Creek may require dewatering due to signs of groundwater inflow. This additional proximity to Baden Creek results in a lower scoring than Alternative 5. Similar to Alternative 5, the alignment should be outside the erosion hazard setback due to the reach being either stressed or unstable.

<u>Alternative 2</u> received a score of 4. This alternative has an open-cut trench at the Baden Creek pond and at the Huehn Award Drain. Flow diversion structures increase the risk of erosion during construction. Mitigation would be required at all locations to prevent an increase of TSS to Baden Creek.

<u>Alternative 3</u> received a score of 3. This alternative is similar to Alternative 2 except that it requires a deeper trench through Baden Creek at the pond, leading to increased potential erosion concerns.

<u>Alternative 1</u> received a score of 1. This alternative is similar to Alternative 3 except for an additional stream crossing. This alternative also requires dewatering at the trenchless crossing north of Snyder's Road West (Tributary B). The added flow diversion structure increases the risk of erosion during construction. Mitigation would be required at all locations to prevent an increase of TSS to Baden Creek.



Natural Environmental Assessment Report (April 2022)

> Evaluation of Alternatives Natural Environment Criteria Technical Memorandum (July 11, 2022)





Baden Trunk Sanitary Sewer Environmental Assessment

Natural Environment Assessment

Prepared for:

MTE Consultants Inc. 520 Bingemans Centre Drive Kitchener, Ontario N2B 3X9

Project No. 2561 | November 2022



Baden Trunk Sanitary Sewer Environmental Assessment

Natural Environment Assessment

Project Team

| Ryan Archer | Project Manager; Senior Terrestrial and Wetland Biologist |
|----------------|---|
| Andrew Dean | Terrestrial and Wetland Biologist |
| Steve Burgin | Aquatic Biologist |
| GIS Technician | Gerry Schaus |

Report submitted on November 3, 2022

Hom and

Ryan Archer, M.Sc. Project Manager Senior Terrestrial and Wetland Biologist

Table of Contents

| 1.0 | Introduction | 1 |
|-----|---|----|
| 2.0 | Background Information Review and Significant Habitat Screening | |
| 2.1 | Background Information Secondary Sources | |
| 2.2 | Significant Species Habitat Screening | 2 |
| 3.0 | Relevant Policies, Legislation and Regulations | 4 |
| 3.1 | Waterloo Region Official Plan Policies | |
| 3.2 | Wilmot Township Official Plan Policies | 4 |
| 3.3 | Grand River Conservation Authority Regulations and Policies | 5 |
| 3.4 | Provincial Policy Statement | 6 |
| 3.5 | Fisheries Act | 7 |
| 3.6 | Endangered Species Act | 8 |
| 3.7 | Migratory Birds Convention Act | |
| 4.0 | Fieldwork Methodology | 9 |
| 5.0 | Existing Conditions | 12 |
| 5.1 | Soils, Terrain and Drainage | 12 |
| 5.2 | Terrestrial Features | 13 |
| 5. | 2.1 Vegetation Communities | 13 |
| 5. | 2.2 Vascular Flora | 17 |
| 5. | 2.3 Birds | 18 |
| 5. | 2.4 Herpetofauna | 20 |
| 5. | 2.5 Mammals | 20 |
| 5. | 2.6 Insects | 21 |
| 5.3 | Aquatic Features | 21 |
| 6.0 | Natural Environment Significance and Sensitivity | 27 |
| 6.1 | Watercourses and Fish Habitat | 27 |
| 6. | 1.1 Watercourse Buffers | 28 |
| 6.2 | Woodlands | 29 |
| 6.3 | Wetlands | |
| 6. | 3.1 Wetland Buffers | |
| 6.4 | Species at Risk | 31 |
| 6. | 4.1 Confirmed Habitat | 31 |
| 6. | 4.2 Potential Habitat | |
| 6.5 | Significant Wildlife Habitat | |

| 6. | 5.1 | Candidate Significant Wildlife Habitat | 33 |
|-----|-----|---|----|
| 6. | 5.2 | Significant Wildlife Habitats Considered Absent | 34 |
| 6.6 | | Regionally Significant Species | 34 |
| 6.7 | | Ecological Linkages | 35 |
| 7.0 | Ref | ferences | 36 |

List of Tables

| Table 1. | Vegetation communities present within the Primary Study Area | .14 |
|----------|--|-----|
| Table 2. | Vegetation communities present within the Northeast Study Area Section | .15 |
| Table 3. | Fish Species Identified within Baden Creek | .24 |

List of Appendices

Appendix I Schmidt Estate EIS Study Area Appendix II Plant Species Inventoried within the Study Area Appendix III Bird Species Reported from the Study Area and Vicinity Appendix IV Herpetofauna Species Reported from the Study Area and Vicinity Appendix V Mammal Species Reported from the Study Area and Vicinity Appendix VI Butterfly Species Reported from the Study Area and Vicinity Appendix VII Odonata Species Reported from the Study Area and Vicinity Appendix VIII Odonata Species Reported from the Study Area and Vicinity Appendix VIII Species at Risk/Species of Conservation Concern Habitat Assessment Appendix IX Significant Wildlife Habitat Assessment

Maps

- Map 1. Natural Environment Assessment Study Area
- Map 2. Vegetation Communities
- Map 3. Natural Environment Constraints

1.0 Introduction

Natural Resource Solutions Inc. (NRSI) was retained by MTE to complete a Natural Environment Assessment (NEA) to inform the Municipal Class Environmental Assessment (EA) associated with the planned installation of a trunk sanitary sewer in the community of Baden. The sanitary sewer installation is required as part of plans to service future development lands in the west end of Baden. The sanitary sewer will extend from Snyder's Road in the north to the Baden Wastewater Pumping Station in the south. Multiple alternative alignments were considered as part of the EA. See Map 1 for the NEA study area.

The study area is predominantly comprised of actively cultivated agricultural lands, with lesser areas of fallow, regenerating fields. Smaller areas of rural residential and commercial land cover also fall within the study area. Natural feature coverage within the study area is dominated by Baden Creek and its tributaries, along with adjacent cultural meadow features. Small, isolated woodlands fall within the study area. Baden Creek passes through an anthropogenic pond located within the northeast end adjacent to Foundry Street.

This report summarizes background information on natural heritage features within the study area as well as the results of field surveys completed to accurately characterize the existing natural environment conditions. Information on existing conditions within the study area was also gathered from background and field data collected for the Schmidt Estate Environmental Impact Study (EIS) collected by NRSI in 2017 for one of the participating landowners. The Schmidt Estate lands cover a large portion of the NEA study area; see Appendix I for an illustration of the Schmidt Estate EIS study area. Although data collected for the Schmidt Estate EIS was used to inform land use constraints for the previous re-zoning of those lands, the EIS itself has not yet been completed. Additional supplementary background and field survey data was collected by NRSI in 2021 for the purposes of this study.

Existing conditions data was used to assess the significance and sensitivity of natural features and ecological functions within the study area. This analysis was completed with consideration for applicable Township and Provincial policies and legislation as well as the regulations and associated policies of the Grand River Conservation Authority (GRCA). The results of this assessment will be used to inform the evaluation of alternative alignments for the planned trunk sanitary sewer installation.

2.0 Background Information Review and Significant Habitat Screening

2.1 Background Information Secondary Sources

A review of existing natural heritage information was completed to identify key natural heritage features and species that are known or have potential to occur within the study area. Requests for background information were sent to the Ontario Ministry of Northern Development, Mines, Natural Resources and Forestry (NDMNRF) Guelph District and the GRCA. Relevant background information for the study area was received from the GRCA on May 10, 2021 and from the NDMNRF on June 18, 2021.

Background information relevant to the study area was also collected and reviewed from sources including the following:

- Land Information Ontario Natural Heritage Make-a-Map base mapping and Natural Heritage Information Centre (NHIC) online database (MNRF 2021a);
- Region of Waterloo Official Plan (2015);
- Township of Wilmot Official Plan (2019);
- GRCA online mapping (GRCA 2021);
- Background information and NRSI field data collected for the Schmidt Estate EIS in 2017;
- Department of Fisheries and Oceans Canada (DFO) Species at Risk Mapping (DFO 2021);
- Atlas of the Mammals of Ontario (Dobbyn 1994);
- Ontario Reptile and Amphibian Atlas (Ontario Nature 2019);
- Ontario Breeding Bird Atlas (BSC et al. 2008);
- eBird (eBird 2021); and,
- iNaturalist (iNaturalist 2021).

2.2 Significant Species Habitat Screening

Species at Risk (SAR) are those listed on the Species at Risk in Ontario List (MECP 2020). These include species identified by the Committee on the Status of Species at Risk in Ontario (COSSARO) as provincially Endangered, Threatened, or Special Concern. Species listed as Endangered or Threatened are protected under the *Endangered Species Act* (ESA), which includes protection to their habitat. Species considered Special Concern are included in the definition of Species of Conservation Concern (SCC), which includes the following:

- species designated provincially as Special Concern,
- species that have been assigned a conservation status (S-Rank) of S1 to S3 or SH by the Natural Heritage Information Centre (MNRF 2021a), and
- species that are designated federally as Threatened or Endangered by the Committee for the Status of Endangered Wildlife in Canada (COSEWIC) but not provincially by the COSSARO. These species may be protected by the federal *Species at Risk Act* (SARA) if they are listed as Threatened or Endangered on Schedule 1 of the SARA.

Habitat for SCC is considered Significant Wildlife Habitat (SWH) (OMNR 2010), which is afforded protection under the Provincial Policy Statement (OMMAH 2020) and municipal natural heritage protection policies. For the purposes of this report, the term "SAR" will refer to provincially Threatened and Endangered species regulated under the ESA while provincial species of Special Concern will be considered SCC.

Based on NRSI's examination of background sources and federally or provincially significant species with occurrence records in the study area vicinity (within 10km), an assessment of SAR and SCC suitable habitat presence within the study area was completed. Assessments of habitat suitability in the study area were made by cross-referencing each species' known habitat preferences or requirements (e.g., OMNR 2000) with existing natural features based on review of recent satellite imagery of the study area lands and based on previous NRSI site characterization within the Schmidt Estate lands. The results of this significant species habitat screening are described in the context of SAR and SCC in Sections 6.4 and 6.5, below.

A preliminary screening for the presence of SWH was also completed for the study area. The Significant Wildlife Habitat Technical Guide (SWHTG) is a guideline document that outlines the types of habitats that the NDMNRF considers significant in Ontario as well as criteria to identify these habitats (OMNR 2000, MNRF 2015a). The SWHTG groups SWH into four broad categories: seasonal concentration areas, rare vegetation communities and specialized wildlife habitat, habitats of SCC, and animal movement corridors. This screening involved the comparison of NDMNRF criteria outlined for Ecoregion 6E, in which the study area is located, against habitats known to occur in the study area. The results of the SWH screening are described in Section 6.5.

3.0 Relevant Policies, Legislation and Regulations

3.1 Waterloo Region Official Plan Policies

Waterloo Region's environmental features and the linkages among them have been mapped as part of a broad Greenlands Network. The Greenlands Network is comprised of Landscape Level Systems, Core Environmental Features, Supporting Environmental Features, fish habitat, and the linkages among these elements (Region of Waterloo 2015). Landscape Level Systems and Core Environmental Features are mapped by the Region in its Official Plan (OP). Neither of these Greenlands Network elements are mapped within the EA study area. However, Significant Habitat of Endangered and Threatened Species, which is a category of Core Environmental Feature, is not mapped on Greenlands Network OP mapping. Regional policies associated with this habitat type will apply where applicable.

With respect to fish habitat, Section 7.D.1 of the Regional OP states that "development and site alteration will not be permitted within fish habitat, except in accordance with Provincial and Federal requirements to the satisfaction of the Federal Department of Fisheries and Oceans, or its delegate".

Supporting Environmental Features comprise environmental features not meeting criteria for recognition as being Regionally significant. It is the role of Area Municipalities to identify and set policies for the protection of Supporting Environmental Features (see Section 3.2). Linkages that are identified through environmental studies are also to be incorporated into site development plans in accordance with the policies and requirements of Area Municipalities.

3.2 Wilmot Township Official Plan Policies

Criteria that define the presence of Supporting Environmental Features are summarized in Section 8.1.5.2 of the Township OP. These include the following that may be of relevance to the study area:

- All non-provincially significant wetlands as designated by the MNRF or the GRCA;
- Locally significant woodland;
- Locally significant environmental linkage as identified by an environmental study;
- Permanent or intermittent watercourses;
- Features that provide a connection or act as a buffer or supporting area to environmental features, thus maintaining ecological connectivity between communities; and,

• Performing vital hydrogeological functions, such as groundwater recharge or discharge areas (Township of Wilmot 2019).

Development within or adjacent to (within 50m of) Supporting Environmental Features will only be permitted under Township policy where a suitable environmental study can demonstrate that adverse environmental impacts will not occur on the natural features and ecological functions, or otherwise subject to other conditions being met as outlined in Section 8.1.5.5 of the OP. Where Environmental Linkages have been identified through environmental studies and in consultation with the appropriate agencies, the Township will require that Environmental Linkages be incorporated into land use planning.

In conjunction with the GRCA, the Township has mapped the locations of Environmentally Constrained Lands, which include areas of flooding, erosion and slope hazards. For reasons of public safety and property damage/disruption, development is generally directed away from areas of Environmentally Constrained Lands. In accordance with OP Section 8.2.5, an exception to development prohibitions within these lands is made for wastewater management infrastructure provided flood flows are not affected and applicable GRCA policies are met (Township of Wilmot 2019).

3.3 Grand River Conservation Authority Regulations and Policies

The study area is regulated by the GRCA under the *Development, Interference with Wetlands and Alterations to Shorelines and Watercourses Regulation* (Ontario Regulation 150/06). Development and site alteration within the regulated lands is prohibited unless permitted by the GRCA under the policies of its regulation. Regulated features within the study area include floodplain, valley and slope erosion hazards, and their regulatory safety allowances, associated with Baden Creek and its tributaries. As described below, small, previously unmapped wetland features have also been confirmed within the study area. In accordance with the Regulation, it must be demonstrated that an undertaking will not affect the control of flooding, erosion, dynamic beaches, or pollution, or the conservation of land, in order to be permitted (GRCA 2015).

Section 8.1.15 of the GRCA's *Policies for the Administration of Ontario Regulation 150/06* (GRCA 2015) states that public infrastructure, including but not limited to sanitary sewers, may be permitted within the Riverine Flooding Hazard zone provided there is no feasible alternative outside of that zone as determined through an EA, and where various conditions are met.

These conditions include but are not limited to the limitation of adverse hydraulic or fluvial impacts with avoidance or appropriate mitigation of upstream or downstream flooding risks, no loss of flood storage where possible, and where unavoidable intrusions into significant natural features or hydrological or ecological functions occur that these features/functions can be adequately restored or enhanced. The maintenance or repair of public infrastructure within the Riverine Flooding Hazard zone may also be permitted where it can be demonstrated that unavoidable intrusions into significant natural heritage features or ecological/hydrological functions can be properly restored or enhanced.

Similarly, Section 8.2.21 states that public infrastructure such as sanitary sewers may be permitted within Riverine Erosion Hazard zones where it is determined that there is no feasible alternative outside of the regulated area as determined through an EA, and where a geotechnical or engineering study identifies a more precise hazard limit, and where additional conditions are met. These additional conditions include, but are not limited to, no impacts on existing or future slope stability occurring, no risk of creating new riverine erosion hazards, and where unavoidable intrusions into significant natural features or hydrological or ecological functions occur that these features/functions can be adequately restored or enhanced (GRCA 2015).

3.4 Provincial Policy Statement

The Provincial Policy Statement (OMMAH 2020) provides direction on matters of provincial interest with respect to land use planning and developments that are subject to the provincial *Planning Act*. It supports improved land use planning and management, contributing toward a more effective and efficient land use planning system while protecting resources of provincial interest including, but not limited to, the natural environment. Section 2.1 of the Provincial Policy Statement identifies policies for the protection of Natural Heritage; these policies should serve as guidance in the design of the preferred sanitary sewer alignment. Of relevance to this project and its study area natural features are the following policies:

- Section 2.1.5:
 - Development and site alteration shall not be permitted in Significant Wildlife Habitat unless it has been demonstrated that there will be no negative impacts on the natural features or their ecological functions.
- Section 2.1.6:

- Development and site alteration shall not be permitted in fish habitat except in accordance with provincial and federal requirements.
- Section 2.1.7:
 - Development and site alteration shall not be permitted in habitat of endangered species and threatened species, except in accordance with provincial and federal requirements.
- Section 2.1.8:
 - Development and site alteration shall not be permitted on adjacent lands to the natural heritage features and areas identified in policies 2.1.5 and 2.1.6 unless the ecological function of the adjacent lands has been evaluated and it has been demonstrated that there will be no negative impacts on the natural features or on their ecological functions.

3.5 Fisheries Act

The *Fisheries Act* provides for the conservation and protection of fish and fish habitat and pollution prevention by prohibiting the "harmful alteration, disruption or destruction of fish habitat" and protects against the "death of fish, other than by fishing". To ensure compliance with relevant provisions included in the *Fisheries Act*, Fisheries and Oceans Canada (DFO) facilitates review of proposed works, undertaking, and activities that may affect fish and fish habitat through the Fish and Fish Habitat Protection Program. Proponents are responsible for assessing their projects for their potential to negatively impact fish and fish habitat and several codes of practice that may be incorporated into project design. If there is the potential to contravene the provisions within the *Fisheries Act* then the project needs to be submitted to DFO for review and approval, which may include a Letter of Advice or a *Fisheries Act* Authorization.

The study area contains several watercourses that provide, or have the potential to provide, direct fish habitat. Project design will need to consider these features and should attempt, to the extent possible, to mitigate against the harmful alteration, disruption or destruction of fish habitat and the death of fish.

3.6 Endangered Species Act

The *Endangered Species Act* (ESA) prohibits the killing, harming, harassing or capturing of species that are listed as Endangered or Threatened in the Species at Risk in Ontario list. It also protects the regulated or general habitats of these species from damage or destruction. Activities that would otherwise be in violation of the ESA may be permitted by the Ontario Ministry of Environment, Conservation and Parks (MECP), such as in accordance with Section 17(2)(c) of the ESA. Activities may also be authorized in accordance with Ontario Regulation 242/08, as applicable to certain SAR, certain activity types, and provided various conditions are met.

As described in Section 6.4, various SAR have potential to occur within the study area based on the results of the SAR/SCC screening described in Section 2.2.

3.7 Migratory Birds Convention Act

The federal *Migratory Birds Convention Act* (MBCA) protects populations of, and individual, migratory bird species as listed under the Act, as well as their eggs and nests. The MBCA enacts various prohibitions, including but not limited to the disturbance, destruction, or taking of a nest or eggs of migratory birds. In the context of this EA, this primarily applies to the removal of trees, other vegetation, or structures that may contain migratory birds or their active nests.

4.0 Fieldwork Methodology

NRSI terrestrial and aquatic field surveys were previously completed for a large portion of the study area (see Appendix I) in 2017 for the Schmidt Estate EIS. The results of these surveys have been considered in this study. Additional field surveys were completed between May and July 2021 to supplement and update the earlier field investigations, as well as to account for the larger EA study area. As an addition to the NEA study area, an additional terrestrial and aquatic site investigation was also completed in February 2022 for an area to the northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection (Map 1). The following is a summary of the field tasks completed for this NEA in 2021 and 2022.

Vegetation Community Mapping and Species Inventories

Ecological Land Classification (ELC) (Lee et al. 1998) mapping previously completed for the Schmidt Estate lands was updated and expanded to cover the EA study area. Updates to vegetation community mapping were completed on May 17 and July 20, 2021. During each of these site visits, a comprehensive inventory of vascular flora was completed to inform the ELC vegetation community classifications. The timing of these visits corresponded to spring- and summer-season vegetation inventories. The vegetation inventory work included an emphasis on the identification of any federally, provincially, or regionally significant vegetation species within the study area. Vegetation species within the watercourse channels that were indicative of groundwater influences were also noted and mapped.

A supplementary site visit was completed on February 10, 2022 to characterize natural features within the area northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection. The focus of this assessment was characterize the features using ELC and to identify the presence of wetland, as could be determined at that time of year in conjunction with air photo interpretation.

Aquatic Habitat Assessment

An NRSI aquatic biologist completed a survey on May 19, 2021 to characterize the aquatic habitats within the study area. This included an updated assessment of Baden Creek and the Huehn Award Drain within the study area, as well as a new assessment of Baden Creek Tributary A (Map 1) up to its outlet from a stormwater management facility in the north end. The description of existing conditions combines observations made during the initial aquatic habitat assessment that was completed on July 20, 2017 and the May 19, 2021 assessment.

During these assessments, the following information was recorded:

- riparian and aquatic vegetation;
- channel dimensions;
- general bank stability;
- cover type and quality;
- substrate type;
- flow conditions; and
- water temperature.

A supplementary site visit was completed on February 2, 2022 to inspect and characterize Baden Creek Tributary B where it flows northeast of the intersection of Snyder's Road, Foundry Street and Livingston Boulevard. Assessment parameters were limited to what could be observed at that time of year, but included wetted widths, water depth, bank height, vegetation species presence, and assessment of fish habitat function.

Breeding Bird Surveys

Two supplemental early morning breeding bird surveys were completed on June 9 and 24, 2021 in accordance with Ontario Breeding Bird Atlas (OBBA) protocol (Birds Canada 2021). Surveys were completed between a half-hour before sunrise and 10:00am and were timed to occur at least 10 days apart. Surveys were completed through comprehensive area searches of each of the study area vegetation communities. Standard breeding evidence codes were recorded based on OBBA protocol. Any observed structures were inspected for the presence of bird nesting where suitable nesting habitat existed.

Wildlife Habitat Assessment

Natural features within the study area were investigated for the presence of potentially significant habitats based on the screening exercise results presented in Sections 6.4 and 6.5. This included a search for potential burrows of the SAR American Badger (*Taxidea taxus*). A reptile area search was also completed under suitable conditions (warm, sunny) on May 17, 2021, timed to coincide with the period of spring reptile emergence. Observations of multiple snakes in a localized area would suggest the possible location of a nearby snake hibernaculum. The survey was targeted to suitable areas of snake basking habitat within the study area.

Assessments of other significant wildlife habitat suitability were made based on the natural feature characterization.

Incidental Observations

During the field work program, all incidental observations of mammals, herpetofauna, butterflies and odonates (dragonflies and damselflies) were documented during all field visits. This included direct observations of individuals, as well as signs of wildlife presence (i.e. tracks, scat, dens, nests, etc.).

5.0 Existing Conditions

5.1 Soils, Terrain and Drainage

The study area is located within a physiographic region of southern Ontario known as the Waterloo Hills (Ontario Ministry of Northern Development and Mines 2017). This region occupies approximately 777 km², centred primarily in Waterloo Region. It is dominated by sandy hills, with sandy till ridges, kames or kame moraines, and sandy outwash in the intervening hollows (Chapman and Putnam 1984). The well-drained soils of the Waterloo Hills may be characterized as mature Grey Brown Luvisolic soils (Chapman and Putnam 1984). The Waterloo Interlobate Moraine overlays the majority of Wilmot Township and contains extensive ice-contact and glaciofluvial stratified deposits either at the surface or buried beneath till units (Ontario Geological Survey and Planning and Engineering Initiatives Limited 1998).

The study area soils east of Baden Creek are characterized by the imperfectly-drained loam and loamy sand of the Grand-Kirkland soil series (Chapman and Putnam 1984). This soil series formed on recent alluvial deposits. The soils west of Baden Creek are characterized by the imperfectly draining silty-clay loam Huron-St. Clements soil series (Chapman and Putnam 1984). This soil series is formed on till or lacustrine deposits.

The topography of the study area is relatively level (~347m elevation) with a gentle, approximately 1m slope towards Baden Creek (~342m elevation).

The subject lands are located within the Grand River watershed. Baden Creek, which drains from a dammed pond within the northeast end of the study area, widely meanders (west, south, east, and again southwards) through the study area from north to south along historically channelized alignments to optimize agricultural land use opportunities. Baden Creek flows into the Nith River downstream of the study area, which is a tributary of the Grand River. Three tributaries confluence with Baden Creek within the study area: The Huehn Award Drain, which drains lands to the east, and two unnamed tributaries to Baden Creek. Baden Creek Tributary A drains lands to the northwest and originates from a stormwater management pond located within a subdivision northwest of the study area. Baden Creek Tributary B drains lands to the northwest of the study area.
5.2 Terrestrial Features

5.2.1 Vegetation Communities

The study area is predominantly comprised of active agricultural fields along with some fallow, early successional regenerating meadow communities. Natural feature coverage primarily comprises Baden Creek and its tributaries, along with adjacent meadow (narrowly fringing in some locations), as well as small, isolated woodland communities. A dammed open water pond, online with Baden Creek, is located in the northeast end of the study area. Small areas of existing rural residential and commercial land use exist within the study area. A smaller disjunct portion of the study area northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection contains open culturally influenced lands including small pockets of wetland, woodland and meadow adjacent to a railway corridor and public pedestrian trail.

Vegetation communities observed within the study area are shown on Map 2 and described in Tables 1 and 2 below. For the purposes of presenting the ELC vegetation communities, the small northeastern portion of the study area as shown on Map 2 (Inset A) is referred to herein as the Northeast Study Area Section, and is separated from the remaining study area, referred to herein as the Primary Study Area. The Northeast Study Area Section was coarsely characterized through a single winter season site investigation, whereas the Primary Study Area was characterized through various multi-season field surveys as described in Section 4.0 of this report. Accordingly, vegetation communities within the Northeast Study Area Section are presented separately to reflect the relatively lower level of site characterization in comparison to the Primary Study Area.

Table 1. Vegetation communities present within the Primary Study Area.

| ELC | | |
|----------|---------------------------|--|
| Ecosite | | |
| Туре | ELC Description | Environmental Characteristics |
| Cultural | 1 | |
| CUM1 | Mineral Cultural Meadow | An early successional community that has established on fallow, previously farmed agricultural field parcels. This community also narrowly fringes the top of bank along Baden Creek and Baden Creek Tributary A in locations where fields are actively cultivated on either side. In some areas this community occurs more broadly around the watercourses within their respective floodplains. It also occurs along the rail corridor and adjacent areas in the northwest study area. |
| | | <u>Canopy</u> : N/A <u>Sub-canopy</u> : N/A <u>Understorey</u> : Manitoba Maple (<i>Acer negundo</i>), Common Apple (<i>Malus pumila</i>), Black Walnut (<i>Juglans nigra</i>) <u>Groundcover</u> : Various cool season grasses, Smooth Bedstraw (<i>Galium mollugo</i>), Garlic Mustard (<i>Alliaria petiolata</i>) |
| | | This community contains two small meadow marsh inclusions as shown on Map 2: a Mineral Meadow Marsh (MAM2) and a Reed Canary Grass Mineral Meadow Marsh (MAM2-2) feature, both within the north end of the study area. Both appear to have anthropogenic origins and are the result of previous earth-moving activities or water impounding. The MAM2 feature is largely dominated by Narrow-leaved Cattail (<i>Typha angustifolia</i>), and it appears to have been dug historically for the purposes of stormwater management to service the former industrial property to the northeast of the feature. The MAM2-2 feature is dominated by Reed Canary Grass (<i>Phalaris arundinacea</i>), and it appears to have been formed by water impoundment caused by the adjacent residential areas that are slightly higher in grade. |
| CUW1 | Mineral Cultural Woodland | This community occurs as two small and isolated woodland features at the north end of the study area. These are young woodland features (<50 years old) and show signs of ecological disturbance through human usage (e.g., informal walking and biking paths, scattered garbage and debris, firepit). The features are comprised of a high proportion of non-native plant species. |
| | | <u>Canopy</u> : Manitoba Maple <u>Sub-canopy</u> : Manitoba Maple, Common Apple, hawthorns (<i>Crataegus</i> spp.) Understorey: Hawthorns, Manitoba Maple, Choke Cherry (<i>Prunus virginiana</i>) |

| ELC Ecosite | | |
|----------------|---|--|
| Туре | ELC Description | Environmental Characteristics |
| | | <u>Groundcover</u> : Garlic Mustard, Woodland Forget-me-not (<i>Myosotis sylvatica</i>), Wild Strawberry (<i>Fragaria virginiana</i>) |
| Woodland | · | |
| FOD7-3 | Fresh-Moist Willow Lowland Deciduous Forest | This community occurs in two locations along the Baden Creek corridor: downstream of the pond in the north end of the study area and as a small isolated feature at the southwest bend in the creek. Both are ecologically mid- aged features. These features, particularly the northern community, may experience periodic seasonal flooding from Baden Creek during periods of high flow. |
| | | <u>Canopy</u> : Crack Willow (<i>Salix euxina</i>), Manitoba Maple, Weeping Willow (<i>Salix alba</i>) |
| | | Sub-canopy: Manitoba Maple, Crack Willow, Black Walnut |
| | | Understorey: Manitoba Maple, Common Apple, Choke Cherry |
| | | Groundcover: Garlic Mustard, Woodland Forget-me-not, Dame's Rocket |
| | | (Hesperis matronalis) |

Table 2. Vegetation communities present within the Northeast Study Area Section.

| ELC Ecosite | | |
|-------------|---------------------------|--|
| Туре | ELC Description | Environmental Characteristics |
| Cultural | | |
| CUM1 | Mineral Cultural Meadow | Two separate polygons of cultural meadow communities located along the pedestrian trails. Dominant species observed include: Tall Goldenrod (<i>Solidago altissima</i>), Smooth Brome (<i>Bromus inermis</i>), Wild Carrot (<i>Daucus carota</i>), Common Buckthorn (<i>Rhamnus cathartica</i>). |
| CUT1 | Mineral Cultural Thicket | Two separate polygons of cultural thicket communities located along the rail corridor, and along the Baden Creek Tributary B. Dominant species observed include: Black Walnut, Staghorn Sumac (<i>Rhus typhina</i>), Red-osier Dogwood (<i>Cornus sericea</i>), Black raspberry (<i>Rubus occidentalis</i>), Common Apple, Common Buckthorn, Tall Goldenrod. |
| CUW1 | Mineral Cultural Woodland | Three separate polygons of cultural woodland communities throughout the Northeast Study Area Section. Dominant species observed include: Manitoba |

| ELC Ecosite | | |
|-------------|-----------------|---|
| Туре | ELC Description | Environmental Characteristics |
| | | Maple, Black Walnut, Green Ash (<i>Fraxinus pennsylvanica</i>), Balsam Poplar |
| | | (Populus tremuloides), White Spruce (Picea glauca), Red-osier Dogwood. |
| MAM | Meadow Marsh | Two separate polygons of meadow marsh communities located along the |
| | | Baden Creek Tributary B, and along the flow route and associated with a |
| | | stormwater discharge outlet. Dominant species observed include: Red-osier |
| | | Dogwood, Reed Canary Grass (Phalaris arundinacea), Narrow-leaved Cattail |
| | | (Typha angustifolia), Panicled Aster (Symphyotrichum lanceolatum). |

5.2.2 Vascular Flora

In total, 141 plant species were inventoried within the study area. Of these, 57 species (40%) are classified as non-native in Ontario (MNRF 2021a). See Appendix II for a complete list of the inventoried plant species.

One federally and provincially significant plant species, Butternut (*Juglans cinerea*), was inventoried within the study area. Butternut is listed as Endangered both federally and provincially (Government of Canada 2021, MECP 2020). Butternut may be subject to the protections afforded by the ESA, depending on the health status of the individual tree as discussed further below. Three Butternut trees were documented within the Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3) adjacent to Baden Creek as shown on Map 3.

These Butternuts fall within the Schmidt Estate EIS study area. As part of earlier work completed for that EIS, an NRSI certified Butternut Health Assessor completed a Butternut Health Assessment (BHA) for these three trees on October 23, 2020. Foliar aspects of the assessment could not be considered based on the timing of the survey. A conservative approach to the assessment was therefore taken to take this limitation into account, and is consistent with agency guidance for conducting BHAs (MNRF 2015b). Based on this assessment, each of the three trees were determined to represent Category 2 Butternuts. Based a field hybridity test, all were determined to be genetically pure Butternuts. A BHA Report has not yet been submitted to the MECP, so these results may be audited by the MECP to confirm these findings. For the purposes of this study, they will be assumed to represent Category 2 Butternuts.

One other provincially significant species was observed: Rhombic-leaved Sunflower (*Helianthus pauciflorus* ssp. *subrhomboideus*). This species is ranked S2S3 ("Imperiled" or "Vulnerable" in Ontario), which renders it a provincial SCC (OMNR 2010). However, this species was observed to be established as adventive growth along the railroad tracks within the north end of the study area. These individuals are far outside of their natural range (northwest Ontario) and seeds of this species may have been carried by trains. According to the NHIC, this species is considered native to the Lake of the Woods area in northwest Ontario, and is considered likely to be a non-native introduction within southern Ontario (MNRF 2020b). In this context, the Rhombic-leaved Sunflower individuals that were observed are not considered to be provincially significant and do not represent a development constraint. This species is also not listed as regionally significant.

One regionally significant species (Richardson and Martin 1999), Running Serviceberry (*Amelanchier spicata*), was observed within the study area. This species was observed within the Mineral Cultural Meadow (CUM1) community along the rail corridor in the northwest study area. Several (approximately 10-20) small colonies of this species were observed, consisting of many suckering stems. Another Regionally-listed species, Black Walnut, was also documented on-site, although, in the context of the study area it is not considered significant as the individuals are not demonstrably indigenous and suspected to be of anthropogenic origin (Richardson and Martin 1999).

The coefficient of conservatism (CC) is a value ranging from 0 (low) to 10 (high), which is based on a species' tolerance of disturbance and fidelity to a specific habitat integrity (Oldham et al. 1995). Higher values are assigned to species that have specific environmental growing requirements and are less tolerant of disturbance. Overall, the inventoried plant species on the subject lands have an average CC value of 1.79, which is low and is indicative of species that are generalist in their habitat preferences and are typically adapted to ecologically disturbed conditions. Among the vegetation communities, average CC values ranged from 2.20 for the FOD7-3 woodland and only 1.13 for the CUM1 meadow including the meadow marsh inclusions. These results generally indicate a low sensitivity of the existing vegetation species to ecological disturbance that may be caused by site alteration. The highest CC value for a particular plant species was 7 among all plant species observations; two of these were observed within the FOD7-3 woodland while one was observed within the Mineral Cultural Woodland (CUW1). Non-native/invasive species growth is widespread within the study area, with proportions of non-native species ranging from 38% of species within the FOD7-3 community to 49% for the CUM1 community including meadow marsh inclusions.

5.2.3 Birds

In total, 74 bird species are reported from within 10km of the study area based on the OBBA (BSC et al. 2008). Thirty-six (36) bird species were documented within the study area during site investigations. All of these species were documented with at evidence of at least possible breeding within the study area. Refer to Appendix III for a list of bird species recorded within in the study area and vicinity.

One bird SAR, Bank Swallow (*Riparia riparia*), was observed within the study area during site investigations. Bank Swallow is listed as Threatened both federally and provincially (Government of Canada 2021, MECP 2020). Multiple individuals of this species were observed

foraging over the floodplain Mineral Cultural Meadow and adjacent agricultural field at the north end of the study area, just downstream of the pond. This species was only observed during one of the breeding bird surveys (June 9, 2021). Consequently, this observation is considered to represent evidence of "possible" breeding within the study area vicinity (Birds Canada 2021). However, no suitable Bank Swallow nesting habitat exists within the study area. Although Bank Swallow was only observed during one breeding bird survey in 2021, this species was also observed during a breeding bird survey completed for the Schmidt Estate EIS. Based on this evidence and taking a precautionary approach, Bank Swallow foraging habitat is considered present within the study area.

An old shed structure that was observed on-site was inspected for the presence of SAR Barn Swallow (*Hirundo rustica*) nests. No nests were observed on the exterior, or within the interior as could be observed through an open door, and no Barn Swallow activity was observed onsite.

Five regionally significant bird species (Martin 1996) were observed within the study area. These included Cliff Swallow (*Petrochelidon pyrrhonota*) foraging over the Mineral Cultural Meadow (CUM1) and Brown Thrasher (*Toxostoma rufum*) within the CUM1 feature, Redbreasted Nuthatch (*Sitta canadensis*) within the Fresh-Moist Lowland Willow Deciduous Forest (FOD7-3), and Hooded Merganser (*Lophodytes cucullatus*) and Great Blue Heron (*Ardea herodias*) within the pond. However, Cliff Swallow is only considered significant when nesting in natural circumstances (Martin 1996). No Cliff Swallow nests were observed in the study area. All of these species were observed with evidence of "possible" breeding within the study area.

As part of previous field studies for the Schmidt Estate EIS, NRSI biologists also previously documented SAR Barn Swallow individuals foraging over that study area. Although this species was not observed during 2021 NRSI field investigations, the continued use of these lands by Barn Swallow for foraging activities cannot be ruled out. As stated above, no Barn Swallow nesting evidence was observed on-site.

Other bird species observed within the study area are relatively common, ubiquitous on the landscape, and have provincially secure populations. These include assemblages of common, generalist species that are often found in agricultural woodlots, open land/agricultural settings and in urbanized areas or developed lands.

5.2.4 Herpetofauna

According to the Ontario Amphibian and Reptile Atlas (Ontario Nature 2019), 14 species of herpetofauna are reported from within 10km of the study area. No herpetofauna species were observed within the study area, including during the targeted snake emergence survey.

During previous NRSI field studies for the Schmidt Estate EIS, two herpetofauna species, Eastern Gartersnake (*Thamnophis sirtalis sirtalis*) and Green Frog (*Lithobates clamitans*), were observed within that study area. Both of these species are common with secure populations in Ontario.

No features typical of good quality thermoregulatory or cover habitat, or potential snake hibernaculum habitat, were observed on site. A previously open and sun-exposed area that comprised cracked and broken concrete floor slabs associated with a previously removed barn, as observed during 2017 NRSI surveys, had completely grown over with herbaceous growth by 2021, and no longer provided good quality snake basking habitat. A complete list of herpetofauna species reported from the study area and vicinity is provided in Appendix IV.

5.2.5 Mammals

According to the Mammal Atlas of Ontario (Dobbyn 1994), 34 mammal species are reported from within 10km of the study area. Of these, direct observation or evidence of three mammal species was observed within the study area: Eastern Cottontail (*Sylvilagus floridanus*), Woodchuck (*Marmota monax*), and Eastern Gray Squirrel (*Sciurus carolinensus*). None of the observed species are considered federally, provincially or regionally significant. No mammal burrows that are characteristic of American Badger use (e.g., D-shaped, lateral claw marks) were observed within the study area. A complete list of mammals reported from the study area and vicinity is included in Appendix V.

An additional seven mammal species were observed directly, or evidence of their presence was observed, within the Schmidt Estate EIS study area during previous NRSI surveys. These included Virginia Opossum (*Didelphis virginiana*), Star-nosed Mole (*Condylura cristata*), Coyote (*Canis latrans*), Striped Skunk (*Mephitis mephitis*), American Mink (*Mustela vison*), Northern Raccoon (*Procyon lotor*), and White-tailed Deer (*Odocoileus virginianus*). It should be assumed that all of these species still occur within the study area. Of these observed species, American Mink, Coyote and Virginia Opossum are considered regionally significant. Specifically,

American Mink and Coyote are considered "Scarce", while Virginia Opossum is considered "Rare" (Region of Waterloo 1985).

A bat habitat assessment was not completed as part of this study, since it is anticipated that the small woodland features will be avoided when planning the alignment of the sanitary sewer. Tree growth is absent or very sparse within the remainder of the study area. See below for further discussion about SAR bat habitat.

5.2.6 Insects

According to the Ontario Butterfly Atlas (McNaughton et al. 2020), 38 butterfly species are known to occur within 10km of the study area. No butterfly species were incidentally recorded during field investigations. A complete list of butterfly species reported from the study area vicinity is provided in Appendix VI.

During previous field surveys completed within the Schmidt Estate study area, two butterfly species were observed: Cabbage White (*Pieris rapae*) and European Skipper (*Thymelicus lineola*). These species should be assumed to still use the study area lands. Neither of these species is considered federally, provincially or regionally significant.

According to the Ontario Odonate Atlas (MNRF 2017), 18 odonate (dragonfly and damselfly) species are recorded within 10km of the subject property. No odonate species were incidentally recorded within the study area during site visits. As well, no odonate species were recorded during previous fieldwork for the Schmidt Estate EIS. A complete list of odonate species reported from the surrounding 10km squares vicinity is provided in Appendix VII.

5.3 Aquatic Features

Four prominent aquatic features exist within the study area. These include Baden Creek, Huehn Award Drain, and the Baden Creek Tributaries A and B, as shown on Map 1. Each of these is described below. Baden Creek and the lower portion of Huehn Award Drain were previously assessed on July 20, 2017 and the full length of Baden Creek Tributary A within the study area was assessed on May 19, 2021. The reach of Baden Creek Tributary B northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection was assessed on February 2, 2022.

Baden Creek

Baden Creek is a permanent watercourse within the study area. The feature flows southwest from a dammed reservoir located in a northeast portion of the study area before turning east towards Foundry Street. It then turns and flows southeast and parallel to Foundry Street, crossing Gingerich Road, and then eventually exiting the study area.

Baden Creek predominantly flows through actively cultivated agricultural lands within the study area, with lesser areas of fallow, regenerating fields. Small areas of rural residential land are located to the north and east while commercial land cover also occurs to the south. The floodplain and extent of natural vegetation is greater than 10m on both sides of the channel throughout the upper portion of the subject property and is comprised of a naturalized meadow and included Spotted Jewelweed (Impatiens capensis), Reed Canary Grass, and other various terrestrial grasses and herbaceous plants. Downstream of the dam and throughout the upper section, log jams, fallen trees, and overhanging trees across Baden Creek were found to be common, particularly within the forested area. Overhanging bank vegetation (variety of grass species) was found throughout the middle and lower sections. Aquatic vegetation was observed throughout Baden Creek including Slender Pondweed (Potamogeton pusillus) which was the most common and abundant growing throughout the channel in sand, silt and gravel. Arrowhead (Sagittaria latifolia) was observed in moderate abundance in silt and sand substrates along the banks. Watercress (*Nasturtium officinale*) and iron staining, which are generally associated with groundwater influence to a feature, were observed in low to moderate abundance along the banks and throughout shallow areas of the creek.

Baden Creek appears channelized (straightened) throughout the lower portion of the feature from where the channel flows in an easterly direction to where it runs alongside Foundry Street and under Gingerich Road. Active and historical erosion was also noted, which suggests that the channel experiences high water levels and velocities at times throughout the year. At the time of the assessment, Baden Creek was flowing, clear, had a low-medium gradient and gently meandered within the confines of the corridor. The gradient increased where the channel becomes channelized, although even within the channelized section the creek was starting to re-establish some gentle meanders. The quality and amount of shade provided through Baden Creek decreases throughout the study area from the dam to Gingerich Road. The small woodlot features provide upwards of 90% shade through mature deciduous trees. The channel banks had poor to fair stability and the bank height also varied from 0.2m to over 2.5m. The high channel banks were not stable and were often found on outside meander bends with

evidence of erosion. Riffles, shallow run habitats, deep run habitats, and pool features were all observed within Baden Creek. The wetted width of the channel also varied depending on location and type of feature present (pool, riffle, run) and ranged from 1.0m to 5.2m. The substrates throughout varied from gravel and cobble to silt and sand, to exposed hardpan clay. An old crossing (two failing corrugated steel pipes (CSP)) could act as a barrier to fish, although water was observed flowing under the structure, which may allow for fish passage. In-stream habitat and cover are provided through pools, riffles, backwater areas, undercut banks, woody debris, and cobble/boulder.

During the July 20, 2017 field assessment water quality parameters were taken at three locations within Baden Creek. Dissolved oxygen ranged from 4.67ppm (51.6%) to 10.12ppm (114.6%), conductivity ranged from 0.52mS to 0.67mS, pH ranged from 7.61 to 8.00, and total dissolved solids ranged from 0.28ppt to 0.31ppt. All of these water quality measurements fall within the recommended Canadian Environmental Quality Guidelines (CEQG) for the protection of aquatic life, and the pH level is typically seen in warm freshwater habitats (CCME 2003). Water temperature ranged from 19.5°C (air temperature 20.0°C) to 20.0°C (25.0°C). On May 19, 2021 the water temperature ranged from 21°C at the confluence with the Baden Creek Tributary to 24°C (air temperature 29°C) at the Gingerich Road culvert. Based on these field measurements and available background information, Baden Creek is considered to have a warmwater thermal regime.

Baden Creek provides direct fish habitat including suitable spawning, foraging and rearing habitats for a variety of species. This was confirmed through an electrofishing survey conducted on July 20, 2017, which utilized the single pass screening level assessment method based on the Ontario Stream Assessment Protocol (OSAP, Stanfield 2013). This survey confirmed the presence of 12 species of fish within Baden Creek in the study area, summarized in Table 2. Data provided by the NDMNRF/GRCA indicates the presence of two additional species as well. The species known from Baden Creek exhibit either cool or warmwater thermal preferences and include a combination of highly tolerant and moderately tolerant species. None of the fish species known from Baden Creek are considered to be SAR and no SAR fish or mussels were identified during background review.

| Common Name | Scientific Name | Thermal Preference ¹ | Tolerance ¹ | NRSI (July 20, 2018) | MNRF ² |
|------------------------|----------------------------|------------------------------------|------------------------|----------------------------|-------------------|
| Blacknose Dace | Rhinichthys obtusus | Coolwater | Intermediate | Х | Х |
| Bluntnose Minnow | Pimephales notatus | Warmwater | Intermediate | Х | |
| Creek Chub | Semotilus atromaculatus | Coolwater | Intermediate | х | х |
| Longnose Dace | Rhinichthys cataractae | Coolwater | Intermediate | х | x |
| Northern Hog Sucker | Hypentelium nigricans | Warmwater | Intermediate | х | х |
| Fathead Minnow | Pimephales promelas | Warmwater | Tolerant | х | х |
| Brook Stickleback | Culaea inconstans | Coolwater | Intermediate | Х | X |
| White Sucker | Catostomus commersonii | Coolwater | Tolerant | х | |
| Largemouth Bass | Micropterus salmoides | Warmwater | Tolerant | х | |
| Johnny Darter | Etheostoma nigrum | Coolwater | Tolerant | Х | Х |
| Common Shiner | Luxilus cornutus | Coolwater | Intermediate | х | |
| Central Stoneroller | Campostoma anomalum | Coolwater | Intermediate | х | |
| Emerald Shiner | Notropis atherinoides | Coolwater | Intermediate | | x |
| Rainbow Darter | Etheostoma caeruleum | Coolwater | Intolerant | | X |

Table 3. Fish Species Identified within Baden Creek

¹Eakins R.J. 2018

²GRCA Online Mapping Tool

Huehn Award Drain

The Huehn Award Drain is a permanent watercourse that flows west through residential land and crosses Foundry Street before flowing to Baden Creek at the east side of the study area. Only a small portion (approximately 40m) of the feature occurs within the study area boundary. The drain flows through a perched CSP culvert that extends under Foundry Street, which prevents any fish from moving upstream through the culvert and east of Foundry Street. However, the channel does provide fish habitat between Foundry Street and Baden Creek, confirmed based on the presence of fish with the plunge pool at the outflow from the culvert on both July 20, 2017 and May 19, 2021. Watercress was noted to be abundant both upstream and downstream from Foundry Street within the Huehn Award Drain and iron staining was also observed, suggesting groundwater input to the channel. The Huehn Award Drain is considered to be a coldwater watercourse by the GRCA.

Baden Creek Tributary A

Baden Creek Tributary A originates from a stormwater management facility just beyond the northwest corner of the study area. From the outlet of the facility the tributary crosses under the railway tracks through a perched culvert and into a small plunge pool. The perched culvert would act to prevent fish from moving upstream from the channel towards the stormwater management facility, but could facilitate downstream movement of fish that are flushed downstream from the pond. During the May 19 site visit a small standing pool of water was observed within the plunge pool. A single small, unidentified fish was observed within the pool, potentially from the stormwater pond. Downstream of this pool, the tributary is channelized and straightened for approximately 350m downstream from the railway tracks to a field crossing and cuts through active agricultural field with a narrow (approximately 10m) meadow riparian corridor. The channel is incised with bank heights ranging from 0.45 to 0.9m and bankfull width ranging from 0.8 to 1.5m. The channel was dry during the assessment, although exposed sand and gravel substrates throughout indicates seasonal flows, most likely attributed to spring freshet and large precipitation events. As such, this section of the tributary should be considered ephemeral and is not direct fish habitat.

The lower portion of Baden Creek Tributary A, downstream and east of the field crossing, becomes more incised with higher banks and a wider bankfull width. Some water was observed within this section of the tributary, originating from a tile drain outlet immediately west of the crossing, although flow was very limited. This tile drain appears to drain a large area of active agricultural land to the west. The water flows through a 0.6m diameter corrugated plastic pipe under the crossing and to the channel eastward. From the crossing to its outlet at Baden Creek the bank height ranged from 0.65 to 1.4m and bankfull width ranged from 1.25 to 5.0m. At the time of the assessment wetted widths ranged from 0.15 to 0.95m and water depths ranged from 0.02 to 0.14m. The bank and riparian areas were heavily vegetated with terrestrial grasses and forbs, which grow within, and overhang, the channel. One small-bodied fish was observed within the lower portion of the channel (approximately 20m upstream from Baden Creek) indicating that fish are able to access the lower portion of the tributary when water levels are elevated, but, overall, the tributary does not provide suitable year-round habitat. Generally, water depths were very low throughout the lower portion of the tributary, and in combination with a moderate gradient this would inhibit upstream movement and habitat for fish from Baden Creek. Further, no signs of groundwater input were noted during the assessment and it appears that baseflow to the channel may be provided by the tile drain located at the field

crossing. During the visit on May 19, 2021 the water temperature was 20°C at 13:00 with and air temperature of 29°C.

Baden Creek Tributary B

Baden Creek Tributary B enters the study area from the north and generally flows in a south/southeasterly direction before its confluence with Baden Creek approximately 120m east of Foundry Street. This watercourse primarily drains a combination of urbanized-residential and agricultural lands within the north end of Baden.

At the time of the February site investigation, the watercourse was open and flowing both upstream and downstream of the railway crossing and pedestrian trail. It featured a defined channel and sorted substrates consisting of sand, gravel, silt and some cobble, suggesting that it is a permanent watercourse feature. The wetted width was approximately 1-2m. Water depths ranges from approximately 5cm (riffles and shallow runs) to 22-30cm (pools). Bank height was approximately 0.4-1.3m and there was evidence of erosion, particularly upstream of the trail.

Small amounts of Watercress were observed, which would suggest some groundwater influence. However, due to the relatively low temperature of the water at the time of the assessment (1°C), the amount of groundwater input to the watercourse may be minimal. No fish were observed during the assessment, but the watercourse contained suitable habitat to support fish. The south (downstream) end of the culvert under Snyder's Road could not be accessed to determine whether it provides upstream passage for fish past the road, railway line and pedestrian trail. However, water depth and flows are sufficient to convey fish within the study area watercourse reach if there are no barriers to movement under those features.

Ephemeral Aquatic Habitat Features

Several other ephemeral aquatic habitat features occur across the study area, which were generally poorly-defined and occur within actively tilled agricultural fields. These would act to divert surface runoff from the surrounding landscape to Baden Creek and Baden Creek Tributary A but are generally dry and do not provide fish habitat. These would only carry water following snow melt and heavy precipitation events.

6.0 Natural Environment Significance and Sensitivity

Analysis of the significance of existing natural features was used to identify those features and habitats that are sensitive to disturbance based on the rarity or sensitivity of the feature or the functions/processes that contribute toward their significance. This assessment also considered the policies, legislation, and regulations that apply to the study area natural features which must be considered in the evaluation of the sanitary sewer preferred alignment. The following summarizes significant and sensitive natural features and habitats present within the study area which may represent constraints to site alteration or areas requiring mitigative measures. Policy-based constraints were identified in accordance with the Regional and Township OPs, and the GRCA regulations and policies as summarized in Section 3.0. This assessment incorporates and updates relevant findings and constraints that were identified by NRSI for the purposes of re-zoning of the Schmidt Estate lands, which comprise a large portion of the EA study area.

6.1 Watercourses and Fish Habitat

Baden Creek and the lower portion of the Huehn Award Drain, up to the perched culvert at Foundry Street, provide direct year-round habitat for a variety of cool and warmwater species. Baden Creek is considered a warmwater feature, but water temperatures are maintained at the lower range due to groundwater influence, evidenced by the presence of Watercress and iron staining throughout the channel. The Huehn Award Drain is a coldwater feature, providing coldwater inputs, which also helps to maintain cooler water temperatures within Baden Creek downstream from the confluence.

Baden Creek Tributary B appears to represent a permanently flowing watercourse that likely provides direct fish habitat provided there are no downstream barriers to the study area reach northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection. The thermal regime of this watercourse has not been determined. However, there is evidence of groundwater inputs through the presence of some Watercress within the feature.

Baden Creek Tributary A appears to exhibit both ephemeral and intermittent characteristics. The upper portion of the tributary appears ephemeral, with flows largely provided by outflow from the stormwater pond and snow melt. The lower portion of the tributary, east of the field tractor crossing, appears to be intermittent with baseflow provided by a tile drain, which drains the agricultural lands to the west. No obvious seeps or springs were noted along the channel and no Watercress was observed, suggesting little to no connection to the groundwater table. The tributary near its confluence with Baden Creek provides direct fish habitat for fish accessing from Baden Creek when flows are elevated within the channel, but this is expected to only be seasonal. Overall, Baden Creek Tributary A does not provide suitable habitat for fish due to the very low water levels and extensive portions of the channel that dry out.

The study area watercourses are considered Supporting Environmental Features as described by the policies of the Township OP. Baden Creek, Baden Creek Tributary B and the Huehn Award Drain also represent fish habitat and fall under the protections afforded by the Regional and Township OPs and the federal *Fisheries Act*. Finally, as regulated features, these four watercourses and their adjacent floodplain and slope hazards and associated allowances are subject to the GRCA's Ontario Regulation 150/06 and policies.

Erosion and sedimentation was noted within the aquatic features on site due to their close proximity to active agricultural lands and location downstream from highly developed areas, including stormwater management facilities, which influence the natural hydrologic regimes of each feature. In addition, historical channelization along Baden Creek and Baden Creek Tributary A would also influence the flow regimes of the features and increase erosion and sedimentation. Attempts should aim to manage these issues, where possible, to improve the aquatic habitats within the watercourse features on site. Further, groundwater influence has been noted within Baden Creek, Baden Creek Tributary B and the Huehn Award Drain and should be maintained and improved, if possible.

6.1.1 Watercourse Buffers

In recognition of the ecological sensitivity of the watercourses and fish habitat to disturbances and to mitigate potential for impacts, it is recommended that watercourse buffers be applied to the Baden Creek, Huehn Award Drain and Baden Creek tributary watercourses where they occur within the study area (Map 3). Variable-width buffers from the watercourses are recommended based on their ecological sensitivities. A 30m buffer is recommended from the Huehn Award Drain due to its status as a coldwater watercourse, while 15m buffers are proposed around the majority of the other watercourse reaches. A 10m buffer from the upper reach of Baden Creek Tributary A (upstream of the tractor crossing) is proposed since this reach is ephemeral in nature. It is recommended that this 20m wide corridor spanning the channel upstream of the tractor crossing (10m on each side) should be preserved for future restoration opportunities if possible. Evaluation of a preferred alignment for the trunk sanitary sewer should consider options that aim to avoid the watercourse buffers, or minimize encroachment into these areas, to the extent feasible. This includes avoiding the need for watercourse crossings if possible. Establishing the trunk sanitary sewer alignment further from the watercourses lessens the potential for future impacts in the event of a leakage from the sewer into the adjacent watercourse. If encroachment into these areas cannot be avoided, proper rationale should be identified as part of an evaluation of alternative designs, such as in the consideration of other design evaluation criteria. If encroachments must occur, they should be maintained as well away from the watercourses as possible.

6.2 Woodlands

The study area woodlands comprise the two Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3) communities and the two Mineral Cultural Woodland (CUW1) communities. The woodlands present within the study area do not meet the criteria to be considered Regionally Significant Woodlands in part due to their relatively small sizes (Region of Waterloo 2015). Section 8.1.5.2 (b) of the Township OP identifies "locally significant woodlands" as a category of Supporting Environmental Features (Township of Wilmot 2019). However, locally significant woodlands are not defined in the OP. Based on a precautionary approach, the FOD7-3 woodlands within the study area are assumed to represent Locally Significant Woodlands, and therefore would represent forms of Supporting Environmental Features (Map 3). By contrast, the CUW1 woodlands are of relatively lower ecological quality and functional value. It is therefore suggested that the CUW1 woodlands do not represent Locally Significant Woodlands, but nonetheless should be preserved to the extent feasible.

Negative impacts to these woodland communities should be avoided by selecting a preferred sewer alignment that bypasses these features. If encroachment into woodland communities is unavoidable due to other design, cost or logistical considerations, this must be demonstrated as part of the evaluation of alternatives. Furthermore, see below with respect to potential SAR bat habitat. Impacts to woodlands would require mitigative actions to minimize, restore, and compensate for areas of disturbance. In avoiding woodlands, it is further recommended that sanitary sewer alignment and associated construction zones be maintained at least 5m from the woodland edge dripline in order to minimize potential for impact to tree root zones.

6.3 Wetlands

Two small, previously unmapped wetland features were identified within the north end of the study area. These are shown as small inclusions to the Mineral Cultural Meadow (CUM1) communities on Map 2. These wetlands are anticipated to be of anthropogenic origin, having been created through previous earth-moving activities and water impounding. Nevertheless, as wetland features, they fall under Township OP natural heritage protection as per Section 8.1.5.2 (a), which identifies all non-provincially significant wetlands as "locally significant" wetlands (Township of Wilmot 2019).

Site investigations confirmed the presence of two small and previously unmapped meadow marsh wetland features immediately northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection. These wetlands are associated with Baden Creek Tributary B or its floodplain, and although they may have resulted in part due to anthropogenic impounding of water caused by adjacent pedestrian trail embankments they would also be considered "locally significant". All of the observed wetlands within the study area would also therefore represent Supporting Environmental Features.

It is anticipated that the two small wetland inclusions located near the railway tracks do not meet the definition of "wetland" as defined under the *Conservation Authorities Act* and included in O. Reg. 150/06, as they do not meet the criterion of directly contributing to the hydrological function of a watershed through connection with a surface watercourse (GRCA 2015). However, this would need to be confirmed by the GRCA.

If impacts to any of the study area wetlands cannot be avoided, these wetlands may be eligible for removal in accordance with Sections 8.4.4 (if confirmed to be natural features) or 8.4.5 (if confirmed to be anthropogenic features) due to their sizes being <0.5ha (GRCA 2015). Development or interference with these wetlands may be permitted provided that a series of conditions can be met as listed under these policies, and as demonstrated to the GRCA.

6.3.1 Wetland Buffers

If study area wetlands are to be preserved and do not require removal to accommodate sewer installation, it is recommended that installation and construction activities be set-back from the wetland features to avoid direct or indirect disturbances. Ecological buffers of 15m are typically applied to non-provincially significant wetlands to adequately mitigate adjacent land use effects

and to provide opportunities for buffer enhancement. It is recommended that site alteration be maintained outside of these buffers to the extent feasible.

6.4 Species at Risk

6.4.1 Confirmed Habitat

Based on background information review and field surveys completed within the study area, three SAR, Butternut, Bank Swallow and Barn Swallow, have been confirmed to occur.

Butternut

Based on background information review and field surveys completed within the study area, one SAR, Butternut, has been confirmed to occur.

Three Butternut trees were documented in the study area as shown on Map 3. As described in Section 5.2.2, these trees were each assessed as Category 2 ("retainable") Butternuts, although these results are subject to MECP audit following the submission of a BHA report. As Category 2 trees, these Butternuts are subject to the protections of the ESA. The killing or harming of these trees is prohibited under the ESA unless permitted by the MECP. However, impact to these trees may be exempt from permitting requirements in accordance with Part V of Ontario Regulation 830/21. This regulation allows for a proponent-led process in which activities that would cause Butternut impacts are registered with the MECP, and a mitigation and compensation plan is developed and implemented.

Ontario Regulation 830/21 Section 30 also states that, in regards to the construction or installation of infrastructure, that no permanent structure should be constructed or installed within 25m of a Butternut tree that is to be retained. Twenty-five metre (25m) radius areas around each Butternut tree are shown on Map 3. In accordance with O. Reg. 830/21, if it is not possible to complete the larger servicing project without constructing or installing permanent infrastructure within 25m of a Butternut to be retained, the person or entity carrying out the activity must demonstrate to the MECP why it is not possible to avoid the 25m zone(s), and which alternative options were considered to avoid the 25m zone(s) and determined to not be feasible. MECP must approve in writing the allowance of infrastructure within 25m of the Butternuts.

Bank Swallow and Barn Swallow

Multiple Bank Swallows were observed foraging over the study area lands during one of the 2021 breeding bird surveys, and an individual of this species was observed during one breeding bird survey completed for the Schmidt Estate EIS. Although in each survey year this species was only observed with evidence of "possible" breeding within the study area vicinity, taken together and following a precautionary principle it is assumed that the study area provides confirmed foraging habitat for Bank Swallow.

Although no Barn Swallows were observed foraging during the 2021 breeding bird surveys, multiple Barn Swallows were observed foraging over the Schmidt Estate lands during 2017 surveys. Following a precautionary approach, it is assumed that Barn Swallows still use the study area for foraging.

All of the open study area lands are assumed to represent existing foraging habitat for Bank Swallow and Barn Swallow. This includes the existing agricultural fields, CUM1 meadow and open water pond features, but excludes the FOD7-3 and CUW1 woodland features. Due to the large expanse of these areas within the study area, they are not shown on Map 3. However, as buried infrastructure that will only require temporary construction activities along linear alignment segments, the proposed undertaking will not negatively impact foraging habitat for these species. Bank Swallow and Barn Swallow foraging habitats therefore do not represent development constraints for this project.

6.4.2 Potential Habitat

Bat Species at Risk

Detailed surveys of potential bat roosting or maternity colony habitat were not completed within the study area. All wooded areas have potential to provide suitable habitat for SAR bats. In general, it is recommended that impacts to woodlands be avoided in selecting a preferred design. If impacts to woodlands are unavoidable, additional detailed assessments of bat cavity tree presence may be required subject to MECP consultation. Since individual trees or small stands of trees outside of woodland features may also represent SAR bat roosting habitat, all trees that are ≥10cm diameter-at-breast-height should be considered potential SAR bat habitat until such time that a cavity tree assessment is undertaken to confirm or rule out habitat suitability. Impacts to individual or woodland trees (e.g., removal or pruning) without proper consideration for avoidance or mitigation measures, in consultation with the MECP, may therefore constitute contravention of the ESA. As assumed habitat for SAR, impacts to these

trees is also prohibited under Regional and Township policies except where permitted or authorized by the MECP pursuant to the ESA.

Other Species at Risk Records

Appendix VIII lists SAR that have occurrence records in the study area vicinity (up to 10km) based on review of background information sources. An assessment of habitat suitability was completed for these species based on a review of each species' habitat requirements (e.g., OMNR 2000) versus habitat availability within the study area as confirmed through site investigations. In addition to the SAR described above, three other SAR, American Badger, Bobolink (*Dolichonyx oryzivorus*) and Eastern Meadowlark (*Sturnella magna*) were determined to have suitable habitat within the study area. American Badger habitat suitability was based on the presence of agricultural lands within the general range of where the species is expected to occur. Based on targeted searches for burrow types that are diagnostic of the species, no evidence of badger presence was observed within the study area. Therefore, no impacts to American Badgers or their habitat are expected. The larger blocks of Mineral Cultural Meadow (CUM1) that occur within the study area have potential to provide breeding habitat for Bobolink and Eastern Meadowlark. However, neither of these species was observed during field studies. Suitable habitats for other SAR with occurrence records in the surrounding vicinity are considered absent (Appendix VIII).

6.5 Significant Wildlife Habitat

An assessment of SWH was completed for the study area based on the results of previous natural environment studies undertaken within the study area as well as NRSI's supplementary field surveys completed for the EA. Based on the results of this work, two forms of Candidate SWH were identified for the study area: Bat Maternity Colonies and Turtle Wintering Area. No confirmed SWH was identified within the study area. See Appendix IX for the complete SWH assessment tables.

6.5.1 Candidate Significant Wildlife Habitat

Bat Maternity Colonies

As stated above, targeted assessments of bat habitat suitability were not completed as part of this study based on the general recommendation that woodland features be avoided in the alignment of a trunk sanitary sewer. The Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3) and Mineral Cultural Woodland (CUW1) communities present the potential to support bat maternity colony habitat. SWH for this habitat function is determined based on a minimum density of suitable habitat trees occurring within the feature. In the absence of targeted surveys, these features should be considered potentially significant pending additional survey work. However, the potential for significance is greater within the larger and more ecologically mature FOD7-3 features relative to the CUW1 features.

Turtle Wintering Area

The large pond in the northeast end of the study area, adjacent to Foundry Street, was identified as Candidate SWH for Turtle Wintering Area. Although no turtles were observed within this feature incidentally during field surveys, this feature was not targeted for turtle overwintering survey work. Depending on its depth and substrates, it may provide an unfrozen lower water column and muddy substrates during winter to allow for turtle overwintering.

The SCC Snapping Turtle was also identified as having suitable habitat within the study area (Appendix VIII). This is in part based on the presence of the pond, as well as the Baden Creek riparian channel which may function as a movement corridor across the lands. However, the potentially significant habitat for this SCC is considered to be the pond and potential overwintering habitat area. The significance of the Baden Creek channel as a general wildlife movement corridor is further discussed below.

6.5.2 Significant Wildlife Habitats Considered Absent

Based on species records known from the vicinity, one additional SCC was identified as having suitable habitat within the study area: Monarch (*Danaus plexippus*). This species was not observed on site during field surveys, although it is likely that the species does use the property due to the presence of its host plant Common Milkweed (*Ascpelias syriaca*). However, the most sensitive form of this species' habitat within Ontario is considered to be its migratory stop-over habitat, which is specifically categorized as a SWH type (MNRF 2015a). Only suitable habitat areas within 5km of Lake Ontario are considered SWH within Ecoregion 6E, ruling out potential habitat significance in the study area. However, measures should be taken to minimize or mitigate Monarch habitat loss within the study area nonetheless.

6.6 Regionally Significant Species

In total, one regionally significant vegetation species, five regionally significant bird species, and three regionally significant mammal species were observed within the study area, either as part of field studies completed for this EA or for the previous Schmidt Estate lands study. Of the

regionally significant mammal species, Coyote and Virginia Opossum may no longer be considered significant on the basis of their being regionally "scarce" or "rare", respectively, relative to the age of the regional mammal significance rankings (1985).

While regionally rare vegetation species in and of themselves do not represent policy-based constraints to sewer design or construction plans, their presence renders additional significance to the features in which they are located. Measures should be taken to avoid or minimize impacts to habitats that support regionally significant species. Efforts should be made to protect individual significant plants during construction where necessary. If impacts cannot be avoided, the individual plants should be relocated to an appropriate site when feasible.

6.7 Ecological Linkages

Ecological connectivity within the study area is primarily associated with the Baden Creek riparian corridor. This corridor represents the primary naturalized feature within the study area that is otherwise dominated by active agriculture. It therefore provides a potential corridor function for land-based wildlife that may avoid the open and sun-exposed agricultural and cultural meadow lands (e.g., amphibians).

The significance of this ecological corridor as a linkage feature may be limited to a local scale for wildlife movement purposes. Although it connects the northeast study area pond, and potentially further upstream riparian features along Baden Creek, downstream habitats are very limited due to existing land developments and ultimately the barrier presented by Highway 7/8 to the south. The study area may provide connectivity to upstream wooded and riparian wetland habitats along the Huehn Award Drain depending on the ability and inclination of wildlife to cross through the Foundry Street culvert or over top of the road. Minor upstream connectivity may also be provided along Baden Creek Tributary B, although the road, railway and pedestrian trail crossings at/adjacent to Snyder's Road may represent a movement barrier. A minor ecological connection may also extend along the Baden Creek Tributary A corridor toward the stormwater management pond to the northwest.

Linkages within the study area may be of minor or localized ecological significance. However, efforts should be made to preserve these connections as part of future site alteration activities. Where feasible these corridors should be enhanced in size and quality to promote their functional value, such as through the enhancement of recommended watercourse buffers.

7.0 References

- Bird Studies Canada (BSC), Environment Canada's Canadian Wildlife Service, Ontario Nature, Ontario Field Ornithologists and Ontario Ministry of Natural Resources. 2008. Ontario Breeding Bird Atlas Database. http://www.birdsontario.org/atlas/aboutdata.jsp?lang=en
- Birds Canada. 2021. Ontario Breeding Bird Atlas. Instructions for General Atlassing. Updated April 2021. https://www.birdsontario.org/atlassing/.
- Department of Fisheries and Oceans (DFO). 2021. Aquatic Species at Risk Map. Prepared by Fisheries and Oceans Canada. Available online at: http://www.dfo-mpo.gc.ca/species-especes/sara-lep/map-carte/index-eng.html.
- Dobbyn, J.S. 1994. Atlas of the Mammals of Ontario. Don Mills, Federation of Ontario Naturalists.
- eBird. 2020. eBird records (all years, year-round) for vicinity of subject property. Accessed June 2021.
- Government of Canada. 2021. Species at Risk Public Registry. https://www.canada.ca/en/environment-climate-change/services/species-risk-public-registry.html. Last updated September 16, 2021.
- Grand River Conservation Authority (GRCA). 2015. Policies for the Administration of the Development, Interference with Wetlands and Alterations to Shorelines and Watercourses Regulation. Ontario Regulation 150/06. Amended October 23, 2015.
- Grand River Conservation Authority (GRCA). 2021. Grand River Conservation Network: Interactive Mapping Tool. https://www.grandriver.ca/en/Planning-Development/Map-Your-Property.aspx? mid =21
- iNaturalist. 2020. Manual search of records within approximately 1km of the study area. https://www.inaturalist.org/observations?place_id=132052. Accessed June 2021.
- Lee, H.T., W.D. Bakowsky, J. Riley, J. Bowles, M. Puddister, P. Uhlig and S. McMurray. 1998. Ecological Land Classification for Southern Ontario: First Approximation and its Application. Ontario Ministry of Natural Resources, Southcentral Science Section, Science Development and Transfer Branch. SCSS Field Guide FG-02.
- MacNaughton, A., R. Layberry, C. Jones and B. Edwards. 2020. Ontario Butterfly Atlas Online. http://www.ontarioinsects.org/atlas_online.htm.
- Ontario Ministry of Environment, Conservation and Parks (MECP). 2020. Species at Risk in Ontario List. https://www.ontario.ca/page/species-risk-ontario. Last updated November 9, 2020.
- Ontario Ministry of Municipal Affairs and Housing (OMMAH). 2020. Provincial Policy Statement. Approved by the Lieutenant Governor in Council, Order in Council No. 229/2020.
- Ontario Ministry of Natural Resources (OMNR). 2000. Significant Wildlife Habitat Technical Guide. October 2000.

- Ontario Ministry of Natural Resources (OMNR). 2010. Natural Heritage Reference Manual for Policies of the Provincial Policy Statement, Second Edition. March 18, 2010.
- Ontario Ministry of Natural Resources and Forestry (MNRF). 2015a. Significant Wildlife Habitat Criteria Schedules for Ecoregion 6E. January 2015.
- Ontario Ministry of Natural Resources and Forestry (MNRF). 2015b. Butternut Health Assessor's Field Guide. Produced for the Ontario Ministry of Natural Resources and Forestry by the Forest Gene Conservation Association. 2015 Edition.
- Ontario Ministry of Natural Resources and Forestry (MNRF). 2017. Ontario Odonata Atlas Database. Natural Heritage Information Centre, Ontario Ministry of Natural Resources and Forestry. Species list from atlas squares 17NJ20.
- Ontario Ministry of Natural Resources and Forestry (MNRF). 2021a. Make a Map: Natural Heritage Areas. Natural Heritage Information Centre (NHIC). http://www.gisapplication.lrc.gov.on.ca/mamnh/Index.html?site=MNR_NHLUPS_Natural Heritage&viewer=NaturalHeritage&locale=en-US.
- Ministry of Natural Resources and Forestry (MNRF). 2021b. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2021-06-03. Available: https://www.ontario.ca/page/get-natural-heritage-information
- Ontario Nature. 2019. Reptiles and Amphibians of Ontario Range Maps. http://www.ontarioinsects.org/herpatlas/herp_online.html.
- Region of Waterloo. 1985. Appendix 4: Mammals in Environmentally Sensitive Policy Areas Technical Appendix. Approved by Council: 1986.

Region of Waterloo. 2015. Regional Official Plan. June 18, 2015.

Richardson, W. and V. Martin. 1999. Regional Municipality of Waterloo. Revisions to Waterloo Region's Significant Species List: Native Vascular Plant Component.

Township of Wilmot. 2019. Township of Wilmot Official Plan. April 2019 Consolidation.

Maps











NAD83 - UTM Zone 17 Size: 11x17" **1:5,000** Project: 2561 Date: August 18, 2022 300 Metres

Appendix I Schmidt Estate EIS Study Area



Appendix II Plant Species Inventoried within the Study Area

Plant Species Reported from the Study Area - Beckdale Sanitary Sewer EA (Project #2561)

| Scientific Name | Common Name | сс | SRANK | SARO | COSEWIC | SARA | SARA Schedule | Waterloo Region | NRSI Observed | CUM1 (w/ MAM2 incl.) | FOD7-3 | CUW1 |
|--|------------------------------|--------------------|------------|------------|-------------------|-------------------|-------------------|--------------------|------------------|-------------------------|----------|------|
| | | Oldham et al. 1998 | MNRF 2020a | MNRF 2020a | ernment of Canada | ernment of Canada | ernment of Canada | hardson & Martin 1 | 999 | | | |
| Gymnosperms | Conifers | | | | | | | | | | | |
| Cupressaceae | Cypress Family | | | | | | | | | | | |
| Juniperus virginiana | Eastern Red Cedar | 4 | S5 | | | | | | Х | X | | |
| Thuja occidentalis | Eastern White Cedar | 4 | S5 | | | | | | Х | X | | |
| Pinaceae | Pine Family | | | | | | | | | | | |
| Picea abies | Norway Spruce | 0 | SE3 | | | | | | Х | X | | Х |
| Pinus strobus | Eastern White Pine | 4 | S5 | | | | | | Х | | | Х |
| Dicotyledons | Dicots | | | | | | | | | | | |
| Aceraceae | Maple Family | | | | | | | | | | | |
| Acer negundo | Manitoba Maple | 0 | S5 | | | | | | Х | Х | Х | х |
| Acer platanoides | Norway Maple | 0 | SE5 | | | | | | Х | | | х |
| Acer saccharinum | Silver Maple | 5 | S5 | | | | | | Х | Х | | |
| Asclepiadaceae | Milkweed Family | | | | | | | | | | | |
| Asclepias svriaca | Common Milkweed | 0 | S5 | | | | | | Х | X | | |
| Asteraceae | Composite or Aster Family | - | | | | | | | | | | |
| Ambrosia trifida | Great Ragweed | 0 | S5 | | | | | | Х | X | X | |
| Bidens cernua | Nodding Beggarticks | 2 | S5 | | | | | | X | | X | |
| Erigeron annuus | Annual Fleabane | 0 | S5 | | | | | | X | | ~ | x |
| Erigeron strigosus | Rough Eleabane | 4 | S5 | | | | | | X | x | | |
| Helianthus pauciflorus ssp. subrhomboide | Rhombic-leaved Sunflower | 0 | \$2\$3 | | | | | | X | x | | |
| Solidago altissima | Tall Goldenrod | 1 | \$5 | | | | | | X | X | | |
| Solidago canadensis | Canada Goldenrod | 1 | <u> </u> | | | | | | X | X | | |
| Solidago nemoralis | Grav-stemmed Goldenrod | 2 | <u> </u> | | | | | | X | X | | |
| Sonchus arvensis | Field Sow-thistle | 0 | SE5 | | | | | | X | X | | |
| Symphyotrichum ericoides | White Heath Aster | 0 | <u>S5</u> | | | | | | X | X | | |
| Symphyotrichum lancoolatum | Papieled Aster | | 85 | | | | | | X | × | | |
| Symphyotrichum lateriflorum | Calico Aster | 3 | 85 | | | | | | X | ~ | | v |
| Symphyotrichum novae angliae | Now England Actor | 2 | 55 | | | | | | × | v | | ^ |
| Symphyotrichum niosum | Old Field Aster | 0 | 85 | | | | | | X | × | | |
| Tragonagon protonoio | Maadaw Caat's baard | 0 | | | | | | | × | ^ | × | |
| | Coltio fact | 0 | SED | | | | | | × | | ~ | |
| Palsaminaceae | Touch me not Family | U | 3E5 | | | | | | ^ | | | |
| | Spotted Jowelwood | 4 | \$5 | | | | | | × | | × | |
| Peregineeee | Spotted Jewelweed | 4 | | | | | | | ^ | | ^ | |
| Boraginaceae | | 0 | 055 | | | | | | Y | × | | |
| | Common Viper's Bugioss | 0 | SED | | | | | | X | ^ | ~ | |
| | | 0 | 55 | | | | | | X | × | X | × |
| Brassianana | Mustand Forget-me-not | 0 | 3E4 | | | | | | ^ | ^ | ^ | ^ |
| Brassicaceae | Mustard Family | 0 | 055 | | | | | | Y | × | | × |
| Alliaria petiolata | Garlic Mustard | 0 | SE5 | | | | | | X | X | <u>×</u> | X |
| Barbarea vulgaris | | 0 | SED | | | | | | X | × | Χ. | |
| Capsella bursa-pastoris | Common Snephera's Purse | 0 | SE5 | | | | | | X | X | | |
| Erysimum cheiranthoides | Wormseed Walifiower | 0 | 55 | | | | | | X | X | X | × |
| Hesperis matronalis | Dame's Rocket | 0 | SE5 | | | | | | X | × × | X | X |
| Rorippa paiustris | | 3 | 55 | | | | | | X | X | | |
| Caprifoliaceae | Honeysuckle Family | | 055 | | | | | | | | | |
| Lonicera tatarica | | 0 | SE5 | | | | | | X | | | X |
| I riosteum aurantiacum | Orange-truited Horse-gentian | 7 | S4S5 | | | | | | X | | | X |
| Viburnum opulus ssp. opulus | Cranberry Viburnum | 0 | SE3? | | | | | | Х | | Х | |
| Caryophyllaceae | Pink Family | | 0=- | | | | | | | | | |
| Saponaria officinalis | Bouncing-bet | 0 | SE5 | | | | | | X | | | X |
| Convolvulaceae | Morning-glory Family | | | | | | | | | | | |
| Calystegia sepium | Hedge False Bindweed | 2 | S5 | | | | | | Х | | Х | |
| Convolvulus arvensis | Field Bindweed | 0 | SE5 | | | | | | Х | X | | |
| Cornaceae | Dogwood Family | | | | | | | | | | | |

| | 1 | - | | 1 | | 1 | | | | | | 1 |
|------------------------|-------------------------------------|---|-----------|-----|-----|------------|------------|-----|---|----------|---|---|
| Cornus alternifolia | Alternate-leaved Dogwood | 6 | S5 | | | | | | X | | X | |
| Cornus sericea | Red-osier Dogwood | 2 | S5 | | | | | | Х | X | X | |
| Crassulaceae | Stonecrop Family | | | | | | | | | | | |
| Penthorum sedoides | Ditch-stonecrop | 4 | S5 | | | | | | Х | | Х | |
| Cucurbitaceae | Gourd Family | | | | | | | | | | | |
| Echinocystis lobata | Wild Mock-cucumber | 3 | S5 | | | | | | Х | | Х | |
| Elaeagnaceae | Oleaster Family | | | | | | | | | | | |
| Elaeagnus umbellata | Autumn Olive | 0 | SE3 | | | | | | Х | X | | |
| Fabaceae | Pea Family | | | | | | | | | | | |
| Melilotus albus | White Sweet-clover | 0 | SE5 | | | | | | Х | X | | |
| Trifolium repens | White Clover | 0 | SE5 | | | | | | Х | X | | |
| Vicia cracca | Tufted Vetch | 0 | SE5 | | | | | | Х | X | | |
| Quercus macrocarpa | Bur Oak | 5 | S5 | | | | | | Х | | | X |
| Geraniaceae | Geranium Family | | | | | | | | | | | |
| Geranium robertianum | Herb-Robert | 2 | S5 | | | | | | Х | | Х | Х |
| Grossulariaceae | Currant Family | | | | | | | | | | | |
| Ribes americanum | Wild Black Currant | 4 | S5 | | | | | | Х | | Х | Х |
| Ribes cynosbati | Prickly Gooseberry | 4 | S5 | | | | | | Х | | Х | |
| Ribes rubrum | Northern Red Currant | 0 | SE5 | | | | | | Х | | х | х |
| Hippocastanaceae | Buckeve Family | | | | | | | | | | | |
| Aesculus hippocastanum | Horse Chestnut | 0 | SE2 | | | | | | Х | X | | |
| Juglandaceae | Walnut Family | | | | | | | | | | | |
| Juglans cinerea | Butternut | 6 | S2? | END | END | Endangered | Schedule 1 | | Х | | X | |
| Juglans nigra | Black Walnut | 5 | S4? | | | | | R+* | X | × | X | x |
| Lamiaceae | Mint Family | | | | | | | | | | | |
| Clinopodium vulgare | Field Basil | 4 | S5 | | | | | | X | | | X |
| Glechoma hederacea | Ground by | 0 | SE5 | | | | | | X | × | | x |
| | Common Dead-nettle | 0 | SE3 | | | | | | X | X | | |
| | Common Motherwort | 0 | SE5 | | | | | | X | ~ | x | x |
| Leonards cardiaca | American Water-borebound | 4 | <u>S5</u> | | | | | | X | | x | X |
| | Northern Water-borebound | 5 | <u> </u> | | | | | | X | | X | |
| Mentha canadensis | Canada Mint | 3 | | | | | | | X | | X | |
| Mentha x piperita | (Mentha aquatica X Mentha spicata) | 0 | SNA | | | | | | X | | X | |
| Malvaceae | Mallow Family | 0 | SINA | | | | | | ^ | | ^ | |
| Malva poglocta | Dworf Choosewood | 0 | SE5 | | | | | | × | v | | |
| Maragaza | Mulberry Family | 0 | 0L3 | | | | | | ~ | ~ | | |
| Morus alba | | 0 | SE5 | | | | | | × | v | | |
| Opagraceae | Evoning primroso Family | 0 | 3E3 | | | | | | ^ | <u> </u> | | |
| Cireana considensia | Broad logyed Englanter's Nightshade | 2 | C.E. | | | | | | × | - | × | × |
| | Broad-leaved Enchanter's Nightshade | 2 | 33 | | | | | | X | × | ^ | ^ |
| | Small-nowered willowherb | 0 | 364 | | | | | | ^ | ^ | | |
| Oxalidaceae | | 0 | 05 | | | | | | X | | | × |
| Dians stricta | Dennes Family | 0 | | | | | | | ^ | | | ^ |
| Chalidanium maiua | Creater Calandina | 0 | 055 | | | | | | V | v | | |
| | Blantain Family | 0 | SE5 | | | | | | X | X | | |
| Plantaginaceae | Figure Plantain Family | 0 | 055 | | | | | | V | ~ | | |
| Plantago lanceolata | | 0 | SED | | | | | | X | × | | |
| Plantago major | | 0 | SE5 | | | | | | X | X | | |
| Polygonaceae | Smartweed Family | | 075 | | | | | | | | | |
| Persicaria maculosa | Spotted Lady's-thumb | 0 | SE5 | | | | | | X | | X | |
| Reynoutria japonica | Japanese Knotweed | 0 | SE5 | | | | | | X | X | | |
| Rumex britannica | vvater Dock | 6 | \$5 | | | | | | X | X | | |
| Rumex obtusifolius | Bitter Dock | 0 | SE5 | | | | | | Х | | X | |
| Ranunculaceae | Buttercup Family | | | | | | | | | | | |
| Caltha palustris | Yellow Marsh Marigold | 5 | S5 | | | | | | Х | | X | |
| Ranunculus repens | Creeping Buttercup | 0 | SE5 | | | | | | Х | X | X | |
| Ranunculus sceleratus | Cursed Buttercup | 2 | S5 | | | | | | Х | | x | |
| Rhamnaceae | Buckthorn Family | | | | | | | | | | | |
| Rhamnus cathartica | Common Buckthorn | 0 | SE5 | | | | | | Х | | X | X |
| Rosaceae | Rose Family | | | | | | | | | | | |
| Amelanchier spicata | Running Serviceberry | 4 | S4 | | | | | R | Х | X | | |
| Crataegus sp. | Hawthorn sp. | 0 | 0 | 0 | 0 | | 0 | | Х | | | x |

| | - | | | - | | | | - | | |
|--------------------------------|---------------------------------|---|-----|---|------|---|--------------|---|----------|----------|
| Crataegus punctata | Dotted Hawthorn | 4 | S5 | | | | Х | X | | Х |
| Fragaria virginiana | Wild Strawberry | 2 | S5 | | | | Х | X | | х |
| Geum canadense | White Avens | 3 | S5 | | | | Х | | | Х |
| Geum laciniatum | Rough Avens | 4 | S4 | | | | Х | Х | | |
| Geum urbanum | Wood Avens | 0 | SE3 | | | | Х | | Х | Х |
| Geum x catlingii | (Geum canadense X Geum urbanum) | 0 | SNA | | | | Х | | | Х |
| Malus pumila | Common Apple | 0 | SF4 | | | | X | x | x | x |
| Potentilla indica | Mock-strawberry | 0 | SE2 | | | | × × | ~ | Y | ~ |
| Prunus avium | Sweet Cherry | 0 | SE4 | | | | × | × | ~ | |
| Brunus acreting | Black Charny | 2 | 014 | | | | × | ~ | | × |
| Prunus serolina | | 3 | 35 | | | | X | × | × | × |
| Prunus virginiana | | 2 | 55 | | | | <u> </u> | × | <u> </u> | <u> </u> |
| Rosa multiflora | Multiflora Rose | 0 | SE5 | | | | <u> </u> | X | X | X |
| Rubus idaeus ssp. strigosus | Wild Red Raspberry | 2 | \$5 | | | | X | | X | X |
| Rubus occidentalis | Black Raspberry | 2 | S5 | | | | X | | X | |
| Rubiaceae | Madder Family | | | | | | | | | |
| Galium aparine | Cleavers | 4 | S5 | | | | Х | | X | X |
| Galium mollugo | Smooth Bedstraw | 0 | SE5 | | | | Х | X | X | |
| Salicaceae | Willow Family | | | | | | | | | |
| Populus balsamifera | Balsam Poplar | 4 | S5 | | | | Х | X | | |
| Populus tremuloides | Trembling Aspen | 2 | S5 | | | | Х | X | | |
| Salix alba | White Willow | 0 | SE4 | | | | Х | X | Х | Х |
| Salix euxina | Crack Willow | 0 | SE | | | | Х | | Х | |
| Scrophulariaceae | Figwort Family | | | | | | | | | |
| Veronica anagallis-aquatica | Water Speedwell | 0 | SE | | | | X | | X | |
| | Fim Family | Ŭ | 02 | | | | Χ | | ~ | |
| | Amorican Elm | 2 | 85 | | | | v | | | × |
| | Nettle Femily | 5 | | | | | ^ | | | ^ |
| | Nettie Family | 0 | 05 | | | - | V | X | | |
| | | 2 | 55 | | | | <u> </u> | | × | |
| Unica dioica ssp. dioica | European Stinging Nettie | 0 | SE2 | | | | X | | X | |
| Verbenaceae | Vervain Family | | | | | | | | | |
| Verbena hastata | Blue Vervain | 4 | \$5 | | | | X | | X | |
| Verbena urticifolia | White Vervain | 4 | S5 | | | | Х | | X | |
| Violaceae | Violet Family | | | | | | | | | |
| Viola odorata | English Violet | 0 | SE2 | | | | Х | | | X |
| Viola sororia | Woolly Blue Violet | 4 | S5 | | | | Х | | X | Х |
| Monocotyledons | Monocots | | | | | | | | | |
| Alismataceae | Water-plantain Family | | | | | | | | | |
| Sagittaria latifolia | Broad-leaved Arrowhead | 4 | S5 | | | | Х | | Х | |
| Araceae | Arum Family | | | | | | | | | |
| Arisaema triphyllum | Jack-in-the-pulpit | 5 | S5 | | | | Х | | Х | |
| Cyperaceae | Sedge Family | | | | | | | | | |
| Carex cristatella | Crested Sedge | 3 | S5 | | | | Х | | X | |
| Carex granularis | Limestone Meadow Sedge | 3 | S5 | | | | Х | Х | | |
| Carex pallescens | Pale Sedge | 5 | S4 | | | | Х | | | Х |
| Carex spicata | Spiked Sedae | 0 | SE5 | | | | Х | | x | |
| Carex stipata | Awl-fruited Sedge | 3 | S5 | | | | X | x | | |
| Carex vulpinoidea | Fox Sedge | 3 | S5 | | | | X | | x | |
| Schoenonlectus tabernaemontani | Soft-stemmed Bulrush | 5 | S5 | | | | X | | x | |
| Iridaçõa | | Ű | 00 | | | | Χ | | ~ | |
| | Vollow Iric | 0 | SE4 | | | | v | | × | |
| luneases | Puch Femily | 0 | 364 | | | | ^ | | ^ | |
| Juncaceae | | 6 | 05 | | | | X | | N N | |
| Juncus articulatus | | 5 | 55 | | | | X | | X | |
| Juncus tenuis | | U | 55 | | | | Х | X | | |
| Liliaceae | Lily Family | | | | | - | | | | |
| Allium tricoccum | Wild Leek | 7 | S4 | | | | Х | | X | |
| Poaceae | Grass Family | | | | | | | | | |
| Agrostis gigantea | Redtop | 0 | SE5 | | | | Х | X | | |
| Agrostis stolonifera | Creeping Bentgrass | 0 | SE5 | | | | Х | | X | |
| Bromus inermis | Smooth Brome | 0 | SE5 | | | | Х | X | | |
| Dactylis glomerata | Orchard Grass | 0 | SE5 | | | | Х | X | X | х |
| Danthonia spicata | Poverty Oatgrass | 5 | S5 | | | | Х | X | | |
| | | | | | | | | | | |

| Festuca rubra | Red Fescue | 0 | S5 | | | Х | Х | | |
|-------------------------------------|-----------------------|---|-----|--|--|-----|----|----|----|
| Hordeum jubatum | Foxtail Barley | 0 | S5? | | | Х | Х | | |
| Leersia oryzoides | Rice Cutgrass | 3 | S5 | | | Х | | Х | |
| Lolium pratense | Meadow Fescue | 0 | SE5 | | | Х | Х | | |
| Phleum pratense | Common Timothy | 0 | SE5 | | | Х | Х | | |
| Phragmites australis ssp. australis | European Reed | 0 | SE5 | | | Х | Х | | |
| Poa annua | Annual Bluegrass | 0 | SE5 | | | Х | Х | | |
| Poa compressa | Canada Bluegrass | 0 | SE5 | | | Х | Х | | |
| Poa palustris | Fowl Bluegrass | 5 | S5 | | | Х | | X | |
| Poa pratensis | Kentucky Bluegrass | 0 | S5 | | | Х | Х | | |
| Potamogetonaceae | Pondweed Family | | | | | | | | |
| Potamogeton nodosus | Long-leaved Pondweed | 7 | S5 | | | Х | | X | |
| Stuckenia pectinata | Sago Pondweed | 4 | S5 | | | Х | | X | |
| Typhaceae | Cattail Family | | | | | | | | |
| Typha angustifolia | Narrow-leaved Cattail | 0 | SE5 | | | X | x | | |
| Total | | | | | | 141 | 72 | 66 | 41 |

*NHIC Atlas Squares: Square #

References

Oldham, M.J., W.D. Bakowsky and D.A. Sutherland. 1995. Floristic quality assessment for southern Ontario. OMNR, Natural Heritage Information Centre, Peterborough. 68 pp.

Ministry of Natural Resources and Forestry (MNRF). 2020a. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2020-01-17. Available: https://www.ontario.ca/page/get-natural-heritage-information Government of Canada. 2020. Species at Risk Public Registry: Species Search. COSEWIC Last Assessment Date: 2020-11-04. Available: https://species-registry.canada.ca/index-en.html#/species?sortBy=commonNameSort&sortDirection=asc&pageSize=10 Richardson, W. and V. Martin. 1999. Regional Municipality of Waterloo. Revisions to Waterloo Region's Significant Species List: Native Vascular Plant Component.

Ministry of Natural Resources and Forestry (MNRF). 2020b. Natural Heritage Information Centre (NHIC): Make a Natural Heritage Area Map Application. Published: 2014-07-17. Updated 2020-01-28. Available: https://www.ontario.ca/page/make-natural-heritage-area-map
Appendix III Bird Species Reported from the Study Area and Vicinity

Bird Species Reported from the Study Area - Baden Sanitary Sewer EA (Project #2561)

| Scientific Name | Common Name | SRANK | SARO | COSEWIC | SARA | SARA Schedule | Region of Waterloo Status | OBBA* | NRSI Observed: Highest Level of Breeding Evidence | Aq | CUM1 | FOD7-3 | Res. | cuw | Off-site |
|--|-------------------------------|------------|-------------|---------------|---------------|------------------|---------------------------------|---------------|---|------|------|--------|------|-----|----------|
| | | MNRF 2021a | MNRF 2021a | Government of | Government of | Government of | Martin 1996 | Cadman et al. | NRSI Results from | 2021 | | | | | |
| Archidae | Duska Casaa & Surana | | MININ LOLIG | Canada 2021 | Canada 2021 | Canada 2021 | Maran 1000 | 2007 | | 1 | 1 | | | | 1 |
| Anatidae | Mallard | 85 | - | - | - | | | C0 | 0.0 | PO | DP | | | | <u> </u> |
| Anas platymynchos Branta canadensis | Canada Goose | 30 | | | | | | 00 | 80 | FU | FR | | | | PO |
| Lophodutes cucultatus | Hooded Merganser | | | | | | al | 00 | PO | | | | | | PO |
| Phasianidae | Partridges Grouse & Turkeys | | | | | | | | 10 | | | | | | 10 |
| Meleagris gallopavo | Wild Turkey | S5 | | | | | | PR | | | | | | | |
| Phasianus colchicus | Ring-necked Pheasant | SNA | | | | | | CO | | | | | | | |
| Columbidae | Pigeons & Doves | | | | | | | | | | | | | | |
| Columba livia | Rock Pigeon | SNA | | | | | | со | PR | | PR | | | | PO |
| Zenaida macroura | Mourning Dove | S5 | | | | | | со | PR | PR | PO | | PO | PO | PO |
| Cuculiformes | Cuckoos & Anis | | | | | | | | | | | | | | |
| Coccyzus erythropthalmus | Black-billed Cuckoo | S4S5B | | | | | V | CO | | | | | | | |
| Trochilidae | Hummingbirds | | | | | | | | | | | | | | |
| Archilochus colubris | Ruby-throated Hummingbird | S5B | | | | | V | CO | | | | | | | |
| Charadriidae | Plovers & Lapwings | | | | | | | | | | | | | | |
| Charadrius vociferus | Killdeer | S4B | | | | | | со | PO | PO | | | | | |
| Scolopacidae | Sandpipers & Allies | | | | | | | | | | | | | | |
| Actitis macularia | Spotted Sandpiper | S5B | | | | | | PO | | | | | | | |
| Ardeidae | Herons & Bitterns | | | | | | - | | | | | | | | |
| Ardea herodias | Great Blue Heron | S4 | | | | | √ | co | PO | | | | | | PO |
| Cathartidae | Vultures | 055,000 | | | | | 1 | | | | | | | | |
| Cathartes aura | Turkey Vulture | 55B, 53N | | | | | N | PO | | | | | | | |
| Accipitridae | Hawks, Kites, Eagles & Allies | 85 | NAR | NAR | NC | No oshodulo | | | - | | | | | | |
| Strigidag | Typical Owle | | INAR | INAR | 113 | NO SCHEQUIE | | 0 | | | | | | | |
| Sulgidae | Great Horned Owl | \$4 | | - | - | | | DP | - | | | | | | |
| Menascons asio | Eastern Screech-Owl | 54 S4 | NAR | NAR | NS | No schedule | | PR | | | | | | | |
| Alcedinidae | Kingfishers | | - IV-U | 10/03 | 110 | No Schedule | | | | | | | | | |
| Megacervle alcvon | Belted Kingfisher | S5B, S4N | | | | | V | PO | | | | | | | |
| Picidae | Woodpeckers | | | | | | | | | | | | | | |
| Colaptes auratus | Northern Flicker | S5 | 1 | | | | | со | | | | | | | |
| Dryobates pubescens | Downy Woodpecker | S5 | | | | | | со | PR | | | PR | | PO | |
| Dryobates villosus | Hairy Woodpecker | S5 | | | | | | CO | | | | | | | |
| Dryocopus pileatus | Pileated Woodpecker | S5 | | | | | \checkmark | PO | | | | | | | |
| Melanerpes carolinus | Red-bellied Woodpecker | S5 | | | | | 1 | PR | | | | | | | |
| Melanerpes erythrocephalus | Red-headed Woodpecker | S3 | SC | E | E | Schedule 1 | V | со | | | | | | | |
| Sphyrapicus varius | Yellow-bellied Sapsucker | S5B, S3N | | | | | 1 | со | | | | | | | |
| Tyrannidae | Tyrant Flycatchers | | | | | | | | | | | | | | |
| Contopus virens | Eastern Wood-Pewee | S4B | SC | SC | SC | Schedule 1 | | PR | | | | | | | |
| Empidonax traillii | Willow Flycatcher | S4B | | | | | | | РО | PO | | | | | |
| Mylarchus crinitus | Great Crested Flycatcher | 55B | | | | | | PR | | | | | | | |
| Sayornis phoebe | Eastern Phoebe | S5B | | | | | | 00 | | | DD | | | | |
| Viroopidao | Viroos | 340 | | | | | | 0 | FR | | FR | | FU | | |
| Vireo flavifrons | Vellow-throated Vireo | S4B | | | | | N | PO | | | | | | | |
| Vireo alvus | Warbling Vireo | S5B | | - | | | | PR | PO | | PO | PO | | PO | PO |
| Vireo olivaceus | Red-eved Vireo | S5B | | | | | | PR | PO | | PO | PO | | | |
| Corvidae | Crows & Javs | | | | | | | | | | | | | | |
| Corvus brachyrhynchos | American Crow | S5 | | | | | | со | PO | PO | | | | | |
| Cyanocitta cristata | Blue Jay | S5 | | | | | | со | PO | | PO | PO | PO | | |
| Alaudidae | Larks | | | | | | | | | | | | | | |
| Eremophila alpestris | Horned Lark | S4 | | | | | | со | PR | PR | | | | | |
| Hirundinidae | Swallows | | | | | | | | | | | | | | |
| Hirundo rustica | Barn Swallow | S4B | THR | SC | Т | Schedule 1 | | CO | | | | | | | |
| Petrochelidon pyrrhonota | Cliff Swallow | S4S5B | | | | | √* | CO | PO | | PO | | | | |
| Riparia riparia | Bank Swallow | S4B | THR | Т | Т | Schedule 1 | | | PO | PO | PO | | | | |
| Stelgidopteryx serripennis | Northern Rough-winged Swallow | S4B | | | | | | | PO | | PO | | | | |
| Tachycineta bicolor | Tree Swallow | S4S5B | | | | | | со | | | | | | | ļ |
| Paridae | Chickadees & Titmice | | | | | | | | | | | | | | |
| Poecile atricapillus | Black-capped Chickadee | S5 | | | | | | CO | CO | | CO | PR | | | |

| Sittidae | Nuthatches | | | | | | | | | | | | | | |
|---------------------------|----------------------------------|----------|-----|------|------|-------------|----|----|-----|----|----|----|----|-----------|----|
| Sitta canadensis | Red-breasted Nuthatch | \$5 | | | | | A | | PO | | | PO | | | |
| Sitta carolinensis | White-breasted Nuthatch | 85 | | | | | | 00 | 10 | | | 10 | | | |
| Conthiidao | Croopers | | | | | | | 00 | | | | | | | |
| Certhia americana | Brown Creeper | 85 | | | | | al | PO | | | | | | | |
| Troglodutidao | Wrone | | | | | | v | FO | | | | | | | |
| Troglodytidae | | 050 | | - | | | | 60 | | | | PO | | | |
| Troglodytes aedon | House wren | SOB | | | | | .1 | | PU | | P0 | P0 | | | |
| Turdidae | Thrushes | 55B, 54N | | | | | N | PR | | | | | | | |
| Hulocichle musteline | Wood Thruph | C4P | 80 | - | - T | Sabadula 1 | | DD | | | | | | | |
| Piplia sistia | Faster Diversit | OFD CAN | | NAD | I NC | Scriedule I | .1 | | | | | | | | |
| Siana sians | Lastern Bluebird | 55B, 54N | NAR | INAR | IN5 | No schedule | N | 00 | | | | | | DO | |
| | American Robin | 55 | | | | | | 00 | 0.0 | PU | PO | PR | 00 | PO | |
| Mimidae | Mockingbirds, Inrasners & Allies | | | | | | | | | | | | | | |
| Dumetella carolinensis | Gray Catbird | S5B, S3N | | | | | 1 | CO | PO | | | PO | | | |
| Toxostoma rufum | Brown Thrasher | S4B | | | | | N | PR | PO | | PO | | | | |
| Sturnidae | Starlings | | | | | | | | | | | | | | |
| Sturnus vulgaris | European Starling | SNA | | | | | | CO | PR | PO | | PR | | | |
| Bombycillidae | Waxwings | | | | | | | | | | | | | | |
| Bombycilla cedrorum | Cedar Waxwing | S5 | | | | | | со | PO | | PO | | PO | PO | |
| Passeridae | Old World Sparrows | | | | | | | | | | | | | | |
| Passer domesticus | House Sparrow | SNA | | | | | | со | PO | | | | | | PO |
| Fringillidae | Finches & Allies | | | | | | | | | | | | | | |
| Haemorhous mexicanus | House Finch | SNA | | | | | | со | | | | | | | |
| Spinus tristis | American Goldfinch | S5 | | | | | | PR | PR | PR | PR | | | PO | |
| Emberizidae | New World Sparrows & Allies | | | | | | | | | | | | | | |
| Melospiza melodia | Song Sparrow | S5 | | | | | | CO | PR | PO | PR | PR | PO | PO | |
| Passerculus sandwichensis | Savannah Sparrow | S5B, S3N | | | | | | CO | | | | | | | |
| Pipilo erythrophthalmus | Eastern Towhee | S4B, S3N | | | | | | PR | | | | | | | |
| Pooecetes gramineus | Vesper Sparrow | S4B | | | | | | PO | | | | | | | |
| Spizella passerina | Chipping Sparrow | S5B, S3N | | | | | | PR | | | | | | | |
| Spizella pusilla | Field Sparrow | S4B, S3N | | | | | | со | | | | | | | |
| Icteridae | Troupials & Allies | | | | | | | | | | | | | | |
| Agelaius phoeniceus | Red-winged Blackbird | S5 | | | | | | со | PR | PO | PR | | PO | PO | PO |
| Dolichonyx oryzivorus | Bobolink | S4B | THR | Т | Т | Schedule 1 | | PR | | | | | | | |
| Icterus galbula | Baltimore Oriole | S4B | | | | | | со | PO | | PO | PO | PO | PO | |
| Molothrus ater | Brown-headed Cowbird | S5 | | | | | | PR | PO | PO | | PO | | | |
| Quiscalus quiscula | Common Grackle | S5 | | | | | | со | со | PO | CO | | CO | | |
| Sturnella magna | Eastern Meadowlark | S4B, S3N | THR | Т | Т | Schedule 1 | | со | | | | | | | |
| Parulidae | Wood Warblers | | | | | | | | | | | | | | |
| Geothlypis philadelphia | Mourning Warbler | S5B | | | | | V | PR | | | | | | | |
| Geothlypis trichas | Common Yellowthroat | S5B, S3N | | | | | | PR | PR | | PR | | | | |
| Seiurus aurocapilla | Ovenbird | S5B | | | | | ~ | PR | | | | | | | |
| Setophaga pensylvanica | Chestnut-sided Warbler | S5B | | | | | 1 | PO | | | | | | | |
| Setophaga petechia | Yellow Warbler | S5B | | | | | | PR | PO | | | PO | | | |
| Setophaga ruticilla | American Redstart | S5B | | | | | 1 | PR | - | | | - | | | |
| Setophaga virens | Black-throated Green Warbler | S5B | | | | | V | PO | | | | | | | |
| Cardinalidae | Cardinals, Grosbeaks & Allies | | | | | | | | | | | | | | |
| Cardinalis cardinalis | Northern Cardinal | S5 | | | | | | со | со | | | | | | со |
| Passerina cvanea | Indigo Bunting | S5B | | | | | | co | | | | | | | |
| Pheucticus Iudovicianus | Rose-breasted Grosbeak | S5B | | | | | | PR | | | | | | | |
| Piranga olivacea | Scarlet Tanager | S5B | | 1 | 1 | 1 | 1 | PR | | | | | | | |
| Total | | | | | | | | 74 | 36 | 14 | 21 | 14 | 9 | 9 | 10 |

*OBBA Atlas Square: 17NJ20

**NHIC Atlas Squares: 17NJ2605, 17NJ2705, 17NJ2604, 17NJ2704

References

Ministry of Natural Resources and Forestry (MNRF). 2021a. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2021-03-18. Available: https://www.ontario.ca/page/get-natural-heritage-information Government of Canada. 2021. Species at Risk Public Registry: Species Search. COSEWIC Last Assessment Date: 2021-05-05. Available: https://species-registry.canada.ca/index-en.html#/species?sortBy=commonNameSort&sortDirection=asc&pageSize=10 Martin, V. 1996. Revisions to Waterloo Region's Significant Species List: Breeding Birds Component. Report to The Regional Municipality of Waterloo, April 16, 1996.

Bird Studies Canada (BSC), Environment Canada's Canadian Wildlife Service, Ontario Nature, Ontario Field Ornithologists and Ontario Ministry of Natural Resources. 2006. Ontario Breeding Bird Atlas Database, 31 January 2008. https://www.birdsontario.org/jsp/datasummaries.jsp Ministry of Natural Resources and Forestry (MNRF). 2021b. Natural Heritage Information Centre (NHIC): Make a Natural Heritage Area Map Application. Published: 2014-07-17. Updated 2021-05-27. Available: https://www.ontario.ca/page/make-natural-heritage-area-map

Appendix IV Herpetofauna Species Reported from the Study Area and Vicinity

Reptile and Amphibian Species Reported from the Study Area - Beckdale Sewer EA (Project #2561)

| | | | | | | SARA | Region of Waterloo | | NRSI |
|---------------------------------------|-------------------------------|------------|------------|------------------------------|------------------------------|------------------------------|-----------------------|------------------------|---------------------------|
| Scientific Name | Common Name | SRANK | SARO | COSEWIC | SARA | Schedule | Status | ORAA* | Observed |
| | | MNRF 2021a | MNRF 2021a | Government of Canada 2021 | Government of Canada 2021 | Government of Canada 2021 | RMOW 1985 | Ontario Nature 2019 | NRSI Results from 2021 |
| Turtles | | | | | | | | | |
| Chelydra serpentina | Snapping Turtle | S4 | SC | SC | SC | Schedule 1 | С | Х | |
| Chrysemys picta marginata | Midland Painted Turtle | S4 | | SC | SC | Schedule 1 | С | Х | |
| Snakes | | | | | | | | | |
| Lampropeltis triangulum | Milksnake | S4 | NAR | SC | SC | Schedule 1 | 1 | Х | |
| Thamnophis sirtalis sirtalis | Eastern Gartersnake | S5 | | | | | С | Х | |
| Salamanders | | | | | | | | | |
| Necturus maculosus | Mudpuppy | S4 | NAR | NAR | NS | No schedule | \checkmark | Х | |
| Notophthalmus viridescens viridescens | Red-spotted Newt | S5 | | | | | \checkmark | Х | |
| Plethodon cinereus | Eastern Red-backed Salamander | S5 | | | | | С | Х | |
| Frogs and Toads | | | | | | | | | |
| Anaxyrus americanus | American Toad | S5 | | | | | С | Х | |
| Hyla versicolor | Gray Treefrog | S5 | | | | | С | Х | |
| Pseudacris crucifer | Spring Peeper | S5 | | | | | С | Х | |
| Lithobates clamitans | Green Frog | S5 | | | | | С | Х | |
| Lithobates palustris | Pickerel Frog | S4 | NAR | NAR | NS | No schedule | \checkmark | Х | |
| Lithobates pipiens | Northern Leopard Frog | S5 | NAR | NAR | NS | No schedule | С | Х | |
| Lithobates sylvaticus | Wood Frog | S5 | | | | | С | х | |
| Total | | | | | | | | 14 | 0 |

*ORAA Atlas Square: 17NJ20

**NHIC Atlas Squares: 17NJ2605, 17NJ2705, 17NJ2604, 17NJ2704

References

Ministry of Natural Resources and Forestry (MNRF). 2021a. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2021-03-18. Available: https://www.ontario.ca/page/get-natural-heritage-information Government of Canada. 2021. Species at Risk Public Registry: Species Search. COSEWIC Last Assessment Date: 2021-05-05. Available: https://species-registry.canada.ca/index-en.html#/species?sortBy=commonNameSort&sortDirection=asc&pageSize=10 Regional Municipality of Waterloo (RMOW). 1985. Appendix 3: Reptiles and Amphibians in Environmentally Sensitive Policy Areas Technical Appendix. Approved by Council: 1986.

Ontario Nature. 2019. Ontario Reptile and Amphibian Atlas Program: Interactive Range Maps. Accessed October 2019.

Ministry of Natural Resources and Forestry (MNRF). 2021b. Natural Heritage Information Centre (NHIC): Make a Natural Heritage Area Map Application. Published: 2014-07-17. Updated 2021-05-27. Available: https://www.ontario.ca/page/make-natural-heritage-area-map

Appendix V Mammal Species Reported from the Study Area and Vicinity

Mammal Species Reported from the Study Area - Beckdale Sewer EA (Project #2561)

| | | | | | | | Region of | Ontario | |
|-------------------------|---------------------------------------|------------|------------|------------------------------|------------------------------|------------------------------|-----------|-------------|-----------|
| | | | | | <u> </u> | SARA | Waterloo | Mammal | NRSI |
| Scientific Name | Common Name | SRANK | SARO | COSEWIC | SARA | Schedule | Status | Atlas | Observed |
| | | MNRF 2021a | MNRF 2021a | Government of Canada 2021 | Government of Canada 2021 | Government of Canada 2021 | RMOW 1985 | Dobbyn 1994 | from 2021 |
| Didelphimorphia | Opossums | | | | | | | | |
| Didelphis virginiana | Virginia Opossum | S4 | | | | | R | Х | |
| Eulipotyphla | Shrews, Moles, Hedgehogs, and Allies | | | | | | | | |
| Blarina brevicauda | Northern Short-tailed Shrew | S5 | | | | | | Х | |
| Condylura cristata | Star-nosed Mole | S5 | | | | | | Х | |
| Sorex cinereus | Masked Shrew | S5 | | | | | G | Х | |
| Sorex fumeus | Smoky Shrew | S5 | | | | | R | Х | |
| Chiroptera | Bats | | | | | | | | |
| Eptesicus fuscus | Big Brown Bat | S4 | | | | | | Х | |
| Lasiurus borealis | Eastern Red Bat | S4 | | | | | | Х | |
| Lasiurus cinereus | Hoary Bat | S4 | | | | | | Х | |
| Myotis lucifugus | Little Brown Myotis | S3 | END | E | E | Schedule 1 | | Х | |
| Myotis septentrionalis | Northern Myotis | S3 | END | E | E | Schedule 1 | | Х | |
| Lagomorpha | Rabbits and Hares | | | | | | | | |
| Lepus europaeus | European Hare | SNA | | | | | | Х | |
| Sylvilagus floridanus | Eastern Cottontail | S5 | | | | | | Х | Х |
| Rodentia | Rodents | | | | | | | | |
| Castor canadensis | Beaver | S5 | | | | | S | Х | |
| Erethizon dorsatum | Porcupine | S5 | | | | | S | Х | |
| Glaucomys sabrinus | Northern Flying Squirrel | S5 | | | | | R | Х | |
| Marmota monax | Woodchuck | S5 | | | | | | Х | Х |
| Microtus pennsylvanicus | Meadow Vole | S5 | | | | | | Х | |
| Mus musculus | House Mouse | SNA | | | | | | Х | |
| Ondatra zibethicus | Muskrat | S5 | | | | | | Х | |
| Peromyscus leucopus | White-footed Mouse | S5 | | | | | | Х | |
| Peromyscus maniculatus | Deer Mouse | S5 | | | | | | Х | |
| Rattus norvegicus | Norway Rat | SNA | | | | | | Х | |
| Sciurus carolinensis | Eastern Gray Squirrel | S5 | | | | | | Х | Х |
| Tamias striatus | Eastern Chipmunk | S5 | | | | | | Х | |
| Tamiasciurus hudsonicus | Red Squirrel | S5 | | | | | | Х | |
| Zapus hudsonius | Meadow Jumping Mouse | S5 | | | | | | Х | |
| Canidae | Canines | | | | | | | | |
| Canis latrans | Coyote | S5 | | | | | S | Х | |
| Vulpes vulpes | Red Fox | S5 | | | | | | Х | |
| Mephitidae | Skunks and Stink Badgers | | | | | | | | |
| Mephitis mephitis | Striped Skunk | S5 | | | | | | Х | |
| Mustelidae | Weasels and Allies | | | | | | | | |
| Mustela erminea | Ermine | S5 | | | | | | Х | |
| Mustela frenata | Long-tailed Weasel | S4 | | | | | S | Х | |
| Neovison vison | American Mink | S4 | | | | | S | Х | |
| Taxidea taxus jacksoni | American Badger (Southwestern Ontario | S1 | END | E | E | Schedule 1 | | Х | |
| Procyonidae | Raccoons and Allies | | | | | | | | |
| Procyon lotor | Northern Raccoon | S5 | | | | | | X | |
| Total | | | | | | | | 34 | 3 |

*Mammal Atlas Square Numbers: 17NJ20 **NHIC Atlas Squares: Square #

References

Ministry of Natural Resources and Forestry (MNRF). 2021a. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2021-03-18. Available: https://www.ontario.ca/page/get-natural-heritage-information Government of Canada. 2021. Species at Risk Public Registry: Species Search. COSEWIC Last Assessment Date: 2021-05-05. Available: https://species-registry.canada.ca/index-en.html#/species?sortBy=commonNameSort&sortDirection=asc&pageSize=10 Regional Municipality of Waterloo (RMOW). 1985. Appendix 4: Mammals in Environmentally Sensitive Policy Areas Technical Appendix. Approved by Council: 1986. Dobbyn, J.S. 1994. Atlas of the Mammals of Ontario. Don Mills, Federation of Ontario Naturalists. 120p.

Appendix VI Butterfly Species Reported from the Study Area and Vicinity

Butterfly Species Reported from the Study Area - Beckdale Sewer EA (Project #2561)

| | | | | | | SARA | Region of | Ontario | NPSI |
|-----------------------------|---------------------------------------|------------|------------|------------|------------|------------|-------------|-------------|--------------|
| Scientific Name | Common Name | SRANK | SARO | COSEWIC | SARA | Schedule | Status | Δtlas* | Observed |
| | | Olount | - CARO | Government | Government | Government | Olulus | | Obscived |
| | | MNRF 2021a | MNRF 2021a | of Canada | of Canada | of Canada | Linton 2012 | Macnaughton | NRSI Results |
| | | | | 2021 | 2021 | 2021 | | et al. 2020 | from 2021 |
| Hesperiidae | Skippers | | | | | | | | |
| Carterocephalus palaemon | Arctic Skipper | S5 | | | | | R | Х | |
| Epargyreus clarus | Silver-spotted Skipper | S4 | | | | | UC | X | |
| Erynnis baptisiae | Wild Indigo Duskywing | S4 | | | | | UK | X | |
| Euphyes conspicua | Black Dash | S3 | | | | | UC | X | |
| Poanes hobomok | Hobomok Skipper | S5 | | | | | С | Х | |
| Thymelicus lineola | European Skipper | SNA | | | | | VC | X | |
| Papilionidae | Swallowtails | | | | | | | | |
| Papilio cresphontes | Giant Swallowtail | S4 | | | | | UC | X | |
| Papilio glaucus | Eastern Tiger Swallowtail | S5 | | | | | VC | X | |
| Papilio polyxenes | Black Swallowtail | S5 | | | | | VC | X | |
| Pieridae | Whites and Sulphurs | | | | | | | | |
| Colias eurytheme | Orange Sulphur | S5 | | | | | VC | X | |
| Colias philodice | Clouded Sulphur | S5 | | | | | | X | |
| Pieris rapae | Cabbage White | SNA | | | | | VC | X | |
| Lycaenidae | Harvesters, Coppers, Hairstreaks, Blu | es | | | | | | | |
| Celastrina lucia | Northern Spring Azure | S5 | | | | | | X | |
| Celastrina sp. | Azure species | SNA | | | | | | X | |
| Cupido comyntas | Eastern Tailed Blue | S5 | | | | | UC | X | |
| Lycaena hyllus | Bronze Copper | S5 | | | | | VC | X | |
| Satyrium calanus | Banded Hairstreak | S4 | | | | | UC | X | |
| Nymphalidae | Brush-footed Butterflies | | | | | | | | |
| Aglais milberti | Milbert's Tortoiseshell | S5 | | | | | UC | X | |
| Asterocampa celtis | Hackberry Emperor | S3 | | | | | | Х | |
| Asterocampa clyton | Tawny Emperor | S3 | | | | | UC | X | |
| Cercyonis pegala | Common Wood-Nymph | S5 | | | | | VC | X | |
| Coenonympha tullia | Common Ringlet | S5 | | | | | С | X | |
| Danaus plexippus | Monarch | S2N,S4B | SC | E | SC | Schedule 1 | VC | X | |
| Lethe anthedon | Northern Pearly-Eye | S5 | | | | | С | X | |
| Libytheana carinenta | American Snout | SNA | | | | | R | X | |
| Limenitis archippus | Viceroy | S5 | | | | | VC | Х | |
| Limenitis arthemis astyanax | Red-spotted Purple | S5 | | | | | С | Х | |
| Megisto cymela | Little Wood-Satyr | S5 | | | | | VC | X | |
| Nymphalis antiopa | Mourning Cloak | S5 | | | | | VC | X | |
| Nymphalis I-album | Compton Tortoiseshell | S5 | | | | | UC | X | |
| Phyciodes cocyta | Northern Crescent | S5 | | | | | UC | Х | |
| Polygonia comma | Eastern Comma | S5 | | | | | VC | Х | |
| Polygonia interrogationis | Question Mark | S5 | | | | | VC | X | |
| Polygonia progne | Gray Comma | S5 | | | | | UC | X | |
| Speyeria cybele | Great Spangled Fritillary | S5 | | | | | VC | Х | |
| Vanessa atalanta | Red Admiral | S5B | | | | | VC | Х | |
| Vanessa cardui | Painted Lady | S5B | | | | | С | Х | |
| Vanessa virginiensis | American Lady | S5 | | | | | C | X | |
| Total | | | | | | | | 38 | 0 |

*TEA Atlas Square: 17NJ20

**NHIC Atlas Square: Square #

References

Ministry of Natural Resources and Forestry (MNRF). 2021a. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2021-03-18. Available: https://www.ontario.ca/page/get-natural-heritage-information Government of Canada. 2021. Species at Risk Public Registry: Species Search. COSEWIC Last Assessment Date: 2021-05-05. Available: https://species-registry.canada.ca/index-en.html#/species?sortBy=commonNameSort&sortDirection=asc&pageSize=10 Linton, J.E. 2012. The Butterflies of Waterloo Region: An Annotated Checklist, a Species Scarcity Ranking, and a Transect-Based Analysis of Urban vs. Rural Populations. Toronto Entomologists Association. 42p. Macnaughton A., Layberry R., Cavasin R., Edwards B., and C., Jones. Ontario Butterfly Atlas. Updated February 2020. Available: https://www.ontarionisco.org/atlas/index.html

Ministry of Natural Resources and Forestry (MNRF). 2021b. Natural Heritage Information Centre (NHIC): Make a Natural Heritage Area Map Application. Published: 2014-07-17. Updated 2021-05-27. Available: https://www.ontario.ca/page/make-natural-heritage-area-map

Appendix VII Odonata Species Reported from the Study Area and Vicinity

Odonate Species Reported from the Study Area - Beckdale Sewer EA (Project #2561)

| | | | | | | | Region of | | |
|----------------------------|---------------------------|------------|------------|------------|------------|------------|-----------|-----------|----------|
| | | | | | | SARA | Waterloo | Odonate | NRSI |
| Scientific Name | Common Name | SRANK | SARO | COSEWIC | SARA | Schedule | Status | Atlas* | Observed |
| | | | | Government | Government | Government | Region of | | |
| | | MNRF 2021a | MNRF 2021a | of Canada | of Canada | of Canada | Waterloo | OOAD 2021 | |
| | | | | 2021 | 2021 | 2021 | 1985 | | |
| Calopterygidae | Broadwinged Damselflies | | | | | | | | |
| Calopteryx aequabilis | River Jewelwing | S5 | | | | | Х | Х | |
| Hetaerina americana | American Rubyspot | S4 | | | | | Х | Х | |
| Lestidae | Spreadwings | | | | | | | | |
| Lestes dryas | Emerald Spreadwing | S5 | | | | | Х | Х | |
| Lestes eurinus | Amber-winged Spreadwing | S4 | | | | | Expected | X | |
| Lestes forcipatus | Sweetflag Spreadwing | S4 | | | | | Х | X | |
| Lestes unguiculatus | Lyre-tipped Spreadwing | S5 | | | | | Х | X | |
| Coenagrionidae | Narrow-winged Damselflies | | | | | | | | |
| Argia apicalis | Blue-fronted Dancer | S4 | | | | | Х | Х | |
| Argia moesta | Powdered Dancer | S5 | | | | | Х | X | |
| Coenagrion resolutum | Taiga Bluet | S5 | | | | | Х | X | |
| Enallagma annexum | Northern Bluet | S4 | | | | | Expected | Х | |
| Enallagma exsulans | Stream Bluet | S5 | | | | | Х | Х | |
| Ischnura verticalis | Eastern Forktail | S5 | | | | | Х | | |
| Nehalennia irene | Sedge Sprite | S5 | | | | | Х | | |
| Aeshnidae | Darners | | | | | | | | |
| Anax junius | Common Green Darner | S5 | | | | | Х | Х | |
| Gomphidae | Clubtails | | | | | | | | |
| Ophiogomphus rupinsulensis | Rusty Snaketail | S4 | | | | | Х | X | |
| Libellulidae | Skimmers | | | | | | | | |
| Ladona julia | Chalk-fronted Corporal | S5 | | | | | Х | Х | |
| Leucorrhinia intacta | Dot-tailed Whiteface | S5 | | | | | Х | Х | |
| Libellula pulchella | Twelve-spotted Skimmer | S5 | | | | | Х | Х | |
| Libellula quadrimaculata | Four-spotted Skimmer | S5 | | | | | Х | X | |
| Plathemis lydia | Common Whitetail | S5 | | | | | Х | Х | |
| Total | | | | | | | | 18 | 0 |

*Odonate Atlas Square Numbers: 17NJ20

**NHIC Atlas Squares: Square(s)

References

Ministry of Natural Resources and Forestry (MNRF). 2021a. Natural Heritage Information Centre (NHIC): Species List for Ontario. Published: 2014-07-17. All Species List Updated: 2021-03-18. Available: https://www.ontario.ca/page/get-natural-heritage-information Government of Canada. 2021. Species at Risk Public Registry: Species Search. COSEWIC Last Assessment Date: 2021-05-05. Available: https://species-registry.canada.ca/index-en.html#/species?sortBy=commonNameSort&sortDirection=asc&pageSize=10 Regional Municipality of Waterloo. 1985. Environmentally Sensitive Policy Areas. Approved by Council: 1986.

Ministry of Natural Resources and Forestry (MNRF). 2021b. Natural Heritage Information Centre (NHIC): Make a Natural Heritage Area Map Application. Published: 2014-07-17. Updated 2021-05-27. Available: https://www.ontario.ca/page/make-natural-heritage-area-map

Appendix VIII Species at Risk/Species of Conservation Concern Habitat Assessment

| Scientific Name | Common Name | S-RANK ¹ | SARO | COSEWIC ² | SARA Schedule | Background Source | Habitat Preference ^{4,5} | Suitable Habitats within Study Area | Carried Forward to EA? | Rationale | Observed within Study Area (NRSI 2018) |
|-----------------------------------|---------------------------|---------------------|------|----------------------|---------------|-----------------------------------|--|---|---------------------------|--|---|
| Birds | | - | | - | - | | | | - | | |
| Riparia riparia | Bank Swallow | S4B | THR | т | Schedule 1 | MNRF 2021, BSC et al. 2006 | Sand, clay or gravel river banks or steep riverbank cliffs; lakeshore bluffs of easily crumbled sand or gravel; gravel pits, road-cuts, grassland or cultivated fields that are close to water; nesting sites are limiting factor for species presence. | Yes (foraging habitat only) | Yes | Cultivated fields present offering forage, though limited by lack of nesting habitat. | Yes - Possible Breeding Evidence |
| Hirundo rustica | Barn Swallow | S4B | THR | т | Schedule 1 | BSC et al. 2006 | Farmlands or rural areas; cliffs, caves, rock niches; buildings or other man-made structures for nesting; open country near body of water. | Yes | Yes | Agricultural farmlands and cultural meadow provide suitable foraging habitat. Agricultural outbuildings may provide opportunities for nesting in proximity to foraging habitat. | Yes - Possible Breeding Evidence |
| Chlidonias niger | Black Tern | S3B | SC | NAR | No Schedule | MNRF 2021, BSC et al. 2006 | Breed mainly imarshes along the edges of the Great Lakes. Builds floating nests in loose colonies in shallow marshes, especially in cattails. | No | No | Suitable habitat not present in Subject Property. No shallow marshes exist within Subject Property. | No |
| Dolichonyx oryzivorus | Bobolink | S4B | THR | т | Schedule 1 | MNRF 2021, BSC et al. 2006 | Large, open expansive grasslands with dense ground cover; hayfields, meadows or fallow fields; marshes; requires tracts of grassland >50 ha. | Yes | Yes | Suitable habitat may be present depending on presence of suitable hayfield crop cover in agricultural fields, or if fields are fallow. | Νο |
| Sturnella magna | Eastern Meadowlark | S4B | THR | т | Schedule 1 | MNRF 2021, BSC et al. 2006 | Open, grassy meadows, farmland, pastures, hayfields or grasslands with elevated singing perches; cultivated land and weedy areas with trees; old orchards with adjacent, open grassy areas >10 ha in size. | Yes | Yes | Land use a mixture of agricultural fields, andcultural meadow with likelihood of offering singing perches Larger than 10 ha in size extending beyond Study Area boundaries. | Νο |
| Contopus virens | Eastern Wood-pewee | S4B | SC | SC | Schedule 1 | BSC et al. 2006 | Mid-canopy layer of forest clearings and edges of deciduous and mixed forests. Most abundant in forest stands of intermediate age and in mature stands with little understory vegetation. | No | No | Forest communities do not provide adequate habitat within or adjacent to Study Area. | No |
| Melanerpes erythrocephalus | Red-headed Woodpecker | S4B | SC | E | Schedule 1 | BSC et al. 2006 | Open, deciduous forest with little understory; fields or pasture lands with scattered large trees; wooded swamps; orchards, small woodlots or forest edges; groves of dead or dying trees; feeds on insects and stores nuts or acorns for winter; loss of habitat is limiting factor; requires cavity trees with at least 40 cm dbh; require about 4 ha for a territory. | No | No | Forest communities do not provide adequate habitat within or adjacent to Study Area. | Νο |
| Hylocichla mustelina | Wood Thrush | S4B | SC | т | Schedule 1 | BSC et al. 2006 | Carolinian and Great Lakes-St. Lawrence forest zones; undisturbed moist mature deciduous or mixed forest with deciduous sapling growth; near pond or swamp; hardwood forest edges; must have some trees higher than 12 m. | No | No | Required habitat not present within or adjacent to Study Area. | Νο |
| Herpetofauna | | | - | | | - | | - | - | | |
| Chelydra serpentina serpentina | Common Snapping Turtle | S4 | SC | SC | Schedule 1 | MNRF 2021, Ontario Nature 2019 | Permanent or semi-permanent fresh water; marshes, swamps or bogs; rivers and streams with soft muddybanks or bottoms. The species often uses soft soil or clean dry sand on south-facing slopes for nest sites and may nest at some distance from water. | Yes | Yes | Suitable travel corridor habitat may be provided by Baden Creek | No |
| Mammals | | 1 | T | T | 1 | T | | - | | | |
| Myotis lucifungus | Little Brown Myotis | S3 | END | E | Schedule 1 | Dobbyn 1994 | Uses caves, quarries, tunnels, hollow trees or buildings for roosting; winters in humid caves; maternity sites in dark warm areas such as attics and barns; feeds primarily in wetlands, forest edges | Yes | Yes | Presence of roosting sites potentially present within barns and other structures as well as trees. Foraging offered over wetlands and along forest edges. | No - Targeted surveys not conducted |

| Scientific Name | Common Name | S-RANK ¹ | SARO | COSEWIC ² | SARA Schedule | Background Source | d Habitat Preference ^{4,5} Ha | | Carried Forward to EA? | Rationale | Observed within Study Area (NRSI 2018) |
|---------------------------|----------------------------|---------------------|------|----------------------|---------------|-----------------------------|--|-----|---------------------------|--|---|
| Myotis septentrionalis | Northern Myotis | S3 | END | E | Schedule 1 | Dobbyn 1994 | Hibernates during winter in mines or caves; during summer males roost alone and females form maternity colonies of up to 60 adults; roosts in houses, man-made structures but prefers hollow trees or under loose bark; hunts within forest, below canopy | Yes | Yes | Man-made structures within Subject Property may offer suitable roosts. Potential for tree cavities and foraging habitat present within wooded areas of nearby forest communities. | No - Targeted surveys not conducted |
| Taxidea taxus jacksoni | American Badger | S2 | END | E | Schedule 1 | SAR Ontario | In Ontario, badgers are found in a variety of habitats, such as tall grass prairie, sand barrens and farmland. These habitats provide badgers with small prey, including groundhogs, rabbits and small rodents. Since badgers are primarily nocturnal and quite wary of people, not many people are fortunate enough to spot one in the wild. | Yes | Yes | Farmlands provide potential habitat , presence of small mammals provides forage. | No - Dens not observed |
| | - | | - | | - | | | - | - | | |
| Danaus plexippus | Monarch | S2N, S4B | SC | E | Schedule 1 | Macnaughton et al. 2020 | Open areas with milkweed species (<i>Asclepias spp.</i>). | Yes | Yes | Milkweed observed within Cultural Meadow in 2018. Offers suitable habitat for Monarch | No - Larval food plant (Milkweed) observed |
| Odonates (Dragon/Damsel F | lies) | | - | | - | | | - | - | | |
| Lestes eurinus | Amber-winged Spreadwing | S3 | N/A | N/A | N/A | Blust and Pfeiffer, 2015 | Associated with, but not exclusive to, bogs and other acidic wetlands and ponds with boggy shoreline. Adapted to fishless, permanent water habitats. | No | No | Suitable habitat is not present within the Study Area | No |
| Fish | | | | | | | | | | | |
| Freshwater Mussels | | | | | | | | | | | |
| Plants | | | | | | | | | | | |

Appendix IX Significant Wildlife Habitat Assessment

Significant Wildlife Habitat Assessment Tables

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|---|--|--|--|--|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Waterfowl Sto | pover and Staging Areas (Terres | trial) | | | |
| Rationale: Habitat important to migrating waterfowl. | American Black Duck Wood Duck Green-winged Teal Blue-winged Teal Mallard Northern Pintail Northern Shoveler American Wigeon Gadwall | CUM1 CUT1 - Plus evidence of annual spring flooding from melt water or run-off within these Ecosites. | Fields with sheet water during Spring (mid March to May). Fields flooding during spring melt and run-off provide important invertebrate foraging habitat for migrating waterfowl. Agricultural fields with waste grains are commonly used by waterfowl, these are not considered SWH unless they have spring sheet water available^{extviii.} Information Sources Anecdotal information from the landowner, adjacent landowners or local naturalist clubs may be good information in determining occurrence. Reports and other information available from Conservation Authorities (CAs) Sites documented through waterfowl planning processes (eg. EHJV implementation plan) Field Naturalist Clubs Ducks Unlimited Canada Natural Heritage Information Centre (NHIC) Waterfowl Concentration Area | Studies carried out and verified presence of an annual concentration of any listed species, evaluation methods to follow "Bird and Bird Habitats: Guidelines for Wind Power Projects" ^{codi} • Any mixed species aggregations of 100 or more individuals required. • The area of the flooded field ecosite habitat plus a 100-300m radius buffer dependent on local site conditions and adjacent land use is the significant wildlife habitat ^{colvii} . • Annual use of habitat is documented from information sources or field studies (annual use can be based on studies or determined by past surveys with species numbers and dates). • SWHMIST ^{codix} Index #7 provides development effects and mitigation measures. | Presence of flooding in fields inadequate to support provincially significant congregations of waterfowl. Not SWH |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|---|--|---|---|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Waterfowl Sto | pover and Staging Areas (Aqua | ic) | • | • | |
| Rationale: Important for local and migrant waterfowl populations during the spring or fall migration or both periods combined. Sites identified are usually only one of a few in the eco-district. | Canada Goose Cackling Goose Snow Goose American Black Duck Northern Pintail Northern Shoveler American Wigeon Gadwall Green-winged Teal Blue-winged Teal Hooded Merganser Common Merganser Lesser Scaup Common Merganser Lesser Scaup Long-tailed Duck Surf Scoter White-winged Scoter Black Scoter Ring-necked Duck Common Goldeneye Bufflehead Redhead Ruddy Duck Red-breasted Merganser Brant Canvasback | MAS1 MAS2 MAS3 SAS1 SAM1 SAF1 SWD1 SWD2 SWD3 SWD4 SWD5 SWD6 SWD7 | Ponds, marshes, lakes, bays, coastal inlets, and watercourses used during migration. Sewage treatment ponds and storm water ponds do not qualify as a SWH, however a reservoir managed as a large wetland or pond/lake does qualify. These habitats have an abundant food supply (mostly aquatic invertebrates and vegetation in shallow water). Information Sources Environment Canada Naturalist clubs often are aware of staging/stopover areas. OMINRF Wetland Evaluations indicate presence of locally and regionally significant waterfowl staging. Sites documented through waterfowl planning processes (eg. EHJV implementation plan) Ducks Unlimited projects Element occurrence specification by Nature Serve: http://www.natureserve.org Natural Heritage Information Centre (NHIC) Waterfowl Concentration Area | Studies carried out and verified presence of: • Aggregations of 100 ^I or more of listed species for 7 days ¹ , results in > 700 waterfow use days. • Areas with annual staging of ruddy ducks, canvasbacks, and redheads are SWH ^{cdix} • The combined area of the ELC ecosites and a 100m radius area is the SWH ^{cdix} • Wetland area and shorelines associated with sites identified within the SWHT ^{Cdivii} Appendix K ^{cdix} are significant wildlife habitat. • Evaluation methods to follow "Bird and Bird Habitats: Guidelines for Wind Power Projects ^{wcodi} • Annual Use of Habitat is Documented from Information Sources or Field Studies (Annual can be based on completed studies or determined from past surveys with species numbers and dates recorded). • SWHMiST ^{cdix} Index #7 provides development effects and mitigation measures. | The on-site watercourse is relatively small and other suitable forms of habitat are absent from the Study Area. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--------------------------------------|-------------------------------|--------------------------------|--|--|---------------------------------|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Shorebird Mig | ratory Stopover Area | | | | |
| Rationale: | Greater Yellowlegs | BBO1 | Shorelines of lakes, rivers and wetlands, including beach | Studies confirming: | Suitable habitat is not present |
| High quality shorebird stopover | Lesser Yellowlegs | BBO2 | areas, bars and seasonally flooded, muddy and un- | Presence of 3 or more of listed species and > | within or adjacent to the |
| habitat is extremely rare and | Marbled Godwit | BBS1 | vegetated shoreline habitats. Great Lakes coastal | 1000 shorebird use days during spring or fall | subject lands. |
| typically has a long history of use. | Hudsonian Godwit | BBS2 | shorelines, including groynes and other forms of armour | migration period. (shorebird use days are the | |
| | Black-bellied Plover | BBT1 | rock lakeshores, are extremely important for migratory | accumulated number of shorebirds counted per | Not SWH. |
| | American Golden-Plover | BBT2 | shorebirds in May to mid-June and early July to October. | day over the course of the fall or spring | |
| | Semipalmated Plover | SDO1 | Sewage treatment ponds and storm water ponds do not | migration period) | |
| | Solitary Sandpiper | SDS2 | qualify as a SWH. | Whimbrel stop briefly (<24hrs) during spring | |
| | Spotted Sandpiper | SDT1 | | migration, any site with >100 Whimbrel used for | |
| | Semipalmated Sandpiper | MAM1 | Information Sources | 3 years or more is significant. | |
| | Pectoral Sandpiper | MAM2 | Western hemisphere shorebird reserve network. | The area of significant shorebird habitat | |
| | White-rumped Sandpiper | MAM3 | Canadian Wildlife Service (CWS) Ontario Shorebird | includes the mapped ELC shoreline ecosites | |
| | Baird's Sandpiper | MAM4 | Survey. | plus a 100m radius area ^{cxtviii} | |
| | Least Sandpiper | MAM5 | Bird Studies Canada | Evaluation methods to follow "Bird and Bird | |
| | Purple Sandpiper | | Ontario Nature | Habitats: Guidelines for Wind Power Projects"ccxi | |
| | Stilt Sandpiper | | Local birders and naturalist clubs | • SW/HMiST ^{cxlix} Index #9 provides development | |
| | Short-billed Dowitcher | | Natural Heritage Information Center (NHIC) Shorebird | effects and mitigation massures | |
| | Red-necked Phalarope Whimbrel | | Migratory Concentration Area | enects and mitigation measures. | |
| | Ruddy Turnstone | | | | |
| | Sanderling | | | | |
| | Dunlin | | | | |
| | Whimbrel | | | | |
| | | | | | |
| | | | | | |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|---|--|---|---|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Raptor Winter | ing Area | | | | |
| Rational: Sites used by multiple species, a high number of individuals and used annually are most significant | Rough-legged Hawk Red-tailed Hawk Northern Harrier American Kestrel Snowy Owl <u>Special Concern:</u> Short-eared Owl Bald Eagle | Hawks/Owls: Combination of ELC Community Series; need to have present one Community Series from each land class: Forest: FOD, FOM, FOC Upland: CUM, CUT, CUS, CUW | The habitat provides a combination of fields and woodlands that provide roosting, foraging and resting habitats for wintering raptors. Raptor wintering sites need to be > 20 ha ^{culvili, cxlix} with a combination of forest and upland. ^{XVI, XVII, XVII, XVII, XXI, XXI, XXI, XX} | Studies confirm the use of these habitats by: • One or more Short-eared Owls or; One or more Bald Eagles or; At least 10 individuals and two listed hawk/owl species • To be significant a site must be used regularly (3 in 5 years) ^{colix} for a minimum of 20 days by the above number of birds • The habitat area for an Eagle winter site is the shoreline forest ecosites directly adjacent to the prime hunting area • Evaluation methods to follow "Bird and Bird Habitats: Guidelines for Wind Power Projects" ^{codi} • SWHMIST ^{Codix} Index #10 and #11 provides development effects and mitigation measures. | Woodland habitat is too small and limiting on the subject lands and surrounding landscape to provide significant winter raptor habitat. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|---|-----------------------------------|---|--|--|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Bat Hibernacu | ıla | | | | |
| <u>Rationale</u> Bat hibernacula are rare habitats in Ontario landscapes. | Big Brown Bat Tri-coloured Bat | Bat Hibernacula may be found in these ecosites: CCR1 CCR2 CCA1 CCA2 (Note: buildings are not considered to be SWH) | Hibernacula may be found in caves, mine shafts, underground foundations and Karsts. Active mine sites should not be considered as SWH The locations of bat hibernacula are relatively poorly known. Information Sources OMNRF for possible locations and contact for local experts Natural Heritage Information Center (NHIC) Bat Hibernaculum Ministry of Northern Development and Mines for location of mine shafts. Clubs that explore caves (eg. Sierra Club) University Biology Departments with bat experts. | All sites with confirmed hibernating bats are SWH. The habitat area includes a 200m radius around the entrance of the hibernaculum^{cxlviii}, ccvii for most. Studies are to be conducted during the peak swarming period (Aug. – Sept.). Surveys should be conducted following methods outlined in the "Bats and Bat Habitats: Guidelines for Wind Power Projects^{«CCV} SWHMiST^{cdix} Index #1 provides development effects and mitigation measures. | Big Brown Bats have been documented in the vicinity of the study area. However, No known hibernacula are present within 200 m of the subject property. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|------------------------------------|---|---|---|--|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Bat Maternity | Colonies | | | | |
| <u>Rationale:</u> Known locations of forested bat maternity colonies is extremely rare in all Ontario landscapes. | Big Brown Bat Silver-haired Bat | Maternity colonies considered SWH are found in forested Ecosites. All ELC Ecosites in ELC Community Series: FOD FOM SWD SWM | Maternity colonies can be found in tree cavities, vegetation and often in buildings ^{xxii, xxv, xxvi, xxvi, xxvii} (buildings are not considered to be SWH). • Maternity roosts are not found in caves and mines in Ontario ^{xii} • Maternity colonies located in Mature deciduous or mixed forest stands ^{ccx, cxx} with >10/ha large diameter (>25cm dbh) wildlife trees ^{crvii} • Female Bats prefer wildlife tree (snags) in early stages of decay, class 1-3 ^{ccxvi} or class 1 or 2 ^{ccxii} • Silver-haired Bats prefer older mixed or deciduous forest and form maternity colonies in tree cavities and small hollows. Older forest areas with at least 21 snags/ha are preferred ^{ccx} • OMNRF for possible locations and contact for local experts • University Biology Departments with bat experts. | Maternity Colonies with confirmed use by: >10 Big Brown Bats >5 Adult Female Silver-haired Bats The area of the habitat includes the entire woodland or a forest stand ELC Ecosite or an Ecoelement containing the maternity colonies. Evaluation methods for maternity colonies should be conducted following methods outlined in the "Bats and Bat Habitats: Guidelines for wind Power Projects^{ccv} SWHMIS T^{colix} Index #12 provides development effects and mitigation measures. | Woodland features are limited within the study area, but may contain suitable concentrations of standing dead or declining trees that provide suitable bat habitat. Candidate SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|--|--|---|--|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Turtle Winteri | ng Area | | | | |
| Rationale: Generally sites are the only known sites in the area. Sites with the highest number of individuals are most significant | Midland Painted Turtle <u>Special Concern</u> : Northern Map Turtle Snapping Turtle | Snapping and Midland Painted Turtles - ELC Community Classes: SW, MA, OA and SA; ELC Community Series: FEO and BOO Northern Map Turtle - Open Water areas such as deeper rivers or streams and lakes with current can also be used as over-wintering habitat. | For most turtles, wintering areas are in the same general area as their core habitat. Water has to be deep enough not to freeze and have soft mud substrates. • Over-wintering sites are permanent water bodies, large wetlands, and bogs or fens with adequate Dissolved Oxygen ^{CK, CX, CX, CX, CX, CX, CX, CX, CX, CX, CX} | Presence of 5 over-wintering Midland Painted Turtles is significant. One or more Northern Map Turtle or Snapping Turtle over-wintering within a wetland is significant. The mapped ELC ecosite area with the over wintering turtles is the SWH. If the hibernation site is within a stream or river, the deep-water pool where the turtles are over wintering is the SWH. Over wintering areas may be identified by searching for congregations (Basking Areas) of turtles on warm, sunny days during the fall (Sept. – Oct.) or spring (Mar. – May)^{cvii} Congregation of turtles is more common where wintering areas are limited and therefore significant^{cit, cx, cxi, cxii}. SWHMIST^{codit} Index #28 provides development effects and mitigation measures for turtle wintering habitat. | Turtle overwintering habitat may be provided by the pond located beyond limits of Study Area. However, no suitable overwintering habitat occurs on the subject lands. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|------------------------------------|----------------------------------|--------------------------------|--|---|----------------------------------|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Snake Hibern | aculum | | | | |
| Rationale: | Snakes: | For all snakes, habitat may | For snakes, hibernation takes place in sites located | Studies confirming: | Potential hibernaculum |
| Generally sites are the only known | Eastern Gartersnake | be found in any ecosite | below frost lines in burrows, rock crevices and other | Presence of snake hibernacula used by a | features, such as south-facing |
| sites in the area. Sites with the | Northern Watersnake | other than very wet ones. | natural locations. The existence of features that go | minimum of five individuals of a snake sp. or; | slopes, rock piles, and fissured |
| highest number of individuals are | Northern Red-bellied Snake | Talus, Rock Barren, Crevice | below the frost line; such as rock piles or slopes, old | individuals of two or more snake spp. | bedrock are absent on the |
| most significant | Northern Brownsnake | and Cave, and Alvar sites | stone fences, and abandoned crumbling foundations | Congregations of a minimum of five individuals | subject lands. However, |
| | Smooth Green Snake | may be directly related to | assist in identifying candidate SWH. | of a snake sp. or; individuals of two or more | remnants of the concrete slab |
| | Northern Ring-necked Snake | these habitats. | Areas of broken and fissured rock are particularly | snake spp. near potential hibernacula (eg. | footprint of the former barn |
| | | | valuable since they provide access to subterranean sites | foundation or rocky slope) on sunny warm days | may provide hibernaculum |
| | Special Concern: | Observations of | below the frost line ^{xliv, I, Ii, Iii, cxii.} | in Spring (Apr/May) and Fall (Sept/Oct). | access. |
| | Milksnake | congregations of snakes on | Wetlands can also be important over-wintering habitat | <u>Note</u>: If there are Special Concern Species | |
| | Eastern Ribbonsnake | sunny warm days in the | in conifer or shrub swamps and swales, poor fens, or | present, then site is SWH | Candidate SWH. |
| | | spring or fall is a good | depressions in bedrock terrain with sparse trees or | <u>Note</u>: Sites for hibernation possess specific | |
| | Lizard: | indicator. | shrubs with sphagnum moss or sedge hummock ground | habitat parameters (e.g. temperature, humidity, | |
| | Special Concern (Southern Shield | | cover. | etc.) and consequently are used annually, often | |
| | population): | For Five-lined Skink, ELC | Five-lined skink prefer mixed forests with rock outcrop | by many of the same individuals of a local | |
| | Five-lined Skink | Community Series of FOD | openings providing cover rock overlaying granite bedrock | population [i.e. strong hibernation site fidelity]. | |
| | | and FOM and Ecosites: | with fissures cciii. | Other critical life processes (e.g. mating) often | |
| | | FOC1 | | take place in close proximity to hibernacula. The | |
| | | FOC3 | Information Sources | feature in which the hibernacula is located plus a | |
| | | | In spring, local residents or landowners may have | 30m buffer is the SWH ⁱ | |
| | | | observed the emergence of snakes on their property | SWHMiST ^{cxiix} Index #13 provides development | |
| | | | (e.g. old dug wells). | effects and mitigation measures for snake | |
| | | | Reports and other information from CAs. | hibernacula | |
| | | | Local Field naturalists and experts, as well as university | Presence of any active hibernaculum for skink | |
| | | | herpetologists may also know where to find some of | is significant | |
| | | | these sites. clubs | • SW/HMiST ^{cxlix} Index #27 provides development | |
| | | | Natural Heritage Information Center (NHIC) | effects and mitigation measures for five lined | |
| | | | OMNRF ecologist or biologist may be aware of | skink wintering babitat | |
| | | | locations of wintering skinks | Stant wintering habitat. | |
| | | | , , , , , , , , , , , , , , , , , , , | | |
| | | | | | |
| | | | | | |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|---|--|--|--|--|--|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Colonially - Ne | esting Bird Breeding Habitat (Ba | nk and Cliff) | | | |
| Rationale: Historical use and number of nests in a colony make this habitat significant. An identified colony can be very important to local populations. All swallow populations are declining in Ontario. | Cliff Swallow Northern Rough-winged Swallow (this species is not colonial but can be found in Cliff Swallow colonies) | Eroding banks, sandy hills, borrow pits, steep slopes, and sand piles Cliff faces, bridge abutments, silos, barns Habitat found in the following ecosites: CUM1 CUT1 CUS1 BLO1 BLS1 BLT1 CLO1 CLS1 CLT1 | Any site or areas with exposed soil banks, undisturbed or naturally eroding that is not a licensed/permitted aggregate area. Does not include man-made structures (bridges or buildings) or recently (2 years) disturbed soil areas, such as berms, embankments, soil or aggregate stockpiles. Does not include a licensed/permitted Mineral Aggregate Operation. Information Sources Reports and other information available from CAs Ontario Breeding Bird Atlas ^{cov} Bird Studies Canada; <i>NatureCounts</i> http://www.birdscanada.org/birdmon/ Field Naturalist clubs | Studies confirming: • Presence of 1 or more nesting sites with 8 ^{cxt/vix} or more cliff swallow pairs and/or rough-winged swallow pairs during the breeding season. • A colony identified as SWH will include a 50m radius habitat area from the peripheral nests ^{ccvii} • Field surveys to observe and count swallow nests are to be completed during the breeding season Evaluation methods to follow "Bird and Bird Habitats: Guidelines for Wind Power Projects ^{"cccii} • SWHMiST ^{ccdix} Index #4 provides development effects and mitigation measures | Cliff Swallow and Northern Rough-winged Swallow have both been idenitified in the vicinity of the subject lands. Suitable habitat not idenitified within the subject lands, but they could be located within the vicinity. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|---|--|--|--|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Colonially - Ne | sting Bird Breeding Habitat (Tre | e/Shrubs) | | | |
| Rationale: Large Colonies are important to local bird population, typically sites are only known colony in area and are used annually. | Great Blue Heron Black-crowned Night-heron Great Egret Green Heron | SWM2 SWM3 SWM5 SWM6 SWD1 SWD2 SWD3 SWD4 SWD5 SWD6 SWD7 FET1 | Nests in live or dead standing trees in wetlands, lakes, islands, and peninsulas. Shrubs and occasionally emergent vegetation may also be used. Most nests in trees are 11 to 15m from ground, near the top of the tree. Information Sources Ontario Breeding Bird Atlas^{ccv}, colonial nest records. Ontario Heronry Inventory 1991 available from Bird Studies Canada or NHIC (OMNR). NHIC Mixed Wader Nesting Colony Aerial photographs can help identify large heronries Reports and other information available from CAs MNRF District Offices Local naturalist clubs | Studies confirming: • Presence of 5 ¹ or more active nests of Great Blue Heron or other listed species. • The habitat extends from the edge of the colony and a minimum 300m radius or extent of the Forest Ecosite containing the colony or any island <15.0ha with a colony is the SWH ^{cc, coli} • Confirmation of active heronries are to be achieved through site visits conducted during the nesting season (April to August) or by evidence such as the presence of fresh guano, dead young and/or eggshells • SWHMiST ^{colik} Index #5 provides development effects and mitigation measures. | Suitable habitat is not present on the subject property. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|---|---|--|--|---|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Colonially - No | esting Bird Breeding Habitat (Gro | ound) | | | |
| Rationale: Colonies are important to local bird populations, typically sites are only known colony in area and are used annually. | Herring Gull Great Black-backed Gull Little Gull Ring-billed Gull Common Tern Caspian Tern Brewer's Blackbird | Any rocky island or peninsula (natural or artificial) within a lake or large river (two-lined on a 1:50,000 NTS map). Close proximity to watercourses in open fields or pastures with scattered trees or shrubs (Brewer's Blackbird) MAM1 – 6 MAS1 – 3 CUM CUT CUS | Nesting colonies of gulls and terns are on islands or peninsulas associated with open water or in marshy areas. Brewers Blackbird colonies are found loosely on the ground in or in low bushes in close proximity to streams and irrigation ditches within farmlands. Information Sources Ontario Breeding Bird Atlas^{cov}, rare/colonial species records. Canadian Wildlife Service Reports and other information available from CAs Natural Heritage Information Center (NHIC) Colonial Waterbird Nesting Area MNRF District Offices Field naturalist clubs | Studies confirming: • Presence of >25 active nests for Herring Gulls or Ring-billed Gulls, >5 active nests for Common Tern or >2 active nests for Caspian Tern ¹ . • Presence of 5 or more pairs for Brewer's Blackbird. • Any active nesting colony of one or more Little Gull, and Great Black-backed Gull is significant. • The edge of the colony and a minimum 150m area of habitat, or the extent of the ELC ecosites containing the colony or any island <3.0ha with a colony is the SWH ^{cc, cotii} • Studies would be done during May/June when actively nesting. Evaluation methods to follow "Bird and Bird Habitats: Guidelines for Wind Power Projects" ^{cotal} • SWHMiST ^{Cutlix} Index #6 provides development effects and mitigation measures. | Islands or peninsula not present within subject property. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|---|--|---|--|--|--|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Migratory But | terfly Stopover Areas | | | | |
| Rationale: Rationale: Butterfly stopovers areas are extremely rare habitats and are biologically important for butterfly species that migrate south for the winter. | Painted Lady Red Admiral <u>Special Concern</u> : Monarch | Combination of ELC Community Series: Need to have present one Community Series from each landclass: <u>Field:</u> CUM CUS CUT <u>Forest:</u> FOC FOM FOD CUP Anecdotally, a candidate sight for butterfly stopover will have a history of butterflies being observed. | A butterfly stopover area will be a minimum of 10 ha in size with a combination of field and forest habitat present, and will be located within 5 km of Lake Ontario ^{cdix} . • The habitat is typically a combination of field and forest, and provides the butterflies with a location to rest prior to their long migration south ^{XXXII} , XXXIV, XXXV, XXXV. • The habitat should not be disturbed, fields/meadows with an abundance of preferred nectar plants and woodland edge providing shelter are requirements for this habitat cxlviii, cxlix. • Staging areas usually provide protection from the elements and are often spits of land or areas with the shortest distance to cross the Great Lakes ^{XXXVII, XXXVII, XXXVII, XXXVII, XXXVII, XXXVII, XXXVII, XXXVII, XXXVII, XII. Information Sources • OMNRF (NHIC) • Agriculture Canada in Ottawa may have list of butterfly experts. • Field Naturalist Clubs • Toronto Entomologists Association • Conservation Authorities} | Studies confirm: • The presence of Monarch Use Days (MUD) during fall migration (Aug/Oct) ^{XIII} . MUD is based on the number of days a site is used by Monarchs, multiplied by the number of individuals using the site. Numbers of butterflies can range from 100-500/day ^{xowii} , significant variation can occur between years and multiple years of sampling should occur ^{xI, xIII} . • Observational studies are to be completed and need to be done frequently during the migration period to estimate MUD • MUD of >5000 or >3000 with the presence of Painted Ladies or Red Admiral's is to be considered significant. • SWHMIST ^{cMIX} Index #16 provides development effects and mitigation measures. | Subject property not within 5 km of Lake Ontario. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|---|--|--|--|--|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Landbird Migr | atory Stopover Areas | | | | |
| Rationale: Sites with a high diversity of species as well as high number are most significant | All migratory songbirds. Canadian Wildlife Service Ontario website: http://www.on.ec.gc.ca/wildlife_e.html All migrant raptors species: Ontario Ministry of Natural Resources: Fish and Wildlife Conservation Act, 1997. Schedule 7: Specially Protected Birds (Raptors) | All Ecosites associated with these ELC Community Series: FOC FOM FOD SWC SWM SWD | Woodlots need to be >10 ha¹ in size and within 5km ^{Iv, v, vi.} V^{II, VII, IX, XI, XII, XII, XII, XII, XV of Lake Ontario.} If multiple woodlands are located along the shoreline, those woodlands <2km from Lake Ontario are more significant^{CXIX} Sites have a variety of habitats; forest, grassland and wetland complexes^{CXIX}. The largest sites are more significant^{CXIX} Woodlots and forest fragments are important habitats to migrating birds^{CCXVIII}, these features located along the shore and located within 5km of Lake Ontario are Candidate SWH^{CCVVIII}. Information Sources Bird Studies Canada Ontario Nature Local birders and naturalist club Ontario Important Bird Areas (IBA) Program | Studies confirm: • Use of the woodlot by >200 birds/day and with >35 spp. with at least 10 bird spp. recorded on at least 5 different survey dates. This abundance and diversity of migrant bird species is considered above average and significant. • Studies should be completed during spring (Apr/May) and fall (Aug/Oct) migration using standardized assessment techniques. Evaluation methods to follow "Bird and Bird Habitats: Guidelines for Wind Power Projects" ^{codl} • SWHMiST ^{cdlix} Index #9 provides development effects and mitigation measures. | Subject property not within 5 km of Lake Ontario. Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|--|-------------------------------|---|---|---|--|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Deer Yarding | Areas | | | | |
| Rationale: Winter habitat for deer is considered to be the main factor for northern deer populations. In winter, deer congregate in "yards" to survive severe winter conditions. Deer yards | White-tailed Deer | Note: OMNRF to determine this habitat. ELC Community Series providing a thermal cover component for a deer yard | Deer yarding areas or winter concentration areas (yards) are areas deer move to in response to the onset of winter snow and cold. This is a behavioural response and deer will establish traditional use areas. The yard is composed of two areas referred to as Stratum I and Stratum II. Stratum II covers the entire winter yard area | No Studies Required: • Snow depth and temperature are the greatest influence on deer use of winter yards. Snow depths > 40cm for more than 60 days in a typically winter are minimum criteria for a deer yard to be considered as SWH ^{Wi, Wii, Mii, Iix, Ix, I} . | White-tailed Deer have been documented within the vicinity of the subject lands. Deer yarding habitat not identified within or adjacent to the subject lands by the MNRF. |
| uppically have a long history of annual use by deer, yards typically represent 10-15% of an areas summer range. | | Wourd Include: FOM, FOC, SWM and SWC. Or these ELC Ecosites: CUP2 CUP3 FOD3 CUT | arro is usually a mixed or declauous forest with plenty of browse available for food. Agricultural lands can also be included in this area. Deer move to these areas in early winter and generally, when snow depths reach 20cm, most of the deer will have moved here. If the snow is light and fluffy, deer may continue to use this area until 30cm snow depth. In mild winters, deer may remain in the Stratum II area the entire winter. • The Core of a deer yard (Stratum I) is located within the Stratum II area and is critical for deer survival in areas where winters become severe. It is primarily composed of coniferous trees (pine, hemlock, cedar, spruce) with a canopy cover of more than 60% ^{cxciv} . • OMNRF determines deer yards following methods outlined in "Selected Wildlife and Habitat Features: Inventory Manual ^{mCxcv} • Woodlots with high densities of deer due to artificial feeding are not significant. | Deer Yards are mapped by OMNRF District offices. Locations of Core or Stratum 1 and Stratum 2 Deer yards considered significant by OMNRF will be available at local MNRF offices or via Land Information Ontario (LIO). Field investigations that record deer tracks in winter are done to confirm use (best done from an aircraft). Preferably, this is done over a series of winters to establish the boundary of the Stratum 1 and Stratum II yard in an "average" winter. MNRF will complete these field investigations^{COCV}. If a SWH is determined for Deer Wintering Area or if a proposed development is within Stratum II yarding area then Movement Corridors are to be considered as outlined in Table 1.4.1 of this Schedule. SWHMiST^{Codix} Index #2 provides development effects and mitigation measures. | Not SWH. |

| | Wildlife Species ¹ | | Candidate SWH | Confirmed SWH | Study Area |
|---|-------------------------------|--|--|--|---|
| | | ELC Ecosite Codes ¹ | Habitat Criteria and Information Sources ¹ | Defining Criteria ¹ | Assessment Details |
| Wildlife Habitat: Deer Winter C | ongregation Areas | | | | |
| Rationale: Deer movement during winter in the southern areas of Ecoregion 6E are not constrained by snow depth, however deer will annually congregate in large numbers in suitable woodlands to reduce or avoid the impacts of winter conditions ^{extrill} | White-tailed Deer | All Forested Ecosites with these ELC Community Series: FOC FOM FOD SWC SWM SWD Conifer plantations much smaller than 50ha may also be used. | Woodlots will typically be >100 ha in size. Woodlots <100ha may be considered as significant based on MNRF studies or assessment. Deer movement during winter in the southern areas of Eco-region 6E are not constrained by snow depth, however deer will annually congregate in large numbers in suitable woodlands^{colviii}. If deer are constrained by snow depth refer to the Deer Yarding Area habitat within Table 1.1 of this Schedule. Large woodlots > 100ha and up to 1500 ha are known to be used annually by densities of deer that range from 0.1-1.5 deer/ha^{ccolvi}. Woodlots with high densities of deer due to artificial feeding are not significant. Information Sources MNRF District Offices LIO/NRVIS | Studies confirm: • Deer management is an MNRF responsibility, deer winter congregation areas considered significant will be mapped by MNRF ^{cotvill} . • Use of the woodlot by white-tailed deer will be determined by MNRF, all woodlots exceeding the area criteria are significant, unless determined not to be significant by MNR ¹ . • Studies should be completed during winter (Jan/Feb) when >20cm of snow is on the ground using aerial survey techniques ^{cocxiv} , ground or road surveys, or a pellet count deer density survey ^{cocxv} . • If a SWH is determined for Deer Wintering Area of if a proposed development is within Stratum II yarding area then Movement Corridors are to be considered as outlined in Table 1.4.1 of this Schedule. • SWHMIST ^{codix} Index #2 provides development effects and mitigation measures. | White-tailed Deer have been documented within the vicinity of the subject lands. Deer overwintering habitat not identified within or adjacent to the subject lands by the MNRF. Not SWH. |



Memo

Project #2561

| To: | Carla IIIman, MTE |
|-------|---|
| From: | Ryan Archer |
| Date: | July 11, 2022 |
| Re: | Baden Trunk Sanitary Sewer Environmental Assessment |
| | Evaluation of Alternatives – Natural Environment Criteria |

1.0 NATURAL ENVIRONMENT

This component of the assessment of alternatives is based on the potential for direct or indirect impacts on the natural environment features including terrestrial, wetland and aquatic ecosystems. Details of the existing natural features and their functions, including assessments of their ecological significance and sensitivity, are provided in the draft Natural Environment Assessment (NEA) report (NRSI 2022).

1.A Terrestrial Ecosystems

The Terrestrial Ecosystems factor includes the sub-factors "Vegetation Communities and Flora" and "Wildlife Habitat", each of which were assessed and scored separately to inform the alternatives evaluation. This factor group includes assessment of potential direct or indirect impacts to terrestrial habitat features and functions, including the woodland and cultural meadow vegetation communities that were mapped within the study area. The ecological sensitivities of these natural features vary as described further in the NEA (NRSI 2022), and this has been accounted for in the evaluation scores. Significant features that were considered in this factor group include Locally Significant Woodlands, which represents a form of Supporting Environmental Feature as defined by the Township of Wilmot (Township of Wilmot 2019), as well as non-significant woodlands. Indirect impacts to these features were also considered where alternative designs are proposed on adjacent lands.

Significant and sensitive vegetation species are also considered in the assessment, including habitat for Species at Risk (SAR) (defined herein as species designated as Endangered or Threatened in Ontario (MECP 2022)). As described in the NEA, the SAR Butternut (*Juglans cinerea*) was confirmed within the study area. The regionally significant species Running Serviceberry (*Amelanchier spicata*) was also identified within cultural meadow at the north end of the study area (NRSI 2022). None of the five options will directly impact the Butternuts or require removal of their 25m radius habitat zones.

This factor group also assesses direct impacts to wildlife habitats, including habitat for SAR, and Significant Wildlife Habitat (SWH) as defined by provincial criteria (OMNR 2000, MNRF 2015). As described in the NEA, habitat for bat SAR may occur within the study area woodlands. The study area woodlands were also identified as Candidate SWH for Bat Maternity Colonies, while the mill pond was identified as Candidate SWH for Turtle Wintering Areas. Direct impacts to these and other wildlife habitats may occur where physical removal of the habitat is required.

Indirect impacts may occur where disturbances to wildlife arise through construction or operation of the infrastructure such that the habitat function is effectively eliminated, degraded or otherwise reduced in ecological value.

Where route alignment options fall within existing municipal road right-of-ways (ROW) it is anticipated that direct natural environment impacts will be limited to potential roadside construction disturbance areas. Common to all five options, a lateral connection of the infrastructure between Foundry Street and the Baden wastewater pumping station will be required. This small lateral section comprises open manicured lawn and does not provide important wildlife habitat; it is therefore not considered in the assessment below.

1) Vegetation Communities and Flora

Option A1 (Score 4)

Portions of this alignment occur entirely within the ROW of Snyder's Road and Foundry Street, limiting its impact on natural features along these sections other than to potential roadside construction disturbances. However, a portion of the alignment northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection will occur off-road where it will align with the route of an existing pedestrian trail. Construction of this segment of the infrastructure may directly impact Mineral Cultural Woodland (CUW1) and Mineral Cultural Meadow (CUM1) (NRSI 2022). Indirect construction-related impacts to adjacent natural features in this northeast portion of the study area, including to areas of Mineral Cultural Thicket (CUT1) and adjacent areas of CUM1, may also occur unless appropriately mitigated (e.g., through construction and/or silt fencing to demarcate construction limits, mitigate erosion and sedimentation impacts).

Impacts to terrestrial natural features along the Foundry Street segment will be negligible and mainly confined to the ROW.

Option A2 (Score 10)

This alignment occurs entirely within the Snyder's Road and Foundry Street ROWs. Impacts to terrestrial natural features along the route of Option 2 will be negligible and mainly confined to the ROW.

Option A3 (Score 8)

This alignment occurs entirely within the ROWs of Christian Street, Charlotta Street, Mill Street, Charles Street and Foundry Street. Impacts to terrestrial natural features along the route of Option 3 will be negligible and mainly confined to the ROW. However, if roadside disturbances occur due to construction there is potential for damage or destruction of individuals of the regionally significant Running Serviceberry that was documented within the CUM1 meadow south of Charlotta Street, unless appropriately mitigated. For this reason, it is scored somewhat lower than Option A2 above.

Option A4 (Score 1)

Among the assessed options, Option A4 requires one of the largest extents of construction work within natural/naturalizing land cover. This option will require temporary construction disturbance within areas of CUM1 meadow, although this vegetation community type is already reflective of ecologically disturbed conditions and is therefore resilient to disturbance effects.

Option A4 will follow the south boundary of the Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3) located at the southwest study area bend of Baden Creek. This woodland was identified as a Locally Significant Woodland in the NEA (NRSI 2022). Construction of the infrastructure in this area may potentially require edge tree removal, or disturbance to tree root zones, if not appropriately mitigated. It is recommended that the sewer alignment be located such that construction disturbances are located outside of the woodland dripline, and offset from the dripline to the extent feasible.

Option A4 will also include a segment that runs in proximity to the FOD7-3 Locally Significant Woodland that is located along Baden Creek south of Queen Street and St. George Street. However, direct impacts to this woodland are not anticipated. Indirect impacts to the adjacent woodland as caused through construction activities may occur if not appropriately mitigated (e.g., demarcation of construction limits). This option will not negatively impact the Butternut trees and is situated outside of their 25m-radius habitat zones (see NEA Map 3 (NRSI 2022)).

This option will require a small number of tree removals where the alignment passes through agricultural hedgerows and isolated tree growth. It may directly impact individuals of the regionally significant Running Serviceberry that were documented within the CUM1 meadow south of Charlotta Street.

Option A5 (Score 5)

This option has a similar alignment to Option A4 except for its northern extent, where it will extend approximately southeast from Christian Street before traversing the south side of Baden Creek. This option will traverse less CUM1 meadow and more agricultural land than Option A4.

Option A5 will follow the south boundary of the FOD7-3 woodland located at the southwest study area bend of Baden Creek. This woodland was identified as a Locally Significant Woodland in the NEA (NRSI 2022). Construction of the infrastructure in this area may potentially require edge tree removal, or disturbance to tree root zones, if not appropriately mitigated. It is recommended that the sewer alignment be located such that construction disturbances are located outside of the woodland dripline, and offset from the dripline to the extent feasible.

2) Wildlife Habitat

Certain wildlife habitat impacts are general in nature and apply to all five of the alternative alignment options within the study area. Impacts that are specific to individual options are presented below.

Construction of the sewer alignments may cause general disturbance to wildlife occupying the immediate area of the construction zone. However, these wildlife species are generally urbanadapted and tolerant of some degree of disturbance. Certain wildlife species may temporarily avoid the area but will return post-construction.

Vegetation removal can cause injury or mortality to birds, and damage or destruction of their nests, if not appropriately mitigated. Vegetation removals should be time to occur outside of the period April 1-August 31 in order to avoid contravention of the federal *Migratory Birds Convention Act.* Where "simple" bird habitats require removal (e.g., isolated trees or hedgerows), removal within the bird nesting period may be possible subject to prior inspection of the vegetation for the presence of active nests by a qualified biologist.

Option A1 (Score 1)

This alignment will require removal of, or disturbance to, wetland features located northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection. These wetlands were not investigated for wildlife habitat use as part of the NEA. However, they likely provide habitat for breeding amphibians as well as other wildlife. The wetland wildlife habitats need to be considered potentially significant unless demonstrated otherwise. This alignment will therefore directly and/or indirectly impact these wildlife habitat functions.

It is NRSI's understanding that construction of the sewer within the Foundry Street ROW will require some in-water work, including isolated dewatering of Baden Creek with flow bypass. In-water construction may extend into the eastern edge of the mill pond, which is considered Candidate SWH for Turtle Wintering Areas. In-water work within this area may therefore directly impact this habitat function unless appropriately mitigated (e.g., minimizing disturbance area, maintaining work outside of turtle overwintering period where suitable habitat may be impacted). Provided appropriate measures are followed, and given the small size of the potential disturbance zone, negative impacts to turtle overwintering habitat are considered unlikely.

A small amount of tree removal may be required to accommodate the sewer installation, primarily to the northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection where Mineral Cultural Woodland (CUW1) occurs. Trees requiring removal may represent roosting habitat for SAR bats. These trees should be inspected for the presence of suitable bat habitat prior to removal. If suitable bat habitat is present, MECP consultation may be required to ensure that steps are taken to avoid or mitigate impacts to bat SAR or their habitat. At a minimum, potential bat habitat trees should be removed outside of the bat active period (April 1-September 30).

Option A2 (Score 6)

This alignment occurs entirely within the Snyder's Road and Foundry Street ROWs. Impacts to wildlife habitats along the majority of the Option A2 route will be negligible. Under this option no woodland habitats will be impacted.

Consistent with Options A1 and A3, this option may require isolation and dewatering of an eastern edge of the mill pond that represents Candidate SWH for Turtle Wintering Areas. However, provided appropriate measures are followed, and given the small size of the potential disturbance zone, negative impacts to turtle overwintering habitat are considered unlikely.

Option A3 (Score 5)

This alignment occurs entirely within the ROWs of Christian Street, Charlotta Street, Mill Street, Charles Street and Foundry Street. Impacts to wildlife habitats along the majority of the Option A3 route will be negligible. Under this option no woodland habitats will be impacted.

Consistent with Options A1 and A2, this option may require isolation and dewatering of an eastern edge of the mill pond that represents Candidate SWH for Turtle Wintering Areas. However, provided appropriate measures are followed, and given the small size of the potential disturbance zone, negative impacts to turtle overwintering habitat are considered unlikely.

Option A4 (Score 4)

Among the assessed options, Option A4 requires one of the largest extents of construction work within natural/naturalizing land cover. This alignment will require the removal of habitat for

wildlife occupying the small, isolated wetland features located south of the railroad tracks and adjacent to Queen Street, identified as Mineral Meadow Marsh (MAM2) and Reed Canary Grass Mineral Meadow Marsh (MAM2-2) in the NEA (NRSI 2022). These wetland features may be used for amphibian breeding and as habitat for other disturbance-tolerant wetland species (e.g., Red-winged Blackbird (*Agelaius phoeniceus*)). If feasible, the alignment of the sewer should be designed to avoid these habitat features.

Option A4 will follow the south boundary of the Fresh-Moist Willow Lowland Deciduous Forest (FOD7-3) located at the southwest study area bend of Baden Creek. It will also come into proximity of the FOD7-3 Locally Significant Woodland that is located along Baden Creek south of Queen Street and St. George Street. Both of these features are identified as potential bat SAR habitat as well as Candidate SWH for Bat Maternity Colonies. The sewer alignment should be designed to avoid direct impacts to these woodland habitats, and should be offset to the extent feasible from the woodland edge to minimize potential for root zone impacts.

This option will require a small number of tree removals where the alignment passes through agricultural hedgerows and isolated tree growth. Trees requiring removal may represent roosting habitat for SAR bats. These trees, as well as any trees that may be impacted within the FOD7-3 woodlands, should be inspected for the presence of suitable bat habitat prior to removal. If suitable bat habitat is present, MECP consultation may be required to ensure that steps are taken to avoid or mitigate impacts to bat SAR or their habitat. At a minimum, potential bat habitat trees should be removed outside of the bat active period (April 1-September 30).

Direct removal of habitat for wildlife occupying CUM1 meadow will be required (e.g., small mammals, snakes, open habitat birds). However, this habitat removal will be temporary, as CUM1 meadow habitats are resilient to disturbance and will readily renaturalize. The majority of species occupying this habitat type are generally disturbance tolerant and mobile, and will not be negatively impacted.

Option A5 (Score 10)

This option has a similar alignment to Option A4 except for its northern extent, where it will extend approximately southeast from Christian Street before traversing the south side of Baden Creek. This option will traverse less Mineral Cultural Meadow (CUM1) wildlife habitat and more agricultural land than Option A4.

Option A5 will not require removal or cause indirect disturbance to wetland habitats.

This option may impact a portion of a Mineral Cultural Woodland (CUW1) located south of Christian Street and the railroad tracks. Trees requiring removal may represent roosting habitat for SAR bats. These trees should be inspected for the presence of suitable bat habitat prior to removal. If suitable bat habitat is present, MECP consultation may be required to ensure that steps are taken to avoid or mitigate impacts to bat SAR or their habitat. At a minimum, potential bat habitat trees should be removed outside of the bat active period (April 1-September 30).

1.B Wetland Ecosystem

Direct impacts assessed for this factor include loss of wetland habitat features and functions. Wetland cover within the study area is limited to small, isolated features that are anthropogenic in nature and/or have been highly altered by past land uses. Wetlands that are not provincially significant are generally considered "locally significant" in the Township. Locally Significant Wetlands are considered a form of Supporting Environmental Feature (Township of Wilmot 2019). Indirect impacts to these features were also considered where alternative designs are proposed on adjacent lands, including within ecological buffers to the features where these have been recommended (NRSI 2022).

This factor also includes the subfactor of "watercourses", which, for the purposes of this assessment, is integrated into the scoring under the "wetlands" factor. For the purposes of this subfactor, impacts to watercourses are addressed without consideration for aquatic ecosystems, which are addressed in a separate subfactor (see below). In the context of this subfactor, impacts are considered in the scoring relative to the number of watercourse crossings and potential for abiotic effects (e.g., changes in drainage or groundwater flow). See below (Aquatic Ecosystems) for a more fulsome description of watercourses present within the study area.

Option A1 (Score 1)

Portions of this alignment occur entirely within the ROW of Snyder's Road and Foundry Street, limiting its impact on natural features along these sections other than to potential roadside construction disturbances. However, a portion of the alignment northeast of the Snyder's Road/Foundry Street/Livingston Boulevard intersection will occur off-road where it will align with the route of an existing pedestrian trail. Construction of this segment of the infrastructure may directly impact wetland features. The wetland in this area (mapped as Meadow Marsh (MAM) (NRSI 2022)) is considered to represent Locally Significant Wetland. Indirect construction-related impacts to adjacent natural features in this northeast portion of the study area, including to areas of meadow marsh, may also occur unless appropriately mitigated (e.g., through construction and/or silt fencing to demarcate construction limits, mitigate erosion and sedimentation impacts). Construction of the infrastructure may also disrupt the drainage regime in this area, which may alter the hydrological conditions that currently support the wetland in this location.

Option A2 (Score 8)

This alignment occurs entirely within the Snyder's Road and Foundry Street ROWs. No impacts to wetlands will occur as a result of this option.

However, this alignment will require two permanent watercourse crossings: a crossing of Baden Creek within the Foundry Street ROW, and a crossing of the Huehn Award Drain within the Foundry Street ROW. These crossings will require a temporary construction-stage re-routing of flows via a bypass culvert for each watercourse. Relative to Option A5, which only requires one crossing of a permanent watercourse (see below), this alternative therefore imposes slightly more impact potential.

Option A3 (Score 9)

This alignment occurs entirely within the ROWs of Christian Street, Charlotta Street, Mill Street, Charles Street and Foundry Street. No wetland removals will be required to accommodate this alignment. Although this route occurs in proximity to a small Mineral Meadow Marsh (MAM2) wetland located south of the railroad tracks, nearby excavation for the sewer installation is not anticipated to indirectly affect this wetland's form or function since it likely results from a small surface runoff catchment.

This alignment will require two permanent watercourse crossings: a crossing of Baden Creek within the Foundry Street ROW, and a crossing of the Huehn Award Drain within the Foundry Street ROW. These crossings will require a temporary construction-stage re-routing of flows via
a bypass culvert for each watercourse. Relative to Option A5, which only requires one crossing of a permanent watercourse (see below), this alternative therefore imposes slightly more impact potential.

Option A4 (Score 4)

This option will require the removal of the small, isolated wetland features located south of the railroad tracks and adjacent to Queen Street, identified as Mineral Meadow Marsh (MAM2) and Reed Canary Grass Mineral Meadow Marsh (MAM2-2) in the NEA (NRSI 2022). If feasible, the alignment of the sewer should be designed to avoid these wetland features.

Construction that is located immediately adjacent to the wetlands may results in an alteration of the surface drainage catchments that likely sustain them, at least during the construction period. This may result in increased or decreased flows to the wetlands. Construction adjacent to the wetlands may also lead to water quality impairment, such as sedimentation and vegetation damage. This can be mitigated through measures such as construction limit demarcation and erosion and sediment control measures.

Option A5 (Score 10)

This option has a similar alignment to Option A4 except for its northern extent, where it will extend approximately southeast from Christian Street before traversing the south side of Baden Creek.

Option A5 will not require removal or cause indirect disturbance to wetland features. As described further below, it will require a crossing of Baden Creek just north of Gingerich Road, as well as a crossing of Tributary A. However, the crossing of a single permanent watercourse (Baden Creek) presents less potential for impact relative to other alternatives. The crossing of the upstream ephemeral reach of Tributary A can occur with no impacts if done during dry conditions. These considerations, in addition to the lack of wetland impacts, presents the least potential for impact within this factor.

1.C Aquatic Ecosystem

As described in the NEA (NRSI 2022), the study area includes the permanent watercourses Baden Creek, Huehn Award Drain and Baden Creek Tributary B. One watercourse, Baden Creek Tributary A, was characterized as an ephemeral/intermittent stream. The Huehn Award Drain is a coldwater watercourse. Baden Creek is a warmwater watercourse, but is at the cooler end of the thermal range due to localized groundwater inputs. Tributary B also shows evidence of some groundwater inputs. The study area watercourses provide direct fish habitat with the exception of Baden Creek Tributary A. The large mill pond online with Baden Creek is an anthropogenic feature resulting from historic damming of the watercourse.

Direct impacts assessed for this factor include impacts associated with watercourse crossings of the sewer alignment, including the construction activities necessary to accommodate the sewer. This is anticipated to require in-water work in certain locations of permanent watercourse flow, such as at the crossings of Baden Creek and the Huehn Award Drain along Foundry Street. It is NRSI's understanding that temporary watercourse isolation and dewatering, with flow bypass, will be required at locations where flow is present at the Foundry Street ROW watercourse crossings. At Baden Creek, this in-water work zone may also extend into the far east end of the mill pond. As described further below, trenchless construction techniques will be utilized at the crossings of Baden Creek at Gingerich Road, and at Tributary A.

Watercourse crossings and construction adjacent to watercourses is common to all five of the alternative design options. Indirect impacts are possible where construction may occur in close proximity, such that water quality impairment may result if not appropriately mitigated. Construction-stage impacts will be temporary in nature. However, longer-term impacts can result if sewer leakages occur in proximity to aquatic features.

Construction activities required for sewer installation are not anticipated to influence the thermal regime of the adjacent watercourse, since it is not going to result in additional warmwater flows into the watercourse and will not result in impounding or additional sun exposure of the water.

All five of the route options will require crossing Baden Creek between the Foundry Street ROW and the Baden wastewater pumping station, south of Gingerich Road. As such, this watercourse crossing is not considered in the evaluation of alternatives since potential impacts associated with this crossing are common to all.

Option A1 (Score 1)

This alignment will require three permanent watercourse crossings: a crossing of Baden Creek Tributary B north of Snyder's Road, a crossing of Baden Creek within the Foundry Street ROW, and a crossing of the Huehn Award Drain within the Foundry Street ROW. This represents the greatest number of watercourse crossings relative to the other alternatives, which increases the risks of potential negative impacts to the aquatic features as a result of construction activities or post-construction leakage of the infrastructure.

All three of these watercourses provide direct fish habitat. Any activities that may cause the Harmful Alteration, Disruption or Destruction (HADD) of fish or fish habitat is subject to the requirements of the federal *Fisheries Act*. Details of the proposed sewer installation activities will be subject to review by the federal Department of Fisheries and Oceans (DFO) through submission of a Request for Review prior to construction. The DFO will provide a response with recommended Best Management Practices to suitably avoid or mitigate impacts to fish and fish habitat, or will determine that the proposed undertaking will require a Fisheries Act Authorization. Acquisition of an authorization under the *Fisheries Act* will require additional measures to provide an Overall Benefit to fish or fish habitat, such as through the restoration or enhancement of aquatic ecosystems.

All three of these watercourses receive shallow groundwater inputs, potentially including within the required construction zones. Trenching and installation of sewer infrastructure may intercept the shallow groundwater level and cause alterations to the flow path that may affect the locations and volumes of groundwater inputs into the watercourses. Measures may be required to ensure that groundwater inputs to the watercourses are not altered as a result of the undertaking.

Watercourse isolation and dewatering activities will require the completion of a fish capture and relocation prior to drawdown. A portion of this isolation and dewatering zone may need to extend into the east end of the mill pond at the Baden Creek crossing. Flow bypass will be required to maintain flow volumes into downstream reaches and receiving features. Water quality control measures will be required to ensure the appropriate removal of suspended sediments prior to discharge of bypass flows.

Construction activities immediately adjacent to watercourses increases the risk of water quality degradation through inputs of sediments and other deleterious substances. This risk can be effectively mitigated through design and implementation of an acceptable Erosion and Sediment

Control Plan. This plan would include, but not be limited to, installation of silt fencing keyed into the ground, regularly inspected and repaired when necessary; limiting exposure of bare soils within the construction site; and avoiding removal of vegetation and work within natural riparian zones to the extent possible.

Option A2 (Score 4)

This alignment will require two permanent watercourse crossings: a crossing of Baden Creek within the Foundry Street ROW, and a crossing of the Huehn Award Drain within the Foundry Street ROW. This is one less watercourse crossing than is required for Option A1.

Potential impacts associated with this option are otherwise the same as Option A1, described above. These generally include the following:

- Potential impacts to fish and fish habitat, requiring DFO review and possible authorization under the *Fisheries Act*;
- Potential for alteration of shallow groundwater flow inputs to the watercourses;
- Need for watercourse isolation and de-watering, with associated fish capture and relocation; need for water quality treatment of flow bypass discharge;
- General construction-stage impact potential to adjacent watercourse features

The potential for these effects are generally lessened in comparison to Option A1 due to a lower number of required watercourse crossings.

Option A3 (Score 3)

This alignment will require two permanent watercourse crossings: a crossing of Baden Creek within the Foundry Street ROW, and a crossing of the Huehn Award Drain within the Foundry Street ROW. This is one less watercourse crossing than is required for Option A1, and the same number as Option A2. Although both Options A2 and A3 require deep sewer installation under Foundry Street, Option A3 is deeper than Option A2. It is NRSI's understanding that this may present slightly more impact potential relative to Option A2.

Option A4 (Score 7)

This alignment will require a crossing of Baden Creek just north of Gingerich Road, as well as a crossing of Tributary A just upstream of its confluence with Baden Creek.

The reach of Tributary A to be crossed, along or just east of the existing tractor crossing path, was characterized as an area of intermittent flow (NRSI 2022). During higher flow periods this reach of the watercourse may accessible to fish from Baden Creek, and therefore may function as direct fish habitat. A crossing of this watercourse, in addition to the direct fish habitat provided by Baden Creek, will be subject to review by the DFO including potential requirement for a Fisheries Act Authorization as described for Option A1. If possible, work within the Tributary A drainage channel should be done during periods of no flow/no standing water to mitigate potential impact to fish and fish habitat.

Sewer installation at the crossings of Baden Creek at Gingerich Road and at Tributary A will use trenchless methodologies. Rather than an open trench cut for installation, trenchless technologies will be employed to install a sleeve under each watercourse crossing, within which the sewer will be installed. At the Baden Creek crossing, the sewer will be installed below the elevation of the existing box culvert. No impacts to creek flows within the culvert will occur as a

result of this installation. No watercourse isolation, dewatering or fish capture/relocation will be required at these locations as a result of this construction plan.

Although Baden Creek is a warmwater watercourse, evidence of groundwater inputs was observed in the vicinity of the proposed crossing. However, the sewer installation described above is not anticipated to cause alteration to any shallow groundwater flows or to cause changes in groundwater inputs to Baden Creek. Tributary A is not connected to the shallow groundwater system; as such, no groundwater-related impacts to the watercourse will occur at this crossing.

Portions of the proposed alignment will occur in close proximity to Baden Creek. Increased proximity to the watercourse may result in greater potential for construction-stage disturbances or potential future leakage of the sewer infrastructure that may affect the watercourse. To the extent possible, the sewer alignment should be maintained outside of the recommended 15m watercourse buffer identified in the NEA (NRSI 2022) to mitigate these potential impacts.

Option A5 (Score 10)

This alignment will require a crossing of Baden Creek within the Gingerich Road ROW, as well as a crossing of Tributary A. However, the reach of Tributary A to be crossed, just south of its source at the stormwater management pond outlet, has been characterized as an ephemeral flow path (NRSI 2022). During periods of seasonal flow this reach of the tributary may represent indirect habitat for fish (e.g., by providing nutrients to downstream direct fish habitat). Provided work is undertaken outside of periods immediately following heavy precipitation or snow melt when flow may occur, no impacts to fish or fish habitat will occur at the Tributary A crossing.

The crossing of Baden Creek, which provides direct fish habitat, will be subject to review by the DFO including potential requirement for a Fisheries Act Authorization as described for Option A1.

As described for Option A4, no watercourse isolation, dewatering, or fish capture/relocation will be required for the Baden Creek crossing due to the planned trenchless installation method. These measures will also not be required for Tributary A.

Although Baden Creek is a warmwater watercourse, evidence of groundwater inputs was observed in the vicinity of the proposed crossing. However, the sewer installation described above is not anticipated to cause alteration to any shallow groundwater flows or to cause changes in groundwater inputs to Baden Creek. Tributary A is not connected to the shallow groundwater system; as such, no groundwater-related impacts to the watercourse will occur at this crossing.

Portions of the proposed alignment will occur in close proximity to Baden Creek. Increased proximity to the watercourse may result in greater potential for construction-stage disturbances or potential future leakage of the sewer infrastructure that may affect the watercourse. To the extent possible, the sewer alignment should be maintained outside of the recommended 15m watercourse buffer identified in the NEA (NRSI 2022) to mitigate these potential impacts.

References

- Natural Resource Solutions Inc. (NRSI). 2022. Baden Trunk Sanitary Sewer Environmental Assessment, Natural Environment Assessment (Draft). Prepared for MTE Consultants Inc. April 2022.
- Ontario Ministry of Environment, Conservation and Parks (MECP). 2022. Species at Risk in Ontario List. https://www.ontario.ca/page/species-risk-ontario. Last updated April 1, 2022.
- Ontario Ministry of Natural Resources (OMNR). 2000. Significant Wildlife Habitat Technical Guide. October 2000.
- Ontario Ministry of Natural Resources and Forestry (MNRF). 2015a. Significant Wildlife Habitat Ecoregion 6E Criterion Schedule: Addendum to Significant Wildlife Habitat Technical Guide. MNRF, January 2015.



The Baden Trunk EA Geotechnical Site Characterization Report (May 3, 2022)





Baden Sanitary Trunk EA

Geotechnical Site Characterization

Project Location: Baden, Township of Wilmot, ON

Prepared for: Township of Wilmot 60 Snyder's Road West Baden, ON

Prepared by: MTE Consultants 520 Bingemans Centre Drive Kitchener, ON N2B 3X9

Date: November 8, 2022

MTE File No.: 48730-100



Engineers, Scientists, Surveyors.



Contents

| 1.0 | Intr | oduction | 1 |
|-----|-------|--|---|
| 2.0 | Pre | evious Studies | 1 |
| 3.0 | Fie | ld and Laboratory Program | 1 |
| 4.0 | Soi | I and Groundwater Conditions | 2 |
| 4.1 | 1 | 012 Snyder's Road West | 3 |
| 4.2 | S | Snyder's Road West | 3 |
| 4.3 | F | Foundry Street | 3 |
| 4.4 | S | Schmidt Lands | 3 |
| 2 | 4.4.1 | Northwest Field Area | 3 |
| 2 | 4.4.2 | Development Portion – North of Baden Creek | 4 |
| 2 | 4.4.3 | Development Portion – South of Baden Creek | 4 |
| 5.0 | Lim | nitations of Report | 5 |

Appendices

| Appendix A | Figures |
|------------|-------------------------|
| Appendix B | Borehole Logs |
| Appendix C | Laboratory Test Results |

1.0 Introduction

MTE Consultants Inc. (MTE) was retained by the Township of Wilmot to conduct a geotechnical site characterization investigation for a proposed sanitary trunk sewer in Baden, Ontario. The sanitary trunk is proposed to extend from the northwest edge of Baden to the Baden wastewater pumping station, located at 1044 Foundry Street in the south portion of Baden. Multiple options are considered, as shown on **Figure 1 in Appendix A**.

The purpose of this geotechnical site characterization investigation is to determine the general soil and groundwater conditions in the area of the proposed sanitary trunk sewer options.

2.0 Previous Studies

In April 2021, MTE conducted a Development Hydrogeology Study for the property located at 1012 Snyder's Road West. The fieldwork included the drilling of eleven (11) monitoring wells (Monitoring Wells MW101-19 to MW106C-19, and MW201-20 to MW203-20) and one (1) mini piezometer (Mini Piezometer MP1-20).

In June 2021, MTE conducted a Phase II Environmental Site Assessment (ESA) for the property located at 1012 Snyder's Road West. The fieldwork included the drilling of five (5) boreholes (Boreholes BH303-20, BH305A-20, and BH306-20 to BH308-20) and six (6) monitoring wells (Monitoring Wells MW301-20, MW302-20, MW304-20, MW305B-20, MW309-20, and MW310-20).

In March 2021, Englobe Corporation (Englobe) conducted a Draft Soil Characterization Report along Snyder's Road. Four (4) boreholes (Boreholes BH-01-21 to BH-04-21) were conducted within the subject area of this study.

In January 2020, Pinchin Ltd. (Pinchin) conducted a Geotechnical Investigation along Snyder's Road (Pinchin File No. 244512). One (1) monitoring well (Monitoring Well BH1) and six (6) boreholes (Boreholes BH2 to BH7) were conducted within the subject area of this study.

In April 2020, Pinchin conducted a Draft Geotechnical Investigation for the property located northwast of the Foundry Street and Gingerich Road (Pinchin File No. 249839). The fieldwork included the drilling of seven (7) boreholes (Boreholes BH1, BH3, BH5 to BH7, BH9, and BH11) and six (6) monitoring wells (Monitoring Wells BH2, BH4, BH8, BH10, BH12, and BH13).

The previous studies boreholes and monitoring wells will be referenced throughout this report and are shown on **Figure 1 in Appendix A** and the logs are provided in **Appendix B**.

3.0 Field and Laboratory Program

The fieldwork for this investigation was carried out on August 12, September 30, and October 14, 2021 and involved the drilling of five (5) boreholes (Boreholes MW101-21 to MW105-21) to depths of 5.0 to 8.2 m. The locations of the boreholes are shown on the Site Plan, **Figure 1 in Appendix A**.

Private and public utility companies were contacted prior to the start of drilling activities in order to isolate underground utilities near the boring locations.

On August 12 and October 14, 2021 the boreholes were advanced with a Diedrich D50T track mounted drill rig equipped with continuous flight hollow stem augers, supplied and operated by London Soil Test Ltd.

MTE Consultants | 48730-100 | Baden Sanitary Trunk EA | November 8, 2022

On September 30, 2021, the boreholes were advanced with a Diedrich D50T track mounted drill rig equipped with continuous flight hollow stem augers, supplied and operated by Envirocore.

Representative soil samples were recovered throughout the depths explored. Standard Penetration Tests (SPT) were carried out during sampling operations in the boreholes using conventional split spoon equipment. Approximate shear strengths of the cohesive deposits were measured with a handheld pocket penetrometer. The SPT N-values and approximate shear strengths recorded are plotted on the borehole logs in **Appendix B**.

Upon completion of drilling, 50 mm diameter monitoring wells were installed in the boreholes to allow measurement of stabilized groundwater levels and groundwater sampling and testing. The installations comprised 1.5 m and 3.0 m filtered screens and bentonite seals above and below the screens. Details of the installation and groundwater observations and measurements are provided on the appended borehole logs.

The monitoring wells were installed in accordance to Ontario Regulation 468/10. The construction, maintenance and abandonment of the wells are regulated under the province's Water Resources Act. The monitoring well network located on the site must be maintained or decommissioned in accordance with regulatory requirements.

The fieldwork was monitored throughout by a member of our geotechnical engineering staff, who directed the drilling procedures; recorded SPT and approximate shear strength values; documented the soil stratigraphies; monitored the groundwater conditions and monitoring well installations; and transported the recovered soil samples to our office for further classification.

The borehole coordinates and ground surface elevations were surveyed by MTE with a Trimble Global Navigation Satellite System (GNSS) model R10 rover. The borehole locations are referenced to Canadian Spatial Reference System (CSRS 1997) coordinates. The geodetic ground surface elevations are based on GNSS and local base station telemetry and have a vertical root mean squared error of less than 20 mm.

The collected soil samples were submitted for moisture content testing with the results provided on the borehole logs in **Appendix B**. Additionally, five (5) soil samples were submitted for particle size distribution analyses and the results are provided in **Appendix C**. The remaining soil samples will be stored for a period of 1 month and will be discarded of at that time without prior request from the client to extend storage time.

4.0 Soil and Groundwater Conditions

For the purpose of this report the site area has been broken into four regions. The four regions and the alternatives located within the regions are as follows:

- 1012 Snyder's Road West Alternatives 2, 3, 4, and 5
- Snyder's Road West Alternatives 1 and 2
- Foundry Street Alternatives 1, 2, and 3
- Schmidt Lands Alternatives 4 and 5

Additionally, the Schmidt Lands were further divided into three subregions as follows:

- Northwest Field Area
- Development Portion North of Baden Creek
- Development Portion South of Baden Creek

MTE Consultants | 48730-100 | Baden Sanitary Trunk EA | November 8, 2022

4.1 1012 Snyder's Road West

MTE Boreholes BH303-20, BH305A-20, and BH306-20 to BH308-20, MTE Monitoring Wells MW101-19 to MW106C-19, MW201-21 to MW203-21, MW301-20, MW302-20, MW304-20, MW305B-20, MW309-20, and MW310-20, and MTE Mini Piezometer MP1-20 were advanced in the 1012 Snyder's Road West portion of the study area.

Topsoil overlying silt and sand deposits were encountered in the majority of the boreholes and monitoring wells with glacial till encountered in MTE Boreholes MW102-19, MW103-19, MW202-20, MW203-20, MP1-20, MW301-20, MW302-20, MW304-20 to MW305B-20, and MW310-20.

Groundwater was measured in the installed MTE monitoring wells at depths of 1.0 to 12.6 m (Elevation 351.4 to 359.6 m) on February 2, 2021. It is noted that MTE Monitoring Wells MW104-19 and MW304-20 were dry on February 2, 2021.

4.2 Snyder's Road West

Pinchin (File No. 244512) Monitoring Well BH1, Pinchin (File No. 244512) Boreholes BH2 to BH7, and Englobe Boreholes BH-01-21 to BH-04-21 were advanced along Snyder's Road West between Christian Street to Foundry Street.

The soil conditions encountered in the boreholes and monitoring well along Snyder's Road West typically include pavement structure and fill overlying native cohesive glacial till deposits.

Groundwater was measured in the Pinchin (File No. 244512) Monitoring Well BH1 at a depth of 5.1 m (Elevation 353.6 m) on August 21, 2019. Saturated soil conditions were encountered in Pinchin (File No. 244512) Borehole BH7 at a depth of 7.6 m (Elevation 344.6 m). Free groundwater was reportedly not encountered in the remaining Pinchin and Englobe boreholes advanced along Snyder's Road West.

4.3 Foundry Street

MTE Monitoring Well MW105-21 was advanced off the west side of Foundry Street at the south side of Baden Creek and fill overlying native glacial till and silt deposits was encountered.

Groundwater was measured in MTE Monitoring Well MW105-21 at a depth of 2.7 m (Elevation 345.7 m) on November 16, 2021.

4.4 Schmidt Lands

4.4.1 Northwest Field Area

MTE Monitoring Wells MW101-21 to MW103-21 were advanced in the northwest field area of the Schmidt Lands. Topsoil overlying native glacial till deposits was encountered in MTE Monitoring Wells MW101-21 and MW103-21. The glacial till in MTE Monitoring Well MW103-21 was intercepted by sand deposits at depths of 4.0 m and 7.6 m (Elevation 347.7 m and 344.1 m). Topsoil overlying native silt and sand deposits was encountered in MTE Monitoring Well MW102-21.

Groundwater was measured in the MTE monitoring wells at depths of 1.0 to 6.0 m (Elevation 344.5 to 349.3 m) on November 16, 2021.

4.4.2 Development Portion – North of Baden Creek

MTE Monitoring Well MW105-21, Pinchin (File No. 249839) Monitoring Wells BH10, BH12, and BH13, and Pinchin (File No. 249839) Boreholes BH9 and BH11 were advanced north of Baden Creek in the portion of the Schmidt Lands currently proposed to be developed.

Topsoil overlying cohesive native silt deposits were predominately encountered within the Pinchin (File No. 249839) boreholes and monitoring wells. Fill overlying native glacial till and silt deposits was encountered in MTE Monitoring Well MW105-21.

Groundwater was measured in Pinchin (File No. 249839) Monitoring Wells BH10, BH12, and BH13 at depths of 0.3 to 5.3 m (Elevation 341.6 to 345.0 m) on January 7, 2020.

Groundwater was measured in MTE Geotechnical Monitoring Well MW105-21 at a depth of 2.7 m (Elevation 345.7 m) on November 16, 2021.

4.4.3 Development Portion – South of Baden Creek

MTE Monitoring Well MW104-21, Pinchin (File No. 249839) Monitoring Wells BH2, BH4, and BH8, and Pinchin (File No. 249839) Boreholes BH1, BH3, and BH5 to BH7 were advanced south of Baden Creek in the portion of the Schmidt Lands currently proposed to be developed.

Topsoil overlying cohesive native silt deposits were predominately encountered within the Pinchin (File No. 249839) boreholes and monitoring wells and MTE Geotechnical Monitoring Well MW104-21.

Groundwater was measured in Pinchin (File No. 249839) Monitoring Wells BH2, BH4, and BH8 at depths of 0.5 to 2.0 m (Elevation 342.9 to 346.0 m) on January 7, 2020.

Groundwater was measured in MTE Monitoring Well MW104-21 at 0.7 m above the ground surface (Elevation 340.4 m) on November 16, 2021. It is noted that surface water was also observed around MTE Monitoring Well MW104-21 on November 16, 2021.

5.0 Limitations of Report

Services performed by MTE Consultants Inc. (MTE) were conducted in a manner consistent with the level of care and skill ordinarily exercised by members of the Geotechnical Engineering & Consulting profession practicing under similar conditions in the same geographic area were the services are provided. No other warranty or representation expressed or implied as to the accuracy of the information, conclusions or recommendations is included or intended in this report.

This report was completed for the sole use of the Client. This report is not intended to be exhaustive in scope or to imply a risk-free site. As such, this report may not deal with all issues potentially applicable to the site and may omit aspects which are or may be of interest to the reader.

In addition, it should be recognized that a soil sample result represents one distinct portion of a site at the time it is collected, and that the findings of this report are based on conditions as they existed during the time period of the investigation. The material in the report reflects our best judgment using the information available at the time the report was written. The soil and groundwater conditions between and beyond the test holes may differ from those encountered in the test holes. Should subsurface conditions arise that are different from those in the test holes MTE should be notified to determine whether or not changes should be made as a result of these conditions.

It should be recognized that the passage of time may affect the views, conclusions and recommendations (if any) provided in this report because groundwater conditions of a property can change, along with regulatory requirements. All design details were not known at the time of submission of this report and it is recommended MTE should be retained to review the final design documents prior to construction to confirm they are consistent with our report recommendations. Should additional or new information become available, MTE recommends that it be brought to our attention in order that we may determine whether it affects the contents of this report.

Any use which another party makes of this report, or any reliance on, or decisions to be made based upon it, are the responsibility of such parties. MTE accepts no responsibility for liabilities incurred by or damages, if any, suffered by another party as a result of decisions made or actions taken, based upon this report. Others with interest in the site should undertake their own investigations and studies to determine how or if the condition affects them or their plans. The contractors bidding on this project or undertaking the construction should make their own interpretation of the factual information and draw their own conclusions as to how subsurface conditions may affect their work.

The benchmark and elevations provided in this report are primarily established to identify differences between the test hole locations and should not be used for other purposes such as, planning, development, grading, and excavation.

All of which is respectfully submitted, **MTE Consultants Inc.**



Ben Heinbuch, P.Eng. Geotechnical Engineer 519-271-7952 ext. 2333 bheinbuch@mte85.com



Manager, Geotechnical 519-271-7952 ext. 2343 dgonser@mte85.com

DMG: bgh

M:\48730\100\Reports\MTE Reports\Geotechnical Investigation\Report\48730-100_Geotechnical Site Characterization Report_2022-11-08_Final.docx



Figures











Borehole Logs



ID No.: MW101-21

Project Name: Baden Sanitary Trunk EA

MTE File No.: 48730-100

Client: Township of Wilmot

Site Location: Baden, ON

Date Completed: 8/12/2021

Drilling Contractor: London Soil Test Ltd.

Drill Rig: D50T Track Mounted

Drill Method: Hollow Stem Augers

Protective Cover: Monument Casing

| Subsurface Profile | | | Sa | mple | | | | | | |
|---|--------|--|-------------------------------|--------|------|---|---|---------------------|----------------------------------|---|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | | Dynamic Cone × × Standard Penetration 20 40 60 80 | Shear Strength (PP) | Water Content % • 10 20 30 | Groundwater Observations and Standpipe Details |
| ft m | | Ground Surface | 347.4 | | | t | | | | |
| | | TOPSOIL dark brown sandy silt, some organics, very moist (610mm) SILT TILL | 0.0 346.8 0.6 | 1 | ss | | 7 | | 1 5 | Concrete |
| 4 4 | | compact to dense mottled dark/light brown to brown silt, trace to some clay and sand, trace gravel, very moist | 345.9 | 2 | SS | | 13 | | 18 | /C Riser |
| 6 1 1 1 1 1 1 1 2 | | occasional wet seams | 1.5 | 3 | SS | - | 14 | | 1 9 | Bentonite |
| 8 | | | | 4 | SS | _ | 20 | | 1 6 | |
| 10 | | | 344.4 | | | | | | 16 | |
| 12 - | | 150mm saturated silty sand seam | 343.6 | 5 | SS | - | | | 22 | Generation (|
| 14 | | CLAYEY SILT TILL very stiff to hard grey clayey silt, some sand, trace gravel, DTPL to APL | 3.8 | 6 | ss | | 24 | 150 | 1 3 | and Pack |
| 16 | | | 342.4 | 7 | SS | _ | 22 | >200 | •14 | S 51mm S |
| 18 20 | | Drilling Terminated | | | | | | | | |

Field Technician: B. Heinbuch

Drafted by: B. Heinbuch

Reviewed by: D. Gonser



Notes:

Water encountered at 3.3mbgs (Elevation 344.1masl) during drilling. Water measured at 1.0mbgs (Elevation 346.4masl) on November 16, 2021.

ID No.: MW102-21

Project Name: Baden Sanitary Trunk EA

MTE File No.: 48730-100

Client: Township of Wilmot

Site Location: Baden, ON

Date Completed: 9/30/2021

Drilling Contractor: Envirocore

Drill Rig: D50T Track Mounted

Drill Method: Hollow Stem Augers

Protective Cover: Monument Casing

| | Subsurface Profile | | | | | | | | | |
|-------------|--------------------|--|-------------------------------|--------|----------|---|---------------------|------------------------------------|---|--|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Dynamic Cone × × Standard Penetration 20 40 60 80 | Shear Strength (PP) | Water Content • % • 10 20 30 | Groundwater Observations and Standpipe Details | |
| ft m | | Ground Surface | 350.5 | | | | | | | |
| 2 | | TOPSOIL dark brown sandy silt, trace organics, very moist (330mm) SILT loose mottled grey/brown to brown silt, | 0.0 <u>350.2</u> 0.3 | 1 | SS SS | 9 | | _11 15 | Concrete | |
| 4 | | trace to some clay, trace sand, very moist | | _ | | | | | | |
| | | | 349.0 | | | | | | | |
| 6 1 2 | | wet/dilatant | 1.5 | 3 | SS | 4 | | 1 9 | C Riser | |
| 。 + - | | saturated | 2.3 | | | 1 | | 21 | | |
| | | | | 4 | SS | | | | Bentonii 51mm | |
| | | | | 5 | SS | 4 | | _20 | | |
| 12 | | | 346 7 | | | | | | | |
| 14 4 | | SILT AND SAND very loose to loose brown silt and sand, saturated | 3.8 | 6 | SS | 3 | | 1 5 | | |
| 16 16 | | | | 7 | SS | 6 | | _ 16 | | |
| | | | | | | | | | | |
| 18 | | | 344.9 | 8 | SS | 2 | | 13 | d Sc | |
| | | SAND | 5.6 | | | | | •'` | otte | |
| 20 6 | 111 | saturated | 344.4 | | | | | | ມ 🔁 ເບັ | |
| | | SILT AND SAND compact brown silt and sand, some | 0.1 | 9 | SS | \ 29 | | 9 | 51mr | |
| 22 | | gravel, occasional cobbles, very moist to wet | | | | | | | | |
| 24 | | | | | | | | | entonite | |
| 26 - | | | 0.40 6 | 10 | SS | 27 | | -7 | ă | |
| | | Drilling Terminated | 342.4 8.1 | | | | | | | |
| | | | | | | | | | | |
| Field | Tec | hnician: B. Heinbuch | | | | | Notes: | | | |

Field Technician: B. Heinbuch

Drafted by: B. Heinbuch





Water encountered at 2.3mbgs (Elevation 348.2masl) during drilling. Water measured at 6.0mbgs (Elevation 344.5masl) on November 16, 2021.

ID No.: MW103-21

Project Name: Baden Sanitary Trunk EA

MTE File No.: 48730-100

Client: Township of Wilmot

Site Location: Baden, ON

Date Completed: 9/30/2021

Drilling Contractor: Envirocore

Drill Rig: D50T Track Mounted

Drill Method: Hollow Stem Augers

Protective Cover: Monument Casing

| Image: Strength (PP) Image: Strength (PP) Git | | |
|--|---|--|
| Image: Soil Description | Groundwater Observations and Standpipe Details | |
| ft m Ground Surface 351.7 | | |
| 2 TOPSOIL 0.0 1 SS 7 2 350.9 10 350.9 10 | | |
| 4 loose mottled dark/light brown to brown silt, some clay, trace to some sand, trace gravel, very moist compact to dense | | |
| | n PVC Rise | |
| 4 SS 22 10 348.7 30 | -1 51mi | |
| 12 12 12 12 12 12 12 12 12 12 | | |
| 14 SAND 4.0 brown sand, trace silt and clay, saturated 347.1 | | |
| $16 \frac{1}{18} \frac{1}{18$ | tted Screen | |
| CLAYEY SILT TILL 345.9 6 CLAYEY SILT TILL 6 5.8 10 5.8 10 5.8 10 5.8 10 5.8 11 >200 12 >200 | 51mm Slo | |
| | | |
| 24 344.1 26 8 SAND 7.6 99/250mm 9 | | |
| Impost 8.0 Drilling Terminated | | |

Field Technician: B. Heinbuch

Drafted by: B. Heinbuch

Reviewed by: D. Gonser



Notes: Water encountered at 4.0mbgs (Elevation 347.7masl) during drilling. Water measured at 2.4mbgs (Elevation 349.3masl) on November 16, 2021.

ID No.: MW104-21

Project Name: Baden Sanitary Trunk EA

MTE File No.: 48730-100

Client: Township of Wilmot

Site Location: Baden, ON

Date Completed: 10/14/2021

Drilling Contractor: London Soil Test Ltd.

Drill Rig: D50T Track Mounted

Drill Method: Hollow Stem Augers

Protective Cover: Monument Casing

| Subsurface Profile | | | | Sai | mple | | | | | |
|--------------------|-----------------|--|-------------------------------|--------|------|---|---------------------|----------------------------------|---|--|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Dynamic Cone × × Standard Penetration 20 40 60 80 | Shear Strength (PP) | Water Content • % 10 20 30 | Groundwater Observations and Standpipe Details | |
| ft m | | Ground Surface | 220.7 | | | | | | H | |
| | | TOPSOIL dark brown sandy silt, trace to some gravel, wet (360mm) | 0.0 339.3 0.4 | 1 | ss | 4 | | 2 2 | ncrete | |
| 4 | ···· ···· | SANDY SILT loose light brown sandy silt, wet | 0.8 | 2 | ss | 10 | | 26 | Co Riser | |
| 6 | 111 | CLAYEY SILT very stiff brown clayey silt, DTPL | <u>338.2</u> 1.5 | 3 | SS | 12 | 150 | 1 6 | Bentonite ⁻ 51mm PVC | |
| 8 | <u>e .</u> 1 | stiff, grey, occasional sand seams, WTPL | 337.4 2.3 | 4 | SS | 8 | 15 | 2 3 | | |
| 10 | art art | | | 5 | ss | 7 | 75 | 23 | × | |
| 12 | | soft to firm | <u>335.9</u> 3.8 | 6 | ss | 4 | 25 | 2 5 | Screen | |
| 16 | | CLAY AND SILT firm to stiff grey clay and silt, numerous sand seams, WTPL | <u>335.1</u> 4.6 | 7 | SS | 8 | 50 | 2 4 | Sand Pack ⁻ | |
| 18 | e el el | | | | | | | | 2 | |
| 20 - 6 | ar ar | | 333.0 | 8 | ss | 8 | | _23 | | |
| 24 24 268 | | Drilling Terminated | 6.7 | | | | | | | |

Field Technician: M. Dalgliesh

Drafted by: B. Heinbuch





Notes: Water encountered at 2.3mbgs (Elevation 337.4masl) during drilling. Water measured at 0.7mags (Elevation 34

Water measured at 0.7mags (Elevation 340.4masl) on November 16, 2021.

ID No.: MW105-21

Project Name: Baden Sanitary Trunk EA

MTE File No.: 48730-100

Client: Township of Wilmot

Site Location: Baden, ON

Date Completed: 10/14/2021

Drilling Contractor: London Soil Test Ltd.

Drill Rig: D50T Track Mounted

Drill Method: Hollow Stem Augers

Protective Cover: Monument Casing

| | Subsurface Profile | | | Sar | nple | | | | | | | | | | | | | | | | |
|---|--------------------|---|-------------------------------|--------|------|--------|---|----|---------------------|---|-------------|------------------------------------|----|--------------|---|-------------|------------|-----------|----------|----------|------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | D × | Dynamic Cone × × Standard Penetration 20 40 60 80 | | Shear Strength (PP) | | PP) FV) 00 | Water Content • % • 10 20 30 | | G O ar | Groundwater Observations and Standpipe Details | | | | | | |
| ft m | | Ground Surface | 348.4 | | | | | | | | | | | | | | | | | | |
| | | FILL (TOPSOIL) dark brown sandy silt, very moist (150mm) | 0.0 | 1 | SS | 4 | | | | | | | | | | | 24 | oncrete | | | |
| 4 | | FILL loose brown clayey silt, some sand, trace to some gravel, wet | 346.9 | 2 | SS | 5 | | | | | | | | | | | 2 4 | Ö | 4 | | |
| | | very loose brown silt and sand, saturated | 1.5 | 3 | SS | 3 | | | | | | 100 | | | | - 16 | 6 _27 | | - | - | Riser — |
| 8 | | stiff to very stiff brown/grey clayey silt, trace to some gravel and sand, WTPL/ SILT AND CLAY TILL very stiff to hard brown/grey silt and | <u>346.1</u> 2.3 | 4 | SS | | 11 | | | | | | | 75 | | • | 19 | Bentonite | | | 51mm PVC F |
| 10 ⁺ +++++++++++++++++++++++++++++++++++ | | clay, trace sand and gravel, DTPL to APL | | 5 | SS | | 16 | 3 | | | | | > | 200 | | | 24 | _ | - | - | |
| 14 14 | | | 344.0 | 6 | SS | | 2 | 1 | | | | | > | 200 | | | 2 4 | | | | |
| | | grey | 4.4 | | | | | | | | | | / | | | | | - | | | ¥ |
| 16 | | CLAY AND SILT TILL very stiff grey clay and silt, trace sand and gravel, APL with occasional saturated sand seams | | 7 | SS | | 2 | 23 | | | | | 25 | | | | _24 | - | | | |
| | | | | | | | | | | | | | | | | | | | • | | l Screen |
| 20 = 6 | | | <u>342.3</u> 6 1 | | | | | | | _ | | | | | | | | act | | | ottec |
| 22 | | stiff to very stiff grey clayey sandy silt, WTPL | 0.1 | 8 | SS | | 13 | | | | | 100 | | | | | 21 | Sand P | | | 51mm Slo |
| 24-1- | | | 340 8 | | | | | | | | | | | | | | | - | | | |
| | | SANDY SILT | 7.6 | | | | | | | | | | | | | | 00 | | <u> </u> | | T |
| 26 8 | | compact grey sandy silt, some clay, saturated | 340.2 | 9 | SS | | 14 | | | | | | | | | | 26 | | | | |
| | | Drilling Terminated | 0.2 | | | | | | | | | | | | | | | | | | |

Field Technician: M. Dalgliesh

Drafted by: B. Heinbuch





Notes:

Water encountered at 1.5, 4.6, and 6.1mbgs (Elevation 346.9, 343.8, and 342.3masl) during drilling. Water measured at 2.7mbgs (Elevation 345.7masl)

Water measured at 2.7mbgs (Elevation 345.7mas on November 16, 2021.

ID Number: MW101-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 5/3/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME75 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument



Field Technician: TXG

Drafted by: TXG

Reviewed by: ATD



Notes: Water level measured August 21, 2020.

ID Number: MW102-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 5/3/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME75 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument



Field Technician: TXG

Drafted by: TXG

Reviewed by: ATD



Notes: Water level measured August 21, 2020.

ID Number: MW103-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 5/2/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME75 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument



Field Technician: TXG

Drafted by: TXG

Reviewed by: ATD



Notes: Water level measured August 21, 2020.

ID Number: MW104-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 5/2/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME75 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument

| | Subsurface Profile | | | | | nple | | | SPT | Moisture | |
|-------------|--------------------|----------|--|-------------------------------|--------|------|--------------|---|---|--------------------------------------|-------------------------|
| Denth Scale | | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | • | N-Value Blows/305mm ● 0 40 60 80 | Moisture Content % 10 20 30 | Well Completion Details |
| π | m | | Ground Surface | 361.2 | | | | | | | |
| 0 | — 0 _ _ | | TOPSOIL dark brown, clayey silt, moist | 0.0 | 1 | ss | 75 | 6 | | 27 | |
| 2 4 1 | - | , | CLAYEY SILT brown, clayey, trace to some sand, moist | 0.8 | 2 | ss | 100 | | 19 | 1 4 | Concr |
| 6 | - - - 2 | | sandy, trace clay SILTY SAND brown trace gravel maint | 359.6 1.5 | 3 | ss | 90 | | 19 | 1 3 | PVC R |
| 8 | - | | brown, trace graver, moist | | 4 | ss | 75 | | 24 | _14 | 51 m Bei |
| 10 | - - - | | SILT brown, some sand, trace clay, trace gravel, damp | 358.1 3.0 | 5 | ss | 50 | | 23 | _15 | |
| 12 | _ 4 | | | | | | | | | | |
| 14 | - | | grey, clayey, wet | 356.6 4.6 | 6 | ss | 95 | | 19 | 20 | d Scree |
| 18 | - | | | | | | | | | | and Pa |
| 20 | - 6 | | | 355.1 | | | | | 22 | 21 | |
| 22 | - | | brown, saturated | | 7 | SS | 85 | | | | |
| 24 | - | | | 353.5 | | | | | | | - |
| 26 | - 8 - | | trace silt, moist | 7.6 352.9 | 8 | ss | 80 | | 21 | 6 | - |
| 28 | - | | Drilling Terminated | 8.2 | | | | | | | - |
| 30 | - | | | | | | | | | | - |
| 32 | - - 10 | | | | | | | | | | |
| 34 | - | | | | | | | | | | - |
| 38 | | | | | | | | | | | - |

Field Technician: TXG

Drafted by: TXG

Reviewed by: ATD



Notes: Well dry August 21, 2020.

ID Number: MW105-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 5/2/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME 850 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument



Reviewed by: ATD



ID Number: MW106A-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 4/29/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME 850 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument

| | Subsurface Profile | | | | | | SPT | Moisture | |
|----------------------------|--------------------|---|-------------------------------|--------|------|--------------|--|--------------------------------------|--|
| Depth Scale | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | N-Value ● Blows/305mm ● 20 40 60 80 | Moisture Content % 10 20 30 | Well Completion Details |
| ft m | | | | | | | | | |
| 0 <u> </u> 0 2 <u> </u> | | Ground Surface TOPSOIL dark brown, silty sand, moist | 360.7 0.0 359.9 | 1 | SS | 90 | • ⁵ | •14 | |
| 4 | ĥ | SAND | 0.8 359.2 | 2 | SS | 60 | | 20 | |
| 6 2 | | trace silt, moist | 1.5 358.4 | 3 | SS | 50 | | 19 | Ŭ 1 |
| 8 | | wet | 2.3 | 4 | SS | 60 | 12 | 20 | |
| 10 | | SILT | 3.0 | 5 | SS | 75 | 14 | 20 | |
| 12 = 4 | | brown, trace sand, varying degrees of clay content, \saturated | 356.9 | 6 | 88 | 75 | /11 | 20 | |
| | | trace gravel, moist | 356.1 | - | 00 | 10 | 12 | _24 | |
| 10 | | grey | | - / | 55 | 100 | | | |
| 20 6 | • | | 354.6 | | | | 32 | 13 | |
| 22 | | SAND brown fine-grained sand, trace silt, moist | 6.1 | 8 | SS | 50 | | •13 | B B B B B B B B B B B B B B B B B B B |
| 24 | | - | 353.1 | | | | | | |
| 26 8 | | SANDY SILT | 7.6 | 9 | SS | 100 | 26 | _25 | |
| 28 | | brown, trace clay, moist | 054.0 | | | | | | <u>3</u> <u></u> |
| 30 | | SAND | <u>351.6</u> 9.1 | 10 | 99 | 75 | 41 | 19 | 21 |
| ³² 10 | | brown, trace silt, saturated | | 10 | 00 | 15 | | | |
| 34 | | | 350.0 | | | | 13 | 16 | |
| 36 | | coarse-grained sand, saturated | 10.7 | 11 | SS | 90 | | ••• | |
| 38 ± 12 | | | 348.5 | | | | | 10 | |
| 40 | | medium-grained sand | 12.2 | 12 | SS | 65 | 32 | 19 | |
| 44 | | | | | | | | | |
| 46 14 | | | | 13 | SS | 75 | 37 | 1 7 | e e . |
| 48 | | | | | | | | | <u>.</u> |
| 50 | | | | 11 | 66 | 65 | 48 | _22 | |
| 52 16 | | | | 14 | 33 | 05 | I I I I I I I I I I I I I I I I I I I | | |
| 54 | | | | | | | 42 | 18 | ⊃ack |
| 56 | | | | 15 | SS | 65 | | •••• | Sol |
| 58 18 | | | | | | | | | Sa and a second se |
| 60 | | | 341.8 | 16 | SS | 50 | 48 | 1 6 | fo |
| 02 | | Drilling Terminated | 18.9 | | | | | | |

Field Technician: TXG

Drafted by: TXG

Reviewed by:



Notes: Water level measured August 21, 2020.

ID Number: MW106B-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 4/26/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME 850 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument

| | Subsurface Profile | | | | Sam | nple | | SPT | Moisture | |
|---|--------------------|--------|---|--|---|--|---|--|--|-------------------------|
| Donth Coolo | | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | N-Value • Blows/305mm • 20 40 60 80 | Moisture Content % 10 20 30 | Well Completion Details |
| ft 0 2 4 10 10 12 14 16 10 12 14 16 10 10 10 10 10 10 10 10 10 10 10 10 10 | m 0 2 4 | | Ground Surface TOPSOIL dark brown, silty sand, moist SAND brown, some silt, trace gravel, moist trace silt, moist wet saturated SILT brown, trace sand, varying degrees of clay content, saturated trace gravel, moist grey SAND brown fine-grained sand, trace silt, moist | 360.6 0.0 359.9 0.8 359.1 1.5 358.3 2.3 357.6 3.0 356.8 3.8 356.0 4.6 354.5 6.1 | 1 2 3 4 5 6 7 7 8 | SS SS SS SS SS SS SS SS | 90 60 50 60 75 75 100 50 | | 14 20 19 20 20 20 20 20 20 20 20 20 20 | Bentonite Concrete |
| 22 24 26 30 32 34 34 | 8 | | brown fine-grained sand, trace silt, moist SANDY SILT brown, trace clay, moist SAND brown, trace silt, saturated | 353.0 7.6 351.5 9.1 349.3 11.3 | 9 10 11 | SS SS SS | 100 75 90 | 41 | 19 16 | Sand Pack |
| 38_ | _ | | Drilling Terminated | 11.5 | | | | | | |

Field Technician: TXG

Drafted by: TXG

Reviewed by: ATD



Notes: Borehole lithology, N-values and moisture contents inferred from MW 106A-19.

Water level measured August 21, 2020.

ID Number: MW106C-19

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 4/26/2019

Drilling Contractor: Aardvark Drilling Inc.

Drill Rig: CME 850 Track

Drill Method: Hollow Stem Auger

Protective Cover: Monument

| | Subsurface Profile | | | | | | SPT | Moisture | |
|--|--------------------|--|--|--------|--|----------------------|--|--------------------------------------|--------------------------------|
| Depth Scale | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | N-Value ● Blows/305mm ● 20 40 60 80 | Moisture Content % 10 20 30 | Well Completion Details |
| $\begin{array}{c} 0 \\ 2 \\ 4 \\ 10 \\ 12 \\ 14 \\ 16 \\ 12 \\ 14 \\ 16 \\ 12 \\ 14 \\ 16 \\ 11 \\ 10 \\ 11 \\ 10 \\ 11 \\ 10 \\ 11 \\ 10 \\ 11 \\ 10 \\ 11 \\ 10$ | | Ground Surface TOPSOIL dark brown, silty sand, moist SAND brown, some silt, trace gravel, moist trace silt, moist wet saturated SILT brown, trace sand, varying degrees of clay content, saturated Drilling Terminated | 360.6 0.0 359.8 0.8 359.1 1.5 358.3 2.3 357.5 3.0 356.9 3.7 | | SS SS SS SS SS | 90 60 50 75 | | | Sand Pack Slotted Screen |

Field Technician: TXG

Drafted by: TXG

Reviewed by: ATD



Notes:

Borehole lithology, N-values and moisture contents inferred from MW106A-19. Water level measured August 21, 2020.

ID Number: MW201-20

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 6/10/2020

Drilling Contractor: London Soil Test Ltd.

Drill Rig: D-50 Turbo

Drill Method: Hollow Stem Auger

Protective Cover: Monument



Field Technician: KLW

Drafted by: KLW

Reviewed by: ATD



Water level measured August 21, 2020.

ID Number: MW202-20

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 6/10/2020

Drilling Contractor: London Soil Test Ltd.

Drill Rig: D-50 Turbo

Drill Method: Hollow Stem Auger

Protective Cover: Monument

| | | Subsurface Profile | 5 | Sam | ple | | SPT | Moisture | |
|--|--------|--|--|--------|----------------------|--------------------------------|--|--------------------------------------|------------------------------|
| Depth Scale | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | N-Value • Blows/305mm • 20 40 60 80 | Moisture Content % 10 20 30 | Well Completion Details |
| $ \begin{array}{c} \text{ft} & \text{m} \\ 0 & 2 & \text{m} \\ 2 & 4 & 6 & \text{m} \\ 10 & 12 & 14 & 16 & \text{m} \\ 10 & 12 & 14 & 16 & \text{m} \\ 10 & 12 & 14 & 16 & \text{m} \\ 10 & 12 & 14 & 16 & \text{m} \\ 20 & 22 & 24 & 26 & \text{m} \\ 22 & 24 & 26 & 28 & 30 & \text{m} \\ 33 & 34 & 36 & \text{m} \\ 34 & 36 & 38 & 40 & \text{m} \\ 42 & 42 & \text{m} \\ 42 & \text{m} \\ 12 &$ | | Ground Surface Drown silt topsoil, moist SILTIL brown silt till, some clay to clayey, minor gravel, mottled trace to some gravel brownish-grey, clayey, moist to wet very moist to wet | 357.7 0.0 356.2 1.5 355.4 2.3 353.1 4.6 | | SS SS SS SS | 100 100 85 100 100 | | | Sand Pack Bentonite Concrete |
| Field [·] | Tec | hnician: KLW | | | | | Notes | 5: | |

Field Technician: KLW

Drafted by: KLW

Reviewed by: ATD



Water level measured August 21, 2020.

ID Number: MW203-20

Project Name: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date Completed: 6/10/2020

Drilling Contractor: London Soil Test Ltd.

Drill Rig: D-50 Turbo

Drill Method: Hollow Stem Auger

Protective Cover: Monument



Field Technician: KLW

Drafted by: KLW

Reviewed by: ATD



Water level measured August 21, 2020.

ID Number: MP1-20

Project: 1012 Snyder's Rd. W. Development Hydrogeology Study

Project No: 40777-800

Client: Nideva Properties Inc.

Site Location: 1012 Snyder's Rd. W., Baden, ON

Date: 6/18/2020

Construction Materials: Stainless Steel

Installation Method: Drive Point

| | | Subsurface Profile | | | | |
|--|--------|---|--|-----------------------------|------------|--|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Soil Sample Lab Analysis | Piezometer | |
| ft m 0 0 - - - - - - - - - - - - - | | FAT drk forwinsh-black muck/peat, saturated SIT ILL The saturated is a satur | 359.6 0.0 359.5 0.1 358.5 1.1 | | | |
| Field Technician: KLW Notes: Drafted by: KLW Notes: | | | | | | |

Reviewed by: ATD



ID Number: MW301-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| Subsurface Profile | | Sample | | | | e | Headspace | | |
|--|--------|---|-------------------------------|--------|------|--------------|--|--|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| $\int_{0}^{\text{ft}} m$ | | Ground Surface | 360.5 | | | | | | |
| 2 1 4 1 4 | 7 | SAND Medium brown sand with trace gravel, moist. CLAYEY SILT TILL Brown clayey silt till, moist. | 359.2 1.2 | 1 | DP | 75 | Metals, As, Sb, Se, OCPs, PCBs, PAHs, PHCs F1-F4, VOCs | β | |
| | X | Till becomes wet. | 357.7 2.7 357.4 | 2 | DP | 100 | | | Bentonite |
| | | SAND Medium brown fine sand, damp. | 3.0 | 3 | DP | 95 | | | |
| 16 16 18 18 20 20 | | | | 4 | DP | 95 | | ρ | |
| Field Technician: JMS Image: Second seco | | | | | | | | | |

Reviewed by: RMR



ID Number: MW301-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| Subsurface Profile | | Sample | | | | е | Headspace | | |
|---|--------|---------------------|-------------------------------|--------|------|--------------|-----------------------------|---|---|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| 22 24 24 24 | | Sand becomes wet. | <u>353.1</u> 7.3 | 5 | DP | 100 | PHCs F1-F4, VOCs | ρ | measured Oct 15 2020.111111 measured Oct 15 2020.111111 |
| 26 26 11 28 11 28 11 11 11 11 11 11 11 11 11 1 | | | 351.3 | 6 | DP | 100 | | | Sand Pack |
| 30 31 32 32 34 10 34 10 34 10 34 11 12 40 | | Drilling Terminated | 9.1 | | | | | | |
| Field Technician: JMS | | | | | | | | | |

Drafted by: JMS

Reviewed by: RMR



otherwise noted.
ID Number: MW302-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 11/9/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| | Subsurface Profile | | | | | mpl | e | Headspace | |
|---|--------------------|---|-------------------------------|--------|------|--------------|-----------------------------|--|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| 0 ft m 0 - 0 | | Ground Surface | 358.6 0.0 | | | | | 0 | |
| 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | Y THINK | Dark brown silty topsoil with trace organics and some gravel, damp. SILTY CLAY TILL Medium brown silty clay till, trace gravel, damp. | <u>358.3</u> 0.3 | 1 | DP | 100 | | 0 | Concrete |
| 6 1 1 1 10 10 | A) A) | | | 2 | DP | 100 | | | Bentonite |
| 12 12 14 14 14 | X | | | 3 | DP | 100 | | | 2020. |
| 16 16 18 18 18 18 18 18 18 11 10 1 6 20 1 | | Till becomes grey, wet. | <u>352.7</u> 5.9 | 4 | DP | 100 | | | Sand Pack- |
| | | huisiana IMO | | | | | Na | a davin an ataini | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless

ID Number: MW302-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 11/9/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| | | Subsurface Profile | | | Sa | ampl | е | Headspace | |
|--|--------|---------------------|-------------------------------|--------|------|--------------|---|--|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| 22 24 24 | | Drilling Terminated | <u>351.0</u> 7.6 | 5 | DP | 100 | Metals, As, Sb, Se, PHC F1-F4, VOCs | ρ | |
| 26 1 8 28 1 1 1 28 30 1 1 1 1 1 28 30 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | | | | | | | | | |
| 40 1 2 | | | | | | | | | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted.

ID Number: BH303-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/19/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: None

| | Subsurface Profile | | | | | ampl | e | Headspace | |
|--|--------------------|--|--|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| $ \begin{array}{c} m_{0} \\ m_{1} \\ m_{2} \\ m_{1} $ | | Ground Surface TOPSOIL Dark brown topsoil with trace organics, moist. SAND Medium brown coarse sand moist. Sand turns grey and is saturated. Drilling Terminated | 360.6 0.0 359.6 0.9 358.1 2.4 357.5 3.0 | 2 | DP | 100 | See Notes | ρ | Bentonite |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted. Soil sample lab analysis at 1-2': Metals, As, Sb, Se, OCPs, PCBs,

ID Number: MW304-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| | Subsurface Profile | | | | Sa | mpl | e | Headspace | |
|--|--------------------|---|-------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| $\int_{0}^{\text{ft m}} 0$ | | Ground Surface | 360.7 | | | | | | |
| ° 111111111111111111111111111111111111 | ĨX X | TOPSOIL Dark brown silty topsoil with trace organics, moist. CLAYEY SILT TILL Medium brown clayey silt till, moist. | 0.0 | 1 | DP | 75 | PHCs F1-F4, BTEX | ρ | Concrete |
| 4 6 8 10 | X | SAND Medium brown fine sand, moist. | <u>358.3</u> 2.4 | 2 | DP | 100 | | | Bentonite |
| 12 14 14 14 14 | | Sand becomes slightly silty wet | <u>356.2</u> 4.6 | 3 | DP | 100 | PHCs F1-F4, VOCs, PAHs | ρ | ofted Screen |
| 16 16 18 18 18 18 18 10 10 10 10 10 10 10 10 10 10 10 10 10 | | Medium brown coarse sand, moist. | <u>355.0</u> 5.8 | 4 | DP | 75 | | | Sand Pack |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted. Water level measured Oct. 15, 2020.

ID Number: MW304-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| | | Subsurface Profile | | | Sa | impl | е | Headspace | |
|--|--------|---------------------|-------------------------------|--------|------|--------------|-----------------------------|--|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| 22 | | | 353.1 | 5 | DP | 100 | | | Cave |
| 28 28 30 32 30 32 31 32 10 34 10 34 11 12 40 | | Drilling Terminated | 7.6 | | | | | | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted. Water level measured Oct. 15, 2020.

ID Number: BH305A-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: None

| | Subsurface Profile | | | | | ampl | е | Headspace | | | |
|--|--|--|-------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|--|--|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details | | |
| 0 ft m 0 0 0 | | Ground Surface | 360.6 | | | | | | F | | |
| | | TOPSOIL Dark brown silty topsoil with trace organics, moist. SAND AND GRAVEL Medium brown sand and gravel, damp. CLAY TILL Dark grey clay till moist | 360.3 0.3 359.2 1.4 | 1 | DP | 100 | PHCs F1-F4, BTEX | ρ | tonite | | |
| | | Slight hydrocarbon-like odour noted, clay till becomes wet to saturated. | 358.7 1.8 357.5 | 2 | DP | 100 | PHCs F1-F4, BTEX | 0 0 | Ber | | |
| $ \begin{array}{c} 10 \\ 12 \\ 12 \\ 14 \\ 14 \\ 14 \\ 16 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 16 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18$ | | Drilling Terminated | 3.0 | | | | | | | | |
| Field | Field Technician: JMS No odour or staining unless otherwise noted. | | | | | | | | | | |

Drafted by: JMS

Reviewed by: RMR



ID Number: MW305B-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 11/9/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument

| | | Subsurface Profile | | | Sa | mpl | е | Headspace | |
|--|--------|---|-------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| $0 \frac{\text{ft}}{1} 0$ | A | Ground Surface | 360.6 | | | | | 0 | |
| | | Dark brown silty topsoil with trace organics and some gravel, damp. SAND AND GRAVEL Medium brown sand and gravel, moist. | 360.3 0.3 359.2 1.4 | 1 | DP | 50 | PHCs F1-F4 | ρ | Bentonite Concrete |
| | X | Medium brown silty clay till, moist. | 358.4 2.1 357.5 | 2 | DP | 75 | BTEX | P | Sand Pack |
| 10 11 12 12 14 14 14 14 | | Till becomes brown-grey, damp. | 3.0 | 3 | DP | 90 | | | |
| 16 16 18 18 18 18 18 18 18 18 10 10 10 10 10 10 10 10 10 10 10 10 10 | | SAND Grey coarse sand, wet. | 355.1 5.5 354.5 6.1 | 4 | DP | 90 | | | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted. Water level measured Feb. 2, 2021.

ID Number: BH306-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: None

| | Subsurface Profile | | | | Sa | ampl | е | Headspace | |
|--|--------------------|---|-------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 400 100 200 400 400 | Well Completion Details |
| $0 \frac{\text{ft m}}{1} 0$ | XXXX | Ground Surface | 360.7 0.0 | | | | | | 88 |
| | | Sandy topsoil mixed with gravel fill, concrete, and demolition debris. | 360.3 | | | | See Notes, pH | o | |
| | | CLAY Medium brown clay, moist. | 0.5 | 1 | DP | 100 | See Notes | ρ | |
| | | | | | | | | | tonite - |
| 6 6 8 8 8 1 1 1 1 1 1 1 1 1 1 1 1 1 | | | 358.0 | 2 | DP | 100 | | | Bent |
| | | SAND Medium brown fine sand, moist. | 2.7 357.7 | | | | | o | |
| 10 12 12 14 14 14 14 14 14 14 14 14 14 | | Drilling Terminated | 3.0 | | | | | | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted. Soil sample lab analysis for 0.5-1.5' & 2-3': Metals, As, Sb, Se, PAHs, PHCs F1-F4, BTEX

ID Number: BH307-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: None

| | Subsurface Profile | | | | | ampl | е | Headspace | |
|--|--------------------|---|-------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| 0 <u>ft</u> m | ww | Ground Surface | 360.7 | | | | | 0 | |
| | | FILL Silty topsoil mixed with gravel fill and demolition debris, trace organics. CLAY Medium brown clay, moist. | 360.4 0.3 | 1 | DP | 100 | See notes | Υ | ntonite |
| 6 6 8 8 8 | | SAND Medium brown fine sand, moist. | 358.3 2.4 357.7 | 2 | DP | 100 | рН | ρ | Ber |
| $ \begin{array}{c} 10 \\ 10 \\ 11 \\ 12 \\ 14 \\ 14 \\ 14 \\ 16 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 18 \\ 16 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18 \\ 18$ | | Drilling Terminated | 3.0 | | | | | | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted. Soil analysis at 0-1': Metals, As, Sb, Se, PHCs F1-F4, PAHs, VOCs, pH

ID Number: BH308-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/15/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: None

| | Subsurface Profile | | | | Sa | mpl | е | Headspace | |
|--|--------------------|---|--|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| $\begin{array}{c} 0 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\$ | | Ground Surface TOPSOIL Dark brown topsoil with trace organics, moist. SAND Medium brown coarse sand, moist. Sand becomes wet. Drilling Terminated | 360.2 0.0 359.9 0.3 359.0 1.2 358.7 1.5 | 1 | DP | 100 | PCBs, PHCs F1-F4, BTEX | ρ | Bentonite |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted.

ID Number: MW309-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/19/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push, Hollow Stem Auger

Protective Cover: Monument

| | Subsurface Profile | | | | Sa | mpl | e | Headspace | |
|--|--------------------|--|-------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details |
| $\int_{0}^{\text{ft m}} 0$ | | Ground Surface | 362.6 | | | | | | |
| ° 111111111111111111111111111111111111 | 2222 | TOPSOIL Dark brown silty topsoil with trace organics, most. SAND Medium brown coarse sand, damp. | 0.0 <u>361.7</u> 0.9 | 1 | DP | 50 | PHCs F1-F4, BTEX, PAHs | ρ | Concrete |
| 10 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | | | | 2 | DP | 90 | | | Bentonite |
| 12 14 14 14 14 14 14 14 14 14 14 14 14 14 | | SILTY SAND | <u>358.2</u> 4.4 | 3 | DP | 100 | | | 32mm PV |
| 16 16 18 18 18 18 10 10 10 10 10 10 10 10 10 10 10 10 10 | | SAND Medium brown coarse sand, damp. | | 4 | DP | 100 | | | |

Field Technician: JMS

Drafted by: JMS

Reviewed by: RMR



No odour or staining unless otherwise noted.

ID Number: MW309-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/19/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push, Hollow Stem Auger

Protective Cover: Monument

| | | Subsurface Profile | | | Sa | mpl | е | Headspace | | | |
|---|---|-------------------------|--------------------------------|--------|------|--------------|-----------------------------|---|-------------------------|--|--|
| Depth | Symbol | Soil Description | Elevation (masl) Depth (m) | Number | Type | Recovery (%) | Soil Sample Lab Analysis | PID ppm 20 40 60 80 Hydrocarbon ppm 100 200 300 400 | Well Completion Details | | |
| 22 24 24 | | | | 5 | DP | 100 | | ρ | | | |
| 26 1 28 1 28 1 1 1 1 1 1 1 1 1 1 1 1 1 | | | | | | | | | 19, 2020 | | |
| 30 + | | | | 6 | HSA | 75 | | | and Pack | | |
| 36 1 36 1 37 38 1 38 1 1 1 1 2 | | Sand becomes saturated. | 351.9 10.7 351.0 11.6 | 7 | HSA | 75 | PHCs F1-F4, BTEX, PAHs | ρ | Se Se | | |
| 40 – [–] [–] [–] – – – – – – – – – – | Field Technician: JMS No odour or staining unless | | | | | | | | | | |

Drafted by: JMS

Reviewed by: RMR



otherwise noted.

ID Number: MW310-20

Project: Phase II ESA

Project No: 40777-200

Client: Snyders Road (Baden) Developments Inc.

Site Location: 1012 Snyders Road West, Baden, ON

Drill Date: 10/19/2020

Drilling Contractor: Direct Environmental Drilling Ltd

Drill Rig: Geoprobe

Drill Method: Direct Push

Protective Cover: Monument





Ground Elevation: 359.08 m

4805588.6 m

526255.1 m

Northing:

Easting:

Borehole Number: BH-01-21

D.Brenner

2100862.100

2021-02-23

Drill Date:

Field Tech:

Job N°:

Can Classe c

Project: Snyder's Road Reconstruction

| Loca | ation: Cristian Street to Ging | erich | Road, | Bader | n, Ontario | | | | | | Dr | | etho | d: | Geo-Sieeve Sampler |
|------------------------|--|--------|--------------------------------|-----------------|-----------------|--|---|---|----|------------|---------------|------------|------|----|---|
| | SOIL PROFILE | | | | SAMPLE | | _ | | | | | | | | |
| Depth (m) | Description Ground Elevation | Symbol | 送 Elevation (m) G Depth (m) | Type and Number | Headspace (ppm) | Environmental Testing | 1 | 0 | 20 | WP Wate | er Con (%) | WL tent | 60 | 70 | Groundwater Observations and Standpipe Details |
| - | PAVEMENT STRUCTURE: Asphaltic Concrete: 200 mm Granular Fill: 560 mm, compact brown sand and gravel, moist, odour or staining | | 0.00 358.32 | SS-1 | 0/0 | - | | | | | | | | | asphaltic concrete |
| 1- 1- - - | <i>SILT TILL:</i> brown silt, some clay, trace sand, moist | | 0.76 | SS-2 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH | | | | | | | | | |
| 2- | very stiff silt and clay | | _357 <u>.10</u> 1.98 | SS-3 | 0/0 | - | | | | | | | | | |
| - - - - - | | | | SS-4 | 0/0 | - | | | | | | | | | |
| - | | | | SS-5 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH, PAHs | _ | | | | | | | | |
| 4- - - - - | | | | SS-6 | 0/0 | - | | | | | | | | | bentonite seal |
| - - 5- - - | | 8 | | SS-7 | 0/0 | - | | | | | | | | | |
| - - 6- | grey, wet | | _353.29_ 5.79 | SS-8 | 0/0 | - | | | | | | | | | |
| - | | | | SS-9 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH | _ | | | | | | | | |
| 7- | | | 351.46 | SS-10 | 0/0 | - | | | | | | | | | |
| 8- | Borehole terminated at 7.62 m | | 7.62 | | | | | | | | | | | | - |
| | | | | | D. 6 | | | | | | | | | | |
| Note | iewed by: D.Brenner | as par | ts per n | nillion (p | ppm). | E.Clochon | | | | | | | | | Sneet: 1 of 1 |



Ground Elevation: 358.37 m

4805629.6 m

526462.7 m

Northing:

Easting:

Borehole Number: BH-02-21

2100862.100

D.Brenner

2021-02-23

Drill Date:

Job N°:

Field Tech:

Drill Method:

Geo-Sleeve Sampler

Project: Snyder's Road Reconstruction

Cristian Street to Gingerich Road Baden Ontario

| LOC | ation. Cristian Street to Ginge | ricn | Road, | ваае | n, Untario | | | | | | | | we | mot | | |
|----------------------------------|---|--------|----------------------------|-----------------|-----------------|--|---|-----|----|----------|--------------|-------------------|----------------------|-----|----|---|
| | SOIL PROFILE | | | | SAMPLE | | - | | | | | | | | | |
| Depth (m) | Description Ground Elevation | Symbol | Elevation (m) Depth (m) | Type and Number | Headspace (ppm) | Environmental Testing | 1 |) : | 20 | WP Wa | iter C (% | ۷ conter ون | - VL nt 50 | 60 | 70 | Groundwater Observations and Standpipe Details |
| | PAVEMENT STRUCTURE: Asphaltic Concrete: 200 mm Granular Fill: 560 mm, comopact brown sand and gravel, trace silt, moist | | 0.00 | SS-1 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH | - | | | | | | | | | asphaltic concrete |
| - 1- - - - - | <i>SILT TILL:</i> stiff brown silt, trace sand, trace to some clay, moist | 0 | 0.91 | SS-2 | 0/0 | - | | | | | | | | | | |
| 2- | some clay | | _3 <u>56.69</u> 1.68 | SS-3 | 10/10 | PHCs, BTEX, Metals, SAR, EC, pH, PAHs | | | | | | | | | | |
| - - - - - - | | 0 0 0 | | SS-4 | 0/0 | - | | | | | | | | | | _ bentonite seal |
| | grey silt and clay | 0 | 354.86 3.50 | SS-5 | 0/0 | - | - | | | | | | | | | |
| 4 - - - - | | 0 0 | 353 65 | SS-6 | 0/0 | - | - | | | | | | | | | |
| 5- | wet | | 4.72 | SS-7 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH | | | | | | | | | | |
| - - - - 6- - | Parabala terminated at 6.10 m | 8 | 352.27 | SS-8 | 0/0 | - | | | | | | | | | | |
| | | | 0.10 | | | | | | | | | | | | | |
| 7- - - - - - - | | | | | | | | | | | | | | | | |
| 8- | | | | | | | | | | | | | | | | |
| Rev | iewed by: <i>D.Brenner</i> | | | | Drafted bv: | E.Ciochon | | | | | | | | | | Sheet: 1 of 1 |
| Note | - es: Headspace readings expressed a | as pa | rts per r | nillion (j | opm). | | | | | | | | | | | |

Vertical Scale = 1 : 50.0



Ground Elevation: 354.00 m

4805672 .0 m

526662.8 m

Northing:

Easting:

Borehole Number: BH-03-21

0400000 4

D.Brenner

2100862.100

2021-02-23

Drill Date:

Job N°:

Field Tech:

Drill Method:

Geo-Sleeve Sampler

Project: Snyder's Road Reconstruction

Location: Cristian Street to Gingerich Road, Baden, Ontario

| | energe | | , | | | | | |
|--|---|--------|--------------------------------|-----------------|-----------------|--|---|---|
| | SOIL PROFILE | 1 | 1 | | SAMPLE | | - | |
| Depth (m) | Description Ground Elevation | Symbol | 뜻 Elevation (m) S Depth (m) | Type and Number | Headspace (ppm) | Environmental Testing | WP WL Water Content (%) 10 20 30 40 50 60 70 | Groundwater Observations and Standpipe Details |
| | PAVEMENT STRUCTURE: | | 0.00 | | | | | asphaltic concrete |
| - - - - - | Asphaltic Concrete: 250 mm Granular Fill: 815 mm, compact brown sand and gravel, trace silt, moist | | | SS-1 | 0/0 | - | | |
| 1 | <i>SILT TILL:</i> brown silt, trace sand, trace to some clay, moist | 0 | 352.93 1.07 | SS-2 | 5/5 | PHCs, BTEX, Metals, SAR, EC, pH | | |
| 2- | | 0 0 0 | | SS-3 | 0/0 | - | | |
| | | | | SS-4 | 0/0 | - | | bentonite seal |
| | grey, wet | 0 | 350.34 3.66 | SS-5 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH, PAHs | | |
| 4- - - - - | | 000 | 349.43 | SS-6 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH | | |
| - | Borehole terminated at 4.57 m | | 4.57 | | | | | |
| 5_ | | | | | | | | |
| - | | | | | | | | |
| 7- - - - - | | | | | | | | |
| - 8- - - - - - - - - - - - - - - - - - | | | | | | | | |
| Rov | iewed by: D Brenner | L | | | Drafted by: | E Ciochon | | Sheet: 1 of 1 |
| Nev | iewea by. D.Dielillei | | | | Draileu Dy. | 2.010011011 | | |
| Note | es: Headspace readings expressed a | is pa | rts per n | nillion (| ppm). | | | |



Ground Elevation: 352.69 m

Northing: Easting:

4805681.2 m

526709.2 m

Borehole Number: BH-04-21

| 2 | 1000 | \$2 11 |
|---|------|--------|

D.Brenner

2100862.100

2021-02-23

Drill Date:

Job N°:

Field Tech:

~ ~ .

Project: Snyder's Road Reconstruction

| LOC | ation: Cristian Street to Ginge | ericn | Road, | ваде | n, Ontario | | 1 | | | | ietnoa | • | Geo-Sieeve Sampler |
|--|--|--------|-----------------------------------|-----------------|-----------------|--|----|----|------------------|---------------------|--------|----|---|
| | SOIL PROFILE | Τ_ | | | SAMPLE | | - | | | | | | |
| Depth (m) | Description Ground Elevation | Symbol | 555 Elevation (m) 89 Depth (m) | Type and Number | Headspace (ppm) | Environmental Testing | 10 | 20 | WP Water (| WL Content %) | 60 | 70 | Groundwater Observations and Standpipe Details |
| | PAVEMENT STRUCTURE: Asphaltic Concrete: 150 mm Granular Fill: 1070 mm, compact brown sand and | | 0.00 | SS-1 | 0/0 | - | | | | | | | asphaltic concrete |
| - - - - - - - - - - - - - | gravel, trace slit and clay, moist SILT TILL: stiff brown silt some clay, trace | | 351.47 1.22 | SS-2 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH | _ | | | | | | |
| 2- | sand, moist | • • • | | SS-3 | 0/0 | PHCs, BTEX, Metals, SAR, EC, pH, PAHs | | | | | | | |
| | | 0 | | SS-4 | 0/0 | - | | | | | | | bentonite seal |
| - - - - - - - - - - - - - - - - - - - | | 9 | 348.88 | SS-5 | 5/5 | PHCs, BTEX, Metals, SAR, EC, pH | | | | | | | |
| 4- - - - | wet | 0 | 3.81 | SS-6 | 0/0 | - | | | | | | | |
| 5- - - - - - - - - - - - - - - - - - - | Borehole terminated at 4.57 m | | 4.57 | | | | | | | | | | |
| 6 | | | | | | | | | | | | | |
| 7- | | | | | | | | | | | | | |
| 8- - - - - - - - - - - - - | | | | | | | | | | | | | - |
| Rev | iewed by: D.Brenner | - | | | Drafted by: | E.Ciochon | | | | | | | Sheet: 1 of 1 |
| Note | es: Headspace readings expressed | as pai | ts per n | nillion (j | ppm). | | | | | | | | |



Logged By: NW

Project: Geotechnical Investigation

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 15, 2019

Project #: 244512

Project Manager: VM

SAMPLE SUBSURFACE PROFILE Recovery (%) Elevation (m) Sample Type N-Value Monitoring Well Details Description Sampler # Standard Penetration Depth (m) Shear Strength Symbol Water Content N-Value SPT I Δ kPa • % 20 40 00 100 200 10 20 358.67 Ground Surface 0 Pavement Structure SS 1 79 28 ┍╋ Asphaltic Concrete (175 mm) Riser -Granular Fill (580 mm) - Brown 357.91 sand and gravel, trace silt, compact, 00 1 damp 2 7 SS 67 Fill 357.14 Bentonite⁻ Brown silt, some sand, trace gravel, loose, damp SS 3 79 10 2 Silt Till 356.38 Dark brown clayey silt, trace sand and gravel, very stiff, DTPL 100 SS 4 25 Brown 3 Hard SS 5 100 25 4 4 Sand⁻ 354.10 Silica 8 Grey/brown, firm, WTPL SS 6 100 9 5 Screen 352.57 6 DTPL SS 7 100 23 • Water 7 level = μ SS 8 100 26 351.20 5.12 mbgs, as End of Borehole measured on August Borehole terminated at 7.47 8 21, 2019 mbgs. At drilling completion dry cave observed at 6.71 mbgs. 9 Contractor: Altech Drilling and Investigative Services Inc. Grade Elevation: 358.67 masl

Drilling Method: Split Spoon/Solid Stem

Well Casing Size: 51 mm

Top of Casing Elevation: 358.60 masl



Project #: 244512

Project: Geotechnical Investigation

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 15, 2019

Logged By: NW

Project Manager: VM

| | | SUBSURFACE PROFILE | | | | | | | SAMPLE | | |
|-------------------------|--------|---|---------------|----------------------------|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content ● % ● 10 20 |
| 0- | | Ground Surface | 358.48 | T | | | | | | | |
| - | | Pavement Structure Asphaltic Concrete (150 mm) Granular Fill (610 mm) - Brown sand and gravel, trace silt, compact, | 357.72 | | SS | 1 | 71 | 28 | | | |
| 1 | | damp | | | SS | 2 | 63 | 9 | E. | A A | |
| | | Brown silt, some gravel, loose, damp | 356.96 | | SS | 3 | 100 | 15 | | A | |
| - | | Clayey silt, trace sand and gravel, | 356.20 | | | | | | | | |
| - | | Occasional silt seams | 355.43 | talled | SS | 4 | 100 | 15 | | | |
| - | | APL WTPL | | ell Ins | SS | 5 | 100 | 14 | φ | | |
| 4 | | | 353.91 | Monitoring W | | | | | | | |
| 5 | | very sum | | 2 | SS | 6 | 38 | 16 | | * | |
| 6- | | | 352.39 | | | | | | | | |
| - | | Grey/brown, nard, DTPL | 351.62 | | SS | 7 | 100 | 28 | | | |
| 7 | | Stiff Silt, trace sand and gravel, dense, moist | 351.01 | × | SS | 8 | 100 | 35 | | | |
| 8- - - - 9- | | Borehole terminated at 7.47 mbgs. At drilling completion no cave observed. | | | | | | | | | |
| | | | | | | | | | | | |
| | С | ontractor: Altech Drilling and Inv | /estigat | ive Servio | ces Ind | D. | | | Grade Elevation | : 358.48 masl | |
| | D | rilling Method: Split Spoon/Solid | d Stem | | | | | | Top of Casing E | levation: NA | |

Well Casing Size: NA



Project #: 244512

Project: Geotechnical Investigation

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 15, 2019

Logged By: NW

| | | SUBSURFACE PROFILE | | | | | | | SAMPLE | | |
|--------------|--------|--|---------------|----------------------------|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content • % • 10 20 |
| 0 | | Ground Surface | 357.20 | | | | | | | | |
| - | | Pavement Structure Asphaltic Concrete (175 mm) Granular Fill (1325 mm) - Brown sand and gravel trace silt dense to | | | SS | 1 | 71 | 32 | P | | |
| 1 | | compact, damp to moist | 355.67 | | SS | 2 | 50 | 13 | | | |
| 2- | | Silt Till Clayey silt, trace sand and gravel, hard, DTPL | 354.91 | ر م | SS | 3 | 100 | 7 | | Ţ. | |
| 3 | | Occasional saturated silt seams | | ll Installe | SS | 4 | 100 | 11 | - | | |
| - | | | | ring Wel | SS | 5 | 100 | 10 | • | 4 | |
| 4- | | | | o Monito | | | | | | | |
| | | | 352.63 | ž | | | | | | | |
| - 5- - | | Grey/brown, WTPL | | | SS | 6 | 100 | 10 | - | \$ | |
| - - 6- | | | 351.10 | | | | | | | | |
| - | | DTPL | 350.49 | • | SS | 7 | 100 | 15 | | | |
| 7 | | End of Borehole Borehole terminated at 6.71 mbgs. At drilling completion wet cave observed at 6.55 mbgs. | | | | | | | | | |
| 8- | | | | | | | | | | | |
| 9 | | | | | | | | | | | |
| | | | | | | | | | | | |
| | С | ontractor: Altech Drilling and In | vestigat | ive Servic | es Ind | D. | | | Grade Elevation | : 357.20 masl | |
| | D | rilling Method: Split Spoon/Soli | d Stem | | | | | | Top of Casing E | levation: NA | |
| | и | /ell Casing Size: NA | | | | | | | Sheet: 1 of 1 | | |



Logged By: NW

Project: Geotechnical Investigation

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 15, 2019

Project #: 244512

Project Manager: VM

SAMPLE SUBSURFACE PROFILE Recovery (%) Elevation (m) Sample Type N-Value Monitoring Well Details Description Sampler # Standard Penetration Depth (m) Symbol Shear Strength Water Content N-Value SPT I Δ kPa • % 20 40 00 100 200 10 20 354.84 Ground Surface 0 Pavement Structure SS 1 75 28 Asphaltic Concrete (200 mm) Granular Fill (560 mm) - Brown 354.07 sand and gravel, compact, trace silt, 1 damp 2 SS 63 10 Fill 353.31 Brown sand, loose, moist Silt Till SS 3 15 88 2 Clayey silt, trace sand and gravel, 352.55 No Monitoring Well Installed hard, Occasional sand seams, APL 100 SS 4 27 DTPL 3 SS 5 100 28 351.27 Silt, trace to some clay, compact, moist 4 350.26 Grey/brown clayey silt, hard, DTPL SS 6 100 20 5 6 SS 7 100 17 ф 348.13 End of Borehole 7 Borehole terminated at 6.71 mbgs. At drilling completion no cave observed. 8 9 Contractor: Altech Drilling and Investigative Services Inc. Grade Elevation: 354.84 masl Top of Casing Elevation: NA Drilling Method: Split Spoon/Solid Stem Sheet: 1 of 1 Well Casing Size: NA



Project #: 244512
Project: Geotechnical Investigation

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 16, 2019

Logged By: NW

Project Manager: VM

SAMPLE SUBSURFACE PROFILE Recovery (%) Elevation (m) Sample Type N-Value Monitoring Well Details Description Sampler # Standard Penetration Depth (m) Symbol Shear Strength Water Content N-Value kPa SPT I Δ • % 20 40 00 100 200 10 20 354.50 Ground Surface 0 Pavement Structure SS 1 75 31 Asphaltic Concrete (200 mm) Granular Fill (560 mm) - Brown 353.74 sand and gravel, trace to some silt, όó 1 dense, damp 2 SS 63 5 Fill 352.98 Sand, some silt, loose, damp SS 3 75 8 Silt Till 2 Clayey silt, trace gravel, very stiff, 352.21 No Monitoring Well Installed APL Hard, DTPL SS 4 88 14 APL 351.45 3 DTPL SS 5 100 23 4 349.93 APL SS 6 100 15 5 348.40 6 WTPL SS 7 42 15 ф 347.79 End of Borehole 7 Borehole terminated at 6.71 mbgs. At drilling completion no cave observed. 8 9 Contractor: Altech Drilling and Investigative Services Inc. Grade Elevation: 354.5 masl Drilling Method: Split Spoon/Solid Stem Top of Casing Elevation: NA Sheet: 1 of 1 Well Casing Size: NA



Project #: 244512 Project: Geotechnical Investigation Logged By: NW

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 16, 2019

Project Manager: VM

SAMPLE SUBSURFACE PROFILE Recovery (%) Elevation (m) Sample Type N-Value Monitoring Well Details Description Sampler # Standard Penetration Depth (m) Symbol Shear Strength Water Content N-Value SPT I Δ kPa • % 20 40 00 100 200 10 20 352.84 Ground Surface 0 Pavement Structure SS 1 75 27 Asphaltic Concrete (200 mm) Granular Fill (560 mm) - Brown 352.08 sand and gravel, trace to some silt, compact, damp 2 SS 75 10 Fill 351.31 Brown sandy silt, trace to some gravel, loose, moist SS 3 79 10 2 Dark brown with black organic staining, some clay, trace gravel, No Monitoring Well Installed occasional wood pieces 100 SS 4 22 Silt Till 349.79 3 Brown clayey silt, trace sand and 10.11 gravel, stiff, DTPL SS 5 75 12 Very stiff to hard, WTPL to DTPL 4 SS 6 100 19 5 346.74 6 Grey/brown, very stiff, APL SS 7 100 12 ф 346.13 End of Borehole 7 Borehole terminated at 6.71 mbgs. At drilling completion no cave observed. 8 9 Contractor: Altech Drilling and Investigative Services Inc. Grade Elevation: 352.84 masl Drilling Method: Split Spoon/Solid Stem Top of Casing Elevation: NA Sheet: 1 of 1 Well Casing Size: NA



Project: Geotechnical Investigation

Client: Regional Municipality of Waterloo

Location: Snyder's Road, Baden, Ontario

Drill Date: August 16, 2019

Project #: 244512

Logged By: NW

Project Manager: VM

| | | SUBSURFACE PROFILE | | | | | | | SAMPLE | | |
|-----------|--------|--|---------------|----------------------------|-------------|-----------|--------------|-------------|--------------------------------------|--------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value Shea | ar Strength kPa △ 00 200 | Water Content ● % ● 10 20 |
| 0- | | Ground Surface | 352.20 | Ŧ | | | | | | | |
| - | | Pavement Structure Asphaltic Concrete (330 mm) Granular Fill (430 mm) - Brown | 351.44 | Ī | SS | 1 | 71 | 20 | | | |
| 1 | | damp | | | SS | 2 | 67 | 6 | φ. | Î | |
| | | Dark brown clayey silt, trace gravel, hard, DTPL | 350.67 | | SS | 3 | 67 | 9 | | | |
| - | | | 349.91 | | | | | | | | |
| | | Brown, nard | 349.15 | | SS | 4 | 100 | 22 | | | |
| - | | Silt and clay, trace sand and gravel | | alle | 00 | - | 400 | | | | |
| | | | | Inst | 55 | 5 | 100 | 29 | | | |
| _ | | | | /ell | | | | | | | |
| 4- | | | | d ∂ | | | | | | | |
| _ | | | | orin | | | | | | | |
| _ | | | | onito | | | | | | | |
| 5- | | | | Ň | SS | 6 | 100 | 17 | ф П | 4 | |
| _ | | | | ž | | | | | | | |
| _ | | | | | | | | | | | |
| - | | | 346 10 | | | | | | | | |
| -0 | | Grev/brown, WTPL | 540.10 | | | | | | | | |
| _ | | | | | SS | 7 | 100 | 11 | ф | 4 | |
| _ | | | | | | | | | | | |
| 7- | | | | | | | | | | | |
| _ | | | 044.50 | | | | | | | | |
| _ | | Silt compact saturated | 344.58 | | | | | | | | |
| 8- | | | 242.07 | | SS | 8 | 71 | 17 | F | | |
| - | | End of Develope | 343.97 | .★ | | | | | | | |
| - | | End of Borehole | | | | | | | | | |
| 9- | | At drilling completion no cave observed. | | | | | | | | | |
| - | | | | | | | | | | | |
| - | | | | | | | | | | | |
| | С | ontractor: Altech Drilling and Inv | /estigat | ive Servic | es Inc | D. | 1 | 1 | Grade Elevation: 352 | .20 masl | 1 |

Drilling Method: Split Spoon/Solid Stem

Well Casing Size: NA

Top of Casing Elevation: NA



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 17, 2019

| | | SUBSURFACE PROFILE | | | | | | | SAMPLE | | | | |
|-----------|--------|--|---------------|----------------------------|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|--|--|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content ● % ● 10 20 | | |
| 0- | | Ground Surface | 358.60 | Ŧ | | | | | | | | | |
| - | | Topsoil Dark brown sandy silt, verv loose. | 358.30 | | SS | 1 | 50 | 2 | R. | | • | | |
| - | dia. | frozen | 357.84 | | | | | | | | | | |
| 1- | | sand, wet | | | SS | 2 | 60 | 12 | ф l | | | | |
| - | 1 | Sand | | | | | | | | | | | |
| - | | Brown silty fine sand, compact, wet | | alled | | ~ | 70 | 10 | | | | | |
| 2- | 12 | | | Insta | - 55 | 3 | 70 | 18 | l IIIII | | | | |
| - | | | | Well | | | | | | | | | |
| - | | | 355.78 | ring | SS | 4 | 80 | 14 | ф | | • | | |
| 3- | | Silt Grev clavev silt_trace sand_stiff | 355.40 | onito | | | | | | <u>↑</u> | | | |
| - | | WTPL | | lo Mo | SS | 5 | 70 | 12 | ф (| | l f | | |
| | | APL | | | | | | | | | | | |
| 4- | | | | | | | | | | | | | |
| _ | | | | | | | | | | | | | |
| | | | | | SS | 6 | 70 | 10 | | | • | | |
| 5- | | | 353.42 | ¥ | | | | | | | | | |
| - | | End of Borehole | | | | | | | | | | | |
| 6 | | Borehole terminated at 5.2 mbgs. | | | | | | | | | | | |
| - | | | | | | | | | | | | | |
| - | | | | | | | | | | | | | |
| 7- | | | | | | | | | | | | | |
| - | | | | | | | | | | | | | |
| - | | | | | | | | | | | | | |
| 8- | | | | | | | | | | | | | |
| | | | | | | | | | | | | | |
| - | | | | | | | | | | | | | |
| 9- | | | | | | | | | | | | | |
| | С | ontractor: Strata Drilling Group | | | | | | | Grade Elevation | : 358.60 masl | | | |
| | D | rilling Method: Split Spoon / Dir | ect Pus | sh | | | | | Top of Casing E | levation: NA | | | |
| | И | /ell Casing Size: NA | | | | | | | Sheet: 1 of 1 | | | | |
| | | - | | | | | | | | | | | |



Well Casing Size: 51 mm

Log of Borehole: BH2

Project #: 249839

Logged By: NW

Project: Geotechnical Investigation

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Sheet: 1 of 1

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 17, 2019

| | | SUBSURFACE PROFILE | 8 | | SAMPLE | | | | | | | | | | | |
|-------------|--------|--|------------------|---|------------------------------------|-------------|-----------|--------------|-------------|--------|--|---------|--------------------------|----------------------|------|-----------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring | Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Stand: | ard Penel N-Value 0 1 | tration | Shear St △ kPa 100 | rength a △ 200 | Wate | er Content % ● 0 <u>2</u> 0 |
| 0_ | | Ground Surface | 346.03 | | | | | | | | | | | | | |
| - | | Topsoil Dark brown sandy silt, very loose, frozen | 345.26 | kiser | onite | SS | 1 | 60 | 4 | - \ | | | | | | Ţ |
| 1- - | | Wet Silt Brown silt with grey mottling, trace | 344.50 | | Bento | SS | 2 | 60 | 8 | | | | | | | + |
| 2- | | Clay and sand, loose, wet | | | | SS | 3 | 90 | 16 | | | | | | | Ţ |
| - | | | 342.98 | L | € 1 p | SS | 4 | 100 | 11 | | | | | | | • |
| 3 | | Grey clayey silt, trace sand, very stiff, APL | | Scree | Silica San | SS | 5 | 90 | 8 | | | | * | | | |
| 4 | | WTPL | 341.76 341.45 | | | SS | 6 | 100 | 14 | | | | | | | $\left\langle \right\rangle$ |
| - - 5 | | Stiff | 340.85 | | | SS | 7 | 90 | 11 | | | | | | | |
| | | End of Borehole Borehole terminated at 5.2 mbgs. | | Water level = mbgs, measi on Ja 7, 202 | = 2.0 , as ured n. 20. | | | | | | | | | | | |
| | С Л | ontractor: Strata Drilling Group | | | <u> </u> | | <u> </u> | Gra | de Elev | vation | : 346.03 | masl | 80 m | asl | | |
| | | | oorr ui | 511 | | | | | | 100 | | | | . 0-10.0 | 50 m | |



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 17, 2019

| | | SUBSURFACE PROFILE | E | | | | | | SAMPLE | |
|--------------|--------|---|------------------|----------------------------|-------------|-----------|--------------|-------------|--|---------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value N | ıt • |
| 0 | | Ground Surface Topsoil Dark brown sandy silt, very loose, frozen | 349.19 348.97 | T | SS | 1 | 20 | 1 | | |
| - 1- - | | Brown silt with grey mottling, some sand, wet | 340.43 | | SS | 2 | 100 | 10 | | |
| - - 2- | | Brown silt with grey mottling, trace sand and clay, compact, wet | | | SS | 3 | 90 | 13 | | |
| - | | Grey clayey silt, trace sand, hard, | 346.45 | Installed | SS | 4 | 80 | 17 | | |
| 3 | | APL | | oring Well | SS | 5 | 70 | 23 | | |
| 4 | | | 344.62 | - No Monito | | | | | | |
| - 5- - | | Stiff, WTPL | | | SS | 6 | 90 | 13 | | |
| - - 6- | | WTPL to APL | 343.10 | | | | | | | |
| - | | End of Borehole | 342.49 | ¥ | SS | 7 | 60 | 6 | | |
| - | | Borehole terminated at 6.7 mbgs. At drilling completion, a dry cave was measured at 6.1 mbgs. | | | | | | | | |
| - 8 | | | | | | | | | | |
| - - 9 | | | | | | | | | | |
| | С | ontractor: Strata Drilling Group | | | | 1 | | | Grade Elevation: 349.19 masl | |
| | D | rilling Method: Split Spoon / Dir | ect Pus | sh | | | | | Top of Casing Elevation: NA | |
| | И | ell Casing Size: NA | | | | | | | Sheet: 1 of 1 | |



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 18, 2019

Project Manager: VM

| | | SUBSURFACE PROFILE | Ξ | | | | | | SAMPLE | | |
|------------------|--------|--|---------------|---|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content • % • 10 20 |
| 0_ | | Ground Surface | 346.46 | | | | | | | | |
| - | | Topsoil Dark brown sandy silt, very loose, frozen | 346.16 | kiser | SS | 1 | 50 | 1 | | | • |
| 1- 1- - | | Brown silt, some sand, wet Silt Brown silt with grey mottling, trace sand and clay, compact, wet Brown | 344.94 | Bento | SS | 2 | 80 | 10 | | | |
| 2- | | sand and clay, compact, wet | | | SS | 3 | 90 | 13 | | | |
| - | | | 343 41 | | SS | 4 | 60 | 19 | | | |
| 3 - - | | Grey clayey silt, trace sand, very stiff, APL | 040.41 | Screel | ss | 5 | 60 | 19 | | <u>^</u> | • |
| - 4 - - | | | 341.89 | | | | | | | | |
| 5- | | Stiff, WTPL | 341.28 | | SS | 6 | 70 | 7 | | | • |
| | | End of Borehole Borehole terminated at 5.2 mbgs. | | Water level = 0.51 mbgs, as measured on Jan. 7, 2020. | | | | | | | |
| | С | ontractor: Strata Drilling Group | 1 | 1 | 1 | <u> </u> | I | <u> </u> | Grade Elevation | <i>:</i> 346.46 masl | L |
| | D | rilling Method: Split Spoon / Dir | ect Pu | sh | | | | | Top of Casing E | levation: 347 | 32 masl |

Well Casing Size: 51 mm



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 19, 2019

Project Manager: VM

SUBSURFACE PROFILE SAMPLE Sample Type Recovery (%) SPT N-Value Elevation (m) Monitoring Well Details Description Sampler # **Standard Penetration** Depth (m) Symbol Shear Strength Water Content N-Value Δ kPa • % 20 40 00 100 200 10 20 344.39 Ground Surface 0 Topsoil 344.08 SS 1 40 1 Dark brown sandy silt, very loose, frozen 343.62 Brown silt with grey mottling, some 1 sand, wet SS 2 90 12 Silt Brown silt with grey mottling, trace No Monitoring Well Installed sand and clay, compact, wet SS 3 100 14 2 341.95 Brown SS 4 90 8 341.34 3 Grey clayey silt, trace sand, very stiff, WTPL SS 5 70 6 4 339.81 Stiff SS θ 6 80 6 5 339.20 End of Borehole Borehole terminated at 5.2 mbgs. At drilling completion, a dry cave was 6 measured at 4.9 mbgs. 7 8 9 Contractor: Strata Drilling Group Grade Elevation: 344.39 masl Drilling Method: Split Spoon / Direct Push Top of Casing Elevation: NA Sheet: 1 of 1 Well Casing Size: NA



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 19, 2019

| | | SUBSURFACE PROFILE | Ξ | | | | | | SAMPLE | | | |
|--------------|--------|--|---------------|----------------------------|-------------|------------------------------|--------------|-------------|--|--|--|--|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value Shear Strength A kPa A 100 200 Water Content • % • 10 20 | | | |
| 0- | | Ground Surface | 343.75 | Ŧ | | | | | | | | |
| - | | Topsoil Dark brown sandy silt, very loose, frozen | 343.52 | Î | SS | 1 | 30 | 0 | | | | |
| 1- - | | Brown silt, some sand, wet Silt Brown silt with grey mottling, trace | | | SS | 2 | 70 | 11 | | | | |
| 2- | | sand and day, compact, wet | | | SS | 3 | 90 | 16 | | | | |
| - | | Grey clayey silt, trace sand, very stiff, APL | 341.31 | Installed | SS | 4 | 70 | 15 | | | | |
| 3- | | Stiff, WTPL | | oring Well | SS | 5 | 90 | 6 | | | | |
| 4 | | | 339.18 | - No Monito | | | | | | | | |
| 5- | | Very stiff | | | SS | 6 | 60 | 5 | | | | |
| - - 6- | | | 337.65 | | | | | | | | | |
| - | | Stiff | 337.04 | | SS | 7 | 80 | 9 | | | | |
| | | End of Borehole Borehole terminated at 6.7 mbgs. At drilling completion, the borehole was open and dry. | | - | | | | | | | | |
| | С | ontractor: Strata Drilling Group | <u> </u> | | 1 | Grade Elevation: 343.75 masl | | | | | | |
| | D | rilling Method: Split Spoon / Dir | ect Pus | sh | | Top of Casing Elevation: NA | | | | | | |
| | И | /ell Casing Size: NA | | | | Sheet: 1 of 1 | | | | | | |



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 19, 2019

| | | SUBSURFACE PROFILE | • | | | | | | SAMPLE | | |
|-------------------|--------|--|------------------|----------------------------|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content • % • 10 20 |
| | | Ground Surface | 343.71 | | | | | | | | |
| 0 - - | | Topsoil Dark brown sandy silt, very loose, frozen | 343.40 342.95 | T | SS | 1 | 10 | 2 | | | |
| - 1- - | | Brown silt, some sand, moist Silt Brown silt with grey mottling, trace | | | SS | 2 | 100 | 16 | | | Ţ |
| - - 2 | | sand and clay, compact, wet | | ll Installed | SS | 3 | 70 | 12 | | ^ | |
| - - - 3- | | | | toring Wel | SS | 4 | 90 | 20 | | | |
| - | | Grey clayey silt, trace sand, very stiff, WTPL | 340.35 | - No Moni | SS | 5 | 90 | 5 | | | • |
| 4 | | | 339.14 | | | | | | | | |
| - - 5- | | Stiff | 338.53 | ▼ | SS | 6 | 90 | 6 | | | • |
| - - 6- | | End of Borehole Borehole terminated at 5.2 mbgs. | | | | | | | | | |
| - - 7- | | | | | | | | | | | |
| | | | | | | | | | | | |
| 8 | | | | | | | | | | | |
| 9- - | | | | | | | | | | | |
| | C | ontractor: Strata Drilling Group | | | | | | | Grade Elevation | : 343.71 masl | |
| | D | rilling Method: Split Spoon / Dir | ect Pu | sh | | | | | Top of Casing E | levation: NA | |
| | W | ell Casing Size: NA | | | | | | | Sheet: 1 of 1 | | |



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 19, 2019

Project Manager: VM

| SUBSURFACE PROFILE | | | | | | | SAMPLE | | | | | | | | |
|--|--------|---|------------------|---|----------------|-------------|-----------|--------------|-------------|--|--|--|--|--|--|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value Shear Strength A kPa A 100 200 Water Content • % • 10 20 | | | | | |
| 0 | | Ground Surface Topsoil Dark brown sandy silt, very loose, frozen | 343.41 343.18 | ier - | ite 🕈 | SS | 1 | 50 | 2 | | | | | | |
| - 1- - | | Brown silt, some sand, very moist Silt Brown silt with grey mottling, trace | 342.65 | Ϋ́ | Benton | SS | 2 | 80 | 7 | | | | | | |
| - - 2 | | sand and clay, loose, moist to wet | 341.13 | | | SS | 3 | 80 | 19 | | | | | | |
| - - 3- | | Wet Grey clayey silt, trace sand, very | 340.67 | eu | ↓ pu | SS | 4 | 100 | 15 | | | | | | |
| - | | | | Scre | Silica Sa | SS | 5 | 90 | 9 | | | | | | |
| 4 | | WTDI | 338.84 | | | | | | | | | | | | |
| - 5- | | | 338.23 | | | SS | 6 | 100 | 6 | | | | | | |
| | | End of Borehole Borehole terminated at 5.2 mbgs. | | Water level = 0.47 mbgs, a measur on Jan. 2020. | as ed 7, | | | | | | | | | | |
| | С | ontractor: Strata Drilling Group | | <u>.</u> | | | | 1 | <u>.</u> | Grade Elevation: 343.41 masl | | | | | |
| Drilling Method: Split Spoon / Direct Push | | | | | | | | | | Top of Casing Elevation: 344.15 masl | | | | | |

Well Casing Size: 51 mm

Top of Casing Elevation: 344.15 masl



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 20, 2019

Project Manager: VM

| | | SUBSURFACE PROFILE | Ξ | | | | | | SAMPLE | | |
|--------------------|---|--|---------------|----------------------------|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content • % • 10 20 |
| 0 | | Ground Surface | 347.94 | | | | | | | | |
| - | | Topsoil Dark brown sandy silt, very loose, frozen | 347.63 | | SS | 1 | 40 | 2 | | | ſ |
| - 1- - | | Brown, silt, some sand, wet Silt Brown silt with grey mottling, trace | | | SS | 2 | 90 | 9 | | | • |
| 2- | | sand and clay, loose, wet | | | SS | 3 | 100 | 18 | | | • |
| | | | | | SS | 4 | 100 | 24 | т. \ ф | | • |
| | | | | ell Installeo | ss | 5 | 100 | 24 | | | • |
| 4 | | | 343.37 | nitoring We | | | | | | | |
| - 5- - | | Grey clayey silt, trace sand, stiff, WTPL | | No Mo | SS | 6 | 80 | 13 | | | |
| - - 6- | | | | | | | | | | | |
| - | | | 340.93 | | SS | 7 | 80 | 6 | | | |
| 7 - - | ŀ | Silt Till Brown sandy silt, some clay, trace gravel, compact, very moist | 340.32 | | SS | 8 | 70 | 28 | | | |
| | 141 | Very dense, moist | 339.91 | | SS | 9 | 60 | >50 | 3 | | • |
| 8— - - 9— | | End of Borehole Borehole terminated at 8.0 mbgs. | | ¥ | | | | | | | |
| | ~ | | 1 | | 1 | 1 | I | 1 | Ouesle Els st | . 0.47.04 | I |
| | Contractor: Strata Drilling Group Grade Elevation: 347.94 masi Drilling Method: Split Spoon / Direct Push Top of Casing Elevation: NA | | | | | | | | | | |

Well Casing Size: NA



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 20, 2019

Project Manager: VM

SUBSURFACE PROFILE SAMPLE Sample Type Recovery (%) SPT N-Value Elevation (m) Monitoring Well Details Description Sampler # Depth (m) **Standard Penetration** Water Content Symbol Shear Strength N-Value Δ kPa • % 2 40 00 100 200 10 20 346.58 Ground Surface 0 Topsoil 346.28 SS 1 40 0 Dark brown sandy silt, very loose, Riser 5 frozen 345.82 Brown silt, some sand, wet 1 100 SS 2 11 Silt Brown silt with grey mottling, trace 345.06 Bentonite sand and clay, compact, wet Dense SS 3 100 33 2 344.30 Brown SS 4 90 12 343.53 3 Moist SS 5 90 8 Silica Sand ┍₽ 4 Screen⁻ 342.01 Wet SS 6 20 14 5 341.40 Saturated 340.49 6 Grey, very loose SS 7 50 4 ф 339.88 End of Borehole Water 7 level = Borehole terminated at 6.7 mbgs. 4.99 mbgs, as measured on Jan. 7, 8 2020. 9 Contractor: Strata Drilling Group Grade Elevation: 346.58 masl Drilling Method: Split Spoon / Direct Push Top of Casing Elevation: 347.34 masl Well Casing Size: 51 mm Sheet: 1 of 1



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 23, 2019

| | | SUBSURFACE PROFILE | = | | | | | | SAMPLE | | | |
|-----------------------|--------|--|---------------|----------------------------|-------------|-----------|------------------------------|-------------|---------------------------------|--------------------------------------|---------------------------------|--|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content • % • 10 20 | |
| | | Ground Surface | 346.86 | | | | | | | | | |
| | | Topsoil Dark brown sandy silt, trace rootlets and gravel, very loose, wet | 346.09 | Ĩ | SS | 1 | 37 | 2 | | | f | |
| - 1- - | | Silt Brown silt, some sand, trace clay, compact, wet | 345.33 | | SS | 2 | 75 | 12 | | | • | |
| - 2- - | | Brown with grey mottling | | I Installed | SS | 3 | 100 | 21 | | | • | |
| - - 3- | | | 343.81 | toring Wel | SS | 4 | 100 | 19 | | | | |
| - | | Grey clay and silt, trace sand, hard, APL | | - No Mon | SS | 5 | 100 | 19 | - | Â | | |
| 4 | | | 342.28 | | | | | | | | | |
| - 5- - | | Clayey silt, trace sand, very stiff, WTPL | 341.67 | × | SS | 6 | 100 | 21 | | | | |
| - - 6 - - | | End of Borehole Borehole terminated at 5.2 mbgs. | | | | | | | | | | |
| - 7- - | | | | | | | | | | | | |
| - 8- - | | | | | | | | | | | | |
| - 9- - | | | | | | | | | | | | |
| | С | ontractor: Strata Drilling Group | | | | | Grade Elevation: 346.86 masl | | | | | |
| | D | rilling Method: Split Spoon / Dir | | | | | Top of Casing Elevation: NA | | | | | |
| | И | ell Casing Size: NA | | Sheet: 1 of 1 | | | | | | | | |



Logged By: NW

Project: Geotechnical Investigation

Project #: 249839

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 23, 2019

Project Manager: VM

SUBSURFACE PROFILE SAMPLE Sample Type Recovery (%) SPT N-Value Elevation (m) Monitoring Well Details Description Sampler # Depth (m) **Standard Penetration** Water Content Symbol Shear Strength N-Value Δ kPa • % 2 40 00 100 200 10 20 350.28 Ground Surface 0 Topsoil 349.97 SS 1 54 8 Q Dark brown sandy silt, trace rootlets Riser 2 and gravel, loose, moist 349.52 Brown silt, some sand, compact 1 100 SS 2 16 Silt Brown silt with grey mottling, trace Bentonite sand and clay, compact, moist Ē SS 3 68 15 2 SS 4 100 16 Ŀ Sand 3 Silica 8 5 100 SS 18 Screen ¹ 4 345.71 Grey/brown clayey silt, trace sand, SS 6 79 12 Δ stiff, WTPL 5 344.18 6 Silt, some clay, trace sand and Water gravel, compact, saturated SS 7 100 15 ф level = 343.57 5.31 End of Borehole mbgs, as 7 measure d on Jan. Borehole terminated at 6.7 mbgs. 7, 2020. 8 9 Contractor: Strata Drilling Group Grade Elevation: 350.28 masl Drilling Method: Split Spoon / Direct Push Top of Casing Elevation: 351.03 masl

Well Casing Size: 51 mm


Log of Borehole: BH13

Project #: 249839

Logged By: NW

Project: Geotechnical Investigation

Client: Estate of Enid Schmidt c/o Kelly & Co. Barristers & Solicitors

Location: Foundry Street at Gingerich Road, Baden, ON

Drill Date: December 23, 2019

Project Manager: VM

| SUBSURFACE PROFILE SAMPLE | | | | | | | | | | | |
|---------------------------|--------|--|---------------|----------------------------|-------------|-----------|--------------|-------------|---------------------------------|--------------------------------------|---------------------------------|
| Depth (m) | Symbol | Description | Elevation (m) | Monitoring Well Details | Sample Type | Sampler # | Recovery (%) | SPT N-Value | Standard Penetration N-Value | Shear Strength △ kPa △ 100 200 | Water Content • % • 10 20 |
| 0- | | Ground Surface | 343.72 | | | | | | | | |
| - | | Topsoil Dark brown sandy silt, trace rootlets and gravel, very loose, wet | 342.96 | er - | SS | 1 | 37 | 3 | - | | |
| 1- | | Brown, some clay | 342.50 | ntonite | SS | 2 | 84 | 4 | - - | | • |
| _ | | Silt, some clay, trace gravel | 342.20 | | | | | | $+$ \setminus $ $ $ $ $ $ $ $ | | |
| - 2- - | | Silt Brown silt, trace sand and clay, compact, wet | 341.44 | | SS | 3 | 100 | 13 | | | F |
| - | | Grey/brown | | ca Sand | SS | 4 | 100 | 12 | | | • |
| 3— - - | | Clayey silt, trace sand, very stiff, WTPL | 340.67 | Screen | SS | 5 | 72 | 12 | | ^ | • |
| 4- - - | | | | | | | | | | | |
| 5- | | Silt Jooso wet | 338.69 | Water level = 0.26 | SS | 6 | 84 | 9 | | | • |
| - | | End of Borehole | | measured on Jan. 7, | | | | | | | |
| - 6- - | | Borehole terminated at 5.2 mbgs. | | 2020. | | | | | | | |
| - | | | | | | | | | | | |
| 7 | | | | | | | | | | | |
| - - 8- | | | | | | | | | | | |
| - | | | | | | | | | | | |
| - 9- | | | | | | | | | | | |
| | С | ontractor: Strata Drilling Group | I | 1 | 1 | 1 | 1 | 1 | Grade Elevation | : 343.72 masl | 1 |
| | D | rilling Method: Split Spoon / Dir | ect Pus | sh | | | | | Top of Casing E | levation: 344. | 56 masl |

Well Casing Size: 51 mm

Sheet: 1 of 1



Geotechnical Lab Testing





Particle Size Distribution Analysis Test Results



and a second second second

NOTES:



MTE File No.: 48730-100

Engineers, Scientists, Surveyors.



Hydrogeological Assessment of Alternatives Technical Memorandum (April 26, 2022)





| To: | Dot Roga, C.E.T., MTE Consultants | MTE File No.: | C48730-100 |
|-----|-----------------------------------|---------------|---|
| | | Date: | June 27, 2022 |
| | | From: | Fraser Cummings, P. Geo. |
| | | Project Name: | Baden Trunk Sewer Environmental Assessment |
| | | | |

Re: Baden Trunk Sewer Class B Environmental Assessment Hydrogeological Assessment of Alternatives

1.0 Background

MTE Consultants Inc. has prepared this technical memorandum as input to the Baden Trunk Sewer Class B Environmental Assessment, Category 1.E – Groundwater Resources. The purpose of the Hydrogeology overview is to provide geological and hydrogeological technical information and analysis as input to assess different sewer alignment alternatives with respect to the impacts related to groundwater resources in the following factor categories:

- Groundwater Recharge areas: Encroachment into the Regional Recharge Area.
- Groundwater Quality: Impacts to Vulnerable Areas (IPZ, SGRA, HVA and WHPA):
 - IPZ Intake Protection Zone;
 - SGRA Significant Groundwater Recharge Areas;
 - HVA Highly Vulnerable Aquifers; and
 - WHPA Well Head Protection Areas.
- Shallow Groundwater Movement: Impacts to existing shallow flow and interflow patterns relative to surface water features and groundwater discharge areas.

This technical memorandum presents and discusses each of the above criteria with the aim of applying numeric scoring between 1 (least favourable) and 10 (most favourable).

The Primary and Detailed Study Areas (PSA and DSA, respectively) are presented on Figure 1.

1.1 Previous Investigations

Several hydrogeological and geotechnical investigations have been conducted in the PSA, as follows (the locations of monitoring wells installed in the PSA are presented on **Figure 1**):

• MTE Consultants. 2021a. 1012 Snyder's Road West, Development Hydrogeology Study.

Installation of 11 monitoring wells to depths ranging from 3.0 to 18.2 metres below ground surface (mbgs) equipped with electronic water level monitors (data loggers). The Regional shallow groundwater table is located at depths ranging from 1.7 to 11.8 mbgs with shallower groundwater encountered in the central portion of the site and adjacent to surface water features. Monitoring of these locations is ongoing.



• MTE Consultants. 2021b. 1012 Snyder's Road West, Phase II Environmental Site Assessment Report.

Drilling of 11 boreholes to a maximum depth of 12.2 mbgs; installation of monitoring wells within 6 of the boreholes. Groundwater levels in the monitoring wells ranged from 0.9 to 11.0 mbgs.

 Peto MacCallum Ltd. 2020. Geotechnical Investigation, Proposed Residential Subdivision, 1012 Snyder's Road West, Baden, Ontario.

Drilling of 12 boreholes to a maximum depth of 8.1 mbgs.

 Pinchin Ltd. 2020a. Geotechnical Investigation - Proposed Snyder's Road Reconstruction, Christian Street to Gingerich Road, Baden, Ontario.

Drilling of 29 boreholes to depths ranging from 5.5 to 8.4 mbgs; installation of monitoring wells within 6 of the boreholes. Groundwater levels in the monitoring wells ranged from 1.3 to 5.1 mbgs.

• Pinchin Ltd. 2020b. Geotechnical Investigation, Proposed Residential and Employment Lands Development, Foundary Street at Gingerich Road, Baden, Ontario.

Drilling of 13 boreholes to depths ranging from 5.2 to 8.0 mbgs; installation of monitoring wells within 6 of the boreholes. Groundwater levels in the monitoring wells ranged from 0.3 to 5.3 mbgs.

1.2 Current investigation

To fill-in data gaps in the local geological and hydrogeologic understanding, MTE drilled five boreholes in August of 2021 and completed each as a monitoring well. The locations of the monitoring wells are presented on **Figure 1**. Details of the drilling program including borehole logs are provided in the concurrent Geotechnical Site Characterization (MTE, 2021c).

2.0 Regional Hydrogeological Setting

2.1 Physiography

The PSA is located within the physiographic region known as the Waterloo Hills. The surface is comprised of sandy hills consisting of ridges of sandy till, kames or kame moraines. The soils in the area are generally well-drained loams and loamy tills (Chapman and Putnam, 2007). Physiographic mapping of the area illustrates the PSA as largely being located within an undrumlinized till plain physiographic landform. The southern half of the DSA is located within a kame moraine landform. (**Figure 2**).

2.2 Quaternary Geology

Ontario Geological Survey (OGS) published map sheet (Map 2559) (**Figure 3**) shows the surficial geology of much of the PSA includes clay to silt textured till, which is characterized as the Maryhill Till and consists of clay till and clayey diamicton (OGS, 2010). Small areas consist of ice-contact sand comprised of fine to coarse sand and gravel that is poorly to well sorted. Along Baden Creek in the DSA, the surficial geology is expected to consist of gravel, sand, silt and clay stream alluvium deposits.



2.3 Regional Hydrostratigraphic Units

The Region of Waterloo obtains approximately 80% of its water supply from groundwater within bedrock and sedimentary aquifers, with approximately half being supplied from the Waterloo Moraine (Bajc et. al., 2014). The Waterloo Moraine is located almost entirely within the Region of Waterloo and is comprised of three separate units, which grade from coarse-grained sediments (gravel) to finer-grained sediments (silt and clay) with decreasing depth (Bajc et. al., 2014).

The fine-grained sediments of the Waterloo Moraine are referred to as the Upper, Middle, and Lower Maryhill Tills. These units represent regionally significant aquitards and will "act as a significant barrier to vertical water movement where present in a thickness greater than 5 metres" (Terraqua Ltd., 1995). The Maryhill Till consists primarily of sandy silt to silty clay and clayey silt diamictons (Bajc et. al., 2014). The Waterloo Moraine Aquifer Units are characterized by interbedded layers of silt, sand and gravels. Clay-rich glaciolacustrine deposits with gravels and pebbles are also present.

These Waterloo Moraine sediments overlie the Catfish Creek Till, characterized as a stoney to sandy diamicton, which is often over-consolidated and forms an important marker horizon within the Region of Waterloo (Karrow, 1987). The Catfish Creek Till acts as an important, relatively continuous regional aquitard that is compositionally distinct and widely distributed across the Region of Waterloo (Terraqua Ltd., 1995; Bajc and Shirota, 2007; Aquaresource Inc., 2009).

The PSA is located on the western flank of the Waterloo Moraine. The units most likely to be present in the vicinity are **bolded** in the table below (AquaResource Inc., 2012; Stantec Consulting Ltd, 2012).

| OGS Name | Geologic Formation | Geologic Material | Regional Aquifer Name | |
|--|--|--|-----------------------------|--|
| ATB1 | Upper Maryhill Till, Port Stanley Till, Tavistock Till, Mornington Till Silty to clayey till | oper Maryhill Till, Port Stanley Till, vistock Till, Mornington Till Silty to Silty to clayey till ayey till | | |
| AFB1 Upper Waterloo Moraine stratified sediments and equivalents | | Mainly fine sand, some gravel | | |
| ATB2 Middle Maryhill Till and equivalents | | Silty to clayey till, silt, clay | Aquifer 1 | |
| AFB2 | Middle Waterloo Moraine Stratified Sediments and equivalents | Mainly fine sand, some gravel | | |
| ATB3 | Lower Maryhill Till and stratified equivalents | Silty to clayey till, silt, clay | Aquitard 2 | |
| AFB3 | Lower Waterloo Moraine Aquifer or Catfish Creek Till outwash | Sand and gravel | Aquifer 2 | |
| ATC1 | Upper / Main Catfish Creek Till | Stoney, silty to sandy till | Aquitard 3 | |
| AFD1 | Pre-Catfish Creek coarse-textured glacio fluvial/lacustrine deposits | Sand and gravel | Aquifer 3 | |

Geological Formations within the Waterloo Moraine:



| | OGS Name Geologic Formation | | Geologic Material | Regional Aquifer Name |
|--|-----------------------------|--|-------------------|-----------------------------|
| ATE1 Canning Drift (till and associated fine- textured glaciolacustrine deposits) | | Silty to clayey till, silt, clay | Aquitard 4 | |
| AFF1 Pre-Canning coarse-textured glaciofluvial/glaciolacustrine deposit | | Pre-Canning coarse-textured glaciofluvial/glaciolacustrine deposit | Sand and gravel | Aquifer 4 |
| ATG1 Pre-Canning coarse-textured till | | Stony, silty to sandy till | Aquitard 4 | |

Regional shallow groundwater elevations, provided by the Grand River Conservation Authority (GRCA; published April 2016), indicate that the regional shallow groundwater elevations in the PSA range between approximately 346 to 353 metres above mean sea level (mAMSL). The regional shallow groundwater flow direction is south-westerly towards the Nith River.

2.4 Paleozoic Geology

Map sheets in publication regarding the Paleozoic geology illustrate bedrock beneath the Subject Lands consists of the upper Silurian Salina Formation which is characterized by irregularly bedded shale interbedded with thin layers of brown dolostone (Armstrong and Dodge, 2007). The bedrock is expected to be encountered at depths in the order of 40 metres below grade in the vicinity of the Subject Lands (AquaResource Inc., 2012).

2.5 Source Water Protection

2.5.1 Groundwater Recharge Areas

Groundwater recharge occurs where precipitation and snowmelt infiltrate into the ground to feed aquifers, watercourses, and wetlands. SGRAs are typically associated with coarse-grained sediments (i.e. sands and gravels) or very shallow overburden material covering upland areas on the landscape. In the Region of Waterloo two different recharge areas have been mapped; Significant Recharge Areas and Regional Recharge Areas.

Significant Groundwater Recharge Areas (SGRAs) are presented in the Lake Erie Region Source Protection Committee Assessment Report (LERSPC, 2020).GRCA mapping indicates that small portions in the north of the PSA are located within a SGRA. The SGRAs are illustrated on **Figure 4**.

The Region of Waterloo Official Plan (Region of Waterloo, 2010) includes mapping of the Regional Recharge Area (RRA) which is generated using a method which differs from that used for the SGRA delineation. Region of Waterloo mapping presented on **Figure 5** indicates that neither the PSA nor the DSA are located within the RRA.

In addition, the Source Protection Mapping (MECP, 2021) generated under the Clean Water Act to protect municipal water supplies indicates that neither the PSA nor the DSA are located above a Highly Vulnerable Aquifer.



2.5.2 Municipal Well Fields

The nearest municipal wellfields are Foxboro Green, located approximately 950 m north of the northern boundary of the PSA; Wilmot Centre, located approximately 3 km southeast from the southeast corner of the PSA; and New Hamburg; located approximately 3.5 km southeast of the southeast corner of the PSA.

The water supply for the Foxboro Green and New Hamburg wellfields is obtained from bedrock production wells, which are open to the Salina Formation at depths ranging from 47 to 76 mbgs. The water supply for the Wilmot Centre wellfield is obtained from production wells screened within the semiconfined Middle Waterloo Moraine Sands (AFB2) at depths ranging from 30 to 40 mbgs.

No portion of PSA or DSA are located within a wellhead protection area (WHPA) for a municipal supply well (MECP, 2021). The location of the WHPAs is illustrated on **Figure 6**.

3.0 Local Hydrogeological Setting

Based on the previous and current investigations outlined in **Section 1**, the surficial sediments in the DSA are consistently described as till and are interpreted to represent the Upper Maryhill Till (Regional Aquitard 1). Throughout the larger PSA, some surficial sediments are sands, which are interpreted to represent the Upper Waterloo Moraine (Aquifer 1).

Based on the borehole logs from the previous and current investigation, the water table is present in the DSA within the Upper Maryhill Till. As this represents a Regional aquitard, shallow groundwater flow is expected to be predominately vertical. In isolated locations where sand seams are present (at MW103-21, for example), there may be a stronger horizontal component to groundwater flow.

Using water levels collected from the previous and current investigations, a map of theoretical water table contours has been generated and is presented as **Figure 7**¹. Note that this representation of the water table is theoretical only as it has been compiled using water levels taken at different times over a 3-year period. Regional water table mapping available from the GRCA is presented in **Figure 8**. Both of these figures suggest that some component of shallow groundwater flow is directed towards Baden Creek.

Mapping available from the Ministry of Natural Resources and Forestry (MNRF) through their Fish ON-Line application indicates that cold water fish species (Rainbow Trout and Largemouth Bass) have been identified in the through-pond of Baden Creek located immediately west of Foundry Street. White Suckers, a cool water species, have been identified in Baden Creek further west of the pond. These findings suggest a groundwater contribution to these surface water features.

4.0 Assessment of Alternatives

Commentary on each alternative, in relation to the criteria under the Class B Environmental Assessment, Category 1.E – Groundwater Resources, based on the regional and local information presented above is provided in **Table 1**.

¹ MTE acknowledges that a representative water table surface would require water levels to be measured from monitoring locations on the same date.



Table 1: Evaluation of Alternatives

| Alternative | Encroachment into the Regional Recharge Area | Groundwater Quality: Impacts to Vulnerable Areas | Shallow Groundwater Movement | | |
|--|---|--|--|--|--|
| Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing) | | | | MW105-21 was drilled along this route. The absence of a sand and gravel unit suggests that dewatering volumes would be low at this location. | |
| 2 Snyders/ Foundry | | None of the | Groundwater levels above ground surface were observed at MW104-21 which is located along | MW105-21 was drilled along this route. The absence of a sand and gravel unit suggests that dewatering volumes would be low at this location. | |
| Charlotta/Mill/ 3 Charles/ Foundry | None of the alternatives encroach into the Regional Recharge Area | alternatives encroach into an Intake Protection | | MW105-21 was drilled along this route. The absence of a sand and gravel unit suggests that dewatering volumes would be low at this location. | |
| Charlotta/ 4 Queen and Baden Creek | | b the ea Ea Ea Ea Ea Ea Ea Ea Ea Ea E | all routes. It is MTE's opinion MW104-21 is located in fine grained material (clay and silt) and as such is not anticipated to yield significant | MW103-21 and MW104-21 were drilled along this route. Greater thicknesses of sand at MW103-21 compared to other locations may result in higher dewatering volumes during construction at this location. Further south, at MW104-21, no sand unit was present suggesting lower dewatering volumes at this location despite the groundwater levels being observed above surface. This route passes through an area where groundwater potentially discharges to Baden Creek (in the vicinity of BH2, BH8, and BH13) and has been observed at depths as shallow as 0.25 mbgs. | |
| 5 Christian and Baden Creek | | | water during excavation. | MW101-21, MW102-21, and MW104-21 were drilled along this route. Of the three, only MW102 contained a sand layer which may indicate higher dewatering volumes may be observed during construction at this location. This route passes through an area where groundwater potentially discharges to Baden Creek (in the vicinity of BH2, BH8, and BH13) and has been observed at depths as shallow as 0.25 mbgs. MTE recommends cut-off collars be installed during construction to maintain groundwater flow towards Baden Creek post construction. | |



5.0 References

AquaResouce Inc. 2012. Region of Waterloo Tier Three Water Budget and Local Area Risk Assessment – Rural Well Fields Characterization Report. Prepared for the Regional Municipality of Waterloo.

Armstrong, D.K. and Dodge, J.E.P. 2007. *Paleozoic geology of southern Ontario*; Ontario Geological Survey, Miscellaneous Release – Data 219.

Chapman, L.J. and Putnam, D.F. 2007. *Physiography of Southern Ontario*; Ontario Geological Survey, Miscellaneous Release – Data 228.

Lake Erie Region Source Protection Committee. 2020. Grand River Source Protection Area Approved Assessment Report.

Ministry of the Environment, Conservation and Parks (MECP). 2021. Source Protection Information Atlas. Retrieved July, 2021, from

https://www.gisapplication.lrc.gov.on.ca/SourceWaterProtection/Index.html?viewer=SourceWaterProtection.SWPViewer&locale=en-US

MTE Consultants. 2021a. 1012 Snyder's Road West, Development Hydrogeology Study

MTE Consultants. 2021b. 1012 Snyder's Road West, Phase II Environmental Site Assessment Report.

MTE Consultants. 2021c. Baden Trunk Sewer Class Environmental Assessment, Geotechnical Characterization.

Ontario Geological Survey 2010: *Surficial geology of southern Ontario*; Ontario Geological Survey, Miscellaneous Release – Data 128, Revised.

Peto MacCallum Ltd. 2020. Geotechnical Investigation, Proposed Residential Subdivision, 1012 Snyder's Road West, Baden, Ontario

Pinchin Ltd. 2020a. Geotechnical Investigation - Proposed Snyder's Road Reconstruction, Christian Street to Gingerich Road, Baden, Ontario.

Pinchin Ltd. 2020b. Geotechnical Investigation, Proposed Residential and Employment Lands Development, Foundary Street at Gingerich Road, Baden, Ontario.

TFC:smk

M:\48730\100\Reports\MTE Reports\HydroG\48730-100_2022-06-27_tech memo_Hydrogeology Technical Memo.docx





















Ŀ,





Fluvial Geomorphological Assessment Technical Memorandum (August 30, 2022)









Baden Trunk Sanitary Sewer Environmental Assessment

Fluvial Geomorphological Assessment, Environmental

Wilmot Township

August 30, 2022

August 30, 2022 WE 21038

Mr. Steve Peterson, CET Director, Land Development MTE Consultants, Inc. 520 Bingemans Centre Dr., Kitchener, ON N2B 3X9

Dear Mr. Peterson:

RE: Baden Trunk Sanitary Sewer Environmental Assessment Fluvial Geomorphological Assessment

1. INTRODUCTION

Water's Edge Environmental Solutions Team (WEEST) was authorized by MTE to complete a geomorphic characterization on Baden Creek and two of it's tributaries as well as the Huehn Award Drain in the Township of Wilmot. This is to support the future location of the Baden Trunk Sanitary Sewer within the area of Snyder's Road and Gingerich Road. MTE is assessing multiple alternatives for the alignment of the trunk sewer and these alternatives could impact the noted channels. WEEST has assessed these channels in order to advise on fluvial geomorphological impacts that may occur to the channels both during construction and in the future.

vater's e

This report will first characterize the existing watercourses that drain into the main branch of Baden Creek and the general watershed conditions within the study area. Relevant literature on the site was reviewed and confirmed and as necessary, updated based on a desktop analysis and synoptic geomorphic survey(s). Results from the desktop assessments were used to determine cross-sections to undertake detailed field surveys sufficient to determine channel characterization and erosion thresholds.

The subject area is located in the southwest section of the village of Baden. The main branch of Baden Creek originates the northeast corner of Baden, while one tributary collects drainage from a residential development to the north, and another one from the northwest corner. The main branch is dammed on the west side of Foundry St. but continues on as it drains generally in a southerly direction out of Baden. Downstream of Gingerich Road, Baden Creek eventually drains into the Nith River. The Huehn Award Drain drains flows from the east side of Baden and is south of the main Baden Creek branch. **Figure 1** shows the Study Area as well as the subject reaches.

We have acquired data and background information as part of our assessment of the Study Area creeks, these data sources for the analysis include:

- Physiography of Southern Ontario by Chapman & Putnam (digital data from Ministry of Northern Development and Mines (MNDM));
- Site Survey and Field Assessments; and,
- Discussions with MTE staff.



Figure 1: Study Reaches

2. BACKGROUND REVIEW

Physiography and Surficial Geology

Understanding the surficial and underlying geology for the study area provides insight into the geological influence on channel geometry, expected rates of erosion, and helps define the quantity and type of sediment available for watercourses to transport and deposit.

The studied watercourse is located within the Waterloo Hills physiographic region. This region is known to contain sandy hills, kames, and moraines. The areas around Baden are particularly known to have large kames that are made up of gravel and sand. The physical landform which the study site is located in is split between kame moraines and till plains (shown in **Figure 2**). The surficial geology of the site is mostly till that is clay to silt textured and is from glaciolacustrine deposits or shale.

General Stream Characteristics

The tributary of Baden Creek originates from a SWM facility on the north side of railway tracks (**Figure 1**). The SWM facility is for a residential development south of Snyder's Rd. The main branch of Baden Creek in the study area is a third order tributary to the Nith River, which it joins into roughly 5 km downstream of the site. Upstream of the study area along Baden Creek is a large pond that is controlled by a dam, this dam backwaters a large portion of Baden Creek.





Figure 2: Local Surficial Geology (data from MNDM)

3. FIELD INVESTIGATIONS

A geomorphic assessment of Baden Creek including two tributaries and a small section of the Huehn Award Drain were completed within the Study Area. **Figure 3** shows the reach breakdown for the Study Area channels. A series of cross-sections were surveyed along with longitudinal profiles of the creeks. Bed and bank materials were recorded to come up with a general characterization of the channel. During the site visit conducted in January 2022, staff gathered a synoptic level understanding of the trends and channel condition through the application of rapid assessments. It is noted that the field investigations took place during the coldest time of the year and the site was partially obscured with ice and snow. However, a follow up investigation in June 2022 provided additional information such as substrate data. Rapid field assessments aid in determining the sensitive sections of the creek which are the focus of an erosion assessment. Locations for detailed surveys of profile, cross-section, and particle analysis were selected based at an appropriate location downstream from the SWM discharge based on distinguishing features observed within the reach.

Reach Delineation

Channel morphology and substrate characteristics can change along a watercourse. Hence, it becomes imperative to account for this variation by delineating lengths of a watercourse that exhibit similar planform, sediment substrate, land use, local geology, valley confinement, hydrology and gradient. For this study, five separate reaches have been identified, two in the west tributary to Baden Creek (Tributary A), two along the main branch of Baden Creek, and one in the Huehn Award Drain. The west tributary is split into two reaches due to the slope change, channel size, substrate change, and entrenchment of the channel. The main branch of Baden Creek varied in the same ways from the other two reaches of the tributary, however it is split into separate reaches for discussion purposes from upstream of the dam and downstream of the dam. The results of the geomorphic survey and analysis are presented in **Table 3.3a-c.**





Figure 3: Reach Map

Reach 1 – Baden Creek Tributary A

Reach 1 extends from the railway track at the north end of the subject reach down to the farm lane crossing approximately 320 metres south. The reach runs between two row crop fields and collects runoff from the fields through tile drainage. The riparian zone along the tributary is exclusively grasses, which obscure the creek from view. The channel itself is small and generally straight, likely placed in this alignment for agricultural purposes. The tributary likely has intermittent flow although it was flowing at the time of the site visit, however it is likely dry for some of the summer months. The banks of the channel are loose soils but protected by shallow vegetation roots. The substrate of the channel is silty with some small sands and gravels. No large material was noted during the inspection. The channel characteristics from the geomorphic survey are listed in **Table 3.3a**.



Photograph 1: View of Reach 1



| Table 3.3a. Dankidii (Dki) channel characteristics | | | | | | |
|--|-------|------|------|-------|------|---------|
| Parameter | XS1 | XS2 | XS3 | XS4 | XS5 | Average |
| Bkf Width (m) | 0.65 | 0.80 | 0.77 | 0.48 | 1.12 | 0.76 |
| Bkf Mean Dep. (m) | 0.22 | 0.20 | 0.19 | 0.18 | 0.12 | 0.18 |
| Bkf Max Dep. (m) | 0.31 | 0.27 | 0.24 | 0.28 | 0.20 | 0.26 |
| Bkf Area (m ²) | 0.15 | 0.16 | 0.14 | 0.09 | 0.14 | 0.14 |
| Wetted Perim. (m) | 1.00 | 1.07 | 1.02 | 0.78 | 1.27 | 1.03 |
| Hyd. Rad. (m) | 0.15 | 0.15 | 0.14 | 0.11 | 0.11 | 0.13 |
| Width-Depth Ratio | 2.95 | 4.00 | 4.05 | 2.67 | 9.33 | 4.60 |
| Entrench. Ratio | 76.9 | 62.5 | 64.9 | 104.2 | 44.6 | 70.6 |
| Bankfull Slope (m/m) | 0.012 | | | | | |
| Riffle Slope (m/m) | | | (| 0.025 | | |
| Substrate D ₅₀ | 0.75 | | | | | |
| Substrate D ₈₄ | | | | 7.08 | | |

Table 3.3a: Bankfull (Bkf) channel characteristics

Reach 2 – Baden Creek Tributary A

Reach 2 is part of the tributary and is 220 m in length. It is located downstream of the farm lane crossing and ties into Baden Creek. The reach is separated from the upper portion of the tributary because it has a steeper slope, it is more entrenched, and the substrate particle size is larger. The riparian area along this reach is also exclusively grasses, however the adjacent fields are grazing fields, not row crops. The channel shows significant signs of bank erosion and lateral migration. Multiple steep outer banks with heights greater than 1 metre were noted. The section of the reach upstream of the confluence with Baden Creek is particularly unstable as it is actively scouring the bed of the creek and a few small knickpoints were present. The shallow rooting depth of the riparian grasses plays a part in this erosion susceptibility. The sinuosity of the channel also increased through this reach. Substrate found in the creek are silts, sands and gravels which are sourced **3.3b**.



Photograph 2: View of Reach 2



| Parameter | XS6 | XS7 | XS8 | Average | |
|---------------------------|-------|------|------|---------|--|
| Bkf Width (m) | 1.18 | 2.08 | 0.74 | 1.33 | |
| Bkf Mean Dep. (m) | 0.23 | 0.21 | 0.29 | 0.24 | |
| Bkf Max Dep. (m) | 0.34 | 0.45 | 0.39 | 0.39 | |
| Bkf Area (m²) | 0.27 | 0.43 | 0.21 | 0.30 | |
| Wetted Perim. (m) | 1.48 | 2.39 | 1.20 | 1.69 | |
| Hyd. Rad. (m) | 0.18 | 0.18 | 0.18 | 0.18 | |
| Width-Depth Ratio | 5.13 | 9.90 | 2.55 | 5.86 | |
| Entrench. Ratio | 42.4 | 24.0 | 67.6 | 44.7 | |
| Bankfull Slope (m/m) | 0.021 | | | | |
| Riffle Slope (m/m) | 0.028 | | | | |
| Substrate D ₅₀ | 1.60 | | | | |
| Substrate D ₈₄ | 8.66 | | | | |

Table 3.3b: Bankfull (Bkf) channel characteristics

Reach 3 – Baden Creek

Reach 3 is the main branch of Baden Creek and the study reach is 680 m long. The study reach starts at the confluence with the Baden Cr tributary and ends near Gingrich Rd. Two hundred and thirty metres of the creek were surveyed for the geomorphic and erosion assessment, although other sections such as downstream of the confluence with Huehn Award Drain were visually assessed as well. Approximately 400 m upstream from the start of the study reach is a large mill pond. The main channel of Baden Creek is approximately 4.5 m wide and meanders through agricultural grazing land. The alignment of the channel indicates historical modification for agricultural purposes. Even with those changes the current state of the creek is natural and the creek has generally re-established in its current alignment. The sinuosity of this reach is moderate at 1.25, with a straight channel being 1.0. The reach has sections that are lined with grasses similar to the tributary, but also has large pockets of wooded areas as well. The creek has many signs of active bank erosion and lateral migration, particularly in grassed areas where the root depth is not sufficient to protect the loose banks from eroding. The bed slope through the reach is very low. Substrate found within the reach ranges from silts to boulders. Typically, riffles are sands and gravels with some cobble, while pools are heavily silted.



Photograph 3: View of Reach 3 typical conditions



| | · / | | | | |
|----------------------------|-------|------|-------|---------|--|
| Parameter | XS9 | XS10 | XS11 | Average | |
| Bkf Width (m) | 3.68 | 5.79 | 4.41 | 4.87 | |
| Bkf Mean Dep. (m) | 0.45 | 0.70 | 0.51 | 0.57 | |
| Bkf Max Dep. (m) | 0.57 | 1.06 | 0.79 | 0.88 | |
| Bkf Area (m ²) | 1.66 | 4.03 | 2.27 | 2.86 | |
| Wetted Perim. (m) | 4.07 | 6.53 | 4.83 | 5.40 | |
| Hyd. Rad. (m) | 0.41 | 0.62 | 0.47 | 0.52 | |
| Width-Depth Ratio | 8.18 | 8.27 | 8.65 | 8.52 | |
| Entrench. Ratio | 13.6 | 8.64 | 11.3 | 10.4 | |
| Bankfull Slope (m/m) | | 0 | .004 | | |
| Riffle Slope (m/m) | 0.006 | | | | |
| Substrate D ₅₀ | 47.71 | | | | |
| Substrate D ₈₄ | | 1(| 03.16 | | |

Table 3.3c: Bankfull (Bkf) channel characteristics

Reach 4 – Baden Creek (East of Foundry St.)

Reach 4 of Baden Creek is from Foundry St. to approximately Brewery St. This reach of the main branch of Baden Creek is heavily impacted by the dam west of Foundry St. The dam backwaters the creek approximately up to Brewery St., after which it regains a typical sinuous form. This reach also includes a small tributary (Tributary B) which drains lands at the northeast corner of Baden. The tributary was not assessed north of Snyder's Rd. The short section of Baden Creek between Brewery St. and Foundry St. has a very slow flow velocity. The typical depth of water through this section was 0.3 - 0.5 m, while bankfull depth is approximately 1.5 m. The section is heavily sedimented with no distinctive channel features such as riffles. No geomorphic survey was completed of this reach, however the typical width of the channel is 12 to 15 m. From a practical fluvial geomorphological perspective this section of Baden Creek could be viewed as a pond rather than an active channel. As such no bed scour is likely to occur unless the downstream dam is removed.



Photograph 4: View of Reach 4 typical conditions



Reach 5 – Heuhn Award Drain

The Huehn Award Drain runs east to west and is approximately 1.2 kms in length. It originates south of Gingerich Rd. however after it crosses the road it collects drainage from a series of SWM ponds in the residential areas on the north side of Gingerich Rd. It crosses Foundry St. just before joining into the main Baden Creek branch upstream of Gingerich Rd. The small section between Foundry St. and Baden Creek was the only section of the drain that was assessed as the upstream section flows along private property. The length of this section is approximately 50 m. It is lined on all sides by tall grasses with mid to low rooting depth for bank protection. The bottom half of the channel banks show this as scour along the toe of the bank is present along the majority of this reach. The slope from the culvert to the confluence with the main channel is low (~0.5%). The bankfull dimensions are approximately 1.5 m in width and 0.80 m in depth. Substrate found within the reach ranges from gravels to cobbles with some silt. The underlying tills were exposed in some areas. The channel characteristics from the geomorphic survey are listed in **Table 3.3d**.



Photograph 5: View of the Huehn Drain typical conditions

| Table 3.3d: Bankfull (E | Bkf) channel | characteristics |
|-------------------------|--------------|-----------------|
|-------------------------|--------------|-----------------|

| Parameter | Average |
|----------------------------|---------|
| Bkf Width (m) | 1.55 |
| Bkf Mean Dep. (m) | 0.75 |
| Bkf Max Dep. (m) | 0.80 |
| Bkf Area (m ²) | 1.16 |
| Wetted Perim. (m) | 2.96 |
| Hyd. Rad. (m) | 0.39 |
| Width/Depth Ratio | 2.07 |
| Entrench. Ratio | 64.5 |
| Bankfull Slope (m/m) | 0.005 |
| Riffle Slope (m/m) | 0.012 |
| Substrate D ₅₀ | 25.16 |
| Substrate D ₈₄ | 55.27 |



Rapid Assessments

Rapid field assessments provide an indication of the channel stability and ecological stream condition, while also identifying primary processes in action (e.g. widening). The Rapid Geomorphic Assessment (RGA) and the Rapid Stream Assessment Technique (RSAT) together provide a thorough description of the existing channel conditions.

The RGA is a check sheet that documents indicators of different modes of channel adjustment: widening, aggradation, degradation, and planform adjustment. These observations are quantified to produce a value that indicates the state of channel stability: "In Regime/Stable" (<0.20), "Transitional/Stressed" (0.21-0.40), or "In Adjustment/Unstable" (>0.40).

RSAT employs a semi-quantitative approach to characterize stream conditions whereby the user assigns a score to 6 different evaluation criteria (MOE, 2003). Abiotic and biotic indicators which influence overall stream quality have been streamlined and weighted appropriately within each of the evaluation criteria. The six criteria are:

- 1. Channel stability;
- 2. Channel scouring and sediment deposition;
- 3. Physical in-stream habitat;
- 4. Water quality;
- 5. Riparian habitat conditions; and
- 6. Biological conditions

The scores are then added to communicate the final index of the stream condition. These values can describe the stream quality as "Excellent" (42-50), "Good" (30-41), "Fair" (16-29), or "Poor" (<16).

Results for the RGA (**Table 3.4**) show that each reach is either stressed or unstable. Reach 1 had poor scores in Widening and Planform Adjustment, meaning there is considerable lateral changes occurring along the reach. Indicators such as basal scour are the prime reason for this. Reach 2 had poor scores in each category, however the Widening and Planform Adjustment were again very poor. Eroding banks, fracture lines along the top of banks, basal scour, and cut-off channels are common in the reach. Reach 3 is categorized as Stressed due to the poor scores in Aggradation and Widening categories. Indicators such as accretion on point bars, siltation in pools, fallen trees, and exposed tree roots are common in this reach. Reach 4 was reviewed using the RGA form however it is not applicable due to the backwater effect of the dam on the geomorphic characteristics of the channel. Reach 5 received a score of 0.25 with most indicators within the Degradation and Widening categories.



| | | Stability | | | | |
|-------------------------------|-------------|-------------|----------|------------------------|-------|--------------|
| | Aggradation | Degradation | Widening | Planform Adjustment | Index | Condition |
| Reach 1 (Trib A U/S) | 0 | 0.10 | 0.30 | 0.43 | 0.21 | Transitional |
| Reach 2 (Trib A D/S) | 0.29 | 0.30 | 0.40 | 0.43 | 0.35 | Unstable |
| Reach 3 (Baden Cr. D/S) | 0.57 | 0 | 0.60 | 0 | 0.29 | Transitional |
| Reach 4 (Baden Cr. U/S) | NA | NA | NA | NA | NA | NA |
| Reach 5 (Huehn Drain) | 0 | 0.67 | 0.33 | 0 | 0.25 | Transitional |

Table 3.4: Reach-averaged RGA Results

The RSAT revealed that stream condition in Reach 1 and 2 are Fair. These low scores are attributed to poor scores in Channel Stability, Riparian Conditions, and Biological Indicators. Since these channels are small and act more as drains than year-round creeks this can be expected. Reach 3 is described as in Good condition, with generally higher than average scores in all categories. Categories that were low are bank stability and biological indicators. Reach 4 was assessed. Reach 5 received a rating of Good as well although the same issues as Reach 3 with poor scores in Bank Stability and Biological Indicators.

| Reach | Score | Condition |
|-------|-------|-----------|
| 1 | 24 | Fair |
| 2 | 22 | Fair |
| 3 | 32 | Good |
| 4 | NA | NA |
| 5 | 31 | Good |

Table 3.5: Reach-averaged RSAT Summary Results

4. EROSION THRESHOLD ASSESSMENT

General

For the purposes of an erosion threshold assessment, bankfull cross-sections were surveyed at riffle or run locations because it can be expected that channel velocities and shear stresses on the bed are greatest through these sections therefore providing the most representative values. The longitudinal profile was also surveyed to determine the local energy gradient (slope). Substrate measurements were made in order to determine the typical D₅₀ for each reach.

This detailed field data (cross-section, gradient, and particle distribution) is used to estimate the bankfull discharge, shear stress, and critical discharge values. Specifically, the critical discharge indicates the point at which sustained flows will tend to entrain and transport sediment. In this analysis, the median particle size (D_{50}) was used as an index size.

To determine the critical shear stress for particle entrainment, the methods presented by Komar (1987), Julien (1995) and Fischenich (2001) were used. These methods adapt and update the work



of Shields (1936). The Komar method is most appropriate to gravel sized material, while Fischenich also incorporates finer material (sands). Based on the critical shear stress determined by each method, a critical depth is back-calculated and a critical discharge is determined. Then, if required, this critical discharge is applied as an erosion threshold target when controlling the flow rate and timing for stormwater releases for a range of design storm events. Otherwise, the critical discharge provides an understanding on the sensitivity of the channel to erosion.

Erosion Threshold Considerations and Discussion

Each cross section used in the analysis was selected to provide representative erosion thresholds. Attempts were made to locate naturally formed riffles or runs for cross-sectional surveys as these provide locations where flows are concentrated, and their composition is indicative of the type of material that becomes mobilized under frequent flow conditions below and up to the bankfull discharge - before spilling onto the floodplain.

Using the data collected during the field investigations and desktop analysis, bankfull characteristics for cross-sections were summarized. The bankfull energy gradient and bed materials are also summarized (**Table 3.3a-d**). Related hydraulic parameters were determined including stream power, unit stream power, bed shear stress at each riffle cross-section. Results are presented for all four reaches (**Table 4.1**).

Estimated bankfull hydraulics reveals that under bankfull conditions, available shear stresses are not great enough to entrain particles larger than the median grainsize in Reaches 3 and 5 but they are great enough to entrain particles in Reaches 1 and 2. This suggests that bankfull conditions are highly favourable for channel erosion in Reaches 1 and 2 but not in 3 and 5.

Erosion threshold values were determined for representative cross-sections from each reach (**Table 4.2**). Essentially, flows are modeled through each cross-section until the critical depth to mobilize the median particle size (D_{50}) is achieved.

| | Cross-Sections | | | | | | |
|--|--------------------------------|--------------------------------|--------------------------------------|------------------------------------|--|--|--|
| Parameter | Reach 1 (Trib A U/S) | Reach 2 (Trib A D/S) | Reach 3 (Baden Cr. D/S) | Reach 5 (Huehn Drain) | | | |
| Relative Roughness (m) | 169.6 | 103.4 | 8.7 | 15.1 | | | |
| Shear Velocity (m/s) | 0.14 | 0.14 | 0.16 | 0.16 | | | |
| Velocity based on ff/RR (m/s) | 2.1 | 1.99 | 1.27 | 1.54 | | | |
| Bankfull Discharge (m ³ /s) | 0.31 | 0.54 | 2.86 | 1.79 | | | |
| Froude # | 1.55 | 1.32 | 0.57 | 0.57 | | | |
| Stream Power (W/m) | 45.5 | 63.5 | 168.3 | 122.8 | | | |
| Unit Stream Power (W/m ²) | 59.1 | 53.8 | 38.2 | 79.2 | | | |
| Mean Bed Shear (N/m²) | 18.7 | 19.5 | 24.4 | 26.2 | | | |

Table 4.1: Summary hydraulics



| | | Cross-Sections | | | | |
|----------------------|--------------------------------------|----------------------------|-----------------------------------|--------------------------------------|------------------------------------|--|
| Method | Parameter | Reach 1 (Trib A U/S) | Reach 2 (Trib A D/S) | Reach 3 (Baden Cr. D/S) | Reach 5 (Huehn Drain) | |
| Komar (1987) | Critical Shear (N/m2) | 0.55 | 1.17 | 34.75 | 18.33 | |
| | Ratio of Critical Shear/Bed Shear | 34.27 | 29.25 | 0.70 | 1.02 | |
| | Critical Depth (m) | 0.004 | 0.006 | 0.59 | 0.37 | |
| | Critical Flow (m ³ /s) | 0.001 | 0.001 | 6.89 | 1.35 | |
| | Ratio of Critical Flow/Bankfull Flow | 0.004 | 0.002 | 2.41 | 0.89 | |
| Julien (1995) | Critical Shear (N/m2) | 0.40 | 0.75 | 38.61 | 19.14 | |
| | Ratio of Critical Shear/Bed Shear | 46.73 | 45.39 | 0.63 | 0.98 | |
| | Critical Depth (m) | 0.003 | 0.004 | 0.66 | 0.39 | |
| | Critical Flow (m ³ /s) | 0.001 | 0.001 | 8.59 | 1.76 | |
| | Ratio of Critical Flow/Bankfull Flow | 0.003 | 0.001 | 3.01 | 1.16 | |
| Fischenich (2001) | Critical Shear (N/m2) | 0.44 | 0.97 | 38.88 | 19.09 | |
| | Ratio of Critical Shear/Bed Shear | 42.77 | 35.11 | 0.63 | 0.98 | |
| | Critical Depth (m) | 0.003 | 0.005 | 0.66 | 0.39 | |
| | Critical Flow (m3/s) | 0.001 | 0.001 | 8.71 | 1.52 | |
| | Ratio of Critical Flow/Bankfull Flow | 0.003 | 0.002 | 3.045 | 1.005 | |

Table 4.2: Threshold results

Overall, the different methods (Komar, Julien, and Fischenich) produce nearly the same result. For both reaches in the tributary a value of 0.001 m³/s or 1 L/s was calculated. For the main branch of Baden Creek, the average result is 8.06 m³/s or 8,060 L/s while the short section of Heuhn Award Drain produces 1.54 m³/s or 1,540 L/s. The very low critical flow of the tributaries reflects the sensitive nature of the reaches, particularly the sandy substrate. The larger particles found within the main branch of Baden Creek as well as in the study reach of Huehn Award drain are large enough to protect the channel bed from constant erosion. However, the banks of the Baden Creek and Huehn Award Drain are still susceptible to erosion due to the bank composition and riparian vegetation. The provided critical flow values can be used as the target flow for any further erosion threshold assessments.

5. TRUNK SEWER ALIGNMENT ALTERNATIVE REVIEW

In order to aid in assessing the alternatives for the trunk sewer alignment a background review, geomorphic assessment, rapid field assessments, and an erosion threshold analysis were completed. Five alternatives have been provided by the study team, each with a slight variation in the trunk sewer alignment. Since the wastewater pumping station is located south of Gingerich Rd. all of the alternatives require the crossing of Baden creek near the facility. Therefore, this is not considered when comparing the alternatives.

After crossing Gingerich Rd. the alignments for Alternatives 1, 2, and 3 run along Foundry St. and then along Snyder's Rd. This alignment requires the crossing of Huehn Award Drain and Baden Creek at the pond. In addition, Alternative 1 runs east along Snyder's Rd. and would also require crossing of Tributary B. All of these crossings would require open-cut construction activity to install the trunk sewer. This activity would require diversion of the watercourse during construction and rehabilitation of the channel. The disturbed section could create weaknesses in the bed and banks of the creek and lead to erosion if not properly rehabilitated. Erosion and sediment controls would



be required for any channel modifications and be properly installed and maintained in order to avoid any excessive sediment loads from entering the watercourse.

The locations of the crossings for Alternatives 1-3 through Heuhn Award Drain and Baden Creek were not identified as sensitive by the erosion threshold analysis. The substrate of Huehn Award Drain is gravels and cobbles and is therefore better equipped for dealing with bed scour. The Baden Creek crossing was not assessed with the erosion threshold analysis because it is a pond, and therefore no bed scour is likely to occur. However, the depth of the trunk sewer is a concern, and should be placed at a depth that the channel is not able to scour to even if creek conditions change in the future. Therefore, a minimum depth of 2 times the bankfull depth should be the goal. For the Heuhn Award Drain this would be 1.5 m below the creek bed and for Baden Creek this would be approximately 3 m below the creek bed.

The alignments for Alternatives 4 and 5 generally follow along Baden Creek through the open space between Gingerich Rd. and Snyder's Rd. They both cross Baden Creek immediately north of Gingerich Rd. by way of a trenchless crossing. Alternative 5 follows along Tributary A and crosses it near the railway by way of a trenchless crossing. Alternative 4 cuts to the east and crosses Tributary A near the confluence with the main branch of Baden Creek. This is a trenchless crossing as well, however the location of the crossing is a highly sensitive section of the creek with knick-pointing and bed scour actively occurring. Due to this, if selected, the depth of the trenchless crossing at this location should be supported by a detailed bed scour analysis.

Baden Creek and Tributary A are noted to be actively migrating laterally. Since Alternatives 4 and 5 generally follow the alignment of the creeks it is important to offset the trunk sewer at a distance that will ensure the creek will not be able to meander to it in the future. The MNR Technical Guide – River and Stream Systems notes that for unconfined systems a meander hazard of 20 times the bankfull width centred over the channel is acceptable without a detailed study. For Baden Creek this would result in an almost 100 metre total beltwidth or 50 m on either side of the creek. This is seen as excessive, and it is unlikely that Baden Creek would ever meander this far. The existing beltwidth of Baden Creek is approximately 25 m and considering the topography as well as the ongoing lateral migration an additional 25 m buffer for a total beltwidth of 75 m is likely to be sufficient if not conservative for an erosion hazard offset. However, a detailed meander beltwidth study may be required to confirm the historical and future meander beltwidth of Baden Creek. Tributary A has a narrower beltwidth (~10 m) and would require an offset of 10 m based on 100% of the overall preliminary beltwidth.

With all this considered, the alternative that is likely to cause the least disruption to creek form and function is Alternative 5. This alternative has only 2 crossings of a watercourse and it is via trenchless installation. Lateral proximity to the creeks is a concern but if a satisfactory offset can be achieved then Alternative 5 has the least erosion potential associated with it and should be the preferred alternative as it has the least immediate impact from construction activity as well as a low likelihood of future impact to the creek.



Respectfully submitted,

Ed Gazendam, Ph.D., P.Eng., President, Sr. Geomorphologist Water's Edge Environmental Solutions Team Ltd.

Nik Gazendam, C.Tech. CAN-CISEC Project Manager and Senior Technician

APPENDICES:

Appendix A: Photo Inventory

REFERENCES:

Fischenich, C. 2001. Stability Thresholds for Stream Restoration Materials. EMRRP Technical Notes Collection (ERDC TN-EMRRP-SR-29), U.S. Army Engineer Research and Development Center, Vicksburg, M.S. www.wes.army.mil/e/emrrp

Galli, J. 1996. *Development and Application of the Rapid Stream Assessment Technique (RSAT) in the Maryland Piedmont.* Paper presented at American Engineering Foundation Conference, Snow Bird, UT

Komar, Paul D. 1987 Selective grain entrainment and the empirical evaluation of flow competence. Sedimentology, 34: 1165-1176.

Ontario Flow Assessment Tool (OFAT) 2014. Ontario Ministry of Natural Resources and Forestry. Copywrite: Queen's Printer for Ontario 2014. Obtained from: http://www.giscoeapp.lrc.gov.on.ca/web/mnr/wrip/ofat/Viewer/viewer.html

Ontario Ministry of the Environment (OMOE). 2003. Stormwater Management Planning and Design Manual. Ontario Ministry of the Environment, Toronto, Ontario

Ontario Ministry of Natural Resources. Water Resources Section. 2002. Technical Guide – River and Stream Systems: Erosion Hazard Limit.






Finance Germanphology

Natural & bannel Cesign

Stream Restoration

Monitoring

Erosian Assessment

Sed ment Transport.

apple - Marija at www.wite.co.ps.co.ps

APPENDIX A:

Photo Inventory

BADEN CREEK - TRIBUTARY A - REACH 1



PHOTOGRAPH NO.: 1 FROM: Left side of creek at north end of Reach 1 LOOKING: Upstream towards railway tracks. Creek crosses to the left.



PHOTOGRAPH NO.:2 FROM: Left bank of creek at top end of Reach 1. LOOKING: Downstream at Reach 1 which is well vegetated and is between two fields.





PHOTOGRAPH NO.: 3 FROM: Centre of creek, middle of Reach 1. LOOKING: At uncovered portion of creek. Creek typically hidden by grasses.



PHOTOGRAPH NO.: 4 FROM: Centre of creek at lower end of Reach 1. LOOKING: Downstream towards farm crossing. Bed slope has increased.





BADEN CREEK - TRIBUTARY A – REACH 2

PHOTOGRAPH NO.: 5 FROM: On top of farm crossing between Reach 1 and 2. LOOKING: Downstream over Reach 2.



PHOTOGRAPH NO.: 6 FROM: Centre of creek, in middle of Reach 2. LOOKING: At the substrate composition in Reach 2.





PHOTOGRAPH NO.: 7 FROM: Middle of Reach 2. LOOKING: Channel knickpointing into undisturbed tills at a few locations.



PHOTOGRAPH NO.: 8 FROM: Left side of creek at lower end of Reach 2. LOOKING: Downstream at heavily eroded bank.





PHOTOGRAPH NO.: 9 FROM: End of Reach 2. LOOKING: Downstream at confluence with main channel of Baden Creek.

BADEN CREEK – REACH 3



PHOTOGRAPH NO.: 10 FROM: Right bank at top end of Reach 3. LOOKING: Upstream at typical channel conditions.





PHOTOGRAPH NO.: 11 FROM: Middle of creek at upper portion of Reach 3. LOOKING: At channel substrate composition. Mostly sands with some gravel/cobble.



PHOTOGRAPH NO.: 12 FROM: Right bank at upper portion of Reach 3. LOOKING: Downstream at open section of creek, high right bank, low left bank.





PHOTOGRAPH NO.: 13 FROM: Left bank of creek in middle of Reach 3. LOOKING: Upstream at typical channel conditions. Steep eroding right bank.



PHOTOGRAPH NO.: 14 FROM: Left bank near middle of Reach 3. LOOKING: Downstream towards open section of channel, steep eroding banks throughout.





BADEN CREEK – REACH 4

PHOTOGRAPH NO.: 15 FROM: Roadside culvert on Brewery St. LOOKING: Upstream at typical channel conditions, backwater ends at top of view.



PHOTOGRAPH NO.: 16 FROM: Centre of channel between Brewery St. and Foundry St. LOOKING: Downstream at typical backwatered conditions, heavy siltation.





PHOTOGRAPH NO.: 17 FROM: Centre of channel between Brewery St. and Foundry St. LOOKING: Across channel at confluence with Tributary B.



PHOTOGRAPH NO.: 18 FROM: Top of culvert at Foundry St. LOOKING: Downstream at pond.





HEUHN AWARD DRAIN – REACH 5

PHOTOGRAPH NO.: 19 FROM: Sidewalk on Foundry St. LOOKING: Upstream at Heuhn Award Drain, running through private property.



PHOTOGRAPH NO.: 20 FROM: Edge of road on Foundry St. LOOKING: Downstream at channel and confluence with the Baden Creek.





PHOTOGRAPH NO.: 21

FROM: Centre of channel

LOOKING: Downstream at typical channel conditions, scoured banks and tall grasses.



PHOTOGRAPH NO.: 22 FROM: Centre of channel LOOKING: At bed of channel with cobbles, gravels and exposed underlying till.



Technical Memo 2 Socio-Cultural Environment



| Drai | Bad | en Trunk |
|------|----------------------|------------------|
| Proj | San | itary Sewer EA |
| To: | Jeff Molenhuis, To | wnship of Wilmot |
| | Bryan Bishop, To | wnship of Wilmot |
| | Trevor Heywood, GRCA | |
| | Kaoru Yaiimi ROV | V |

cc: Pierre Chauvin, MHBC Paul Britton, MHBC MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENTAL ASSESSMENT Socio-Cultural Assessment of Alternatives

2.0 Socio-Cultural Environment

2.A Land Use Policy

This Factor Group reviews the degree of consistency with the Provincial Policy Statement (PPS) of 2020 and conformity with A Place to Grow – Growth Plan for the Greater Horseshoe (2020), the amended Regional Official Plan (2015) and the Township of Wilmot Official Plan (2019). It supports the coordination and integration of planning for infrastructure with growth management and land use planning, and provision of infrastructure in an efficient manner.

There are a number of policies that implement the vision and underlying objectives of the PPS. Some of the relevant policies include:

- Focus growth and development within the settlement of Baden.
- Manage and direct land use to achieve efficient and resilient development and land use patterns.
- Direct the efficient provision of infrastructure, public service facilities and sewage water.
- Require infrastructure and public service facilities to be integrated and coordinated with land use planning.
- Support energy conservation, air quality and consider the impacts of a changing climate.
- Require that provincially significant natural features and areas are to be protected for the long term.
- Direct the protection, improvement or restoration of the quality and quantity of water.

Land Use Designation and Policies

MHBC was retained by MTE to undertake a Land Use Planning Evaluation for the Municipal Class EA. The technical memo (dated April 28, 2022) was prepared to provide input into the evaluation of the five alternatives presented, and identify the potential land use impacts these alternatives have on adjacent properties. Refer to the *Technical Memo – Land Use Planning Evaluation, Baden Trunk Sanitary Sewer EA (April 28, 2022)* in **Appendix A**.

This Factor Group evaluates the degree to which each alternative is aligned with Provincial, Regional and Township growth, taking into account development goals, objectives or policies (high, medium, low), and input from the MHBC technical memo. This was performed by identifying upper and lower tier infrastructure as this is a limiting factor with respect to development patterns and timing.



<u>Alternative 4</u> received a maximum score of 10. This alternative scored the highest due to its autonomy to implement infrastructure improvement without upper tier regulations, formulating development and growth independently. The location and depth of sewer would provide serviceability to all lands within the Primary Study Area and help facilitate the Township's long-term vision of growing New Hamburg and Baden together. Timing of construction requires coordination to avoid impacting the growing season for areas located on agricultural lands just north of Gingerich Road.

<u>Alternative 5</u> received a score of 8. This alternative is similar to Alternative 4 except a larger portion of the alignment is located on agricultural land outside of the urban boundary. Disruptions to cropping or other agricultural activities would require mitigation by avoiding construction during the growing season. Following construction, the below grade sewer would not impact agricultural activities/uses. The location and depth of sewer would provide serviceability to all lands within the Primary Study Area and help facilitate the Township's long-term vision of growing New Hamburg and Baden together.

<u>Alternative 3</u> received a score of 4. This alternative has a low degree of compatibility due to the timing and prioritization of combined regional and local projects. This alignment consists of one (1) regional project and one (1) local project. This alignment would not provide sanitary services to all lands west of Baden and would not facilitate the Township's long-term vision of growing New Hamburg and Baden together.

<u>Alternative 2</u> received a score of 2. This alternative is similar to Alternative 3 except it is entirely dependent upon the timing and prioritization of regional projects. This alignment consists of two (2) separate regional projects. This alignment also would not provide sanitary services to all lands west of Baden and would not facilitate the Township's long-term vision of growing New Hamburg and Baden together.

<u>Alternative 1</u> received the minimum score of 1. This alternative is the least compatible due to the timing and prioritization of this alignment. It consists of two (2) regional projects and one (1) local project. It would also be a temporary disturbance to the access for the agricultural/livestock operation along existing farm lane during construction. This alignment would not provide sanitary services to all lands west of Baden and would not facilitate the Township's long-term vision of growing New Hamburg and Baden together.

Impact of Sanitary Sewer on Adjacent Lands

This Factor Group assess the impacts of the sanitary sewer construction on the existing land uses, residences and business utilizing the right-of-ways.

<u>Alternative 5</u> received a maximum score of 10. This alternative impacts the fewest residences due to the alignment traversing along only Christian Street (**Figure 8.1**). Coordination is required within the Schmidt Lands to reduce the impact on cropping activities.

<u>Alternative 4</u> received a score of 8. This alternative is similar to Alternative 5 except it impacts the additional residential properties along Charlotta Street and Queen Street (**Figure 7.1**). Coordination is required within the Schmidt Lands to reduce the impact on cropping activities.

<u>Alternative 3</u> received a score of 4. This alternative significantly impacts many residential properties along Christian Street, Charlotta Street, Mill Street, and Charles Street (**Figure 6.1**). In addition, businesses and residential properties along Foundry Street would be significantly impacted.

<u>Alternative 2</u> received a score of 2. This alternative would significantly impact businesses and residential properties along Snyder's Road West and Foundry Street (**Figure 5.1**). Local residences that utilize Snyder's Road West and Foundry Street for access would also need to be rerouted.



<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 except it would also impact farming operations along the private farm road north of Snyder's Road East (**Figure 4.1**).

2.B Archaeological Resources

Detritus Consulting Ltd. was retained by MTE Consultants Inc. to conduct a Stage 1-2 archaeological assessment for the Municipal Class EA. The Stage 1 background research indicated that the Detailed Study Area (DSA) exhibited moderate to high potential for the identification and recovery of archaeological resources. The Stage 2 investigation consisted of a pedestrian survey at 3m intervals of most of the agricultural land throughout the DSA, and a test pit assessment of the grassy and treed areas, and the seasonal creek extending from Baden Creek to the west.

The various road allowances throughout the Study Area, the Canadian National Railway bed, and the pumping house driveway were evaluated as having no find potential based on the identification of extensive and deep land alteration that has severely damaged the integrity of archaeological resources.

Their report, *Stage 1-2 Archaeological Assessment Baden Trunk Sanitary Sewer Replacement, Baden (July 19, 2022)* along with its supplemental document (**Appendix B**), was used to evaluate each of the five alternatives and their impacts according to the archeological findspots and their buffers. Refer to Tile 3, Tile 7 and Tile 8 of the supplemental document in **Appendix B**.

This Factor Group was assessed based on the proximity of the alignment in relation to the archeological fndspot's 20m protective buffer, 50m monitoring buffer or if further archeological investigation is required.

<u>Alternative 2</u> received a maximum score of 10. This alternative scored the highest due to it be located entirely within an existing right-of-way and surrounded by previously disturbed areas. No archeological assessment is required (**Figure 5.1**).

<u>Alternative 3</u> received a score of 9. This alternative is similar to Alternative 2 in that the entirety of the alignment is within previously disturbed areas. No archeological assessment is required (**Figure 6.1**).

<u>Alternative 1</u> received a score of 8. This alternative is similar to Alternative 2 along Snyder's Road West and Foundry Street. However, the forcemain portion of the alignment to service Parcel O could potentially bring about another find (**Figure 4.1**).

<u>Alternative 5</u> received a score of 5. This alternative is within the 50m construction monitoring buffer of archeological findspots along the Baden Creek. Archeological monitoring would be required during construction (**Figure 8.1**).

<u>Alternative 4</u> received the minimum score of 1. This alternative is similar to Alternative 5 however it is in close proximity to another 50m construction monitoring buffer of Findspot AiHe-469 along the Baden Creek, just south of Queen Street. (**Figure 7.1**).



2.C Built Heritage

MHBC completed a *Cultural Heritage Evaluation Report for the Municipal Class EA (December 2021)* on behalf of MTE (**Appendix C**). This report was prepared to provide input into the evaluation of the five alternatives with respect to their impact to local Cultural Heritage Resources. Eleven Cultural Heritage Resources were identified as a result of this evaluation:

Listed Heritage Properties:

- 1316 Snyder's Road West
- 53 Snyder's Road West
- 55 Snyder's Road West
- 8 Foundry Street
- 76 Mill Street
- 75 Charles Street
- 130 Queen Street

Designated Heritage Properties:

- 44 Beck Street
- 60 Snyder's Road West
- 1138 Snyder's Road West
- 39 Snyder's Road West

Baden has also been identified as a potential Cultural Heritage Landscape by the Region of Waterloo as well as identifying Foundry Street and Snyder's Road as scenic roads. Adverse impacts are designated in the Ontario Heritage Toolkit (OHTK). It states that impacts may be direct or indirect, occurring over the short term or long term, and occurring during a pre-construction phase, construction phase or post-construction phase.

This Factor Group was evaluated based on the potential for adverse impacts on designated heritage properties identified in the Township Heritage Register and in accordance with the Ontario Heritage Act (OHA), and as described within the MHBC Cultural Heritage Evaluation Report

Though none of the Alternatives would likely result in the removal of identified cultural heritage resources, scoring is based on the potential to impact listed and designated heritage properties.

Each alternative impact assessment is related to construction work, which is both subterranean and will be undertaken for a limited period of time. It assesses the potential impacts due to land disturbances and vibrations and are therefore based on proximity to listed properties. Refer to Figure 2 of the *Cultural Heritage Evaluation Report for the Municipal Class EA (December 2021)* in **Appendix C** for Heritage locations.

<u>Alternative 5</u> received a maximum score of 10. This alternative scored the highest because it had the least impact on built heritage. It is sited furthest away from one (1) listed property, 130 Queen Street.

<u>Alternative 4</u> received a score of 9. This alternative is similar to Alternative 5 except that it is closer in proximity to 130 Queen Street.



<u>Alternative 3</u> received a score of 5. This alternative is adjacent to four (4) listed and two (2) designated properties. Construction would affect the scenic route along Foundry Street during construction.

<u>Alternative 1</u> received a score of 2. This alternative is similar to Alternative 3 except it would also impact the scenic route along Snyder's Road West as well as Foundry Street during construction.

<u>Alternative 2</u> received the minimum score of 1. This alternative is the same as Alternative 1 except that it would impact the scenic route for a longer period of time due to sewer depth. Both Alternative 1 and 2 affect the road immediately in front of the Castle Kilbride.

2.D Noise/Odour

This Factor Group was evaluated based on the potential noise impacts during construction and the generated noise and odour associated with wastewater pumping stations.

<u>Alternative 5</u> received a maximum score of 10. This Alternative would eliminate the noise and odour associated with the removal of two (2) existing pumping stations. Construction noise would be limited to residences along Christian Street (**Figure 8.1**).

<u>Alternative 4</u> received a score of 9. This Alternative is similar to Alternative 5 except that construction noise would impact additional residences along Charlotta Street and Queen Street (**Figure 7.1**).

<u>Alternative 3</u> received a score of 8. This Alternative is similar to Alternative 4 except construction noise would impact additional residences and businesses along Mill Street, Charles Street and Foundry Street (**Figure 6.1**).

<u>Alternative 2</u> received a score of 2. This Alternative would eliminate the noise and odour associated with the removal of one (1) pumping station located at the school along Snyder's Road West, however, a large number of business and residences would be impacted by construction noise along Snyder's Road West and Foundry Street (**Figure 5.1**).

<u>Alternative 1</u> received the minimum score of 1. This alternative would increase noise and odour due to the additional pumping station required to service Parcel O. Construction noise would impact the greatest number of business and residences along Snyder's Road West, Foundry Street and the farm road (**Figure 4.1**).



Land Use Planning Evaluation, Baden Trunk Sanitary Sewer EA Technical Memorandum (April 28, 2022)





KITCHENER WOODBRIDGE LONDON KINGSTON BARRIE BURLINGTON

| То: | Jeff Martens, Dot Roga, Carla Illman |
|----------|---|
| From: | Pierre Chauvin |
| Date: | April 28, 2022 |
| File: | 0941CD |
| Subject: | Technical Memo – Land Use Planning Evaluation, Baden Trunk Sanitary Sewer EA |

INTRODUCTION

MHBC was retained by MTE to undertake a Land Use Planning Evaluation for the Baden Trunk Sanitary Sewer Municipal Class Environmental Assessment. The study area is located in the Town of Baden, in the general area located east of Nafziger Road, west of Foundry Street on either side of Snyder's Road West.

The purpose of this technical memo is to provide input into the evaluation of each of the five Alternatives of the Baden Trunk Sanitary Sewer through the identification of potential land use impacts within the Detailed Study Areas and provide a preliminary evaluation of the proposed Alternatives. This information will feed into the Environmental Assessment process along with other technical reports so that a Preferred Alternative can be selected.

The purpose of the waste water/sanitary sewer line is to provide municipal services for growth areas within the community of Baden. There are currently 5 alignment Alternatives proposed. The attached figure prepared by MTE, identifies the location of the Alternative routes. **Figures 1** and **2** illustrate the Alternative routes in relation to the Township Official Plan Land Use and Countryside Schedules.

A wastewater servicing study for the growth areas of Baden and New Hamburg was prepared in 2014. This study evaluated sanitary infrastructure needs which were part of the 2011 Baden and New Hamburg Water and Waste Water Master Plan. The anticipated growth for the Township of Wilmot prompted a servicing strategy for Baden and New Hamburg. As a result, the Township initiated a Schedule B Municipal Class Environmental Assessment for the Baden Trunk Sanitary Sewer, which commenced in June 2021. The existing sanitary infrastructure lacks the capacity and depth for the anticipated growth within Baden. The proposed sanitary trunk will accommodate current and future growth while optimizing existing infrastructure which consists of trunk sewers, gravity sewers, and pressure sewers.

POLICY CONTEXT

The Places to Grow Act, 2005 provides for the development and approval of Regional Growth Plans that are to guide government investments, land use planning policies and decisions. Waterloo Region is located within the Greater Golden Horseshoe and **A Place to Grow - Growth Plan for the Greater Golden Horseshoe (2020)** applies as does the 2020 Provincial Policy Statement (PPS).

The Growth Plan (2020) builds upon the initial Growth Plan (2006) and subsequent policy amendments. The Growth Plan (2020) came into effect on August 28, 2020. All decisions in respect of the exercise of any authority that affects a planning matter are to conform with the Growth Plan. The Growth Plan includes a growth concept as well as a policy framework relating to: "Where and How to Grow"; infrastructure planning; climate change; and the protection of water resources, the natural environment, hydrologic features and areas, public open space, agricultural systems, as well as cultural heritage and aggregate resources. The Growth Plan directs the vast majority of growth to settlement areas and prioritizes intensification both generally and in strategic growth areas as the basis to focus investments in transit, other types of infrastructure and public service facilities.

The policies of the Growth Plan support an integrated approach to land use planning, infrastructure investments, and environmental protection to achieve the outcomes of the Plan, which among other matters supports the achievement of complete communities and ensuring the efficient use of land, and infrastructure. Municipal water and wastewater systems are to be planned, designed, constructed, or expanded to serve growth in a manner that supports achievement of the minimum intensification and density targets of the Growth Plan.

The **2020 Provincial Policy Statement (PPS)** was issued by the Province of Ontario in accordance with Section 3 of the <u>Planning Act</u> and replaces the PPS that was issued on April 30, 2014. The Provincial Policy Statement (PPS) has similar policy thrusts and provides overall policy direction on matters of provincial interest related to land use and development in Ontario. The PPS applies province-wide except where the PPS or another provincial plan provide otherwise.

The PPS provides policy direction on matters of provincial interest related to land use planning and development. All decisions affecting planning matters are to be consistent with the PPS. Similar to the Growth Plan, the PPS is to be read in its entirety and the relevant policies are to be applied to each situation.

The PPS outlines a vision for Ontario's land use planning system which among other matters: focuses growth and development within urban and rural settlement areas; provides for the management of land use change; directs the achievement of appropriate development that meets current and future needs; contemplates efficient development patterns; and provides for the protection of significant or sensitive resources.

Development patterns are to optimize the use of land, resources and public investment in infrastructure and public service facilities. The province's long-term prosperity, environmental health and social well-being depend on wisely managing change and promoting efficient land use and development patterns. There are a number of policies that implement the vision and underlying objectives of the PPS. Some of the relevant policies include:

- Focus growth and development within settlement areas.
- Manage and direct land use to achieve efficient and resilient development and land use patterns.
- Direct the efficient provision of infrastructure and public service facilities, sewage water and stormwater services.

- Require infrastructure and public service facilities to be integrated and coordinated with land use planning.
- Support energy conservation, air quality and consider the impacts of a changing climate.
- Require that provincially significant natural features and areas are to be protected for the long term.
- Direct the protection, improvement or restoration of the quality and quantity of water.

The **Regional Official Plan (ROP)** was approved in June of 2015 and as an implementation of the 2006 Places to Grow – Growth Plan. The ROP is currently being reviewed to implement the 2020 PPS and Growth Plan. An overarching goal of the ROP is to promote balanced growth by directing new growth to occur within the Urban Area and Township Urban Area designations.

The ROP outlines policies to address the Region's infrastructure needs for servicing, stormwater management and transportation. Urban Areas, such as Baden, are a top priority for servicing in the Region. Further priority is given to areas that are currently serviced or can be extended at minimal costs. Full municipal water and sanitary sewage systems are the preferred form of servicing within the Region.

The **Township of Wilmot Official Plan (OP)** was adopted by Township Council on July 21, 2003 and approved with modifications by the Regional Municipality of Waterloo on August 5, 2004 The Township OP is deemed to be consistent with the PPS and implements/conforms to the policies of the Growth Plan and ROP. Similar to the ROP, the OP policies prioritize the utilization and extension of municipal wastewater services to service growth and development.

The proposed EA is consistent with the PPS and conforms with the policies of the Growth Plan, ROP and OP, all of which, support the coordination and integration of planning for infrastructure with growth management and land use planning, and provision of infrastructure in an efficient manner.

PROPOSED ALTERNATIVES

The Broader Study Area extends from Erbs Road to the north towards Gingerich Road and the Baden Wastewater Pumping Station to the south. The Broader Study Area extends from Nafziger Road to the west to Foundry Street, and extends along the railway corridor turning north near Erhhardt Wagner Place and Kropf Drive. The only Alternative which extends into the broader study area is Alternative 1 (do nothing). The detailed study area can be defined as those lands located along Snyder's Road West between Christian Street to the west and Foundry Street to the east. The detailed study area extends south to Gingerich Road and terminates at the Baden Waste Water Pumping Station at 1044 Foundry Street (see Figure 1).

The following table summarizes the land use impacts and considerations relative to each Alternative.

| Alternatives and Descriptions | Land Use Considerations | |
|---|--|--|
| Alternative 1 | • Sewer alignment is on east side of Baden | |
| Modified 2014 Environmental Assessment Preferred | along an existing farm lane, which serves as | |
| Alternative (Do Nothing) | a secondary access to an existing | |
| • This Alternative examines what may happen if no | agricultural/livestock operation. | |
| alternative is selected; | Construction of the sewer would be a | |
| • If no alternatives are selected, then the current | temporary disturbance to the access to the | |
| wastewater strategy of 2014 will be carried out; | agricultural/livestock operation. | |

| This Alternative begins at the north east urban boundary of Baden and runs south west along the railroad and down Foundry Street to the Baden Waste Water Pumping Station. It will also service Snyder's Road West from Foundry to Christian Street; This Alternative would require the following: A pumping station, and a pressure sewer along the eastern Urban Boundary; A sewer which crosses the railroad at the intersection of Foundry Street and Snyder's Road West; and Upgrading existing sewer and integrating it along the Regional Road. | Location of sewer would not provide for servicing of lands to the west of Baden and would not facilitate the Township's long-term vision of growing New Hamburg and Baden together. Alignment of sewer would cross the Goderich-Exeter railway at the intersection of Snyder's Road West and Foundry Street, creating a temporary disruption at a very busy intersection in the community with an at-grade rail-crossing. A number of residential and commercial uses along Foundry Street, including the Township's Fire Hall, would be directly impacted by the temporary construction activities, and require alternative access arrangements. There may be potential disruptions to emergency services. |
|--|--|
| Alternative 2 Snyder's/Foundry Force main Crossing at Snyder's/Foundry Predominantly lies within the existing right-of-way of Snyder's Road West and Foundry Street. This Alternative would require the following: Reconstructing large portions of existing road infrastructure; Crossing the railroad; Upgrading existing sewers; and Upgrading and integrating along a Regional Road. | Alignment of sewer would cross the Goderich-Exeter railway at the intersection of Snyder's Road West and Foundry Street, creating a temporary disruption at a very busy intersection in the community with an at-grade rail-crossing. A number of residential and commercial uses along Foundry Street, including the Township's Fire Hall, would be directly impacted by the temporary construction activities, and require alternative access arrangements. There may be potential disruptions to emergency services. Location and depth of sewer would provide for the servicing of lands to the west of Baden and would help facilitate the Township's long-term vision of growing New Hamburg and Baden together. |
| Alternative 3 Christian/Charlotta/Mill/Charles/Foundry Lies within the existing right-of-ways of Christian Street, Charlotta Street, Mill St., Charles Street and Foundry Street. This Alternative would require the following: Reconstructing large portions of existing road infrastructure; Crossing a railroad; Upgrading existing sewers; and Upgrading and integrating sewers along a Regional Road. | Location and depth of sewer would provide for the servicing of lands to the west of Baden and would help facilitate the Township's long-term vision of growing New Hamburg and Baden together. A number of residential uses on Christian Street, Charlotta Street, Mill Street, Charles Street and Foundry Street and some commercial uses along Foundry Street would be directly impacted by the temporary construction activities, and require alternative access arrangements. Access to the Township's Fire Hall may also |

| Alternative 4 Christian/Charlotta/Queen and Baden Creek Predominantly lies within the existing right ways of Christian Street, Charlotta St., Que Street and Foundry Street. This Alternative would require the following: Reconstructing existing Reinfrastructure; Crossing a railroad; Multiple stream crossings; Constructing new sewers; Upgrading existing sewers; and Upgrading and integrating sewers alor a Regional Road. | be temporarily disrupted which may lead to delays to emergency services response times. Alignment of sewer would cross the Goderich-Exeter railway at Mill Street, where there is an existing at-grade crossing. Some residential uses on Christian Street, Charlotta Street and Queen Street would be directly impacted by the temporary construction activities, and require alternative access arrangements. A portion of the alignment is outside of the settlement area and on agricultural land. Depending on the timing of construction disruptions to cropping or other agricultural activities would need to be mitigated by avoiding construction during the growing season. Following construction, the sewer would not impact agricultural activities/uses. Location and depth of sewer would provide for the servicing of lands to the west of Baden and would help facilitate the Township's long-term vision of growing New Hamburg and Baden together. Alignment of sewer would cross the Goderich-Exeter railway at the terminus of Queen Street, through an open area along the railway, and not impact a railway/road |
|--|---|
| Alternative 5 Christian and Baden Crock | Some residential uses on Christian Street would be directly impacted by the |
| Predominantly lies within the existing right | of temporary construction activities, and |
| ways of Christian Street and Foundry Street. | require alternative access arrangements. |
| Inis Alternative would require the following: Reconstructing existing relation | A portion of the alignment is outside of the settlement area and on agricultural land |
| infrastructure; | Depending on the timing of construction |
| o Crossing a railroad; | disruptions to cropping or other agricultural |
| o A stream crossing; | activities would need to be mitigated by |
| o Upgrading existing sewers; and | season, Following construction, the sewer is |
| Upgrading and integrating sewers alo | ng below grade and would not impact |
| a Regional Road. | agricultural activities/uses. |
| | Location and depth of sewer would provide for the servicing of lands to the west of Baden and would help facilitate the Township's long-term vision of growing New Hamburg and Baden together. |

| • | Alignment of sewer would cross the |
|---|--|
| | Goderich-Exeter railway at the terminus of |
| | Christian Street, where there is an existing |
| | pumping station and open space lands, and |
| | will not impact a railway/road crossing. |

EVALUATION OF ALTNERNATIVES

None of the proposed Alternatives are anticipated to result in the removal of any buildings or structures (other than the Charlotta Pumping Station) or have significant land use planning impacts. Alternatives 1, 2, and 3 are primarily located along right-of-ways and will result in temporary construction disturbances for the various uses fronting the right-of-ways. Alternatives 1 and 2 are the least preferred as they have the most disruption to the downtown core of Baden as their alignments would interrupt the main intersection in the Village (Snyder's Road and Foundry Street). From a broader growth management perspective, Alternative 1 is the least preferred as it would not provide for sanitary servicing to lands west of Baden, where the Township's vision is to grow the communities of Baden and New Hamburg together and where lands are available within the Regional Countryside Line.

Alternatives 4 and 5 would have the least impact on existing land uses. Fewer residents and businesses would be impacted by the construction activity with these Alternatives and the crossing of the railway are in locations that would not disrupt vehicular traffic crossings. Additional measures will need to be taken to mitigate any impacts on the agricultural lands, but in the end, there would be no loss of agricultural lands as a result of the project. Construction will need to take place during periods of time that avoid the loss of crops or disrupt cropping activities (late fall/winter), and soils will need to be properly managed/handled in order to ensure the disturbed lands are properly restored to generally the same preconstruction agricultural soil capability.

Respectfully submitted,

MHBC

Pierre J. Chauvin, BSc(Agr), MA, MCIP, RPP Partner



Figure 1: Sewer Alternatives Overlaid on Township of Wilmot Official Plan (2019): Map 4.1 Baden Urban Area

Township of Baden Township of Wilmot Region of Waterloo

LEGEND

Sewer Alternatives

- Alternative 1: Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing)
- Alternative 2: Snyder's / Foundry (with and without pump station on Activa Lands) ----Force Main Crossing at Snyder's / Foundry
- Alternative 3: Christain/ Charlotta / Mill / Charles / Foundry
- Alternative 4: Christian/ Charlotta/ Queen and Baden Creek
- ---- Alternative 5: Christain and Baden Creek

| - | inter Ana Booties |
|-------|--|
| 1122 | Eriemmentaly Ceretaries Lamb |
| | Island Pality Arrest Schert Factoristic |
| | Johan Cara Anta |
| 1.000 | Lafe balance |
| | Treas Riperio |
| - | MILC: PROBABILI |

Notes:

 For information purposes only
 Source: Parcel data from Region of Waterloo and Teranet Land Information Services and its licensors (2019); Map from Township of Wilmot (2019), Sewer Alternatives from Township of Wilmot (2022)



DRAWN: CCF

K:\0941CD - BADEN TRUNK SEWER EA\RPT\LANDUSE_MAP4_1_02MAY2022.DWG







Figure 2: Sewer Alternatives Overlaid on Township of Wilmot Official Plan (2019): Map 10 The Countryside

Township of Baden Township of Wilmot Region of Waterloo

LEGEND

Sewer Alternatives

- Alternative 1: Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing)
- Alternative 2: Snyder's / Foundry (with and without pump station on Activa Lands) Force Main Crossing at Snyder's / Foundry
- Alternative 3: Christain/ Charlotta / Mill / Charles / ----Foundry
- Alternative 4: Christian/ Charlotta/ Queen and Baden Creek
- Alternative 5: Christain and Baden Creek

| | Courseseeting |
|-----|---------------------------------|
| 382 | Partnerind Countyrailer |
| | Paral Areas? |
| | Persilipitukat |
| | What train (Third Dettemorities |

Notes:

 For information purposes only
 Source: Parcel data from Region of Waterloo and Teranet Land Information Services and its licensors (2019); Map from Township of Wilmot (2019), Sewer Alternatives from Township of Wilmot (2022)

DATE: May 2, 2022 FILE: 0941CD **SCALE:** 1: 10,000

DRAWN: CCF

K:\0941CD - BADEN TRUNK SEWER EA\RPT\LANDUSE_MAP10_02MAY2022.DWG







Stage 1-2 Archaeological Assessment Beckdale Sanitary Sewer Replacement, Baden

(November 8, 2022)

Supplementary Documentation Stage 1-2 Archaeological Assessment Beckdale Sanitary Sewer Replacement, Baden

(November 8, 2022)



Stage 1-2 Archaeological Assessment Baden Sanitary Sewer Replacement, Baden

Part of Lots 15 and 16, German Block South of Snyder Road; Part of Lots 15 and 16, German Block North of Snyder Road; and Part of the Snyder Road Allowance between Lot 16 German Block South of Snyder Road and Lot 16 German Block North of Snyder Road, Geographic Township of Wilmot, Historical County of Waterloo, now the Regional Municipality of Waterloo, Ontario

Submitted to:

MTE Consultants Inc. 520 Bingeman's Centre Drive, Kitchener, ON N2B 3X9

and

Ontario's Ministry of Heritage, Sport, Tourism and Culture Industries

Submitted by:



196 Westhheights Drive, Kitchener Ontario, N2N 1J9 Mobile/Office: 519-744-7018 www.detritusconsulting.ca

Licensee: Walter McCall License Number: P389 PIF Number: P389-0551-2021 CP Number: 2021-048

ORIGINAL REPORT

November 8, 2022

Executive Summary

Detritus Consulting Ltd. was retained by MTE Consultants Inc. (the 'Proponent') to conduct a Stage 1-2 archaeological assessment on part of Lots 15 and 16, German Block South of Snyder Road, and part of the Snyder Road Allowance between Lot 16 South of Snyder Road and Lot 16 North of Snyder Road within the within the Geographic Township of Wilmot and historical County of Waterloo, now the Regional Municipality of Waterloo, Ontario (Figure 1). This investigation was conducted in advance of the proposed Baden sanitary sewer replacement along a winding, narrow corridor (the 'Study Area') running north of Michael Myers Road in the northwest to the Baden Wastewater Pumping Station at 1044 Foundry Street to the southeast within the community of Baden (Figure 3).

An archaeological assessment was triggered by the Provincial Policy Statement ('PPS') that is informed by the *Planning Act* (Government of Ontario 1990a), which states that decisions affecting planning matters must be consistent with the policies outlined in the larger *Ontario Heritage Act* (Government of Ontario 1990b). According to Section 2.6.2 of the PPS, "development and site alteration shall not be permitted on lands containing archaeological resources or areas of archaeological potential unless significant archaeological resources have been conserved." To meet this condition, a Stage 1-2 assessment of the Study Area was conducted during the pre-approval phase of the development under archaeological consulting license P389 issued to Dr. Walter McCall by the Ministry of Heritage, Sport, Tourism and Culture Industries ('MHSTCI') and adheres to the archaeological license report requirements under subsection 65 (1) of the *Ontario Heritage Act* (Government of Ontario 1990b) and the MHSTCI's *Standards and Guidelines for Consultant Archaeologists* ('*Standards and Guidelines*'; Government of Ontario 2011a).

The Study Area is a long, narrow corridor measuring approximately 2.9 kilometres ('km') long and 10m wide (Figure 3 and Tile 3 in the Supplementary Documentation). The northwestern end begins approximately 154 metres ('m') to the northwest of the residential lot at 105 Michael Myers Road, in the southwestern corner of the large unassigned parcel to the southeast of the Foxwood Country Club. The corridor turns sharply to the southwest before accessing the large adjacent agricultural property at 1012 Snyder's Road West. It then travels the length of the property until it meets Snyder's Road West to the southeast. At this point, the Study Area crosses over Snyder's Road West and carries on for the length of Christian Street. It then turns to the southwest to follow Charlotta Street for approximately 35m, before veering again to the southwest to cross over the small unassigned parcel to the northeast of 193 Charlotta Street, and the Canadian National Railway line. From here, the Study Area winds a path to Gingerich Road, generally following the course of Baden Creek, which traverses the unassigned agricultural land to the southwest of Foundry Street. At Gingerich Road it turns to the northeast to meet Foundry Street. Finally, the Study Area extends down Foundry Street for 228m until it meets the Baden Wastewater Pumping Station at 1044 Foundry Street.

Most of the Study Area consists of either agricultural land that was accessible to ploughing or overgrown grass to the west of Baden Creek. The southeastern end of 1012 Snyder's Road West featured a small swathe of manicured lawn between the houses at 182 and 186 Snyder's Road West. Another small grassy area was observed between Charlotta Street and the Canadian National Railway line. Finally, an area of woodlot was encountered at the bend in the Study Area, where it veers to the northeast to approach Foundry Street. The remainder of the Study Area included all or portions of the Snyder's Road West, Christian Street, Charlotta Street, Gingerich Road, and Foundry Street road allowances; the Canadian National Railway railbed; and the paved driveway of the Baden Wastewater Pumping Station. Baden Creek transects the southeastern end of the Study Area from southeast to northwest adjacent to Foundry Street and under the driveway of the Baden Wastewater Pumping Station.

The Stage 1 background research indicated that the Study Area exhibited moderate to high potential for the identification and recovery of archaeological resources. This investigation also revealed that portions of the Study Area overlap areas that were subject to a previous Stage 1-2 assessment conducted by Detritus in 2020 (Detritus 2020). This earlier investigation was conducted on two large irregular parcels of primarily agricultural land located to the southwest of

Foundry Street and to the northwest of Gingerich Road. The two parcels were generally separated by Baden Creek.

This investigation resulted in the documentation of one pre-contact Aboriginal site, AiHe-469, as well as 19 pre-contact Aboriginal findspots, and one Euro Canadian findspot (Tile 3 of the Supplementary Documentation). Seven of the pre-contact Aboriginal findspots were located within 50m of the current Study Area, including five to the northwest of the long stretch running parallel to Gingerich Road and another two to the southeast.

Among the latter group, Findspot 16 (AiHd-178) extends into the current Study Area (Tile 3 of the Supplementary Documentation). Findspot 16 (AiHd-178) comprised a single chert flake and a projectile point fragment, both manufactured from Onondaga chert. Although fragmentary, the point fragment was identified as a Late Archaic Crawford Knoll variety. Findspot 18 (AiHe-467) to the northeast of the Study Area, meanwhile, comprised an isolated projectile point fragment that shared characteristics common among varieties from the Late Paleo-Indian period. Both sites were recommended for a Stage 3 assessments, neither of which had been completed at the time of this investigation. Although Findspot 18 (AiHe-467) does not extend into the current Study Area, a portion of its surrounding 20m protective buffer overlaps it.

The remaining findspots that were identified within 50m of the Study Area yielded isolated chert flakes, including eight from Findspot 9, Findspot 10, Findspot 12, and Findspot 19 (AiHd-177) to the northwest of the Study Area corridor, and another four from Findspot 11 (AiHe-470) to the southwest.

The Stage 2 field survey of the Study Area was conducted between August 16th and 18th, 2021 and on May 26, 2022. This investigation consisted of a pedestrian survey at a 3m interval of most of the agricultural land throughout the Study Area, and a test pit assessment of the grassy and treed areas, and the seasonal creek extending from Baden Creek to the west. Prior to the completion of the Stage 2 assessment, the Proponent adjusted the development area to exclude the portions of the Study Area corridor to the northwest of Snyder's Road West.

The various road allowances throughout the Study Area, the Canadian National Railway bed, and the pumping house driveway were evaluated as having no potential based on the identification of extensive and deep land alteration that has severely damaged the integrity of archaeological resources, as per Section 2.1, Standard 2b of the Standards and Guidelines (Government of Ontario 2011a). As mentioned above Baden Creek transects the southeastern portion of the Study Area. This area was determined to retain no archaeological potential due to the identification of physical features of low archaeological potential, including a permanently wet area, as per Section 2.1, Standards 2.a.i of the Standards and Guidelines (Government of Ontario 2011a). Additionally, a grass area on the southern side of the driveway to the Baden Wastewater Pumping Station was determined to be steeply sloping. This area was determined to retain no archaeological potential due to the identification of physical features of low archaeological potential, including steep slopes greater than 20 degrees, as per Section 2.1, Standards 2.a.iii of the Standards and Guidelines (Government of Ontario 2011a). The previously disturbed, permanently wet, and steeply sloped areas, as confirmed during a Stage 2 property inspection, were mapped and photo documented in accordance with Section 2.1, Standard 6 and Section 7.8.1, Standards 1a and 1b of the Standards and Guidelines (Government of Ontario 2011a).

The Stage 2 field survey resulted in the documentation of two pre-contact Aboriginal findspots: Findspot 1 and Findspot 2. Findspot 1 is an Onondaga chert thinning flake recovered during the test pit assessment of the lawn area between 182 and 186 Snyder's Road West prior to its removal from the development area. Although eight cardinal test pits were excavated around the positive test pit, all of which were sterile, no 1m unit was excavated over it. Findspot 2, meanwhile, comprises one Onondaga chert thinning flake. The eight cardinal test pits and a single 1m test unit yielded no additional artifacts.

Based on the available evidence, Findspot 1 and Findspot 2 have been identified as isolated pieces of debitage that likely originated from one of the larger sites or findspots identified during the earlier Stage 2 assessment of the agricultural land located to the southwest of Foundry Street and to the northwest of Gingerich Road.

The Stage 2 assessment of Findspot 1, however, remains incomplete. Therefore, **Findspot 1 is recommended for additional Stage 2 investigation** in the form of a 1m metre test unit, excavated above the positive test unit, as per Section 2.1.3, Section 2a of the *Standards and Guidelines* (Government of Ontario 2011a). This test unit will be excavated by hand in systematic levels and into the first five centimetres ('cm') of subsoil as per Section 3.2, Standards 3 to 5 of the *Standards and Guidelines* (Government of Ontario 2011a).

Given the isolated nature of the non-diagnostic artifact, Findspot 2, which was subject to full intensification, does not fulfill any of the criteria for a Stage 3 site specific assessment as per Section 2.2 of the *Standards and Guidelines* (Government of Ontario 2011a); therefore **Findspot 2 is not recommended for a Stage 3 assessment**.

Additionally, according to the results of the previous Stage 1-2 assessment conducted by Detritus in 2020, the portion of Findspot 16 (AiHd-178) that extends into the current Study Area is recommended for Stage 3 assessment (Detritus 2020). Likewise, the portion of the 20m protective buffer surrounding Findspot 18 (AiHe-467) that intersects the Study Area is also recommended for Stage 3 assessment (Detritus 2020).

Typically, a Stage 3 assessment begins with an intensive controlled surface pickup ('CSP') across the Stage 2 limits of the sites, conducted as per Section 3.2.1 of the *Standards and Guidelines* (Government of Ontario 2011a). The Stage 2 pedestrian survey at the sites, however, consisted of an intensive surface collection across the entire limits of each site. All artifacts were mapped digitally and collected for laboratory analysis. Thus, the conditions for a Stage 3 CSP were met during the Stage 2 assessment.

Given that it is not yet evident that the level of cultural heritage value or interest will result in a recommendation to proceed to Stage 4, the Stage 3 assessments at Findspot 16 (AiHd-178) and the protective buffer around Findspot 18 (AiHe-467) will consist of the hand excavation of 1m square test units every 5m in systematic levels and into the first 5cm of subsoil as per Table 3.1, Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a). Additional 1m test units, amounting to 20% of the grid total, will be placed in areas of interest within the site extent as per Table 3.1, Standard 2 of the *Standards and Guidelines* (Government of Ontario 2011a). All excavated soil will be screened through six-millimetre mesh; all recovered artifacts will be recorded by their corresponding grid unit designation and collected for laboratory analysis. If a subsurface cultural feature is encountered, the plan of the exposed feature will be recorded and geotextile fabric will be placed over the unit before backfilling the unit.

Additionally, **the Stage 1 investigation revealed that portions of the Study Area exhibited moderate to high potential for the identification and recovery of archaeological resources. These portions corresponded with the agricultural field, and the grass area surrounding the silo to the northwest of Snyder's Road West, which were not assessed as part of the Stage 2 assessment. All active agricultural land that retains archaeological potential and is accessible for ploughing will be subject to a typical Stage 2 pedestrian survey at a 5m interval**, conducted according to Section 2.1.1 of the *Standards and Guidelines* (Government of Ontario 2011a). The fields will be ploughed until 80% surface visibility is attained, then allowed to weather prior to assessment. As per Section 2.1.1, Standard 7 of the *Standards and Guidelines* (Government of Ontario 2011a), if archaeological resources are found, the survey transects will be decreased to 1m intervals over a 20m radius around each find to determine whether it is an isolated find or part of a larger scatter. All formal artifact types and diagnostic categories will be collected for laboratory analysis and cataloguing, including all refined ceramic sherds for 19th century archaeological sites.

The grass area, which is inaccessible for ploughing, will be subject to a typical Stage 2 test pit assessment at a 5m interval, conducted according to Section 2.1.2 of the *Standards and Guidelines* (Government of Ontario 2011a). The test pit survey will be conducted until test pits show evidence of disturbance according to Section 2.1.2, Standard 4 of the *Standards and Guidelines* (Government of Ontario 2011a). Each test pit must be approximately 30 centimetres ('cm') in diameter and excavated 5cm into sterile subsoil. The soil and test pits will then be examined for stratigraphy, cultural features, or evidence of fill. All soil will be screened through

six-millimetre mesh hardware cloth to facilitate the recovery of small artifacts and then used to backfill the pit.

In accordance with Section 2.1.3 Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a), if archaeological resources are encountered during the Stage 2 test pit survey, the test pit excavation will continue on the survey grid to determine the extent of further positive test pits. If insufficient archaeological resources are found through a continued survey of the grid to meet the criteria for continuing to Stage 3, the survey coverage will be intensified around the positive test pits using either Option A or Option B of Section 2.1.3, Standard 2 of the *Standards and Guidelines* (Government of Ontario 2011a). Universal Transverse Mercator ('UTM') coordinates will then be recorded for all positive test pit in addition to a fixed reference landmark using a Garmin eTrex 10 GPS unit with a minimum accuracy 1-2.5m (North American Datum 1983 ('NAD83') and UTM Zone 17T). All artifacts will be collected and recorded according to their associated positive test pit or 1m test unit.

Lastly, if any of the areas recommended for Stage 2 assessment are determined to be previously disturbed, or permanently wet during the course of the Stage 2 assessment, they will be photo documented as per Section 2.1, Sections 2a and 2b and Section 6 of the *Standards and Guidelines* (Government of Ontario 2011a).

The Executive Summary highlights key points from the report only; for complete information and findings, the reader should examine the complete report.

Table of Contents

| Executive Summary | ii | |
|---|------|--|
| 1.0 Project Context | 1 | |
| 1.1 Development Context | 1 | |
| 1.2 Historical Context | 2 | |
| 1.2.1 Post-Contact Aboriginal Resources | 2 | |
| 1.2.2 Euro-Canadian Resources | 3 | |
| 1.3 Archaeological Context | 4 | |
| 1.3.1 Property Description and Physical Setting | 4 | |
| 1.3.2 Pre-Contact Aboriginal Land Use | 5 | |
| 1.3.3 Previous Identified Archaeological Work | 5 | |
| 1.3.4 Archaeological Potential | 8 | |
| 2.0 Field Methods | .10 | |
| 3.0 Record of Finds | . 12 | |
| 3.1 Cultural Material | . 12 | |
| 3.2 Findspot 1 | . 13 | |
| 3.3 Findspot 2 | . 13 | |
| 4.0 Analysis and Conclusions | .14 | |
| 5.0 Recommendations | . 15 | |
| 6.0 Advice on Compliance with Legislation | .16 | |
| 7.0 Bibliography and Sources | . 17 | |
| 8.0 Maps | . 19 | |
| 9.0 Images | .27 | |
| 9.1 Field Photos27 | | |
| 9.2 Artifact Photos | | |

Project Personnel

| Garth Grimes P017 |
|--|
| Jon Cousins, R296 |
| Melanie Chan, Eric Cousins, Rob Brown, Jemmett Kirk; Isaac Bender, R1325 Ryan Eden, R1302; Michaela Todd; Tim Wenn |
| Eric Laforme (Mississaugas of the Credit First Nation) Jessica Williams, Steven Markle, Owen Greene, Danton Miller, Wayne Hill (Haudenosaunee Development Institute) Davis Jacobs (Six Nations of the Grand River) |
| Walter McCall, P389; Amanda McCall, R470 |
| Amanda McCall, R470 |
| Garth Grimes P017 |
| |

Acknowledgments

Generous contributions by the following individuals and agencies made this report possible.

• Chelsea Kochany, MTE Consultants Inc.

1.0 Project Context

1.1 Development Context

Detritus Consulting Ltd. was retained by MTE Consultants Inc. (the 'Proponent') to conduct a Stage 1-2 archaeological assessment on part of Lots 15 and 16, German Block South of Snyder Road, and part of the Snyder Road Allowance between Lot 16 South of Snyder Road and Lot 16 North of Snyder Road within the within the Geographic Township of Wilmot and historical County of Waterloo, now the Regional Municipality of Waterloo, Ontario (Figure 1). This investigation was conducted in advance of the proposed Baden sanitary sewer replacement along a winding, narrow corridor (the 'Study Area') running north of Michael Myers Road in the northwest to the Baden Wastewater Pumping Station at 1044 Foundry Street to the southeast within the community of Baden (Figure 3).

An archaeological assessment was triggered by the Provincial Policy Statement ('PPS') that is informed by the *Planning Act* (Government of Ontario 1990a), which states that decisions affecting planning matters must be consistent with the policies outlined in the larger *Ontario Heritage Act* (Government of Ontario 1990b). According to Section 2.6.2 of the PPS, "development and site alteration shall not be permitted on lands containing archaeological resources or areas of archaeological potential unless significant archaeological resources have been conserved." To meet this condition, a Stage 1-2 assessment of the Study Area was conducted during the pre-approval phase of the development under archaeological consulting license P389 issued to Dr. Walter McCall by the Ministry of Heritage, Sport, Tourism and Culture Industries ('MHSTCI') and adheres to the archaeological license report requirements under subsection 65 (1) of the *Ontario Heritage Act* (Government of Ontario 1990b) and the MHSTCI's *Standards and Guidelines for Consultant Archaeologists* ('*Standards and Guidelines*'; Government of Ontario 2011a).

The purpose of the Stage 1 assessment is to compile all available information about the known and potential archaeological heritage resources within the Study Area and to provide specific direction for the protection, management and/or recovery of these resources. In compliance with the *Standards and Guidelines* (Government of Ontario 2011a), the objectives of the following Stage 1 assessment were as follows:

- To provide information about the Study Area's geography, history, previous archaeological fieldwork and current land conditions;
- to evaluate in detail, the Study Area's archaeological potential which will support recommendations for Stage 2 survey for all or parts of the property; and
- to recommend appropriate strategies for Stage 2 survey.

To meet these objectives Detritus archaeologists employed the following research strategies:

- A review of relevant archaeological, historic and environmental literature pertaining to the Study Area;
- a review of the land use history, including pertinent historic maps; and
- an examination of the Ontario Archaeological Sites Database ('ASDB') to determine the presence of known archaeological sites in and around the Study Area.

The purpose of the Stage 2 assessment is to provide an overview of any archaeological resources within the Study Area, and to determine whether any of the resources might be archaeological sites with cultural heritage value or interest ('CHVI'), and to provide specific direction for the protection, management and/or recovery of these resources. In compliance with the *Standards and Guidelines* (Government of Ontario 2011a), the objectives of the following Stage 2 Property Assessment were as follows:

- To document all archaeological resources within the Study Area;
- to determine whether the Study Area contains archaeological resources requiring further assessment; and
- to recommend appropriate Stage 3 assessment strategies for archaeological sites identified.
The licensee received permission from the Proponent to enter the land and conduct all required archaeological fieldwork activities, including the recovery of artifacts.

1.2 Historical Context

1.2.1 Post-Contact Aboriginal Resources

Prior to the arrival of European settlers, much of the central and southern Ontario was occupied by Iroquoian speaking linguistic groups that had united to form confederacies, including the Huron-Wendat, the Neutral (or Attawandaran), and the Petun in Ontario, as well as the Five Nations Iroquois Confederacy in Upper New York State (Birch 2010; Warrick 2013). Of these groups, the Huron-Wendat established themselves to the east of the Niagara escarpment and the Neutral, to the west (Warrick 2000).

Throughout the middle of the 17th century, the Iroquois Confederacy sought to expand upon their territory and to monopolize the fur trade between the European markets and the tribes of the western Great Lakes region. A series of bloody conflicts followed known as the Beaver Wars or the French and Iroquois Wars, contested between the Iroquois Confederacy and the Algonkian speaking communities of the Great Lakes region. Many communities were destroyed including the Huron, Neutral, Susquehannock and Shawnee leaving the Iroquois as the dominant group in the region. By 1653 after repeated attacks, the Niagara peninsula and most of Southern Ontario had been vacated (Heidenreich 1990).

At this same time, the Anishinaabeg Nation, an Algonkian-speaking community situated inland from the northern shore of Lake Huron, began to challenge the Haudenosaunee for dominance in the Lake Huron and Georgian Bay region in order to advance their own role in the fur trade (Gibson 2006). The Algonkian-speaking groups that settled in the area bound by Lake Ontario, Lake Erie, and Lake Huron were referred to by the English as the Chippewas or Ojibwas. By 1680, the Ojibwa began expanding into the evacuated Huron-Wendat territory, and eventually into Southern Ontario. By 1701, the Haudenosaunee had been driven out of Ontario completely and were replaced by the Ojibwa (Gibson 2006; Schmaltz 1991).

The late 17th and early 18th centuries also mark the arrival of an Ojibwa band known as the Mississaugas into Southern Ontario and, in particular, the watersheds of the lower Great Lakes. 'The Mississaugas' is the name that the Jesuits had used in 1840 for the Algonquin community living near the Mississaugas, as recounted by Chief Robert Paudash and recorded in 1904, suggest that the Mississaugas defeated the Mohawk Nation, who retreated to their homeland south of Lake Ontario. Following this conflict, a peace treaty was negotiated between the two groups (Praxis Research Associates n.d.).

From the beginning of the 18th century until the end of the Seven Year War in 1763, the Ojibwa nation, including the Mississaugas, experienced a golden age in trade holding no alliance with either the French or the British (Schmaltz 1991). At the end of the 17th century, the Mississaugas' settled permanently in Southern Ontario (Praxis Research Associates n.d.). Around this same time, in 1722, the Five Nation Iroquois Confederacy adopted the Tuscarora in New York becoming the Six Nations (Pendergast 1995).

The Study Area first entered the Euro-Canadian historical record on December 7th, 1792 as part of Treaty No. 3, which included land acquired in the 'Between the Lakes Purchase' dating to May 22, 1784. According to the terms of the treaty, the Mississaugas ceded to the Crown approximately 3,000,000 acres of land between Lake Huron, Lake Erie, and Lake Ontario in return for trade goods valued at £1180.

The limits of the Treaty 3 lands are documented as comprising,

Lincoln County excepting Niagara Township; Saltfleet, Binbrook, Barton, Glanford and Ancaster Townships, in Wentworth County; Brantford, Onondaga, Tusc[a]r[o]ra, Oakland and Burford Townships in Brant County; East and West Oxford, North and South Norwich, and Dereham Townships in Oxford County; North Dorchester Township in Middlesex County; South Dorchester, Malahide

Detritus Consulting Ltd.

and Bayham Township in Elgin County; all Norfolk and Haldimand Counties; Pelham, Wainfleet, Thorold, Cumberland and Humberstone Townships in Welland County.

Morris 1943: 17-18

One of the stated objectives of the Between the Lakes Purchase was "to procure for that part of the Six Nation Indians coming into Canada a permanent abode" (Morris 1943: 17). Shortly after the transaction had been finalised in May of 1784, Sir Frederick Haldimand, the Governor of Québec, made preparations to grant a portion of land to those Six Nations who remained loyal to the Crown during the American War of Independence. More specifically, Haldimand arranged for the purchase of approximately 550,000 acres of land adjacent to the Treaty 3 limits from the Mississaugas. This tract of land, referred to as either the Haldimand Tract or the 1795 Crown Grant to the Six Nations, was provided for in the Haldimand Proclamation of October 25th, 1784 and was intended to extend a distance of six miles on each side of the Grand River from mouth to source (Weaver 1978). By the end of 1784, representatives from each constituent nation of the Six Nations, as well as other allies, relocated to the Haldimand Tract with Joseph Brant (Weaver 1978; Tanner 1987).

Throughout southern Ontario, the size and nature of the pre-contact settlements and the subsequent spread and distribution of Aboriginal material culture began to shift with the establishment of European settlers. By 1834 it was accepted by the Crown that losses of portions of the Haldimand Tract to Euro-Canadian settlers were too numerous for all lands to be returned. Lands in the Lower Grand River area were surrendered by the Six Nations to the British Government in 1832, at which point most Six Nations people moved into Tuscarora Township in Brant County and a narrow portion of Oneida Township (Page & Co. 1879; Weaver 1978; Tanner 1987). Following the population decline and the surrender of most of their lands along the Credit River, the Mississaugas were given 6000 acres of land on the Six Nations Reserve, establishing the Mississaugas of New Credit First Nation, now the Mississaugas of the Credit First Nation ('MCFN'), in 1847 (Smith 2002).

Despite the encroachment of European settlers on previously established Aboriginal territories, "written accounts of material life and livelihood, the correlation of historically recorded villages to their archaeological manifestations, and the similarities of those sites to more ancient sites have revealed an antiquity to documented cultural expressions that confirms a deep historical continuity to Iroquoian systems of ideology and thought" (Ferris 2009: 114). As Ferris observes, despite the arrival of a competing culture, First Nations communities throughout Southern Ontario have left behind archaeologically significant resources that demonstrate continuity with their pre-contact predecessors, even if they have not been recorded extensively in historical Euro-Canadian documentation.

1.2.2 Euro-Canadian Resources

The Study Area is located in the Geographic Township of Wilmot in the Regional Municipality of Waterloo Ontario.

On July 24, 1788, Sir Guy Carleton, the Governor-General of British North America, divided the Province of Québec into the administrative districts of Hesse, Nassau, Mecklenburg and Lunenburg (Archives of Ontario 2012-2015). Further change came in December 1791 when the Province of Québec was rearranged into Upper Canada and Lower Canada under the Constitutional Act. Colonel John Graves Simcoe was appointed as Lieutenant-Governor of Upper Canada; he initiated several initiatives to populate the province including the establishment of shoreline communities with effective transportation links between them (Coyne 1895:33).

In July 1792, Simcoe divided Upper Canada into 19 counties, including Waterloo County, stretching from Essex in the west to Glengarry in the east. Later that year, the four districts originally established in 1788 were renamed as the Western, Home, Midland and Eastern Districts (Archives of Ontario 2012-2015).

The township of Wilmot was surveyed in 1842. Mennonites from Waterloo County and the Amish from Europe began claiming lots and clearing roadways and farms. The first three roads of the

township were Upper Street, Middle Street, and Lower Street, now referred to as Erb's Road, Snyder's Road and Bleams Road. Most early settlers of the area were of German descent, although some Scottish, English and Irish were also amongst the first settlers. The village of Baden was founded and established in 1855 and was originally names Weissenburg, after a bachelor living in the area. Baden is the home of Adam Beck, the founder of Ontario's public hydro-electric system (The Corporation of the Township of Wilmot 2015).

The *Illustrated Historical Atlas of Waterloo and Wellington Counties* ('*Historical Atlas*'), demonstrates the extent to which Wilmot Township had been settled by 1877 (Walker & Miles 1877; Figure 2). An increasing population throughout the late 19th century is evident from the number of villages and small towns indicated, the town lots for the early communities of Baden, New Hamburg, and Petersburg, as well as a branch of the Grand Trunk Railway, which transect the township from east to west.

The *Historical Atlas* map of Wilmot lists no landowners and illustrates no structures for Lots 15 and 16 German Block South of Snyder Road or for Lots 15 and 16 German Block North of Snyder Road. Additionally, the early community of Baden is illustrated to the east and northeast of the Study Area. The Grand Trunk Railway is illustrated transecting the Study Area from southwest to northeast, connecting the early communities of Baden and New Hamburg, which is located to the southwest of the Study Area. It should be recognized that historical county atlases were funded by subscriptions fees and were produced primarily to identify factories, offices, residences and landholdings of subscribers. Landowners who did not subscribe were not always listed on the maps (Caston 1997:100). Moreover, associated structures were not necessarily depicted or placed accurately (Gentilcore and Head 1984).

1.3 Archaeological Context

1.3.1 Property Description and Physical Setting

The Study Area is a long, narrow corridor measuring approximately 2.9 kilometres ('km') long and 10m wide (Figure 3 and Tile 3 in the Supplementary Documentation). The northwestern end begins approximately 154 metres ('m') to the northwest of the residential lot at 105 Michael Myers Road, in the southwestern corner of the large unassigned parcel to the southeast of the Foxwood Country Club. The corridor turns sharply to the southwest before accessing the large adjacent agricultural property at 1012 Snyder's Road West. It then travels the length of the property until it meets Snyder's Road West to the southeast. At this point, the Study Area crosses over Snyder's Road West and carries on for the length of Christian Street. It then turns to the southwest to follow Charlotta Street for approximately 35m, before veering again to the southwest to cross over the small unassigned parcel to the northeast of 193 Charlotta Street, and the Canadian National Railway line. From here, the Study Area winds a path to Gingerich Road, generally following the course of Baden Creek, which traverses the unassigned agricultural land to the southwest of Foundry Street. At Gingerich Road it turns to the northeast to meet Foundry Street. Finally, the Study Area extends down Foundry Street for 228m until it meets the Baden Wastewater Pumping Station at 1044 Foundry Street.

Most of the Study Area consists of either agricultural land that was accessible to ploughing or overgrown grass to the west of Baden Creek. The southeastern end of 1012 Snyder's Road West featured a small swathe of manicured lawn between the houses at 182 and 186 Snyder's Road West. Another small grassy area was observed between Charlotta Street and the Canadian National Railway line. Finally, an area of woodlot was encountered at the bend in the Study Area, where it veers to the northeast to approach Foundry Street. The remainder of the Study Area included all or portions of the Snyder's Road West, Christian Street, Charlotta Street, Gingerich Road, and Foundry Street road allowances; the Canadian National Railway railbed; and the paved driveway of the Baden Wastewater Pumping Station.

The majority of the region surrounding the Study Area has been subject to European-style agricultural practices for over 100 years, having been settled by Euro-Canadian farmers by the mid-19th century. Much of the region today continues to be used for agricultural purposes.

The Study Area is located within the Waterloo Hills physiographic region (Chapman and Putnam 1984). This is a region of sandy hills and ridges broken by the alluvial terrace system of the Grand River spillway.

The closest source of potable water is Baden Creek, which runs adjacent to the northeast of the Study Area and transects the southeastern end of the Study Area.

1.3.2 Pre-Contact Aboriginal Land Use

This portion of Southwestern Ontario was occupied by people as far back as 11,000 years ago as the glaciers retreated. For the majority of this time, people were practicing hunter gatherer lifestyles with a gradual move towards more extensive farming practices. Table 1 provides a general outline of the cultural chronology of Wilmot Township (Ellis and Ferris 1990).

| Time Period | Cultural Period | Comments |
|-----------------|-------------------------------------|--|
| 9500 – 7000 BC | Paleo-Indian | first human occupation hunters of caribou and other extinct Pleistocene game nomadic, small band society |
| 7500 - 1000 BC | Archaic | ceremonial burials increasing trade network hunter gatherers |
| 1000 - 400 BC | Early Woodland | large and small camps spring congregation/fall dispersal introduction of pottery |
| 400 BC – AD 800 | Middle Woodland | kinship based political system incipient horticulture long distance trade network |
| AD 800 - 1300 | Early Iroquoian (Late Woodland) | limited agriculture developing hamlets and villages |
| AD 1300 - 1400 | Middle Iroquoian (Late Woodland) | shift to agriculture complete increasing political complexity large palisaded villages |
| AD 1400 - 1650 | Late Iroquoian | regional warfare and political/tribal alliances destruction of Huron and Neutral |

Table 1: Cultural Chronology for the Wilmot Township

1.3.3 Previous Identified Archaeological Work

In order to compile an inventory of archaeological resources, the registered archaeological site records were consulted. In Ontario, information concerning archaeological sites stored in the ASDB (Government of Ontario n.d.) is maintained by the MHSTCI. This database contains archaeological sites registered according to the Borden system. Under the Borden system, Canada is divided into grid blocks based on latitude and longitude. A Borden Block is approximately 13km east to west and approximately 18.5km north to south. Each Borden Block is referenced by a four-letter designator and sites within a block are numbered sequentially as they are found. The Study Area under review is within Borden Blocks AiHe and AiHd.

Information concerning specific site locations is protected by provincial policy, and is not fully subject to the *Freedom of Information and Protection of Privacy Act* (Government of Ontario 1990c). The release of such information in the past has led to looting or various forms of illegally conducted site destruction. Confidentiality extends to all media capable of conveying location, including maps, drawings, or textual descriptions of a site location. The MHSTCI will provide information concerning site location to the party or an agent of the party holding title to a property, or to a licensed archaeologist with relevant cultural resource management interests.

An examination of the ASDB has shown that there are 26 archaeological sites registered within a 1km radius of the Study Area (Table 2). Of the 26 sites, 19 are pre-contact Aboriginal, 2 of which date to the Early Archaic period, and 6 are Euro-Canadian. No information is available for the remaining site.

| Borden | | | | |
|----------|-----------------|-----------------------------------|---------------|--------------------------|
| Number | Site Name | Time Period | Affinity | Site Type |
| AiHe-38 | Castle Kilbride | Post-Contact | Euro-Canadian | farmstead |
| AiHe-44 | AiHe-44 | Archaic, Early | Aboriginal | findspot |
| AiHe-45 | | Pre-Contact | Aboriginal | scatter |
| AiHe-46 | | Post-Contact | Euro-Canadian | homestead, midden |
| AiHe-47 | | Post-Contact | Euro-Canadian | privy, homestead, midden |
| AiHe-463 | | Archaic | Aboriginal | findspot |
| AiHe-464 | | Woodland, Early | Aboriginal | camp / campsite |
| AiHe-465 | | Archaic, Late, Archaic, Middle | Aboriginal | camp / campsite |
| AiHe-466 | | Woodland, Early | Aboriginal | camp / campsite |
| AiHe-467 | Kwaya-si | Paleo-Indian, Late | Aboriginal | findspot |
| AiHe-468 | | Archaic, Middle | Aboriginal | hunting loss |
| AiHe-469 | Asyeh | Archaic, Late, Woodland, Late | Aboriginal | short term |
| AiHe-470 | | Pre-Contact | Aboriginal | scatter |
| AiHe-471 | | Post-Contact | Euro-Canadian | dump |
| AiHe-472 | | Archaic, Early | Aboriginal | scatter |
| AiHe-473 | | Archaic, Early | Aboriginal | scatter |
| AiHe-474 | | Pre-Contact | Aboriginal | scatter |
| AiHd-17 | Spongy Lake | Woodland | Aboriginal | camp/campsite |
| AiHd-2 | Baden | - | - | - |
| AiHd-20 | Wolf | Woodland, Late | Iroquoian | village |
| AiHd-69 | Schout | Pre-Contact | Aboriginal | - |
| AiHd-91 | Baden Brewery | Post-Contact | Euro-Canadian | brewery, creamery |
| AiHd-110 | Michel | Archaic, Early | Aboriginal | findspot |
| AiHd-162 | | Post-Contact | Euro-Canadian | dump |
| AiHd-177 | | Pre-Contact | Aboriginal | scatter |
| AiHd-178 | Skiwa | Archaic, Late | Aboriginal | findspot |

Table 2: Registered Archaeological Sites within 1km of the Study Area

Portions of the Study Area overlap areas that were subject to a previous Stage 1-2 assessment conducted by Detritus in 2020 (Detritus 2020). This earlier investigation was conducted on two large irregular parcels of primarily agricultural land located to the southwest of Foundry Street and to the northwest of Gingerich Road. The two parcels were generally separated by Baden Creek.

This investigation resulted in the documentation of one pre-contact Aboriginal site, AiHe-469, as well as 19 pre-contact Aboriginal findspots, and one Euro Canadian findspot (Tile 3 of the Supplementary Documentation). Seven of the pre-contact Aboriginal findspots were located within 50m of the current Study Area, including five to the northwest of the long stretch running parallel to Gingerich Road and another two to the southeast.

Among the latter group, Findspot 16 (AiHd-178) extends into the current Study Area (Tile 3 of the Supplementary Documentation). Findspot 16 (AiHd-178) comprised a single chert flake and a projectile point fragment, both manufactured from Onondaga chert. Although fragmentary, the point fragment was identified as a Late Archaic Crawford Knoll variety. Findspot 18 (AiHe-467) to the northeast of the Study Area, meanwhile, comprised an isolated projectile point fragment that shared characteristics common among varieties from the Late Paleo-Indian period. Both sites were recommended for a Stage 3 assessments, neither of which had been completed at the time of this investigation. Although Findspot 18 (AiHe-467) does not extend into the current Study Area, a portion of its surrounding 20m protective buffer overlaps it.

The remaining findspots that were identified within 50m of the Study Area yielded isolated chert flakes, including eight from Findspot 9, Findspot 10, Findspot 12, and Findspot 19 (AiHd-177) to the northwest of the Study Area corridor, and another four from Findspot 11 (AiHe-470) to the southwest.

The Stage 2 assessment of AiHe-469 resulted in the documentation of eleven lithic artifacts including nine pieces of chipping detritus and two projectile points. The artifacts were scattered across an area of approximately 62m northwest-southeast by 38m northeast-southwest, in the northwestern corner of the northern agricultural field within the assessment area. One of the projectile points is reminiscent of a Hind type variety dating to the Late Archaic period. The remaining projectile point is reminiscent of a Nanticoke Triangular variety dating to the Late Woodland period. Morphological analysis of the chipping detritus suggests that late stages of lithic reduction occurred at the site a propensity towards formal tool preparation, and maintenance, as supported by the presence of two projectile points. Given the results of the Stage 2 assessment, AiHe-469 was interpreted as a medium size activity area of unknown function, occupied seasonally by Aboriginal people during the Late Archaic and Late Woodland periods, and characterised by late stages of lithic reduction. AiHe-469 fulfilled the criteria for a Stage 3 archaeological investigation. To further evaluate the site's CHVI, a Stage 3 archaeological assessment was recommended for AiHe-469.

The Stage 2 assessment of Findspot 1 resulted in the documentation of two pieces of chipping detritus manufactured from Onondaga and Haldimand chert, discovered along the western edge of the southern agricultural field. The Stage 2 assessment of Findspot 2 resulted in the documentation of one multi-tool manufactured from Onondaga chert, discovered along the southeastern edge of the Study Area in the southern agricultural field. The Stage 2 assessment of Findspot 3 resulted in the documentation of three pieces of chipping detritus manufactured from Onondaga and Haldimand chert, discovered along the northwestern edge of the southern agricultural field. The Stage 2 assessment of Findspots 4 and 5 each resulted in the documentation of a single piece of Onondaga chert chipping detritus, discovered in the southern agricultural field. The Stage 2 assessment of Findspot 6 resulted in the documentation of three pieces of chipping detritus manufactured from Onondaga chert, discovered in the central portion of the southern agricultural field. The Stage 2 assessment of Findspot 7 (AiHe-468) resulted in the documentation of a single Brewerton side-notched projectile point manufactured from Onondaga chert, discovered along the northwestern edge of the southern agricultural field. The Stage 2 assessment of Findspot 8 resulted in the documentation of two pieces of Onondaga chert chipping detritus and a biface manufactured from Onondaga chert, discovered along the northeastern edge of the northern agricultural field. The Stage 2 assessment of Findspot 13 resulted in the documentation of three pieces of Onondaga chert chipping detritus, discovered in the southern half of the northern agricultural field. The Stage 2 assessment of Findspots 14 and 15 each resulted in the documentation of a single piece of Onondaga chert chipping detritus. discovered in the northern agricultural field. The Stage 2 assessment of Findspot 17 resulted in the documentation of a single piece of Onondaga chert chipping detritus, discovered in the northcentral portion of the northern agricultural field. Given the isolated nature of the artifacts, the CHVI of Findspots 1-6, 7 (AiHe-468), 8, 13 to 15, and 17 is judged to be sufficiently documented. Findspots 1-6, 7 (AiHe-468), 8, 13 to 15, and 17 did not fulfill the criteria for a Stage 3 archaeological investigations. Therefore, no further archaeological assessment was recommended for Findspots 1-6, 7 (AiHe-468), 8, 13 to 15, and 17.

Lastly, H1 was identified during the test pit assessment of an overgrown grassy clearing located north of the northern agricultural field, in the northern portion of the assessment area. The Stage 2 assessment of the site resulted in the documentation of 31 Euro-Canadian artifacts from six test pit and a test unit, spanning an area of approximately 5m by 5m. The Stage 2 artifact assemblage comprised 13 structural artifacts, 8 ceramic sherds, 7 household artifacts, 2 pieces of recent material, and 1 piece of undetermined metal. Analysis of the assemblage has suggested a period of occupation at the site spanning the late 19th to 20th century. This conclusion is supported by ceramic assemblage, which was dominated by sherds of red earthenware (n=4; 50%), and also featured 2 pieces of refined white earthenware, 1 piece of ironstone, and 1 piece of porcelain. Furthermore, the nails within the Stage 2 assemblage were entirely wire nails and all of the window glass shards are greater than 1.6mm suggesting a post-1845 date of occupation. The piece of plastic, the piece of aluminum foil, and the piece of porcelain indicate 20th century occupation. Based on the results of the Stage 2 investigation, H1 was interpreted as a small size, late 19th to 20th century domestic scatter. Given the presence of less than 20 artifacts that date the period of use to before 1900, H1 does not meets the criteria for a Stage 3 Site Specific Assessment and retains no CHVI.

It was recommended that the Stage 3 archaeological assessments for AiHe-469, Findspot 16 (AiHd-178), and Findspot 18 (AiHe-467) will be conducted according to the procedures outlined in the *Standards and Guidelines* (Government of Ontario 2011a). Typically, a Stage 3 assessment begins with an intensive controlled surface pickup ('CSP') across the Stage 2 limits of the sites, conducted as per Section 3.2.1 of the *Standards and Guidelines* (Government of Ontario 2011a). The Stage 2 pedestrian survey, however, consisted of an intensive surface collection across the entire site limits within the agricultural fields; all artifacts were mapped digitally and collected for laboratory analysis. Thus, the conditions for a Stage 3 CSP were met during the Stage 2 assessment.

Given that it is not yet evident that the level of CHVI at the sites will result in a recommendation to proceed to Stage 4 (see Section 4.19), the Stage 3 assessment of AiHe-469, Findspot 16 (AiHd-178), and Findspot 18 (AiHe-467) will consist of the hand excavation of 1m square test units every 5m in systematic levels and into the first 5cm of subsoil as per Table 3.1, Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a). Additional 1m test units, amounting to 20% of the grid total, will be placed in areas of interest within the sites extents as per Table 3.1, Standard 2 of the *Standards and Guidelines* (Government of Ontario 2011a). All excavated soil will be screened through six-millimetre mesh; all recovered artifacts will be recorded by their corresponding grid unit designation and collected for laboratory analysis. If a subsurface cultural feature is encountered, the plan of the exposed feature will be recorded and geotextile fabric will be placed over the unit before backfilling the unit.

To the best of Detritus' knowledge, no additional assessments have been conducted adjacent to the Study Area, nor are any sites registered within 50m.

1.3.4 Archaeological Potential

Archaeological potential is established by determining the likelihood that archaeological resources may be present on a subject property. Detritus applied archaeological potential criteria commonly used by the MHSTCI to determine areas of archaeological potential within Study Area. According to Section 1.3.1 of the *Standards and Guidelines* (Government of Ontario 2011a), these variables include proximity to previously identified archaeological sites, distance to various types of water sources, soil texture and drainage, glacial geomorphology, elevated topography, and the general topographic variability of the area.

Distance to modern or ancient water sources is generally accepted as the most important determinant of past human settlement patterns and, when considered alone, may result in a determination of archaeological potential. However, any combination of two or more other criteria, such as well-drained soils or topographic variability, may also indicate archaeological potential. When evaluating distance to water it is important to distinguish between water and shoreline, as well as natural and artificial water sources, as these features affect site locations and types to varying degrees. As per Section 1.3.1 of the *Standards and Guidelines* (Government of Ontario 2011a), water sources may be categorized in the following manner:

- Primary water sources, lakes, rivers, streams, creeks;
- secondary water sources, intermittent streams and creeks, springs, marshes and swamps;
- past water sources, glacial lake shorelines, relic river or stream channels, cobble beaches, shorelines of drained lakes or marshes; and
- accessible or inaccessible shorelines, high bluffs, swamp or marshy lake edges, sandbars stretching into marsh.

As was discussed above, the closest source of potable water is Baden Creek, which runs adjacent to the northeast of the Study Area.

Soil texture is also an important determinant of past settlement, usually in combination with other factors such as topography. The primary soils within the Study Area have been documented as being suitable for pre-contact Aboriginal practices. Considering also the length of occupation of Wilmot Township prior to the arrival of Euro-Canadian settlers and the 19 pre-contact Aboriginal sites registered within 1km, and the pre-contact and post-contact Aboriginal archaeological potential of the Study Area is judged to be moderate to high.

For Euro-Canadian sites, archaeological potential can be extended to areas of early Euro-Canadian settlement, including places of military or pioneer settlements; early transportation routes; and properties listed on the municipal register or designated under the *Ontario Heritage Act* (Government of Ontario 1990b) or property that local histories or informants have identified with possible historical events.

Historical Atlas (Walker & Miles 1877) demonstrates that Wilmot Township was occupied by Euro-Canadian farmers by the late 19th century. Much of the established road system and agricultural settlement from that time is still visible today. The early communities of Baden and Petersburg are illustrated to the east of the Study Area and New Hamburg to the west. Furthermore, the Grand Trunk Railway is illustrated transecting the Study Area from southwest to northeast. Considering also the six Euro-Canadian sites registered within 1km of the Study Area, the archaeological potential for Euro-Canadian sites is deemed to be moderate to high.

Finally, despite the factors mentioned above, extensive land disturbance can eradicate archaeological potential within a Study Area, as per Section 1.3.2 of the *Standards and Guidelines* (Government of Ontario 2011a). Current aerial imagery of the Study Area identified a number of potential disturbance areas within the Study Area, including all or portions of the Snyder's Road West, Christian Street, Charlotta Street, Gingerich Road, and Foundry Street road allowances; the Canadian National Railway railbed; and the paved driveway of the Baden Wastewater Pumping Station. It is recommended that these areas be subject to a Stage 2 property inspection, conducted according Section 2.1.8, Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a), to confirm and document the disturbed areas.

2.0 Field Methods

The Stage 2 assessment was conducted between August 16, 2021 and May 26, 20221, under archaeological consulting license P389 issued to Dr. Walter McCall by the MHSTCI. The limits of the Study Area were determined using mapping provided by the Proponent and a handheld GPS unit.

During the Stage 2 field work, assessment conditions were excellent and at no time were the field, weather, or lighting conditions detrimental to the recovery of archaeological material. Table 3 displays the field and weather conditions. Photos 1 to 24 demonstrate the land conditions throughout the Study Area, including areas that met the requirements for a Stage 2 archaeological assessment, as per Section 7.8.6, Standards 1a and b of the *Standards and Guidelines* (Government of Ontario 2011a). Figures 4 to 8 provides an illustration of the Stage 2 assessment methods as well as photograph locations and directions. Figure 9 provides an illustration of the Stage 2 assessment methods in relation to the development.

| Date | Activity | Weather | Field Conditions |
|-----------------|--|-------------------|--|
| August 16, 2021 | test pit survey | sunny, high 25°C | soil dry and screens easily |
| August 17, 2021 | test pit survey | cloudy, high 20°C | soil dry and screens easily |
| August 18, 2021 | test pit survey and test unit excavation | cloudy, high 27°C | soil dry and screens easily |
| May 26, 2022 | pedestrian, test pit survey | cloudy, high 17°C | soil visibility 90%; soil dry and screens easily |

Table 3: Field and Weather Conditions

Approximately 30% of the Study Area consisted of grassy areas, treed areas, and a dry seasonal creek bed running west of Baden Creek, all of which were deemed inaccessible for ploughing. The majority of these areas were subject to a typical Stage 2 test pit survey, conducted at 5m intervals in accordance with Section 2.1.2, Standards 1 and 2 of the *Standards and Guidelines* (Government of Ontario 2011a; Photos 6 to 12). The remainder of these areas was subject to a test pit survey at 10m intervals. The test pits in these areas were excavated to greater than 60cm deep and no subsoil was encountered. The test pits contained rounded pebbles and silty sand, which is indicative of a creek bed (Photos 13, 14, 15, 16). Additionally, the woodlot within the Study Area was densely covered and test pits were excavated at 10m intervals or wherever possible as per Section 2.1.2, Guideline 1 of the *Standards and Guidelines* (Government of Ontario 2011a).

The test pit survey was conducted to within 1m of the built structures (intact or ruins) or until disturbance was encountered according to Section 2.1.2, Standard 4 of the *Standards and Guidelines* (Government of Ontario 2011a). Each test pit was at least 30 centimetres ('cm') in diameter and excavated 5cm into sterile subsoil as per Section 2.1.2, Standards 5 and 6 of the *Standards and Guidelines* (Government of Ontario 2011a). The soils were examined for stratigraphy, cultural features, or evidence of fill.

The test pits ranged in depth from 33cm to 64cm. All test pits contained a single stratigraphic layer. Considering that each test pit was excavated 5cm into sterile subsoil, this observed soil layer ranged in depth from 28cm to 59cm. All soil was screened through six-millimetre mesh hardware cloth to facilitate the recovery of small artifacts and then used to backfill the pit as per Section 2.1.2, Standards 7 and 9 of the *Standards and Guidelines* (Government of Ontario 2011a).

When archaeological resources were encountered, the test pit excavation was continued on the survey grid, as per Section 2.1.3, Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a). Given that insufficient resources were recovered through the continued survey on the grid, the survey coverage was intensified to determine whether a Stage 3 assessment could be supported using Section 2.1.3, Standard 2, Option A of the *Standards and Guidelines* (Government of Ontario 2011a). After the excavation of cardinal test pits, insufficient resources were produced to support a recommendation to carry out a Stage 3 assessment, therefore a stage 2 test unit was excavated on top of the positive test pit at Findspot 2 (Photos 23, 24). The test unit produced no artifacts; therefore, it was determined that the findspot did not support a recommendation to carry out a Stage 3 assessment methods were

employed. The test pit survey at Findspot 1 resulted in a single positive test pit producing a single pre-contact Aboriginal artifact. No test unit was excavated.

All artifacts were recorded with reference to their associated test pit and test unit and were retained for laboratory analysis. In accordance with Section 5, Standard 2b of the *Standards and Guidelines* (Government of Ontario 2011a) a Universal Transverse Mercator ('UTM') coordinate was recorded for each positive test pit. All coordinates were taken using a Garmin eTrex 10 GPS unit with a minimum accuracy 1-2.5m (North American Datum 1983 ('NAD83') and UTM Zone 17T) and are presented in the Supplementary Documentation to this report.

Approximately 15% of the Study Area comprised active agricultural land that was accessible for ploughing (Photos 10, 17 to 19). As per Section 2.1.1, Standards 2 and 3 of the *Standards and Guidelines* (Government of Ontario 2011a), the agricultural land had been ploughed and allowed to weather prior to assessment. The ploughing was deep enough to provide total topsoil exposure, and provided a minimum of 80% surface visibility, as per Section 2.1.1, Standards 4 and 5 of the *Standards and Guidelines* (Government of Ontario 2011a). The ploughed area was subject to a pedestrian survey at 3m intervals, conducted in accordance with Section 2.1.1, Standard 6 of the *Standards and Guidelines* (Government of Ontario 2011a). No artifacts were observed during the pedestrian survey; therefore, no further survey methods were employed.

Approximately 20% of the Study Area comprised areas identified in the aerial photography of the Study Area as being previously disturbed. These areas were subject to a Stage 2 property inspection, conducted according to Section 2.1.8, Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a). All or portions of the Snyder's Road West, Christian Street, Charlotta Street, Gingerich Road, and Foundry Street road allowances; the Canadian National Railway railbed; and the paved driveway of the Baden Wastewater Pumping Station were evaluated as having no potential based on the identification of extensive and deep land alteration that has severely damaged the integrity of archaeological resources, as per Section 2.1, Standard 2b of the *Standards and Guidelines* (Government of Ontario 2011a). All of the visibly disturbed areas documented within the Study Area were mapped and photo documented in accordance with Section 2.1, Standard 6 and Section 7.8.1, Standard 1b of the *Standards and Guidelines* (Government of Ontario 2011a).

The remaining 35% of the Study Area comprised agricultural field that was not subject to assessment following a decision by the Proponent to exclude the area north of Snyder's Road from the current development.

3.0 Record of Finds

The Stage 2 archaeological assessment was conducted employing the methods described in Section 2.0 above. This investigation resulted in the documentation of two pre-contact Aboriginal findspots (see below). An inventory of the documentary record generated by fieldwork is provided in Table 4 below.

| Document Type | Current Location of | Additional Comments |
|-------------------------|---------------------|----------------------------------|
| | Document Type | |
| 4 Pages of Field Notes | Detritus' office | Stored digitally in project file |
| 3 Maps provided by the | Detritus' office | Stored digitally in project file |
| Proponent | | |
| 6 Field Maps | Detritus' office | Stored digitally in project file |
| 103 Digital Photographs | Detritus' office | Stored digitally in project file |

Table 4: Inventory of Document Record

All of the material culture collected during the Stage 2 survey is contained in one box and will be temporarily housed in the offices of Detritus until formal arrangements can be made for its transfer to Her Majesty the Queen in right of the Province of Ontario or another suitable public institution acceptable to the MHSTCI.

3.1 Cultural Material

As was discussed above, two pre-contact Aboriginal findspots, were identified within the Study Area. Photographs of the artifacts recovered from the Stage 2 assessment is depicted in Section 9.2 of this report.

The pre-contact Aboriginal findspots produced lithic artifacts manufactured from Onondaga chert. Chert type identifications were accomplished visually using reference materials located online or in personal collections.

Onondaga formation chert is from the Middle Devonian age, with outcrops occurring along the north shore of Lake Erie between Long Point and the Niagara River (Eley and von Bitter 1989). Primary outcrops have also been reported along the banks of the Grand River (Ellis and Ferris 1990). It is a high-quality raw material frequently utilized by pre-contact people and often found at archaeological sites in southern Ontario. Onondaga chert occurs in nodules or irregular thin beds. It is a dense non-porous rock that may be light to dark grey, bluish grey, brown or black and can be mottled with a dull to vitreous or waxy lustre (Eley and von Bitter 1989).

Furthermore, all pieces of chipping detritus were subject to morphological analysis following the classification scheme described by Lennox, Dodd and Murphy for the Wiacek Site (Lennox *et al.* 1986) and expanded upon by Fisher for the Adder Orchard site (Fisher 1997). The flake types that were identified during the analysis of current Stage 2 assemblage include thinning flakes.

Primary and secondary flakes, along with cortical removal flakes, are a product of percussion flaking undertaken during the initial reduction phases of raw material into blanks, bifaces and preforms. These early-stage reduction flakes tend to exhibit minimal dorsal flake scarring, and are often characterized by the presence of cortex, or the original unflaked chert exterior, on their dorsal surfaces and proximal ends. For cortical removal flakes, over half of the dorsal surface comprises cortex; for primary flakes, less than half. Secondary flakes, meanwhile, may not contain any cortex. Thinning flakes are produced during the latter stages of lithic reduction, when blanks, bifaces, and preforms are shaped into projectile points and formal tools. They are the result of pressure flaking, where the maker uses a softer material such as antler, wood or bone to apply direct pressure onto a specific part of the tool. Pressure flaking generally produces smaller, thinner flakes than does percussion flaking. Thinning flakes also exhibit more flake scars on their dorsal surface than do primary or secondary flakes. Fragmentary flakes are flakes that may have some identifiable flake characteristic, but cannot be classified with certainty into a specific category.

3.2 Findspot 1

Findspot 1 comprises one piece of chipping detritus from a single test pit in the lawn area between 182 and 186 Snyder's Road West, prior to its removal from the development area. The piece of chipping detritus was identified as an Onondaga chert thinning flake.

Table 5 provides a complete catalogue of the Stage 2 artifact assemblage from Findspot 1. Given the isolated nature of the artifact, it is difficult to draw any useful conclusions regarding site function.

Table 5: Findspot 1 Stage 2 Artifact Catalogue

| Cat # | Context | Artifact | Morphology | Freq. | Chert Type |
|-------|---------|-------------------|------------|-------|------------|
| 1 | PTP1 | chipping detritus | Thinning | 1 | Onondaga |

3.3 Findspot 2

Findspot 2 comprised one piece of Onondaga chert chipping detritus, discovered between the Canadian National Rail Tracks and Gingerich Road, approximately 188m to the southwest of St. George Street. Table 6 provides a complete catalogue of the Stage 2 artifact assemblage from Findspot 2. Given the isolated nature of the artifact, it is difficult to draw any useful conclusions regarding site function.

Table 6: Findspot 2 Stage 2 Artifact Catalogue

| Cat # | Context | Artifact | Frequency | Chert Type |
|-------|---------|-------------------|-----------|------------|
| 1 | PTP2 | chipping detritus | 1 | Onondaga |

4.0 Analysis and Conclusions

Detritus was retained by the Proponent to conduct a Stage 1-2 archaeological assessment in advance of a proposed Baden sanitary sewer replacement within the community of Baden.

The Stage 1 background research indicated that the Study Area exhibited moderate to high potential for the identification and recovery of archaeological resources. This investigation also revealed that portions of the Study Area overlap areas that were subject to a previous Stage 1-2 assessment conducted by Detritus in 2020 (Detritus 2020). This earlier investigation was conducted on two large irregular parcels of primarily agricultural land located to the southwest of Foundry Street and to the northwest of Gingerich Road. The two parcels were generally separated by Baden Creek.

This investigation resulted in the documentation of one pre-contact Aboriginal site, AiHe-469, as well as 19 pre-contact Aboriginal findspots, and one Euro Canadian findspot (Tile 3 of the Supplementary Documentation). Seven of the pre-contact Aboriginal findspots were located within 50m of the current Study Area, including five to the northwest of the long stretch running parallel to Gingerich Road and another two to the southeast.

Among the latter group, Findspot 16 (AiHd-178) extends into the current Study Area (Tile 3 of the Supplementary Documentation). Findspot 16 (AiHd-178) comprised a single chert flake and a projectile point fragment, both manufactured from Onondaga chert. Although fragmentary, the point fragment was identified as a Late Archaic Crawford Knoll variety. Findspot 18 (AiHe-467) to the northeast of the Study Area, meanwhile, comprised an isolated projectile point fragment that shared characteristics common among varieties from the Late Paleo-Indian period. Both sites were recommended for a Stage 3 assessments, neither of which had been completed at the time of this investigation. Although Findspot 18 (AiHe-467) does not extend into the current Study Area, a portion of its surrounding 20m protective buffer overlaps it.

The remaining findspots that were identified within 50m of the Study Area yielded isolated chert flakes, including eight from Findspot 9, Findspot 10, Findspot 12, and Findspot 19 (AiHd-177) to the northwest of the Study Area corridor, and another four from Findspot 11 (AiHe-470) to the southwest.

The Stage 2 field survey of the Study Area was conducted between August 16th and 18th, 2021 and on May 26, 2022. This investigation consisted of a pedestrian survey at a 3m interval of most of the agricultural land throughout the Study Area, and a test pit assessment of the grassy and treed areas, and the seasonal creek extending from Baden Creek to the west. Prior to the completion of the Stage 2 assessment, the Proponent adjusted the development area to exclude the portions of the Study Area corridor to the northwest of Snyder's Road West.

The various road allowances throughout the Study Area, the Canadian National Railway bed, and the pumping house driveway were evaluated as having no potential based on the identification of extensive and deep land alteration that has severely damaged the integrity of archaeological resources, as per Section 2.1, Standard 2b of the *Standards and Guidelines* (Government of Ontario 2011a). The previously disturbed areas, as confirmed during a Stage 2 property inspection, were mapped and photo documented in accordance with Section 2.1, Standard 6 and Section 7.8.1, Standards 1a and 1b of the *Standards and Guidelines* (Government of Ontario 2011a).

The Stage 2 field survey resulted in the documentation of two pre-contact Aboriginal findspots: Findspot 1 and Findspot 2. Findspot 1 is an Onondaga chert thinning flake recovered during the test pit assessment of the lawn area between 182 and 186 Snyder's Road West prior to its removal from the development area. Although eight cardinal test pits were excavated around the positive test pit, all of which were sterile, no 1m unit was excavated over it. Findspot 2, meanwhile, comprises one Onondaga chert thinning flake. The eight cardinal test pits and a single 1m test unit yielded no additional artifacts.

Based on the available evidence, Findspot 1 and Findspot 2 have been identified as isolated pieces of debitage that likely originated from one of the larger sites or findspots identified during the

earlier Stage 2 assessment of the agricultural land located to the southwest of Foundry Street and to the northwest of Gingerich Road.

5.0 Recommendations

The Stage 2 assessment of Findspot 1, however, remains incomplete. Therefore, **Findspot 1 is recommended for additional Stage 2 investigation** in the form of a 1m metre test unit, excavated above the positive test unit, as per Section 2.1.3, Section 2a of the *Standards and Guidelines* (Government of Ontario 2011a). This test unit will be excavated by hand in systematic levels and into the first five centimetres ('cm') of subsoil as per Section 3.2, Standards 3 to 5 of the *Standards and Guidelines* (Government of Ontario 2011a).

Given the isolated nature of the non-diagnostic artifact, Findspot 2, which was subject to full intensification, does not fulfill any of the criteria for a Stage 3 site specific assessment as per Section 2.2 of the *Standards and Guidelines* (Government of Ontario 2011a); therefore **Findspot 2 is not recommended for a Stage 3 assessment**.

Additionally, according to the results of the previous Stage 1-2 assessment conducted by Detritus in 2020, the portion of Findspot 16 (AiHd-178) that extends into the current Study Area is recommended for Stage 3 assessment (Detritus 2020). Likewise, the portion of the 20m protective buffer surrounding Findspot 18 (AiHe-467) that intersects the Study Area is also recommended for Stage 3 assessment (Detritus 2020).

Typically, a Stage 3 assessment begins with an intensive controlled surface pickup ('CSP') across the Stage 2 limits of the sites, conducted as per Section 3.2.1 of the *Standards and Guidelines* (Government of Ontario 2011a). The Stage 2 pedestrian survey at the sites, however, consisted of an intensive surface collection across the entire limits of each site. All artifacts were mapped digitally and collected for laboratory analysis. Thus, the conditions for a Stage 3 CSP were met during the Stage 2 assessment.

Given that it is not yet evident that the level of cultural heritage value or interest will result in a recommendation to proceed to Stage 4, the Stage 3 assessments at Findspot 16 (AiHd-178) and the protective buffer around Findspot 18 (AiHe-467) will consist of the hand excavation of 1m square test units every 5m in systematic levels and into the first 5cm of subsoil as per Table 3.1, Standard 1 of the *Standards and Guidelines* (Government of Ontario 2011a). Additional 1m test units, amounting to 20% of the grid total, will be placed in areas of interest within the site extent as per Table 3.1, Standard 2 of the *Standards and Guidelines* (Government of Ontario 2011a). All excavated soil will be screened through six-millimetre mesh; all recovered artifacts will be recorded by their corresponding grid unit designation and collected for laboratory analysis. If a subsurface cultural feature is encountered, the plan of the exposed feature will be recorded and geotextile fabric will be placed over the unit before backfilling the unit.

Additionally, **the Stage 1 investigation revealed that portions of the Study Area exhibited moderate to high potential for the identification and recovery of archaeological resources. These portions corresponded with the agricultural field to the northwest of Snyder's Road West, which were not assessed as part of the Stage 2 assessment. All active agricultural land that retains archaeological potential and is accessible for ploughing will be subject to a typical Stage 2 pedestrian survey at a 5m interval**, conducted according to Section 2.1.1 of the *Standards and Guidelines* (Government of Ontario 2011a). The fields will be ploughed until 80% surface visibility is attained, then allowed to weather prior to assessment. As per Section 2.1.1, Standard 7 of the *Standards and Guidelines* (Government of Ontario 2011a), if archaeological resources are found, the survey transects will be decreased to 1m intervals over a 20m radius around each find to determine whether it is an isolated find or part of a larger scatter. All formal artifact types and diagnostic categories will be collected for laboratory analysis and cataloguing, including all refined ceramic sherds for 19th century archaeological sites.

6.0 Advice on Compliance with Legislation

This report is submitted to the Minister of Heritage, Sport, Tourism and Culture Industries as a condition of licensing in accordance with Part VI of the *Ontario Heritage Act*, R.S.O. 1990, c 0.18. The report is reviewed to ensure that it complies with the standards and guidelines that are issued by the Minister, and that the archaeological fieldwork and report recommendations ensure the conservation, protection and preservation of the cultural heritage of Ontario. When all matters relating to archaeological sites within the project area of a development proposal have been addressed to the satisfaction of the Ministry of Heritage, Sport, Tourism and Culture Industries, a letter will be issued by the ministry stating that there are no further concerns with regard to alterations to archaeological sites by the proposed development.

It is an offence under Sections 48 and 69 of the *Ontario Heritage Act* for any party other than a licensed archaeologist to make any alteration to a known archaeological site or to remove any artifact or other physical evidence of past human use or activity from the site, until such time as a licensed archaeologist has completed archaeological fieldwork on the site, submitted a report to the Minister stating that the site has no further cultural heritage value or interest, and the report has been filed in the Ontario Public Register of Archaeology Reports referred to in Section 65.1 of the *Ontario Heritage Act*.

Should previously undocumented archaeological resources be discovered, they may be a new archaeological site and therefore subject to Section 48 (1) of the *Ontario Heritage Act*. The proponent or person discovering the archaeological resources must cease alteration of the site immediately and engage a licensed consultant archaeologist to carry out archaeological fieldwork, in compliance with Section 48 (1) of the *Ontario Heritage Act*.

The *Cemeteries Act*, R.S.O. 1990 c. C.4 and the *Funeral, Burial and Cremation Services Act*, 2002, S.O. 2002, c.33 (when proclaimed in force) require that any person discovering human remains must notify the police or coroner and the Registrar of Cemeteries at the Ministry of Consumer Services.

Archaeological sites recommended for further archaeological fieldwork or protection remain subject to Section 48 (1) of the *Ontario Heritage Act* and may not be altered, or have artifacts removed from them, except by a person holding an archaeological license.

7.0 Bibliography and Sources

- Archives of Ontario. 2012-2015. *The Evolution of the District and County System*, *1788-1899*. Electronic document: <u>http://www.archives.gov.on.ca/en/maps/ontario-districts.aspx</u>. Last accessed, July 7, 2022.
- Birch, J. 2010. *Coalescence and Conflict in Iroquoian Ontario*. Electronic resource: <u>http://uga.academia.edu/JenniferBirch/Papers/183903/Coalescence and Conflict in Iroquoia n Ontario</u>. Last accessed January 20, 2022.
- Caston, Wayne A. 1997. Evolution in the Mapping of Southern Ontario and Wellington County. *Wellington County History* 10:91-106.
- Chapman, L.J. and D.F. Putnam. 1984. *The Physiography of Southern Ontario*. Third Edition. Ontario Geological Survey. Special Volume 2. Toronto: Ontario Ministry of Natural Resources.
- The Corporation of the Township of Wilmot. 2015. *History of Wilmot Township*. Electronic Source: <u>https://www.wilmot.ca/en/living-here/History-of-Wilmot-Township.aspx#</u>. Last accessed, August 28, 2020.
- Coyne, J. H. 1895. The Country of the Neutrals (As Far as Comprised in the County of Elgin): From Champlain to Talbot. St. Thomas: Times Print.
- Detritus. 2020. Stage 1-2 Archaeological Assessment, Foundry and Gingrich Lands, Part of Lots 15 and 16, German Block South of Snyder Road, Geographic Township of Wilmot, now the Region of Waterloo, Ontario. Report on file with the MHSTCI.
- Eley, Betty, and Peter H. von Bitter, 1989. *Cherts of Southern Ontario*. Toronto: Royal Ontario Museum.
- Ellis, Chris J. and Neal Ferris (editors). 1990. *The Archaeology of Southern Ontario to A.D. 1650*. Occasional Publication of the London Chapter, Ontario Archaeological Society, Number 5.
- Ferris, Neal. 2009. *The Archaeology of Native-Lived Colonialism: Challenging History in the Great Lakes*. Tucson: University of Arizona Press.
- Fisher, Jacqueline A. 1997. *The Adder Orchard Site: Lithic Technology and Spatial Organization in the Broadpoint Late Archaic*. Occasional Publications of the London Chapter, OAS, Number 3, 1997.
- Gentilcore, R. Louis and C. Grant Head. 1984. *Ontario's History in Maps*. Toronto: University of Toronto Press.
- Gibson, M. M. 2006. In the Footsteps of the Mississaugas. Mississauga Heritage Foundation.
- Government of Ontario. n.d. Archaeological Sites Database Files. MHSTCI.
- Government of Ontario. 1990a. *Ontario Planning Act*, R.S.O. 1990, CHAPTER P.13. Last amendment: 2021, c. 25, Sched. 24. Electronic document: <u>https://www.ontario.ca/laws/statute/90p13</u>. Last accessed January 24, 2022.
- Government of Ontario. 1990b. *Ontario Heritage Act, R.S.O. 1990, CHAPTER O.18*. Last amendment: 2021, c. 4, Sched. 6, s. 74. Electronic document: <u>https://www.ontario.ca/laws/statute/90018</u>. Last accessed January 24, 2022.
- Government of Ontario. 1990c. *Freedom of Information and Protection of Privacy Act, R.S.O.* 1990, CHAPTER F.31. Last amendment: 2021, c. 4, Sched. 11, s. 11. Electronic document: <u>https://www.ontario.ca/laws/statute/90f31</u>. Last accessed January 24, 2022.
- Government of Ontario. 2011a. Standards and Guidelines for Consultant Archaeologists. MHSTCI.
- Government of Ontario. 2011b. Engaging Aboriginal Communities in Archaeology draft technical bulletin. MHSTCI.

- Heidenreich, Conrad. 1990. History of the St. Lawrence-Great Lakes Area to 1650. In Ellis, Chris J. Neal Ferris (editors), *The Archaeology of Southern Ontario to A.D. 1650*. Occasional Publications of the London Chapter OAS No. 5, pp. 475-492. Ontario Archaeological Society Inc.
- Lennox, P., C. Dodd and C. Murphy. 1986. The Wiacek Site: A Late Middleport Component, Simcoe County. London: Ontario Ministry of Transportation and Communications.
- Morris, J.L. 1943. *Indians of Ontario*. 1964 reprint. Toronto: Department of Lands and Forests, Government of Ontario.
- Page, H.R. & Co. 1879. Illustrated Historical Atlas of the County of Haldimand, Ontario. H.R. Page & Co.
- Pendergast, James. 1995. The Identity of Jacques Cartier's Stadaconans and Hochelagans: The Huron-Iroquois Option. In Origins of the People of the Longhouse: Proceedings of the 21st Annual Symposium of the Ontario Archaeological Society, pp. 106-118. André Bekerman and Gary Warrick, editors. North York: Ontario Archaeological Society.
- Praxis Research Associates. n.d. *The History of the Mississaugas of the New Credit First Nation*. Lands, Research and Membership, Mississaugas of the New Credit First Nation. Hagersville, Ontario.
- Schmalz, Peter S. 1991. The Ojibwa of Southern Ontario. University of Toronto Press, Toronto.
- Smith, Donald. 2002. "Their Century and a Half on the Credit: The Mississaugas." In Mississauga: The First 10,000 Years, edited by Frank Dieterman, 107-122. Easted Books, Toronto.
- Tanner, Helen (editor). 1987. *Atlas of Great Lakes Indian History*. Norman: University of Oklahoma Press.
- Walker & Miles. 1877. *Illustrated Historical Atlas of the County of Waterloo, Ont*. Toronto: Walker & Miles Co.
- Warrick, G.A. .2000. The Precontact Iroquoian Occupation of Southern Ontario. *Journal of World Prehistory*, Vol.14, No.4: pp. 415-466.
- Warrick, G. 2013. The Aboriginal Population of Ontario in Late Prehistory. In Munson, M.K. and Jamieson, S.M (editors), *Before Archaeology: The Archaeology of a Province*. McGill Queen's University Press.
- Weaver, Sally. 1978. Six Nations of the Grand River, Ontario. In *Handbook of North American Indians. Volume 15: Northeast*, pp. 525-536. Bruce G. Trigger, editor. Washington: Smithsonian Institution Press.

8.0 Maps

















9.0 Images

9.1 Field Photos

Photo 1: Agricultural Field Not Assessed, Stage 2 Recommended, facing northeast



Photo 3: Typical Test Pit in Grass North of Snyder's Road West

Photo 2: Grass Test Pit Surveyed at 5m Intervals, facing southwest



Photo 4: Disturbed Christian Street and Right-of-Way Not Assessed, facing southeast



Photo 5: Disturbed Charlotta Street Not Assessed, facing northeast





Photo 6: Grass Test Pit Surveyed at 5m Intervals, facing northeast



Photo 7: Typical Test Pit in Grass South of Charlotta Street



Photo 9: Grass Test Pit Surveyed at 5m Intervals, facing southeast



Photo 11: Grass Test Pit Surveyed at 5m Intervals, facing northwest





Photo 10: Pedestrian Survey at 3m Intervals, facing southeast



Photo 12: Typical Test Pit in Grass Southeast of the Canadian National Railway





Photo 13: Grass Covered Dry Seasonal Creek extending from Baden Creek to the Southeast Test Pit Surveyed at 10m Intervals, facing southeast Photo 14: Typical Test Pit Containing No Subsoil of the Grass Covered Dry Seasonal Creek extending from Baden Creek to the Southeast



Photo 15: Grass Test Pit Surveyed at 5m Intervals, facing northwest



Photo 17: Pedestrian Survey at 3m Intervals, facing southwest



Photo 16: Woodlot Test Pit Surveyed at 10m Intervals, facing northeast



Photo 18: Pedestrian Survey at 3m Intervals, facing northwest





Photo 19: Pedestrian Survey at 3m Intervals, facing southeast



Photo 21: Disturbed Foundry Street, facing northwest





Photo 22: Disturbed Baden Wastewater Pumping Station Paved Driveway, facing southwest



Photo 23: Test Unit Excavation at Findspot 2, facing east



Photo 24: Test Unit Excavation at Findspot 2, facing north





9.2 Artifact Photos

Plate 1: Chipping Detritus Recovered from Findspot 1 (left) and Findspot 2 (right)



SUPPLEMENTARY DOCUMENTATION Stage 1-2 Archaeological Assessment Baden Sanitary Sewer Replacement, Baden

Part of Lots 15 and 16, German Block South of Snyder Road; Part of Lots 15 and 16, German Block North of Snyder Road; and Part of the Snyder Road Allowance between Lot 16 German Block South of Snyder Road and Lot 16 German Block North of Snyder Road, Geographic Township of Wilmot, Historical County of Waterloo, now the Regional Municipality of Waterloo, Ontario

Submitted to:

MTE Consultants Inc. 520 Bingeman's Centre Drive, Kitchener, ON N2B 3X9

and

Ontario's Ministry of Heritage, Sport, Tourism and Culture Industries

Submitted by:



196 Westhheights Drive, Kitchener Ontario, N2N 1J9 Mobile/Office: 519-744-7018 www.detritusconsulting.ca

Licensee: Walter McCall License Number: P389 PIF Number: P389-0551-2021 CP Number: 2021-048

ORIGINAL SUPPLEMENTARY DOCUMENTATION

November 8, 2022

Table of Contents

| 1.0 | Maps | 3 |
|-----|-------------------------------|------|
| 2.0 | Universal Transverse Mercator | 11 |
| 3.0 | Aboriginal Engagement | . 12 |
| 4.0 | Email Communications | . 13 |
| 4.1 | Email 1 | . 13 |

1.0 Maps
















2.0 Universal Transverse Mercator

The Universal Transverse Mercator ('UTM') coordinates recorded during the Stage 2 assessment are provided below. The UTM coordinates were recorded using a Garmin eTrex Touch 30x GPS unit with a minimum accuracy 3m (North American Datum 1983 ('NAD83') and UTM Zone 17T). The UTM coordinates for the archaeological findspots are presented in Table 1 below. Landmark UTM coordinates are found in Table 1 below.

| Table 1: UTM Coordinates for the Archaeological | Sites and Findspots recovered during the |
|---|--|
| Stage 2 assessment | |

| Findspot | Description | UTM Coordinates |
|------------|--|------------------------|
| Findspot 1 | PTP1 | 17T 526171mE 4805626mN |
| Findspot 1 | Landmark #1 – Telephone Pole South Side of Snyder's Road West | 17T 526169mE 4805566mN |
| Findspot 1 | Landmark #2 – Telephone Pole South Side of Snyder's | |
| _ | Road West | 17T 526208mE 4805574mN |
| Findspot 2 | PTP2 | 17T 526647mE 4805077mN |
| Findspot 2 | Landmark #1 – Telephone Pole South Side of Gingerich Road | 17T 527077mE 4804751mN |
| Findspot 2 | Landmark #2 – Telephone Pole South Side of Gingerich Road | 17T 527119mE 4804783mN |

3.0 Aboriginal Engagement

According to Section 3.5, Standard 1 of the *Standards and Guidelines for Consultant Archaeologists* (*'Standards and Guidelines'*; Government of Ontario 2011a) and Section 1.1, Standard 2c of the *Engaging Aboriginal Communities in Archaeology* draft technical bulletin (Government of Ontario 2011b), Aboriginal communities must be engaged following a Stage 3 assessment when formulating Stage 4 mitigation strategies for the following types of Aboriginal archaeological sites:

- a. rare Aboriginal archaeological sites
- b. sites identified as sacred or known to contain human remains
- c. Woodland Aboriginal sites
- d. Aboriginal sites where topsoil stripping is being contemplated
- e. undisturbed Aboriginal sites
- f. sites previously identified as being of interest to the Aboriginal community

This section of the Supplementary Documentation provides a description of the engagement practices that were conducted following the Stage 2 assessment of the Study Area. No additional documentation arose from the engagement process.

The Stage 2 assessment was conducted between August 16th and 18th, 2021 and on May 26, 2022. This investigation consisted of a pedestrian survey of the recently ploughed and weathered agricultural land, and a typical test pit survey of the grass and treed areas, conducted at five-metre intervals and in some areas at 10m intervals. This investigation resulted in the documentation of two pre-contact Aboriginal findspots.

Prior to conducting the Stage 2 fieldwork, on July 16, 2021, Mississaugas of the Credit First Nation ('MCFN'), Six Nations of the Grand River First Nation ('Six Nations'), and Haudenosaunee Development Institute ('HDI') were contacted by Detritus to inform them that fieldwork was set to begin at the end of July. Additionally, Detritus passed along the Proponent contact information so that monitoring agreements could be put in place prior to the commencement of the fieldwork.

During the course of the field investigation, representatives from MCFN, Six Nations, and HDI participated as on-site monitors (Table 2). No fieldwork was conducted without the knowledge of all three communities.

| Date | Field Director | First Nations Monitors |
|-----------------|-------------------|--|
| August 16, 2021 | Jon Cousins, R296 | Jessica Williams and Wayne Hill (HDI), Davis Jacobs |
| | | (Six Nations), Eric Laforme (MCFN) |
| August 17, 2021 | Jon Cousins, R296 | Steve Markle (HDI), Davis Jacobs (Six Nations), Eric |
| | | Laforme (MCFN) |
| August 18, 2021 | Jon Cousins, R296 | Steve Markle (HDI), Davis Jacobs (Six Nations), Eric |
| | | Laforme (MCFN) |
| May 26, 2022 | Jon Cousins, R296 | Danton Miller, Owen Greene (HDI) |

Table 2: Fieldwork Dates and Communities Represented during the Fieldwork

4.0 Email Communications

4.1 Email 1

The following is email correspondence from the MHSTCI.

From: Horne, Malcolm (MHSTCI) <<u>Malcolm.Horne@ontario.ca</u>> Date: Mon, Aug 23, 2021 at 4:48 PM Subject: Recommendations for Incomplete Stage 2 Assessment, Beckdale Sanitary Sewer, Wilmot, P389-0551-2021,MHSTCI File 0014639 To: <u>waltermccall.arch@gmail.com</u> <<u>waltermccall.arch@gmail.com</u>> Cc: Archaeology (MHSTCI) <<u>archaeology@ontario.ca</u>>

Hi, Walt. I understand from the below that the part of the original project area which was north of Snyder's Road has now been removed from the project area. However, some archaeological assessment has already been completed in this area but not completed. And, an archaeological site that was identified was not assessed to standard.

To address this situation you have suggested:

- 1. "The Stage 1 identified the entire Study Area as having potential; a Stage 2 was recommended for the entire original Study Area.
- 2. The Area south of Snyder's Road will have typical recommendations for any artifacts we recover (so far we have identified two areas of artifacts in the test pit portion; we are still awaiting ploughing and weathering for the rest)
- The portion of the corridor north of Snyder's Road East will be recommended for Stage 2 pedestrian survey of all agricultural land, as well as a recommendation for intensification for the findspot (which will have a specific findspot designation) according to Section 2.1.3, Standard 2 Option A or B."

This is a valid approach and these recommendations will be acceptable.

Please include a PDF copy of this advice as supplementary documentation to your project report package.

As a standard part of all advice provided to licensees, please note that this advice has been provided by this ministry under the assumption that the information submitted by the licensed archaeologist is complete and accurate. The advice provided applies only to the project in question and is not to be used as a precedent for future projects. Further measures may need to be taken in the event that additional artifacts or archaeological sites are identified or if the information provided by the licensed archaeologist is otherwise found to be inaccurate, incomplete, misleading, or fraudulent.

Sincerely,

Malcolm Horne Archaeology Review Officer Archaeology Program Unit

Detritus Consulting Ltd.

Ministry of Heritage, Sport, Tourism and Culture Industries

Mobile: 437-339-8861

Email: Malcolm.Horne@ontario.ca

From: Walter McCall <<u>waltermccall.arch@gmail.com</u>> Sent: August 19, 2021 4:04 PM To: Hadlari, Wai (MHSTCI) <<u>Wai.Hadlari@Ontario.ca</u>>; Archaeology (MHSTCI) <<u>archaeology@ontario.ca</u>> Subject: Stage 2 assessment, Beckdale Sanitary Sewer Stage 1-2

CAUTION -- EXTERNAL E-MAIL - Do not click links or open attachments unless you recognize the sender.

Good afternoon Wai,

I am contacting you regarding a Stage 1-2 assessment that Detritus has been retained to conduct in advance of the Beckdale Sanitory Sewer installation on the western edge of the community of Baden in the Region of Waterloo (P389-0551-2021). As I understand it, Garth has already been in touch with you regarding a strategy of test-pitting the entire corridor. If this is not the case, can you please direct my inquiry to whomever did provide this advice.

In any case, representatives from 6 Nations, MCFN, and HDI I believe have been in attendance during the fieldwork. They have requested that all areas that can be ploughed, be subject to a Stage 2 pedestrian survey rather than test pitting. We have no issues with this, and have informed the clients accordingly. This pedestrian survey hasn't happened yet, but it is planned for the near future.

A small situation has arisen, however, that requires a bit of guidance.

The original Study Area was based on a map provided by the client (attached here, called Original Study Area). We have just learned that only the portion of the corridor south of Snyder's Road E, that main thoroughfare through Baden, had been cleared for development. Normally in this situation, I would simply contact the MHSTCI to have the mapping from our original PIF request adjusted to match the revised Study Area. However, we conducted the test pitting portion of the stretch north of th Snyder's Road East. As the mapping shows, this test pit area was not extensive since most of the corridor on this side crosses agricultural land. Nevertheless, we found a single positive test pit north of Snyder's Road East containing one chert flake. The test pit was located next to an area to be ploughed. Because the positive test may be part of a larger scatter in the adjacent field that could require Stage 3, our FD did not intensify around the positive test pit.

There are no plans to return to the area north of Snyder's Road East, and it is unclear if the client actually has permission to access the fields in question. With this in mind, I am wondering how I document this flake. We have found other artifacts within the revised Study Area to the south, which have been intensified accordingly allowing us to make valid recommendations. This single flake, however, was never intensified. Even if it was, it is unknown if artifacts extend into the agricultural land to the north, which will not be assessed.

Technically, therefore, I will have a Stage 1 that covers the entire original Study Area; a completed Stage 2 assessment south of Snyder's Road East; and a partial Stage 2 assessment on the portion to the north of the road. Therefore, I am assuming my recommendations will be;

1. The Stage 1 identified the entire Study Area as having potential; a Stage 2 was recommended for the entire original Study Area.

2. The Area south of Snyder's Road will have typical recommendations for any artifacts we recover (so far we have identified two areas of artifacts in the test pit portion; we are still awaiting ploughing and weathering for the rest)

3. The portion of the corridor north of Snyder's Road East will be recommended for Stage 2 pedestrian survey of all agricultural land, as well as a recommendation for intensification for the findspot (which will have a specific findspot designation) according to Section 2.1.3, Standard 2 Option A or B.

I have not encountered this before, so I am unsure how to deal with it. Is this approach valid, or is there another way forward?

Thank you in advance for your help.

Cheers,

Walter McCall, PhD Company Director Consulting Archaeologist P389 www.detcon.net (226) 922-0297



Appendix C

Cultural Heritage Evaluation Report (CHER) (December 2021)



CULTURAL HERITAGE EVALUATION REPORT

Review of Allerthankow 145 for the vachts frumk Samlary Dowor Munimezi Classifins sommer tal Association

ta -: Decembers 2021

Wan. Foat

- :+ - l :-/ MacNaughton Hermsen Britton Clarkson Planning Limited (MHBC)

Best

11. (15%

Steel on genand en et de Obtenet (Constant)s Philosoph (MC Millson (MC)



Sich

MYCETS

Liels

A.A.

Le Alex

Ale a construction de la Succe La construction d'Arrich de la construction de la Constru

Table of Contents

| Project Personnel | |
|---|----------|
| Actnowledgement | Þ |
| Ginesgrych Alapreziations | |
| 176 - Executive Summary | б |
| 7∧ Inπhouedan | s |
| Constrained the solution | 1 |
| Miller Legelice in | |
| 7.4 Folloy context | |
| Contention of and the CCS | I |
| i al contractore d'al experience de | 1 |
| Contraction and Appendix 1 | |
| in example is deal and is the | |
| (i) It dependent Weiger op einer de substantion des la filles des la base des substanties (Comparison et Comp et 2011) | ⊾·I |
| i statistic statistic | • |
| 4.6 - Olgorital averatew | 71 |
| a filme en al energier. | I |
| Recent warman, Reeconomy was enough and many second second. | I |
| 5/A — Description of Proposed Barlen Trunk Senirally Sever Alternatives | |
| a for the second sec | ·: |
| G/A Eval, adion of Proposed Alternative Foures | 71 |
| лі жни сі ни | i |
| n a tagain the fail sources and consider the total sectors and becauses | i. |

| an a sena na selata sena sena di segna sena di sena di sena sena sena sena sena sena sena sena |
|---|
| 7/2 Conclusions and recommendations |
| The full easier with the research of the second off the latter of the State of the |
| 8/A Felerence: |
| - Appendix All Mapin: Droader and Local Study Area and Alternatives |
| - Appendix D - Map or Fulcarel Heritage Properties Hemitikality the Township or Wilmot in Felado pro the Orrader and Local 30.49 Greed |
| Appendix C+Minismy of Tourism, Culture and Sporr Cilleria for Evaluating Potendal to Cultifilier cage Resources also C. Ioural Heritage Landstaper Checklist |
| Appendix D. Fhoto Map |
| Appendix F. Excel proof the Township of 'Allmost Ledrage Register uptoperdes incates "Appendix F. Excel provider 30. dy Gree) |
| Appendix F- Currie, Jum Pilze |

A di basa karang di Kasar Karang Karang Karang Karang di Karang

Project Personnel

| , General and State (State 1997) A Million | una ngagata na séla angat Pangan | r gert star soen |
|---|-------------------------------------|----------------------------------|
| terenation 24 - 4 P | , est a constraint | والموجع والمتعالية المراجع المعر |

Acknowledgements

this eran all novie decide variant lie force in this another all the solution and ends are readed at the life and all others expected on the ty

| ave Constant and Anna a | an California Anna Martin Anna Anna An Anna Anna Anna Anna Anna | $147.0 \le 2$ |
|-------------------------|--|----------------------|
| : 1 | ur († 1995) Staat Weel, Weel, skrie skriet fan de skriet 1990 - De skriet fan de skri 1990 - De skriet fan de skri | 1 (r= 1 1, 1, 5) |
| :1 | an baile anns a' suid March Anns a' Sanns anns anns a' Sanns Anns anns anns an Sanns | The Claim |
| : 1 | an bhi tha ann an Airte Mar a' fha ann. Ann ann an Airte an Airte an Airtean | A. 112.4 |
| : 1 | an foar a sentre ar earling fan Streene Anne Streef fan Kaarne ar bereke fan Kaarne ar ei Angelstere Broom Kaarne Regelstere oar berek terenstere | fersen bestind in de |
| н. | a lana sa magang sang sina sa sa sa sa Na sila sa sa sa | A. 202 |

Ale a construction de la Succe La construction de la Construction de la Const

Glossary of Apbreviations

| 4 | and the second |
|--------------|--|
| 44 | الأنافي محترجون وأنجر ومحرور |
| 1 | ender en en andere andere en |
| ч н. | and the set of the second second |
| 1: | , waari wa gomma na filinga ƙwallon |
| vi . | and an |
| 9 1 : | ala any desirate any fisi sa ani mbaka Anatatea |
| 41 A | et al succession and a specific de la seconda de la se |
| с в | and a second second second second |
| (Geology)# | ana kanalari tahir da Kanalaring Kanal Manalari kanalari |
| · | and the first warmen by the |

1.0 Executive Summary

of the woodship of branched as worthout test cost with a non-spin for the relief through the features. Nexes of the 10 local containing the Assessment the the viewers raised in the invariant Baren. In Transmission and the state Malageer the geven of the units the state of the same of through the state Test.

The compose of the event window decision is the the events of a few model of the first and the event of the events of the events of the event of the even of the event of the even of the even

Conclusions on First update of Barler Truck Sankary Sewer Algematives (* 153)

- Even the function is adhered context, one works are usingle that the power the formation let use the idea take mean representation in the event of the population the present representation is a power of the Alternatives on the racted on the may represent conduct of Appendix Balliance to:
- Mark of the K smallve compared to the represent tells brown on the service of a single rest operation of the same ther to design a service restrict evolution metalway are ben field. The period prevales and
- The septons Weens the decide of each agent option and on the target action paranti-strictly the action methods are seen as the treat as seen to was in the methods are a strictly action of Methods and the tops are methods and on the tensor and operative control for each sign of the tops are to prove and on the tensor action particles of the strictly of the tops are to prove and on the tensor action action.
- Determine the complexity of the exercition of the complexity of the exercise of the complexity of the exercise of the complexity of the exercise of the exercise

The the conduction on the description is even that the view of a contract removal even on a schema schema is the radia to be for a schema in the track the schema is schema in the schema is a dimensionel in each other for a schema report one ther base of a schema is schema to be expressively decreman. Ale a construction de la Reserv Les relations de la Reserve de la reserve de la Reserve

2.0 Introduction

Matter versions from contract die er informatien bezeichen einer versionen auf die formen bezwiehen wird die einer versionerse nationer einer einer die konstant und versionerse die konstant die nationalise record ein version fragen einer die nationalise in wirdsproche tract.

the amove althous a second second second reacting to demonstrative movements of the second second second second second reacting the demonstration of the second sec

a work of a second price of the precisive of Release are set of a three second properties of the next to a second contract the precisive effect the antimate convolt on the contract of second to the Work and Work environmental the antimate convolt on the convolution of the contract a second provide of the action of each of the action of the tracted to the tracted at the contract of the contract of the second of the second of the tracted to the tracted at the contract of the contract of the second of the second of the tracted to the tracted at the contract of the second of the second of the second of the second of the tracted to the rest of the second of the tracted of the rest of the second of the tracted of the second of the tracted of the second of the secon

2.1. Secret of Mary Steal

The study allowing consider that how the trader process reaches a second state of the second base of the trader whether that have the state the scale of the scale of the state the state the scale to be a state of the scale is far scheme that the trade scheme for scheme the scale of the scale of the scale of the scale **Figure 1** (4) w₀



Figure 7 in operations energy on each technological and a discussion entropy (Ase Appendix Appender of the first technological

2.2 - ee mige Service

The considerations are an an energy of the energy of basis and the deviction for the formula of the resonance for the reversion of which is the term of the first energy of the Chinak term Sender. The energy of the first of the major constant basis of the energy and the **Appendix D** r a la contra la succesión destri Esta e la contra de la contra de



Figure 7: movies to be easing the relief tree and the second system of pervention end of the movies the local statement of extended the tree end of the Societ for Societ Markov (Societ Society), 2 of 1

The following provides and on these surgeries is a set within the provement of a set within the effective structure of the set of

| Address | Heritage Status | i otaretAtjacencioAligo reojisu No: |
|--------------------------|-------------------|--|
| and the second second | | • |
| line græfe stær fær | 1. 2 | A A C A PO PO PO PO PO A SU PO PO PO PO PO |
| a fin krienk Greni - Sed | 1. 1 . | Definition de 1200 Definition de 1200 Definition de 1200 Definition de 1200 |
| n in sters bred steri | 1. 1 . | Definition de 1/2 m Definition de 1/2 m Definition de 1/2 m |
| ale elgit u | 1. ph | Marka setto 1 Marka setto 1 |

A de la construcción de la Succe La construcción de la Maria de la Const

| 11 M N H | 1. A A A A A A A A A A A A A A A A A A A | • All-ranke 1:01 |
|--|--|--------------------------------------|
| | | All shares to be |
| | | • All shares to be |
| the first of the sector of the | 1. ph | • All-rankeiter |
| "• _0••0".0 •• | 5 m | • All shares to be |
| | | • All share the first |
| a se de la service de la company en en | | |
| n di seri di peri | Contraction and A | • Alexander : |
| n i fa scherk Greit-Med | Cestra destrata del | • All short sectors 1 |
| | | • All-ranke Vroll |
| Bin generation from | Cestra destrata del | • A A C A PT PT PT A A C |
| | | All short set |
| is he shere the liter | Cost des et al | • Tell - Contraction 1 |
| | | • Alexander 1991 |
| | | • All-ranke tool |

• Constant for the first first the number of each sector sector densities as a sector of the sector of each sector is an each of each sector of the secto

for other since the other and the end of the second since one of the second since the second of the end of Telefont Spectra of operations as end there since all contractions (1) telefont the other efficiency in other prefix exists in a second second since the second since the second states of the second second second

na entre a la seconda en la forma de la seconda en activitation de l'analises de la forma accesa. La seconda de la seconda y a la seconda de la forma de la forma de la seconda de la seconda de la seconda de la La seconda de la seconda de la forma de la seconda de la seconda de la seconda.

Ale a construction de la Reserv Les Martines de la Construction de la Const



Figure 7: we shall the sequence of a second second contract on the second s

word formal inflating on the subscripting the subscripting both ansatz to the subscripting design of the subscription of the

relief foar allef heer hier fried en de de die namelike te een noop efficiel. Noor een alle een ne Graat nij heer Bennand Waler production op die namelike noor geer **Figure 4** heer eg



Figure 4: the shall be any provide the first believed by an University because for the set respectively expected as a 200 p

r a la compositiva de la Succe Esta en la compositiva de la compositiv

3.0 Policy context

3.1 to fairing Arrans in 2010.

The Characterization revealed in the off advance method mathematical and where the screen zontemport of the All prior temporth expecting provide endowing advantid providing for a featimetry from the dation revealed for zone endowing the data detection devices developed proteaction endowing the dation revealed for zone of the method scale for a sequence to the featimetry operation operations and the dation revealed for a scale of the method scale for a sequence to the featimetry operation in the method scale of the method scale of the method scale featimetry operation is the featimetry operation in the method scale of the method scale of the scale of the edge of the scale of the featimetry operations are to be in the method. The scale of the scale of the scale of the edge of the scale of the scale

na an ann an t-fhean ann a' chuir ann an taobh an bha ann an taoir an seannach taoir fan taoir an taoir an tao Taoir an t-fhean a' ann an taoir ann an taoir an taoir an taoir a' ann an taoir an taoir an taoir an taoir an t Taoir an taoir an taoir an taoir an taoir ann an taoir an taoir a' an taoir an taoir an taoir an taoir an taoir

an an an ann an Araba Araba an Arab

the clinication and there are provided for the internal report. This exactly, and the book is the structure to a resoluted through the latitude state of provides.

an di kan di patrika di kana kana kan kan di kana di patri na kana pana bata ng kan di sa Na na sanati

the character is constructed in the activities

SignWeight and the second second residence subject to subject a second state of the second se Second secon second sec Collect of the bogs 1 processing, and the collection of a structure of the detection of the construction of the collection of the colle

$3.2\,{\rm er}$ via the rest of Assessment even

The conversion of Alexandra Alexandra pression which perturbative transpersion of a systransmit over 1 whet was even a subtransmit operation approximation economic solution above against all 1000 With a compare for

n ar energia de la constante d A constante de la constante de l

n var ander fan de beerde oar en de streen in de bester en de bester kerde op een de findaar. Die en die selften die tekense oorde waar eerdst

tian in independential eacher eacher de arte in the categories of Astronomy, for the states that the entropy and eacher (new article) is the former of (

A is the set of the transmission of the transmission of the transmission of the A - A -model of the transmission of transmission of the transmission of tra

and the encode element of the transparanta term the elementation. The creation esta patholic transmission Avec section,

3.3 Considerating CAst

The construction of the set (19) by the by the construction of the

3-1-region en Waren de Cirtefe Bae

Teal on the difference point of the end of the ended to be a subscription of the ended on plate measurement of In continue the temporary many the only of the properties a second and the separative states are the death of the subscription.

alam da ana arista da ana barang barang sa

(2) The subject of the statement of the second statement of the subject function of the statement of the subject of the second statement of the subject o

(2) The second of the second second second second second in the second secon

(2) A second of a substance of the second state of an analysis of the second state of the second state

i gran ann an an an Anna à bha ann ann an tar ann an tha ann an tar ann an taraiste an tar ann an tar an tar a An Anna 1976

Spectra distribution and a state of the part of the state of the state

i a shi shi nengan ke sa shi na na na na na ka shekara a marki sa na shi sa sa sa ka ka ka sa ya sa shi Mani na ƙwallon ƙasar wa ka sa sa sa sa . Na kaominina mandrida mini kaokana dia minina mpikambana amin'ny kaodim-paositra dia mpikambana. Ny INSEE dia mampikambana mpikambana mpikambana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana

المعتقد والمتعاصفة والمعتد والمتعاط والمتعاقبة والمتعادية والمعادية

n palen dari by Siet and Antonio and an anna an grandil an an diffication and an an

t e navenske standen in de stêre. Die steren is Ander Weiter in Marie provinsier. Einstelle street werde tere oorgeneer.

(2) A second second black of the matrix o

(a) which is the constraint of the constraint

(2) If the state of a state of each second state to the state of th

(2) The end of product the number of the second state of the second state of the second state and the second state.

7.4 England at Avera the source Breaks function and united to the test sectors. Deputy new (2001):

ta selena Sterilere i la monta falle el secone directes sue la craena del constructiva di la segura difi Necesia oficienza une planta per a plana per treba se la Seconde do Lamen, Cabita del constituità. ie neuro spans de la centra por des melforativos e la contraes procesarios des consuleros. Les espansiones procesarios per des consules com

- A second s
- Contractions and the contraction of the second states of th
- Receive a second solution of the second secon
- A second second process of a transfer of the second second second second second process of a second process of the second se second sec
- Construction of the product of the second of
- Constants from the constant of th

and the an upper expression expression ender the end of end of the desired or efficiency of the second statements of the

to actual model ensants the second styre

- The second state of a second se
- The first sector is the structure of the sector sector, and the structure of the structure of the structure of the structure of the sector sect

Ale a construction de la Reserv Les relations de la Charl Maria de la Charl

Constantly and a difference of the conference of the constant of

والمحمد والمركب والمتحد والمتحد والمحمد والمحمول

na bernanden en en en en en de service de la constant de la constant de l'entre de la constant de la constant La constant de la cons La constant de la co La constant de la cons La constant de la co La constant de la const Manda de la constant de la constant

n en la secta de la subsecta de la subsecta de la subsecta de la secta de la La secta de la La secta de la s La secta de la La secta de la s

ena da arra

(a) we can set the conductive conductive resolution of a conductive conduc

```
Ale a construction de la Reserv
Les relations de la Construction de la Const
```

direction and the second devices of the solid measurement of the provident of the second second second second s An example in the second second for the second s An example in the second se

5.5.0

(i) a submatrix a constraint for the state of the state of the discription of the state of th

3.5 , considered Witness Official task

e transension i nel razione professione i i finita i sur acta devilta si i sono prione en ediciti e state. Esti response or lloc di l'i patti entropa partia di l'acta di

and the Werth Cleans second in the new enterth

(a) a constraint of the second of the constraint of the second of the state of the second of the

 $\sigma_{\rm eff} = 1000 \, {\rm eff} \, {\rm ef$

en de la servición de la companya de presenten de la complexión de la comp La complexión de la c • A the first of press of the statistic state of the s

a the state of the second

na an tao amin' ny faritr'o amin' Ana amin' amin

 $(g_{i})_{i\in I}$, we consider the transformation of the transfor

a fra a new sector a construction de l'esta da construction de la construction de la construction de la constru La construction de la c

 $\sim 1 - 1.5$ and the elements of the state of the 1 - 1.5

e Martine and an an article and a first and a figures are according to a single of the second second second sec An and the provide structure and article and a second second second second second second second second second se An and a second second second second second for the second second second second second second second second sec Ale a construction de la Reserv Le reference de la Construction de la Const

4.0 It stor call overview

Eregenouerrenert.

In the event watch when we direct the internal plant of the state of a set of the fact the matter of the state of the s

I. Saach, Without Reserving, Write Rep Courtes.

The online is an owner water of "Asterio on the view expression error of the set of the set of the online is a standard (f) the solution of the conversion of the expression better of the online of the standard methods are from the View and the transmission of the solution of the online of the standard methods from the Conversion error water on the solution of the solution of the standard methods from the Conversion error water on the solution of the online of the standard methods from the Conversion error water on the standard methods are solution of the standard from solution of the Conversion of the Solution of the Solution of the solution of the standard from solution of the standard solution of the solution of so

Net no the second case as more beserved to set them as the function of the bulk of the basis of even to reperfused V-an on the form the second second case enrolling them as a more set who ca make the basis of an order of these memores in were developmented them at the theory of much device a schemed memory of these memory ends of the Previous second to the second second of the schemed memory of the second case of the second second to the second second to the second of the schemed memory of the second case of the second to the schemed second second case of the second to the schemed second second case of the second second second second second second second second

Notice the end of the one transformer wave a set of plane are the fermewide -0.1 of the maximum term of the field -1.5 when -1.4 when the velocity terms of the field -1.5 when -1.4 we define the velocity terms of the set of the field -1.5 when -1.4 we define the velocity terms of the set of t

The environality of the control VVV of All increases and Erver control with a control of a field of the control of the environment of a control of the contr

the are aller for the stand for the perfect that can be yndeen to be dear to the structure of an order to a bra Telen of a system of the later the later the restructure for the standard structure of the structure of the dear Telefore structure of the standard that the structure of the standard structure of the structure of the structure Telefore structure of the structure of 1000 the structure of the standard structure of the s



Figure 2: we want her both the entry word lenking statute matter and entry of entry and provide the entry set.

It when she shall be easies that her as this demonstrations wheeling in terms



Figure Conversion the Lemma e bag of which the control of 1.6 and a single or even element. They example a second statement of the solution of the example form the terma king the stage the solution of the Physics

We can find to the transformer end of Sected on a unit. The end describes of a strategistic sector. Some processing to scenario equipant using of the over such the Courteen over the analytic metry recently interview in screecess of each doment, rectaining a radius to be the fighter of Chemical matching and such as country in Rec. (The cellisment end of the assessmpts and your help to see the transformer such you Rec.) The cellisment end of the assessmpts and your help to see the transformer such as some very all therein Chemical and the problem provides the metry end of the provides and end of the second of the provides of the provides of the second of the provides.

| ALPHABITICAL LIST OF I | BOWESSIONS, TRADES, he |
|---|--|
| ALWIARDITICAL LIST OF I Arkel, Arder, willer, Inere, Lewis, Servery, HECK, JACOB, propressor of Re- den Rearing trails and ten- Sonalty. Hec, Garles, w., Islamer, Ras, Charles, w., Islamer, Ras, Charles, w., Islamer, Ras, Charles, w., Islamer, Ras, Charles, w., Islamer, Rasser's, David parter at matter. Cray, Berry Blackwarth, Relinger, Berry Blackwarth, Rome, Gether, Isbaner, Rome, Charles sadder, Rohet, Flaip scoper, Rohet, Berry, Barter, Barter, Rohet, Berry, Barter, Barter, Rohet, Getry, Progen-maker, Rohet, Adorph, Samer, Rohet, Adorph, Samer, Rohet, Adorph, Samer, Rohet, John Adaes, Heidemitt, Rohet, John Adaes, Rohet, Berne, Bernet, Bernet, Rohet, John Adaes, Rohet, John Adaes, Rohet, John Adaes, Rohet, Bernet, Bernet, Rohet, Bernet, Bernet, Rohet, Frank, Bernet, Rohet, Frank, Bernet, Rohet, Bernet, Bernet, Bernet, Bernet, Rohet, Bernet, Bernet, Bernet, <td> BOPESSIONS, TRADES, Im. Caliz, August, farmer. Miller, Ourles, willer. Miller, Ourles, willer. Miller, John, Isharar. MOBLET, MICHAEL, recent are will, and proprietar of pill & stars factory. Masar, Prest, Wesser. Myses, Malanel, tawentile cAr. PERINE, J. H., Farme Brothers, PERINE, J. H., Schereller, Barren, Brother, John, Miller, Barren, Schleicht, North, grownil surveitant, Schleicht, Jahn, & Girmes, Schleicht, Jahn, & Girmes, Stattaborger, Henry, Sommith, ample, Adat, Marten, Martin, Magner, Henry, Schlemater, Walter, Bartert S, elort Wagner, Christian, wei, Senart, Walter, Bartert S, elort Wagner, Henry, Controler, Walter, Bartert S, elort Wagner, Henry, Scherneller, Walter, Beriset S, elort Wangle, San, muchler, Walter, San, muchler, Stattaborger, San, muchler, Stattaborger, Stattaborger, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Marter, Stattaborger, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Wangle, San, muchler, Wangle, San, muchler, Marter, Stattaborger, Schereller, Sch</td> | BOPESSIONS, TRADES, Im. Caliz, August, farmer. Miller, Ourles, willer. Miller, Ourles, willer. Miller, John, Isharar. MOBLET, MICHAEL, recent are will, and proprietar of pill & stars factory. Masar, Prest, Wesser. Myses, Malanel, tawentile cAr. PERINE, J. H., Farme Brothers, PERINE, J. H., Schereller, Barren, Brother, John, Miller, Barren, Schleicht, North, grownil surveitant, Schleicht, Jahn, & Girmes, Schleicht, Jahn, & Girmes, Stattaborger, Henry, Sommith, ample, Adat, Marten, Martin, Magner, Henry, Schlemater, Walter, Bartert S, elort Wagner, Christian, wei, Senart, Walter, Bartert S, elort Wagner, Henry, Controler, Walter, Bartert S, elort Wagner, Henry, Scherneller, Walter, Beriset S, elort Wangle, San, muchler, Walter, San, muchler, Stattaborger, San, muchler, Stattaborger, Stattaborger, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Marter, Stattaborger, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Marter, Schereller, Wangle, San, muchler, Wangle, San, muchler, Marter, Stattaborger, Schereller, Sch |
| Roch Erar, surpretty, | Warnyh, Sara, muchden. Waha, Jaha, seard invare. |
| Knus, Clarist, houst krepet. Kenny, David, farmet. | Weiler, Jahrs, surprotes, Witte, Charlun, neperater, |
| Krapi, Christian, Arrow, Long, Bleary, Moorer, | Zurmaki, Win., rengent Zunn, Guotun, unrperare. |
| Lierada, Hroes, farmer. | |

COMPANY AND A REPORT OF A DATA OF A DATA

Figure 7: we shall be a ther and they you've test shows each and beening 100 million and a second se

Words and built equivalence of some all efforts for deal or 0 and y the mean start end of effective it was an end of the other than the some of the second deal end of the solution of the efforts for a second of the other takes.



Figure 8 to enable the second burner ended water to Chart 1, 2000 and 5, 2000 and 5, 1000 and 5, 1000 ended there ends the early second by developing the second burner end the statemeters (which expected burner end) water ends (1, 1) is

woons not of the site and promotified environment of the state of the


Figure 9: see all of the software questions of enables and substantial distances interesting over intertimentee detailed and a see all or sharing exactly intertimental values of an all to staff applitions generated for it.

5.0 Description of Proposed Backen Trunk Sanitary Server Alternatives

5.1 intersection attack

The invester is an Alexie clean of an isolation of the main operation of the relation of the set water definition of the main of the main of the main operation of the relation of the rela

r a la contra la recipión Succión estre la contra de la Contra de La Contra



Eligine 10, we shall be entered at the Allen shows Very noting the allen's off, Area eliginesis and which entry area of 1, 2015.

al magninume time and an informe composed. Allem subscriptions is a methodown much permana speceric Tende a time entitient of the transference of the scription and tendent of Appendix (Burn Home scriptions) The specific tendence of the perferition of fraction y revenue of the scription.

| Alter varioes and Descriptions | Adjacent to Identified Letitage [| | |
|--|---|--|--|
| | Festives liceated as | | |
| Alter variate 1 | " d Specificat; | | |
| and the state of the second state of the secon | • dir niçile lder | | |
| 14.5%.9j | • в толяй са модел | | |
| Environmental e examinar provincial de la termental de la companya. | | | |
| a sua asin'ny si | is normal particular. | | |
| The off-size vectors set the contract the contract. | | | |
| when every studied environment of the Unexample of the | The other converting to end | | |
| The Alexander result of the normal sector and | The other Classifier Classifier | | |
| to a simple when a public work is well when the | | | |
| where energy costs are needed to be Repeated with | | | |
| Teach Angling Switch I was deplete a street. | | | |
| s we first from the only these and the sectores | | | |

Ale a construction de la Succe La construction d'Arrich de la construction de la Constru

| The second second second second second gradients | |
|--|--|
| o le la marco dobre la la presente secte | |
| a any the easily of their than services. | |
| o ja restar joja – restar ja entrustar Hra | |
| A provide the second | |
| • vit a "I | |
| Experimente order di server la cum esperimente | |
| a sequences a based | |
| Alternative 2 | In the end of second sec |
| and the second second second second second second second | ·- |
| with a second state of the manufacture of the second state. | • References and the first first |
| Technologi By Levie Brouchevers the non-service state | •= |
| Transfer v Steri - Sediens i statistica e el | A state of the state of the state |
| The second second requesters with p | distanti y |
| o verson stranderen verson diese verson | All states have been been |
| e free e charer | -land |
| ○ 	 10151 it 	 1#11+1116#11 | • All only the tales |
| o laterine etc. e cetter a mi | ر مرداد معر با ال 11 - • |
| Extended and marginal contract execution | - |
| • wi* | |
| Alter variate 3 | Schwarz erste sonsteller, en |
| a na shekara na shekara shekar | • |
| A write the structure between a contract | A set to set disc to |
| To each other statistications and the statistication of the statis | a state for a lase |
| statistical frances | distanti di |
| The second second requests of an quantum second seco | all states level level |
| o orași din angelere de profesiola și oraș | Alse . |
| the training | Contracts for each locarth |
| ○ ╹ ''' 4 ' '- ₩ : | Contract to provide a set of the set of th |
| O Istanjakasi kanadaka | |
| o lateries en en en en entre sette a entre | |
| equivally of | |
| Alter partice / | "A presentation of the p |
| encompanya terdenak persenan terdenak di | |
| Setting at the set of the example of the fraction. | |
| | |
| eter of the set of the set of the | |
| n han an an ann an Alban (an Alban) an Alban (an Alban) An Alban (an Alban) | |
| One of the order of the presence of the off of the order of the order | |

Construction of the Subsystem of the Subsyst

| ○ '18'()) 414 ' ; | |
|--|--------------------------|
| o dilassana ora ng | |
| o vandar va extern | |
| o laterine e entre tiere | |
| Exception and interpartic servers estimate | |
| e-gravet Barel | |
| Alter variag 5 | معرر <u>المحالي</u> نا • |
| Processing of the company | |
| Perking an Hyber value in electric on the front of sectors. | |
| chart and new writing in the sec | |
| The second even the time test wing | |
| o ostanti di tareksina tareksina tare | |
| ○ 'IC'(I 414 ' . . | |
| ○ . # \ [#4 • • \ ["] | |
| o vanalar va estesen | |
| o laterie: e entre tiere | |
| Excention and integration servers electric el | |
| •••••••••••••••••••••••••••••••••••••• | |

6.0 Evaluation of Proposed Alternative Routes

$\lesssim 1$ -induction (don's

The following summary of the comprise elements of fills frequency of the result of end of end of end of elements of the following summary subscription of the summary subscription of the summary subscription of the summary subscription.

the following provides on the single frequency of standards and standards in the set with the set we the single function of the single frequency of the

- Description and operations in consideration of the second sec second sec
- Attendion of shell she directly in a complete bound work of a second a process.
- Shadows a set that due to relate explore a full or the anti-sub-transformer to a set as near 1, a or so dues lead test. O so that the set a name.
- Direction indirect Obstructions, and four inservation in equilation from a software end of the feature.
- A change in work uses in a constraint of the end of the end of the set of the set of the set of the determinant of the end of the
- Land disturbances: the variable in taken of all on a minimum equations are ensured as a distribution and the second.

The most shifts have set overlapment of dama by multiplicative tops each or the decision of the test of equilation processes of the most end of a first state of the set of the set of the orbital of the set of the decision of set of the most of the set of these imports to set of the test end test of the measure of the decision for the set of the may have any manetale to high tests of the set of period.

$\lesssim 2$ -values based at a tracker space. Monthly, if using the tragely even track

The althous because of a feat of the electric conservation whether more althous as well they consult to access electron to contract the table Beach desired of the electric electron effective contracts of "electron at the begin to "events of the electron of the electron desired at the table to activate the tage term endes instead in the Data should even on terms of the tage term of the table of the tage term endes instead in the Data should even on terms of the tage.

• I are a the model with 6 terms measured to be with the control there move to and the bit of the operation of the terms of terms o

- Reversely Alleman sector frame in a section of the provincing of the function of the task record enables we have been alleman. The end of the section of the section of the section of the section of the following the performance because of the test result for the section.

- Definition we can be considered if the end of each definition of the first the trapped Block. Next Constitution gale is specified.
- Set even we are a set and the only of each of the environment of the set of the devices.
- The second se Second secon second sec

At a reaction of the next the encoder of very the trace of the maximum vertex that works the direct of the trace is the trace of the tr



Figure 11: C-I and the IX Constraint for the end of a standard standard standard in the second standard stand

La china toa no avaita le can be terra of valence den les partices de line de l'havait (1,2)c) estitues de local des les propositions et anticipationer sub-elliphicoper y mobile to the local definition content de local terra terra de line est le entercopy d'ultre line estat finition a lug typic de line de centrars ales de l'Alema de l'han d'estat estat avait de la complexit estat activation de line de centrars ales de l'Alema de l'han d'estat estat activation de la complexiestat activation de l'hande centrars ales de l'hand estat estat activation d'estat estat complexit a (1,2) (centrare) n a chuidh a' chuidh Guile Baile Curainn an Conaigh an ann a' Conai



Figure 12. Celor of Herlicos, the maint Herley of the target any new endered draw. Mapping the material enders the impact and a set of the se

 \lesssim) we loss that Alternative simples is a power 4 measures. He tagget induces a second system to be a simple space of the unit space of (

ter do the Alexander Southe prost of a solution existing expension parts from the prostant at an expension of several of the execution provide the control of a solution to the existing of the existing and examples of the execution provide the source control of an existing to the total of the State and the existing to the existing for the second of the existing to the solution of the existing to the existing to the existing for the second of the existing to the existing control of existing to the control of existing to the existing to th r a la contra la recipió destructor estas formas en altres Silver a contra da Contra

When shows it a sum to be bounded as an optical depicted on a state of the state. But the term is the final state we don't the existence is the formation of the state of the state of the state of the many state of the state of the existence is the formation of the state of the state of the the state of the state of the existence is the formation of the state of the least of the state of the least of the state of the least of the state o

Ale a construction de la Reserv Les relations de la Reserve de la reserve de la Reserve

7.0 -Conclusions and recommendations

$7\,1\,{\rm submatrix}$ with a constants that a regarder Second encoders.

Harving the Alternatives converses in this end are been presented by the elements of terms of a the net type end many president to be defined on any presentes to the transmission end at an element ferror the net type excision formally there are the source element of the which reprint that terms element operations formally there are the source element of the index to control a net term to be excision for any there are the transmission element of the source of the source element operation to the element of the source of the transmission element of the end of the residue to the element of the source of the transmission element of the other element of the source element operation to the element of the source of the transmission element of the element of the source element of the transmission of the transmission element of the source for the transmission of the source of the element of the transmission element of the averaging the transmission of the source of the transmission of the transmission element of the element element of the source of the transmission of the transmission element of the element element of the source of the transmission element of the averaging the transmission of the transmission element of the transmission element of the element element of the source of the transmission element of the averaging the transmission of the transmission element of the transmission element of the element element of the source of the transmission element of the transmission element of the element element of the source of the element element of the transmission element of the element element element element element element element element element of the element element element element element element element element element of the element element element element element element element element of the element element

In one of which every solution of the end of the value of the control of the term of te

r a la contra la recipión Succión Esta a la contra de la Contra de La Contra



FIRe electric de la contrata de la Academica de la Anna de Contrata de Contrata de Contrata de La Contrata de Contra Contrata de Contrata d Contrata de Contra

The second variable of the second se second sec

of a single for the second single state to the second state of the second second second second second second s

of a single state of the version of the second size of the size of the second size size of the size of

where you'l according to exercise on the according to the second s

chaving of the state of the end of states of the state of the states of the

(c) as yes? Based of the elements of the class factors of the first based on the value of the value of the elements of the second se

el ne gran i la succession de la constructiva de la serie Santa Constructiva de la serie de la serie. El serie que prote

"Received inductions of the other than the

al an an tao ao amin'ny faritr'i Anaton Mary ao Sanarah, ao amin'ny faritr'i Anatona. Ny INSEE dia mampiasa mampiasa

a Karang

Appendix \mathbf{A} – Map of Biologen and Local Study Alea and Alternatives

| LEC | GEND | |
|--------------|---|--|
| | | |
| | PROPERTY LINE | |
| | EXISTING SANITARY TRUNK | |
| | EXISTING SANITARY FORCEMAIN | |
| 326 | EXISTING 2015 GRCA CONTOURS | |
| DR. TRU | AINAGE TO FOUNDRY STREET JNK SEWER | |
| DR. NA | AINAGE BY GRAVITY TO FZIGER ROAD TRUNK SEWER | |
| DR | AINAGE TO FOREST GLEN WWPS | |
| DR. EM | AINAGE TO WILMOT PLOYMENT LANDS TRUNK WER | |
| LAN | NDS NOT TO BE DEVELOPED | |
| GR | CA REGULATORY FLOODPLAIN | |
| GR | CA EXISTING WATERBODY | |
| GR(| CA EXISTING WATERCOURSE | |
| | TOWNSHIP URBAN AREA | |
| | COUNTRYSIDE LINE | |
| | PRIMARY STUDY AREA | |
| | DETAILED STUDY AREA | |
| 0 | PROPOSED WASTEWATER PUMPING STATION | |
| 0 | EXISTING WASTEWATER PUMPING STATION | |
| ALTERNATIVES | | |
| | 10DIFIED 2014 ENVIRONMENTAL ASSESSMENT PREFERRED ALTERNATIVE (DO NOTHING) | |
| 2 | NYDER'S/FOUNDRY with and without pump station n Activa Lands) FORCE MAIN ROSSING AT SNYDER'S/FOUNDRY | |
| 3 | CHRISTIAN/CHARLOTTA/MILL/ CHARLES/FOUNDRY | |
| 4 | CHRISTIAN/CHARLOTTA/QUEEN | |
| 5 | CHRISTIAN AND BADEN CREEK | |



P:\P\48730\100\48730−100−F2.0.dwg



Figure 1: Sewer Alternatives Overlaid on Township of Wilmot Official Plan (2019): Map 4.1 Baden Urban Area

Township of Baden Township of Wilmot Region of Waterloo

LEGEND

Sewer Alternatives

- Alternative 1: Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing)
- Alternative 2: Snyder's / Foundry (with and without pump station on Activa Lands) ----Force Main Crossing at Snyder's / Foundry
- Alternative 3: Christain/ Charlotta / Mill / Charles / Foundry
- Alternative 4: Christian/ Charlotta/ Queen and Baden Creek
- Alternative 5: Christain and Baden Creek



Notes:

 For information purposes only
 Source: Parcel data from Region of Waterloo and Teranet Land Information Services and its licensors (2019); Map from Township of Wilmot (2019), Sewer Alternatives from Township of Wilmot (2022)

DATE:May 2, 2022 FILE: 0941CD

SCALE: 1: 10,000

DRAWN: CCF

K:\0941CD - BADEN TRUNK SEWER EA\RPT\LANDUSE_MAP4_1_02MAY2022.DWG







Figure 2: Sewer Alternatives Overlaid on Township of Wilmot Official Plan (2019): Map 10 The Countryside

Township of Baden Township of Wilmot Region of Waterloo

LEGEND

Sewer Alternatives

- Alternative 1: Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing)
- Alternative 2: Snyder's / Foundry (with and without pump station on Activa Lands) Force Main Crossing at Snyder's / Foundry
- Alternative 3: Christain/ Charlotta / Mill / Charles / _ Foundry
- Alternative 4: Christian/ Charlotta/ Queen and Baden Creek
- Alternative 5: Christain and Baden Creek

| | Country does like i |
|-------|----------------------------|
| 18 | Secure 14 News |
| | Sugar and a second |
| 1.1.5 | er heet toutes |
| 1 | Contract Contraction and A |

Notes:

 For information purposes only
 Source: Parcel data from Region of Waterloo and Teranet Land Information Services and its licensors (2019); Map from Township of Wilmot (2019), Sewer Alternatives from Township of Wilmot (2022)

DATE: May 2, 2022 FILE: 0941CD **SCALE:** 1: 10,000

DRAWN: CCF

K:\0941CD - BADEN TRUNK SEWER EA\RPT\LANDUSE_MAP10_02MAY2022.DWG





Appendix **B** – Maplet Cultural Heritage Properties identified by the Township of Wilmonin Relation to the Broadenand Local Study Areas



Baden Heritage Properties and Trunk Sanitary Sewer EA

Appendix C - Ministry of Tourism, Culture and Sport Offerialter Evaluating Potential for Built Heritage Resources and Cultural Heritage Landscapes Checklist



Manaley al canada. De surs are Spert De surs al Cardona Ba

Desaran ya Shekara Barat Afir 29 Shekara Barat Tanaya ya Shikara a Criterie for Evaluating Potential for Built Herringe Resources and Duitural Herringe Landscapes A Checklister to herr Specialis

- w purpose of the clock link is to care time.
- Te property serve projected at
 - CANNER bio heregy property.
 - may be of a fund harflage value
- The education of a large the operation of a state A, while injuries infinited as:
 - A second second 100A -
 - e i nen sin qu
 - Take spectra with spit AAs;
 - The second second second depends
- -receive covered enter the thethic outries.
 - Years, 44.
 - Devening the Association (3.4).
 - Appropriate According
 - Denote change and a Standardy one transmission that they is the 2⁴ erosite to 1 a tagge Proportion.
- Call and Her Juge Funder and Report (CHEK).

year of increased in an event one on more of the good one of the sink draft you may work the event of a more set of the sign works on vports Orbot,

the sector because pays to

Monify, we use and objective and hereage resources on your map, the manual result.

Intrinsic system in paragraphic factors approach.

Other check has

- Final sector provides that he has you wanted to
 - A state of the second second second second second regulation 3503.5 (Second Second s Second se
 - A stage type process was a second or space-enderse high constrained in Constent).
- Encountrative is the instruction around the second state of the second state of which we provide the instru-

| en al port d'an agentige à d'ant | | |
|--|------------|------|
| Bade, Toni Soliny Seven | | |
| . Never han being after an andra de aparte de la sense en la provinción de la sense de la sense de la sense de A sense de la sense de la sense | | |
| | | |
| Townshi o Wilcom | | |
| Hegen I. and die n. e | | |
| JH | | |
| Screening Greetings | | |
| | Υ | 141 |
| | Г | Ξ |
| H Yang please follow the procedure we watch the electric during the consistence of a | | |
| History commence on Calvasi on A | | |
| Rent A. Scheening for one on the recourteep? Gu burd Herback Value | | |
| | | |
| | <u>т</u> . | 100 |
| In writing openy (or project energies) before and found notice be prive by a liphting water. | | 16. |
| Press donot complete two establishes to a second | | |
| When propose a personal strategy and the approximate the figure is | | |
| aurementes the previous statistics and | | |
| add the checkletic if a project is with the appropriate option of a dense obtains option to protect by available way undertakes. | | |
| B second opport a unit of an effort mental opport y net | | |
| Sate Reprint Sector Control in Sector Se Sector Sector Sect | | |
| methodicytecompacty consultations in tespester (3 n Ac | | |
| B.R.n., and annual the adapt to | | |
| | Ϊ'n. | - Vφ |
| Դերի հատությունը հայուները։ Հումի | | |
| a. I deal les des graves et site et se prefecter et site Chienkov ⁺ site site autoritée les les les les les les value? | * | 11 |
| (a) And and the set of the set | Ξ | |
| ng mingkapan kanan ka | = | Ŧ |
| al tales publicant a tradication (as the Second of Advances Add) (as 200 | | 고 |
| g – sylysia – a start de si Herdan - Haddan festan – kan Hadags Hadda's s-Serve 20 Ma∏ DLAS | | Z |
| j – ligg, y – v ji o o trans. Hat se le lo o se nove end ar satut fan it e particul yn 1941 ô ôr (*4.294). Henny se eddy | | Z |
| 3) Yes to any office above qualifiers, you need to hims a public purporty a product of a. | | |
| 2.35 (gg) (synthetic term in Height), the Streament of Collins. He is a prime of the Ook of the St May effect payons the probability of the end of the manyolic term. | | |
| ng shakan shuk 20 di na 29 di ngana kao na 11 na magi kao poliwin 55, antari na manana na azaking za manini. Jangan shi yana sana ina mana panikan yana sa ni tarangan dag | | |
| End of any discovery and the relation of the relation of the second strategy of the second strategy of the gene improve | | |
| 11 Migra Palenaa yn dynastaan 1 | | |

| Par D: Sersen og So Polendu. Gellerer en Legnaria | | |
|--|-------|-----|
| | Пр., | Hir |
| Construction of very Act project areas constitute particle lanes? A sub- | | |
| a stranubjes, of a near strain power or indexident control of with or the termination dissue? | · • 1 | |
| The prime approximation is brown to be a feature for a set of the set of th | · - [| |
| tu in a Catadian Indheya Reviews Andrews | ļ | ÷. |
| terto ne da dingo enazada en Reito e 42 necesario y ser 112 | - | _ |
| Part C. Ether Considerations | | |
| | 100 | Ma |
| 3. To many local or Above product when the rest of a consensation suggesting trail dis property for the rest. | J | |
| y – y nyawana ili kakata kalendari zamini minyerkendik kany subdiazyon ina balan katalar. Adalah jibar kalendari metangai | | Ē |
| a grant set as a set of a comparison provide the backwards | - | |
| ∠ genjen – en til fer anten hertinge landssatten? | 3 | |
| Hiften to know on more of the other to the test that the and C provided an emittings building it over any extransion of the property of which it is very when we | | |
| Van meet to nive expedit 20 minute and a site | | |
| A Cultural Herlings Bysic aug (Hype 1109-990) | | |
| , ee y so it y technika aan it is de vie tre vel ner age value and eligneikers of An zopenes, so et an still sou deed of Historia and two persons (10 monet tree) | | |
| Hereign ages a Assessment (1, A) – the expected sub-cases 3, where we are interfaced in packs. | | |
| n each stair the shows questions there is go colors at to built the prior of the shape bridatepe on the prior of s | | |
| eta ya Janearo, mogany cenar andorazion-eti sufaza () et 8 | | |
| A second device studies | | |
| As a sub-classification of a polytime postariant data for the polytical data | | |
| The summery and net open a construction rate be | | |
| Administry of a second significant of a guide for the second significant of guide the second s | | |
| a constructive the property carries proposal of polasies) carries. | | |

Process have the following partial doction whereasting information related to the scheming quere and tackets.

- a clear map a rowing the fuel at the behaviory of the horizon project area.
 - A row some and small strate states r_a county over the counter for contrast response.
- The meninipal application of all propagaes within the project area.
- and Wing concentration and percent more (and all properties with the project). Us

En. more information, see the Minutesian measure, Galaxie and Epost's <u>Caracian Rolling</u>, Tuokies, Stormanis (Stormany, 6), 30: Into Int Conservation of Provincial Menage, Sugarkae

In this remarks the is lessing sectorilities apply:

- qualified paragrays prevents hele concerned or share prevent application, or viewas ogidat. Optim 2002 (0.200) revent experience in the concerned on share her region countrate.
- proportion memory demonstrations groups of organization data control of proposed to burry out on beneficially on a few performance of the region region of a state of burry of on one of the bits.

1 is the original provided acressing checkfind, methods ago or process is n and/?

An woleling devokie, me terzology o provos may a roady do in class for conflying poter blocks to this feature resources, extraory

- Diverse detaction y a manipulity.
- an environmente responsant processio provident : proceder (proceder) : proceder (proceder) : 1900
- new text to expressed by the Ministry of Ilbanani, Calure and Spark (NTCS) all, or the Ortaria gave time filts.
 Statewets & Grade new for Galaxy sparks on Now List Horizon. Protection [NEP].

Particle Server ing the between a support of Calleral Hard yes accord

a selar the property for project among seen evaluated patters and to and not to be of apliand to via provide ab

```
Respond yes for las question, including to low the transmission
```

A provinsity that the considered replicible of the little that lags value it.

- F. F. In Will entropy Tvolubility Repart (C. LUK) a negative and share occurs or parent for U. 2. Nov (By end) 110 (dvice 11) is the Meditary on anality has been external to be of outpathing logic XX, 2014.)
- The main right her box committee has evaluated the property to not colored him togo vulue or all its calls of duty minor. Institute property terms of colored terrange vulue or economic

A property may need to be intended to \$2%.

- There is written as institute nerticable situation conversion angled.
- See after a tuo is set lipple.
- Be extensible the Colored Fortupe Value does not previde the information contents y to reprove other prevides.
- the walk of one of a local effect 2007 and distinct meeting other attends on Reputations SOB and Rep0.

Note: Direct government mensures and prote codes (a bounded in doi: Regulation 17915) the continue to use the neo state evaluation processes continue useds. For the protect contraction in 2 wither Standards & Chicalboo for Conservational Invented, the table Proceeding both from previous of an approval by VEC 5.

To believe the lighter or premier projection as here been walk deel was well.

- The upplex state of s
- the projected.
- Je Mojsky of Tours in Culture and Score.

Call to the property comproject analytic design field on otherwise protonics, which is the the two Western Sectors being of others, heritage value sig.

Antigemed under the Octave Surfrege Rof.

- Lividy, Leeignan on (Earl In).
- \sim , by the herizage construction distribution (Poll V) .

an overall configuration - Part M

Ngangan sili di salar padasi

- By a municipal by-domain being of publical renders value on record (§2.7 of 8 × 00 han free free Art).
- Ay not involve Ministeryl Tourism. Cultaneuro Spotted bailly of pullural relation to be well and depleto response (CVLV). Note: To para respectee neuropeer coulque to by viet C in Av.

Healthan Conversion and Investory

A property a long the test of a distance of the stage of the Antonica Provide Antonica States Control (a −4, A line Control with use with

Events of the second seco

- e na agus de
- en statue en el gabiere el
- brailing rephilty allowable out-to solid.

be aubjected as automatic cost and the second second and the first the tracks of the forsely damage and

er e ep. To eas a version of were versionally between the trener the property of the work in part 1, styles is a prevent of the velocity regioneral of the

the print of principal of the appearant links.

- Cremented contractive and an indicate of a roll and option when
- cheering called a contrait and two

Do not a creative of Axe

- Statistical and the statistical production provides the sector of the statistical statisticas statistical statistical statistical statistical statistical statisticae statist
- The step in the back part of an experience of an experience of a sector (i) 37 of the Canado Annage Ma().
- Sector segments and the fillence of the

A state of a state of the state

Man divelog men are the office if any or wood well called its value as wellet. We have a second of the transmedy Regioner to use

 \sim 1.01 property angular system of the standard fractional sector p with p and δ of N_{1}

 according to the system of some short of the sector of the short of the background of pertagal value on the case of the sector of the sector.

```
because of a sould be a sole
```

- contraction (Mathematics)
- The split of square in the second state

```
- consider the constant
```

a subject base is used.

- Transform of the provide the Data District College Auffition of the College Auffi-
- A set of the second of Diemonial asymptotic by merclinder Plattic of Die ContractWork (pr Arc).

e property statile extraction of interstore to response to a sequence of an All sectors and a sector to be 2007 In accordance wills

- Solution (2) of the end of the end of the second secon second sec
- Average gauge a strategy of the facility (Apple 1984)

vene an indigenier two of all sponses for the internation 12 months Coffic Strenge Strenge Strenge Contervaliter The barry and you a

for non-internative, parallel

```
i e la pigne polytica de la companya inclusion de la confection de contribución de Sancola XX. († 25 ano al XX.)
```

I sus relations and solutions

A mentades with C moley of the an encodered Reports Solot coverses the supervised the s

Provincial Refinitye noticealler are noticealler the Assemblication of Castan Samu Satan Satan Satan Satan Sata The Ru

The following of Teerland, Children and Spannfell's Antonione a Latin follow wear of the spectral of the termination of a new deditional lines and pressiling it shift has no even as a read for easy Steams proportion to the state of set mapping as

Example: "for all all, contact the CEO's Regions, its <u>could all borders as</u>

Ib. Is the property (or project area) a Medine at Historic Site (or part n°17

National extension in the method of the back of the off damage points and any first of first endpoint Mary Addi How and a name of the second Advantation of the state of the second first States and North States of Mary data

Processing and the state of the Paris of the House State sector is

Set to the property (or project event needs refer the n'every Reflexy Stations Provedion Act?)

The Meddy's Number's Section on Sector Adipatence in Regime Tray suctors the earboy is the system party and in Reference decide the optimate relative proton of Addate for Netherland Sector Ships to the Reference has been se

How we have been been like If source of Babey as an Harlings Participe Stationary

36. Is the property (or projectionse) designated ander the Nertleye Lighthouse Protection Act7

like for a privation of Mathematical Conference of the second standard standard in the Harden second and a promotion The Conference provides a little standard of the mathematical standard standard standard standard year or the s

Experience of the second state of the second s

3a, to the property to: project eres) identified as a Pederal Haritege Building by the Federal Heritege Buildings Review Office?

The role of the Federal Harlings Buildings Review Offices FHOROsis to he othe leveral programment oversities hereave buildings is same. The policy applies to all federal government dependence the some star wall rendency, the three federal Kimwall Contoralises

Tormory, rformanion, confecting <u>Fadera, Herkage Disipings Rayley, Office</u>,

See a simple/ of all locans has sign dat one one.

 Is the property (or project energy located within a Utilier National Educational, Scientific and Galtural Organization (UNESCO) World Heritage Site?

A CMSSUC World Herberg Site to obtain the World CMSSOC containing problem upper Second as containing under the Chrysochemister any the this technic of the World Calibration of Antaratifier legal in a demonstrated as a containing which in tage Site theory of the method in the name demonstration, taget as

Chief M. J., B. Weit Chief D. 2020. Weiki Heritage States Chief Ka

Francis inducting the the Court of All Meriday, State and All

Part B. Bareering for assentier Colliger Henlage Value

45 Deck like property (unany, occurve) containing parcel of long the lines a state pair provincie, or federal users severally on interpret wipleque T.

Lettings rest, they are often recognized with formal progress or markets.

Flogues a classifiered sys-

- men opsi fitti
- It is a real ministration of the second secon
- Tede of ministrica or ageneres.
- Table her gevennikenderne kerdthorgen zaboral

- a second shad to a when

- inunts gained even on the or boat verlegs viga abor on wrot moreal as vis the kital or to black the something.
- Order bit alofted Seciel /s it arts #-director #+ for a layor related avoid avoid avoid a segurity a bayors.
- The Arithmetic Property of the Arithmetic Property Condition Bases
- Hole reflection Notice to be added Control in a control space operation with p Control in stars.

4b. Cose the property (or project area) contain a percei of land that has or is adjacent to a known buriet alle and/or containanty?

Here was a ferrer Source for which we change a deal to be a set of

- Deriversites Regulations, Charlo Virlayy of Consumer Satupes Iona <u>database of redutered permendias</u>
- Enterior Tenes ogkal Society (DEC) to locale reports of Charle beauteres, both suirently and no biget in a defense being, tenity plots and outs' registers.
- Canadian Courts As suffigued Explanation before additionant text.

to the number, there is even a spin on a subject so the solved so that being a sign of the solu-

(a) Darwing projectly (or no require) is rejuine proceduling all of the Constraint Anthony Konst Advantant

11. Cynan yn Henlange yw y Maria a waar al ann aw 'n ar sawed ar ar gene talgen adar, ar han a star ar de be twanglyn i sûn slên anwel ar ge

Que i pri Mankapi vez estas di frezi sua considera e se compositore e compositore se compositore de la devicit Tabley, com

Kanan Angeler and Angeler a

Bana a se sa kan saya ka baran da sa ƙwallo da sa a d

- your conservation at the re-
- makipateziñ,

4d. Does the property for project enset contain a perceiption that contains buildings or structures that use 40 or more years old?

2.4) Such advantance of a spectry of the order. For order the state of such a flow to the provident the spectra of spectra of a provident to spectra of spectra of spectra of spectra of spectra.

- The region from the subspace of all the region
- I to back and show the second
- A solution data solution
- If the problem of

Hanga ta waa aa ing balaan ni kadi adalah ni puntasi kalanggi ni sérinén én én Béng Agés kalanéné Asépin si Kadi Ang disarih sana mang magalah di sa ta Agés adalah na déné Béng Mérés

References such that the standard manufacture in the standard state of the solution of the state of the state

and the product of the second second

- e de la constan
- A set in the last of the delay
- The set of concentration of the second se Second secon second sec
- A second fit of the fit
- Interacting sector educed in the product theory of

Rener versionen er en en en en gener produktion generge namme er en hitter Anne-Kerker, 100 - 41 Sin Be<u>rkerberg</u> <u>Frenede Lynk af en</u>

Part Cost In a La contest free s

So, in these local to divergence because here an accordance on taking supporting that the property for projections, its constituted a basis of a taken a conservation constraint, any developed or size. The average supporting the characteristic and the according to a conservation of the second constraints of the second second second second characteristic and the second second second second constraints of the second se

uotoi ulivaa yksi la tekesga nugi ulivat varike tegeti otelion sistemetkinis tarte vilistvirtemaa porentelio virasvom dokimyisä: oluppioistollates onistaase

- A subspace between probability of the state of the state
- An optimization of the pro-
- A second sec second sec
-

No feel was traded. Make what know adge to approxible not amends on suggesting that the property (or project math taken were to second due to the community, network or biscort at event?)

Lon for Axe ginal how adge may more that the projection for that it also on a parker of end use international d A direction and you are very well of Nacida Internet for instance

- Acting the increase allocation
- Tarihi Me warmi
- bolk / del

Unlighted over the - due to importance to the contraintly.

So is there are involved and unoviotige or accessible documentation suggesting inside property (or project anul, new one or report of a soft re-invitige isothereas?

. Als Michael (All and a subscription of a collection and sectors and the sign system of a straight subscription MTV 2010 with V inde My a subscription of the community

For experiment for the second second on the second many maximum and the second second

For more information on Oracle braidway (Archive) and the Report of

- Education Alter ginal Constant Report community measurements in the may never information on overthe values and the second second interview participation of the second sec
- nunvije svalega osnit dese pri ozaliternega ogan de pro-
- View syll how ball Society's Heringe Directory for all explinition accurate and heritage cryptimes with seven avelage.

And a small value on may lind helpful resources line using:

- misike call medius
- Nation callwalking fours.

Intribute the tage management plane

- curulal herhage, anderade to effect
- municipal principal stand.

the second an appendic to an an analysis to explore the support to be the

Ale a construction de la Reserv La Referencia de Carlo Maria de La Const

Appendix **D** – Photo Map

Photo Map - Aeris, Photo of Broad and Local Study Areas (Baden Trunk Sanitary Sever)



Photos 1, & 2, 1610 how of stole many blemonite Church et 1616 process Read West Took removalment in the source share share Victor (Victor) in the Victor of diffeomorphic access to 1100 Strater share West Took removal cest from mathetics of Straters and West Took rectified with the king access to 1100 Straters access West Took rectified in the Stater of Straters and West Took rectified with the king access to 2000 Straters access West Took rectified with the strater of West Took rectified with the strate of Straters access with the West Took rectified with the strate of Straters access with the Straters access with the strate of Straters access with the Straters access with the straters access access with the straters access acces



Photos 3.27 (bit) www.ipartouturaliterty beking north west overes. 100 Gryde 2.8000 Vest remine the open Gryde 2.8000 Vest) or 0 Vest is anyon should west beking test towards Bacan Wester Lower (Pource Millou 2021)



Photos 1, 2,8 (1910 Klow or bounds you twoen Bounds vice in control in size of Photos 1,800 Mittal (1910 Klow or texton Bounds vice of new states of Shorts should Well (Source Millio, 2021



Photos - Biol (HB View or Christian screek tooking source) cher Reventrumk Alternatives interface voer screek voer Screek Net Source Nittle 2021



Photos 5 & 10 Told New one tway colling sound on the soulder of Critician Occutand Changua Student from Cklowic Changuas et the student source with Student Students Changuas Students Student



Photos 116, 12 (b)D view one tway colling sourcess from the role Succ. (incr C view or salibuilitie) to deliber na south acres railwest events even munk Alternatives Fand on Nource Million 2021



Anotos 10.8. A Cold view of Chericate Great racking search and Seven hunk Alternative C Charle View Chericate Uniterval of Instance Instance In Early in the Alternative Charles and Seven hunk Alternative C Seven Instance Instanc



Photos 155. It foll device sweet Receives, besing west a 60 Grade viset flast of the other ment and the Receives the end viset of the there is the following sweet the Alternative No. 1 Source Million and the deciver stand what flast of the deciver stand what flast of the deciver stand what flast of the Million and the deciver stand what flast of the deciver stand what along beyond the Alternative No. 1 Source Million and the deciver stand what along beyond the Alternative No. 1 Source Million and the deciver stand what flast of the deci -22 ⁻



Photos 178. U 1910 Now 519 stweet Vestion in the unit of the cruit state will clobe with a CVI an CVIew 510 stweet state Westeries and Westeri



Photos 19 St. C. 1910 New bit she sten Protovenan Church at 14 politik coll coltromorth cast hom sour most control Book Succeand Millarroot 3 and Villarroot 3 and Villarroot 3. De 2021



Photos 21 S. 12 1910 Now o called that Telds beking north from Cheen shoot stene Yewer frunk. A terretive 4-3 on C Klow of 100 Octoor Steel beking west rem Cuiver 3, est selecent to Sever Trunk. A complete 4 Double 14 12 (21)



Photos 206, 04 Told Mexic termination of Gueen street style George Vicet, beking source west since Yewer trunk A termstree A tran 0 View of Creates screet, colding east when you Creates Substand VC Millistreet lesstee in the periodensia ong sector runk Atemative 5 (Source Millia, 2021)



Photos 203, 1010 Yow of Charles shoot leafthe east nem intersection of there. Succeard Millistreet store speer 1, the Alternative 3, the native 3, the first speed whet along beyon number of the Source of the 2021



Photos 278, but followed to nore?: coll colling source on nersection of 2 edge. Read tablent to under stread to ker in one Key for 20 and 20



Photos 23 Stot. (e) 5 % of the new 3, est polyna source on or you satisfy and the new 3 role and 3 were the and 30 mill 3 work to new 1 sources a store sole of the store of t



Photos 31 Stability event outers 3: both optimal sources the intersection of Ghoerler Read and Fourier Streets and Seven Funkt A teratives 1, 1, 9, 4, and 1, 1 and 2 and 5 an


Photos 31, 1989 two illusion Wastewister Perinding station slong sower Alternatives 1, 2, 3, 4 and 5 (Source, M. De 2021)

n a la companya na katalan katalan Katalan katalan katalan biran katalan Katal

Appendix E – Excerpts of the Township of Wilmet. Heritage Register (properties located within the Broader Study Arcs)

Lownship of Wilmod Heritane Register.

Economy of index ties to sted within the bacer in my Sanitary Seven breaker Study Acer

Listed

1316 Snyder's Road West

| | We start | Pat | 3 P.S., they for Ann | Sugar | ADM S OF SECOND SIDE FOR OF MARK |
|-----------------------------|--------------|-----|----------------------|-------|----------------------------------|
| | Manageria | | Torius- | | Stat Service 1 |
| Call Contraction | Dista. | | 57900 | | |
| and the second second | 040.000 F300 | | | | |
| ALC: NO. OF TAXA | 34855972525 | | | | |
| A DESCRIPTION OF THE OWNER. | | | | | |

Calculation (Calculated Free Steen All Bridges

decision in the set of the balance in the plant from the decision in the second of the balance of the balance of the balance in the second of the balance of the balance in the second of the balance of





55 Snyder's Road West



| | Albert Cover 1 | -90 | 20 Sealor's Rein West | Sealer: | PASSES OF ISOTORIES | |
|-----|----------------|-----|-----------------------|---------|---------------------|--|
| | 36.6 | | Sec. | | RIDER DOCK DITIES | |
| 100 | 242-026-02022 | | | | | |
| | | | | | | |
| | | | | | | |

A. Social and the second state of the second s

Control to board and hadren control boards prior over and replan that type does and our part of a second domain or "black and replanted over a toperate where and another toperate and acceleration operate by the effect domain of a second domain, and also the replanted over a system of door to over 10 3 years. Note: the first that the form and the domain of the second door to be to be to be and the second of the second domain of the second domain of the second domain.





8 Foundry Street



75 Charles Street & 76 Mill Street

| | Sprighter (analy) 20 NB / Bally (Peed is Sigury) | 1.0010 | (Schurke Sever geden | (and the | MARTINE MUSIC STRUCTURE CON- CONTRACTOR STRUCTURE CON- NUMBER STRUCTURE |
|-----|---|--------|-------------------------|----------|---|
| No. | Ven Opportunit presidente estat original Sets conservation (sets and conservation) sets and conservation sets and conservation presidente estation | | te Martinen Anten | | ANALY COLORED AND AN AND AN AN AN AN |

A their Brings and a deckage Walkare

Interaction a track of backage theorem interaction to the contraction of target the backage that pass depend in reaching the interaction of the product to be an order of the second and the dependence of target the backage to be a second and the second and th





53 Snyder's Road West



Us Such Participa Nation April 2022 Million State

Construction of the second second





130 Queen Street

| a farmer and | Neller (School) - Ancer | 18% | Life (3, err 30 av). Refer | 1abc | N/F 021 0412 1120 F8 04131 088 840 |
|--------------|-----------------------------|-----|-------------------------------|------|------------------------------------|
| | 0080340208 | | | | |
| | | | | | |

Coll and An Japa Value," Her high 24th Roman

Georgies 12 stores, sales have a loss have been back as been and taken, shows and a first set with 1 A and as from its few personance of Which the set failed of the store of the with endoding to the terms of the term of the version from the second of the second set of the set of the second seco





Designated

60 Snyder's Road West: Castle Kilbride National Historic Site



Characterize in the elements that contribute to the bell wiper value of the test taken include the

- there is a local with the second local recting the money. In twice and rectant with
- The stress base lands on blick and the second region
- La división insurantes.
- So the simple shock for the chirms exercise main reaction of the reacted operations and represent the leads wall on the the wind.
- Testaclene spawning the hipped isoni with spenne name.
- wrate (kallen) te etaet jatwie de tri
- Open provide with kind of tails topp theology with explosive previous previous integrated minute previous previous with the integrate key similar previous million operations and by previous previous previous to the previous million operation.
- We exercise projektes with contractors dealing and contract.
- Type stokes projecting takes can the is gade with the second pairsed by deal.
- Recipient doces de para participar y primerà avin doces
- The doces we append to the easily evolution function to easily or eadly turned public of the cost.
- divise insche zure representation eine in von the sense of the unscheduling of the unscheduling enders.
- We der seis konstruktion ihre ist nie solche end main bezigere erwäher
- The Cost SHORE Apply weakly the main evolution and an edge on the result of set set is depicted.
- There is not plenosities in party including policies electric to bill is timed, while the model of the bill of the system of the solutions of the site.
- Induksie metrolining betreuts und mutatione uted by Lubdicise en
- Exclosion model uniformation of a neukono cept cover del in the stating educing examples as Religibility in the result guilty to build energy
- Encode a los llocalego intropopor que tres dimensions oppres prese
- Reclamated over the line paint report of known by Konting and which is the being to be an edition.
- A bit bit can be a la parte es de parte parte de la faceó eterda variada la parte da parte esta parte e properte da parte e properte da parte e properte da parte esta parte
- The state percent state hits.
- emits og talfovisvinar Mil Moss.





1138 Snyder's Road West









Characterization elements that contribute to the bell age. where of the Wenfer Property include its.

- tati kingga digengaki kental
- Te de stre prio strepris par se un travella de la mérica-•
- redare), an incurran
- where the integral
- shows version involved spars the propher the equation
- to the second second second
- nache 12 scher-en
- public longs for i ding the barry he differenced, the work nonvertible reserver the weak distribution of regi periodic accelettar care the volume of each backgar 218 app.4
- differ of the build report the property.



39 Snyder's Road West



Characterization elements that contribute to the bell agevalue of the range of loss include us

- carved view stores which reserve 0 in the control view in 1922.
 Without the next boxed we walked
- neconality to the 15 little kills on
- Device the class chains real
- Fault outdoor as in carrier of some fault on the fault of the two lines and the fault of the sound of the sou
- Le character insurant de la
- Assessment (Self a Generation)
- DevenServale situation via travel
- These whether is the indiregimental based ball at the procession where
- repleted the induction backet with research and weather even length
- Delicery of period like and the well-of-
- Decision e plain il erectet i ero pionale.
- a contracted process the weld else the





44 Beck Street



Character Selicity Lenenix.

- An the second rank Gravit
- And Mon-14, 15 (real)
- se tide
- Nor 10404-104
- Andrea
- •
- Thush law;
- shine
- restand set :
- Materials englished.





Ale a construction and the Research and the Construction of Research and the Constru-

Appendix F., Curriculum Vitac



EDUCATION

2006 Masters of Arts (Planning) University of Waterloo

1998 Bachelor of Environmental Studies University of Waterloo

1998 Bachelor of Arts (Art History) University of Saskatchewan

CONTACT

540 Bingemans Centre Drive, Suite 200 Kitchener, ON N2B 3X9 T 519 576 3650 x 744 F 519 576 0121 dcurrie@mhbcplan.com www.mhbcplan.com

CURRICULUMVITAE

Dan Currie, BA, BES, MA, MCIP, RPP, CAHP

Dan Currie, a Partner with MHBC, joined MHBC Planning in 2009, after having worked in various positions in the public sector since 1997 including the Director of Policy Planning for the City of Cambridge and Senior Policy Planner for the City of Waterloo.

Dan provides a variety of planning services for public and private sector clients including a wide range of policy and development work. Dan has experience in a number of areas including strategic planning, growth plan policy, secondary plans, watershed plans, housing studies and downtown revitalization plans. Dan specializes in long range planning and has experience in growth plans, settlement area expansions and urban growth studies. He has provided expert planning evidence to the Local Planning Appeals Tribunal and heritage planning evidence to the Conservation Review Board.

Dan holds a Masters degree in Planning from the University of Waterloo, a Bachelors degree (Honours) in Planning from the University of Waterloo and a Bachelor of Arts degree from the University of Saskatchewan. He is a registered Professional Planner and a Member of the Canadian Institute of Planners and a Professional Member of the Canadian Association of Heritage Professionals.

PROFESSIONAL ASSOCIATIONS

Full Member, Canadian Institute of Planners Full Member, Ontario Professional Planners Institute Professional Member, Canadian Association of Heritage Professionals

PROFESSIONAL HISTORY

| 2013 – Present | Partner, MacNaughton Hermsen Britton Clarkson Planning Limited |
|----------------|--|
| 2009 – 2013 | Associate MacNaughton Hermsen Britton Clarkson Planning Limited |
| 2007 - 2009 | Director, Policy Planning, City of Cambridge |
| 2000 - 2007 | Senior Planner, City of Waterloo |



CURRICULUMVITAE

Dan Currie, BA, BES, MA, MCIP, RPP, CAHP

| 1999 - 2000 | Planner, City of Waterloo |
|-------------|-------------------------------------|
| 1997 - 1998 | Research Planner, City of Kitchener |

SELECTED PROJECT EXPERIENCE

MASTER PLANS, GROWTH MANAGEMENT STRATEGIES AND POLICY STUDIES

Township of West Lincoln, Smithville Northwest Quadrant Secondary Plan Township of Tiny Growth Management Strategy and Urban Expansion Analysis Niagara-on-the-Lake Mary Street Streetscape Study Richmond Hill, Bond Crescent Intensification Strategy City of Cambridge Climate Change Adaptation Policy Ministry of Infrastructure Pilot Test of Growth Plan Indicators Study Cambridge West Master Environmental Servicing Plan Township of Tiny Residential Land Use Study Township of West Lincoln Settlement Area Expansion Analysis Port Severn Settlement Area Boundary Review City of Cambridge Green Building Policy Township of West Lincoln Intensification Study & Employment Land Strategy Ministry of the Environment Review of the D-Series Land Use Guidelines Meadowlands Conservation Area Management Plan City of Cambridge Trails Master Plan City of Kawartha Lakes Growth Management Strategy City of Cambridge Growth Management Strategy Cambridge GO Train Feasibility Study City of Waterloo Height and Density Policy City of Waterloo Student Accommodation Study Uptown Waterloo Residential Market Study City of Waterloo Land Supply Study City of Kitchener Inner City Housing Study

CONTACT

540 Bingemans Centre Drive, Suite 200 Kitchener, ON N2B 3X9 T 519 576 3650 x 744 F 519 576 0121 dcurrie@mhbcplan.com www.mhbcplan.com



CURRICULUMVITAE

Dan Currie, BA, BES, MA, MCIP, RPP, CAHP

HERITAGE PLANNING

Port Credit Heritage Conservation District Town of Cobourg Heritage Master Plan Municipality of Chatham-Kent Rondeau Heritage Conservation District Plan City of Markham Victoria Square Heritage Conservation District Study City of Kingston Barriefield Heritage Conservation District Plan Burlington Heights Heritage Lands Management Plan Township of Muskoka Lakes, Bala Heritage Conservation District Study and Plan Municipality of Meaford, Downtown Meaford Heritage Conservation District Plan City of Guelph Brooklyn and College Hill Heritage Conservation District Plan Niagara Peninsula Conservation Authority St John's Master Plan City of Toronto Garden District Heritage Conservation District Study and Plan City of London Western Counties Cultural Heritage Plan City of Cambridge Heritage Master Plan City of Waterloo Mary-Allen Neighbourhood Heritage District Study City of Waterloo Rummelhardt School Heritage Designation

Other heritage consulting services including:

- Heritage Impact Assessments
- Requests for Designations
- Alterations or new developments within Heritage Conservation Districts

DEVELOPMENT PLANNING

Provide consulting services and prepare planning applications for private sector clients for:

- Draft plans of subdivision
- Consent
- Official Plan Amendment
- Zoning By-law Amendment
- Minor Variance
- Site Plan

CONTACT

540 Bingemans Centre Drive, Suite 200 Kitchener, ON N2B 3X9 T 519 576 3650 x 744 F 519 576 0121 dcurrie@mhbcplan.com www.mhbcplan.com



$1.00\times 0.00\times$

COB Medie A. Anstruktur ministra specialisms problem acce that since Conservative Collectory Conservative Collectory Collectory Collectory

<.01

i verteilenet Artsålterneus) Tellis Trikelinterustrial Archeetiker, Vefterensen frederen

CALCENTARY

u ere dis nélositori tiens di Tento generale is prisibilitati member

CURRICULU VVITAE

Vanessa Hicks www.ce *deseed*

Versional Theories Sector Hell was Planter and Associate with MITCL were exactly of which the firm a random by patient agreements was Manager of Hell operational process of the method of the week to the test of test of test of the test of test

PROFESSIONAL EXPLORACE

| en en esta da | , Cristine Frederic Operation for the separation of the |
|---------------|---|
| Francia | Machine of the constant of the school of the second problem. |
| | - construité source d'année d'Année d'Année A search Austria |
| Мау2000 | leur wege life nichteg Aside dan |
| | Arweitigt Anter dag |
| e | Archaileadh a' Bh Archaileadh a Roinneadh Aconailte All Iol. |

 HER.
 C. B. Scherk, and Scherkerter van Scherker.
 Marker V. C. HER. 1998
 Marker V. C. HER. 1998
 Marker V. C. HER. 1998
 Marker V. C. HER. 1997
 Marker V. C. HER. 1997
 Marker V. D. Scherker V.
 Marker V. B. Scherker V.



200-540 BINGEMANS CENTRE DRIVE KITCHENER / ONTARIO /N2B3X9 / T:519.576.3650 / F: 519-576-0121 / WWW.MHBCPLAN.COM

Technical Memo 3 Transportation / Municipal Services & Utilities



Project Name:Baden Trunk
Sanitary Sewer EATo:Jeff Molenhuis, Township of Wilmot
Bryan Bishop, Township of Wilmot
Trevor Heywood, GRCA

Kaoru Yajimi, ROW cc: Pierre Chauvin, MHBC Paul Britton, MHBC MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENTAL ASSESSMENT Transportation/Municipal Services and Utilities Assessment of Alternatives

3.0 Transportation/Municipal Services & Utilities

3.A Transportation Network (Roads and Pedestrians)

Roads and Intersections

This Factor Group was assessed based on the potential impact each alternative would have on local roads, regional roads and intersections during construction. It should be noted that all alternatives would impact the intersection at Snyder's Road West and Christian Street, Foundry Street and Gingerich Road, and Foundry Street at the entrance to the Baden WWPS.

<u>Alternative 5</u> received a maximum score of 10. This alternative would have the least impact on existing roads, both locally and regionally. Impacted roads include Christian Street, a small section of Charlotta Street and a small section on Gingerich Road as well as the intersections mentioned above.

<u>Alternative 4</u> received a score of 8. This alternative is similar to Alternative 5 however a larger section of Charlotta Street and Queen Street would also be impacted. In addition, construction would locally impact eight intersections.

<u>Alternative 3</u> received a score of 5. This alternative would impact local and regional roads. Impacted roads include Christian Street, Charlotta Street, Mill Street, part of Charles Street, and Foundry Street south of Charles Street. In addition, construction would locally impact eleven intersections.

<u>Alternative 2</u> received a score of 2. This alternative would greatly impact regional roads Foundry Street and Snyder's Road West. It would impact eleven (11) intersections along these roads including the major intersection at Snyder's Road West and Foundry Street.

<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 except it would also disturb an existing farm lane impacting the access to the agricultural/livestock operations.



Traffic Operations

This Factor Group reviews road safety and accessibility with respect to sight distance and turning movements. It takes into consideration the capability to accommodate forecast traffic demand and the ability to provide appropriate access to existing properties during construction.

Traffic operations similarly impeded by all alternatives include the intersection at Snyder's Road West and Christian Street and the intersection at Foundry Street and Gingerich Road which impacts traffic operation to, and from, Highway 401 located to the south of the Baden WWPS.

<u>Alternative 5</u> received a maximum score of 10. This alternative impedes traffic operations the least by avoiding the major Regional Roads where possible. The only impact is along the narrow local right-of-way where construction would likely impact the entire width of the road along Christian Street. Sightline would be temporarily impacted at three (3) intersections. This alternative would least impact road accessibility to existing residential properties.

<u>Alternative 4</u> received a score of 9. This alternative is similar to Alternative 5 except it would impact two additional local roads, Charlotta Street and Queen Street, as well as eight (8) intersections that affect sightline. Traffic control measures would be required during construction.

<u>Alternative 3</u> received a score of 4. This alternative would impact the largest number of local roads, Christian Street, Charlotta Street, Mill Street, and Charles Street, and one Regional Road, Foundry Street south of Charles Street. Sightline would be affected at eleven (11) intersections. This alignment would temporarily impede access to several residential homes along the local roads, as well as businesses and additional homes on Foundry Street. Traffic control measures would be required during construction especially when considering Foundry Street is a major access road to, and from, Highway 8. The Township's Fire Hall along Foundry Street would be directly impacted requiring alternative access arrangements during construction.

<u>Alternative 2</u> received a score of 2. This alternative would require significant closures on the heavily trafficked Snyder's Road West and Foundry Street during construction. Significant temporary traffic routing operations through residential neighbourhoods would be required during construction. Traffic operations would be greatly impeded at the busy intersection of Snyder's Road West and Foundry Street requiring traffic control measures for access to Highway 8. The Township's Fire Hall along Foundry Street would be directly impacted requiring alternative access arrangements during construction.

<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 except that it would also impact an existing farm lane which serves as a secondary access to an existing agricultural/livestock operation. The farm lane is a narrow roadway and would be required to be built to a municipal/maintenance access standard.



Transportation Policy Initiatives

This Factor Group evaluates the impact on the capability to support municipal policy initiatives regarding transit and active transportation. Public transit services affected include the Kiwanis specialized transit and the Grand River Transit line 77 bus. The Kiwanis specialized transit services individual residences therefore local roads are impacted greater. The scoring is based on the number of roads impacted and the ability for the public to reach bus stops.

<u>Alternative 5</u> received a maximum score of 10. This alternative would have the least potential obstructions for the Kiwanis transit due to the fewest roads being impacted. There are no direct impacts to public transit routes.

<u>Alternative 4</u> received a score of 8. This alternative is similar to Alternative 5 however additional local roads (Charlotta Street and Queen Street) may impact the Kiwanis transit. There are no direct impacts to public transit routes.

<u>Alternative 3</u> received a score of 5. This alternative would have a greater potential of obstructing the Kiwanis transit due to construction along many local roads (Christian Street, Charlotta Street, Mill Street and Charles Street) and Foundry Street. There are no direct impacts to public transit routes.

<u>Alternative 2</u> received a score of 2. This alternative was not considered to be a viable option due to obstructing the public bus service to and from Kitchener (Bus line 77 - intersection of Snyder's Road West & Foundry Street) as well as for the Kiwanis transit.

<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 however it may additionally impact access of Kiwanis transit along the farm road if ever required.

Railroad Crossings

This Factor Group evaluates the ease of railroad crossing construction based on depth, right-of-way and associated approvals.

<u>Alternative 3</u> received a maximum score of 10. This alignment, though deep, was deemed the easiest to construct due to sufficient access down the local road Mill Street (**Figure 6.1**).

<u>Alternative 5</u> received a score of 7. This alignment would have some challenges due to the length and angled crossing required to trench the railroad and the Baden Creek Tributary 'A' (**Figure 8.1**).

<u>Alternative 4</u> received a score of 5. Though this alternative has plenty of access along Queen Street, there appears to also be a railway turnout that leads to the mill located at 75 Charles Street increasing the required length. There is also a locally significant wetland that may be impacted during the construction of the trench due to dewatering. This alternative would require mitigation (**Figure 7.1**).

<u>Alternative 1</u> received a score of 2. This alternative is not a favourable option due to the depth of the trench and it traversing through the main intersection of Snyder's Road West and Foundry Street (**Figure 4.1**).

<u>Alternative 2</u> received the minimum score of 1. This alternative is similar to Alternative 1, but a deeper trench would be required due to the depth of the sewer (**Figure 5.1**).



3.B Municipal Services and Utilities

Municipal Services

This Factor Group evaluates opportunities for connections to the broader sanitary system. It identifies impacts to existing sanitary, water, and storm infrastructure during construction as well as identifying possible undermining of infrastructure during construction.

<u>Alternative 5</u> received a maximum score of 10. This alternative would have the least impact on existing local services. It may impact services during construction along Christian Street due to the depth of the sewer. The existing 450mm regional watermain along Christian Street and Gingerich Road that would require adequate separation from proposed services.

<u>Alternative 4</u> received a score of 8. This alternative is similar to Alternative 5 with its impacts to Christian Street, but would also impact Charlotta Street and Queen Street. The existing 450mm regional watermain along Christian Street, Charlotta Street and Gingerich Road that would require adequate separation from proposed services.

<u>Alternative 3</u> received a score of 6. This alternative is similar to Alternative 4 but would impact additional local streets (Mill Street, and Saint Charles Street as well as Foundry Street). The existing 450mm regional watermain along Christian Street, Charlotta Street, Mill Street, Charles Street, and Foundry Street would require adequate separation from proposed services.

<u>Alternative 1</u> received a score of 3. This alternative would have significant potential impacts to existing services along Regional Roads Snyder's Road West and Foundry Street. In addition, the existing 450mm regional watermain along Foundry Street would require adequate separation from proposed services.

<u>Alternative 2</u> received the minimum score of 1. This alternative is similar to Alternative 1 except it would potentially impact more municipal infrastructure due to deeper sewer alignment, leading to an increased potential of undermining.

Municipal Infrastructure (Above ground)

This Factor Group reviews the degree of impact that the alternative would have on additional aboveground municipal structures including sanitary pumping stations, acoustic noise walls, rights-of-way and their associated long-term maintenance requirements.

<u>Alternative 5</u> received the maximum score of 10. This alternative would eliminate the need for the Baden West WWPS and the Waterloo-Oxford School WWPS, which would be decommissioned under the ultimate development condition. The acoustic noise wall along Charlotta Street may be removed or remain in-tact for privacy. Rights-of-way impacted along Christian Street, Charlotta Street and Foundry Street.

<u>Alternative 4</u> received a score of 9. This alternative is similar to Alternative 5 except it would impact the Queen Street right-of-way as well as increasing the impacts to the Charlotta Street right-of-way.

<u>Alternative 3</u> received a score of 8. This alternative is also similar to Alternatives 4 and 5, however, this alignment will impact the right-of-way of additional local streets (Mill Street, Charles Street) as well as the Regional Road, Foundry Street.

<u>Alternative 2</u> received a score of 5. Only one WWPS would be decommissioned, the Waterloo-Oxford School WWPS, under the ultimate development condition. This alternative would require the noise wall to remain. The right-of-way along Regional Roads; Snyder's Road West and Foundry Street, would be fully impacted.



<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2, however an additional pumping station is required within Parcel O (**Figure 4.1**). The existing WWPS and noise wall on Charlotta Street would remain. Impacts to existing rights-of-way include Snyder's Road East as well as the private farm road leading to the additional WWPS.

Utilities

This Factor Group reviews the degree of exposure of hydro, gas and bell utilities and/or utility conflicts during construction. Scoring takes into account that all alignments will have the same degree of utility exposure and/or conflicts from the Foundry Street/Gingerich Road intersection to the Baden WWPS.

<u>Alternative 5</u> received a maximum score of 10. This alternative would have the least impact on utilities due to construction only occuring along one local road, Christian Street. Depending upon construction methods, street lights may potentially be impacted along Christian Street during construction.

<u>Alternative 4</u> received a score of 8. This alternative is similar to Alternative 5 in that utilities would be impacted along the additional local roads of Charlotta Street and Queen Street.

<u>Alternative 3</u> received a score of 5. This alternative would impact utilities along the local roads of Christian Street, Charlotta Street, Mill Street and Charles Street as well as the utilities along Foundry Street, a major Regional Road, possibly impacting transmission to additional local neighbourhoods. There would be an increased risk to underground utilities along the narrow rights-of-way of the local roads due to the depth of the sewer.

<u>Alternative 2</u> received a score of 2. This alternative would impact utilities along the major Regional Roads of Snyder's Road West and Foundry Street, possibly impacting transmission to additional local neighbourhoods. There would be an increased risk associated with construction within major rights-of-way containing more significant underground utilites.

<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 except part of the private farm road, north of Snyder's Road East, also has utilities.

Technical Memo 4 Financial



Project Name:Baden Trunk
Sanitary Sewer EATo:Jeff Molenhuis, Township of Wilmot
Bryan Bishop, Township of Wilmot

cc: Trevor Heywood, GRCA Kaoru Yajimi, ROW Pierre Chauvin, MHBC

Paul Britton, MHBC

MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENT ASSESSMENT Financial Evaluation of Alternatives

4.0 Financial

Capital Cost

This Factor Group reviews the estimated Capital Cost associated with constructing, upgrading and/or replacing a sanitary sewer. Each alternative has been itemized and includes manholes, forcemain, pipes, trenchless construction, equipment, labor costs and pumping station as required.

Trenchless construction costs vary and are dependent upon the complexity of the installation. The complexity surrounding trenchless construction are discussed in the Technical section. Please refer to the attached Cost Estimate Table in **Appendix A**.

Quoted costs are preliminary cost estimates and are subject to change with final design.

<u>Alternative 5</u> received the maximum score of 10. This alternative would require approximately 1900m of sewer pipe and manholes, 2 – 30m trenchless stream crossings and a 40m trenchless railway crossing.

Capital Cost = \$3.63M

<u>Alternative 4</u> received a score of 9. This alternative requires approximately 2005m of sewer pipe with manholes, 2 – 30m trenchless stream crossings and a 95m trenchless railway crossing.

Capital Cost = \$4.42M

<u>Alternative 3</u> received the minimum score of 8. This alternative requires approximately 1980m of sewer pipe with manholes and 1 – 30m trenchless railway crossing. Greater cost is due to sewer depth.

Capital Cost = \$4.56M

<u>Alternative 2</u> received a score of 7. This alternative requires approximately 1800m of sewer pipe with manholes and 1 - 60m trenchless railway crossing. Greater cost is due to sewer depth and complexity of installation.

Capital Cost = \$5.07M

<u>Alternative 1</u> received a score of 1. This alternative requires a sanitary pump station, approximately 1560m of forcmain pipe, air release valve chambers, approximately 1800m of sewer pipe and manholes, a 60m trenchless railway crossing and a 20m trenchless stream crossing for the forcemain. Greater cost due to additional forcemain and pumping station as well as the complexity of installation.

Capital Cost = \$7.09M

Engineers, Scientists, Surveyors.



Operational Cost – Sanitary Sewer

This Factor Group qualifies the proposed long-term operations and maintenance requirements for each alignment and is scored accordingly. Costs are associated with maintenance of pumping stations, sewers, easements, rights-of-way and traffic control as a result of maintenance operations.

<u>Alternative 5</u> received a maximum score of 10. This alignment would eliminate two existing pumping stations, Baden West WWPS and School WWPS, requiring only one pumping station, Baden WWPS, to service all of Baden. The maintenance access adjacent to the Baden Creek (within the Schmidt Lands), will require maintenance but is less costly than maintenance within the road network.

<u>Alternative 4</u> received a score of 8. This alternative is the similar to Alternative 5, however there are additional local road rights-of-way to maintain. Costs associated with traffic control may be greater as a result of the additional roads.

<u>Alternative 3</u> received a score of 6. This alternative eliminates two existing pumping stations. The alignment would be entirely within the local and regional road rights-of-way, increasing maintenance costs and traffic control as a result of maintenance operations.

<u>Alternative 2</u> received a score of 5. This alternative would eliminate one pumping station (School Block WWPS). Maintenance costs associated with rights-of-way would be greater along the busy regional roads. Extensive traffic control will be required along Snyder's Road West and Foundry Street during maintenance operations.

<u>Alternative 1</u> received a score of 1. This alternative would significantly increase associated operational costs due to an additional pumping station and associated long forcemain. Additional maintenance costs would be required for forcemain easement. Extensive traffic control will be required along Snyder's Road West and Foundry Street during maintenance operations. Special coordination will be required along the farm road north of Snyder's Road East.



Cost Estimate



| BADEN SANITARY TRUNK SEWER PRELIMINARY COST ESTIMATE | | | | | | | |
|--|---|---------------------|---|--|---------------------------|--|--|
| Alternatives | A1 | A2 | A3 | A4 | A5 | | |
| | Modified 2014 Environmental Assessment Preferred Alternative (Do Nothing) | Snyder's/Foundry | Christian/Charlotta/Mill Charles/Foundry | Christian/Charlotta/Queen and Baden Creek | Christian and Baden Creek | | |
| TOTAL LENGTH (m) | 3355 | 1800 | 1985 | 2095 | 1905 | | |
| Capital Cost | \$ 5,908,859 | \$ 4,222,945 | \$ 3,802,635 | \$ 3,684,232 | \$ 3,025,695 | | |
| Total Construction | \$ 5,908,859 | \$ 4,222,945 | \$ 3,802,635 | \$ 3,684,232 | \$ 3,025,695 | | |
| 20% Contingency | \$ 1,181,772 | \$ 844,589 | \$ 760,527 | \$ 736,846 | \$ 605,139 | | |
| Total Capital Cost | \$ 7,090,631 | \$ 5,067,534 | \$ 4,563,162 | \$ 4,421,078 | \$ 3,630,834 | | |
| TOTAL ESTIMATED COST | \$7,090,630.80 | \$5,067,534.00 | \$4,563,162.00 | \$4,421,078.40 | \$3,630,834.00 | | |
| Operational Cost of Pump Station and Maintenance (WWPS: \$30,000-\$50,000/year) | \$150,000-\$250,000 | \$120,000-\$200,000 | \$50,000.00 | \$50,000.00 | \$50,000.00 | | |

Notes:

1. Cost estimate does not include HST (%13)

2. Estimate does not include environmental mitigation

3. Estimate does not include soft costs (Engineering, Legal, Planning, etc.)

4. Estimate does not include cost of modifying to increase capacity or to replacing pumping station

5. Estimate does not include cost of dewatering

Technical Memo 5 Technical



 Project Name: Baden Trunk Sanitary Sewer EA
 To: Jeff Molenhuis, Township of Wilmot Bryan Bishop, Township of Wilmot Trevor Heywood, GRCA
 Kaoru Yajimi, ROW Pierre Chauvin, MHBC

MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENTAL ASSESSMENT Technical Assessment of Alternatives

5.0 Technical

Paul Britton, MHBC

Constructability

This Factor Group evaluates the alternative alignments according to physical constructability (adequate access to construction area); potential conflicts with existing infrastructure other than utilities; depth of sewer, trenchless crossings, and water table and groundwater dewatering (short term impact). Generally, less interference with existing roads, from the constructability point of view, is considered a benefit.

<u>Alternative 5</u> received a maximum score of 10. A large portion (~60%) of the alignment is within agricultural lands with minimal depths where possible. However, the upstream portion of this alignment, along Christian Street, ranges from 8.3m to 9.3m deep (**Figure 8.1**). This road contains a local sewer and a 450mm regional watermain. The complexity lies within the installation of the deep sanitary trunk while not undermining the watermain and maintaining adequate separation within the narrow right-of-way as well as reconstructing the local sewer. Two trenchless crossings are required, one within the railway and Baden Creek Tributary A, and one at Gingerich Road. Groundwater interaction at these locations is considered negligible.

<u>Alternative 4</u> received a score of 9. This alternative is similar to Alternative 5, but scored lower because there are more local roads along this alignment (Charlotta Street and Queen Street) increasing the complexity of the installation as described in Alternative 5. This alignment has close interactions with locally significant wetlands (**Figure 7.1**) requiring more attention to reduce impacts. Dewatering and additional erosion control measures may be required. The upstream trenchless stream crossing of the Baden Creek Tributary A is close to Baden Creek, adding more complexity by increasing the potential for dewatering at this location.

<u>Alternative 3</u> received a score of 6. This alternative is located within the local rights-of-way. The depth of sewer combined with the narrow width of the rights-of-way increases the complexity of the alignment within these streets (**Figure 6.1**). Two open cut crossings are required along Foundry Street at the Baden Creek pond and Huehn Award Drain crossings. Dewatering and flow rerouting is required during the construction of these crossings. The trenchless railway crossing at Mill Street is considered the most feasible of the alternatives.



<u>Alternative 2</u> received a score of 2. This alternative scored lower than Alternative 3 because of the complexity of constructing a deep sanitary trunk sewer along Snyder's Road West and the trenchless railway crossing at the critical intersection of Snyder's Road West and Foundry Street (**Figure 5.1**).

<u>Alternative 1</u> received a score of 1. This alternative scored lower than Alternative 2 because of the additional 20m stream crossing north of Snyder's Road West. This alignment is just downstream of two (2) local wetlands (**Figure 4.1**), where potential dewatering and additional erosion control measures will be required.

Trenchless (Constructability and Complexity)

This Factor Group assesses the complex issues regarding the installation of trenchless sewer pipes under the railways and at the stream crossings. Installation of trenchless connections become more complicated when dealing with narrow right-of-way, deep sewers, various connections and busy intersections. Sending and receiving pit locations are dependent upon soil structure and will require confirmation by a geotechnical engineer.

<u>Alternative 3</u> received the maximum score of 10. This alternative would require approximately 40m of trenchless connection for the railway crossing along Mill Street. The sewer depth at this crossing increases its complexity, however, Mill Street provides the most direct installation with ample clearance for sending and receiving pits. No trenchless stream crossings required.

<u>Alternative 5</u> received a score of 9. This alternative would require approximately 60m of trenchless railway crossing and 1-40m and 1-30m trenchless stream crossings. Challenges for this alignment include back-to-back sending and receiving pits for the railway crossing and the crossing along Tributary A. No dewatering would be required at this crossing. The trenchless stream crossing north of Gingerich Road, does not present any foreseen issues.

<u>Alternative 4</u> received a score of 8. This alternative would require approximately 95m of trenchless railway crossing and 2-30m trenchless stream crossing. The trenchless crossing is longer than Alternative 5 because of the existing spur line. A local wetland, located between the railway and the spur line, lies directly above the proposed alignment presenting potential challenges. A second wetland also runs along this alignment and would be situated near the sending pit. The trenchless stream crossing, north of Gingerich Road, does not present any foreseen issues.

<u>Alternative 2</u> received a score of 3. This alternative would require approximately 60m of trenchless railway crossing at the intersection of Snyder's Road West and Foundry Street. The railway right-of-way is narrow through the intersection requiring sending and receiving pits to be located outside of the intersection. The depth (>6m) of this trenchless connection poses additional complications through the intersection as well as the connection with the deep trunk sewer along Snyder's Road West and the local sewer at Livingston Boulevard.

<u>Alternative 1</u> received the minimum score of 1. This alternative is similar to Alternative 2 except there would be an additional 20m trenchless crossing north of Snyder's Road East for the forcemain. This crossing is within a regulated floodplain and under a permanent watercourse believed to have groundwater inflow. This crossing is also just downstream of a locally significant wetland.



Property (Land/Acquisition)

This Factor Group reviewed property requirements for the Baden Trunk alternatives and was assessed based on the number and area of properties and ease of acquisition.

<u>Alternative 2</u> received a maximum score of 10. This alternative would be located fully within existing rights-of-way therefore no additional land acquisitions would be required.

<u>Alternative 3</u> received a score of 9. This alternative would be located fully within existing rights-of-way therefore no additional land acquisitions would be required.

<u>Alternative 4</u> received a score of 8. This alternative would require a portion of the right-of-way to be acquired. An understanding has been reached with the property owner that arrangements will be made regarding securing the right-of-way through the open areas.

<u>Alternative 5</u> received a score of 7. This alternative is similar to Alternative 4 except a larger portion of open land would need to be secured.

<u>Alternative 1</u> received a score of 1. Construction of the new pumping station would require approximately one residential lot (~0.16acre). It is anticipated arrangements can be made with the property owner to secure the required area for the pumping station. However, securing the right of way along the private farm lane for the forcemain is uncertain.

Serviceability (Long-term)

This Factor Group evaluates the percentage of Baden to be serviced by the Baden Trunk in support of future growth. Percentages were determined by serviceable drainage area for each alternative.

<u>Alternative 5</u> received a maximum score of 10. This alignment would be able to service all of Baden's future growth areas, including Foxboro. Refer to **Figure 8.2** for contributing sanitary drainage areas.

<u>Alternative 4</u> received a score of 9. This alignment is similar to Alternative 5, refer to **Figure 7.2** for contributing sanitary drainage areas.

<u>Alternative 3</u> received a score of 7. This alignment scored lower than Alternative 4 because it would service approximately 90% of Baden's future growth areas. The remaining area would be serviced through New Hamburg. Refer to **Figure 6.2** for contributing sanitary drainage areas.

<u>Alternative 2</u> received a score of 5. This alignment scored lower than Alternative 3 because it would service approximately 80% of Baden's future growth areas. The remaining area would be serviced through New Hamburg. Refer to **Figure 5.2** for contributing sanitary drainage areas.

<u>Alternative 1</u> received a score of 1. This alignment scored lower than Alternative 2 because it would service approximately 75% of Baden's future growth areas. The remaining area would be serviced through New Hamburg. Refer to **Figure 8.2** for contributing sanitary drainage areas.

M:\48730\100\Reports\MTE Reports\Tech Memos\Tech Memo 5.0 - Text_Figures_Appendices\48730-100 5.0 Technical Baden Trunk Tech Memo.doc

Technical Memo 6 Selection of the Preferred Alternative & Evaluation Matrix



| Drai | Baden Trunk |
|------|------------------------------------|
| FIOJ | Sanitary Sewer EA |
| та | Jeff Molenhuis, Township of Wilmot |
| 10: | Bryan Bishop, Township of Wilmot |
| | Trevor Heywood, GRCA |
| ~~. | Kaoru Yajimi, ROW |

cc: Pierre Chauvin, MHBC Paul Britton, MHBC MTE File No.: 48730-100

Date: November 6, 2022

From: Jeff Martens, P.Eng. Carla Illman, M.Eng.

Re: BADEN TRUNK SANITARY SEWER CLASS B ENVIRONMENTAL ASSESSMENT Selection of the Preferred Alternative

6.0 Selection of the Preferred Alternative

The preferred sanitary sewer trunk alignment was selected by following a Schedule B framework outlined in the Municipal Engineers Class Environmental Assessment manual. Based on the preceding analysis and the outcome of the evaluation matrix (**Appendix A**), the preferred alternative is **Alternative 5 with a total score of 47**. The sanitary sewer construction of the preferred alternative has the following benefits:

- Least impacts existing businesses and residents
- Least impacts to cultural heritage
- Provides timely project completion due to minimal regional road corridors, coordination
- · Provides flexibility for servicing the entire primary study area
- Eliminates two (2) existing pumping stations
- Eliminates noise and odour associated with the removal of existing pumping stations
- Reduces long-term operational/maintenance cost
- Minimal impact to road network
- Minimal impact to traffic
- Minimizes impacts to significant water courses (Baden Creek at the Baden Creek pond and the Huehn Award Drain)
- Avoids wetlands, woodlands, Butternuts
- Does not impact flow regime of any water feature
- Eliminates impacts to the coldwater fisheries tributary (Huehn Award Drain)
- Represents the least cost solution

Other design alternatives ranked as follows:

 Alternative 4 with a total score of 38 – Viable alternative, stays within Township boundaries providing timely project completion, minimal impacts to existing businesses, residents and cultural heritage sites, reduces long-term operational costs, eliminates 2 existing pumping stations and associated noise and odour, minimizes impacts to significant water courses, eliminates impacts to the cold water fisheries tributary and provides flexibility for servicing the entire primary study area

(18.4% behind the first ranked alternative);



- Alternative 3 with a total score of 34 Eliminate 2 existing pumping stations and associated noise and odour, reduces long-term operational/maintenance cost, avoids wetlands, woodlands and butternut trees, has a secure right-of-way. (28.0% behind the first ranked alternative);
- **Alternative 2 with a total score of 23** This alternative avoids wetlands, woodlands, butternuts and has a secure right-of-way

(49.7% behind the first ranked alternative);

• **Alternative 1 with a total score of 9** – It is the current approved alignment. (80.2% behind the first ranked alternative).

M:\48730\100\Reports\MTE Reports\Tech Memos\Tech Memo 6.0 - Text_Figures_Appendices\48730-100 6.0 Preferred Alternative Baden Trunk Tech Memo.doc


Evaluation Matrix



| PRELIMINARY EVALUATION CRITERIA - BADEN SANITARY TRUNK Alternative | | | | | | | | | | | | |
|--|---|-------------------|--|-------------------|---|-------------------|---|---|-------------------|--|--|--|
| Factor Category / Factor Group / Factor | | | A1 | | A2 | | Alternative A3 | A4 | | A5 | | |
| Date: August 2022 | Description | Modified 2014 | Environmental Assessment Preferred Alternative (Do Nothing) | | Snyder's/Foundry | CI | vistian/Charlotta/MII/Charles/Foundry | Christian/Charlotta/Queen and Baden Creek | | Christian and Baden Creek | | |
| | | Score | Comments | Score | Comments | Score | Comments | Score Comments | Score | Comments | | |
| 1. NATURAL ENVIRONMENT 1.A Terrestrial Ecosystems | | | | | | | | | | | | |
| Vasiation Communities and Elona | Impact on feature and function of vegetation communities and flora | | Impact along Foundry St, near pond and area N. of Snyder's Rd West. Consruction of segments may | 10 | Negligible impacts to natural features along Foundry St. | | Similar to Option A2, except more potential impacts along Charlotta St and Foundry St (CUM1) during | Largest extents of construction within natural areas. Temporary disturbances to CUM1 meadow. Construction is in proximity to FOD7-3, potential | 5 | Traverses near less meadow area than A5. | | |
| Vojement Contrante and Frite | -sensitivity/ susceptibility (e.g., mortality, stress, composition change). | - | impact CUW1, CUM1. Indirect potential impacts to CUT1 and areas near SUM1 unless properly miligated. | 10 | and Snyder's Road W. | 0 | construction to regionally significant Running Serviceberry if not properly mitigated. | edge tree removal and tree root disturbance, regionally significant Running Serviceberry disturbance. | | depending on surveyed alignment. | | |
| | Impact on feature and function of wildlife habitat -significance (e.g., size, presence of specialized habitat elements, species of | | Anticipated but unknown impacts to wildlife north of Snyder's Rd W within wetlands which were not part of the NEA (impacts therefore considered potentially | | | | Similar impacts to Baden Creek habitats as A1 and A2. | Largest extent of construction within natural areas. Impacts to amphibian breeding habitat (MAM2-2). | | Aligment runs along outer boundary of wildlife corridor. No impacts to wetland habitats. Less | | |
| Wildlife Habitat | conservation concern) sensitivity (e.g., noise, light) Effects on wildlife movement corridors (e.g., habitat removal, fragmentation, potential increases in animal-vehicle conflicts). | 1 | significant to wildlife). Potential tree removals in CUW1 required which may house SAR bats. Potential impact to Turtle Wintering Areas in Baden Creek Pond habitat due to dewatering. | 6 | Similar impacts to Baden Creek habitats as A1 and A3. | 5 | A3 scored slightly lower than A2 due to construction dewer depth and extent of work. | 4 Closest impact to forested area, potential disturbance to SAR bat habitats (FOD7-3). Temporary impacts to habitats within CUM 1. | 10 | impacts to CUM1 habitats than A4. Potential tree removals in CUW1 required which may house SAR bats. | | |
| 1.B Wetland Ecosystems | Normalized Factor Group Score | 2.5 | | 8.0 | | 6.5 | | 2.5 | 7.5 | | | |
| | Impact on feature and function of wellands (e.g. area, classification and quality | | Construction of sever north of Snyder's Road West & Foundry Street may directly impact Locally Significant Wetland feature. Potential indirect Impacts to adjacent | | No impact to wetlands. Alignment is in slight proximity | | No innact to wetlands. Alignment is in slight provimity | Construction directly impacts wetlands MAM2 & MAM2-2 due to envirolity during construction. May | | | | |
| Wetlands | relative extent, function in broader wetland complex ecosystem). | 1 | meadow marsh unless miligated. Construction may disrupt local drainage regime and hydrological conditions that support the wetland. | 8 | to Locally Significant Wetland north of Snyder's Road West & Foundry Street. | 9 | to MAM2. | 4 impact surface drainage catchments and impair water quality unless properly mitigated. | 10 | Does not impact wellands. | | |
| 1.C Aquatic Ecosystems | Normalized Factor Group Score | 1.0 | | 8.0 | | 9.0 | | 4.0 | 10.0 | | | |
| | -Number of water bodies and watercourse crossings. -Sensitivity of fishtish habitat and thermal regime (warm. cool or cold water). | | 3 watercourse crossings; a stream crossing north of Snyder's Road West to Tributary B, a stream crossing | | | | | 2 watercourse crossings: a stream crossing at Gingerich Road, a stream crossing at Baden Creek Tributary A upsteam of the confluence. Tributary A | | 2 watercourse crossings: a stream crossing at Gingerich Road, a stream crossing at Baden Creek | | |
| Aquatic Habitat | -Extent (area) and function of riparian habitat removed. -Extent and type of fish habitat (in-stream) altered/displaced in watercourse, including importance to aquatic ecosystem (e.g. spawning, nursery areas, habitat for species of conservation concern or less tolerant sparse). | 1 | Hushn Award Drain. All watercourses provide aquatic habitat, subject to federal Fisheries Act. Crossings require fish capture and relocation during construction, | 4 | 2 watercourse crossings; a stream crossing to Baden Creek Pond, a cold water stream crossing at Huehn Award Drain. | 3 | 2 watercourse crossings; a stream crossing to Baden Creek Pond, a cold water stream crossing at Huehn Award Drain. Al is a slightly deeper crossing than A2. | identified as intermittent flow with no groundwater interaction, may be accessible to fish. Potential to construct through tributary during period of no flows. Trenchiese instalation to be utilized for both | 10 | Identified as intermittent flow with no groundwater interaction, may be accessible to fish. Potential to construct through tributary during period of no flows. | | |
| | тот аресона от солзна такон солостт се выя сонотат аресона). | | watercourses need isolation and de-watering, water quality treatment, etc | | | | | crossings. Tributany A crossing is closer to Baden Creek than A5. | | Trenchless installation to be utilized for both crossings. | | |
| 1.D Surface Water (Quality - Impact to Water Quality / T | Normalized Factor Group Score | 1.0 | | 4.0 | Two ones cut stream culved crossings at the Bades | 3.0 | | 7.0 | 10.0 | Proposed trenchiess crossion at Barlen Creek | | |
| Water Quality | Impact on water quality of surface water features (TSS, gasloil fueling, construction operations). | 1 | Similar to A2 with additional upstream impacts at stream crossing north of Snyder's Road West, increased risk of surface water impacts adjacent to Locally Significant Wetland. All utimately can increase | з | Creek Pond Tributary B and Huehn Award Drain. Flow rerouting and dewatering required to prevent increased flows. Increased potential for contaminated sediments from the Deviced Pondersula Contaminated Sediments | 2 | Similar to A2 with deeper sanitary sewer depths, further potential impacts during construction. | Similar to A5, but trenchless crossing is closer to Baden Creek. Increased potential impacts to water quality. | 10 | Tributary A is an intermittent flow tributary with no evidence of groundwater interaction, no dewatering required during construction. No anticipated impacts | | |
| | | | TSS levels in Baden Creek unless properly mitigated. | | Pond and Baden Creek. | | | | ├── | Gingerich. | | |
| Thermal Regime | Impact in the thermal regime of surface water features. | 1 | Similar to A2 with additional upstream impacts at stream crossing north of Snyder's Road West. Additional dewatering and construction can ultimately can affect thermal remine in Barden Creek Prori and | з | Close proximity to HAU which provides coldwater in support of cool water habitat in Baden Creek Pond. Pond has GW connectivity as weld, where dewatering during construction can affect thermal regime. Potential | 2 | Similar to A2 with deeper sanitary sewer depths, further potential impacts during construction. | Similar to A5, but trenchless crossing is closer to Baden Creek. Increased potential impacts to thermal review | 10 | No impact on Baden Creek Pond and Huehn Award Drain. Trenchiess crossing on intermittent tributary, no municipation impact to thermal project | | |
| Chlorides | Impact on water quality of surface water features. | 10 | HAD. | 10 | Impact to vegetation that regulates temperature in Baden Creek | 10 | Not Significant | 10 Not Significant | 10 | Not Significant | | |
| 1.E Groundwater Resources | Normalized Factor Group Score | 4.0 | | 5.3 | | 4.7 | | 8.7 | 10.0 | | | |
| Regional Recharge Areas (Groundwater) | Encroachment into the Regional Recharge Area. (removalidisruption of function – area; depth). | 1 | Similar to A2 and A3 with respect to HAD, with additional proximity to recharge areas near pump station within Parcel O. | 7 | Close proximity to HAD, which is connected to the Regional Recharge Area, depth of trunk may affect GW | 6 | Similar to A2, but sewer is deeper under the HAD. | 9 Similar to A5, alignment is slightly closer to regional recharge area. | 10 | Alignment is furthest away from recharge areas. BH sample shows no sand layer as that of the regional recharge area, will not draw groundwater | | |
| Groundwater Quality - Clean Water Act (WHPA, SGRA, IPZ, HVA) | Potential for impacts to vulnerable areas (Area) as defined in Source Water Protection Areas (SWPA). | 1 | Closest proximity to WHPA north of Parcel O | 5 | Similar to A3 | 6 | WHPA location to the east | 9 Similar to A5, alignment is slightly closer to WHPAs and SGRAs. | 10 | Furthest alignment from WHPA location to east, does not affect it | | |
| Shallow Groundwater Movement/Contributions (Quality and Quantity) | Potential for interference with existing flow patterns (baseflow) relative to proximity to significant surface water feature and significant groundwater feature discharge areas. Consider Short-Term Impact Regional shallow gw elevation 346m - 353m | 1 | Shallow forcemain from pump station and stream crossing N of Snyder's Road West, may affect Baden Creek Pond and HAD. Largest extent of construction and potential impacts to groundwater. | 5 | May affect Baden Creek Pond and HAD during construction. | 4 | Proximity to wetland MAM2, may affect Baden Breek Pond and HAD during construction due to depth of sewer. | 2 Close proximity to wetland MAM2 and MAM2-2, tributary crossing at fish habitat location. | 10 | No evidence of GW interaction at Baden Creek tributary A stream crossing, no sand layer in stream crossing at Gingerich Road. GW impacts minimalineologible. | | |
| 1.F Surface Drainage (Quantity - Erosion) | Normalized Factor Group Score | 1.0 | | 5.7 | | 5.3 | | 6.7 | 10.0 | | | |
| Quantity (flows) | Quantity Control and Flow Regime | 7 | Impacts flows at Baden Creek Tributary B, Baden Creek Pond and Huehn Award Drain crossings. Tributary B is trenchless, others are open cut which requires stream | 6 | Impacts flows at Baden Creek Pond and Huehn Award Drain crossings. Crossings are open cut and requires | 1 | Impacts flows at Baden Creek Pond and Huehn Award Drain crossings. Crossings are open cut and requires stream diversion, mitigation required for sediment | Trenchless railway crossing occuring through a sand layer near local wetlands and section between trenchless crossing and Queen Street may result in the section of the | 10 | No change to flow regime due to tenchless crossings at Baden Creek Tributary A and Gingerich | | |
| · · · · · | ·· · · · · | | diversion. Mitigation required for sediment control and flow rate. | | stream diversion. Mitigation required for sediment control and flow rate. Deeper trench than A1 | | control and flow rate. Significantly deeper trench than A2 | potential deviated in and rerouning to Bacen Creek. No anticipated change to flow regime due to tenchiess crossing north of Gingerich Road. | | Road. No dewatering or flow rerouting anticipated. | | |
| Floodplain Function | Impacts to Regulatory Floodplain and changes to floodplain hydrologic function. (During Construction), including need for GRCA work permit | 7 | Minimal impact to floodplain due to (2) stream crossings occuring along road right-of-way. Imapct to floodplain at stream crossing north of Snyder's Road East. | 10 | Minimal impact to floodplain due to (2) stream crossings occuring along road right-of-way | 9 | Minimal impact to floodplain due to (2) stream crossings occuring along road right-of-way | Alignment along floodplain boundary & impacts 1 GRCA floodplain boundary, closest to floodplain near forest, impact at 2 stream crossing | 6 | Alignment along floodplain boundary & impacts GRCA floodplain boundary, impact at 2 stream crossing | | |
| Stormwater Management Water Balance | Opportunities to enhance roadway stormwater management measures, including coordination with/use of adjacent development facilities. Chances to water balance due to development | 10 | Not Significant Not Significant | 10 10 | Not Significant Not Significant | 10 10 | Not Significant Not Significant | 10 Not Significant 10 Not Significant | 10 10 | Not Significant Not Significant | | |
| Erosion Assessment | Impact on Slope stability due to stream disturbances and from increased TSS during construction. | 1 | Two open cut stream crossings along Foundry Street at the Baden Creek pond and Huehn Award Drain, additional upstream trenchiess crossing at Baden Creek Tributary B. milination required to research correspond | 4 | Two open cut stream crossings along Foundry Street at the Baden Creek pond and Huehn Award Drain, mitigation required to prevent increased TSS loading | 3 | Similar to A2, but sewer is deeper under the HAD, furher erosion potential | 9 Similar to A5, trenchless crossing closer to Baden Creek than A5. | 10 | Two trenchless crossings. No dewatering anticipated, so less erosion concerns are present. Erosion setbacks present, construction to take | | |
| | Normalized Factor Group Score TOTAL CATEGORY SCORE | 7.0 | TSS loading and erosion prevention | 8.0 6.5 | and erosion prevention | 6.6 5.9 | | 7.6 6.1 | 92 9.5 | place outside buffer where possible. | | |
| 2 SOCIO-CIII TURAL ENVIRONMENT | % BEHIND FIRST RANKED ALIGNMENT CATEGORY BANKING | 71% 5 | | 31% 2 | | 38% 4 | | 36% | 0% 1 | | | |
| 2.A Land Use Policy | | | Least compatible due to timing and prioritization of | | | | | | | | | |
| Land Use Designations and Policies | Aligned with Provincial, Regional and Township growth and development goals, objectives or policies (high, medium, low). Upper fer and lower tier infrastructure and comidor owner. | 1 | regional projects (2 separate regional projects, 1 local project). In addition, temporary disturbance to access for local agricultural operation north of Snyder's Road. Deer port foll, integrate Tournshipt claim to loss Raden | 2 | Similar to A3, except 2 Regional projects required to facilitate this alignment. Lower scoring due to complexity of Regional projects. | 4 | Low degree of compatibility due to timing and prioritization of regional projects combined with local projects (1 regional project, 1 local project), does not fully integrate Township's plan to join Baden and New | Autonomy to implement infrastructure improvement 10 without upper tier regulations formulating development and growth independently | 8 | Similar to A4, longer construction time within Schmidt Lands (Parcoel AA) and disruption to agricultural lands, impact to growing season | | |
| | Embersy of resaming development patients and timing. | | and New Hamburg servicability. | | | | Hamburg servicability. | | | | | |
| Impact of the Sanitary Sewer on Adjacent Lands | Impacts to residence and businesses | 1 | Impacts adjacent residential properties and busissess along major regional roads Foundry Street and Snyder's Road West. | 2 | Impacts adjacent residential properties and busissess along major regional roads Foundry Street and Snyder's Road West. | 4 | Christian Street, Charlos Street, Mil Street, Charles Street. Impacts businesses and residents along Foundry Street Regional Road. | More residential properties impacted than A5 along 8 Christian Street and Charlotta Street. Coordination required with Schmidt Lands. | 10 | Least amount of properties impacted primarily along Christian Street. Coordination required with Schmidt Lands. | | |
| 2.B Archaeological Resources | Normalized Factor Group Score | 1.0 | | 2.0 | | 4.0 | | 9.0 Alignment impacts the 50m monitoring buffer at | 9.0 | | | |
| Known and/or Potential Archaeological Sites | Numbertypesignticance of directindirect impacts to known archaeological resources and/or areas of high archaeological potential. Normalized Factor Group Score | 8 8.0 | Furthest and shakowest from found artifact, Baden Creek is a barrier to artifact location | 10 10.0 | Furthest from found artifact but not as snakow, Baden Creek is a barrier to artifact location | 9.0 | Furthest and deepest from found artifact, Baden Creek is a barrier to artifact location | Findspot-16 Findspot-18 and construction may impact AiHe-469 50m monitoring buffer. | 5 5.0 | Augment impacts the monitoring sum butter at Findspot-16 Findspot-18 | | |
| 2.C Built Heritage Built Heritage Features | Number/type/significance of direct/indirect impacts to built heritage resources (based on presence of built heritage resources identified by the Region of Minimized) | 2 | Construction will affect scenic routes along Snyder's Road West and Foundry Street; Adjacent to 4 listed and 2 designed respective listed on OFA | 1 | Similar to A1, but road closures anticipated to be longer due to depth of sewers | 5 | Construction will affect scenic route along Foundry Street; Adjacent to 4 listed and 2 designated properties listed on OHA | 9 Similar to A5, alignment is closer in proximity to 130 Queen Street | 10 | No direct impacts to heritage properties or OHA, alignment is near one listed property: 130 Queen | | |
| 2.D Noise | Normalized Factor Group Score | 2.0 | 2 designated properties insted on OHA | 1.0 | No. additional course and an and paramiter of Darker Wind. | 5.0 | | 9.0 | 10.0 | Sales | | |
| Noise | Noise impacts due to addition/removal of wastewater pumping stations (wwps) Normalized Factor Group Score TOTAL CATEGORY SCORE | 1 1.0 3.0 | Baden West WWPS | 2 2.0 3.8 | WWPS | 8 8.0 6.5 | Removal of noise wall and Baden West WWPS | Removal of noise wall and Baden West WWPS 9.0 7.0 | 10 10.0 8.5 | Removal of noise wall and Baden West WWPS | | |
| 3. TRANSPORTATION/MUNICIPAL SERVICES AND UT | % BENIND FIRST RANKED ALIGNMENT CATEGORY RANKING | 5 | | 4 | | 3 | | 2 | 1 | | | |
| 3.A Transportation Network (Roads and Pedestrians) | | | | | 2 regional marks 11 mark intersection major | | Christian Street, Charlotta Street, Mil Street, Carles | Christian Street, Jamer portion of Charlotta Iban A5 | | Christian Street portion of Charlotta portion of | | |
| Roads and Intersections | Potential impacts to roads/intersections. | 1 | Similar to A2, additional impact to private farm road access north of Snyder's Road East | 2 | Intersection impact of Snyder's Road West and Foundry Street | 5 | Street, large portion of Foundry Street (Regional) / 11 intersections | 8 Queen Street, portion of Gingerich Road, portion of Foundry Street (Regional) / 8 intersections | 10 | Gingerich Road, portion of Foundry Street (Regional) / 3 intersections | | |
| | Road safety and accessibility (sight distance; turning movements). | | Circles to AD and the set language to form have a set of | | Impedes traffic along two Regional Roads Foundry | | Impedes traffic along the most local roads (4: Christian Street, Charlotta Street, Mill Street, Charles Street) and a portion of Regional Road Foundry Street. 11 | Impedes slightly more traffic than A5 along | | Impedes the least amount of traffic throughout construction. Narrow local right of ways, | | |
| Traffic Operations | Provides for appropriate access to existing properties. Impact on access to existing properties. | 1 | Similar to Az, additional impacts to farm and access at Snyder's Road/Foundry intersection. | 2 | traffic operations routing through local roads required during construction. | 4 | Intersections will have line of sight affected during construction. Additional arrangements required for Township's Fire Hall at Foundry Street during construction. | 9 Charlotal Street and Queen Street. Impedes access to more homes than A5. 8 intersections will have sightline affected. | 10 | construction will impact entirety of roadway. 3 intersections will have sightline affected. Least amount of homes have access affected. | | |
| | | | Transit obstruction for bus service to and from | | Transit obstruction for bus service to and from | | Minimal impact to local transit, will affect access to | Similar to A5 with addition impacts to Charlotta | | Least potential transit obstructions (Kiwanis transit) only affects Christian Street and small portion of | | |
| Transportation policy initiatives | Capability to support municipal policy initiatives (transit, active transportation, roundabouts). | 1 | Kitchener (bus line 77-intersection of Snyder's Rd W& Foundry St), additional obstruction of access along Farm Road | 2 | Kitchener (bus line 77-intersection of Snyder's Rd W & Foundry St), | 5 | several residential properties in neighborhoods and along Foundry Street. No direct impact to trnasit routes. | 8 Street and Queen Street. No direct impact to transit routes. | 10 | Charlotta Street, will affect access to residential properties in neighborhoods. No direct impact to trnasit routes. | | |
| Railmad Crossins | Even of mirrord crossing, approval | 2 | Significant trench required through main intersection of | 1 | Similar to A1 but deeper tranch required | 10 | Easiest trench crossing to construct at local road (Mil | Trench crossing at local road (Queen St.), will cross two widths of railway to address additional railway directed to privite burgers (75 Charles Street) | 7 | Trench crossing will cross railway as well as Baden Creek Tributary A. Tributary A is ephemeral, no | | |
| | Neurotect Casto Case Case | - | Snyder's Road W and Foundry Street crossing | | | | St.) | Trench is in close proximity to wetland MAM2 and culterual woodland CUE1 | | complications anticipated if constructed during dry period. | | |
| 3.B Municipal Services and Utilities | Normatized Hactor Group Score | 1.3 | | 1.6 | | 0.0 | Moderate impact to existing Local and Regional | | 2.3 | | | |
| Municipal Services (sanitary, water, storm) | uppunulnities for connections to the broader sanitary system. Impacts to existing sanitary, water, & storm infrastructure during construction. Undermines existing infrastructure. | 3 | Major impact to existing Regional infrastructure | 1 | Similar to A1 but deeper sanitary sewer alignment leads to larger potential impact to existing services | 6 | services, 4oumm regional WTM along Christian Street, Charlotta Street, MII Street, Charles and Foundry. Large portion of Foundry Street services potentially impacted. | Minimal impact to existing Local and Regional services, 450mm regional WTM along Christian Street, Charlotta Street, and Gingerich Road. | 10 | Least impact to existing local services, 450mm regional WTM present along Christian Street and Gingerich Road. | | |
| Municipal Infrastructure | Wasterwater Pumping Stations Noise Wall | 1 | Similar to A2, requires and additional pumping station in Parcel O | 5 | No additional pumping station required, Baden West WWPS will remain, School WWPS to utimately be decomplement with the second statement of the second | 8 | Baden West WWPS and School WWPS to be decommissioned, noise wall removed on Charlotta | Baden West WWPS and School WWPS to be 9 decommissioned (1 total pumping station), noise | 10 | Baden West WWPS and School WWPS to be decommissioned (1 total pumping station), noise unit memory of a constraint of the station of the sta | | |
| Utilities (Bell, Hydro, Gas) | Degree of exposure of utilities and/or utility conflicts with road design. | 1 | Impact along regional roads, transmission line to local neighbourhoods, hydro lines along farm road | 2 | wwwmmissioned. Noise wall stays on Charlotta Street. Impact along regional roads, transmission line to local neighbourhoods | 5 | overs Impact along regional and local roads, including transmission line to local neighbourhoods | 8 Impact along some local roads | 10 | www.removed.on.charlotta Street | | |
| | Normalized Factor Group Score TOTAL CATEGORY SCORE % BEHIND FIRST RANKED ALIGNMENT | 1.7 1.5 85% | | 2.7 2.2 77% | | 6.3 6.2 36% | | 8.3 7.9 18% | 10.0 9.6 0% | | | |
| 4. FINANCIAL | CATEGORY BANKING | 5 | | 4 | | 3 | | 2 | | | | |
| Capital Cost | Total Construction costs | 1 | \$8,222,525 (1800m & 1554m forcemain of pipes, pumping station, 2 trenchless crossings) Traffic costs during maintenance. A superior static | 7 | \$5,741,535 (1800m of pipe, 1 trenchless crossing) | 8 | \$5,142,051 (-1985m of pipe, 1 trenchless crossing) Traffic costs during maintenance 1 outputs | 9 \$4,837,808 (-2095m of pipe, 3 trenchiess crossings) Traffic control during maintenance 4 | 10 | \$3,554,104 (1905m of pipe, 3 trenchiess crossings) Traffic control during maintenance 4 | | |
| Operational Cost (Sanitary sewer) | Normalized Factor Group Score | 1.0 | maintenance of right-of-ways. | 5 6.0 | maintenance of right-of-ways. | 6 7.0 7.0 | maintenance of right-of-ways. | station, maintenance of right-of-ways. 8.5 8.5 | 10 10.0 | station, maintenance of right-of-ways. | | |
| | N BEHIND FIRST RANKED ALIGNMENT CATEGORY RANKING | 90% 5 | | 40% 4 | | 30% 3 | | 15% 2 | 0% | | | |
| B. TEGHNICAL | | | | | | | | | | | | |
| | Physical constructability (adequate access to construction area). | | | | Entirety of construction within two Regional Roads | | Alignment is located within the most local right-of-ways, narrow widths and deep sever depth incrase | Similar to A5, but additional alignment is present within local marks and less with the anticultural | | Large portion of the alignment is within agricultural lands at minimal depths, ease of access throughout lands. Narrow ROW on Christian Street paired with | | |
| Constructability | Potential conflicts with existing infrastructure (other than utilities). Depth, trenchless crossings Water Table and groundwater dewatering (short term impact). | 1 | Similar to A2 with additional trenchless crossing north of Snyder's Road West. | 2 | Foundry Street and Snyder's Road West. Trenchless railway crossing at intersection will be very complex. 2 open cut trenches along Foundry Street. | 6 | complexity of alignment. Two open cut crossings along Regional Road Foundry Street. Dewatering and flow rerouting required. Trenchless crossing at MII Street considered easiest of the 5 alternatives. | 9 lands. Potential interaction with wetlands MAM2 and MAM2-2. Trenchless crossing at Tributary A is closer to Baden Creek than A5. | 10 | deep trunk sewer leads to a complex installation of Trunk, including complete removal and replacement of local sewer down Christian (minimal), regional 450mm watermain down Christian St. and Charlotta | | |
| | | | | | | | | | <u> </u> | St., railway and 2 stream crossings. | | |
| | | | Requires a 60m trenchless railway crossing at the intersection of Snyder's Road West and Foundry Street. | | Requires a 60m trenchless railway crossing at the | | n | Requires 95m of trenchless piping for the railway crossing and 2-30m trenchless stream crossing. | | Requires a 60m trenchless railway crossing and 2 trenchless stream crossing (1-40m, 1-30m). This | | |
| Trenchless (Constructability and Complexity) | The complex issues associated with the construction of the trunk sewer. | 1 | Railway has a narrow right-of-way through the intersection requiring sending and receiving pits to be located outside of intersection. Two additional sever connections at intersection. Requires an artificiant action of the section section and the section of the section of the section and the section section action of the secti | з | Intersection of Snyder's Road West and Foundry Street. Railway has a narrow right-of-way through the Intersection requiring sending and receiving pits to be located outside of intersection. Two arkitional come | 10 | crossing. Though a deep trench, it provides the straightest, most direct installation, sufficient clearance for sending and receiving pits. No trenchless stream | The railway right-of-way is wider to accommodate for spurine. The trenchless connection will pass directly under a local wetland and second wetland is in close proximity to the servition of Trenchlese | 9 | alignment has back to back sending and receiving pits for the railway crossing and the crossing along Tributary A. No groundwater interaction at trenchless location. Trenchless stream crossing | | |
| | | | trenchless forcemain stream crossing north of Snyder's Road East. | | connections at intersection. | | crossings required. | stream crossing upstream of existing culvert north of Gingerich Road. | | upstream of existing culvert north of Gingerich Road. | | |
| Property (Landlacouisition) | Property requirements (number, area) and ease of acquisition | 1 | Understanding with the property owner that arrangements will be made regarding securing the land for the provide the security of the securit | 10 | No land acquisition required, fully within existing right-of- | 9 | No land acquisition required, fully within existing right-of- | Understanding with the property owners that arrangements will be made recording accurate the | 7 | Understanding with the property owners that arrangements will be made regarding securing the cloth of under the security of | | |
| Serviceability (Long-Jerm) | Percentage of Barlen to be serviced by Triek server- | | we are pumping station. Uncertainty surrounding securing the right-of-way along the private farm road. | | WE service approximately 80% of Poston | 7 | Will service approximately 0455 of Poston | right-of-way through open areas. | 40 | High Holl-Wey webugn open areas. Larger area to secure than Alternative 4. Will service approximately 100% of Baden & | | |
| 2 C d miny | Normalized Factor Group Score | 1.0 1.0 | | 5.0 | | 8.0 8.0 | and approximately of a condition | Foxboro 8.5 8.5 8.5 | 9.0 | 1-000070 | | |
| | W BEHINU FIRST RANKED ALIGNMENT CATEGORY RANKING GRAND TOTAL | 90% | | 44% | | 3 | | 38 | 1 | | | |
| | GRAND TOTAL % BEHIND FIRST RANKED ALIGNMENT | 80.2% | | 23 49.6% | | 28.0% | | 18.4% | 0.0% | | | |



Detailed Evaluation Matrix



| PRELIMINARY EVALUATION CRITERIA - BADEN SANITARY TRUNK Alternative | | | | | | | | | | | | |
|---|---|-------------------|--|-------------------|---|-------------------|--|---|--------------------|---|--|--|
| Factor Category / Factor Group / Factor | | | A1 | | A2 | | Alternative A3 | A4 | | A5 | | |
| Date: November 2022 | Description | Modified 2014 | Environmental Assessment Preferred Alternative (Do Nothing) | | Snyder's/Foundry | CI | vistian/Charlotta/MII/Charles/Foundry | Christian/Charlotta/Queen and Baden Creek | | Christian and Baden Creek | | |
| | | Score | Comments | Score | Comments | Score | Comments | Score Comments | Score | Comments | | |
| 1. NATURAL ENVIRONMENT 1.A Terrestrial Ecosystems | | | | | | | | | | | | |
| | Impact on feature and function of vegetation communities and flora | | Impact along Foundry St, near pond and area N. of Snvder's Rd West. Construction of segments may | | Neoligible impacts to natural features along Foundry St. | | Similar to Option A2, except more potential impacts along Charlotta St and Foundry St (CUM1) during | Largest extents of construction within natural areas. Temporary disturbances to CUM1 meadow. Construction is in proximity to FOD7-3. potential | _ | Traverses near less meadow area than A5. | | |
| vegetation communities and Pora | -ecological significance (type, size, composition; species of conservation concern) -sensitivity/ susceptibility (e.g., mortality, stress, composition change). | * | impact CUW1, CUM1. Indirect potential impacts to CUT1 and areas near SUM1 unless properly mitigated. | 10 | and Snyder's Road W. | 0 | construction to regionally significant Running Serviceberry if not properly mitigated. | edge tree removal and tree root disturbance, regionally significant Running Serviceberry disturbance. | 5 | depending on surveyed alignment. | | |
| | Impact on feature and function of wildlife habitat -significance (e.g., size, presence of specialized habitat elements, species of | | Anticipated but unknown impacts to wildlife north of Snyder's Rd W within wetlands which were not part of the NEA (impacts therefore considered potentially | | | | Similar impacts to Baden Creek habitats as A1 and A2. | Largest extent of construction within natural areas. Impacts to amphibian breeding habitat (MAM2-2). | | Aligment runs along outer boundary of wildlife corridor. No impacts to wetland habitats. Less | | |
| Wildlife Habitat | conservation concern) -sensitivity (e.g., noise, light) Effects on wildlife movement corridors (e.g., habitat removal, fragmentation, potential increases in animal-vehicle conflicts). | 1 | significant to wildlife). Potential tree removals in CUW1 required which may house SAR bats. Potential impact to Turtle Wintering Areas in Baden Creek Pond habitat | 6 | Similar impacts to Baden Creek habitats as A1 and A3. | 5 | A3 scored slightly lower than A2 due to construction dewer depth and extent of work. | 4 Closest impact to forested area, potential disturbance to SAR bat habitats (FOD7-3). Temporary impacts to habitats within CUM 1. | 10 | impacts to CUM1 habitats than A4. Potential tree removals in CUW1 required which may house SAR bats. | | |
| 1.B Wetland Ecosystems | Normalized Factor Group Score | 2.5 | due to dewatering. | 8.0 | | 6.5 | | 2.5 | 7.5 | | | |
| | | | Construction of sewer north of Snyder's Road West & Foundry Street may directly impact Locally Significant Workey development and the service of the servi | | No impact to wetlands. Alignment is in slight proximity | | No. Incore da constitución del accordo de la constitución de | Construction directly impacts wetlands MAM2 & | | | | |
| Wetlands | relative extent, function in broader welland complex ecosystem). | 1 | meadow marsh unless miligated. Construction may disrupt local drainage regime and hydrological conditions that support the wetland. | 8 | to Locally Significant Wetland north of Snyder's Road West & Foundry Street. | 9 | to MAM2. | 4 Invance case of potential of the control of th | 10 | Does not impact wetlands. | | |
| 1.C Aquatic Ecosystems | Normalized Factor Group Score | 1.0 | | 8.0 | | 9.0 | | 4.0 | 10.0 | | | |
| | -Number of water bodies and watercourse crossings. | | 3 watercourse crossings; a stream crossing north of Snyder's Road West to Tributary B, a stream crossing | | | | | 2 wateroourse crossings: a stream crossing at Gingerich Road, a stream crossing at Baden Creek Tributen A unchane of the configuree Tributen A | | 2 watercourse crossings: a stream crossing at Gingerich Road, a stream crossing at Baden Creek | | |
| Aquatic Habitat | Extent (area) and function of riparian habitat removed. -Extent (area) and function of riparian habitat removed. -Extent and type of fish habitat (in-stream) altered/displaced in watercourse, Including importance to aquatic ecosystem (e.g. spawning, nursery areas, habitat | 1 | to Baden Creek Pond, a cold water stream crossing at Huehn Award Drain. All watercourses provide aquatic habitat, subject to federal Fisheries Act. Crossings require fish capture and relocation during construction. | 4 | 2 watercourse crossings; a stream crossing to Baden Creek Pond, a cold water stream crossing at Huehn Award Drain. | 3 | 2 watercourse crossings; a stream crossing to Baden Creek Pond, a cold water stream crossing at Huehn Award Drain. Al is a slightly deeper crossing than A2. | 7 interaction, may be accessible to fish. Potential to construct through tributary during period of no | 10 | Tributary A upsteam of the confluence. Tributary A identified as intermittent flow with no groundwater interaction, may be accessible to fish. Potential to construct through tributary during period of no flows. | | |
| | for species of conservation concern or less tolerant species). | | watercourses need isolation and de-watering, water quality treatment, etc | | | | | rows. Trenchess instalation to be usized for both crossings.Tributary A crossing is closer to Baden Creek than A5. | | Trenchless installation to be utilized for both crossings. | | |
| 1.D Surface Water (Quality - Impact to Water Quality / Ti | Normalized Factor Group Score hermal / Chlorides) | 1.0 | | 4.0 | | 3.0 | | 7.0 | 10.0 | | | |
| Water Quality | Impact on water quality of surface water features (TSS, gas/oil fueling, construction operations). | 1 | Similar to A2 with additional upstream impacts at stream crossing north of Snyder's Road West, increased risk of surface water impacts adjacent to a code significant Westerd A1 vitimately considerations. | з | Two open cut stream cutvert crossings at the Baden Creek Pond Tributary B and Huehn Award Drain. Flow reroufing and dewatering required to prevent increased flows. Increased potential for contaminated sediments | 2 | Similar to A2 with deeper sanitary sewer depths, further potential impacts during construction. | Similar to A5, but trenchiess crossing is closer to Baden Creek. Increased potential impacts to water | 10 | Proposed trenchess crossing at Baden Creek Tributary A is an intermittent flow tributary with no evidence of groundwater interaction, no dewatering required during construction. No anticipated impacts | | |
| | | | TSS levels in Baden Creek unless properly mitigated. | | from the Regional Roadway to enter Baden Creek Pond and Baden Creek. | | | quanty. | | at the trenchless crossing underneath the culvert at Gingerich. | | |
| Thermal Regime | Impact in the thermal regime of surface water features. | 1 | Similar to A2 with additional upstream impacts at stream crossing north of Snyder's Road West. Additional dewatering and construction can ultimately | 3 | Close proximity to HAD which provides coldwater in support of cool water habitat in Baden Creek Pond. Pond has GW connectivity as well, where dewatering during construction can affect thermal regime. Potential | 2 | Similar to A2 with deeper sanitary sewer depths, further potential impacts during construction | Similar to A5, but trenchless crossing is closer to 8 Baden Creek. Increased potential impacts to | 10 | No impact on Baden Creek Pond and Huehn Award Drain. Trenchiess crossing on intermittent tributary, | | |
| - | | | can affect thermal regime in Baden Creek Pond and HAD. | | impact to vegetation that regulates temperature in Baden Creek | | · · · · · · · · · · · · · · · · · · · | thermal regime. | | no groundwater interaction/impact to thermal regime | | |
| Chlorides 1.E Groundwater Resources | Impact on water quality of surface water features. Normalized Factor Group Score | 10 4.0 | Not Significant | 10 5.3 | Not Significant | 10 4.7 | Not Significant | 10 Not Significant 8.7 | 10 10.0 | Not Significant | | |
| Regional Recharge Areas (Groundwater) | Encroachment into the Regional Recharge Area. (removalidisruption of function – area; depth). | 1 | Similar to A2 and A3 with respect to HAD, with additional proximity to recharge areas near pump station within Parend O | 7 | Close proximity to HAD, which is connected to the Regional Recharge Area, depth of trunk may affect GW | 6 | Similar to A2, but sewer is deeper under the HAD. | 9 Similar to A5, alignment is slightly closer to regional recharge area. | 10 | Alignment is furthest away from recharge areas. BH sample shows no sand layer as that of the regional recharge area will not draw convolution. | | |
| Groundwater Quality - Clean Water Act (WHPA, SGRA, IPZ, HVA) | Potential for impacts to vulnerable areas (Area) as defined in Source Water Protection Areas (SWPA). | 1 | Closest proximity to WHPA north of Parcel O | 5 | Similar to A3 | 6 | WHPA location to the east | 9 Similar to A5, alignment is slightly closer to WHPAs and SGRAs. | 10 | Furthest alignment from WHPA location to east, does not affect it | | |
| Shallow Groundwater Movement/Contributions (Quality and Quartity) | Potential for interference with existing flow patterns (baseflow) relative to proximity to significant surface water feature and significant groundwater feature discharge areas. Consider Short, Term Impact Resional shallow we levation 345m | 1 | Shallow forcemain from pump station and stream crossing N of Snyder's Road West, may affect Baden Creek Pond and HAD. Largest extent of construction | 5 | May affect Baden Creek Pond and HAD during construction. | 4 | Proximity to wetland MAM2, may affect Baden Breek Pond and HAD during construction due to depth of server | 2 Close proximity to wetland MAM2 and MAM2-2, tributary crossing at fish habitat location. | 10 | No evidence of GW interaction at Baden Creek tributary A stream crossing, no sand layer in stream crossing at Gingerich Road. GW impacts | | |
| 1.F Surface Drainage (Quantity - Erosion) | Normalized Factor Group Score | 1.0 | and potential impacts to groundwater. | 5.7 | | 5.3 | | 6.7 | 10.0 | minimal/negligible. | | |
| | | _ | Impacts flows at Baden Creek Tributary B, Baden Creek Pond and Huehn Award Drain crossings. Tributary B is | | Impacts flows at Baden Creek Pond and Huehn Award Drain crossings. Crossings are open cut and requires | | Impacts flows at Baden Creek Pond and Huehn Award Drain crossings. Crossings are open cut and requires | Trenchiess railway crossing occuring through a sand layer near local wetlands and section between trenchiess crossing and Queen Street may result in | | No change to flow regime due to tenchiess | | |
| Quantity (flows) | Quantity Control and Flow Regime | 7 | trenchless, others are open cut which requires stream diversion. Mitigation required for sediment control and flow rate. | 6 | stream diversion. Mitigation required for sediment control and flow rate. Deeper trench than A1 | 1 | stream diversion, mitigation required for sediment control and flow rate. Significantly deeper trench than A2 | 8 potential dewatering and rerouting to Baden Creek. No anticipated change to flow regime due to tenchiess crossing north of Gingerich Road. | 10 | crossings at Baden Creek Tributary A and Gingerich Road. No dewatering or flow rerouting anticipated. | | |
| Floodplain Function | Impacts to Regulatory Floodplain and changes to floodplain hydrologic function. (During Construction), including need for GRCA work permit | 7 | Minimal impact to floodplain due to (2) stream crossings occuring along road right-of-way. Imapct to floodplain at stream crossing north of Snyder's Road East. | 10 | Minimal impact to floodplain due to (2) stream crossings occuring along road right-of-way | 9 | Minimal impact to floodplain due to (2) stream crossings occuring along road right-of-way | Alignment along floodplain boundary & impacts GRCA floodplain boundary, closest to floodplain near forest, impact at 2 stream crossing | 6 | Alignment along floodplain boundary & impacts GRCA floodplain boundary, impact at 2 stream crossing | | |
| Stormwater Management Water Balance | Opportunities to enhance roadway stormwater management measures, including coordination with/use of adjacent development facilities. | 10 | Not Significant | 10 | Not Significant | 10 | Not Significant | 10 Not Significant | 10 | Not Significant | | |
| Erosion Assessment | Impact on Slope stability due to stream disturbances and from increased TSS | 1 | Two open cut stream crossings along Foundry Street at the Baden Creek pond and Huehn Award Drain, additional upstream trenchiess crossing at Baden Creek | 4 | Two open cut stream crossings along Foundry Street at the Baden Creek pond and Huehn Award Drain, mitigation remember and the stream of the | 3 | Similar to A2, but sewer is deeper under the HAD, further accelon potential | 9 Similar to A5, trenchless crossing closer to Baden | 10 | Two trenchiess crossings. No dewatering anticipated, so less erosion concerns are present. | | |
| | Normalized Factor Group Score | 7.0 | Tributary B, mitigation required to prevent increased TSS loading and erosion prevention | 8.0 | and erosion prevention | 6.6 | | 7.6 | 92 | place outside buffer where possible. | | |
| | 101AL CALEGORY SCORE % BEHIND FIRST RANKED ALIGNMENT CATEGORY RANKING | 71% 5 | | 31% 2 | 1 | 38% | | 36% | 0% | | | |
| 2. SOCIO-CULT URAL ENVIRONMENT 2.A Land Use Policy | | ſ | | 1 | | ſ | | | 1 | | | |
| Land Use Designations and Policies | Aligned with Provincial, Regional and Township growth and development goals, objectives or policies (high, medium, low). | 1 | Least compatible due to timing and prioritization of regional projects (2 separate regional projects, 1 local project). In addition, temporary disturbance to access for local anticititural operation protect of Sovidr's Road | 2 | Similar to A3, except 2 Regional projects required to facilitate this alignment. Lower scoring due to | 4 | Low degree of compatibility due to timing and prioritization of regional projects combined with local projects (1 regional project, 1 local project), does not | Autonomy to implement infrastructure improvement 10 without upper tier regulations formulating | 8 | Similar to A4, longer construction time within Schmidt Lands (Parocel AA) and disruption to | | |
| | Efficiency of resulting development patterns and timing. | | Does not fully integrate Township's plan to join Baden and New Hamburg servicability. | | complexity of Regional projects. | | fully integrate Township's plan to join Baden and New Hamburg servicability. | development and growth independently | | agricultural lands, impact to growing season | | |
| Impact of the Sanitary Sewer on Adjacent Lands | Impacts to residence and businesses | 1 | Impacts adjacent residential properties and busissess along major regional roads Foundry Street and Snyder's Road West. | 2 | Impacts adjacent residential properties and busissess along major regional roads Foundry Street and Snyder's Road West. | 4 | Impacts several residential properties along local roads Christian Street, Charlotta Street, Mill Street, Charles Street, Impacts businesses and residents along Eventry Street Realow Broot | More residential properties impacted than A5 along Christian Street and Charlotta Street. Coordination required with Schmidt Lands. | 10 | Least amount of properties impacted primarily along Christian Street. Coordination required with Schmidt Lands. | | |
| 2.B Archaeological Resources | Normalized Factor Group Score | 1.0 | | 2.0 | | 4.0 | | 9.0 | 9.0 | | | |
| Known and/or Potential Archaeological Sites | Numbertype/significance of direct/indirect impacts to known archaeological resources and/or areas of high archaeological potential. | 8 | Furthest and shallowest from found artifact, Baden Creek is a barrier to artifact location | 10 | Furthest from found artifact but not as shallow, Baden Creek is a barrier to artifact location | 9 | Furthest and deepest from found artifact, Baden Creek is a barrier to artifact location | Alignment Impacts the 50m monitoring buffer at 1 Findspot-16 Findspot-18 and construction may impact AIHe-469 50m monitoring buffer. | 5 | Alignment impacts the monitoring 50m buffer at Findspot-16 Findspot-18 | | |
| 2.C Built Heritage | Numberltypelsignificance of direct/indirect impacts to built heritage resources //based on presence of built heritage resources identified by the Review of | 2 | Construction will affect scenic routes along Snyder's Road West and Foundry Street: Adjacent to 4 listed and | 1 | Similar to A1, but road closures anticipated to be | 5 | Construction will affect scenic route along Foundry Street: Adjacent to 4 listed and 2 designated importies | 9 Similar to A5, alignment is closer in proximity to | 10 | No direct impacts to heritage properties or OHA, alignment is near one listed property: 130 Queen | | |
| 2.D Noise | Waterloo). Normalized Factor Group Score | 2.0 | 2 designated properties listed on OHA | 1.0 | longer due to depth of servers | 5.0 | listed on OHA | 9.0 | 10.0 | Street | | |
| Noise | Noise impacts due to addition/removal of wastewater pumping stations (wwps) Normalized Factor Group Score | 1 | Noise from additional wwps, noise wall remains at Baden West WWPS | 2 | No additional wwps, noise wall remains at Baden West WWPS | 8 | Removal of noise wall and Baden West WWPS | Removal of noise wall and Baden West WWPS 9.0 7.0 | 10 | Removal of noise wall and Baden West WWPS | | |
| | % BEHIND FIRST RANKED ALIGNMENT CATEGORY RANKING | 65% 5 | | 56% 4 | | 24% 3 | | 18% 2 | 0% | | | |
| 3. TRANSPORTATION/MUNICIPAL SERVICES AND UT 3.A Transportation Network (Roads and Pedestrians) | ILITIES | | | 1 | | | | | 1 | | | |
| Roads and Intersections | Potential impacts to roads/intersections. | 1 | Similar to A2, additional impact to private farm road access north of Snyder's Road East | 2 | 2 regional roads, 11 road intersection, major intersection impact of Snyder's Road West and Foundry Street | 5 | Christian Street, Charlotta Street, Mil Street, Carles Street, large portion of Foundry Street (Regional) / 11 intersections | Christian Street, larger portion of Charlotta than A5, Queen Street, portion of Gingerich Road, portion of Foundry Street (Regional) / 8 intersections | 10 | Christian Street, portion of Charlotta, portion of Gingerich Road, portion of Foundry Street (Regional) / 3 intersections | | |
| | | | | | | | Impedes traffic along the most local roads (4: Christian Street, Charlott, Street, Mil Street, Charlott, Street) and | | | Impades the level amount of Imflia throughout | | |
| Traffic Operations | Road safety and accessibility (sight distance; turning movements). Capacity/capability to accommodate forecast traffic demand. Provides for appropriate access to existing properties. Impact on access to existing properties. | 1 | Similar to A2, additional impacts to farm lane access at Snyder's Road/Foundry intersection. | 2 | Impedes traffic along two Regional Roads Foundry Street and Snyder's Road West. Significant temporary traffic operations routing through local roads required during construction. | 4 | a portion of Regional Road Foundry Street. 11 intersections will have line of sight affected during construction. Additional arrangements required for | 9 9 10 10 10 10 10 10 10 10 10 10 | 10 | construction. Narrow local right of ways, construction will impact entirety of roadway. 3 Intersections will have sightline affected. Least | | |
| | | | | | ч - | | construction. | · | | amount of nomes have access attected. | | |
| Transportation policy initiatives | Capability to support municipal policy initiatives (transit, active transportation, roundabouts). | 1 | Transit obstruction for bus service to and from Kitchener (bus line 77-intersection of Snyder's Rd W & Foundry St), additional obstruction of access along Farm Road | 2 | Transit obstruction for bus service to and from Kitchener (bus line 77-Intersection of Snyder's Rd W & Foundry St), | 5 | Minimal impact to local transit, will affect access to several residential properties in neighborhoods and along Foundry Street. No direct impact to transit mates | Similar to A5 with addition impacts to Charlotta 8 Street and Queen Street. No direct impact to transit routes. | 10 | Least potential transit obstructions (kiwanis transit) only affects Christian Street and small portion of Charlotta Street, will affect access to residential properties in neighborhoods. No direct impact to | | |
| | | | | | | | | Trench crossing at local road (Queen St.), will cross | | trnasit routes. Trench crossing will cross railway as well as Baden | | |
| Railroad Crossing | Ease of railroad crossing, approval | 2 | Significant trench required through main intersection of Snyder's Road W and Foundry Street crossing | 1 | Similar to A1 but deeper trench required. | 10 | Easlest trench crossing to construct at local road (Mil St.) | two widths of railway to address additional railway directed to private business (75 Charles Street). Trench is in close proximity to wetland MAM2 and culterual woodland CUE1 | 7 | Creek Tributary A. Tributary A is ephemeral, no complications anticipated if constructed during dry period. | | |
| 3.B Municipal Services and Utilities | Normalized Factor Group Score | 1.3 | | 1.8 | | 6.0 | | 7.5 | 9.3 | | | |
| Municipal Services (sanitary, water, storm) | Opportunities for connections to the broader sanitary system. Impacts to existing sanitary, water, & storm infrastructure during construction. | 3 | Major impact to existing Regional infrastructure | 1 | Similar to A1 but deeper sanitary sewer alignment leads to larger potential impact to existing services | 6 | Moderate impact to existing Local and Regional services, 450mm regional WTM along Christian Street, Charlotta Street, MII Street, Charles and Foundry. Large portfaller, of Evueting Street project protectable. | Minimal impact to existing Local and Regional 8 services, 450mm regional WTM along Christian Street, Christian Street, and Ginaedich Boot | 10 | Least impact to existing local services, 450mm regional WTM present along Christian Street and Generich Road | | |
| | | | | | No additional pumping station registered Radion Wood | | Baden West WWPS and School WWPS to he | Baden West WWPS and School WWPS to be | | Baden West WWPS and School WWPS to be | | |
| Municipal Infrastructure | wasaewater Pumping Stations Noise Wall | 1 | similar to A2, requires and additional pumping station in Parcel O | 5 | WWPS will remain, School WWPS to utimately be decommissioned. Noise wall stays on Charlotta Street. | 8 | decommissioned, noise wall removed on Charlotta Street | 9 decommissioned (1 total pumping station), noise wall removed on Charlotta Street | 10 | decommissioned (1 total pumping station), noise wall removed on Charlotta Street | | |
| Utilities (Bell, Hydro, Gas) | Degree of exposure of utilities and/or utility conflicts with road design. Normalized Factor Group Score TOTAL CATEGORY COOPE | 1 1.7 1.5 | neighbourhoods, hydro lines along farm road | 2 27 22 | neighbourhoods | 5 6.3 6.2 | respend arong regional and local roads, including transmission line to local neighbourhoods | 8 Impact along some local roads 8.3 7.9 | 10 10.0 9.6 | Impact along 1 local road | | |
| | % BEHIND FIRST RANKED ALIGOMENT CATEGORY RANKING | 85% 5 | | 77% | | 36% 3 | | 18% 2 | 0% 1 | | | |
| • • PINANCIAL Capital Cost | Total Construction costs | 1 | \$8,222,525 (1800m & 1554m forcemain of pipes, bumping station, 2 (merchines, concelerent) | 7 | \$5,741,535 (1800m of pipe, 1 trenchless crossing) | 8 | \$5,142,061 (-1985m of pipe, 1 trenchless crossing) | 9 \$4,837,808 (~2095m of pipe, 3 trenchless crossing=1 | 10 | \$3,564,104 (1905m of pipe, 3 trenchiess crossings) | | |
| Operational Cost (Sanitary sewer) | Operational costs due to traffic during maintenance, pumping station | 1 | Traffic costs during maintenance, 4 pumping stations, maintenance of right-of-ways. | 5 | Traffic costs during maintenance, 2 pumping stations, maintenance of right-of-ways. | 6 | Traffic costs during maintenance.1 pumping station, maintenance of right-of-ways. | 8 Traffic control during maintenance, 1 pumping station, maintenance of right-of-ways. | 10 | Traffic control during maintenance, 1 pumping station, maintenance of right-of-ways. | | |
| | Normalized Factor Group Score TOTAL CATEGORY SCORE % BEHIND FIRST RANKED ALIGNMENT | 1.0 1.0 90% | | 6.0 6.0 40% | | 7.0 7.0 30% | | 8.5 8.5 15% 2 | 10.0 10.0 0% | | | |
| 5. TECHNICAL | | | л Т | | | | | | | | | |
| | | | | | | | Alignment is located within the most local rinht of wow | Planta and a second second second | | Large portion of the alignment is within agricultural lands at minimal depths, ease of anness throughout | | |
| Constructability | Physical constructability (adequate access to construction area). Potential conflicts with existing infrastructure (other than utilities). Depth, trenchless crossings | 1 | Similar to A2 with additional trenchless crossing north of Snyder's Road West. | 2 | Entirety of construction within two Regional Roads Foundry Street and Snyder's Road West. Trenchiess railway crossing at intersection will be very complex. 2 | 6 | narrow widths and deep sever depth incrase complexity of alignment. Two open cut crossings along Regional Road Foundry Street. Dewatering and flow | Similar to A5, but additional alignment is present within local roads, and less with the agricultural lands. Potential interaction with wetlands MAM2 and MAM2-2. Trenchless crossing at Tributary A is | 10 | lands. Narrow ROW on Christian Street paired with deep trunk sewer leads to a complex installation of Trunk, including complete removal and replacement | | |
| | YY MME I ADME and groundwater dewatering (short term impact). | | | | open cut trencnes along Foundry Street. | | rerouting required. Trenchless crossing at Mill Street considered easiest of the 5 alternatives. | closer to Baden Creek than A5. | | or sical sewer down Christian (minimal), regional 450mm watermain down Christian St. and Charlotta St., railway and 2 stream crossings. | | |
| | | | Resulters a 60m topoblere miliusuin | | | | | Resulter Ofm of Issochart states to the set | | Requires a film trenchiser miller | | |
| Trenchless (Constructability and Commission) | The complex issues associated with the construction of the test- | | Intersection of Snyder's Road West and Foundry Street. Railway has a narrow right-of-way through the intersection requiring sending and receiving pits to be | | Requires a 60m trenchless railway crossing at the intersection of Snyder's Road West and Foundry Street. Railway has a narrow right-of-way through the | 10 | Requires 40m of trenchless piping for the railway crossing. Though a deep trench, it provides the straightest, most direct instrations of direct and the second s | requires sum or tenchless piping for the railway crossing and 2-30m tenchless stream crossing. The railway right-of-way is wider to accommodate for spur line. The trenchless connection will pass | | trenchess stream crossing (1-40m, 1-30m). This alignment has back to back sending and receiving pits for the railway crossing and the crossing along | | |
| . , | , LONG CONTRACTOR OF UNE DURING STREET. | | connections at intersection. Two additional sever connections at intersection. Requires an additional 20m trenchless forcemain stream crossing north of Snyder's Road East. | | mersection requiring sending and receiving pits to be located outside of intersection. Two additional sewer connections at intersection. | | for sending and receiving pits. No trenchless stream crossings required. | arrectly under a local wetland and second wetland is in close proximity to the sending pit. Trenchless stream crossing upstream of existing outvert north of Gingerich Road. | | trenchess location. Trenchess stream crossing upstream of existing culvert north of Gingerich Road. | | |
| | | | Understanding with the nonnerty owner that | | | | | | | Understanding with the nonnecty current these | | |
| Property (Landlacquisition) | Property requirements (number, area) and ease of acquisition | 1 | arrangements will be made regarding securing the land for the pumping station. Uncertainty surrounding securing the right-of-way along the private farm road. | 10 | No land acquisition required, fully within existing right-of- ways | 9 | No land acquisition required, fully within existing right-of- ways | Understanding with the property owners that arrangements will be made regarding securing the right-of-way through open areas. | 7 | arrangements will be made regarding securing the right-of-way through open areas. Larger area to secure than Alternative 4. | | |
| Serviceability (Long-term) | Percentage of Baden to be serviced by Trunk sewer. Normalized Factor Group Goods | 1 | Will service approximately 75% of Baden | 5 | Will service approximately 80% of Baden | 7 | Will service approximately 94% of Baden | 9 Will service approximately 100% of Baden & Foxboro 8.5 | 10 | Will service approximately 100% of Baden & Foxboro | | |
| | TOTAL CATEGORY SCORE % BEHIND FIRST RANKED ALIGNMENT CATEGORY RANKING | 1.0 89% 90% | | 5.0 44% 4 | | 8.0 11% 3 | | 8.5 6% 2 | 9.0 0% 1 | | | |
| | GRAND TOTAL % BEHIND FIRST RANKED ALIGNMENT | 9 80.2% | | 23 49.6% | | 34 28.0% | | 38 18.4% 2 | 47 | | | |



Preliminary Plan and Profile Drawings and Design Sheet



| BADEN - NEW HAMBURG SANITARY SERV | ICING | | | | | | | | | | | De | sign Para | ameters | | | | | | | | | |
|---|----------------------------------|---------------------------------|----------------------------------|-----------------------------|----------------------------------|--------------------|--|-------------|--------------------------|-----------------|-------------|----------------------------------|--------------|-------------------------|--------------|----------------|----------------|---------------|----------------|--------------|------------|----------------|--------------|
| (ALTERNATIVE #5) | | | SANITAR | | SIGN S | SHEET | Average Daily Flow Mannings "n" 0.0130 | | | | | | | | | | | | | | | | |
| TOWNSHIP OF WILMOT, Ontario | | Ppha) | | | 0.0032 L/s/c Min. Velocity 0.6 m | | | | | | | 6 m/sec | | | | | | - 6.6 | | | | | |
| Project Number: | 48730-104 | | ENGINE | | | | JNKS | - | | | | | Residentia | al Harmon Peaking | Factor (F) | | | | | | $A^{*}A$ | | |
| Date: | November 9, 2022 | | | | | | | Commore | ial | 0.95 | l /c/ba | | Rosidontia | Areas Infiltration | 0.25 | l /c/ba | | | | C.M | | | |
| Checked By: | JEM | | | | | | | Industrial | iai | 0.35 | L/s/ha | | i lesidentia | | 0.25 | L/3/11a | | | | | | | |
| File: | Q:\48730\100\SAN\48730-100 ALT 5 | Sanitary Servicing Strategy.xls | | | | | | Inst. / Sch | | 0.25 | L/s/ha | | | | | | | | | | | | |
| LOCAT | ΓΙΟΝ | | RESIDENTIAL AREAS and POPULATION | | | | | | INSTITUTIONAL COMMERCIAL | | | | INDUSTRIAL | | | INFILTRATION | | | | | DESIGN | | |
| STREET | AREA MANH | OLE LOCATION | | | СОМИ | | PEAK | | 0.25 <i>L</i> | HECTA L/s/ha | RES AND FLC |)W OF EACH).95 <i>L/s/ha</i> | ZONING | 0.40 <i>L/s/ha</i> | TOTALS- | | CUMUL | INFIL | | | | CARACITY | FULL FLOW |
| omeen | NO. FROM MH | TO MH | AREA DENO | | POPU | L. "F" | FLOW | AREA | CUMUL AREA | PEAK FLOW | AREA CUM | UL PEAK | AREA | CUMUL PEAK AREA FLOW | FLOW | AILLA | AREA | FLOW | FLOW | | | | VELOCITY |
| | | | ha ppha | 1000 | s 1000s | 3 | L/sec | ha | ha | L/sec | ha ha | L/sec | ha | ha L/sec | L/sec | ha | ha | L/sec | L/sec | m % | тт | L/sec. | m/s |
| 1 - TO MH22 ON SNYDER'S RD W | | | | | | | | | | | | | | | | | | | | | | | |
| FOXBORO | 1 FOXBORO | P1 (1012 LANDS) | 42.12 | 15 0.0 | 632 0.6 | 32 3.92 | 0 7.88 | 3 | | | | | | | 0.00 | 42.12 | 42.12 | 10.53 | 18.41 | 0.5 | 250 | 42.03 | 0.86 |
| Parcel P2 | 2 P2 (1012 LANDS) | P1 (1012 LANDS) | 30.58 | 60 1.8 | 335 1.8 | 35 3.61 | 5 21.1 | 1 | | | | | | | 0.00 | 30.58 | 30.58 | 7.65 | 28.75 | 0.5 | 250 | 42.03 | 0.86 |
| Parcel O2 Parcel O1 | 3 O2 4 O1 | O1 P1 (1012 LANDS) | 24.91 14.91 | 60 1.4 60 0.4 | 195 1.4 395 2.3 | 95 3.68 89 3.52 | 1 17.5 ⁻ 4 26.80 | 1 | | | | | | | 0.00 0.00 | 24.91 14.91 | 24.91 39.82 | 6.23 9.96 | 23.74 36.76 | 0.5 0.5 | 250 250 | 42.03 42.03 | 0.86 0.86 |
| Confluence through Parcel P1 (1012 Snyder's Road) | 5 P1 (1012 LANDS) | BADEN TRUNK | 20.46 | 60 1.: | 228 6.0 | 83 3.16 | 5 61.28 | 3 | | | | | | | 0.00 | 20.46 | 132.98 | 33.25 | 94.53 | 0.3 | 450 | 168.59 | 1.06 |
| Parcel Y2 Parcel Y1 | 6 Y2 7 Y1 | Y1 SNYDERS RD W | 24.24 18.40 | 60 1.4 60 1.1 | 154 1.4 104 2.5 | 54 3.68 58 3.50 | 9 17.08 0 28.50 | 3 | | | | | | | 0.00 0.00 | 24.24 18.40 | 24.24 42.64 | 6.06 10.66 | 23.14 39.16 | 0.50 0.50 | 250 250 | 42.03 42.03 | 0.86 0.86 |
| Parcel Z | 8 Z | SNYDERS RD W | 7.98 | 60 0.4 | 179 0.4 | 79 3.98 | 4 6.07 | 7 | | | | | | | 0.00 | 7.98 | 7.98 | 2.00 | 8.07 | 0.5 | 200 | 23.18 | 0.74 |
| SCHOOL PS | 9 SCHOOL PS | SNYDERS RD W | | 0.0 | 0.0 0.0 | 00 4.50 | 0 0.00 | 15.35 | 15.35 | 3.84 | | | | | 3.84 | 15.35 | 15.35 | 3.84 | 7.68 | 0.50 | 200 | 23.18 | 0.74 |
| Confluence on Snyder's Road West | - SNYDERS RD W | BADEN TRUNK | | 0.0 | 000 3.0 | 37 3.43 | 8 33.23 | 3 | 15.35 | 3.84 | | | | | 3.84 | 0.00 | 65.97 | 16.49 | 53.56 | 0.5 | 300 | 68.34 | 0.97 |
| Confluence to Baden Trunk @ Snyder's Rd W | - BADEN TRUNK | | | | 9.1 | 21 2.99 | 4 86.92 | 2 | 15.35 | 3.84 | | | | | 3.84 | 0.00 | 198.95 | 49.74 | 140.50 | 1.0 | 450 | 284.96 | 1.79 |
| 2 - CHRISTIAN/CHARLOTTA TO SCHMIDT L | ANDS | | | | | | | | | | | | | | | | | | | | | | |
| Parcel X | 10 X | Q | 13.43 | 60 0.8 | 306 0.8 | 06 3.85 | 9 9.90 | þ | | | | | | | 0.00 | 13.43 | 13.43 | 3.36 | 13.25 | 0.5 | 200 | 23.18 | 0.74 |
| Parcel W | 11 W | Q | 35.85 | 60 2. | 151 2.1 | 51 3.56 | 1 24.38 | 3 | | | | | | | 0.00 | 35.85 | 35.85 | 8.96 | 33.34 | 0.5 | 250 | 42.03 | 0.86 |
| Parcel Q | 12 Q | BADEN WEST | 32.17 | 60 1.9 | 930 4.8 | 87 3.25 | 4 50.62 | 2 | | | | | | | 0.00 | 32.17 | 81.45 | 20.36 | 70.98 | 0.5 | 375 | 123.91 | 1.12 |
| Ex. Baden West Subdivision + Christian Street (incl. 9.16ha from Ex. WWPS per Baden West Subdivision - | 13 BADEN WEST | BADEN TRUNK | 14.62 | 60 0.8 | 377 5.7 | 64 3.18 | 7 58.47 | 7 | | | | | | | 0.00 | 14.62 | 96.07 | 24.02 | 82.49 | 0.50 | 375 | 123.91 | 1.12 |
| Wastewater Pumping Station Design Brief [Stantec, 2005]) | | | | | | | | | | | | | | | | | | | | | | | |
| Confluence to Baden Trunk @ Christian/Charlotta | BADEN TRUNK | | | | 14.8 | 85 2.78 | 2 131.78 | 3 | 15.35 | 3.84 | | | | | 3.84 | 0.00 | 295.02 | 73.76 | 209.37 | 1.0 | 450 | 284.96 | 1.79 |
| 3 - FLOWS THROUGH SCHMIDT LANDS TO | GINGERICH | | | | | | | | | | | | | | | | | | | | | | |
| Parcel AA (N-NW Portion to Trunk) | 14 AA1 | BADEN TRUNK | 67.11 | 60 4.0 |)27 4.0 | 27 3.33 | 1 42.69 | 9 | | | | | | | 0.00 | 67.11 | 67.11 | 16.78 | 59.46 | 0.5 | 300 | 68.34 | 0.97 |
| Confluence to Baden Trunk | BADEN TRUNK | | | | 18.9 | 11 2.67 | 7 161.13 | 3 | 15.35 | 3.84 | | | | | 3.84 | 0.00 | 362.13 | 90.53 | 255.50 | 0.6 | 525 | 346.55 | 1.60 |
| Parcel AA (SW Portion to Trunk) | 15 AA2 16 AA3 | AA3 BADEN TRUNK | 19.60 17.07 | 60 1. ⁻ 60 1. | 176 1.1)24 2.2 | 76 3.75 00 3.55 | 4 14.05 3 24.88 | 5 | | | | | | | 0.00 0.00 | 19.60 17.07 | 19.60 36.67 | 4.90 9.17 | 18.95 34.05 | 0.5 | 200 250 | 23.18 42.03 | 0.74 0.86 |
| Parcel BB (South Industrial Lands to Trunk) | 17 BB2 | BADEN TRUNK | | 0.0 | 0.0 0.0 | 00 4.50 | 0 0.00 | D | | | | | 19.73 | 19.73 7.8920 | 7.89 | 19.73 | 19.73 | 4.93 | 12.82 | 0.6 | 200 | 25.39 | 0.81 |
| Confluence Baden Trunk @ to Gingerich Road | BADEN TRUNK | | | | 21.1 | 12 2.62 | 9 176.65 | 5 | 15.35 | 3.84 | | | | 19.73 7.8920 | 11.73 | 0.00 | 418.53 | 104.63 | 293.01 | 0.2 | 675 | 420.08 | 1.17 |
| 4 - FLOWS TO FOUNDRY STREET TRUNK | | | | | | | | | | | | | | | | | | | | | | | |
| Ex. Activa MDS Lands to Foundry (Area per Stantec Sanitary Sewer Design Sheet [2009]) | 18 BAD1 | FOUNDRY | 53.12 | 60 <i>3.</i> : | 87 3.1 | 87 3.42 | 0 34.69 | 2.25 | 2.25 | 0.5625 | 0.91 0 | 0.91 0.8645 | | | 1.43 | 56.28 | 56.28 | 14.07 | 50.19 | 0.60 | 300 | 74.87 | 1.06 |
| <u>West side to Foundry</u> Ex. Lots to Snyder's Road West (North of CN Rail) | 19 BAD2 | FOUNDRY | 12.06 | 50 O 2 | 724 0.7 | 24 3.88 | 6 8.9.ª | 5 | | | | | | | 0.00 | 12.06 | 12.06 | 3.02 | 11.97 | 0.50 | 200 | 23.18 | 0.74 |
| | | | | 5.7 | | 2.50 | | | | | | | | | | | | | | 5.00 | | | |

| LOCATION | | | | | | RESIDENTIAL AREAS and POPULATION | | | | | | AL | COMMERCIAL INDUSTRIAL | | | | INFILTRATION | | | DESIGN | | | | | |
|--|-----|-----------|------------------|-------------------|--------------|----------------------------------|--------|--------|----------|--------------|-------------|----------------------------|--|---------------------------|-----------------|---------|--------------|--------|--------|-----------------|-----|-----|-----|----------|-----------|
| | | | MANHOLE LOCATION | | | | | CUMUL | PEAK | PEAK | 0.25 L/ | HECTA /s/ha | TARES AND FLOW OF EACH ZONING 0.95 <i>L/s/ha</i> 0.40 <i>L/s/ha</i> | | | TOTALS- | | CUMUL | INFIL | TOTAL | | | | | FULL FLOW |
| SINEE | NO. | | FROM MH | TO MH | AREA DENSITY | POPUL. F | POPUL. | "F" | "F" FLOW | AREA CUMUL F | PEAK LOW | AREA CUMUL PEA AREA FLC | W AREA | A CUMUL PEAK AREA FLOW | FLOW | AREA | AREA | FLOW | FLOW | LENGTH SLOTE TH | | | | VELOCITY | |
| | | | | | ha | ppha | 1000s | 1000s | | L/sec | ha ha l | L/sec | ha ha L/s | ec ha | ha L/sec | L/sec | ha | ha | L/sec | L/sec | m % | I | mm | L/sec. | m/s |
| Ex. Lots west of Foundry Street @ Charles (South of CN Rail) | 20 | BAD3 | | CHARLES @ FOUNDRY | 7.85 | 60 | 0.471 | 0.471 | 3.987 | 5.98 | | | | | | 0.00 | 7.85 | 7.85 | 1.96 | 7.94 | 0. | 50 | 200 | 23.18 | 0.74 |
| Parcel U | 21 | 1 U | | CHARLES @ FOUNDRY | 2.48 | 60 | 0.149 | 0.149 | 4.192 | 1.99 | | | | | | 0.00 | 2.48 | 2.48 | 0.62 | 2.61 | 0. | .50 | 200 | 23.18 | 0.74 |
| Parcel BB1 to Foundry | 22 | 2 BB1 | | FOUNDRY | 13.17 | 60 | 0.790 | 0.790 | 3.864 | 9.72 | 2 | | | | | 0.00 | 13.17 | 13.17 | 3.29 | 13.01 | 0. | .50 | 200 | 23.18 | 0.74 |
| Confluence flows west of Foundry | | | | | | | | 2.134 | 3.564 | 24.20 | | | | | | | | 35.56 | 8.89 | 33.09 | 0. | .50 | 250 | 42.03 | 0.86 |
| East side to Foundry | | | | | | | | | | | | | | | | | | | | | | | | | |
| Parcel S, R, V | 23 | 3 S, R, V | | TRUNK TO FOUNDRY | 18.77 | 60 | 1.126 | 1.126 | 3.766 | 13.50 | | | | | | 0.00 | 18.77 | 18.77 | 4.69 | 18.19 | 0. | .50 | 200 | 23.18 | 0.74 |
| Ex. Lots east of Foundry Street | 24 | 1 | | TRUNK TO FOUNDRY | 99.31 | 60 | 5.959 | 5.959 | 3.174 | 60.19 | | | | | | 0.00 | 99.31 | 99.31 | 24.83 | 85.02 | 0. | 50 | 375 | 123.91 | 1.12 |
| Parcel T | 25 | 5 T | | TRUNK TO FOUNDRY | 0.92 | 60 | 0.055 | 0.055 | 4.306 | 0.76 | ; | | | | | 0.00 | 0.92 | 0.92 | 0.23 | 0.99 | 0. | .50 | 200 | 23.18 | 0.74 |
| Conflulence flows east of Foundry | | | | | | | | 7.140 | 3.098 | 70.41 | | | | | | | | 119.00 | 29.75 | 100.16 | 0. | .50 | 375 | 123.91 | 1.12 |
| Confluence @ Gingerich Road Intersection | | | | | | | | 12.461 | 2.859 | 113.40 | 2.25 | 0.5625 | 0.91 0.8 | 645 | | 1.43 | | 210.84 | 52.71 | 167.54 | 0. | .50 | 450 | 201.50 | 1.27 |
| 5 - TO BADEN WWPS | | | | | | | | | | | | | | | | | | | | | | | | | |
| | | | | | | | | | | | | | | | | | | | | | | | | | |
| Ex. Industrial Lots on Gingerich Road | 26 | 3 | | | | | | | | | | | | 26.9 | 1 26.91 10.7640 | 10.76 | 26.91 | 26.91 | 6.73 | 17.49 | 0. | 50 | 200 | 23.18 | 0.74 |
| TOTAL APPROXIMATE FLOWS TO BADEN TRUNK ALT | 5 | | | | | | | 33.572 | 2.429 | 259.60 | 17.60 4 | .4000 | 0.91 0.86 | 45 | 46.64 18.6560 | 23.92 | | 656.28 | 164.07 | 447.59 | 0. | 30 | 750 | 609.46 | 1.38 |
| | | | | | | | | | | | | | | | | | | | | | | | | | |











lovember 10, 2022 – 12:12:13 AM – Plotted By: Adam





ELER DR OREEK TRIBUTARY 'A' BADEN CREEK TRIBUTARY 'A' BADEN CREEK TRIBUTARY 'A' ONNEOTON PONTO MAY AN AMAZED



ember 10, 2022 – 12:19:09 AM – Plotted By: Adam H

FILE PATH: P:\P\48730\100



SANITARY ALTERNATIVE 5







of







| 1+800 | 1- | +825 1· | +850 1- | +875 | 1+900 | 1+925 | 1+9 |
|---|---|--|---|----------------------------------|---------|----------------|-----------|
| 341.03 | | 341.73 | 341.00 | 340.69 | 340.26 | 339.83 | 339.59 |
| | Ψ GINGERICH PVI STA= 1+787,96 PVI STA= 1+787,96 PVI STA= 341.923 MH127- 1500mm# | SIA=1+824.43 T/G=341.75 | | | ISTING | - 1500mmø | .68 |
| | | | | | \ | (H128- | 7G=1+9 |
| | AN. | INSTALL SAFETY GRA PER OPSD.404.020. ELEV=339.00 | | | | | |
| | | | | SAN. | | | |
| | | | | | | | |
| | | | | | | | |
| | | | | | | | |
| ини и и и и и и и и и и и и и и и и и и | -750mmø @ 0.30% 221-7 221-172 200K) 221 200K) 200K) 200K) 200K 200K) 200K) 200K 200K) 200K 200K 200K 200K 200K 200K 200K 200K | SE 336.622 | 116.0r | n=750mmø SAN. @ 0.30% (TRUNK) | | NW336.274 | SE336.244 |
| | | TOWN Internet Street S | TOW | NSHIP OF V | WILMOT | GEODETIC BM | ELEV. = |
| | SNYDERS RD W R Har | | | | | SITE BENCHMARK | ELEV. = |
| | | energer | 8. 7. 6. 5. 4. 3. 2. 1. | | | | |
| | | KEY PLAN N.T.S. | No. REVI | SION | BY DATE | | |



